Doctoral theses at NTNU, 2022:306

Matteo Favero

Occupant-centric models for thermal comfort in buildings

Theoretical and experimental analysis of methods for enhancing user comfort in dynamic thermal indoor environments

NTNU

Thesis for the Degree of Faculty of Engineering Department of Civil and Environmental Engineering Norwegian University of Science and Technology Philosophiae Doctor



Norwegian University of Science and Technology

Matteo Favero

Occupant-centric models for thermal comfort in buildings

Theoretical and experimental analysis of methods for enhancing user comfort in dynamic thermal indoor environments

Thesis for the Degree of Philosophiae Doctor

Trondheim, October 2022

Norwegian University of Science and Technology Faculty of Engineering Department of Civil and Environmental Engineering



Norwegian University of Science and Technology

NTNU

Norwegian University of Science and Technology

Thesis for the Degree of Philosophiae Doctor

Faculty of Engineering Department of Civil and Environmental Engineering

© Matteo Favero

ISBN 978-82-326-6325-5 (printed ver.) ISBN 978-82-326-6909-7 (electronic ver.) ISSN 1503-8181 (printed ver.) ISSN 2703-8084 (online ver.)

Doctoral theses at NTNU, 2022:306

Printed by NTNU Grafisk senter

Preface

This doctoral thesis is submitted to the Norwegian University of Science and Technology (NTNU) as partial fulfilment of the requirements for the degree of Philosophiae Doctor. The work was funded by the Research Council of Norway (Norges Forskingsrådet) through the Research Centre on Zero Emission Neighbourhoods in Smart Cities (FME ZEN, Grant no. 257660). The research was carried out at the Department of Civil and Environmental Engineering (IBM) at NTNU under the supervision of Prof. Salvatore Carlucci (NTNU/The Cyprus Institute) and co-supervised by Dr Igor Sartori (SINTEF Community).

This work is the culmination of a journey that started in September 2018 after deciding to leave the industrial sector for a change. I decided to pursue and get a PhD. I found my PhD path full of little peaks that were not on my radar but were much more rewarding. I would be lying if I said I had not experienced despair, defeat and tiredness in the previous four years. However, I also felt joy, optimism, trust and brotherhood. This journey has never really been a lonely adventure, and I would like to thank everyone who supported me and helped me cross the finish line.

To begin with, I want to thank my supervisors, Salvatore Carlucci and Igor Sartori, for inspiring me to be ambitious, believing in me, and teaching me how to identify and develop good research.

Many thanks to all the researchers involved in the IEA EBC Annex 79 'Occupant-Centric Building Design and Operation' – especially Subtasks 1 & 2 – for the many fruitful and insightful discussions, which undoubtedly boosted my PhD.

I want to express my heartfelt gratitude to all the people around me for their invaluable ongoing support, both personally and professionally and for making the PhD journey a most enjoyable time in my life. Particularly to Siri, Camilla, Shabnam, Therese, Amin, Atle, David, Ali, Carine, Anna, Artur, Mattia, Rosita, Cristobal, Ramya, Oskar, and Tobias. Thanks for the lunch breaks, conversations at the coffee machine, and social

events. You made it fun to come to work. I would also like to thank the fantastic IBM administration team for their kind support.

Finally, I am forever thankful to my family, my parents, and my sister Jessica, for their unconditional support, understanding and love.

Trondheim, June 2022 Matteo Favero

Abstract

In the last decades, increasing global energy demand, a foreseen reduction of available fossil fuels and increasing evidence for global warming have shown the urgency to rethink the built environment and promote energy transition. Indeed, in most industrialised nations, the building sector accounts for about 40% of the total energy consumption (space heating and cooling, domestic hot water, ventilation, lighting and appliance use). A significant share of this energy is used to thermal control buildings and provide thermally comfortable indoor environments. However, technical building systems are typically designed and operated considering fixed set-point temperatures based on the 'one-size-fits-all' principle – which has been questioned in the last fifty years - assuming universal thermal comfort requirements. Furthermore, the indoor environment frequently changes abruptly across buildings or between various parts within a single building. For instance, manually operating thermostats, windows and solar shades can result in considerable and not systematic changes in the indoor environment. Also, automatic controllers exhibit, to a lesser degree, a similar behaviour. Moreover, individual activity modifies the basal metabolic rate over time, and the addition or removal of clothes affects the heat balance of the human body as well. In other words, the steady-state temperature settings are the exception rather than the rule. Building temperature ranges should instead be based on real-time empirical evidence regarding the needs of its occupants, which is obtained through their feedback (usually on a rating scale). This thesis investigates these topics and relies upon an experimental study to explore the human reaction to dynamic thermal environments.

The general approach utilised in this thesis encompasses a technical/methodological aspect, namely a newer controlled experimental procedure and a robust and replicable methodology for human feedback acquisition, and a statistical aspect, namely an original statistics-enabled occupant-centric modelling. The technical/methodological aspect refers to how thermal comfort data are collected; that

Abstract

is, the specific approach and experimental set-up utilised. The statistical aspect refers to how thermal comfort data are analysed, namely, the specific statistical technique to be adopted and the needed modelling steps.

It was found that the human reaction to dynamic thermal stimuli is asymmetric with respect to heating and cooling processes, and two distinct mechanisms cause discomfort due to overheating and overcooling. Compared to the recommendations regarding temperature cycles, drifts and ramps included in the ASHRAE Standard 55, this result showed that current recommendations underestimate the risk of thermal discomfort during a cooling process while overestimating it during a heating one. Concerning the subjective thermal comfort data analysis, the choice of the statistical method affects the conclusions. While it may seem a trivial consideration, till now, it is common in the thermal comfort field to find studies that use, for example, linear regression on ordinal data following an old approximation used to overcome the lack of statistical tools and computational power (which are not anymore limiting aspects in statistical analysis). Particularly, we showed that applying a linear regression model to ordinal data suggested that there is no difference in means and effect size between genders (female/male). In contrast, an ordinal regression model leaded to the opposite conclusion. This is considered one of the reasons why there is no consensus in the scientific literature on whether gender is an influential factor when assessing the perception of the thermal environment. This result points out that greater attention should be paid to the choice of the statistical method used to analyse subjective data, which should consider the level of measurement used during the data gathering. Furthermore, two different procedures were proposed to facilitate the integration of the occupants and their actual needs into the design and operation of buildings: the former is suitable for a better-informed design phase, where the target is the optimal thermal comfort conditions expressed for an 'average' occupant; the other is appropriate for including the human-in-the-control-loop of a building, where satisfying the needs of a specific occupant is the primary goal.

Through this thesis work, new knowledge concerning the human reactions to a dynamic thermal environment was created, which can improve the understanding of the extent to which the indoor environmental conditions can vary both naturally and artificially. Designing and implementing thermally comfortable set-point modulations

that consider the occupant feedback capabilities would be beneficial to increase perceived thermal comfort and productivity, potentially reduce energy consumption and significantly support the clean energy transition. In addition, several recommendations for future research are presented.

Sammendrag

I løpet av de siste tiårene har økende global energietterspørsel, en forventet reduksjon av tilgjengelig fossilt brensel og økende bevis for global oppvarming, vist at det haster med å tenke nytt om det bygde miljøet og fremme energiomstilling. I de fleste industrialiserte land står byggesektoren for omtrent 40 % av det totale energiforbruket (oppvarming og kjøling av rom, varmtvann til husholdningsbruk, ventilasjon, belysning og bruk av apparater). En betydelig andel av denne energien brukes til å varmekontrollere bygninger og gi termisk komfortable innendørsmiljøer. Imidlertid er tekniske bygningssystemer vanligvis designet og drevet med tanke på faste settpunkttemperaturer som baserer seg på "one-size-fits-all"-prinsippet - hvilket det de siste femti årene har blitt stilt spørsmål ved- som stiller universelle krav til termisk komfort. Videre endrer innemiljøet seg ofte brått på tvers av bygninger eller mellom ulike deler i bygningen. For eksempel kan manuelt betjente termostater, vinduer og solskjermer resultere i betydelige og ikke-systematiske endringer i innemiljøet. Automatiske kontroller viser i mindre grad en lignende oppførsel. Kroppslig aktivitet endrer dessuten basalstoffskiftet over tid, og tilføyelse eller fjerning av klær påvirker varmebalansen i menneskekroppen. Med andre ord er steady-state temperaturinnstillingene unntaket snarere enn regelen. Temperaturområdet i et bygg bør i stedet være basert på sanntids empiriske bevis angående behovene til brukerne, som er innhentet gjennom deres tilbakemeldinger (vanligvis på en vurderingsskala). Denne oppgaven undersøker disse temaene og baserer seg på en eksperimentell studie for å utforske menneskets reaksjon på dynamiske termiske miljøer.

Den generelle tilnærmingen som benyttes i denne oppgaven omfatter et teknisk/metodologisk aspekt, nemlig en nyere kontrollert eksperimentell prosedyre og en robust samt replikerbar metodikk for innhenting av menneskelig tilbakemelding, og et statistisk aspekt, nemlig en original statistikkaktivert occupant-sentrisk modellering. Det tekniske/metodiske aspektet refererer til hvordan termisk komfordata samles inn;

Sammendrag

det vil si den spesifikke tilnærmingen og eksperimentelle oppsettet som benyttes. Det statistiske aspektet refererer til hvordan termisk komfordata analyseres, nemlig den spesifikke statistiske teknikken som skal tas i bruk og de nødvendige modelleringstrinnene.

Det ble funnet at den menneskelige reaksjonen på dynamiske termiske stimuli er asymmetrisk i forhold til oppvarmings- og kjølingsprosesser, og to distinkte mekanismer ble funnet til å forårsake ubehag på grunn av overoppheting og overkjøling. Sammenlignet med anbefalte temperatursykluser, drifter og ramper inkludert i ASHRAE Standard 55, viste dette resultatet at gjeldende anbefalinger undervurderer risikoen for termisk ubehag ved en kjøleprosess mens den overvurderes ved en oppvarmingsprosess. Når det gjelder den subjektive termiske komfordataanalysen, påvirkes konklusjonen av valgt statistisk metode. Selv om det kan virke som en triviell vurdering, er det frem til nå vanlig innen termisk komfort å finne studier som bruker for eksempel lineær regresjon på ordinære data etter en eldre tilnærming brukt for å kompensere for mangelen på statistiske verktøy og beregningskraft (som ikke lenger er begrensende aspekter i statistisk analyse). Spesielt viste vi at bruk av en lineær regresjonsmodell på ordinaldata antydet at det ikke er forskjell i gjennomsnitt og effektstørrelse mellom kjønn (kvinne/mann). Derimot førte en ordinær regresjonsmodell til motsatt konklusjon. Dette anses som en av grunnene til at det ikke er konsensus i vitenskapelig litteratur om hvorvidt kjønn er en påvirkningsfaktor når man vurderer oppfatningen av det termiske miljøet. Dette resultatet peker på at det bør vies større oppmerksomhet til valget av den statistiske metoden som benyttes ved analyse av subjektive data, som bør ta hensyn til målenivået som brukes under datainnsamlingen. Videre ble det foreslått to forskjellige prosedyrer for å lette integreringen av brukerne og deres faktiske behov i design og drift av bygninger: førstnevnte er egnet for en bedre informert designfase, hvor målet er de optimale termiske komforforholdene uttrykt for en "gjennomsnittlig" bruker; sistnevnte er hensiktsmessig for å inkludere mennesket-ikontroll-sløyfen i en bygning, der det å tilfredsstille behovene til en spesifikk bruker er hovedmålet.

Gjennom denne avhandlingen har det blitt skapt ny kunnskap om menneskets reaksjoner på et dynamisk termisk miljø, som kan forbedre forståelsen av hvilken grad innemiljøforholdene kan variere både naturlig og kunstig. Å designe og implementere termisk komfortable settpunktmodulasjoner som tar hensyn til tilbakemeldingsevnen for beboere vil være fordelaktig for å øke opplevd termisk komfort, og produktivitet, samt potensielt redusere energiforbruket og betydelig støtte overgangen til ren energi. I tillegg presenteres flere anbefalinger for fremtidig forskning.

Sommario

Negli ultimi decenni, l'aumento dei consumi energetici mondiali, la minore disponibilità di combustibili fossili e la crescente evidenza del riscaldamento globale hanno manifestato la necessità di ripensare l'ambiente edificato e promuovere la transizione energetica. Infatti, nella maggior parte dei paesi industrializzati, il settore edile rappresenta circa il 40% del consumo totale di energia (riscaldamento e raffrescamento degli ambienti, acqua calda sanitaria, ventilazione, illuminazione e utilizzo di elettrodomestici). Una quota significativa di questa energia viene utilizzata per il controllo termico degli edifici e per fornire ambienti interni termicamente confortevoli. Tuttavia, i sistemi tecnici per l'edilizia sono generalmente progettati e gestiti considerando temperature di setpoint fisse basate sul principio 'one-size-fits-all' - che è stato messo in discussione negli ultimi cinquant'anni - assumendo requisiti di comfort termico universali. Inoltre, le condizioni ambientali interne possono cambiare bruscamente da un edificio all'altro o tra le varie zone di uno stesso edificio. Ad esempio, il funzionamento manuale di termostati, finestre e tende solari può comportare cambiamenti considerevoli e non sistematici nell'ambiente interno. Inoltre, i controllori automatici mostrano, in misura minore, un comportamento simile. In aggiunta, il metabolismo basale cambia a seconda dell'attività fisica della singola persona, ed indossare o togliere gli indumenti influisce sull'equilibrio termico del corpo umano. In altre parole, considerare la temperatura in regime stazionario è l'eccezione piuttosto che la regola. Gli intervalli di temperatura degli edifici dovrebbero invece essere basati su prove empiriche in tempo reale relative ai bisogni dei suoi occupanti, ottenute attraverso i loro feedback (di solito su una scala di valutazione). La presente tesi analizza questi aspetti proponendo uno studio sperimentale per esplorare la reazione umana agli ambienti termici dinamici.

L'approccio generale utilizzato in questa tesi comprende un aspetto tecnico/metodologico, ovvero una procedura sperimentale innovativa e una

Sommario

metodologia robusta e replicabile per l'acquisizione del feedback da parte degli utenti, e un aspetto statistico, ovvero una modellazione statistica originale incentrata sull'occupante. L'aspetto tecnico/metodologico si riferisce alle modalità di raccolta dei dati di comfort termico; cioè lo specifico approccio e l'impostazione sperimentale utilizzata. L'aspetto statistico si riferisce al modo in cui i dati sul comfort termico vengono analizzati, ovvero la specifica tecnica statistica da adottare e le fasi di modellazione necessarie.

È stato riscontrato che la reazione umana agli stimoli termici dinamici è asimmetrica rispetto ai processi di riscaldamento e raffreddamento, e che due meccanismi distinti causano disagio da surriscaldamento e sovraraffreddamento. Rispetto alle raccomandazioni relative a "temperature cycles, drifts and ramps" incluse nello Standard ASHRAE 55, questo risultato ha mostrato che le attuali raccomandazioni sottovalutano il rischio di disagio termico durante il processo di raffreddamento mentre lo sovrastimano durante quello di riscaldamento. Per quanto riguarda l'analisi soggettiva dei dati di comfort termico, la scelta del metodo statistico influisce sulle conclusioni. Anche se può sembrare una considerazione banale, nel campo del comfort termico è comune trovare studi che utilizzino, ad esempio, la regressione lineare su dati ordinali basandosi su una vecchia approssimazione, utilizzata per sopperire alla mancanza di strumenti statistici e potenza computazionale (che non sono più aspetti limitanti nell'analisi statistica). In particolare, abbiamo mostrato che l'applicazione di un modello di regressione lineare su dati ordinali suggerisce che non vi è alcuna differenza nella media e nell'ampiezza dell'effetto tra i sessi (femmina/maschio). Al contrario, un modello di regressione ordinale ha portato alla conclusione opposta. Questo è considerato uno dei motivi per cui non c'è consenso nella letteratura scientifica sul fatto che il sesso sia un fattore influente nella valutazione della percezione dell'ambiente termico. Tale risultato evidenzia la necessità di porre una maggiore attenzione nella scelta del metodo statistico utilizzato per l'analisi dei dati soggettivi, il quale dovrebbe considerare la scala di misura utilizzata durante la raccolta dei dati. Inoltre, sono state proposte due diverse procedure per facilitare l'integrazione degli occupanti e le loro effettive esigenze nella progettazione e nel funzionamento degli edifici: la prima è utile per informare una fase progettuale più consapevole, dove l'obiettivo sono le condizioni ottimali di comfort termico espresse per un occupante 'medio'; l'altra procedura è adatta per includere

l'essere umano nel circuito di controllo dell' edificio, dove soddisfare i bisogni di uno specifico occupante è l'obiettivo primario.

Attraverso questo lavoro di tesi è stata creata nuova conoscenza riguardante la reazione umana ad un ambiente termico dinamico, la quale può migliorare la comprensione dell'entità con cui le condizioni ambientali interne possono variare sia naturalmente che artificialmente. Progettare e implementare modulazioni termicamente confortevoli del setpoint, che tengano conto delle capacità di feedback degli occupanti, sarebbe utile per aumentare il comfort termico e la produttività, ridurre potenzialmente il consumo di energia e supportare in modo significativo la transizione verso l'utilizzo sostenibile di fonti energetiche rinnovabili. Infine, vengono presentate alcune raccomandazioni per la ricerca futura.

Table of Contents

Preface	
Abstract (English/Norsk/Italiano)	v
List of Tables	xix
List of Figures	xxi
Glossary	xxiii
List of Papers	xxv
Introduction	1
Background	5
1.1 Dynamic and non-uniform thermal environments	7
1.2 Human thermal perception and thermoregulation: a brief overview	9
1.2.1 Thermophysiological models	11
1.3 Individual differences	12
1.3.1 Personal comfort paradigm	13
1.4 Rating scales for subjective assessment of thermal environments	14
1.5 Occupant behaviour	16
Knowledge gap and research questions	17
2.1 Research questions	18
Methods	21
3.1 Literature review and theoretical framework	21
3.2 Experimental aspect	21
3.2.1 Experimental set-up	22
3.2.2 Experimental data analysis	26
3.3 Modelling and statistics aspect	27
3.3.1 Statistical modelling	27
3.4 Applicational aspect	29

3.4.1 Statistical modelling2	9
Results	3
4.1 Human thermal comfort under dynamic environmental conditions	3
4.1.1 General observations	3
4.1.2 KM survival curves3	7
4.1.3 Cox-regression	9
4.2 Analysis of subjective thermal comfort data4	2
4.2.1 Unconditional model4	3
4.2.2 Fitting a categorical variable4	4
4.2.3 Fitting a linear predictor4	7
4.2.4 Structured thresholds4	9
4.3 Human-in-the-loop methods for occupant-centric building design and operation	า 51
4.3.1 Ordinal model5	1
4.3.2 Beta model5	3
4.3.3 Models' comparison5	6
Discussion	3
5.1 Limitations	6
5.1.1 Regarding the experimental analysis6	6
5.1.2 Regarding the data analysis6	7
Conclusions7	1
6.1 Concluding remarks7	1
6.1.1 Further conclusions7	3
6.2 Suggested directions for future research7	5
References	7
Appendix A: Main publications8	3
Appendix B: Supplementary publications15	9

List of Tables

Table 1 – Limits on temperature drifts and ramps by ASHRAE 55:2020 [7]	8
Table 2 – List of covariates used in the model for both space heating and cooling processes	39
Table 3 – Regression coefficients for the model with only a categorical variable (allowing the standard deviation to vary by group).	45
Table 4 – Regression coefficients for the model with a categorical and continuous variable (allowing the standard deviation to vary by group).	48
Table 5 – Values of the Leave-One-Out Information Criterion (LOOIC) and their difference for the cumulative probit model with structured and unstructured thresholds.	50
Table 6 – Predictors' relative importance for both the beta and ordinal models	58

List of Figures

Fig. 1 – Predicted percentage dissatisfied (PPD) as a function of predicted mean vote (PMV). From ASHRAE 55:2020 [7]
Fig. 2 – Pathways for autonomous thermoregulatory responses and thermal perception from peripheral tissues. Adapted from Ref. [43]10
Fig. 3 – Comparison of the model segmentation. Adapted from Ref. [55]
Fig. 4 – Outline of research with main research question, objectives, and research activities
Fig. 5 – Journal publications in this thesis20
Fig. 6 – Floor plan of the facility22
Fig. 7 – Extract from the rating scale used to assess the perception of the room temperature; (a) rating scale displayed by the participant and (b) rating scale displayed for the data analysis
Fig. 8 – Digital interface for the discomfort button25
Fig. 9 – Schematic of the three-level hierarchical study: repeated measures within experimental conditions cross-classified by participant and day
Fig. 10 – Thermal ramps endpoint
Fig. 11 – Rating scales for thermal discomfort events
Fig. 12 – KM survival curves for different rates of temperature change
Fig. 13 – Log-log survival chart for heating and cooling based on the rate of temperature changes
Fig. 14 – Penalised spline fit of (a) BMI, (b) initial operative temperature and (c) operative temperature change for heating41
Fig. 15 – Penalised spline fit of operative temperature change for cooling42
Fig. 16 – Posterior prediction for (a) the thresholds-only and (b) intercept-only model.
Fig. 17 – Posterior distributions for the model that include the variable Gender: (a) cumulative probit and (b) gaussian (ordinal-as-metric) model
Fig. 18 – (a) Standardised and (b) 'original' regression coefficient for air temperature for the cumulative probit (green) and gaussian (ordinal-as-metric) (orange)

List of Figures

Fig. 19 – Spacing for (a) structured and (b) unstructured thresholds and (c) their difference.	50
Fig. 20 – Predicted probabilities of a thermal preference vote using the (a) cluster- specific and (b) population-averaged procedures	52
Fig. 21 – Predicted probabilities of a thermal preference vote using the (a) cluster- specific and (b) population-averaged procedures for three different operative temperatures.	53
Fig. 22 – Predicted responses using the cluster-specific (black line) and population- averaged (red line) procedures.	54
Fig. 23 – (a) Probability densities and (b) categorised probabilities of the predicted response using the cluster-specific procedure for three different operative temperatures.	55
Fig. 24 – (a) Probability densities and (b) categorised probabilities of the predicted response using the population-averaged procedure for three different operative temperatures.	56
Fig. 25 – Predicted responses using the cluster-specific (black solid and dashed line and population-averaged (red solid and dashed lines) procedures for the beta and ordinal models, respectively.	es) 60

Glossary

Beta mixed-effects model — A mixed-effect model which assume a beta distribution for the conditional distribution of the response variable.

Central tendency — A measure of a central or typical value for a probability distribution. Mean, median, and mode are examples of possible measures of central tendency.

Cluster-specific procedure — An approach used to handle the group-level residual (i.e., group-level random effect) during the prediction. The probabilities calculated with this approach have a cluster-specific interpretation (i.e., describe only the unit belonging to the specific cluster).

Cumulative probit model — A cumulative link model (belonging to the broad class of ordinal regression models) in which the link function is the probit.

Dispersion — A measure of variability of a distribution, usually around its central tendency. Variance, standard deviation, and interquartile range are example of possible measures of dispersion.

Logit — Sometimes called log-odds, it is the inverse of the cumulative distribution function of the standard logistic distribution. It is a type of function that maps probability values from (0,1) to real numbers in $(-\infty, +\infty)$.

Mixed-effects model — A synonym of multilevel model. See multilevel model.

Multilevel model — A statistical technique suitable to model the relationship between the dependent(s) and independent(s) variables when there is a correlation between observations (e.g., whenever the data are clustered and/or nested) [1].

Ordinal mixed-effect model — A mixed-effect model which assume a multinomial distribution for the conditional distribution of the response variable.

Population-averaged procedure — A simulation-based approach used to handle the group-level residual (i.e., group-level random effect) during the prediction. The probabilities calculated with this approach have a population-averaged interpretation (i.e., averaged across the random effects).

Probability density function — Provides the probability distribution for continuous variables. It allows to determine the probability of an observation being within a set range around a target value [2].

Probability mass function — Provides the probability distribution for discrete variables. It allows to determine the probability of an observation being exactly equal to a target value [2].

Probit — The inverse of the cumulative distribution function of the standard normal distribution. It is a type of function that maps probability values from (0,1) to real numbers in $(-\infty, +\infty)$.

Random effect — A term that refers to the randomness in the probability model for the group-level coefficients of a multilevel model [1].

List of Papers

Main publications

The following journal articles constitutes the main publications of this thesis:

- (I) M. Favero, I. Sartori, S. Carlucci, Human thermal comfort under dynamic conditions: An experimental study, Building and Environment 204 (2021) 108144. doi:10.1016/j.buildenv.2021.108144. Contribution of the PhD candidate: Conceptualisation, Data curation, Formal analysis, Methodology, Software, Validation, Visualisation, Writing original draft, and Writing review & editing.
- (II) M. Favero, A. Luparelli, S. Carlucci, Analysis of subjective thermal comfort data: a statistical point of view [Manuscript submitted for publication] (2022). Contribution of the PhD candidate: Conceptualisation, Formal analysis, Methodology, Software, Visualisation, Writing – original draft, and Writing – review & editing.
- (III) M. Favero, J.K. Møller, D. Cali, S. Carlucci, Human-in-the-loop methods for occupant-centric building design and operation, Applied Energy 325 (2022) 119803. doi:10.1016/j.apenergy.2022.119803. *Contribution of the PhD candidate*: Conceptualisation, Data curation, Formal analysis, Methodology, Software, Validation, Visualisation, Writing – original draft, and Writing – review & editing.

Supplementary publications

The following journal articles constitutes the supplementary publications of this thesis:

(a) M. Schweiker, E. Ampatzi, M.S. Andargie, R.K. Andersen, E. Azar, V.M. Barthelmes, C. Berger, L. Bourikas, S. Carlucci, G. Chinazzo, L.P. Edappilly, M. Favero, S. Gauthier, A. Jamrozik, M. Kane, A. Mahdavi, C. Piselli, A.L. Pisello, A. Roetzel, A. Rysanek, K. Sharma, S. Zhang, Review of multi-domain approaches to indoor environmental perception and behaviour, Building and Environment 176 (2020). doi:10.1016/j.buildenv.2020.106804.

Contribution of the PhD candidate: Data curation, Formal analysis, Writing – original draft, and Writing – review & editing.

(b) A.L. Pisello, I. Pigliautile, M. Andargie, C. Berger, P.M. Bluyssen, S. Carlucci, G. Chinazzo, Z. Deme Belafi, B. Dong, **M. Favero**, A. Ghahramani, G. Havenith, A. Heydarian, D. Kastner, M. Kong, D. Licina, Y. Liu, A. Luna-Navarro, A. Mahdavi, A. Nocente, M. Schweiker, M. Touchie, M. Vellei, F. Vittori, A. Wagner, A. Wang, S. Wei, Test rooms to study human comfort in buildings: A review of controlled

experiments and facilities, Renewable & sustainable energy reviews 149 (2021) 111359. doi:10.1016/j.rser.2021.111359.

Contribution of the PhD candidate: Formal analysis, Investigation, and Writing original draft.

(c) G. Chinazzo, R.K. Andersen, E. Azar, V.M. Barthelmes, C. Becchio, L. Belussi, C. Berger, S. Carlucci, S.P. Corgnati, S. Crosby, L. Danza, L. de Castro, M. Favero, S. Gauthier, R.T. Hellwig, Q. Jin, J. Kim, M.S. Khanie, D. Khovalyg, C. Lingua, A. Luna-Navarro, A. Mahdavi, C. Miller, I. Mino-Rodriguez, I. Pigliautile, A.L. Pisello, R.F. Rupp, A. Sadick, F. Salamone, M. Schweiker, M. Syndicus, G. Spigliantini, N.G. Vasquez, D. Vakalis, M. Vellei, S. Wei, Quality criteria for multi-domain studies in the indoor environment: critical review towards research guidelines and recommendations [Manuscript submitted for publication] (2022).

Contribution of the PhD candidate: Conceptualisation, Formal analysis, Visualisation, and Writing - original draft.

S. Carlucci, M. De Simone, S.K. Firth, M.B. Kjærgaard, R. Markovic, M.S. Rahaman, (d) M.K. Annageeb, S. Biandrate, A. Das, J.W. Dziedzic, G. Fajilla, M. Favero, M. Ferrando, J. Hahn, M. Han, Y. Peng, F. Salim, A. Schlüter, C. van Treeck, Modeling occupant behavior in buildings, Building and environment 174 (2020) 106768. doi:10.1016/j.buildenv.2020.106768.

Contribution of the PhD candidate: Data curation, Investigation, Visualisation, Writing – original draft, and Writing – review & editing.

(e) M. Vellei, E. Azar, K. Bandurski, C. Berger, S. Carlucci, B. Dong, M. Favero, A. Mahdavi, M. Schweiker, Documenting occupant models for building performance simulation: A state-of-the-art, Journal of building performance simulation 15 (2022) 634-655. doi:10.1080/19401493.2022.2061050.

Contribution of the PhD candidate: Conceptualisation, Data curation, Formal analysis, Writing - original draft, and Writing - review & editing.

(f) B. Dong, R. Markovic, S. Carlucci, Y. Liu, A. Wagner, A. Liguori, C. van Treeck, D. Olevnikov, E. Azar, G. Faiilla, J. Drooňa, J. Kim, M. Vellei, M. De Simone, M. Shamsaiee, M. Bavaresco, M. Favero, M. Kjaergaard, M. Osman, M. Frahm, S. Dabirian, D. Yan, X. Kang, A guideline to document occupant behavior models for advanced building controls, Building and Environment 219 (2022) 109195. doi:10.1016/j.buildenv.2022.109195.

Contribution of the PhD candidate: Conceptualisation, Investigation, Methodology, and Writing – original draft.

Additional publication

The following article was published during the doctoral research but is not included in the thesis:

A. Mahdavi, C. Berger, H. Amin, E. Ampatzi, R.K. Andersen, E. Azar, (1) V.M.Barthelmes, M. Favero, J. Hahn, D. Khovalyg, H.N. Knudsen, A. Luna-Navarro, A. Roetzel, F.C. Sangogboye, M. Schweiker, M. Taheri, D. Teli, M. Touchie, S. Verbruggen, The Role of Occupants in Buildings' Energy Performance Gap: Myth Reality?, Sustainability (Basel, Switzerland) 13(6) (2021) or 3146. doi:10.3390/su13063146.

Introduction

In the last decades, increasing global energy demand, a foreseen reduction of available fossil fuels and increasing evidence for global warming have sparked a surge in promoting the energy transition. In most industrialised nations, buildings energy consumption account for a considerable amount (about 40% [3]) of total energy consumption and is used for space heating, cooling, ventilation, and lighting of rooms, as well as the generation of domestic hot water and electric appliances used by inhabitants. Typically, space heating and cooling are the most energy-intensive processes. For instance, in 2018, they accounted for almost 40% of all energy consumed by buildings worldwide [4]. Rethinking how buildings are designed and operated is necessary to reduce their energy consumption. However, the design and operation of buildings have significant implications for people's comfort, well-being, and health, especially considering that humans spend approximately 90% of their time indoors [5]. Consequently, measures to reduce energy consumption in buildings should be centred on the occupants.

The human sensory system is exposed to a variety of environmental stimuli. In buildings, four ecological factors are identified as the main aspects characterising the indoor environment, namely thermal, visual, auditory, and air quality stimuli. Human senses allow people to perceive these stimuli and subsequently evaluate and respond to them, often through behavioural actions. Although not all interactions between occupants and the built environment are motivated by a reaction to an unpleasant stimulus (i.e., discomfort), perception and behaviour have a strong correlation [6]. Nevertheless, there are few studies linking perception and action. Furthermore, the scientific literature in this field frequently treats these various stimuli separately.

The thermal domain is, to date, the most studied in climatic chamber experiments among the four environmental factors typically considered in buildings. However, most focus is on stationary thermal environments, while the experiments on dynamic

Introduction

conditions mainly involve step-changes. Unsurprisingly, technical building systems are typically designed and operated considering fixed set-point temperatures based on the 'one-size-fits-all' principle assuming universal thermal comfort requirements. On the one hand, maintaining a tight temperature range demands more energy than allowing a wider operative temperature shift. On the other, the understanding of the human response to changing thermal environments is still limited. So, to what extent the indoor thermal conditions of buildings can be modulated without sacrificing the thermal comfort perceived by their occupants remains an unresolved research subject.

Therefore, it is evident that indoor environmental monitoring and control strategies play an essential role in the design and operation of a building (both for energy consumption and the thermal comfort of the occupants). Tailoring services such as Heating, Ventilation, and Air Conditioning (HVAC), lighting, and electrical power have the potential to save a significant amount of required energy. In addition to environmental awareness, occupant-aware control schemes have been shown to save between 10–42%, depending on factors such as outdoor climate and control strategy [7,8]. A more detailed view of the building environment and its occupants opens the door to more energy efficient building services tailored to specific purposes and target groups.

Aims and structure of the thesis

The general aim of this thesis is to investigate how indoor thermal conditions may be modulated while guaranteeing satisfactory thermal comfort conditions. The general approach utilised in this thesis encompasses both a technical/methodological and a statistical point of view. The technical/methodological aspect refers to how thermal comfort data are collected; that is, the specific approach and experimental set-up utilised. On the other, the statistical aspect refers to how thermal comfort data are analysed, namely, the specific statistical technique to be adopted and the needed modelling steps. Both of these aspects are essential and interrelated. They compose the research design – the overall strategy for data collection and analysis – which, in turn, is based on the specific question driving the research.

The introduction provides a brief preface to the context and the aims of the thesis. Chapter 1 offers a more detailed theoretical background for the research. It summarises historical research on 'static' thermal comfort, its integration into standards and the most recent development of 'dynamic' thermal comfort. An overview of human thermal perception and thermoregulation is also provided with a hint of thermophysiological models. It also describes the individual differences in thermal comfort and briefly introduces the new paradigm of 'personalised thermal comfort'. In addition, the subjective assessment of the thermal environment and the behaviour of occupants in buildings are discussed. The knowledge gap identified from the theoretical background, as well as the primary research topics addressed in this thesis, are given in Chapter 2. Chapter 3 presents the methodology undertaken in this thesis. The results are presented in Chapter 4 and discussed in Chapter 5. Finally, the conclusions and future developments are presented in Chapter 6.

Chapter 1 Background

Thermal comfort models for the human body have been available for over 40 years as a result of considerable efforts in the 1960s and earlier to develop such models for military and aerospace applications. Currently, the most adopted international thermal comfort standards, ASHRAE 55:2020 [9], ISO 7730:2005 [10] and EN 16798-1:2019 [11] (formerly EN 15251:2007 [12]), propose requirements based on Fanger model (beyond also including other approaches), which solves the heat balance between the human body and its surroundings represented as a uniform environment. Fanger introduced the 'Predicted Mean Vote' (PMV) as the index that predicts the mean thermal sensation vote on a standard scale for a large group of persons exposed to a given combination of activity level, clothing insulation and four thermal environmental variables (dry-bulb air temperature, mean radiant temperature, air velocity and relative humidity) [13]. The PMV-model is generally referred to as a static model because it is only suited to predict thermal sensation under steady-state or slowly changing indoor conditions (i.e., rate of change lower than 2.0K/h) [10]. Based on the PMV, the 'Predicted Percentage of Dissatisfied' (PPD) was determined:

$$PPD = 100 - 95 \cdot e^{(-0.03353 \cdot PMV^4 - 0.2179 \cdot PMV^2)}$$
Eq. (1)



Fig. 1 – Predicted percentage dissatisfied (PPD) as a function of predicted mean vote (PMV). From ASHRAE 55:2020 [9].

In Fig. 1 can be seen that the relationship between PMV and PPD has a minimum of 5% at PMV = 0 – generally referred to in the literature as thermal neutrality. Furthermore, it is also symmetric on the cooler (i.e., for negative PMV values) and the warmer (i.e., for positive PMV values) sides. In this regard, the issue that arises is whether the outcome of Fanger's analysis (specifically, minimum of dissatisfied of 5% and symmetry of the curve centred at PMV = 0) are correct. Different studies have found other relationships than the one described by Fanger [14-17].

In the 1970s, the same period when Fanger developed his model, Nicol and Humphreys [18] hypothesised the presence of 'control mechanisms' (feedback loops) between the occupants' thermal comfort perception and their behaviour in buildings. They suggested that the building design may concentrate on how to allow for the 'control mechanisms' to operate rather than trying to establish optimum indoor climates. In the first meta-analysis of thermal comfort field studies worldwide, Humphreys [19] compiled data from more than 30 field studies of thermal comfort and shows, with compelling evidence, the disparity between physiological model predictions and empirical field study. In a subsequent reanalysis of the data, Humphreys [20] evaluated the influence of climate and found that the indoor temperature required for thermal comfort is related to the prevailing outdoor mean temperature. Later on, the analysis of the two studies was brought together [21]. It can be said that these papers are the ancestors of what is nowadays known as the adaptive thermal comfort model since they have shaped its development (e.g., see [22-25]). The hypothesis of adaptive thermal

comfort predicts that contextual factors and past thermal history modify occupant's thermal expectations and preferences [26]. People in warm climate zones would prefer higher indoor temperatures than people living in cold climate zones, which contrasts with the assumptions underlying comfort standards based on the PMV model [26]. Adaptation is defined as the gradual decrease of the human response to repeated environmental stimulation, and can be both behavioural (e.g., clothing adjustment), physiological (e.g., acclimatisation) as well as psychological (e.g., expectation) [24,26,27]. Therefore, the adaptive model recognises the adaptation of people; however, it focuses on long-term adaptation mechanisms, ignoring short-term ones.

In current standards, Fanger's PMV/PPD model is the prerogative of mechanically heated and/or cooled buildings, while the adaptive thermal comfort model is reserved for free-running buildings. Both are steady-state and whole-body thermal comfort models that predict the mean thermal sensations for a group of people. Therefore, they can account neither for dynamic and non-uniform thermal environments nor for individual differences.

1.1 Dynamic and non-uniform thermal environments

Manually operating thermostats, ventilation fans, windows, and window shades can result in considerable and not systematic changes in the indoor environment. Automatic controllers exhibit, to a lesser degree, a similar behaviour. The indoor environment frequently changes abruptly across buildings or between various parts within a single building. Individual activity modifies the basal metabolic rate over time, and the addition or removal of clothes affects the heat balance as well. In other words, the steady-state temperature settings that define most comfort studies are the exception rather than the rule. In 1981, Rohles [28] stated that 'human response to the thermal environment depends on seven variables' (air temperature, relative humidity, mean radiant temperature, air velocity, clothing, physical activity, and time). He also mentioned that 'even though the thermal conditions to which humans are exposed are never constant for long periods of time, time has received only modest attention as a variable in comfort research' [28]. These days, 'time' continues to be the least understood variable in the field of thermal comfort research.

Nowadays, all thermal comfort standards include definitions of the requirements for indoor thermal conditions in buildings both for design and operational assessment. However, current standards only indicate the maximum variations in operative temperature for non-steady-state thermal environments. ASHRAE 55:2020 [9] and ISO 7730:2005 [10] classify temperature variations as either temperature drifts and ramps or temperature cycles. Drifts and ramps are defined as 'monotonic, non-cyclic changes in operative temperature' [9], and their limits during a period are shown in Table 1. Drifts refer to passive temperature changes in an enclosed space, while ramps denote actively controlled ones. In contrast, cycles refer to 'those situations where the operative temperature repeatedly rises and falls, and the period of these variations is not greater than 15 min' [9]. For these changes, ASHRAE 55 allows a maximum peak-to-peak cyclic variation in operative temperature of 1.1 K and recommends treating cyclic variations with a period greater than 15 min as drifts or ramps.

Time period (h)	Maximum operative temperature to change allowed (K)
0.25	1.1
0.5	1.7
1	2.2
2	2.8
4	3.3

Table 1 – Limits on temperature drifts and ramps by ASHRAE 55:2020 [9].

ISO 7730:2005 [10] provides less detailed indications. For temperature cycles, it sets a maximum peak-to-peak variation of 1 K, whereas, for drifts and ramps with a rate of change lower than 2.0 K/h, it prescribes steady-state methods. These standards also include step-changes, which involve changing the environment (i.e., moving to/from another space) rather than a change within the environment. Consequently, they are not described here because they are out of the scope of this thesis.

The limiting criteria in Table 1 are probably based on early laboratory studies of thermal comfort under transient exposure [29-31]. During the same period (the 1970s and 1980s), other studies were conducted on both cyclical [32,33] and monotonic temperature variations [34-38]. Hensen [39] reviewed these studies meticulously and found inconsistent results. He offered several possible explanations for these
dissimilarities, including the different voting scales and acceptability criteria and the distinct experimental conditions, among others. Despite these discrepancies, Hensen argued that the experimental results support a 2.2 K/h constraint for cyclical variations in operative temperature. As no evidence had been found to the contrary, he also concluded that this limit could also apply to temperature drifts and ramps. Since this review, only a handful of studies have been conducted on cyclical [40-42] and monotonic variations [43]. Under cyclical variations, these recent studies indicate a positive effect on occupants' thermal comfort. In contrast, for monotonic variations, different rates of temperature change result in inconsistent effects. As mentioned earlier, different acceptability criteria and voting scales could plausibly be the main source of the discrepant findings. Another factor that might be responsible for these differences involves human thermal perception and thermoregulation.

1.2 Human thermal perception and thermoregulation: a brief overview

The skin, the largest organ in the human body, is an interface that separates the body from the rest of the world. On a daily basis, its surface processes at least hundreds of physical sensations, among them environmental thermal stimuli. These stimuli are detected by the free nerve endings of the primary sensory neurons in the skin. These neurones, located in the dorsal root ganglia, convert the external stimuli into electrical signals that are then transmitted to second-order neurons (namely dorsal horn neurons), which are located in the spinal cord [44]. At this first relay centre, thermal information is further processed before being sent to the brain. This pathway is illustrated in Fig. 2.



Fig. 2 – Pathways for autonomous thermoregulatory responses and thermal perception from peripheral tissues. Adapted from Ref. [45].

In neurophysiology, significant progress has been made in identifying primary sensory neurons' thermal response profiles [44,46,47]. Researchers have ascertained that the principal detectors of the thermal stimuli in the peripheral nervous system are the ion channels of the transient receptor potential (TRP) family [44]. These thermosensitive TRPs are triggered at specific threshold temperatures and function as dedicated transducers of distinct thermal modes. Among them, TRPM2/TRPV1 and TRPM8 are the primary sensors of hot and cold temperatures, respectively. Conversely, the understanding of spinal cord temperature encoding remained limited until recently, when Ran et al. [48] showed that the representation of heat and cold in the dorsal horn is substantially different from the operation of TRPs. They observed that the response of cold-sensitive spinal neurons is mostly determined by the rate of cooling and rapidly adapts to the steady-state value. Heat-sensitive spinal neurons, on the other hand, respond primarily to absolute temperatures and are not as adaptable. The interested reader is referred to Ref. [45] for a more detail description and further discussions of this topic.

1.2.1 Thermophysiological models

As mentioned previously, Fanger's PMV model predicts thermal responses (i.e., thermal sensation votes) to a given steady-state environment. These predictions are based on formulae obtained from experimental conditions, making this model empirical. However, the environment where humans carry out their everyday tasks is intrinsically dynamic and non-uniform. Furthermore, Fanger's model regards the body as a whole, de facto excluding the human thermoregulatory system. As a consequence, the human being is viewed as a passive recipient of thermal stimuli.

As an alternative to Fanger's steady-state heat balance model, models that include the thermoregulatory system have emerged. These models include a 'passive system' and an 'active system'. The former simulates the human body and its energy transfer physical mechanisms (both within and between the human body and its environment). In contrast, the latter simulates the thermoregulatory mechanisms that regulate the internal body temperature –vasodilation, vasoconstriction, sweating and shivering.

One of the first and well-known examples of these type of models is Gagge twonode thermal model [49], which is a lumped-parameter model. They split the body into two concentric shells, with the inner shell representing internal organs, bone, muscle, and subcutaneous tissue and the outer shell representing the skin layer. As the temperatures of both shells are usually assumed to be uniform, the model thermally consists of two nodes. Human thermoregulation models have evolved in the past 50 years and more advance complex model have been developed (e.g., the Berkeley Comfort Mode [50], Tanabe [51], Fiala [52], ThermoSEM [53] and Takada [54]). According to Fu ([55] cited in [56]), different models can be classified into four categories: (i) one-node thermal models, (ii) two-node thermal models, (iii) multi-node thermal models and (iv) multi-element thermal models. These models either represent the human body as a single segment or divide it into several parts (i.e., multi-segment) (see Fig. 3). The interested reader is referred to Ref. [57] for further discussions on this topic.





While these models are invaluable to deepen the understanding of dynamic thermal comfort, they require more 'invasive' measurements (e.g., skin temperature sensors) that would make them difficult to be applied directly in everyday control strategies (e.g., temperature control in an office). For this reason, these types of models have been not analysed in this thesis.

1.3 Individual differences

In thermal comfort, individual differences refer to situations where distinct people perceive the same thermal environment differently (i.e., inter-individual differences) and/or when the same individual assesses the same environment differently at different times or in different situations (i.e., intra-individual differences). Humphreys and Nicol [58] suggested that inter-individual differences encompass both temperature differences to be considered neutral and differences in the interpretation of the semantic scale categories. In contrast, intra-individual differences refer to personal judgments that differ from time to time.

In an effort to summarise the factors that in thermal comfort might lead to the individual differences and to assess their significance, Wang et al. [59] reviewed 112

papers (only peer-reviewed conference or journal articles) including both chamber and field studies but presenting them separately. They found out that no consistent conclusions could be drawn on the size and significance of inter-group differences in the preferred/neutral temperature between females and males, nor the young and the elderly. However, it is believed that certain groups of occupants (females and the elderly) are more critical towards the indoor thermal environment, and more sensitive to deviations from an optimal environment, than other sub-populations (males and the young) [60]. Furthermore, numerous contextual factors can also influence occupants' thermal comfort perception, including behavioural and cultural aspects, individual preferences, space layout, architectural features, and adaptive opportunities available [61].

1.3.1 Personal comfort paradigm

Nowadays, all thermal comfort standards are based on models that inherently do not account for individual differences. As a consequence, buildings are designed and operated accordingly. For instance, a traditional, centralized air-conditioning system strives to always maintain an ideal indoor environment with uniformly distributed temperature across the occupied zone. However, because of individual variances in comfort requirements, this 'one-size-fits-all' strategy unavoidably disappoints a considerable portion of building inhabitants. Unsurprisingly, the actual satisfaction percentage of existing building occupants frequently does not reach the ASHRAE Standard's goal satisfaction rate of 80% [62-64].

To address individual differences, a new modelling approach called personal comfort model was derived [65]. In these models, the unit of analysis is the individual rather than a group of people. Personal comfort models can be used to understand individual occupants' specific needs and desires, potentially specifying a set of variables that would fulfil their thermal comfort in a given environment. In this setting, a valid solution is using personally-owned thermal control devices, such as personal comfort systems (PCSs). PCSs concentrate on conditioning the micro-environment of each occupant rather than the entire volume of space within a given thermal zone of a building. For example, PCS such as heated and cooled chairs can address the individual differences in thermal comfort while providing potential energy savings [66]. In this situation, the centralised system oversees keeping the ambient temperatures within a

range that allows the PCS to compensate for each individual's thermal comfort demands. While this approach looks promising, most of these PCSs were tested under controlled conditions, and long-term performance studies from real buildings are lacking. Furthermore, there is a knowledge and tool gap that must be bridged in order for these systems to be extensively used in buildings. For this reason, PCSs have been not used and analysed in this thesis.

1.4 Rating scales for subjective assessment of thermal environments

According to a commonly cited definition, thermal comfort is 'the condition of mind that expresses satisfaction with the thermal environment and is assessed by subjective evaluation' [9]. Regardless the apparent simplicity of this definition, determining such conditions is a complex and partially unsolved matter. In general, capturing the volatile nature of constructs (e.g., thermal comfort) is, to any extent, notoriously difficult. It can be very challenging with just one question or item to consider. However, in thermal comfort research single-item scales are commonly used. The most adopted of which is the ASHRAE 7-point thermal sensation scale, consisting of seven verbal anchors: 'cold', 'cool', 'slightly cool', 'neutral', 'slightly warm', 'warm', and 'hot'. This is a perceptual judgement (single-item) scale [67] and is utilised to measure thermal sensation. Other rating scales are also employed in thermal comfort studies: the most common ones being thermal evaluation, preference, and acceptability. ISO 10551:2019 [67], beyond those already mentioned, also introduces a 'tolerance scale', which is rarely used in the scientific literature. When utilising single-item scales, three significant problematics are commonly expressed: (i) it is improbable that they can capture the construct (poor content validity), (ii) they have fewer discrimination points (low sensitivity), and (iii) they do not have an internal consistency metric (reliability). In 1981, Auliciems [68] specified that the scales of thermal comfort have not been tested for validity and reliability. To our knowledge, thermal comfort scales have not yet been tested. Although the verification of the validity, sensitivity and reliability of the thermal comfort scales is of fundamental importance, it is beyond the scope of this thesis and, therefore, not analysed here. Furthermore, we believe this issue should be addressed with a collaborative effort among researchers within the thermal comfort community and not as a 'single' individual.

Each one of these (single-item) scales used in thermal comfort research can be presented in different formats (e.g., discontinuous versus continuous format) and methods (e.g., paper-versus computer-based). Independently of the format and method used, it is common practice to assign a numerical value to each level (i.e., the verbal anchors) of a scale. For instance, the ASHRAE 7-point thermal sensation scale generally varies from -3 ('cold') to +3 ('hot') with one-unit steps. However, different values can be assigned, such as 1 for 'cold' and 7 for 'hot'. This interchangeability is possible because these numbers are merely placeholders without underlying meaning. Nevertheless, it is common practice to calculate the mean of the thermal sensation votes of a group of people (e.g., [13,26]). The reasoning behind this method is that, while the variable is ordinal in nature, a vote created by averaging different responses is continuous. Furthermore, the averaged votes will result in a more normal-looking distribution and, therefore, statistical methods that assume normality (e.g., linear regression and analysis of variance) can be applied. The origin of this approach can be found in early works to measure attitudes, such as in Thurstone [69] and Likert [70]. However, there are two problems with this approach. Firstly, it is not appropriate to calculate the mean of an ordinal variable because its linearity (i.e., equally spaced divisions) is an arbitrary assumption imposed to the original scale values. This assumption was also recently questioned by Schweiker et al. [71,72]. Secondly, this approach conflates the problem of the level of measurement with that of the distribution of a variable. Averaging ordinal data may improve the degree to which the distribution of votes resembles a normal distribution, but it does not change the nature of the observations from ordinal to interval.

Consequently, in the field of thermal comfort, it is common practice to analyse subjective human thermal responses independently of how they have been measured. That is, the statistical analysis is unrelated to the modalities of the data that have been acquired. For example, even if measured on an ordinal scale, thermal sensation vote (TSV) is generally treated as continuous and analysed with linear regression or other statistical tests that assume (conditional) normality. Since thermal comfort 'is assessed by subjective evaluation' [9], the choice of an appropriate method to analyse these data

Chapter 1 Background

is essential. As mentioned in Section 1.1, Hensen [39], in his review, assigned to different voting scales and acceptability criteria as one of the possible explanation for the inconsistent results.

Discussion regarding the different types of rating scales employed (e.g., categorical scale, visual analogue scale, and graphic categorical scale), the number of anchors utilised, and the assumptions underlying their usage are outside the scope of this thesis. The interested reader is referred to previous studies such as [71-74] for further discussions of these topics.

1.5 Occupant behaviour

Occupant behaviour is complicated and requires a multidisciplinary approach to be appropriately comprehended (if at all possible). On the one hand, both external elements like culture, economics, and climate and internal factors such as individual comfort preferences, physiology, and psychology influence occupant behaviour. On the other hand, occupant behaviour involves interactions with building systems (e.g., adjusting the thermostat, switching lights, opening/closing windows), which greatly influence building operations and, therefore, energy consumption, costs, and comfort. This, in turn, affects the behaviour of the occupants, establishing a closed-loop.

The field of occupant modelling emerged over 40 years ago (e.g., [75]) but has increased recently – particularly because of Annex 66 'Definition and Simulation of Occupant Behavior in Buildings' (<u>https://annex66.iea-ebc.org/</u>) promoted by the International Energy Agency's Energy in Buildings and Communities Programme (IEA EBC). Experimental research methodologies, modelling strategies and model validation, and occupant simulation were all formalised in this Annex. However, several open questions about occupant comfort and behaviour and the implementation of advanced occupant modelling paved the way for the follow-up IEA EBC Annex 79 'Occupant-Centric Building Design and Operation' (https://annex79.iea-ebc.org/).

While it is not within the objectives of this thesis to develop occupant behaviour models, to design and operate (low energy) buildings, it is essential to have a thorough understanding of occupant behaviour and the ability to analyse and quantify its influence on the use of building technologies.

Chapter 2

Knowledge gap and research questions

As highlighted in Section 1, although thermal comfort models for the human body are available for over four decades, several research gaps (both from a technical/methodological and statistical point of views) affect the understanding of this topic. The following gaps were identified as relevant for investigation in this thesis:

- Some knowledge gaps still affect understanding of the human response to changing thermal environments. Among them, 'time' continues to be the least understood variable.
- Despite the fact that thermal comfort is, by definition, 'assessed by subjective evaluation' [9], the relevance of the proper use of rating scales for subjective assessment of thermal environments seems to be overlooked. Particularly, the importance of an appropriate processing of these data and model development.
- Nowadays, although not formally considered into standardisation, the presence of individual differences in thermal comfort is a recognised fact.
 However, its integration (e.g., in the form of human feedback) in both the design and operation of buildings is still lacking.
- There is a little knowledge about occupants' interactions with building technologies. For example, on thermal comfort-driven actions caused by multiple and interdependent environmental influences. Moreover, there is a lack of guidelines and documentation for applying occupant behaviour models during building design and operation.

2.1 Research questions

This thesis aimed to close these knowledge gaps by investigating how enhancing user comfort in (dynamic) thermal indoor environments with particular emphasis on the technical/methodological perspective (i.e., 'experimental aspect') and the subsequent data analysis (i.e., 'modelling and statistics aspect'). It also aims to provide more occupant-centric design and control strategies for the buildings' indoor environment (i.e., 'applicational aspect'). This results in the following research questions:

- RQ1. To what extent can the indoor thermal condition be modulated without compromising occupants' thermal comfort?
- RQ2. How are the rating scales analysed for the subjective assessment of thermal environments?
- RQ3. How can the description of occupants' thermal preferences be used to provide more satisfying control strategies?

As shown in Fig. 4, the answers to these research questions are given using the work developed in three peer-reviewed scientific journal articles: Article I, II and III. These articles constitute the 'main' publications of this thesis. In addition, six peer-reviewed scientific journal articles are appended as 'supplementary' publications: Article a, b, c, d, e and f. These publications constitute the groundwork and points of reflection concerning occupant behaviour and thermal comfort in buildings, used for the different elements analysed in this thesis. More details about the scientific publications are provided in Fig. 5.



Fig. 4 – Outline of research with main research question, objectives, and research activities.

Main publications



Journal article:

Favero, Matteo; Sartori, Igor; Carlucci, Salvatore. Human thermal comfort under dynamic conditions: An experimental study. *Building and Environment* (2021).

Journal article:

Favero, Matteo; Luparelli, Antonio; Carlucci, Salvatore. Analysis of subjective thermal comfort data: a statistical point of view [Manuscript submitted for publication] (2022).

Journal article:

Favero *et al.* Human-in-the-loop methods for occupant-centric building design and operation. *Applied Energy* (2022).

Supplementary publications



Journal article:

Schweiker *et al.* Review of multi-domain approaches to indoor environmental perception and behaviour. *Building and Environment* (2020).

Journal article:

Pisello et al. Test rooms to study human comfort in buildings: A review of controlled experiments and facilities. Renewable and Sustainable Energy Reviews (2021).

Journal article:

Chinazzo *et al.* Quality criteria for multi-domain studies in the indoor environment: critical review towards research guidelines and recommendations [Manuscript submitted for publication] (2022).

Journal article:

Carlucci et al. Modeling occupant behavior in buildings. Building and Environment (2020).

Journal article:

Vellei *et al.* Documenting occupant models for building performance simulation: A state-of-the-art. *Journal of Building Performance Simulation* (2022).

Journal article:

Dong *et al.* A Guideline to Document Occupant Behavior Models for Advanced Building Controls. *Building and Environment* (2022).

Literature review & theoretical framework*





Applicational aspect



Chapter 3 Methods

As mentioned in the Introduction, the general aim of this thesis is to investigate how indoor thermal conditions may be modulated while guaranteeing satisfactory thermal comfort conditions. The general approach utilised in this thesis encompasses both a technical/methodological and a statistical point of view. Both aspects are essential and interrelated since they compose the research design which is based on the specific question driving the research. The research methods used are presented below, arranged by each study.

3.1 Literature review and theoretical framework

These reviews and theoretical frameworks were conducted under the umbrella of the IEA EBC Annex 79 'Occupant-Centric Building Design and Operation' (<u>https://annex79.iea-ebc.org/</u>) and were the result of a collaborative effort of international researchers. These activities were performed to identify current trends, experiences, and outcomes of various aspects related to occupant behaviour in buildings, which constitute the groundwork and point of reflections for the different elements analysed in this thesis (Fig. 4). Specifically, the 'experimental aspect' (Article a, b, and c), 'modelling and statistics aspect' (Article d) and 'applicational aspect' (Article e and f).

3.2 Experimental aspect

The core of this PhD thesis is based on the design and implementation of an exploratory study performed in a climatic chamber furnished like a typical single-office. The experimental study was used to answer RQ1 ('To what extent can the indoor thermal

condition be modulated without compromising occupants' thermal comfort?') and benefited from some outcomes of the Article a, b, and c of the 'supplementary publications'.

3.2.1 Experimental set-up

The facility

The experiment was conducted in the ZEB Test Cell Laboratory on the Norwegian University of Science and Technology (NTNU) premises (Trondheim campus) between September 2019 and January 2020. Two identical climatic chambers (Fig. 6), furnished like a typical single office, were used to recreate a change in the environment induced by thermal ramps. Space heating and cooling were provided from a constant air-volume system that supplied 100% fresh air from outside, distributed by a 2m-long perforated fabric tube installed at the ceiling. Further details on the facility's experimental equipment can be found in Article I.



Fig. 6 – Floor plan of the facility.

Participants

Thirty-eight participants (29 females, 9 males) were recruited from the university campus with a targeted age between 20 and 67. Participation in the experiment was voluntary and in agreement with the principles and instructions of the European General Data Protection Regulation (GDPR). To comply with the GDPR, the experiment description was submitted to the Norwegian Centre for Research Data (NSD) and approved with reference code 525790. Further details about the main demographic and anthropometric characteristics of the subjects can be found in Article I.

Experimental conditions

The operative temperature set-point of $22.0 \,^{\circ}\text{C} \pm 1.0 \,^{\circ}\text{C}$ was defined in accordance with the thermal comfort limit for winter according to Category A of ISO 7730:2005 [10]. Both space heating and cooling variations were tested within winter conditions. The rates of temperature changes, derived from the limit in ASHRAE 55:2020 [9] (Table 1), were: (i) $\pm 4.4 \,\text{K/h}$, (ii) $\pm 3.4 \,\text{K/h}$, (iii) $\pm 2.2 \,\text{K/h}$ and (iv) $\pm 1.4 \,\text{K/h}$. The study's design was a randomised crossover trial, a longitudinal study in which participants received a randomised sequence of thermal ramps. The experimental session was seven and a half hours, including a half-hour lunch break. The day could be split into half days to increase participation, meaning one-morning session (8:00–11:30) and one-afternoon session (12:00–15:30). However, participants were required to attend an even number of morning and afternoon sessions. Further details about the experimental procedure can be found in Article I.

Data collection

The indoor environment was monitored during the experiments by measuring ambient and indoor air temperatures, surface temperatures, globe temperature, relative humidity, airspeed, CO₂ concentration, and horizontal and vertical illuminance every minute throughout every session. The mean radiant temperature (MRT) was calculated according to ISO 7726:1998 [76] based on the surrounding surfaces' measured temperature and the angular factor computed for a seated person in the specific climate chamber. The calculated MRT was combined with the measured air temperature and air velocity to calculate the operative temperature. In addition, a weather station installed in proximity to the southern façade of the ZEB Test Cell measured ambient air temperature, relative humidity, wind speed and direction, global solar irradiance on the horizontal plane and precipitation in 10-minute intervals. Further details about the environmental measurements are given in Article I.

Participants were asked to fill out computer-based questionnaires at different scheduled intervals to evaluate the indoor environment during the experiment. By means of graphic categorical scales (see Fig. 7.a), these questionnaires were used to assess perception, evaluation, preference, and acceptability of the thermal, visual, acoustic and air quality of the environment. It is vital to illustrate how the participants vote and how we acquire this information. Participants could manually draw a diagonal line on the rating scale of the computer-based questionnaire. This questionnaire was an interactive PDF file in which a hidden button allowed, if pressed, to change the scale format into the one shown in Fig. 7.b. This graduated scale allowed us to quantify the vote precisely.



Fig. 7 – Extract from the rating scale used to assess the perception of the room temperature; (a) rating scale displayed by the participant and (b) rating scale displayed for the data analysis.

Furthermore, participants were instructed to press a digital button (see Fig. 8) as soon as they felt uncomfortable. Here uncomfortable was defined as the decision to 'take action to restore a comfort condition' (e.g., if the environment is too warm, then regulate the thermostat or open the window). It is essential to point out that participants could press the button for any source of discomfort related to the indoor environment (e.g., stuffy air, noise from the ventilation system, lack of daylight) and not only for temperature-related discomfort. After pressing the digital button, a computer-based questionnaire appeared on the dedicated laptop. This questionnaire was used to assess the environment and record the source(s) of discomfort through multiple-choice answers.



Fig. 8 – Digital interface for the discomfort button.

In addition, there were two other questionnaires, one at the beginning and one at the end of each session. After arrival, participants were asked to fill out a questionnaire related to demographic and anthropometric characteristics, current clothing level and satisfaction with the workplace. At the end of the session, subjects were asked to fill out a questionnaire about their satisfaction with the workplace as a whole, expressed on a Likert scale. All the survey questions and rating scales are given in the appendix of Article I.

3.2.2 Experimental data analysis

Environmental, demographic and anthropometric data were studied using survival analysis. Survival analysis comprises a family of methods that examine and model the time it takes for events to occur. However, its goal is not limited to investigating the effects on the time until the event occurs but also evaluating the relationship of survival time to covariates. Covariates (often referred to interchangeably as predictors or independent/explanatory variables) assess the impact of certain features on the dependent variable. The prototype event is death – hence the name 'survival analysis' and much of its terminology – but the range of applications of survival analysis is much broader. For example, the same methods are known as 'failure-time analysis' in engineering and 'event-history analysis' in sociology. Further details about survival analysis can be found in Article I.

Kaplan-Meier method and Cox regression

Survival analysis is the name for a collection of statistical techniques. These techniques can be summarised into three categories: (i) non-parametric models, (ii) parametric models, and (iii) semi-parametric models. The main difference between the three categories is whether the survival time is assumed to follow a specific distribution. Nonparametric methods are used when no theoretical distribution adequately fits the data; therefore, they are distribution-free. The Kaplan-Meier method is an example of this category. Conversely, in the parametric model, the underlying distribution of the outcome is specified. For survival analysis, several parametric distributions can be used to describe time to event data, such as exponential, Weibull and log-normal distribution, each of which is defined by a different hazard function. Semi-parametric models are a combination of the two previously mentioned categories. Even if these models' regression parameters (the betas) are known, the outcome's distribution remains unknown. The Cox proportional hazards (PH) model belongs to this category. Since the survival time distribution is unknown during the analysis, non-parametric and semiparametric models were utilised, more specifically, the Kaplan-Meier method and Cox regression. The former has been used only to describe and visualise the survival curves at a preliminary stage, while the latter evaluates the relationship of survival time to covariates. Details on the mathematical formulation and assumption underlying the Kaplan-Meier method and Cox regression can be found in Article I.

3.3 Modelling and statistics aspect

This analysis was used to answer the RQ2 ('How are the rating scales analysed for the subjective assessment of thermal environments?') and benefited from some aspects analysed in the Article d of the 'supplementary publications'.

3.3.1 Statistical modelling

One of the goals of thermal comfort research is to establish a relationship between the thermal environment and the human response. In a statistical modelling framework, this is generally achieved through regression analysis. Regression analysis is 'the blanket name for a family of data analysis techniques that examine relationships between variables' [77], which are categorised into a dependent variable ('outcome' or 'response' variable), *Y*, and one or more independent variables ('explanatory variables', 'predictors', 'covariates' or 'features'), *X*. Here, two different modelling strategies to analyse subjective thermal comfort data were compared: the cumulative probit model and the classical linear regression, referred to as gaussian (ordinal-as-metric) model. Further details on the mathematical formulation of these models can be found in Article II.

Bayesian approach

Bayesian approach was used to analyse the data. This approach is not entirely new in thermal comfort studies (e.g., [78-80]); however, it is not an established practice either. Since statistical knowledge in this field generally tends towards 'frequentist' principles, it is essential to explain the Bayesian approach and compare it with the frequentist one. Frequentist statistics have a more recent history than its philosophical adversary, Bayesian statistics. Frequentist statistics were established mostly in the early 20th century and have lately emerged as the dominant paradigm in inferential statistics, whereas Bayesian statistics were invented in the 19th century. Despite this dominance, there is no consensus on whether frequentism or Bayesian statistics are superior. In addition, the statistical nomenclature is not as old as the philosophical interpretations. The modern statistical terminology 'Bayesian' and 'frequentist' consolidated in the second part of the 20th century.

Essentially, the divide between frequentists and Bayesians is in the definition of probability. For frequentists, probabilities are associated with frequencies of events. For

Chapter 3 Methods

Bayesian, probabilities are related to their own understanding (i.e., certainty or uncertainty) of events. This difference has important implications in the analysis of data. Nevertheless, the aim is neither to go into details about their differences, nor to be a full introduction to either approach. For a more complete treatment, see, for example, [81] and [82]. Furthermore, it is essential to emphasise that we are not advocating that the specific approach used is the best way to analyse subjective thermal comfort data measured on an ordinal scale: the approach presented is merely one of the possible ways to do so. For instance, ordinal models in a frequentist framework provide another valid solution for analysing ordinal data (see *ordinal* package [83]). Statistics is a field that is an art as much as it is a science. Although statistical theory is founded on exact assumptions and conditions, the real world is seldom that straightforward. Consequently, the practice of statistics involves a tremendous number of choices, and the challenge is how to make those choices.

Dataset

The analysis was carried out on Indraganti et al.' study [84] dataset, included in ASHRAE Global Thermal Comfort Database II [85]. This dataset comprised 6048 observations (~27 % female) collected during 14 months from 2787 individuals (all Indian nationals within the age group of 18–48 years). More details regarding the field survey can be found in Indraganti et al. [84]. This dataset was selected because there are no missing values for either thermal sensation votes (i.e., the dependent variable), or gender and air temperature (i.e., the independent variables). Furthermore, the data were collected under a wide range of indoor air temperatures. The analysis was not carried out on the entire ASHRAE Global Thermal Comfort Database II for the following reasons:

- The dependent variable needs to be measured on the ordinal scale.
 Unfortunately, the Comfort Database II does not distinguish between scales, and ordinal and continuous measurements are mixed.
- There are conspicuous missing values in the Comfort Database II. This issue does not derive from the database itself but originates from the lack of explicit agreement on measuring the 'essential' variables in thermal comfort studies.
- Different datasets composed the Comfort Database II. Even though all these datasets went through a rigorous quality assurance process to harmonise

their contents, it is reasonable to assume that each dataset has some unique peculiarities – different measurement protocols, questionnaires, or instruments. This aspect of the database would require that analysis of the entire database be carried out with an 'appropriate' method that considers these peculiarities (e.g., multilevel modelling).

3.4 Applicational aspect

This analysis was used to answer the RQ3 ('How can the description of occupants' thermal preferences be used to provide more satisfying control strategies?') and benefited from some outcomes of the Article e and f of the 'supplementary publications'.

3.4.1 Statistical modelling

Multilevel models (commonly referred to as mixed or hierarchical models) were used to deal with clustered and nested data. When individuals form groups or clusters, it is reasonable to expect that two randomly selected individuals from the same group will tend to be more alike than two individuals chosen from different groups; for example, on the same floor of the same building, two people sharing an office facing south compared to two people sharing an office facing north. Following similar reasoning, measurements taken on the same individual on different occasions will be more highly correlated than measurements taken from different individuals. Therefore, whenever data are clustered and/or nested, the assumption of independent errors is violated.

The dataset used during the analysis comes from the experimental investigation described in Section 3.2 (all the information about the experimental design can be found in Article I). This experimental study examined a mixture of nested and crossed relationships. As shown in Fig. 9, different measurements on the thermal environment (level 1) are nested within experiment conditions (level 2), which, in turn, are cross-classified by participant and day (level 3). It is essential to mention that the multilevel structure defined here is not the property of a model but rather the property of the experimental/study design, which is then reflected in the data, which the model then encapsulates.





Within the multilevel framework, different modelling strategies can be used. Here, two different modelling strategies were applied: the beta mixed-effects model (a beta model including random effects) and the ordinal mixed-effects model (an ordinal model including random effects). The beta mixed-effects model is a suitable choice when the thermal preference votes are measured on a continuous, but bounded, scale. In contrast, the ordinal mixed-effects model is appropriate when a categorical scale is used. For the sake of clarity and brevity, the beta mixed-effects model and the ordinal mixed-effects model will hereafter be referred to as simply the beta model and ordinal model, respectively. Variable selection was performed with an automated backward elimination employing the Akaike information criterion (AIC) as the selection criterion. Further details about these models can be found in Article III.

Computing predictions suitable for building design and operation

Two different procedures were used to handle the group-level residual during prediction. For the ordinal model, the first procedure consisted of holding the group-level residual at its mean of zero and calculating the probabilities for some specific values of the predictors. Holding the group-residual at zero means considering only a part of the subject, namely the one whose random effect is zero. Random effects allow accounting for heterogeneity in the data, for instance, the inherent differences among peoples. However, fixing the random effects to zero is not a requirement, and a different estimated value for the random effect could be chosen. The response probabilities thus calculated have a cluster-specific (or subject-specific in this case) interpretation. This procedure is appropriate during the building operation phase, where the focus should be on satisfying the needs of the specific occupant or type of occupants (i.e., the specific

cluster). The second procedure outlined a simulation-based approach, which resulted in probabilities with a population-averaged interpretation. This interpretation derives from the fact that the calculated probabilities are an average of all simulated random effects (i.e., averaged across experimental conditions, participants and days). This procedure is suitable for the occupant-centric building design phase, where the target is the 'general' occupant whose needs represent those of a large group of typical building users. The same two procedures were applied to the beta model, with the difference that the prediction was not a vector (i.e., probabilities of voting in each category) but rather a single number (i.e., predicted mean). Further details about the two procedures can be found in Article III.

Chapter 4 Results

The results of the main research activities are presented in the following subsections, arranged by each study.

4.1 Human thermal comfort under dynamic environmental conditions

The aim of RQ1 was to explore the human response to dynamic thermal environments. The results are grouped according to (i) general results and observations, (ii) descriptive analysis from the KM method, and (iii) results obtained from the extended Cox model.

4.1.1 General observations

A total of 314 thermal ramps were performed, which led to 223 recorded thermal discomfort events. Fig. 10 presents a time course of the discomfort events during exposure to the different thermal ramps for both the space heating and cooling processes. In this figure, the right-censored observations are also represented (dots without the black outline). Right-censored observations were observed when the experimental session was interrupted because the time available for the session was over. For ease of interpretation, the ASHRAE 55-2020 [9] comfort limit (dark grey X-shaped cross) and a fitted line between this limit (grey dashed line) are also plotted in Fig. 10. It can be clearly seen that the thermal discomfort events are not symmetrical. Participants were more sensitive to a cold variation than a warm one. In fact, 83% of the discomfort events for cold are within the ASHRAE comfort limit, while on the warm side, only 30% are within the comfort limit.



Fig. 10 – Thermal ramps endpoint.

An overview of participants' assessment of perception, evaluation, preference, and acceptability of the thermal environment during the discomfort event is presented in Fig. 11. In this figure, participants' votes on the four previously mentioned rating scales are divided between heating and cooling mode. Particularly:

- a) Thermal sensation: Discomfort events are not symmetric. During space heating, thermal behaviours were undertaken mostly when the environment was sensed as 'warm' (+2) with a temperature difference (ΔT) up to 5K. On the other hand, during space cooling, actions were undertaken when the environment was perceived as 'slightly cool' (-1) and 'cool' (-2). Here the same range of operative temperature change (-3K) was perceived differently.
- b) Thermal comfort: The distribution of discomfort events for space heating and cooling is remarkably similar. Most of the thermal behaviours were undertaken when the environment was judged to be 'slightly uncomfortable' (+1) or

'uncomfortable' (+2) for both space heating and cooling processes. This suggests that, indeed, thermal comfort is the driver for thermal behaviour.

- c) Thermal preference: Most of the actions were undertaken with a thermal preference vote different from 'without change' (0). Reasonably, a participant would initiate a thermal behaviour out of a desire for a higher or lower temperature.
- d) Thermal acceptability: For both space heating and cooling processes, discomfort events follow a skewed distribution, specifically a negative skew (or left-skewed) for acceptable environments and a positive skew (or right-skewed) for unacceptable ones. Consequently, most of the actions were undertaken at the boundary between an acceptable and unacceptable environment.



Fig. 11 – Rating scales for thermal discomfort events.

Note. The data shown here represent only the 'right-here-right-now' votes on the questionnaire at the moment of the thermal discomfort event (i.e., when the digital button was pressed).

4.1.2 KM survival curves

Fig. 12 shows the KM curves for the various thermal ramps, where the plus symbol represents the right censoring. In this figure, it is noticeable that the survivability for warm variations was higher than for cold ones. Also, for both space heating and cooling processes, slower variations led to longer survival than faster variations.



Fig. 12 – KM survival curves for different rates of temperature change.

Fig. 13 shows the log-log plot drawn for each slope (in absolute value). The initial distance between the curves for space heating and cooling processes is greater for a ramp slope of 1.4 K/h than a ramp slope of 4.4 K/h, indicating an effect between the temperature change and the direction of the change (i.e., increase or decrease of the temperature). Moreover, on the whole, all the curves show a divergent-convergent shape: that is the curves initially separate but eventually join up.



Fig. 13 – Log-log survival chart for heating and cooling based on the rate of temperature changes.

In the context of monotonic temperature variations (thermal ramps), warm changes induce thermal discomfort with some delay compared to cold ones, but this delay progressively wears off. The underlying process, that is, the discomfort from thermal ramps, is delayed on the warm side, or stated analogously, the survival is prolonged temporarily. However, it is important to point out that the number of participants still at risk decreased towards the curve's end. Therefore, caution is generally required not to over-interpret the right side of this part of the plot.

4.1.3 Cox-regression

Two separate models were developed for the space heating and cooling processes. This choice had the advantage of assessing the selected covariates' significant predictors separately for the two models. Table 2 lists all the covariates used in the inference of the heating and cooling models.

Variable	Code	Туре	Unit
Thermal resistance of clothing	Clothing	Continuous, time- independent	clo
Gender	Gender	Categorical, time- independent	Female (reference) / Male
Age	Age	Continuous, time- independent	Years
Body Mass Index	BMI	Continuous, time- independent	kg/m ²
Time lived in Norway	Time.Norway	Categorical, time- independent	Less than or equal to 3 years (reference) / More than 3 years
Air velocity	Air.vel	Continuous, time- dependent	m/s
Time of day	Time.day	Categorical, time- independent	Morning (reference) / Afternoon
Vapour pressure	Vap.pre	Continuous, time- dependent	N/m ²
Operative temperature change	Top.delta	Continuous, time- dependent	К
Initial operative temperature	Top.start	Continuous, time- independent	٦°
Participant ID-code	ID.subj	Categorical, time- independent	-

Table 2 – List of covariates used in the model for both space heating and cooling processes.

Before proceeding with the analysis, it is essential to briefly explain how survival analysis evaluates the relationship of survival time to covariates. In survival analysis, the measure of effect is called hazard ratio and is expressed as an exponential of one or more regression coefficients in the model. By taking the logarithm of the hazard on each variable is possible to evaluate if the relationship between each variable and the hazard itself (i.e., the functional form) is linear or not. Further details can be found in Article I.

For spacing heating process four significant predictor were identified (Body Mass Index, time lived in Norway, operative temperature variation and initial operative temperature) all positively associated with increased risk of 'warm discomfort'. Gender, even though was not statistically significant, was still maintained in the model. This

Chapter 4 Results

because gender was found to be a confounder for BMI. Among the statistically significant predictors, there are three continuous variables. Their functional form (i.e., their relationship with the hazard) were showed in Fig. 14. BMI, formerly called the Quetelet index, is a measure for indicating nutritional status in adults. Fig. 14.a shows that the hazard increases from low BMI to around 'normal weight' BMI levels, where it flattens out and then rises slightly in the pre-obesity category, but not significantly. This indicates that participants with lower BMI values have a lower hazard of experiencing 'warm discomfort' than participants with normal and pre-obesity BMI values. This result is not completely in line with the literature. While it is true that the underweight population (BMI < 18.5 kg/m²) is associated with a higher comfortable temperature, the overweight population (i.e., BMI > 25.0 kg/m²) is associated with a lower comfort temperature. However, BMI does not actually measure body fat nor the proportion of muscle-to-fat. Therefore, it is possible that some of the participants were incorrectly classified in the pre-obesity category. Concerning the initial operative temperature, Fig. 14.b shows that a linear fit is within the confidence interval; therefore, a linear relationship between the log(hazard) and the initial operative temperature is assumed (red line). Fig. 14.c shows that the hazard increases linearly with the increment in operative temperature until about +4K, where it flattens out. Nevertheless, conceptually, it is hard to believe that the hazard of thermal discomfort associated with a monotonous rise in operative temperature levels off as higher delta temperatures are reached. A more logical fit would be a continuation of the linear relationship before the +4K increment (the solid green line in Fig. 14.c). A possible explanation for the hazard's flattening upon reaching higher delta temperatures is that different individuals have different frailty levels. More frail individuals are more likely to experience the discomfort event early. Consequently, over time, the 'risk set' has an increasing proportion of less frail individuals, and the hazard flattens out.





Note. The red line is the chosen relationship; the green line is the hypothesized relationship (logical fit); rug plots at the bottom of each plot.

For spacing cooling process three significant predictor were identified (time lived in Norway, time of day and operative temperature variation). Time of day and operative temperature variation were negatively associated with an increased risk of 'cold discomfort', while time lived in Norway was positively associated with the same outcome. Gender and thermal resistance of clothing, even though were not statistically significant, were still maintained in the model. In this case, they remained because their presence improved the model's overall fit compared to the model without them. Fig. 15 showed the relationship between the log(hazard) and the operative temperature variation. Here, the hazard decreases fairly linearly with the decrement in the operative temperature. Therefore, a linear relationship was assumed (red line).



Fig. 15 – Penalised spline fit of operative temperature change for cooling. Note. The red line is the chosen relationship; rug plots at the bottom of the plot.

4.2 Analysis of subjective thermal comfort data

The aim of RQ2 was to highlight the implication of the choice of the statistical method used to analyse subjective thermal comfort data. In this field, it is common practice to analyse subjective human thermal responses independently of how they have been measured. Subjective thermal comfort data are usually measured on an ordinal scale but then treated as continuous and analysed with linear regression or other statistical tests that assume (conditional) normality. The results comparing the cumulative probit and gaussian (ordinal-as-metric) model are grouped according to the modelling step. It is important to mention that this analysis should be regarded as an indicative example:

as such only two variables (gender and air temperature) were used as covariates during the analysis. Thermal sensation vote (TSV) was selected as dependent variable.

4.2.1 Unconditional model

The goal of a modelling strategy is to try to reproduce or predict an observable phenomenon via the lens of a model. Before incorporating a predictor, the unconditional model can be used to test the 'goodness' of the modelling technique. The posterior predictions from the two models are shown in Fig. 16. Here the data generated from the models are compared with the empirical data.





Fig. 16 – Posterior prediction for (a) the thresholds-only and (b) intercept-only model.

The posterior predictive distribution for the cumulative probit model (Fig. 16.a) accurately describes the distribution of the outcomes. Conversely, the posterior predictions for the gaussian (ordinal-as-metric) model (Fig. 16.b) are not a good fit, and they also have impossible predictive outcomes (i.e., value below the category '1' that is, 'cold' and above the category '7', that is, 'hot').

4.2.2 Fitting a categorical variable

In this section, the categorical variable gender is added to the unconditional model. Table 3 shows the results of the fitted cumulative probit and gaussian (ordinal-as-metric) model. The six thresholds for the cumulative probit model and the intercept for the Gaussian (ordinal-as-metric) model were omitted for brevity. The full table can be found in Article II (see Table 5). Moreover, the standard deviation (SD) was modelled on the log scale to constrain its value to be 0 or larger. However, there is a difference in the approach to model the unequal SD for the cumulative probit and the conventional Gaussian model. In Table 3, the *Disc.Male* parameter is not related to the standard deviation (SD) itself, but to the inverse of the SD, that is, $\sigma = 1/disc$. Consequently, the estimated SD for male is $\sigma = 1/\exp(0.14) = 0.87$ and $\sigma = \exp(0.28) = 1.32$ for the cumulative probit and gaussian (ordinal-as-metric) model, respectively.
		Estimate	Est. Error	L-95 % CI*	U-95 % CI*
Cumulative probi	t model				
Gender	female	reference			
	male	0.06	0.03	0.00	0.12
Disc.Male		0.14**	0.02	0.09	0.18
Gaussian (ordinal-as-metri		c) model			
Gender	female	reference			
	male	0.06	0.04	-0.02	0.14
Sigma.Female		0.39**	0.02	0.36	0.42
Sigma.Male		0.28**	0.01	0.26	0.31

Table 3 – Regression coefficients for the model with only a categorical variable (allowing the standard deviation to vary by group).

* CI stands for credible interval (based on quantiles).

** Values expressed on the logarithmic scale.

Fig. 17 shows the marginal posterior distribution of the parameters (i.e., the means and standard deviations) and the effect sizes for the cumulative probit (Fig. 17.a) and gaussian (ordinal-as-metric) (Fig. 17.b) models, respectively. The cumulative probit model does not have a distribution for female because this is the reference category, and its mean and standard deviation are fixed. In this figure, the black line and dot at the bottom of each distribution represent the highest density interval (HDI) and the mode, respectively. The HDI is a way to summarise the distribution by defining an interval that spans over the distribution so that every point inside the interval has higher credibility than any point outside it. These intervals (i.e., the black lines) are defined here to span over 95% of the distribution; therefore, they represent the 95% HDIs.

Focusing on effect sizes and differences in means and standard deviations, two different results can be observed from Fig. 17. For the cumulative probit model, it can be seen that zero is outside the 95 % HDI for the effect size and the difference in means and SD. However, in the gaussian (ordinal-as-metric) model, zero is included in the 95 % HDIs for the effect size and the difference in SD, while it is outside the 95 % HDI for the difference in means. As a consequence, the two models convey different conclusions. More details about the result can be founded in Article II.





Fig. 17 – Posterior distributions for the model that include the variable Gender: (a) cumulative probit and (b) gaussian (ordinal-as-metric) model.

4.2.3 Fitting a linear predictor

In this section, the continuous variable air temperature was added to the previous model, that is, the model with the variable gender and unconstraint standard deviation. However, air temperature was standardised (i.e., subtracting the mean and dividing by its standard deviation) before entering the model. The results of fitting this model are presented in Table 4. The six thresholds for the cumulative probit model and the intercept for the Gaussian (ordinal-as-metric) model were omitted for brevity. The full table can be found in Article II (see Table 6). Here it can be seen that after adding the standardise air temperature as a predictor, the upper and lower 95 % CI (i.e., L-95 % CI and U-95 % CI) for the male coefficient of the gaussian (ordinal-as-metric) model does

not include zero. Consequently, the two models now convey the same conclusions for gender.

		Estimate	Est. Error	L-95 % CI*	U-95 % CI*
Cumulative probit model					
Gender	female	reference			
	male	0.09	0.03	0.03	0.14
Tair.s		0.34	0.01	0.31	0.37
Disc.Male		0.12**	0.02	0.07	0.16
Gaussian (ordinal-as-metri		c) model			
Gender	female	reference			
	male	0.09	0.04	0.01	0.16
Tair.s		0.47	0.02	0.44	0.51
Sigma.Female		0.32**	0.02	0.28	0.35
Sigma.Male		0.22**	0.01	0.20	0.25

Table 4 – Regression coefficients for the model with a categorical and continuous variable (allowing the standard deviation to vary by group).

* CI stands for credible interval (based on quantiles).

** Values expressed on the logarithmic scale.

The marginal distribution of the standardised regression coefficient for air temperature is shown in Fig. 18. This is a standardised regression coefficient and represents a sort of effect size for air temperature. The two models give a different distribution for the coefficient, with a distinct mode and 95 % HDIs. The coefficient of the cumulative probit model is expressed on the underlying latent scale, while the gaussian (ordinal-as-metric) coefficient refers to the ordinal scale. As a consequence, the gaussian (ordinal-as-metric) coefficient for air temperature is overestimated.



Fig. 18 – (a) Standardised and (b) 'original' regression coefficient for air temperature for the cumulative probit (green) and gaussian (ordinal-as-metric) (orange).

4.2.4 Structured thresholds

In all the previous cumulative probit models, the thresholds were defined as 'flexible' providing the standard unstructured thresholds. However, restrictions such as equidistance can be imposed on the thresholds, which restricts the distance between consecutive thresholds to be of the same size (i.e., equally spaced). This allows assessing the assumptions that the subjects used the response scale (i.e., TSV) in such a way that the distance between adjacent response categories is the same. The spacing of the equidistant threshold is plotted in Fig. 19.a. Here, the average distance between consecutive unstructured thresholds is also plotted (Fig. 19.b). It can be seen that zero is outside the 95% HDI for the difference between the spacing for structured and

unstructured thresholds (Fig. 19.c), suggesting that, in terms of 'standard decision rules', the thresholds should not be approximated as equidistant.





Furthermore, whether the restriction on the thresholds is warranted by the data can be assessed formally by comparing the relative fit of the computed models to the data. One method to assess relative fit is approximate leave-one-out cross-validation (LOOCV) [86], where smaller values indicate better fit. Table 5 shows the estimated LOO information criterion (LOOIC) for the two models and their differences. The cumulative probit model with unstructured thresholds has a significantly better fit (smaller LOOIC value) than the structured thresholds one since the difference in LOOIC is very large (more than 12 times the corresponding standard error). In the context of model selection, a LOOIC difference higher than twice its associated standard error suggests that the model with the lower LOOIC value fits the data significantly better.

Table 5 –	Values	of	the	Leave-O	ne-Out	Information	Criterion	(LOOIC)	and	their
difference	e for the	cun	nulati	ve probit	model	with structure	ed and uns	tructured	thresh	olds.

Model	LOOIC	SE	LOOIC.diff*	SE.diff**
Cumulative probit model (unstructured thresholds)	19,449.2	100.0	0.0	0.0
Cumulative probit model (structured thresholds)	20,014.0	97.2	564.81	44.39

* LOOIC.diff is the difference between the two LOOIC scores.

** SE.diff is the standard error of the LOOIC.diff.

4.3 Human-in-the-loop methods for occupant-centric building design and operation

The aim of RQ3 was to predict the thermal preference vote of human subjects exposed to a dynamic thermal environment. Two different modelling strategies were applied: the ordinal model and beta model. The results are grouped accordingly.

4.3.1 Ordinal model

Five significant predictors were identified – thermal resistance of clothing, Body Mass Index, air velocity, time of day and operative temperature – all negatively associated with $\text{Logit}(\gamma_k)$. A negative coefficient for β indicates that an increase of the associated variable x_i decreases the thermal preference vote. Stated analogously, votes for higher categories (e.g., prefer 'higher') are less likely. Here, the coefficient estimates are given in units of ordered logits (or ordered log-odds). Fig. 20 shows the predicted probabilities as functions of the operative temperature for the cluster-specific and populationaveraged procedures. The probabilities calculated with the two methods are dissimilar. For example, the maximum predictive probability for 'without change' is about 91 % for the cluster-specific approach, while it is only 55 % for the population-averaged one.



Fig. 20 – Predicted probabilities of a thermal preference vote using the (a) clusterspecific and (b) population-averaged procedures.

Fig. 21.a shows the probability mass for the ordinal model and cluster-specific procedure. These probabilities are plotted as a function of three different operative temperatures while holding the other covariates constant at their centred values and

fixing the random effects at zero. Fig. 21.b shows the population-averaged procedure's results.



Fig. 21 – Predicted probabilities of a thermal preference vote using the (a) clusterspecific and (b) population-averaged procedures for three different operative temperatures.

4.3.2 Beta model

Four significant predictors were identified – thermal resistance of clothing, Body Mass Index, time of day and operative temperature – all negatively associated with $Logit(\mu)$. Here, the coefficient estimates are given in units of ordered logits (or ordered log-odds). Fig. 22 shows the predicted responses as functions of the operative temperature using the cluster-specific and population-averaged procedures. The points are the observed

thermal preference votes. While the predicted central tendency follows the general trend of the data, the predictions do not agree well with the observations, particularly close to the upper (i.e., prefer 'higher') and lower (i.e., prefer 'lower') boundaries.





Note. The points are the observed thermal preference votes.

Fig. 23.a shows the probability density functions (pdfs) generated from the beta model's estimated parameters using the cluster-specific procedure. Each pdf is plotted as a function of three different operative temperatures while the other covariates are held constant at their centred values and the random effects are fixed at zero. It can be observed that the dispersion of the probability densities is relatively high. For instance, for an operative temperature of 26 °C, the probability of voting equal or lower 0.50 (i.e., from 'lower' to 'without change' on the continuous scale) is about 93 %, implying a 7 % probability of voting higher than that. Fig. 23.b shows the categorised probabilities of







Note. The dotted lines in (a) represent the thresholds used for categorisation.





Note. The dotted lines in (a) represent the thresholds used for the categorisation.

4.3.3 Models' comparison

As explain in Section 4.3, the focus of the study is prediction and not inference; therefore, the specific value of the models' coefficients is not of interest. However, it is useful to compare variables selection across the models and contrast the relative importance of these variables. The automated backward elimination selected different sets of predictors for the two models. Four out of five predictors are shared by the two models, while the fifth variables differ. For the ordinal model, automated backward

elimination selected air velocity, whereas for the beta model, vapour pressure was selected.

In any attempt to understand the relative importance of the parameters estimated for the models, a direct comparison between their absolute values would be meaningless because the variables are measured using different units. Furthermore, several units could be used to measure the same variable. For example, if the operative temperature had been measured in degrees Fahrenheit instead of degrees Celsius, its estimated regression coefficient would have been different. However, the importance of the variable would not have changed. The relative importance of the predictors could be obtained via standardisation (i.e., subtracting the mean from each observed variable and dividing by its standard deviation) before conducting the statistical analysis. The resulting parameters estimated by the model are on the same scale and can be directly compared. The results of this procedure are show in Table 6. Here, even though the two models have different predictors, the order of relative importance of the common predictors is the same. The variables that differ between the two models are of minor relative importance. However, this importance is purely statistical. To determine the practical importance of the variables, subject-area expertise is required. Note that pvalues cannot be used directly to assess the importance of the predictors. A predictor can have a small p-value when it has a very precise estimate, low variability, or a large sample size. As a result, even effect sizes that are small in practice might have extremely low p-values. Understanding the practical importance of the predictors is beyond the scope of this study and is not pursued further. However, for inferential purposes, it is of the utmost importance.

Modelling strategy	Predictor		Standardise coeff	Rank*
Ordinal model	Clothing		-0.464	4
	BMI		-0.567	2
	Air.vel		-0.163	5
	Time.day	morning	Reference	
		afternoon	-0.474	3
	Тор		-2.917	1
Beta model	Clothing		-0.172	4
	BMI		-0.199	2
	Time.day	morning	Reference	
		afternoon	-0.177	3
	Vap.pre		-0.102	5
	Тор		-0.793	1

Table 6 – Predictors' relative importance for both the beta and ordinal models.

*the higher the absolute value of the standardise coefficient, the higher the rank.

The AIC was used for variable selection. This metric is based on the maximised log-likelihood value with a penalty for including more parameters; it is a trade-off between goodness of fit (assessed by the likelihood function) and parsimony (the smaller the number of parameters, the lower the penalty). However, the AIC tends to over-parameterised, thus selecting models with a higher number of predictors, which could explain why the first four relatively important predictors were common to the two models, while their least relatively important predictors differed.

The AIC is generally used to compare different possible models and determine which one best fits the data. However, it cannot be used to compare models with different likelihood functions. For example, for a discrete distribution (e.g., ordinal response), the likelihood refers to the joint probability mass of the data, whereas for a continuous distribution (e.g., continuous response), the likelihood refers to the joint probability density of the data. Therefore, models based on continuous and ordinal responses cannot be compared directly. For this reason, the two models are compared graphically in terms of predicted probabilities. However, it is important to point out that this method poses a limitation: a different categorisation of the beta distribution would

4.3 Human-in-the-loop methods for occupant-centric building design and operation

lead to different probabilities. The same applies for the categorisation of the thermal preference vote used to estimate the ordinal model. Nevertheless, by comparing the probabilities estimated by the two models, the following general observations can be made. On the one hand, the ordinal model is more flexible in the sense that it can handle different probability distributions (virtually any probability distribution). For example, in Fig. 21, it can handle the spike in the probabilities for the 'without change' category for an operative temperature of 23 °C. On the other hand, the beta model is more detailed since it provides a probability density function. For example, in Fig. 23.a, the predicted probability of observing a thermal preference vote between 0.45 and 0.55 for an operative temperature of 23 °C is 19.2 %.

An alternative would be to calculate the mean of the estimated probabilities for the ordinal model and compare it with the predicted mean response of the beta model. Here, the category prefer 'lower' was mapped to 1 and the category prefer 'higher' was mapped to 5. The resulting mean probabilities were then rescaled between +0.001 and +0.999 to match the predicted mean response of the beta model. Fig. 25 shows this comparison.





Note. The points are the observed thermal preference votes.

The curve produced by using the cluster-specific procedure for the ordinal model has three inflexion points. This particular behaviour can be explained by looking at the predicted probabilities in Fig. 21.a. Between the operative temperatures of 22–24 °C, the predicted probabilities for 'without change' were much greater than all the others (from 80 % up to more than 90 %). Consequently, within this range, the calculated mean was greatly affected by these probabilities. The same behaviour can be observed for the population-averaged curve to a lesser extent. However, more considerable differences are visible at the tails of the curves, that is, the two extremities. Here, the beta model's mean response curve has tails that are heavier than the mean of the estimated probabilities for the ordinal model.

The predicted thermal preference votes were calculated from the two models using two different approaches: fixing the random effects at their mean of zero (cluster-specific procedure) and using a simulation approach with $N = 1 \cdot 10^4$ (population-

4.3 Human-in-the-loop methods for occupant-centric building design and operation

averaged procedure). Regarding the ordinal model, from Fig. 21 the most evident difference between the cluster-specific and the population-averaged procedures are the predicted probabilities for an operative temperature of 23 °C. Here, the predictive probability for 'without change' is about 91% for the cluster-specific approach, while it is only 55% for the population-averaged one. The reason for the discrepancy lies in the fact that the variances are not close to zero. As the between-cluster variances in the random-intercepts model increase, the curves will be further apart. The advantage of having predictive probabilities as outcomes is that they are their own error measures. In Fig. 21, the predicted probability of 'without change' for the cluster-specific approach is 91%; if one decided not to choose this category as the expected outcome, the probability of this being an error is, by definition, 91%. Following the same reasoning, for the population-averaged approach, not selecting 'without change' as the expected outcome has a 55% probability of being an error. As a standard practice, the ordinal model regards the category with the highest probability as the predicted outcome (i.e., thermal preference vote). However, utilising a hard threshold, such as the automatic selection of the category with the higher probability, does not fully use the information contained in the probabilities. For example, in Fig. 20.b, such a threshold would lead to 'slightly lower' and 'slightly higher' never being selected. Here the necessity of defining a utility/cost function that, for example, maximises the expected utility or minimises the expected cost. Regarding the beta model, from Fig. 23 and Fig. 24 it can be seen that, for both the cluster-specific and population-averaged procedures, the distributions of the probability densities (and analogously, the categorised probabilities) for an operative temperature equal to 23 °C are the same. The predicted mean response of the beta model intersects the thermal preference vote at 0.5, at which the median equals the mean prediction (see Fig. 25). The median (i.e., cluster-specific) curve is lower than the population-averaged curve for a predicted thermal sensation vote lower than 0.5 but is higher for a predicted thermal sensation vote higher than 0.5. Consequently, the clusterspecific probability densities (i.e., the median probabilities) become skewed faster than the population-averaged ones (i.e., the mean probabilities) at operative temperatures higher or lower than 23 °C.

Chapter 5 Discussion

This thesis investigated how to enhance user comfort in (dynamic) thermal indoor environments and how occupant feedback may help improve the design and control strategies of both new and existing buildings' indoor environments.

The first research question (RQ1), 'To what extent can the indoor thermal condition be modulated without compromising occupants' thermal comfort?', was addressed in Section 4.1. The central answer to this question was provided with a 4-months experiment carried out in the ZEB Test Cell Laboratory at NTNU. Three key results concerning thermal comfort in a dynamic environment were identified. The first one implied that the limits for drifts and ramps are not symmetric in winter conditions. The limits on temperature cycles, drifts and ramps defined in ASHRAE 55-2020 [9] are loose for cold temperature variations and conservative for warm ones. The second one concerned the discomfort mechanism for space heating and cooling processes. For warm discomfort, the operative temperature level is a significant predictor, while for cold discomfort, the relative change in operative temperature is the trigger. Finally, the distributions of participants' thermal comfort ratings during warm and cold discomfort events were remarkably similar, despite different temperature changes. This suggests that, indeed, thermal comfort is the driver of thermal behaviour. In addition, some important methodological issues concerning the semantic equivalence of different rating scales were overcome. For instance, a classic hypothesis (rule-of-thumb) is to consider an environment 'satisfactory' when the thermal sensation vote is between 'slightly cold' (-1) and 'slightly warm' (+1). In this study, this conversion is well suited for warmer variations. Still, it is utterly misleading for colder ones. Fig. 11 showed that most of the discomfort events were experienced when the environment was perceived as 'slightly cold'. In the context of multi-domain comfort, the methodology applied here

Chapter 5 Discussion

could be used to analyse the relation between perception and action. It would also be possible to evaluate which contextual and personal factors affecting behaviour influence perception and vice versa.

The second research question (RQ2) highlighted the implication of the choice of the statistical method used to analyse subjective thermal comfort data. The question was posed as 'How are the rating scales analysed for the subjective assessment of thermal environments?'. The main answers to this question are presented in Section 4.2 as an evaluation of the impact of statistical analysis. The first takeaway from this study is that statistical modelling is a highly sophisticated topic. However, some of these complexities are overlooked when applied to thermal comfort research, specifically the check that some assumptions of the methods used in analysing subjective data are not violated. In a statistical setting, but more generally in science, the veracity of the conclusions is exactly the veracity of the assumptions; that is, the conclusions are contingent upon those assumptions. A critical thinker is aware of these assumptions since they might be wrong or misinformed. The work presented in this section of the thesis sought to address this lack of critical thinking by analysing in detail, in an illustrative example, the modelling steps. The method used to assess the reliability of the models is the so-called posterior predictive checks, a commonly used technique in Bayesian analysis. In essence, after computing the posterior distribution of the parameters, many simulated data are generated and compared with the observed ones. This approach has the evident drawback of evaluating a model against the same data used to estimate its parameters. Unsurprisingly, the model predicts the data used to fit the parameters, but even this simple test fails when the model's assumptions are severely violated. These systematic discrepancies were clearly established: the posterior predictive distributions for the cumulative probit model accurately describes the distribution of the TSV, while the ones of the gaussian (ordinal-as-metric) have impossible predictive outcomes. Furthermore, the two models convey different conclusion regarding the significance of the variables gender and the magnitude of the estimated coefficient for air temperature. In addition, it was shown that the estimated thresholds for the cumulative probit model should not be approximated as equidistant, suggesting that, in this sample, the TSV is not interval-scaled. Nonetheless, it is important to point out that we cannot claim that treating ordinal data as continuous always yields a different result or conclusion than treating them as ordinal. However, knowing in advance that a difference exists is impossible; a different result can be detected only if an ordinal analysis is also performed. Therefore, we strongly discourage the use of linear regression for analysing thermal comfort data measured on an ordinal scale. To improve the reliability of the results, we encourage researchers to use ordinal models.

The outcomes from RQ1 and RQ2 were combined to develop a model to predict the thermal preference vote of human subjects exposed to a dynamic thermal environment. This topic was the subject of the third research question (RQ3) 'How can the description of occupants' thermal preferences be used to provide more satisfying control strategies?'. Section 4.3 dealt with this question and also proposed two different procedures to facilitate the integration of the occupants and their actual needs into buildings. This section's main findings were that the two models (ordinal and beta model) used were both valid strategies for modelling thermal preference votes. On the one hand, the ordinal model is more flexible in the sense that it can handle different probability distributions (virtually any probability distribution). On the other hand, the beta model is more detailed because it provides a probability density function. However, the choice between the ordinal and the beta models should be made based on how the response variable is measured. The beta model is a suitable choice when the thermal preference votes are measured on a continuous, but bounded, scale. In contrast, the ordinal model is appropriate when a categorical scale is used. Concerning the integration of the occupants and their actual needs into buildings the two distinct procedures used were the cluster-specific and the population-averaged procedures. The population-averaged approach is suitable for the occupant-centric building design phase, where the target is the 'general' occupant characterised by features representing a larger population. On the other hand, during the building operation phase, the notion of a 'general' occupant is pointless, and the focus should be on satisfying the needs of the specific occupant or a homogenous group of people with respect to the parameters used during the model development. In this case, a clusterspecific procedure is appropriate. Operationally, this procedure can be carried out by collecting and assessing the specific occupant responses to the environment, and consequently updating the probabilities of the population-averaged procedure that is, therefore, adjusted to match the specific pattern of preferences of the specific occupant/cluster.

5.1 Limitations

The reader is referred to each individual articles for a more detailed discussion about the limitations (see Appendix A). In the following the limitations are presented in subsections, arranged by each study.

5.1.1 Regarding the experimental analysis

This study's limitations arise from the relative homogeneity of age and the unbalanced number of male and female participants. Since most of the participants were between 23 and 31 years old, the results are not completely representative of the office worker population. The gender imbalance among participants might be the main cause of non-statistically significant differences between males and females in terms of thermal discomfort (actions). To reduce the effect of the generally heterogeneous initial metabolic rate, participants spent the first half-hour before starting the session in a constant temperature environment. However, previous studies on thermal comfort in climatic chambers have shown that subjects' average thermal sensation decreases during the first two hours, even during exposure to constant temperatures [38]. On the other hand, time and organisational constraints did not allow for such an extension of this study's acclimation period. Therefore, it is possible that the potential carry-over effects influenced the participants' thermal sensation even after the 30-minute acclimation phase.

Even with some constraints (e.g., clothing adjustment), this study aimed to reproduce a typical office environment and, consequently, simulate a typical office activity pattern. Nevertheless, participants were prone to the Hawthorne effect. Typically, the Hawthorne effect is described as a change in research participants' behaviour in response to their awareness of being observed [87,88]. In this study, to avoid potential bias, participants were blinded to the environmental changes; that is, they were not informed about the change in the room temperature. However, if the participants changed their behaviour during the experiment — for example, by

increasing their awareness and, therefore, sensitivity to change in the indoor environmental condition — the Hawthorne effect would have occurred.

Also, the use of the digital button could have introduced a behavioural change. There is a difference in the intention to perform an action and the action itself. It is undeniable that performing an actual action, for example, adjusting the thermostat, would have required more effort than pressing the digital button. On the other hand, the opposite is also true. A specific human-building interface affects the level of interaction that a person has with it, and therefore its usability, which could lead to a different behavioural choice. Furthermore, it would be unfeasible to provide all the real means of possible interaction with the indoor environment (e.g., for the thermal environment alone, these include open/close window, thermostat adjustment, beverage intake, personalised/local cooler/heater, and ceiling/desk fans). Therefore, even with the aforementioned limitations, the discomfort button was adopted.

5.1.2 Regarding the data analysis

Subjective assessment of thermal environments

A fundamental aspect that is usually overlooked is the assumption of independence: residuals, and thus observations, are assumed to be independent. Non-independence can arise, for example, from temporal and spatial autocorrelation. When underlying spatial or temporal processes have the potential to impact a response, the data are autocorrelated - the closer the observations are in space or time, the more highly correlated they are. These sources of non-independence can be apparent or far less so. The response of one sampling unit influencing the response of other sampling units is an example of evident non-independence. The non-independence caused by nonmeasured confounding influences that vary spatially or temporally is less obvious to detect. Dealing with temporal (or spatial) autocorrelation or analysing temporal (or spatial) trends is different. The former endeavours to deal with the lack of independence associated with temporal (or spatial) data, while the latter tries to model the effect of temporal (or spatial) patterns. During the data analysis stage, it was impossible to identify either spatial or temporal autocorrelation to test the assumption of independence because there was no temporal (e.g., subject ID and timestamp) or spatial (e.g., building ID) information available. Consequently, this assumption was not

Chapter 5 Discussion

checked. Given that the analysis was carried out for illustrative purposes only, this issue can be overlooked. However, in a real-world analysis, the assumption of independence needs to be verified.

Furthermore, other issues, such as functional form misspecification, multicollinearity and omitted variable, were not considered during the analysis because they were outside the scope of this article. Nevertheless, when developing a model, depending on the aim of the study, these issues can play an important role and need to be considered.

Human-in-the-loop

This study's limitations arise from some simplifications introduced during the statistical modelling. For both models, the functional form was assumed to be linear for simplicity. Consequently, the models do not account for potential nonlinearities. However, nonlinearities could be considered, for example, by using smoothing splines. Another simplification derives from assuming that all the independent variables were measured exactly, that is, 'error-free'. When covariates are measured with errors, the parameter estimates do not tend to the true values, even in extensive samples. For simple linear regression, this effect is known as the attenuation bias and leads to an underestimation of the coefficient. For more complex methods, such as multilevel models, this issue deserves a proper treatise and is beyond the scope of this study.

For a beta model, the conditional variance is a function of both the mean μ_i , and the precision parameter ϕ . The parameter ϕ is known as the precision parameter because for fixed μ_i , the larger the ϕ , the smaller the variance of Y_i . Therefore, the variance is not constant but rather a function of the mean and the precision parameter, here assumed to be constant. However, the precision parameter can be modelled as a function of some predictors, for example, the operative temperature. In this study, this possibility was not explored and should examined in future studies.

To apply an ordinal model, the dependent variable must be categorical. For this reason, the dependent variable was binned into five categories according to the thresholds -0.6, -0.2, +0.2, and +0.6. However, these cut-off points were arbitrary and indirectly assumed to be the same for all participants. When a categorical scale is used to measure the dependent variable, this choice is made directly by single responders. Consequently, it is unlikely to be the same for all responders. While this study used

categorisation to apply the ordinal model, we do not encourage this practice in 'normal' circumstances. It is more appropriate to measure the variable directly with a categorical scale. As stated earlier, cut-off points are arbitrary and generally do not have practical/scientific meaning.

Chapter 6 Conclusions

The work presented in this thesis aimed to enhance user comfort in (dynamic) thermal indoor environments, starting from a technical/methodological point of view (i.e., 'experimental aspect'), continuing with the subsequent data analysis (i.e., 'modelling and statistical aspect') and ending providing occupant-centric design and control strategies for the buildings' indoor environment (i.e., 'application aspect'). In the next paragraphs, the main conclusions are drafted.

6.1 Concluding remarks

From a scientific perspective, the main research activities presented in this thesis have potential scientific impacts in diverse aspects. The characterisation of the human response to changing thermal environments is derived from critical literature analysis and formally includes 'time' in the analysis (Article I). The new knowledge developed from the experimental work encompasses both the limits of temperature drifts and ramps and the warm and cold discomfort mechanism. However, this work's potential impact goes beyond this particular application. The methodology applied here could be extended to analyse the relation between perception and action in a multi-domain comfort context. It would also be possible to evaluate which contextual and personal factors affecting behaviour influence perception and vice versa. However, further experiments should be performed using the same experimental method to increase the reliability of the results. Moreover, conducting the investigation in real settings (i.e., in field studies) would be beneficial to avoid some limitations that generally affect experimental studies carried out in climatic chambers (e.g., the Hawthorne effect).

One of the aims of thermal comfort research is to establish the relationship between the thermal environment and the human sensation of warmth. This is usually achieved

Chapter 6 Conclusions and future perspectives

by measuring the subjective human thermal response to different thermal environments. Diverse rating scales are generally used to measure different aspects of thermal comfort, such as thermal sensation, thermal comfort, thermal preference, and thermal acceptability. In this thesis, the focus was on analysing the data once they were collected and not on the correctness of the level of measurement utilised to measure them. The statistical issues highlighted in this thesis (Article II) are not usually mentioned because the modelling steps are rarely presented, and only the final model is described. However, this is a limitation because researchers can neither assess the reliability of the model nor completely understand the limits of its applicability. Hopefully, this thesis will spur researchers to analyse these kinds of data more critically. Nevertheless, to our knowledge, the thermal comfort scales have not yet been tested for validity and reliability. While this verification is of fundamental importance, we believe this issue should be addressed with a collaborative effort among researchers within the thermal comfort community. Moreover, more emphasis should be given to the choice of the different type of rating scales employed (e.g., categorical scale, visual analogue scale, and graphic categorical scale), the number of anchors utilised, and the assumptions underlying their usage. In addition, while not of primary interest in this thesis, it emerged that there is a lack of homogeneity in the collection of common variables within the ASHRAE Global Thermal Comfort Database II. Furthermore, the Database II does not distinguish between scales, and ordinal and continuous measurements are mixed. We recommend that guidelines be developed for defining specific variables to measure. Although there is generally no 'one-size-fits-all' method (e.g., questionnaire) valid for all purposes, agreeing a 'minimum set' of variables to be consistently measured, possibly with a standardised protocol, would undoubtedly benefit the thermal comfort research community. Notably, some of the issues encountered during the analysis of the ASHRAE Global Thermal Comfort Database II have contributed to its improvement. An updated version of Database II released in June 2022 (Ref. [89]) addresses some of the issues found during this PhD (the change log with all updates is available at https://github.com/CenterForTheBuiltEnvironment/ashrae-db-II).

A comfortable indoor climate is a highly subjective matter assessed individually by each occupant. Therefore, subjective data from occupants is critical for understanding buildings' indoor environment, and may be valuable for several applications, including building design and operation. Perhaps, even help bridge the performance gap in present and future buildings. In the literature, there are numerous methods for predicting individual thermal comfort responses. Specifically, machine learning/datadriven algorithms have exploded in popularity recently for this purpose. Although these techniques appear to have the potential to improve prediction ability at the level of a single building occupant, their inherent character as 'black box' models render them fundamentally unfit to explain their outputs. In predictive modelling, direct interpretability regarding the relationship between the predictors and the outcome of interest is not required; however, transparency is desirable. One of the main contributions of this thesis (Article III) is the development of a model aimed at predicting the thermal preference vote of human subjects exposed to a dynamic thermal environment. Here, the data-generation process is viewed as a 'transparent' tool for developing good predictions. However, the modelling strategy does not aim to model the effect of temporal patterns directly but rather to account for them. Two different procedures were proposed to facilitate the integration of the occupants and their actual needs into buildings: the cluster-specific and the population-averaged procedures. The population-averaged approach is suitable for the occupant-centric building design phase, where the target is the 'general' occupant. On the other hand, during the building operation phase, the notion of a 'general' occupant is pointless, and the focus should be on satisfying the needs of the specific occupant. In this case, a cluster-specific procedure is appropriate. This procedure can be carried out by measuring the specific occupant response to the environment and consequently updating the probabilities of the population-averaged procedure. It is hoped that the knowledge generated in this doctoral research will contribute to a transformation in how we design and manage buildings. Buildings should be designed and operated to allow occupants to interact with them, engaging the occupants in maintaining, for instance, a comfortable environment rather than relying on physical metrics alone.

6.1.1 Further conclusions

Building occupants are subjected to a variety of environmental stimuli, the combination and interplay of which influence human perception, physiology, behaviour, and performance. To begin with, a review of multi-domain research was carried out with the goal of emphasising motivating origins, essential methodology, and primary findings of

Chapter 6 Conclusions and future perspectives

multi-domain investigations of human perception and behaviour in indoor environments (Article a). Subsequently, a review paper about laboratory experiments with humans in controlled environments was conducted (Article b). The aim was to identify common features of the different climatic chambers so that, in the future, it would be possible to have standardising test procedures, allowing the reproduction of the same experiments in different contexts. To conclude, a paper that aims to define a framework for designing, conducting, and reporting future multi-domain studies was presented (Article c). Hopefully, this would facilitate the integration of our understanding of multi-domain impacts on human reactions into standards and guidelines.

Understanding how people use a space and how their behaviour impacts on a building's energy performance are of fundamental importance. However, comprehending how occupant behaviour is modelled according to different purposes, available computational power, and technical solutions is critical. To this aim, a study that reviews approaches, methods and key findings related to occupant modelling in buildings was carried out (Article d). Numbers of models developed for occupancy and occupant behaviour have been included in building performance simulation (BPS) tools, especially in the last decades. However, their use, in reality, is still restricted. This may be attributable, in part, to the difficulty in comprehending their utility and the obstacles associated with their adoption into BPS. Both issues are caused by a lack of a framework for their definition and communication. To this end, a paper proposing a framework for documenting occupant models was introduced (Article e). Importantly, this approach might be viewed as a guideline to assist researchers in communicating their models in a straightforward manner. Moreover, occupant behaviour models have been developed to include people in the building control loop. Furthermore, depending on the extent of building automation, including occupant-behaviour modelling into building controls may result in improved building operation and lower energy use. Additionally, this may result in increased thermal comfort and occupant satisfaction with the indoor environment. However, occupant behaviour models are rarely included in building controls. To this end, a paper that offered a guideline for complete and harmonised occupant-behaviour model documentation was presented (Article f).

6.2 Suggested directions for future research

The research in this doctoral thesis is a first step in enhancing the understanding of user comfort in dynamic thermal indoor environments. The research methodologies introduced open the door for new investigations, and the findings generate new research questions. In the following paragraphs, some avenues for further research are described.

Controlling the indoor temperature of grid-interacting buildings within a thermal comfort range is a way to provide energy flexibility to the grid, exploiting the slow thermal inertia of a building's envelope in combination with the users' thermal comfort band. The findings presented in this thesis (Article I) have the potential to improve the performance of such controllers by providing a more accurate description of the human thermal response under dynamic conditions. For example, a model predictive control (MPC) with dynamic temperature constraints can be developed where these constraints are given by applying survival analysis. This would allow dynamic set-point modulations without jeopardising the occupant's thermal comfort. In addition, survival analysis could be used to model recurrent events (time-to-event data), such as discomfort events under cyclical variation of temperature, induced by, for instance, demand response (DR) strategies (e.g., DR-activated smart thermostats).

This thesis focuses on analysing subjective thermal comfort data once they were collected (Article II). However, how to measure them deserves a specific treatment. Psychometrics is a field of study concerned with the objective measurement of latent constructs that cannot be directly observed (e.g., intelligence). A psychometric scale (usually) comprises multiple questions or statements, each with its own rating scale, used to measure the construct of interest reliably and validly. However, thermal comfort scales use only one item (i.e., one question with an associated rating scale) to measure the construct of interest. For example, only one item is used to measure thermal sensation, thermal comfort, thermal preference and thermal acceptability. Although using single-item scales may be sufficient to measure the construct of interest, to our knowledge, thermal comfort scales have not yet been tested for validity and reliability. Also, the interplay among the different aspects of thermal comfort, such as thermal sensation, thermal comfort, thermal preference, and thermal acceptability, should be further explored and examined.

The two different procedures proposed to facilitate the integration of the occupants and their actual needs into the design and operation of buildings (Article III) should be applied in other contexts (e.g., other climatic chamber experiments or real building settings). However, the specific model and the variables should depend on the particular research aim. For instance, if the objective is to predict thermal discomfort, it is sufficient to use the variables associated with this outcome in the model. On the contrary, the association is insufficient if the aim is to prevent thermal discomfort; for this purpose, causality is required. Therefore, it is essential to understand that the modelling process, and consequently the resulting model, is only a means to an end, and the same model should not be used 'indiscriminately' for all purposes.

References

- [1] A. Gelman, J. Hill, Data analysis using regression and multilevel/hierarchical models, Cambridge University Press, Cambridge, 2007.
- [2] W. Haynes, Probability Distributions, in: W. Dubitzky, O. Wolkenhauer, K.-H. Cho, H. Yokota (Eds.), Encyclopedia of Systems Biology, Springer New York, New York, NY, 2013, pp. 1752–1754.
- [3] J.H. Wang, Behavioral Policy Modeling: Consumer Behavior Impacts on Residential Energy Consumption, School of Public Policy, Georgia Institute of Technology, Atlanta 177 (2011).
- [4] N.W. Stauffer, Cooling buildings worldwide: Analysis points the way to energyefcient systems, 2018. <u>https://energy.mit.edu/news/cooling-buildings-worldwide/</u>. (Accessed March 2021).
- [5] European Commission, Indoor Air Pollution: New EU Research Reveals Higher Risks than Previously Thought, 2003. <u>https://ec.europa.eu/commission/presscorner/detail/en/IP_03_1278</u>. (Accessed March 2021).
- [6] M. Schweiker, S. Carlucci, R.K. Andersen, B. Dong, W. O'Brien, Occupancy and Occupants' Actions, Cham: Springer International Publishing, Cham, 2017, pp. 7– 38.
- [7] C.Y.H. Chao, J.S. Hu, Development of a dual-mode demand control ventilation strategy for indoor air quality control and energy saving, Building and environment 39(4) (2004) 385-397. doi:10.1016/j.buildenv.2003.11.001.
- [8] M. Jin, N. Bekiaris-Liberis, K. Weekly, C.J. Spanos, A.M. Bayen, Occupancy Detection via Environmental Sensing, IEEE transactions on automation science and engineering 15(2) (2018) 443-455. doi:10.1109/TASE.2016.2619720.
- [9] ANSI/ASHRAE Standard 55, Thermal environmental conditions for human occupancy, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, GA, 2020.
- [10] ISO 7730, Ergonomics of the thermal environment Analytical determination and interpretation of thermal comfort using calculation of the PMV and PPD indices and local thermal comfort criteria, International Organization for Standardization, Geneva, Switzerland, 2005.
- [11] CEN EN 16798-1, Energy performance of buildings Ventilation for buildings Part 1: Indoor environmental input parameters for design and assessment of energy performance of buildings addressing indoor air quality, thermal environment, lighting and acoustics - Module MI-6, European Committee for Standardization, Brussels, Belgium, 2019.
- [12] CEN EN 15251, Indoor environmental input parameters for design and assessment of energy performance of buildings addressing indoor air quality, thermal environment, lighting and acoustics, European Committee for Standardization, Brussels, Belgium, 2007.
- [13] P.O. Fanger, Thermal comfort: analysis and applications in environmental engineering., Danish Technical Press, Copenhagen, Denmark, 1970.

References

- [14] V.M.D. Araújo, E.H.S. Araújo, The applicability of ISO 7730 for the assessment of the thermal conditions of users of the buildings in Natal-Brazil., in: G. Raw, C. Aizlewood, P. Warren (Eds.) Indoor Air '99, Edinburgh, 1999, pp. 148–153.
- [15] A.A. de Paula Xavier, R. Lamberts, Indices of thermal comfort developed from field survey in Brazil, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., Dallas, TX, 2000, pp. 45–58.
- [16] E. Mayer, A new correlation between predicted mean votes (PMV) and predicted percentages of dissatisfied (PPD). in: J.E. Woods, D.T. Grimsrud, N. Boschi (Eds.) Healthy Buildings/IAQ '97, Washington, DC, 1997, pp. 189–194.
- [17] D.W. Yoon, J.Y. Sohn, K.H. Cho, The comparison on the thermal comfort sensation between the results of questionnaire survey and the calculation of the PMV values., in: G. Raw, C. Aizlewood, P. Warren (Eds.) Indoor Air '99, Edinburgh, 1999, pp. 137– 141.
- [18] J.F. Nicol, M.A. Humphreys, Thermal comfort as part of a self-regulating system, Building Research and Practice 1(3) (1973) 174–179. doi:10.1080/09613217308550237.
- [19] M.A. Humphreys, Field studies of thermal comfort compared and applied, The Building Services Engineer 44 (1976) 5–27.
- [20] M.A. Humphreys, Outdoor temperatures and comfort indoors, Batiment International, Building Research and Practice 6(2) (1978) 92–92. doi:10.1080/09613217808550656.
- [21] M.A. Humphreys, Chapter 15 the Dependence of Comfortable Temperatures upon Indoor and Outdoor Climates**The work described has been carried out as part of the research programme at the Building Research Establishment and is contributed by permission of the Director. British Crown Copyright, in: K. Cena, J.A. Clark (Eds.), Studies in Environmental Science, Elsevier1981, pp. 229–250.
- [22] D.A. McIntyre, Indoor Climate, Applied Science Publishers1980.
- [23] A. Auliciems, Towards a psycho-physiological model of thermal perception, International Journal of Biometeorology 25(2) (1981) 109–122. doi:10.1007/BF02184458.
- [24] R.J. de Dear, G.S. Brager, Developing an adaptive model of thermal comfort and preference, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., San Francisco, CA, 1998, pp. 27-49.
- [25] M.A. Humphreys, J.F. Nicol, Understanding the adaptive approach to thermal comfort, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., San Francisco, CA, 1998, pp. 991–1004.
- [26] R.J. de Dear, G.S. Brager, D. Cooper, Developing an adaptive model of thermal comfort and preference Final Report on ASHRAE RP-884, 1997.
- [27] R.J. de Dear, G.S. Brager, Thermal comfort in naturally ventilated buildings: revisions to ASHRAE Standard 55, Energy and buildings 34(6) (2002) 549–561. doi:10.1016/S0378-7788(02)00005-1.
- [28] F.H. Rohles, Thermal Comfort and Strategies for Energy Conservation, Journal of Social Issues 37(2) (1981) 132–149. doi:10.1111/j.1540-4560.1981.tb02629.x.
- [29] F.H. Rohles, G.A. Milliken, D.E. Skipton, I. Krstic, Thermal comfort during cyclical temperature fluctuations, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., Denver, CO, 1980, pp. 125–140.
- [30] C.H. Sprague, P.E. McNall, The effects of fluctuating temperature and relative humidity on the thermal sensation (thermal comfort) of sedentary subjects, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., San Francisco, CA, 1970, pp. 146–156.

- [31] D.P. Wyon, T. Asgeirsdottir, P.K. Jensen, P.O. Fanger, The effects of ambient temperature swings on comfort, performance and behavior, ARCH.SCI.PHYSIOL. 27(4) (1973) 441–458.
- [32] D.P. Wyon, N.O. Bruun, S. Olesen, P. Kjerulf-Jensen, P.O. Fanger, Factors affecting the subjective tolerance of ambient temperature swings, Proc. 5th Int. Congress for Heating, Ventilating and Air-Conditioning, 1971, pp. 87–107.
- [33] R. Nevins, R.R. Gonzalez, L.G. Berglund, A.P. Gagge, Effect of changes in ambient temperature and level of humidity on comfort and thermal sensations, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., Boston, MA, 1975, pp. 169–182.
- [34] I.D. Griffiths, D.A. McIntyre, Sensitivity to temporal variations in thermal conditions, Ergonomics 17(4) (1974) 499–507. doi:10.1080/00140137408931380.
- [35] L.G. Berglund, R.R. Gonzalez, Occupant acceptability of eight-hour-long temperature ramps in the summer at low and high humidities, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., Albuquerque, NM, 1978, pp. 278–284.
- [36] L.G. Berglund, R.R. Gonzalez, Application of acceptable temperature drifts to built environments as a mode of energy conservation, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditioning Eng. Inc., Atlanta, GA, 1978, pp. 110–121.
- [37] F.H. Rohles, J.E. Laviana, R. Wei, R. Wruck, The human response to temperature drifts in a simulated office environment, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., Chicago, IL, 1985, pp. 116–123.
- [38] H.N. Knudsen, R.J. de Dear, J.W. Ring, T.L. Li, T.W. Püntener, P.O. Fanger, Thermal comfort in passive solar buildings. CEC Research Project: EN3S-0035-DK(B), 1989.
- [39] J.L.M. Hensen, Literature review on thermal comfort in transient conditions, Building and Environment 25(4) (1990) 309–316. doi:10.1016/0360-1323(90)90004-B.
- [40] M. Miura, T. Ikaga, Human response to the indoor environment under fluctuating temperature, Science and Technology for the Built Environment 22(6) (2016) 820– 830. doi:10.1080/23744731.2016.1184550.
- [41] F. Zhang, R.J. de Dear, C. Candido, Thermal comfort during temperature cycles induced by direct load control strategies of peak electricity demand management, Building and Environment 103 (2016) 9–20. doi:10.1016/j.buildenv.2016.03.020.
- [42] L. Schellen, W.D. van Marken Lichtenbelt, M.G.L.C. Loomans, J. Toftum, M.H. de Wit, Differences between young adults and elderly in thermal comfort, productivity, and thermal physiology in response to a moderate temperature drift and a steadystate condition, Indoor Air 20(4) (2010) 273–283. doi:10.1111/j.1600-0668.2010.00657.x.
- [43] J. Kolarik, J. Toftum, B.W. Olesen, A. Shitzer, Occupant responses and office work performance in environments with moderately drifting operative temperatures (RP-1269), HVAC and R Research 15(5) (2009) 931–960. doi:10.1080/10789669.2009.10390873.
- [44] J. Vriens, B. Nilius, T. Voets, Peripheral thermosensation in mammals, Nature Reviews Neuroscience 15(9) (2014) 573–589. doi:10.1038/nrn3784.
- [45] M. Vellei, R.J. de Dear, C. Inard, O. Jay, Dynamic thermal perception: A review and agenda for future experimental research, Building and environment 205 (2021) 108269. doi:10.1016/j.buildenv.2021.108269.
- [46] A. Dhaka, V. Viswanath, A. Patapoutian, TRP ion channels and temperature sensation, Annual Review of Neuroscience, 2006, pp. 135–161.
- [47] R. Palkar, E.K. Lippoldt, D.D. McKemy, The molecular and cellular basis of thermosensation in mammals, Current Opinion in Neurobiology 34 (2015) 14–19. doi:10.1016/j.conb.2015.01.010.

- [48] C. Ran, M.A. Hoon, X. Chen, The coding of cutaneous temperature in the spinal cord, Nature Neuroscience 19(9) (2016) 1201–1209. doi:10.1038/nn.4350.
- [49] A.P. Gagge, J.A.J. Stolwijk, Y. Nishi, An effective temperature scale based on a simple model of human physiological regulatiry response, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., Philadelphia, PA, 1972, pp. 247– 262.
- [50] C. Huizenga, H. Zhang, E. Arens, A model of human physiology and comfort for assessing complex thermal environments, Building and Environment 36(6) (2001) 691–699. doi:10.1016/S0360-1323(00)00061-5.
- [51] S.-i. Tanabe, K. Kobayashi, J. Nakano, Y. Ozeki, M. Konishi, Evaluation of thermal comfort using combined multi-node thermoregulation (65MN) and radiation models and computational fluid dynamics (CFD), Energy and Buildings 34(6) (2002) 637– 646. doi:10.1016/S0378-7788(02)00014-2.
- [52] D. Fiala, G. Havenith, P. Bröde, B. Kampmann, G. Jendritzky, UTCI-Fiala multi-node model of human heat transfer and temperature regulation, International Journal of Biometeorology 56(3) (2012) 429–441. doi:10.1007/s00484-011-0424-7.
- [53] B.R.M. Kingma, L. Schellen, A.J.H. Frijns, W.D. van Marken Lichtenbelt, Thermal sensation: a mathematical model based on neurophysiology, Indoor Air 22(3) (2012) 253–262. doi:10.1111/j.1600-0668.2011.00758.x.
- [54] S. Takada, S. Matsumoto, T. Matsushita, Prediction of whole-body thermal sensation in the non-steady state based on skin temperature, Building and Environment 68 (2013) 123–133. doi:10.1016/j.buildenv.2013.06.004.
- [55] G. Fu, A transient, 3-D mathematical thermal model for the clothed human, Kansas State University, 1995, p. 540.
- [56] R. Holopainen, A human thermal model for improved thermal comfort, Aalto University, 2012, p. 143.
- [57] K. Katić, R. Li, W. Zeiler, Thermophysiological models and their applications: A review, Building and Environment 106 (2016) 286–300. doi:10.1016/j.buildenv.2016.06.031.
- [58] M.A. Humphreys, J. Fergus Nicol, The validity of ISO-PMV for predicting comfort votes in every-day thermal environments, Energy and buildings 34(6) (2002) 667– 684. doi:10.1016/S0378-7788(02)00018-X.
- [59] Z. Wang, R.J. de Dear, M. Luo, B. Lin, Y. He, A. Ghahramani, Y. Zhu, Individual difference in thermal comfort: A literature review, Building and environment 138 (2018) 181–193. doi:10.1016/j.buildenv.2018.04.040.
- [60] S. Karjalainen, Thermal comfort and gender: a literature review, Indoor Air 22(2) (2012) 96–109. doi:10.1111/j.1600-0668.2011.00747.x.
- [61] R.F. Rupp, N.G. Vásquez, R. Lamberts, A review of human thermal comfort in the built environment, Energy and buildings 105 (2015) 178–205. doi:10.1016/j.enbuild.2015.07.047.
- [62] C. Huizenga, S. Abbaszadeh, L. Zagreus, E.A. Arens, Air quality and thermal comfort in office buildings: results of a large indoor environmental quality survey, Healthy Buildings 2006, Lisbon, Portugal, 2006, pp. 393–397.
- [63] E. Arens, M.A. Humphreys, R.J. de Dear, H. Zhang, Are 'class A' temperature requirements realistic or desirable?, Building and environment 45(1) (2010) 4–10. doi:10.1016/j.buildenv.2009.03.014.
- [64] C. Karmann, S. Schiavon, L.T. Graham, P. Raftery, F. Bauman, Comparing temperature and acoustic satisfaction in 60 radiant and all-air buildings, Building and environment 126 (2017) 431–441. doi:10.1016/j.buildenv.2017.10.024.
- [65] J. Kim, Y. Zhou, S. Schiavon, P. Raftery, G.S. Brager, Personal comfort models: Predicting individuals' thermal preference using occupant heating and cooling
behavior and machine learning, Building and environment 129 (2018) 96–106. doi:10.1016/j.buildenv.2017.12.011.

- [66] H. Zhang, E. Arens, Y. Zhai, A review of the corrective power of personal comfort systems in non-neutral ambient environments, Building and environment 91 (2015) 15–41. doi:10.1016/j.buildenv.2015.03.013.
- [67] ISO 10551, Ergonomics of the physical environment Subjective judgement scales for assessing physical environments, International Organization for Standardization, Geneva, Switzerland, 2019.
- [68] A. Auliciems, Psycho-physiological criteria for global thermal zones of building design, International Journal of Biometeorology 26 (1981) 69–86.
- [69] L.L. Thurstone, Attitudes Can Be Measured, Am J Sociol 33(4) (1928) 529–554. doi:10.1086/214483.
- [70] R. Likert, A technique for the measurement of attitudes, Archives of Psychology 22(140) (1932) 55.
- [71] M. Schweiker, M. André, F. Al-Atrash, H. Al-Khatri, R.R. Alprianti, H. Alsaad, R. Amin, E. Ampatzi, A.Y. Arsano, M. Azadeh, E. Azar, B. Bahareh, A. Batagarawa, S. Becker, C. Buonocore, B. Cao, J.-H. Choi, C. Chun, H. Daanen, S.A. Damiati, L. Daniel, R.D. Vecchi, S. Dhaka, S. Domínguez-Amarillo, E. Dudkiewicz, L.P. Edappilly, J. Fernández-Agüera, M. Folkerts, A. Frijns, G. Gaona, V. Garg, S. Gauthier, S.G. Jabbari, D. Harimi, R.T. Hellwig, G.M. Huebner, Q. Jin, M. Jowkar, J. Kim, N. King, B. Kingma, M.D. Koerniawan, J. Kolarik, S. Kumar, A. Kwok, R. Lamberts, M. Laska, M.C.J. Lee, Y. Lee, V. Lindermayr, M. Mahaki, U. Marcel-Okafor, L. Marín-Restrepo, A. Marquardsen, F. Martellotta, J. Mathur, I. Mino-Rodriguez, D. Mou, B. Moujalled, M. Nakajima, E. Ng, M. Okafor, M. Olweny, W. Ouyang, A.L.P.d. Abreu, A. Pérez-Fargallo, I. Rajapaksha, G. Ramos, S. Rashid, C.F. Reinhart, M.I. Rivera, M. Salmanzadeh, K. Schakib-Ekbatan, S. Schiavon, S. Shooshtarian, M. Shukuya, V. Soebarto, S. Suhendri, M. Tahsildoost, F. Tartarini, D. Teli, P. Tewari, S. Thapa, M. Trebilcock, J. Trojan, R.B. Tukur, C. Voelker, Y. Yam, L. Yang, G. Zapata-Lancaster, Y. Zhai, Y. Zhu, Z. Zomorodian, Evaluating assumptions of scales for subjective assessment of thermal environments - do laypersons perceive them the way, we Energy researchers believe?. and **Buildings** 211 (2020). doi:10.1016/j.enbuild.2020.109761.
- [72] M. Schweiker, X. Fuchs, S. Becker, M. Shukuya, M. Dovjak, M. Hawighorst, J. Kolarik, Challenging the assumptions for thermal sensation scales, Building Research & Information (2016) 1–18. doi:10.1080/09613218.2016.1183185.
- [73] J.-Y. Lee, E.A. Stone, H. Wakabayashi, Y. Tochihara, Issues in combining the categorical and visual analog scale for the assessment of perceived thermal sensation: Methodological and conceptual considerations, (2010) 282–290. doi:10.1016/j.apergo.2009.07.007.
- [74] G.A. Miller, The magical number seven, plus or minus two: some limits on our capacity for processing information, Psychol Rev 63(2) (1956) 81–97. doi:10.1037/h0043158.
- [75] P.R. Boyce, Observations of the manual switching of lighting, Lighting Research & Technology 12(4) (1980) 195–205. doi:10.1177/096032718001200403.
- [76] ISO 7726, Ergonomics of the thermal environment Instruments for measuring physical quantities, International Organization for Standardization, Geneva, Switzerland, 1998.
- [77] P. Lavrakas, Encyclopedia of Survey Research Methods, (2008). doi:10.4135/9781412963947.n461.

- [78] S. Crosby, A. Rysanek, Predicting thermal satisfaction as a function of indoor CO2 levels: Bayesian modelling of new field data, Building and Environment 209 (2022) 108569. doi:10.1016/j.buildenv.2021.108569.
- [79] J. Langevin, J. Wen, P.L. Gurian, Modeling thermal comfort holistically: Bayesian estimation of thermal sensation, acceptability, and preference distributions for office building occupants, Building and Environment 69 (2013) 206–226. doi:10.1016/j.buildenv.2013.07.017.
- [80] L.T. Wong, K.W. Mui, C.T. Cheung, Bayesian thermal comfort model, Building and Environment 82 (2014) 171–179. doi:10.1016/j.buildenv.2014.08.018.
- [81] A. Gelman, J.B. Carlin, H.S. Stern, D.B. Dunson, A. Vehtari, D.B. Rubin, Bayesian Data Analysis, third ed.2013.
- [82] L. Wasserman, All of Statistics: a Concise Course in Statistical Inference, Springer, New York, NY2004.
- [83] R.H.B. Christensen, ordinal---Regression Models for Ordinal Data, 2019. https://CRAN.R-project.org/package=ordinal
- [84] M. Indraganti, R. Ooka, H.B. Rijal, G.S. Brager, Adaptive model of thermal comfort for offices in hot and humid climates of India, Building and Environment 74 (2014) 39–53. doi:10.1016/j.buildenv.2014.01.002.
- [85] V. Földváry Ličina, T. Cheung, H. Zhang, R.J. de Dear, T. Parkinson, E. Arens, C. Chun, S. Schiavon, M. Luo, G.S. Brager, P. Li, S. Kaam, ASHRAE Global Thermal Comfort Database II, (2018). doi:10.6078/D1F671.
- [86] A. Vehtari, A. Gelman, J. Gabry, Practical Bayesian model evaluation using leaveone-out cross-validation and WAIC, Statistics and Computing 27(5) (2017) 1413– 1432. doi:10.1007/s11222-016-9696-4.
- [87] R. McCarney, J. Warner, S. Iliffe, R. Van Haselen, M. Griffin, P. Fisher, The Hawthorne Effect: A randomised, controlled trial, BMC Med Res Methodol 7(1) (2007) 30–30. doi:10.1186/1471-2288-7-30.
- [88] N.S. Fox, J.S. Brennan, S.T. Chasen, Clinical estimation of fetal weight and the Hawthorne effect, Eur J Obstet Gynecol Reprod Biol 141(2) (2008) 111–114. doi:10.1016/j.ejogrb.2008.07.023.
- [89] T. Parkinson, F. Tartarini, V. Földváry Ličina, T. Cheung, H. Zhang, R.J. de Dear, P. Li, E. Arens, C. Chun, S. Schiavon, M. Luo, G. Brager, ASHRAE global database of thermal comfort field measurements, (2022). doi:10.6078/D1F671.

Appendix A: Main publications

The following journal articles constitutes the main publications of this thesis.

- M. Favero, I. Sartori, S. Carlucci, Human thermal comfort under dynamic conditions: An experimental study, Building and Environment 204 (2021) 108144. doi:10.1016/j.buildenv.2021.108144.
- (II) M. Favero, A. Luparelli, S. Carlucci, Analysis of subjective thermal comfort data: a statistical point of view [Manuscript submitted for publication] (2022).
- (III) M. Favero, J. Kloppenborg Møller, D. Cali, S. Carlucci, Human-in-the-loop methods for occupant-centric building design and operation, Applied Energy 325 (2022) 119803. doi:10.1016/j.apenergy.2022.119803.



Building and Environment 204 (2021) 108144

Contents lists available at ScienceDirect



Building and Environment

journal homepage: www.elsevier.com/locate/buildenv



Human thermal comfort under dynamic conditions: An experimental study



Matteo Favero^{a,*}, Igor Sartori^b, Salvatore Carlucci^{c,a}

^a Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway

^b SINTEF Community, P.O.Box 124, Blindern 0314, Oslo, Norway

^c Energy, Environment and Water Research Center, The Cyprus Institute, Nicosia, Cyprus

ARTICLEINFO

Keywords: Temperature ramps Dynamic and non-uniform thermal environments Thermal comfort Experimental study Survival analysis

ABSTRACT

Although thermal comfort has been a research topic since the 1960s, some knowledge gaps still affect understanding of the human response to changing thermal environments. To enhance knowledge in this regard, an exploratory study is presented, which aims to understand human response to monotonic thermal variations by describing its relationship with covariates of interest. Thirty-eight participants (29 females, 9 males) worked in an office-like climate chamber and were exposed to dynamic and controlled heating and cooling ramps of the operative temperature with different speeds. Participants' perception, evaluation, preference and acceptability of the indoor thermal environment were recorded by filling in dedicated questionnaires. Additionally, participants could indicate when an uncomfortable event occurred during these temperature ramps by clicking a digital button on a dedicated app. This discomfort event was defined in behavioural terms as the decision to "take action to restore a comfort condition". Survival analysis was used to study participants' reactions to the dynamic thermal stimuli. It showed that two distinct mechanisms caused discomfort events due to overheating and undercooling: warm discomfort is driven by the absolute value of the achieved operative temperature, while the relative change in operative temperature mainly causes cold discomfort. Compared to the current recommendations regarding temperature cycles, drifts and ramps, this result shows that current standard recommendations underestimate the risk of thermal discomfort during a cooling process while overestimating it during a heating one. The new knowledge of human reaction to a dynamic thermal environment can lead to more energy-efficient and satisfactory building control strategies.

15 min as drifts or ramps.

temperature repeatedly rises and falls, and the period of these variations is not greater than 15 min" [2]. For these changes, ASHRAE 55 allows a

maximum peak-to-peak cyclic variation in operative temperature of 1.1

K and recommends treating cyclic variations with a period greater than

ture cycles, it sets a maximum peak-to-peak variation of 1 K, whereas,

for drifts and ramps with a rate of change lower than 2.0 K/h, it pre-

scribes steady-state methods. These standards also include step-changes,

which involve changing the environment (i.e., moving to/from another

space) rather than a change within the environment. Consequently, they

tory studies of thermal comfort under transient exposure [4-6]. During

the same period (the 1970s and 1980s), other studies were conducted on

both cyclical [7,8] and monotonic temperature variations [9-13].

Hensen [14] reviewed these studies meticulously and found inconsistent

The limiting criteria in Table 1 are probably based on early labora-

are not described here because out of the scope of this study.

ISO 7730-2005 [3] provides less detailed indications. For tempera-

1. Introduction

Thermal comfort is a consolidated research subject, first incorporated into standardisation in 1966 [1]. After that, standardisation bodies produced standards dedicated to thermal comfort in moderate and severe thermal environments and indoor environmental quality. Nowadays, all thermal comfort standards include definitions of the requirements for indoor thermal conditions in buildings both for design and operational assessment. However, current standards only indicate the maximum variations in operative temperature for non-steady-state thermal environments. ASHRAE 55-2017 [2] and ISO 7730-2005 [3] classify temperature variations as either temperature drifts and ramps or temperature cycles. Drifts and ramps are defined as "monotonic, non-cyclic changes in operative temperature" [2], and their limits during a period are shown in Table 1. Drifts refer to passive temperature changes in an enclosed space, while ramps denote actively controlled ones. In contrast, cycles refer to "those situations where the operative

https://doi.org/10.1016/j.buildenv.2021.108144

Available online 16 July 2021

0360-1323/© 2021 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Corresponding author. 7a Høgskoleringen, 7034, Trondheim, Norway. *E-mail address:* matteo.favero@ntnu.no (M. Favero).

Received 23 April 2021; Received in revised form 8 July 2021; Accepted 11 July 2021

M. Favero et al.

Table 1

Limits on temperature drifts and ramps by ASHRAE 55-2017 [2].

Time period (h)	Maximum operative temperature to change allowed (K)
0.25	1.1
0.5	1.7
1	2.2
2	2.8
4	3.3

results. He offered several possible explanations for these dissimilarities, including the different voting scales and acceptability criteria and the distinct experimental conditions, among others. Despite these discrepancies, Hensen argued that the experimental results support a 2.2 K/h constraint for cyclical variations in operative temperature. As no evidence had been found to the contrary, he also concluded that this limit could also apply to temperature drifts and ramps. Since this review, only a handful of studies have been conducted on cyclical [15-17] and monotonic variations [18]. Under cyclical variations, these recent studies indicate a positive effect on occupants' thermal comfort. In contrast, for monotonic variations, different rates of temperature change result in inconsistent effects. As mentioned earlier, different acceptability criteria and voting scales could plausibly be a main source of the discrepant findings. Another factor that might be responsible for these differences involves human thermal perception and thermoregulation, described in the following sections.

1.1. Thermal perception and thermoregulation

The skin, the largest organ in the human body, is an interface that separates the body from the rest of the world. On a daily basis, its surface processes at least hundreds of physical sensations, among them environmental thermal stimuli. These stimuli are detected by the free nerve endings of the primary sensory neurons in the skin. These neurones, located in the dorsal root ganglia, convert the external stimuli into electrical signals that are then transmitted to second-order neurons (namely dorsal horn neurons), which are located in the spinal cord [19]. At this first relay centre, thermal information is further processed before being sent to the brain.

In neurophysiology, significant progress has been made in identifying primary sensory neurons' thermal response profiles [19-21]. Researchers have ascertained that the principal detectors of the thermal stimuli in the peripheral nervous system are the ion channels of the transient receptor potential (TRP) family [19]. These thermosensitive TRPs are triggered at specific threshold temperatures and function as dedicated transducers of distinct thermal modes. Among them, TRPV1 and TRPM8 are the primary sensors of hot and cold temperatures, respectively. Conversely, the understanding of spinal cord temperature encoding remained limited until recently, when Ran et al. [22] showed that the representation of heat and cold in the dorsal horn is substantially different from the operation of TRPs. They observed that heat-responding neurons are activated gradually with incremental increases in temperature, where higher temperatures activate more neurons. Therefore, higher absolute temperatures induce stronger neuron responses. Furthermore, if a steady heat stimulus persists, these neurons are not able to adapt and thus persistently respond to it. These results combined suggest that heat-responding spinal neurons encode the absolute temperature. Conversely, cold-responding neurons' reaction reaches its highest point during the cooling phase but rapidly adapts to steady cold stimuli. This behaviour allows these neurons to signal changes over a wide range of environmental temperatures. Therefore, they communicate a relative drop in absolute skin temperature rather than absolute skin temperature. As a result, from a neurophysiological point of view [22], the response to heat (i.e., an increase in temperature) in the spinal cord is encoded in absolute terms (i.e., a certain temperature level), whereas the response to cold (i.e., a decrease in temperature)

Building and Environment 204 (2021) 108144

is coded in relative terms (i.e., a certain temperature difference).

1.2. Thermal alliesthesia

Skin receptors (thermoreceptors), although ideal for sensing changes in the environmental temperature, do not perform well in detecting increases in core temperature, for example, during exercise. This is because the body's internal temperature would increase to an unbearably high level before the skin thermoreceptors could detect it. Not surprisingly, the body is provided with other temperature-sensitive neurons, located throughout the body core (e.g., in the liver, kidneys, and stomach) and in the brain (i.e., the preoptic hypothalamus), that play a major role in detecting changes in deep-body temperature. Nevertheless, given the body's thermal inertia, these neurons are not suitable for detecting changes in the environment. The lag time of using body core temperature-sensitive neurons would be too large to perform effective regulation. Therefore, if the body's core temperature falls within the thermoneutral zone (TNZ), peripheral inputs play the most significant role in thermoregulation. Inside the TNZ, body temperature regulation is accomplished only through the control of sensible heat loss [23] and therefore involves only autonomic thermoregulatory mechanisms. Anticipating this line of reasoning, Marks and Gonzales [24] predicted "that pleasantness and unpleasantness of thermal stimuli depend on the temperature of the skin before stimulation - which itself reflects environmental conditions - given constant internal body temperatures". Only after 40 years, Parkinson and De Dear [25] formalised this concept as "spatial alliesthesia", where the term alliesthesia, first introduced by Cabanac, is "the property of a given stimulus to arouse pleasure or displeasure according to the internal state of the subject" [26]. In spatial alliesthesia in particular, the perceptual changes are detected by cutaneous thermoreceptors, not the body core, which drive pleasure sensations. This notion becomes more relevant when considering that thermal behaviour is driven by thermal comfort [27] and is regarded as the primary influencing factor in body temperature homeostasis [28]. Also, it is essential to notice that the indoor environment's transient conditions are commonly within the TNZ, where the influence of thermal behaviour is omitted. Kingma et al. [29] analysed the relationship between the TNZ and the thermal comfort zone (TCZ). They concluded that the ambient temperature associated with the thermoneutral zone is greater than that of thermal comfort. This finding implies that thermal behaviour could be initiated even before the thermoneutral zone boundaries are reached. In terms of spatial alliesthesia, negative alliesthesia (i.e., thermal displeasure) can be viewed as thermal discomfort [25], which in turn prompts human beings to counter the thermal environment accordingly.

Following this logic, Vellei and Le Dréau [30] proposed a modified version of Fanger's predicted percentage of dissatisfied (PPD) index that considers both a static and a dynamic component. The former is based on thermal sensation derived from the predicted mean vote (PMV), while the latter includes thermal alliesthesia and thermal habituation/adaptation. Utilising the data from Zhang's experiment on cyclical temperature variations induced by demand response events [16], the authors showed the impact of these psycho-physiological phenomena on dynamic thermal perception.

1.3. Research aims

Despite the existence of previous studies on temperature cycles, drifts and ramps, their inconsistent results limit the knowledge of dynamic thermal comfort limits. Regarding the processes driving dynamic thermal perception in temperature cycles, the previously mentioned study by Vellei and Le Dréau is noteworthy. However, the dynamics of temperature cycles differ from the dynamics of temperature drifts and ramps. The latter, being monotonic changes, do not have the same stimulus repeated over time. Furthermore, this study has some potential issues related to the use of different scales to assess satisfaction. In Zhang's experiment, the percentage of dissatisfied is calculated from actual observed data, measured using a binary acceptability scale. In contrast, Fanger's PPD index is inferred from the 7-point ASHRAE thermal sensation scale (assumed to be ≥ 2 or ≤ -2 ; see page 130 of [31]). Therefore, there is a problem with the semantic equivalence of these scales. In truth, this is a problem that extends to other psychometric scales (e.g., thermal comfort and thermal preference), and that the thermal comfort research community has yet to address adequately.

The present research is an exploratory study whose goal is to understand human reaction to monotonic thermal variations by describing the relationship between their response and the covariates of interest. Therefore, the emphasis of this work is to derive some insight into the relationships that exist rather than to test hypotheses that certain relationships hold. This is achieved through a laboratory experiment with "office-like subjects", simulating office settings in a ramp-induced thermal environment. In this configuration, the relationship between environmental and demographic factors (with their potential interactions) to participants' thermal discomfort event was analysed. We considered the actual thermal behaviour as the thermal comfort limit, that is, the action prompt from the discomfort event. By doing so, we avoid the issue of semantic equivalence between different psychometric scales. Nevertheless, participants' perception, evaluation, preference and acceptability of the environment were collected.

2. Methods

2.1. Participants

Participants were recruited from the university campus with a targeted age between 20 and 67. A summary of the main demographic and anthropometric characteristics of the subjects is listed in Table 2. Participation in the experiment was voluntary, and participants were informed about the possibility of withdrawing their consent at any time, without giving a reason in agreement with the principles and instructions of the European General Data Protection Regulation (GDPR). A printed information letter was distributed, and the participants signed a consent form prior to participation. The letter included information on data protection measures and general information about questionnaires and measurements. However, it did not inform the subjects about specific changes in environmental variables, such as changes in temperature. To comply with the GDPR, the experiment description was submitted to the Norwegian Centre for Research Data (NSD) and approved with reference code 525790.

2.2. Experimental set-up

The experiment was conducted in the ZEB Test Cell Laboratory on the Norwegian University of Science and Technology (NTNU) premises (Trondheim campus) between September 2019 and January 2020 (see Appendix A for a summary of the outdoor climatic conditions). Two identical climatic chambers (2.4 m \times 4.2 m x 3.3 m in height, surrounded by two guard rooms kept at 22 °C) (Fig. 1), furnished like a typical single office, were used to recreate a change in the environment induced by thermal ramps. Space heating and cooling were provided from a constant air-volume system that supplied 100% fresh air from outside, distributed by a 2 m long perforated fabric tube installed at the ceiling. The temperature of the supplied air was controlled through a PID controller (implemented in LabVIEW) utilising a Class A Pt100 temperature sensor located in the extraction air duct. Chamber's walls, ceiling, and floor consist of prefabricated sandwich panels with a low thermal mass; therefore, the surface temperatures almost instantly follow the air temperature. The climatic chambers were illuminated with office pendant and task lighting, as well as natural lighting through a south-facing window with a window-to-wall ratio of 0.56. The shading configuration was composed of 13 louvres tilted at 15° mounted on the external side of the window. Further details on the facility's experimental equipment, as well as the properties of the ZEB laboratory, can be found in Goia et al. [32].

During the experiments, the indoor environment was monitored by measuring air temperatures (at 0.10, 0.60 and 1.10 m), surface temperatures (five on the two side walls, three on the floor and the ceiling, four on the window and one above the door), globe temperature (at 1.70 m), relative humidity (at 1.75 m), airspeed (at 0.10, 0.60, 1.10 and 1.70 m), CO₂ concentration (at 1.75 m), horizontal and vertical illuminance (on the work-plane and at eye levels, respectively) every minute throughout every session. In addition, a weather station installed in proximity to the southern façade of the ZEB Test Cell measured ambient air temperature, relative humidity, wind speed and direction, global solar irradiance on the horizontal plane and precipitation in 10-min intervals. The accuracy of the sensors used, both for indoor and outdoor measurements, are shown in Table 3.

2.3. Experimental conditions and procedure

The operative temperature set-point of 22.0 \pm 1.0 °C was defined in accordance with the thermal comfort limit for winter according to Category A of ISO 7730-2005 [3]. Both space heating and cooling variations were tested within winter conditions. The rates of temperature changes were derived from the limit in ASHRAE 55-2017 (Table 1) [2]. Given the limit of 3 h for each experimental session (Fig. 2) and compatible with a typical office occupancy schedule, only the following thermal ramps were implemented: (i) \pm 4.4 K/h, (ii) \pm 3.4 K/h, (iii) \pm 2.2 K/h and (iv) \pm 1.4 K/h.

The study's design was a randomised crossover trial; a longitudinal study in which participants received a randomised sequence of different exposure (i.e., thermal ramps). The schematic of the experimental session, illustrated in Fig. 2, was composed of seven and a half hours with a half-hour lunch break included (as a typical standard Norwegian workday). To increase participation, the day could be split into half days, meaning one morning session (8:00–11:30) and one afternoon session (12:00–15:30). However, participants were required to attend an even number of morning and afternoon sessions. Subjects could choose to join the experiment for two or four days and were offered compensation, upon completion of the agreed days, of 200 or 600 NOK, respectively. In addition, a lottery was set up: one lucky participant, selected from among those who successfully completed the agreed-upon days, received an Apple iPad.

After arrival, participants were asked to take a seat at the workplace assigned beforehand by the researchers. At this time, they were asked to fill out a first questionnaire consisting of questions related to demographic and anthropometric characteristics, current clothing level and satisfaction with the workplace (q1 in Fig. 2). During the first 30

Table 2

Demographic and anthropometric characteristics of participants.

Gender	Number	Age (year)	Height (cm)	Weight (kg)	BMI (kg/m ²)
		Median (IQR*)	Median (IQR*)	Median (IQR*)	Median (IQR*)
Male	9	28.0 (30.0-25.0)	174.0 (184.0–170.0)	70.0 (85.0-67.0)	24.2 (26.3-22.1)
Female	29	26.0 (31.0-22.0)	170.0 (172.0-165.0)	63.0 (70.0-53.0)	21.6 (23.4-20.7)
Total	38	26.5 (30.8-23.0)	170.5 (173.0–165.0)	65.0 (70.0–58.3)	21.8 (24.2 20.8)

*IQR is the interquartile range, that is, the difference between upper and lower quartiles.



Fig. 1. Floor plan of the facility.

Table 3

Characteristics of the sensors for the measurement of indoor and outdoor conditions.

Physical variable	Type of sensor	Accuracy
Indoor		
Air temperature	Pt100	±0.3 °C
Surface temperature	T-type thermocouple	±0.5 °C
Globe temperature	Pt100	±0.3 °C
Relative humidity	Capacitive	±5%
Airspeed	Hot wire	$0 \div 0.1 \text{ m/s} = \text{NA}$
		$0.1 \div 0.5 \text{ m/s} = \pm 0.083 \text{ m/s}$
		$0.5 \div 1 \text{ m/s} = \pm (0.05 + 0.05)$
		va*) m/s
		$>1 \text{ m/s} = \pm (0.1 + 0.05 \text{ va*})$
		m/s
CO2 concentration	Non-dispersive infrared	$\pm 70 \text{ ppm} + 5 \%_{\text{measured}}$
Horizontal illuminance	Photodiode	$\pm 5\%$
Vertical illuminance	Photodiode	$\pm 3\%$
Outdoor		
Air temperature	Pt100**	±0.1 °C
Relative humidity	Capacitive**	$\pm 1.5\%$
Wind speed	N.32 step optoelectronic	$0 \div 3 \text{ m/s} = 1.5\%$
	disk	>3 m/s = 1%
Wind direction	See above	1%
Global solar irradiance	Thermopile pyranometer	10%
Precipitation	Tipping bucket***	$0\div 20~mm/h=\pm~0.2~mm$
		$20 \div 240 \text{ mm/h} = 1\%$

*va is mean airspeed.

**Thermohygrometer with multi-plate natural ventilation radiant screen.

***Rain gauge equipped with heater and siphon.

min, participants acclimatised to the constant set-point temperature and were free to adjust their clothing ensemble. After this period, the experimental session started. At this time, the subjects were instructed to report the final clothing level (i.e., if any change in the initial clothing level occurred during the acclimation period) and to maintain the adopted garment level throughout the experimental session. Furthermore, participants were not allowed to interact with the environment (e.g., open the window/door, regulate the thermostat). However, due to the long sessions, participants were allowed to stand up and move around the climate chamber, leave it for a short period (to visit the restroom), and consume refreshments. No specific tasks or tests were carried out during the experiment, and participants were asked to carry out their typical office activity. This contributed to the simulation of a typical office activity pattern. Nevertheless, subjects had to fill out computer-based questionnaires at different scheduled intervals (q2 in Fig. 2). By means of graphic categorical scales, these questionnaires were used to assess perception, evaluation, preference, and acceptability of the thermal, visual, acoustic and air quality of the environment. These scales, derived from the standard ISO 10551-2019 [33], are shown in Appendix B.

During the experimental session, the participants were instructed to press a digital button (available on a dedicated laptop situated on the desk, see Fig. 3) as soon as they felt uncomfortable. Here uncomfortable was defined as the decision to "take action to restore a comfort condition" (e.g., if the environment is too warm, then regulate the thermostat or open the window). It is important to point out that participants could press the button for any source of discomfort related to the indoor environment (e.g., stuffy air, noise from the ventilation system, lack of daylight) and not only for temperature-related discomfort. After pressing the digital button, a computer-based questionnaire appeared on the dedicated laptop (q3 in Fig. 2). This questionnaire was used to assess the environment (in the same manner as q2) and record the source(s) of discomfort through multiple-choice answers (shown in Appendix C). Participants were also requested to rank, from 1 to 3 (with one being the most important), the strategies (among a predefined set of listed options) that they would use to restore comfort. These strategies varied from simple actions (such as adding/removing a clothing layer and opening/closing the window) to more complex ones (such as adjusting the cooling/heating set-point temperature and plugging-in a local/personal cooler/heater).

The thermal ramp was interrupted when one of the two following conditions was met: (i) the session ended (i.e., at 11:30 and 15:30); (ii)

M. Favero et al.

Building and Environment 204 (2021) 108144



Fig. 2. Schematic of the experimental procedure (bottom) with an example of a possible scenario (top).



Fig. 3. View of the workstation in one of the two single offices.

the participant pressed the digital button. In the latter case, the thermal ramp was stopped only if the discomfort event was related to the temperature level, that is, the participant selected "temperature too high" or "temperature too low". At the end of every session, subjects were asked to fill out a questionnaire (q4 in Fig. 2) about their satisfaction with the workplace as a whole, expressed on a Likert scale.

3. Statistical analysis

Environmental, demographic and anthropometric data were studied using a survival analysis. Survival analysis comprises a family of methods that examine and model the time it takes for events to occur. However, its goal is not limited to investigating the effects on the time until the event occurs, but also to evaluate the relationship of survival time to covariates. Covariates (often referred to interchangeably as predictors or independent/explanatory variables) assess the impact of certain features on the dependent variable.

The prototype event is death – hence the name "survival analysis" and much of its terminology – but the range of applications of survival analysis is much broader. For example, the same methods are known as "failure-time analysis" in engineering and "event-history analysis" in sociology.

3.1. Survival analysis

t

In survival analysis, there are two crucial quantities that need to be introduced, namely the survivor function, denoted by S(t), and the hazard function, denoted by $\lambda(t)$. Let T be a non-negative random variable representing the waiting time until the occurrence of an event. The survival function S(t) can be written as the probability that the random variable T is larger than a specified time t, that is

$$S(t) = \Pr(T \ge t)$$
 Eq. (1)

More generally, it is the probability that the event of interest has not occurred by duration *t*.

An alternative characterisation of the distribution of *T* is given by the hazard function, defined as

$$\lambda(t) = \lim_{dt \to 0} \frac{\Pr(t \le T < t + dt | T \ge t)}{dt}$$
 Eq. (2)

which gives the instantaneous rate of occurrence of the event at time t, given survival up to time t.

The two functions in Eq. (1) and Eq. (2) express, in essence, opposing concepts: while the survivor function focuses on surviving, the hazard function focuses on failing, given survival up to a certain point in time. Moreover, there is a clear relationship between these two quantities. Knowing the form of S(t), the corresponding $\lambda(t)$ can be derived, and vice versa. More generally, this relationship can be expressed equivalently in either of the two formulae:

$$S(t) = \exp\left(-\int_{0}^{t} \lambda(u)du\right)$$
 Eq. (3)

$$\lambda(t) = -\frac{d}{dt} \log S(t)$$
 Eq. (4)

The integral in the round brackets in Eq. (3) is called the cumulative hazard (or cumulative risk) and is denoted as

$$\Lambda(t) = \int_{0}^{t} \lambda(u) du \qquad \qquad \text{Eq. (5)}$$

Furthermore, censoring and its assumptions need to be mentioned as well. Censoring is a form of missing data problem in which the time-to-event is not observed. Therefore, there is only partial information about individual survival time. There are three different types of censoring, as graphically illustrated in Fig. 4:

- Left-censored: the event occurs between t_{start} and t_3 , but the exact time is unknown.
- Interval-censored: the event occurs within t₁ and t₄, a specified time interval, but the exact time is unknown.
- Right-censored: the event does not occur before the end of the study, $t_{\rm end}.$

There are three assumptions about censoring for survival data: independent censoring, random censoring, and non-informative censoring. These assumptions, even though they have similarities, are different and should not be used interchangeably. Among the three, independent censoring is the most relevant since it affects validity.¹ Many of the analytical techniques discussed in the next paragraph rely on this assumption for valid inference in the presence of right-censored data. For mathematical definitions of these three assumptions, the reader is referred to Kalbfleisch and Prentice [34] and Klein and Moeschberger [35], and for more intuitive definitions and examples to Building and Environment 204 (2021) 108144

Kleinbaum and Klein [36].

As mentioned before, survival analysis is the name for a collection of statistical techniques. These techniques can be summarised into three categories: (i) non-parametric models, (ii) parametric models, and (iii) semi-parametric models. The main difference between the three categories is whether the outcome, namely the survival time, is assumed to follow a specific distribution. Non-parametric methods are used when no theoretical distribution adequately fits the data; therefore, they are distribution free. The Kaplan-Meier method is an example from this category. Conversely, in the parametric model, the underlying distribution of the outcome is specified. Typical examples of parametric models in a regression-type framework are linear regression, logistic regression, and Poisson regression. The outcome is assumed to follow some distribution with these models, such as the normal, binomial, or Poisson distribution. For survival analysis, several parametric distributions can be used to describe time to event data, such as exponential, Weibull and log-normal distribution, each of which is defined by a different hazard function. Semi-parametric models are a combination of the two previously mentioned categories. Even if the regression parameters (the betas) are known in these models, the outcome's distribution remains unknown. The Cox proportional hazards (PH) model belongs to this category.

In this investigation, since the outcome distribution (i.e., the survival time distribution) is unknown, non-parametric and semi-parametric models were utilised, more specifically, the Kaplan-Meier method and Cox regression. The former has been used in this study only to describe and visualise the survival curves at a preliminary stage, while the latter evaluates the relationship of survival time to covariates.

3.1.1. Kaplan-Meier method

The Kaplan-Meier (KM) estimator of a survival function at time t, $\hat{S}(t)$, is given by [37].

$$\widehat{S}(t) = \prod_{i: t_i \leq t} \left(1 - \frac{d_i}{n_i} \right)$$
 Eq. (6)

where d_i is the number of events at time t_i and n_i is the number at risk at time t_i . This method is based on individual survival times and assumes independence between censoring and survival, that is, the reason an observation is censored is unrelated to the cause of failure. From Eq. (6), it can be seen that Kaplan-Meier requires a minimal feature set. Kaplan-Meier only needs the time when the event (or censorship) occurred and the duration between the onset and the event. Also, as mentioned before, it is distribution-free. However, it cannot estimate the magnitude of the survival-predictor relationship of interest, nor control for multiple covariates. Therefore, it has been used only to describe and visualise the survival curves at a preliminary stage.

3.1.2. Cox proportional hazards model

The Cox proportional hazards (PH) model is the most used procedure for modelling covariates' relationship to survival or other censored outcomes. It is mainly popular because it does not require any assumptions about the shape of the hazard function (that is, the specific way that risk changes over time); however, it allows for estimating the regression coefficients.

The Cox PH model is usually written in terms of the hazard model formula

$$\lambda(t) = \lambda_0(t)e^{X\beta}$$
 Eq. (7)

where $\lambda_0(t)$, is an unspecified non-negative function of time called the baseline hazard, while $e^{X\beta}$, is the time-independent exponential expression that involves the covariates *X*.

A fundamental assumption of the Cox model is proportional hazards, which implies that the hazard ratio for any two subjects i and j is constant over time.

¹ Validity is meant as lack of bias. The presence of non-independent censoring will result in a biased estimated effect.





If this assumption holds, each covariate's effect can be summarised with a single number. Since in practice this assumption is never 100% confirmed (for example, this is the case for any Cox model that include time-dependent variables), there are various strategies to deal with this. Models that rely upon this strategy are called "extended Cox models" and can be generally written as

$$\lambda(t) = \lambda_0(t) e^{X(t)\beta}$$
 Eq. (9)

$$\lambda(t) = \lambda_0(t)e^{X\beta(t)}$$
 Eq. (10)

where Eq. (9) is a time-dependent covariate, and Eq. (10) has a time-dependent coefficient. Note that the PH assumption presumes that the coefficient does not change over time: $\beta(t) = c$.

In the literature, there is another approach called the "stratified Cox procedure", in which the variable that does not meet the PH assumption is stratified. Stratification is suitable only for categorical variables and implies different baselines for each level of the variable being stratified. It can be written as

where *g* denotes the levels of the variable. Note, however, that the stratified variable is not included in the model, and it is not possible to obtain a hazard ratio value for the stratified variable adjusted for the other variables. Nevertheless, the same coefficients (the β s) are assumed for each level of the stratified variable.

The Cox model relies upon other assumptions that need to be verified, which derive from the fact that this model is a regression-type model. These assumptions state that the relationship between the covariate and the response (the logarithm of the hazard in this case) is additive and linear. The former means that the effect of changes in a covariate X_k on $\log \lambda(t)$ is independent of the values of the other covariates, while the latter states that the change in the $\log \lambda(t)$ due to a one-unit change in X_k is constant, regardless of the value of X_k . For further detail, the reader is referred to Refs. [36,38].

3.1.3. Data preparation and analysis

Data gathered during the acclimation² period were excluded from the data analysis. The mean radiant temperature (MRT) was calculated according to ISO 7726-1998 [39] based on the surrounding surfaces' measured temperature and the angular factor computed for a seated person in the specific climate chamber. Following the aforementioned standard, the calculated MRT was used, combined with the measured air temperature and air velocity, to calculate the operative temperature. Due to a technical problem with the air conditioning during space cooling processes, data from two female participants were excluded from the analysis. This led to a difference in the female sample size between space heating and cooling processes, from 29 to 27, respectively.

All statistical analyses were performed using R [40] with the RStudio integrated development environment (RStudio Inc., Boston, MA, USA). Survival analyses, using both the Kaplan-Meier method and Cox regression, were performed with the *survival* package [41] and the respective graphs were created with the *ggplot2* package [42] and the *survniner* package [43]. The significance level for all analyses was set at 0.05.

4. Results

The results are grouped according to (i) general results and observations, (ii) descriptive analysis from the KM method, and (iii) modelling step and results obtained from the extended Cox model.

4.1. General observations

A total of 314 thermal ramps were performed, which led to 223 thermal discomfort events. Specifically, 104 thermal discomfort events occurred during heating processes (with 155 thermal ramps), while 119 thermal discomfort events occurred during cooling processes (with 159 thermal ramps). Table 4 summarises the results for the different thermal ramps.

Fig. 5 presents a time course of the discomfort events during exposure to the different thermal ramps for both the space heating and cooling processes. In this figure, the right-censored observations are also represented (dots without the black outline). Right-censored

² Acclimation and acclimatisation, although etymologically indistinguishable, define two distinct processes. The former describes "adaptive changes that occur within an organism in response to experimentally induced changes in particular climatic factors" (e.g., the ambient temperature in a controlled environment). The latter denotes "adaptive changes that occur within an organism in response to changes in the natural climate" [23].

M. Favero et al.

Table 4

Number of therma	l discomfort	events	for	each	ramp	and	each	process.
------------------	--------------	--------	-----	------	------	-----	------	----------

Ramp description	Total number of thermal ramps	Thermal ramps with a thermal discomfort event
Heating		
$3.4 \text{ K/h} < \text{ramp} \leq 4.4 \text{ K/h}$	41	25
$2.2 \text{ K/h} < \text{ramp} \leq 3.4 \text{ K/h}$	36	28
$1.4 \text{ K/h} < \text{ramp} \leq 2.2 \text{ K/h}$	35	26
$0.0 \text{ K/h} < ramp \leq 1.4 \text{ K/h}$	43	25
Cooling		
$0.0 \text{ K/h} > \text{ramp} \ge -1.4 \text{ K/h}$	46	36
$-1.4 \text{ K/h} > \text{ramp} \ge -2.2 \text{ K/h}$	40	34
$-2.2 \text{ K/h} > \text{ramp} \ge$ -3.4 K/h	33	24
$-3.4 \text{ K/h} > \text{ramp} \ge$ -4.4 K/h	40	25
Total	314	223

observations were observed when the experimental session was interrupted because the time available for the session was over – condition (i) in section 2.3. For ease of interpretation, the ASHRAE 55-2017 [2] comfort limit (dark grey X-shaped cross) and a fitted line between this limit (grey dashed line) are also plotted in Fig. 5. It can be clearly seen that the thermal discomfort events are not symmetrical. Participants were more sensitive to a cold variation than a warm one. In fact, 83% of the discomfort events for cold are within the ASHRAE comfort limit, while on the warm side, only 30% are within the comfort limit. An overview of participants' assessment of perception, evaluation, preference, and acceptability of the thermal environment during the discomfort event is presented in Fig. 6. In this figure, participants' votes on the four previously mentioned psychometric scales are divided between heating and cooling mode. Particularly:

- a) Thermal sensation: Discomfort events are not symmetric. During space heating, thermal behaviours were undertaken mostly when the environment was sensed as "warm" (+2) with ΔT up to 5 K. On the other hand, during space cooling, actions were undertaken when the environment was perceived as "slightly cool" (-1) and "cool" (-2). Here the same range of operative temperature change (-3 K) was perceived differently.
- b) Thermal comfort: The distribution of discomfort events for space heating and cooling is remarkably similar. Most of the thermal behaviours were undertaken when the environment was judged to be "slightly uncomfortable" (+1) or "uncomfortable" (+2) for both space heating and cooling processes. This suggests that, indeed, thermal comfort is the driver for thermal behaviour.
- c) Thermal preference: Most of the actions were undertaken with a thermal preference vote different from "without change" (0). Reasonably, a participant would initiate a thermal behaviour out of a desire for a higher or lower temperature.
- d) Thermal acceptability: For both space heating and cooling processes, discomfort events follow a skewed distribution, specifically a negative skew (or left-skewed) for acceptable environments and a positive skew (or right-skewed) for unacceptable ones. Consequently, most of the actions were undertaken at the boundary between an acceptable and unacceptable environment.



* ASHARE comfort limit for drifts and ramps

-- Fitted line between ASHRAE comfort limit for drifts and ramps

Fig. 5. Thermal ramps endpoint.

M. Favero et al.



Fig. 6. Psychometric scales for thermal discomfort events. Please note that the data shown here represent the right-here right-now votes on the questionnaire at the moment of the thermal discomfort event (i.e., when the digital button was pressed).



Fig. 7. KM (a) and log-log (b) survival curves for different rates of temperature change.

4.2. KM survival curves

As mentioned in section 3.1.1, the KM method has the advantage of being distribution-free, but, at the same time, it cannot estimate the magnitude of the survival predictor relationship of interest nor control for multiple covariates. Therefore, this method has been used only to describe and visualise the survival curves at a preliminary stage. Fig. 7.a shows the KM curves for the various thermal ramps, where the plus symbol represents the right censoring. In this figure, it is noticeable that the survivability for warm variations was higher than for cold ones. Also, for both space heating and cooling processes, slower variations led to longer survival than faster variations.

In Fig. 7.b, the eight thermal ramps are plotted on a log-log survival scale against time on the log scale. This plot, usually referred to as a log-log plot, is a graphical approach to evaluating the PH assumptions. If the hazards cross or are not parallel in some other way, the PH assumptions for the predictor of interest are not met. In this specific case, since the rates of temperature change for heating and cooling processes intersect, the PH assumptions for this predictor are not satisfied. On the other hand, when considering space heating and cooling separately (plot not shown), the curves for the different rates of change are roughly parallel. However, this is a necessary condition but not a sufficient condition. In fact, even if the hazards do not cross, it is still possible that the PH assumption is not met. Thus, checking for crossing hazards is not

sufficient, and other approaches to evaluate the reasonableness of the PH assumption must be used.

In Fig. 8, the log-log plot has been drawn with each slope (in absolute value) plotted separately to increase readability. From this plot, it can be noticed that there is some indication of non-parallelism after 70 min for slope 3.4 K/h (Fig. 8c) and before 15 min for slope 2.2 K/h (Fig. 8b). Also, the initial distance between the curves for space heating and cooling processes is greater for a ramp slope of 1.4 K/h than a ramp slope of 4.4 K/h, indicating an effect between the temperature change and the direction of the change (i.e., increase or decrease of the temperature). Moreover, on the whole, all the curves show a divergent-convergent shape; that is the curves initially separate but eventually join up.

In the context of monotonic temperature variations (thermal ramps), warm changes induce thermal discomfort with some delay compared to cold ones, but this delay progressively wears off. The underlying process, that is, the discomfort from thermal ramps, is delayed on the warm side, or stated analogously, the survival is prolonged temporarily. However, it is important to point out that the number of participants still at risk decreased towards the curve's end. Therefore, caution is generally required not to over-interpret the right side of this part of the plot.

4.3. Cox-regression

The descriptive analysis carried out in the previous section showed



Heating process — Cooling process

Fig. 8. Log-log survival chart for heating and cooling based on the rate of temperature changes.

Table 5

List of covariates used in the model for both space heating and cooling processes.

Variable	Code	Туре	Unit
Thermal resistance of clothing	Clothing	Continuous, time- independent	clo
Gender	Gender	Categorical, time- independent	Female (reference)/Male
Age	Age	Continuous, time- independent	Years
Body Mass Index	BMI	Continuous, time- independent	kg/m ²
Time lived in	Time.	Categorical, time-	Less than or equal to 3
Norway	Norway	independent	years (reference)/More than 3 years
Air velocity	Air.vel	Continuous, time- dependent	m/s
Time of day	Time.day	Categorical, time- independent	Morning (reference)/ Afternoon
Vapour pressure	Vap.pre	Continuous, time- dependent	N/m ²
Operative temperature change	Top.delta	Continuous, time- dependent	К
Initial operative temperature	Top.start	Continuous, time- independent	°C
Participant ID- code	ID.subj	Categorical, time- independent	-

that if heating and cooling are considered in the same model, the PH assumption is not met. Even though there are methods to deal with this (as mentioned in section 3.1.2), it was decided to develop separate models for the space heating and cooling processes. This choice also had the advantage of assessing the selected covariates' significant predictors separately for the two models. Table 5 lists all the covariates used in the inference of the heating and cooling models.

The rate of temperature change (i.e., ± 4.4 , ± 3.4 , ± 2.2 and ± 1.4 K/h) was only considered in the descriptive analysis (Figs. 7 and 8) and not incorporated directly into the Cox-regression model. In this model, the rate of temperature change (K/h) is indirectly implied in the operative temperature change (K), a time-dependent covariate.

The variation of the operative temperature (*Top.delta*) and the initial operative temperature (*Top.start*) are the decomposition of the operative temperature. *Top.start* is defined as the operative temperature at time t = 0. In contrast, *Top.delta* is the difference between the operative temperature at t > 0 and t = 0. This division aims to verify whether the operature temperature level affects dynamic thermal disconfort.

The covariate *ID.subj* was used to account for correlated observations since the same subject appears in overlapping intervals. This variable was used in the analysis to create a robust variance, allowing the computation of an infinitesimal jackknife variance estimate [38].

The following modelling steps were undertaken:

 Purposeful selection of covariates: After performing a first fit of the initial multivariable model, the p values of the individual coefficient were used to ascertain covariates that might be deleted from the model. This procedure is commonly known as backwards eliminations. The reduced model was evaluated to check if the elimination of a covariate produced a "relevant" change in the parameter estimates of the model's remaining variables. A change of about 20% was used as an indicator of the relevant change. If an important confounder was removed, it was incorporated back into the model.

- 2. Define the correct functional form (i.e., test the linearity assumption): With the previous model, the scale of the continuous variable was analysed to determine whether or not the effect of the covariates was linear in the log hazard (and therefore check if the data support this initial hypothesis). In this analysis, smoothing splines³ were utilised for this purpose.
- 3. Check for interaction terms (i.e., test the additivity assumptions): In this step, it was determined whether interactions between predictors needed to be added to the model. Each individual interaction was introduced separately and assessed by comparing the model with the interaction term to the main effect model. This assessment was carried out by examining any changes in the main effect's coefficients and checking the partial likelihood ratio test. All significant interactions were added jointly to the main effects model.
- 4. Check the PH assumption: In this step, the model was carefully evaluated by performing model diagnostics. Also, to avoid overfitting, the number of variables that can be included in the model should be confined. It is not trivial to make a general statement about this, but an approximate criterion is to have one covariate per ten events [44].

4.3.1. Initial models

In this section, both the initial multivariable models for heating and cooling are presented. In this step, the backwards elimination has not been yet applied.

At this point in the analysis, four main significant predictors had been detected for space heating while only two had been detected for space cooling (Table 6). Between the two models, the only common significant predictor is the operative temperature variation. As expected, its coefficient is positive for heating processes and negative for cooling processes. The cooling coefficient is greater than that for heating in absolute value, suggesting that cooling variations are more threatening to thermal comfort. It is important to remember that this coefficient represents the overall effect of the corresponding time-dependent variable, considering all times at which this variable has been measured in the study. Also, at this point, the linearity assumption between the risk and the covariate had yet to be verified.

In the following two sections, only the main results of applying the modelling steps mentioned above are illustrated.

4.3.2. Space heating process

Table 7 summarises the results of the multivariable model for heating after applying backwards elimination. Four significant predictors were identified – BMI, time lived in Norway, operative temperature variation and initial operative temperature – all positively associated with increased risk of "warm discomfort". Among them, three are continuous variables (and will be discussed later), while *Time.Norway* is categorical. This variable has been used as a proxy for inferring a long-term adaptation⁴ to the Norwegian environment. In a recent study, Luo et al. [45] investigated the long-term thermal adaptation of building occupants by conducting two comparative field studies on thermal comfort in China. They observed for some years two groups of people, one that moved

³ Splines are mathematical constructs made up of polynomial functions joined together to form a smooth curve, where the joining points are called "knots". An effective way to find a smoothing spline in survival analysis is with "penalised partial likelihood". When this quantity is maximized, it balances the goodness of fit against complexity [38].

⁴ According to the glossary of terms for thermal physiology [23], adaptation is defined as "changes that reduce the physiological strain produced by stressful components of the total environment". It includes both genotypic (genetic selection) and phenotypic adaptation (changes that may occur within the lifetime of an organism, such as changes in the thermoregulatory system).

M. Favero et al.

Table 6

Regression coefficients for the predictors in the initial multivariable model (before applying back	ds elimination).
---	------------------

8	*				11 5 6				
Predictor		Heating proc	ess			Cooling proc	ess		
		coeff	se (coeff)	Z	<i>p</i> -value	coeff	se (coeff)	Z	p-value
Clothing		0.693	0.677	1.023	.306	-1.538	1.443	-1.066	.287
Gender	female	Reference				Reference			
	male	-0.507	0.459	-1.103	.270	-0.642	0.439	-1.462	.144
Age		-0.006	0.026	-0.242	.809	0.016	0.019	0.834	.404
BMI		0.174	0.067	2.600	.009*	0.009	0.068	0.128	.898
Time.Norway	\leq 3 years	Reference				Reference			
	>3 years	1.142	0.325	3.511	<.001*	0.502	0.337	1.491	.136
Air.vel		3.098	4.934	0.628	.530	-4.470	6.048	-0.739	.460
Time.day	morning	Reference				Reference			
	afternoon	0.004	0.233	0.018	.986	-0.668	0.260	-2.572	.010*
Vap.pre		-0.001	0.001	-0.626	.531	0.000	0.001	-0.537	.591
Top.delta		0.784	0.110	7.146	<.001*	-0.894	0.203	-4.400	<.001*
Top.start		0.906	0.333	2.721	.007*	0.340	0.294	1.157	.247
		Likelihood ra	atio test = 98.35	on 10 df, $p \leq 2$.2E-16	Likelihood r	atio test = 56.92	on 10 df, $p = 1$.378E-08

* indicates a significant term.

Table 7

Regression coefficients for predictors in the multivariable heating model (after applying backwards elimination).

Predictor		Heating process			
		coeff	se (coeff)	z	p-value
Gender	female	Reference			
	male	-0.564	0.470	-1.200	.230
BMI		0.168	0.068	2.476	.013*
Time.Norway	\leq 3 years	Reference			
	>3 years	1.067	0.316	3.373	<.001*
Top.delta		0.756	0.101	7.476	<.001*
Top.start		0.871	0.259	3.359	<.001*
Likelihood ratio	test = 96.78 o	n 5 df, $p \le 2.2$	2E-16		

*indicates a significant term.

from southern China (Shanghai) to northern China (Beijing) and one that moved in the opposite direction. The authors concluded that thermal adaptation exhibits asymmetric trajectories: the southern origin groups accepted neutral and warm indoor temperatures in less than one year, while the northern origin groups took three years to adjust to colder indoor temperatures. Based on this result, it was assumed that participants who had lived in Norway for more than three years had adapted to different indoor temperatures and heating/cooling strategies. The estimated hazard ratio (HR) is $\exp(1.067) = 2.907$ for participants living in Norway for more than 3 years (95% CI [1.564, 5.403], p < .001). Therefore, individuals who had lived in Norway for more than three years were, at any given time during this study, 2.907 times as likely to experience "warm discomfort" as those who had lived in Norway for less than three years. In other words, they had an increased risk of 190%.

It can be noticed that the covariate *Gender*, even though not statistically significant, still remains in the model. This because its elimination caused a relevant change in the BMI variable. Therefore, in this study, gender is a confounder for BMI. This can be explained by looking at the participants' anthropometric characteristics in Table 2. Female participants were, generally, shorter and lighter than their male counterparts.

The next step is to verify whether the linearity assumption for continuous variables in the model has been met. Initially, for each continuous variable, a smoothing spline with four degrees of freedom was fitted, and the resulting plot was checked for significant nonlinearity. If non-linearity was detected, the correct functional form was derived by refitting a smoothing spline but with an "optimal" degree of freedom based on the Akaike information criterion (AIC). Otherwise, a linear relationship was assumed. It is important to mention that this flexibility comes at a price. The interpretation of the estimated coefficients resulting from splines is, in fact, meaningless. However, the linear combinations of these coefficients can be used to obtain predicted values that can be plotted and interpreted. Fig. 9 shows this analysis for BMI, operative temperature change, and initial operative temperature.

BMI, formerly called the Quetelet index, is a measure for indicating nutritional status in adults. For adults over 20 years old, the World Health Organization has divided the BMI into: (i) "Underweight" if BMI $< 18.5 \text{ kg/m}^2$; (ii) "Normal weight" if 18.5 kg/m² \leq BMI \leq 24.9 kg/m²; (iii) "Pre-obesity" if 25.0 kg/m² \leq BMI \leq 29.9 kg/m²; (iv) "Obesity class I" if 30.0 kg/m² \leq BMI \leq 34.9 kg/m²; (iv) "Obesity class II" if 35.0 kg/ $m^2 \le BMI \le 39.9 \text{ kg/m}^2$; and (vi) "Obesity class III" if $BMI \ge 40 \text{ kg/m}^2$ [46]. Fig. 9.a shows that the hazard increases from low BMI to around "normal weight" BMI levels, where it flattens out and then rises slightly in the pre-obesity category, but not significantly. This indicates that participants with lower BMI values have a lower hazard of experiencing "warm discomfort" than participants with normal and pre-obesity BMI values. However, there is no significant difference between the normal and pre-obesity category. This result is not completely in line with the literature. While it is true that the underweight population (BMI < 18.5 kg/m²) is associated with a higher comfortable temperature, the overweight population (i.e., $BMI > 25.0 \text{ kg/m}^2$) is associated with a lower comfort temperature. For instance, Indraganti et al. [47]'s field investigation in India found this difference to be 0.7 K. However, BMI does not actually measure body fat nor the proportion of muscle-to-fat. Therefore, it is possible that some of the participants were incorrectly classified in the pre-obesity category. A smoothing spline with three degrees of freedom has been selected for the functional form (the purple line in Fig. 9a).

Concerning the initial operative temperature, Fig. 9.b shows that a linear fit is within the confidence interval; therefore, a linear relationship between the log(hazard) and the initial operative temperature is assumed (purple line). Interestingly, the hazard increases with a higher value of initial operative temperature even if these values are within 22.0 ± 1.0 °C, which are the comfort limits for Category A from ISO 7730-2005 [3]. This is in agreement with Ran's neuroscience experiment [22], discussed in section 1.1. Since heat-responding spinal neurons encode absolute temperature, higher initial operative temperature values for the same increment in temperature.

Fig. 9.c shows that the hazard increases linearly with the increment in operative temperature until about +4 K, where it flattens out. A smoothing spline with two degrees of freedom was selected for the functional form (the purple line in Fig. 9c). Nevertheless, conceptually, it is hard to believe that the hazard of thermal discomfort associated with a monotonous rise in operative temperature levels off as higher delta temperatures are reached. A more logical fit would be a continuation of the linear relationship before the +4 K increment (the green line



Fig. 9. Penalised spline fit of (a) BMI, (b) initial operative temperature and (c) operative temperature change for heating.

in Fig. 9c). A possible explanation for the hazard's flattening upon reaching higher delta temperatures is that different individuals have different frailty levels. More frail individuals are more likely to experience the discomfort event early. Consequently, over time, the "risk set" has an increasing proportion of less frail individuals, and the hazard flattens out. In addition, looking at Fig. 5, it can be noticed that around a +4 K increment, some participants do not experience the "warm discomfort event" before being censored.

No significant interactions have been found, and the PH assumption

for the time-independent variables has been met. It is important to remember that, implicitly, time-dependent predictors do not satisfy it. For these variables, the hazard ratio is a function of time. Consequently, the coefficient of the time-dependent variable represents the overall effect of that predictor, considering all times at which this predictor has been measured in the study (for more details, the reader is referred to section 3.1.2 of this article).

4.3.3. Space cooling process

Table 8 summarises the results of the multivariable model for space cooling processes after applying backwards elimination. Three significant predictors were identified - time lived in Norway, time of day and operative temperature variation - one more compared with the initial model (see Table 6). Time.day and Top.delta are negatively associated with an increased risk of "cold discomfort", while Time.Norway is positively associated with the same outcome. As with the heating model, Time.Norway is positively associated with an increased risk of discomfort, but this time the relationship is weaker (HR = 1.854, 95% CI [1.060, 3.241], p = .030). The Time.day predictor is a categorical variable used to distinguish between the morning (8:00-11:30) and afternoon (12:00-15:30) sessions' thermal ramps. It was used to account for the circadian rhythm's influence on the risk of discomfort induced by variation in operative temperature. A circadian rhythm is a natural, internal process that regulates the sleep-wake cycle. This system also modulates other physiological functions, such as the body's core temperature, with a periodic variability over the 24 h (with maximal values in the late afternoon and minimal in the early morning). During periods of decreasing core temperatures, the average skin temperature rises to promote heat loss, and the reverse occurs during periods of rising core temperatures [48]. A circadian rhythm of heat loss from the distal limbs has been observed in humans: skin temperature and blood flow rhythms in these regions show peaks in the late evening and minima in the morning [49,50]. Consequently, an individual is in a "heat gain" mode in the morning (a rise in core temperature) and in a "heat loss" mode in the evening (a decrease in core temperature). Previous studies by Fanger et al. [51,52] found that, although the mean skin and rectal temperatures were slightly higher in the evening than in the morning, subjects did not prefer a different ambient temperature. They concluded that the same thermal comfort conditions can be used independently of the time of day or night. In this study, the hazard ratio for the time until "cold discomfort" for morning versus afternoon was 0.597 (95% CI [0.417, 0.853], p = .005), showing a difference between the two parts of the day. However, this result does not necessarily disagree with Fanger's previous findings. A lower risk of "cold discomfort" during the afternoon than in the morning does not inevitably imply a preferred lower temperature. It only suggests that, at any time during this study, participants during the afternoon were 0.597 times as likely to have a "cold discomfort" as during morning (that is, they experienced a reduction in risk of 40%).

It can be noticed that the covariates *Clothing* and *Gender*, even though not statistically significant, are still maintained in the model. In this

Table 8

Regression coefficients for predictors in the multivariable cooling model (after applying backwards elimination).

Predictor		Cooling pro	ocess		
		coeff	se (coeff)	z	p value
Clothing		-1.535	1.422	-1.079	.280
Gender	female	Reference			
	male	-0.666	0.361	-1.844	.065
Time.Norway	\leq 3 years	Reference			
	>3 years	0.617	0.285	2.164	.030*
Time.day	morning	Reference			
	afternoon	-0.517	0.183	-2.828	.005*
Top.delta		-0.956	0.182	-5.241	<.001*
Likelihood ratio	test = 54.17 or	$15 ext{ df, } p = 1.93$	33E-10		

*indicates a significant term.

case, however, they remain not because they are confounders but because their presence improves the model's overall fit compared to the model without them ($\chi^2(2) = 8.438$, p = .01471).

The next step is to verify whether the linearity assumption for the continuous variables in the model is met. The same procedure as described previously was applied here, but only the statically significant continuous variable (i.e., operative temperature variation) is shown (Fig. 10). Here, the hazard decreases fairly linearly with the decrement in the operative temperature. Therefore, a linear relationship between the log(hazard) and the operative temperature variations was assumed (purple line).

No significant interactions were found, and the PH assumption for the time-independent variables has been met.

5. Discussion

In this study, the results obtained from the observed thermal discomfort events were precautionary for both space heating and cooling processes. The possibility of undertaking voluntary adaptation mechanisms or actions was precluded, including such simple actions as clothing adjustment. This approach agrees with the one used in the ASHARE standard 55 for temperature variation with time. The objective of ASHARE limits on temperature cycles, drifts and ramps (Table 1) is mainly to prevent occupants from experiencing discomfort due to temperature variations. Its applicability is limited to temperature fluctuations that are not under the individual occupant's direct control. Moreover, an occupant's clothing adaptation is implicitly considered only for occupant-controlled naturally conditioned spaces that meet specific criteria (see section 5.4.1 of [2]). Nevertheless, these limits seemed both loose and conservative compared with the results of this study. Cold temperature variations were perceived to be uncomfortable earlier than the standard prescribed. On the other hand, the limits for can also be directly observed from this study's results, from both a descriptive and analytical perspective. The KM method gave a descriptive point of view. Fig. 7 showed that the survival probability is higher for heating processes compared with cooling ones, even for the same rate of temperature change (thermal ramp). Concerning cooling variations, it can be observed that different temperature change rates initially affected survivability similarly. The contrast between the different cooling ramps is more marked at a later time. This can also be explained using the findings of Ran's neuroscience experiment [22]. In their study, the authors observed that a larger delta temperature produced greater responses than smaller delta temperature values. Nevertheless, they noted no observable cooling rate effects on either the percentage of cold-responding neurons or their response amplitudes. Of course, faster temperature variation results in a greater delta temperature, assuming the same amount of time. However, when the amount of time is small, the temperature difference with distinct temperature change rates is smaller. Therefore, some (perhaps more sensitive) participants experienced a thermal discomfort event regardless of the rate of temperature change for cooling. In turn, this explains why, at an early phase, the survivability for a different rate of temperature change was similar. An analytical point of view was given by the Cox-regression. The model for cooling showed no statistically significant effect on the starting operative temperature. Conversely, in the heating model, the risk of experiencing a warm discomfort event increased with higher starting operative temperatures.

Furthermore, if considering elevated air movement, it is reasonable to assume that the observed thermal discomfort events for warm temperature variation could be postponed. Elevated air movement is a recognised factor that increases the acceptable range of operative temperatures [2]. In this study, the air was kept practically still to avoid local discomfort (a draft) during cooling. This resulted in air movement being an insignificant predictor.



Fig. 10. Penalised spline fit of operative temperature change for cooling.

warm temperature variations were found to be excessively restrictive. This asymmetric behaviour is supported by neurophysiological findings.

5.1. Limitations

As mention in section 1.1, in the spinal cord, cold-responding neurones react to temperature changes, while heat-responding ones react to the absolute temperature. Consequently, humans are more sensitive to cooling than heating, meaning that they react more quickly or more than usual to cooling than to heating. This neurophysiological interpretation This study's limitations arise from the relative homogeneity of age and the unbalanced number of male and female participants. Since most of the participants were between 23 and 31 years old, the results are not completely representative of the office worker population. The gender imbalance among participants might be the main cause of nonstatistically significant differences between males and females in terms of thermal discomfort. To reduce the effect of the generally heterogeneous initial metabolic rate, participants spent the first half-hour before starting the session in a constant temperature environment. However, previous studies on thermal comfort in climatic chambers have shown that subjects' average thermal sensation decreases during the first 2 h, even during exposure to constant temperatures [13]. On the other hand, time and organisational constraints did not allow for such an extension of this study's acclimation period. Therefore, it is possible that the potential carry-over effects influenced the participants' thermal sensation even after the 30-min acclimation phase.

Even with some constraints (e.g., clothing adjustment), this study aimed to reproduce a typical office environment and, consequently, simulate a typical office activity pattern. Nevertheless, participants were prone to the Hawthorne effect.⁵ Typically, the Hawthorne effect is described as a change in research participants' behaviour in experimental or observational studies. In this study, to avoid potential bias, participants were blinded to the environmental changes; that is, they were not informed about the change in the temperature. However, if the participants changed their behaviour during the experiment - for example, by increasing their awareness and, therefore, sensitivity to change in the indoor environmental condition - the Hawthorne effect would have occurred. Also, the use of the digital button could have introduced a behavioural change. Schweiker et al. [53], in their review of multi-domain approaches to indoor environmental perception and behaviour, pointed out that there is a difference in the intention to perform an action and the action itself. It is undeniable that performing an actual action, for example, adjusting the thermostat, would have required more effort than pressing the digital button. On the other hand, the opposite is also true. A specific human-building interface affects the level of interaction that a person has with it, and therefore its usability, which could lead to a different behavioural choice [54]. For example, even in a more familiar context, such as a residential setting, a common usability barrier for a thermostat is its complexity or the buttons' reduced size and comprehensibility [55]. Furthermore, it would be unfeasible to provide all the real means of possible interaction with the indoor environment (e.g., for the thermal environment alone, these include open/close window, thermostat adjustment, beverage intake, personalised/local cooler/heater, and ceiling/desk fans). Therefore, even with the aforementioned limitations, the discomfort button was adopted.

5.2. Conclusion and future perspectives

An experimental study has been conducted to explore the effects of ramp-induced temperature variations in an office setting. The purpose was to understand human reaction to monotonic thermal variations by describing the relationship between human response and covariates of interest. The study's design was a randomised crossover trial, a longitudinal study in which participants received a randomised sequence of different exposure (i.e., thermal ramps). Based on the analysis carried out, the following conclusions can be drawn:

- The distributions of participants' thermal comfort ratings during warm and cold discomfort events were remarkably similar, despite different temperature changes. This suggests that, indeed, thermal comfort is the driver for thermal behaviour. Thermal sensation votes were found to be asymmetric during discomfort events, while most of the thermal acceptability votes were at the boundary between an acceptable and unacceptable environment. This could indicate that thermal acceptability has a broader meaning, which, in a general sense, might be interpreted as tolerance.

- A distinct discomfort mechanism for space heating and cooling processes was observed in this experiment. For warm discomfort, the operative temperature level is a significant predictor, while for cold discomfort, the relative change in operative temperature is the trigger. This result agrees with the recent research evidence from neuroscience experiments [22].
- During heating variations, in addition to operative temperature (that is, the operative temperature variation plus the initial operative temperature), BMI and time lived in Norway significantly predicted participants' warm discomfort. For cooling processes, besides operative temperature variation, time lived in Norway and time of day were significant predictors of cold discomfort. For both space heating and cooling processes, gender and age did not significantly affect discomfort. Furthermore, no significant interaction has been identified.
- The current experimental results imply that the limits for drifts and ramps are not symmetric in winter conditions. The limits on temperature cycles, drifts and ramps defined in ASHRAE 55-2017 [2] are loose for cold temperature variations and conservative for warm ones.

In addition, this paper overcomes some important methodological issues concerning the semantic equivalence of different psychometric scales, highlighting, at the same time, the practical implications. For instance, a classic hypothesis (rule-of-thumb) is to consider an environment "satisfactory" when the thermal sensation vote is between "slightly cold" (-1) and "slightly warm" (+1). In this study, this conversion is well suited for warmer variations. Still, it is utterly misleading for colder ones. Fig. 6 shows that the majority of the discomfort events were experienced when the environment was perceived as "slightly cold".

In the context of multi-domain comfort, the methodology applied in this study could be used to analyse the relation between perception and action. It would also be possible to evaluate which contextual and personal factors affecting behaviour influence perception and vice versa.

Furthermore, the new knowledge of human reaction to a dynamic thermal environment can be used to design more energy-efficient and satisfying control strategies to enable buildings' thermal flexibility. Indeed, controlling the indoor temperature of buildings within a comfort range is a way to provide energy flexibility to the grid [56], exploiting the slow thermal inertia of a building's envelope in combination with the users' comfort band. However, the comfort band is usually assumed symmetric for space heating and cooling purposes and defined solely by absolute values of the indoor operative temperature, such as in the ASHRAE 55-2017 standard [57]. The findings presented in this paper have the potential to improve the performance of such controllers by providing a more accurate description of the human thermal response under dynamic conditions.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgement

The authors would like to express their gratitude to A. Nocente and O.A. Oksavik for their technical assistance, and to A. Marchenko and V. Rizzardi for their help in carrying out the experiment. Furthermore, they would like to thank Prof. Dr S. Martino for the fruitful discussions on survival analysis methods. Appreciation is also expressed to the subjects who participated in the experiments. This work has been performed

⁵ The Hawthorne effect got its name from a study of psychological aspects and physical and environmental influences in the workplace at the Western Electric Company's Hawthorne facility in Cicero, Illinois, during the 1920s. There, the workers increased their productivity when under observation but decreased it at the end of the study.

M. Favero et al.

within the framework of the International Energy Agency – Energy in Buildings and Communities Program (IEA-EBC) Annex 79 "Occupantcentric building design and operation". The authors would like to thank the Research Centre on Zero Emission Neighbourhoods in Smart Cities (FME ZEN, Grant n. 257660) and the Research Council of Norway (*Norges Forskingsrådet*) for their support.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.buildenv.2021.108144.

References

- ANSI/ASHRAE Standard 55, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, GA, 1966.
- [2] ANSI/ASHRAE Standard 55, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, GA, 2017.
- [3] ISO 7730, International Organization for Standardization, Geneva, Switzerland, 2005.
- F.H. Rohles, G.A. Milliken, D.E. Skipton, I. Krstic, Thermal comfort during cyclical temperature fluctuations, Build. Eng. 86 (1980) 125–149.
 C.H. Sprague, P.E. McNall, The effects of fluctuating temperature and relative
- [3] Ch. optague, E. Meran, the effects of internating temperature and relative humidity on the thermal sensation (thermal comfort) of sedentary subjects, Build. Eng. 76 (1970) 146–157.
- [6] D.P. Wyon, T. Asgeirsdottir, P.K. Jensen, P.O. Fanger, The effects of ambient temperature swings on comfort, performance and behavior, Arch. Sci. Physiol. 27 (1973) 441-458.
- [7] Wyon, D. P., Bruun, N. O., Olesen, S., Kjerulf-Jensen, P. & Fanger, P. O. in Proc. 5th Int. Congress for Heating, Ventilating and Air-Conditioning. 87-107.
- [8] R. Nevins, R.R. Gonzalez, L.G. Berglund, A.P. Gagge, Effect of changes in ambient temperature and level of humidity on comfort and thermal sensations, Build. Eng. 81 (1975) 169–182.
- [9] I.D. Griffiths, D.A. McIntyre, Sensitivity to temporal variations in thermal conditions, Ergonomics 17 (1974) 499–507, https://doi.org/10.1080/ 00140137408931380.
- [10] L.G. Berglund, R.R. Gonzalez, Occupant acceptability of eight-hour-long temperature ramps in the summer at low and high humidities, Build. Eng. 84 (1978) 278–284.
- [11] L.G. Berglund, R.R. Gonzalez, Application of acceptable temperature drifts to built environments as a mode of energy conservation, Build. Eng. 84 (1978) 110–121.
 [12] F.H. Rohles, J.E. Laviana, R. Wei, R. Wruck, The human response to temperature
- [12] F.H. Rolles, J.E. Laviala, K. Wei, K. Witck, the human response to temperature drifts in a simulated office environment, Build. Eng. 91 (1985) 116–123.
 [13] H.N. Knudsen, et al., Thermal comfort in passive solar buildings. CEC Research
- [13] H.N. Knudsen, et al., Thermal comfort in passive solar buildings. CEC Research Project: EN3S-0035-DK(B), 1989.
- [14] J.L.M. Hensen, Literature review on thermal comfort in transient conditions, Build. Environ. 25 (1990) 309–316, https://doi.org/10.1016/0360-1323(90)90004-B.
- [15] M. Miura, T. Ikaga, Human response to the indoor environment under fluctuating temperature, Sci. Technol. Built. Environ. 22 (2016) 820–830, https://doi.org/ 10.1080/23744731.2016.1184550.
- [16] F. Zhang, R. de Dear, C. Candido, Thermal comfort during temperature cycles induced by direct load control strategies of peak electricity demand management, Build. Environ. 103 (2016) 9–20, https://doi.org/10.1016/j. buildenv.2016.03.020.
- [17] L. Schellen, W.D. van Marken Lichtenbelt, M.G.L.C. Loomans, J. Toftum, M.H. de Wit, Differences between young adults and elderly in thermal comfort, productivity, and thermal physiology in response to a moderate temperature drift and a steady-state condition, Indoor Air 20 (2010) 273–283, https://doi.org/ 10.1111/j.1600-0668.2010.00657.x.
- [18] J. Kolarik, J. Toftum, B.W. Olesen, A. Shitzer, Occupant responses and office work performance in environments with moderately drifting operative temperatures (RP-1269), HVAC R Res. 15 (2009) 931–960, https://doi.org/10.1080/ 10789669_2009.10390873.
- [19] J. Vriens, B. Nilius, T. Voets, Peripheral thermosensation in mammals, Nat. Rev. Neurosci. 15 (2014) 573–589, https://doi.org/10.1038/nrn3784.
- [20] A. Dhaka, V. Viswanath, A. Patapoutian, Annual Review of Neuroscience, vol. 29, 2006, pp. 135–161.
- [21] R. Palkar, E.K. Lippoldt, D.D. McKemy, The molecular and cellular basis of thermosensation in mammals, Curr. Opin. Neurobiol. 34 (2015) 14–19, https:// doi.org/10.1016/j.conb.2015.01.010.
- [22] C. Ran, M.A. Hoon, X. Chen, The coding of cutaneous temperature in the spinal cord, Nat. Neurosci. 19 (2016) 1201–1209, https://doi.org/10.1038/nn.4350.
- [23] T.C. IUPS, Glossary of terms for thermal physiology, in: Revised by the Commission for Thermal Physiology of the International Union of Physiological Sciences (IUPS Thermal Commission), third ed., vol. 51, Japanese Journal of Physiology, 2001, pp. 245–280.
- [24] L.E. Marks, R.R. Gonzalez, Skin temperature modifies the pleasantness of thermal stimuli, Nature 247 (1974) 473–475, https://doi.org/10.1038/247473a0.
- [25] T. Parkinson, R. De Dear, Thermal pleasure in built environments: physiology of alliesthesia, Build. Res. Inf. 43 (2015) 288–301, https://doi.org/10.1080/ 09613218.2015.989662.

Building and Environment 204 (2021) 108144

- [26] M. Cabanac, Sensory pleasure, Q. Rev. Biol. 54 (1979) 1–29, https://doi.org/ 10.1086/410981.
- [27] Z.J. Schlader, S.E. Simmons, S.R. Stannard, T. Mündel, The independent roles of temperature and thermal perception in the control of human thermoregulatory behavior, Physiol. Behav. 103 (2011) 217–224, https://doi.org/10.1016/j. physbeh.2011.02.002.
- [28] A. A. Thermoregulation Romanovsky, Some concepts have changed. Functional architecture of the thermoregulatory system, Am. J. Physiol. Regul. Integr. Comp. Physiol. 292 (2007) R37–R46, https://doi.org/10.1152/ajpregu.00668.2006.
- [29] B.R.M. Kingma, A.J.H. Frijns, L. Schellen, W.D. van Marken Lichtenbelt, Beyond the classic thermoneutral zone: including thermal comfort, Temperature 1 (2014) 142-149, https://doi.org/10.4161/temp.29702.
- [30] M. Vellei, J. Le Dréau, A novel model for evaluating dynamic thermal comfort under demand response events, Build. Environ. 160 (2019), https://doi.org/ 10.1016/j.buildenv.2019.106215.
- [31] P.O. Fanger, Thermal Comfort: Analysis and Applications in Environmental Engineering, Danish Technical Press, 1970.
- [32] Goia, F., Schlemminger, C. & Gustavsen, A. in Energy Procedia. 531-536.
- [33] ISO 10551, International Organization for Standardization, Geneva, Switzerland, 2019.
- [34] J.D. Kalbfleisch, R.L. Prentice, The Statistical Analysis of Failure Time Data, 2011.
 [35] J.P. Klein, M.L. Moeschberger, Survival Analysis, Techniques for Censored and
- Truncated Data. Statistics Biology Heal, 2003, https://doi.org/10.1007/b97377. [36] D.G. Kleinbaum, M. Klein, Survival Analysis, A Self-Learning Text. Statistics
- [30] D.G. Kleinbaum, M. Klein, Survival Analysis, A Sen-Learning Text. Statistics Biology Heal, 2012, https://doi.org/10.1007/978-1-4419-6646-9.
 [37] E.L. Kaplan, P. Meier, Nonparametric estimation from incomplete observations,
- [57] E.L. Kapian, P. Weier, Nonparametric estimation from incomplete observations, J. Am. Stat. Assoc. 53 (1958) 457–481, https://doi.org/10.1080/ 01621459.1958.10501452.
- [38] T.M. Therneau, P.M. Grambsch, Modeling Survival Data: Extending the Cox Model, 2000.
- [39] ISO 7726, International Organization for Standardization, Geneva, Switzerland, 1998.
- [40] R, A Language and Environment for Statistical Computing, R Foundation for Statistical Computing, Vienna, Austria, 2020.
- [41] A Package for Survival Analysis in R, 2020.
- [42] ggplot2, Elegant Graphics for Data Analysis, Springer-Verlag New York, 2016.
- [43] survminer, Drawing Survival Curves Using, ggplot2, 2020.
 [44] F.E.J. Harrell, Regression Modeling Strategies, with Applications to Linear Models, Logistic and Ordinal Regression, and Survival Analysis, Springer Ser Statistics, 2015, https://doi.org/10.1007/978-3-319-19425-7.
- [45] M. Luo, et al., The time-scale of thermal comfort adaptation in heated and unheated buildings, Build. Environ. 151 (2019) 175–186, https://doi.org/ 10.1016/j.buildenv.2019.01.042.
- [46] World Health Organization, Body mass index BMI (n.d.), https://www.euro.who. int/en/health-topics/disease-prevention/nutrition/a-healthy-lifestyle/body-massindex-bmi.
- [47] M. Indraganti, R. Ooka, H.B. Rijal, Thermal comfort in offices in India: behavioral adaptation and the effect of age and gender, Energy Build. 103 (2015) 284–295, https://doi.org/10.1016/j.enbuild.2015.05.042.
- [48] E.J.W. Van Someren, R.J.E.M. Raymann, E.J.A. Scherder, H.A.M. Daanen, D. F. Swaab, Circadian and age-related modulation of thermoreception and temperature regulation: mechanisms and functional implications, Ageing Res. Rev. 1 (2002) 721–778. https://doi.org/10.1016/S1568-1637(02)00030-2.
- [49] J. Smolander, M. Härmä, A. Lindgvist, P. Kolari, L.A. Laitinen, Circadian variation in peripheral blood flow in relation to core temperature at rest, Eur. J. Appl. Physiol. Occup. Physiol. 67 (1993) 192-196, https://doi.org/10.1007/ BE000376666
- [50] K. Krauchi, A. Wirz-Justice, Circadian rhythm of heat production, heart rate, and skin and core temperature under unmasking conditions in men, Am. J. Physiol. Regul. Integr. Comp. Physiol. 267 (1994) R819–R829, https://doi.org/10.1152/ ajpregu.1994.267.3.r819.
- [51] P.O. Fanger, J. Højbjerre, J.O.B. Thomsen, Thermal comfort conditions in the morning and in the evening, Int. J. Biometeorol. 18 (1974) 16–22, https://doi.org/ 10.1007/JBF0145061.
- [52] P.O. Fanger, O. Östberg, McK, A.G. Nicholl, N.O. Breum, E. Jerking, Thermal comfort conditions during day and night, Eur. J. Appl. Physiol. Occup. Physiol. 33 (1974) 255–263, https://doi.org/10.1007/BF00430233.
- [53] M. Schweiker, et al., Review of multi-domain approaches to indoor environmental perception and behaviour, Build. Environ. 176 (2020), https://doi.org/10.1016/j. buildenv.2020.106804.
- [54] J.K. Day, et al., A review of select human-building interfaces and their relationship to human behavior, energy use and occupant comfort, Build. Environ. 178 (2020), https://doi.org/10.1016/j.buildenv.2020.106920.
- [55] A. Meier, C. Aragon, T. Peffer, D. Perry, M. Pritoni, Usability of residential thermostats: preliminary investigations, Build. Environ. 46 (2011) 1891–1898, https://doi.org/10.1016/j.buildenv.2011.03.009.
- [56] A.J. Marszal-Pomianowska, et al., Principles of Energy Flexible Buildings: Energy in Buildings and Communities Programme Annex 67 Energy Flexible Buildings, 2020.
- [57] J. Drgoña, et al., All you need to know about model predictive control for buildings, Annu. Rev. Contr. 50 (2020) 190–232, https://doi.org/10.1016/j. arcontrol.2020.09.001.

Paper II

Analysis of subjective thermal comfort data: a statistical point of view

Matteo Favero¹, Antonio Luparelli², Salvatore Carlucci^{3,*}

¹ Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway

² Automation and Information Systems Area, CETMA - European Research Centre for Technologies Design and Materials, Brindisi, Italy

³ Energy, Environment and Water Research Center, The Cyprus Institute, Nicosia, Cyprus *Corresponding author address: Konstantinou Kavafi 20, 2121 Aglantzia, Cyprus; email: s.carlucci@cyi.ac.cy

Abstract

Thermal comfort research aims to determine the relationship between the thermal environment and the human sense of warmth. This is usually achieved by measuring the subjective human thermal response to different thermal environments. However, it is common practice to use simple linear regression to analyse data collected using ordinal scales. This practice may lead to severe errors in inference. This study first set the methodological foundations to analyse subjective thermal comfort data from a statistical perspective. Subsequently, we show the practical consequences of fallacious assumptions by utilising a Bayesian approach and show that, at least with one dataset, a linear regression model applied to ordinal data suggests results different from those obtained using ordinal regression. Specifically, given the specified dataset, linear regression found no difference in means and effect size between genders, while the ordinal regression model led to the opposite conclusion. In addition, compared to the ordinal model, the linear regression model distorts the estimated regression coefficient for air temperature. Finally, the ordinal model shows that the distance between adjacent response categories of the ASHRAE 7-point thermal sensation scale is not equidistant. Given the abovementioned issues, we advocate utilising ordinal models instead of metric models to analyse ordinal data.

Keywords

Subjective thermal comfort data; Rating scales; Level of measurement; Ordinal regression; Bayesian analysis; Statistical thinking.

1 Introduction

One of the aims of thermal comfort research is to establish the relationship between the thermal environment and the human sensation of warmth. The sensation of warmth is quantified by rating scales, the most adopted of which is the ASHRAE 7-point thermal sensation scale, which consists of seven verbal anchors: 'cold', 'cool', 'slightly cool', 'neutral', 'slightly warm', 'warm', and 'hot'. This is a perceptual judgement scale [1] and is utilised to measure thermal sensation. Other rating scales are also employed in thermal comfort studies: the most common ones being thermal evaluation, preference, and acceptability. ISO 10551:2019 [1], beyond those already mentioned, also introduces a 'tolerance scale', which is rarely used in the scientific literature. Each one of these scales can be presented in different formats (e.g., discontinuous versus continuous format) and methods (e.g., paper- versus computer-based). Independently of the format and method used, it is common practice to assign a numerical value to each level (i.e., the verbal anchors) of a scale. For instance, the ASHRAE 7-point thermal sensation scale generally varies from -3 ('cold') to +3 ('hot'). However, different values can be assigned, such as 1 for 'cold' and 7 for 'hot'. This interchangeability is possible because these numbers are merely placeholders without an underlying meaning. Nevertheless, it is common practice to calculate the mean of the thermal sensation votes of a group of people (e.g., [2,3]). The reasoning behind this method is that, while the variable is ordinal in nature, a vote created by averaging different responses is continuous. Furthermore, the averaged votes will result in a more normal-looking distribution and, therefore, statistical methods that assume normality (e.g., linear regression and analysis of variance) can be applied. The origin of this approach can be found in early works to measure attitudes, such as in Thurstone [4] and Likert [5]. However, there are two problems with this approach. Firstly, it is not appropriate to calculate the mean of an ordinal variable because its linearity (i.e., equally spaced divisions) is an arbitrary assumption imposed on the original scale values. This assumption was also recently questioned by Schweiker et al. [6,7]. Secondly, this approach conflates the problem of the level of measurement with that of the distribution of a variable. Averaging ordinal data may improve the degree to which the distribution of votes resembles a normal distribution, but it does not change the nature of the observations from ordinal to interval.

Concerning the analyses of subjective thermal comfort data, ISO 10551:2019 [1] gives guidance to the analysis of ordinal data. Unfortunately, it uses disputed arguments, based on McIntyre's work [8], to legitimise treating ordinal data from the ASHRAE 7-point thermal sensation scale as a continuous variable. In his paper published in 1978, McIntyre clearly stated that the 7-point warmth scale is ordinal and that, therefore, nonparametric statistics are the appropriate method. However, McIntyre also said that non-parametric statistics are generally related to hypothesis testing and are guite limiting for thermal comfort analysis. Therefore, utilising the method of graded dichotomies, he investigates whether these scales can be treated as intervals (i.e., if the psychological width of the categories can be approximated to be of equal spacing). McIntyre concluded that there is 'no reason to suppose that we are not dealing with an equal interval scale', even if nothing can be said to the extreme categories, that is 'cold' and 'hot' [8]. In addition, performing a Kolmogorov-Smirnov test¹ (K-S test), he found no significant deviation from normality and deduced that it is appropriate to use statistical methods that presuppose normality. However, checking whether an ordinal variable can be assumed to be interval for analytic purposes may work in some cases, but it does not constitute general proof. Nevertheless, this practice seemed reasonable at the time, considering that, until the 1960s, there was relatively little development of models for categorical responses (see page 1 of [9]). Furthermore, the K-S test can be applied only to continuous distributions, which is not the case analysed by McIntyre. In addition, the distribution used to compare the sample must be fully specified, that is, the location and scale parameters (i.e., mean and standard deviation) of the normal distribution must be known a priori and not estimated from the data. If these parameters are calculated from the data, the critical region of the K-S test is no longer valid and should be determined by simulation.

¹ The K–S test is used to test if a sample comes from a population with a specific distribution.

In the following two sections, the notion of 'level of measurement' is introduced (Section 1.1), and the issue of analysing ordinal data as metric is discussed (Section 1.2.3). Discussion regarding the different types of scales employed (e.g., categorical scale, visual analogue scale, and graphic categorical scale), the number of anchors utilised, and the assumptions underlying their usage are outside the scope of this study. The interested reader is referred to previous studies such as [6,7,10,11] for further discussions of these topics.

1.1 Level of measurement

A level of measurement is a classification that represents the nature of the information contained in the values assigned to the variables [12]. A widespread measurement classification is Stevens's typology [13], which is divided into four classes: nominal, ordinal, interval, and ratio. The nominal scale identifies or categorises the values of the variables but cannot order the categories; the ordinal scale, in which the values of the variables are ranked or ordered, is used for this purpose. For the interval scale, the intervals between the values of the variables are equally spaced, and the zero on the scale is arbitrary (i.e., the zero on the scale is a matter of convention or convenience). Conversely, the ratio scale has a true zero point, which defines the absence of the quantity being measured. As a consequence, ratios of magnitudes can be defined.

In Stevens's view, it is important to know which kind of scale one is dealing with because 'to each of these types of scales certain statistics are appropriate and others are not' [14], and a scale that retains meaning under a certain class of transformations should be limited to statistics whose meaning would not change if those transformations were applied to the data. Table 1 shows the different types of scales with their empirical operations, invariant mathematical transformations, and (permissible) measures of central tendency.

Scale	Empirical operations	Permissible transformations	Permissible measures of central tendency
Nominal	Determination of equality	Any one-to-one substitution	Mode
Ordinal	Determination of greater or lesser (rank-order)	Any increasing monotonic transform	Median
Interval	Determination of the equality of intervals or of differences	Multiplication by and addition of a constant	Arithmetic mean
Ratio	Determinations of the equality of ratios	Multiplication by a constant	Geometric mean Harmonic mean

Table 1 – Types of measurement scales (from [14])

Stevens went beyond his simple typology and classified not just simple operations but also statistical procedures according to the scales for which they were permissible. The idea that a particular level of measurement prescribes or proscribes statistical methods has been strongly criticised by statisticians [15-17], and alternative taxonomies have been proposed. Mosteller and Tukey's typology [18] and Chrisman's typology [19] introduced an expanded list of levels of measurement to account for various measurements that do not fit well into Stevens's framework. The difference is that they do not prescribe statistical methods nor even suggest that statistical methods should depend on the levels of measurement. Statistical analyses make assumptions about the distributions of variables and/or errors, not about measurement levels. Of course, it is necessary to verify that these assumptions comply with the data at hand. However, to conclude that there is no value in the data types would be inaccurate. The notion of scale type is important, and Stevens's nomenclature is frequently used. For example, any designed experiment must distinguish between

categorical factors (usually nominal or ordinal in Stevens's terminology) and metric/continuous covariates (usually intervals or ratios) [15]. However, these scale types derive from how the data were measured rather than being fundamental characteristics of the data themselves.

1.2 Statistical methods: a brief overview

As stated previously, one of the goals of thermal comfort research is to establish a relationship between the thermal environment and the human response. In a statistical modelling framework, this is generally achieved through regression analysis. Regression analysis is 'the blanket name for a family of data analysis techniques that examine relationships between variables' [20], which are categorised into a dependent variable ('outcome' or 'response' variable), *Y*, and one or more independent variables ('explanatory variables', 'predictors', 'covariates' or 'features'), *X*.

The most common approach utilised in thermal comfort research for the analysis of subjective thermal comfort data is linear regression. Another approach, even if far less common, is ordinal regression (e.g., [21]). The main difference between the two is that linear regression requires the dependent variable to be continuous, while ordinal regression requires it to be ordinal. Even though different regression models have different mathematical underpinnings, they share a general form that can be expressed as the function of a random component, $g(\cdot)$, which refers to the conditional probability distribution of the response variable, and a systematic component, $h(\cdot)$, which refers to the explanatory variables. The systematic component is used as the predicted tendency of *Y* given the predictors. Nevertheless, *Y* is not predicted to be exactly $h(\cdot)$, but near $h(\cdot)$. That is, the best that can be done is to predict the probability that *Y* will have any particular value, given *x*. This probability density function (PDF) is the random component $g(\cdot)$. This is a more general notation that encompasses different models (i.e., it extends more easily to other models by focusing on the conditional distribution of the response rather than the distribution of the error term [22]). The following sections briefly describe two regression-type models utilised to model subjective thermal comfort data.

1.2.1 General linear model

The most common approach utilised in thermal comfort research for the analysis of subjective thermal comfort data is the general linear model (GLM), which usually refers to the linear regression model. In this model, the continuous response variable is modelled given some predictors, generally assuming a conditional normal distribution of the response:

$$Y_i \sim \text{Normal}(\mu_i, \sigma^2)$$
$$\mu_i = \eta_i$$
Eq. (1)
$$\eta_i = \mathbf{x}_i^T \boldsymbol{\beta}$$

where μ_i is the mean, σ is the standard deviation, and η_i is the predictor term function of some predictors $\mathbf{x}_i^{\mathrm{T}}$. The subscript *i* is to stress the dependency on the *i*th observation.

1.2.2 Cumulative link model

As mentioned previously, if the response variable is assumed to be ordinal (and therefore measured as ordinal), it is proper to analyse it with ordinal models. Cumulative link models (CLMs) belong to a broad class

of models known as ordinal regression models. Following the categorisation of Bürkner and Vuorre [23], in addition to the cumulative models, other two distinct model classes belong to the ordinal regression models: sequential and adjacent-category models. Each of these models has a different rationale behind it and, consequently, a different application.

The rationale behind choosing a CLM lies in the fact that this model has a latent variable representation, which is in line with the general assumption underlying the rating scales. The idea is that the dependent variable *Y* is the categorisation of a latent (not observable) continuous variable \tilde{Y} . Fig. 1 illustrates this concept. The categorical outcome, *Y* (Fig. 1.a and 1.b) is a categorised version of an unobservable (latent) continuous variable, \tilde{Y} (Fig. 1.c and 1.d). The dotted lines in the bottom figures divide the continuous latent variable into K + 1 bins according to the threshold parameters $\{\tau_k\}$, with $k \in \{1, ..., K\}$. Consequently, the area under the curve in each bin represents the probability of the corresponding observed ordinal response (Fig. 1.a and 1.b). In Fig. 1, the thresholds are shown as not equidistant and equidistant (Fig. 1.c and 1.d, respectively) for illustrative purposes only. In practice, the thresholds are determined by nature; they are parameters to be estimated.





equidistant (d).

The conditional distribution of the response variable *Y* is assumed to follow a multinomial distribution where its probability vector is $\pi = {\pi_1, ..., \pi_k}$ with $\pi_k = \Pr(Y = k)$. The cumulative probability corresponding to π_k is $\gamma_k = \Pr(Y \le k)$ so that $\gamma_k = \pi_1 + ... + \pi_k$. The cumulative probabilities are then mapped to the real numbers through a link function. In this study, the probit function was chosen as the link function. The reason is that the probit link assumes the latent variable to be normally distributed² around the predicted central tendency (i.e., the mean of the latent scale) and is therefore comparable with linear regression. The mathematical form of the model can be written as:

² Technically, the distributional assumption should be made on the error term, not the response variable. However, in linear regression, to assume the error as normally distributed around zero is equivalent to assuming the response to be normally distributed around the regression line.

$$Y_i \sim \text{Multinomial}(n, \boldsymbol{\pi}_i)$$

Probit $(\gamma_{ik}) = \tau_k - \eta_i$
Eq. (2)
$$\eta_i = \mathbf{x}_i^T \boldsymbol{\beta}$$

where τ_k are the thresholds parameters and η_i is the linear predictor term without an intercept³. The subscript *i* is to stress the dependency on the *i*th observation.

For more explanations and practical guidelines for using this and other methods (i.e., sequential and adjacentcategory models), along with detailed mathematical derivations and discussions, the reader is referred to Bürkner and Vuorre [23].

1.2.3 Ordinal-as-metric

While it is generally recognised that ordinal data are not metric, it is commonplace to analyse them with methods that assume metric responses. This is inappropriate for the following reasons. First and foremost, the ordinal variable's categories may not be equidistant since it is unknown the psychological distance between adjacent categories and whether these distances are the same across subjects. In a survey respondent's thinking, the difference between 'neutral' and 'slightly warm', for example, may be considerably smaller than the difference between 'warm' and 'hot', as demonstrated by Schweiker et al. [6,7]. Second, the distribution of ordinal categories can be nonnormal, especially if low (e.g., 'cold') or high (e.g., 'hot') values are commonly chosen. Third, the variances of the unobserved variables underlying the observed ordinal categories can vary, for example, between periods (e.g., seasons) and groups (e.g., gender). The ordinal-as-metric method cannot account for such uneven variances.

The issue of examining ordinal data as metrics was analysed in great detail by Liddell and Kruschke [24], whose arguments are summarised hereafter. To facilitate their understanding and explanation, Fig.3 and Fig.4 in [24] have been adapted and reproduced here as Fig. 2. In this figure, the mean of the ordinal values (i.e., when the ordinal values are treated as metric) is plotted as a function of the latent mean, μ , and standard deviation (SD), σ . The four letter-labelled points represent a specific combination of μ and σ on the underlying latent scale that, if used as parameters in a cumulative probit model, would generate a particular pattern in the ordinal data. For instance, the point indicated by (i.e., group B) has a latent mean and standard deviation of $\mu = 2$ and $\sigma = 1$, respectively (Fig. 2.c) and an ordinal mean of 1.93 (Fig. 2.b).

³ Omitting the intercept term allows the full set of thresholds $\tau_1, ..., \tau_k$ to be identified.



Fig. 2 – Mean of the ordinal values as a function of latent mean, μ , and SD, σ (a); ordinal level (b) and latent scale value (c) for group (B) (adapted from [24]).

Note. The thresholds τ_k (the dotted lines in (c)) are defined here as being equidistant.

From Fig. 2.a, four different 'effects' can be observed:

- 1 Points (A) and (B) illustrate a false-alarm rate (Type I error): these two groups have the same latent means, but the ordinal means are estimated as very different.
- 2 Points (B) and (D) illustrate a low correct-detection rate (Type II error): for these two groups, the latent means are quite different, but the ordinal means are estimated as equal.
- 3 Points (A) and (D) illustrate a distorted effect-size estimate: here, the two groups have identical latent variances, but the difference in means on the horizontal axis (i.e., on the underlying latent scale) is larger than the corresponding difference on the vertical axis (i.e., on the ordinal-as-metric scale).
- 4 Points © and D illustrate a reversed effect-size estimate: here, the latent mean of group D is greater than that of group ©, but the ordinal means for group © are incorrectly estimated to be greater than those of group D.

Liddell and Kruschke [24] posited that there are infinite combinations of underlying parameter values (μ and σ) that lead to inflated false-alarm rates, or low rates of correct detection, or distorted effect-size estimates, or inversions of differences between groups. Consequently, analysing ordinal data with metric methods (i.e., methods that assume continuous response variables), such as *t*-test, analysis of variance (ANOVA) and linear regression, could lead to the aforementioned issues. Furthermore, linear regression applied directly to ordinal values can misestimate regression coefficients, leading to incorrect inferences about differences or non-differences in slopes across conditions, as well as the existence or absence of non-linear trends. For further discussion and examples, the reader is referred to Liddell and Kruschke [24].

1.3 Objective and relevance of this study

Establishing the link between the thermal environment and the human sense of warmth is one of the goals of thermal comfort research. This is usually achieved by measuring the subjective human thermal response to different thermal environments. In this field, it is common practice to analyse subjective human thermal responses independently of how they have been measured. That is, the statistical analysis is unrelated to the

modalities of the data that have been acquired. For example, Zhang and de Dear [25] state that thermal sensation vote 'although it is essentially an ordinal variable, the thermal comfort research community has usually regarded it as a continuous variable'. From this statement, the authors (i) highlight that there is a difference between ordinal and continuous variables but (ii) specify that, within the thermal comfort research community, there is the tendency to consider it as continuous. In other words, linear regression is widely used to analyse TSV measured on an ordinal scale. Liddell and Kruschke [24] showed that analysing ordinal data as if they were continuous could lead to misleading results. This is particularly relevant for thermal comfort research, since thermal comfort models are mainly based on ordinal data analysed as if they were continuous (e.g., Ref [25] citing [26-30]). This might be a concurrent factor to explain why conflicting results were found in previous research where, for example, gender was shown to be or not an influential factor in determining human responses to the thermal environment. Furthermore, these models are included in international standards, such as EN 15251:2007 [31], replaced by EN 16798-1:2019 [32], and ASHRAE 55:2020 [33], which are used in the design and operation of buildings all around the world.

This paper focuses on analysing the data once they have been collected and not on the correctness of the level of measurement utilised to measure them (see Refs [6] and [7] for further discussions of this topic). For this purpose, this study leverages the largest global thermal comfort database to date. The aim of the paper is twofold. The first aim is to overview the methods commonly used to analyse subjective thermal comfort data from a statistical perspective. The second aim is to highlight the ordinal-as-metric issue that is often not considered and to spur researchers to analyse these kinds of data more critically. It is essential to emphasise that we are not advocating that the specific approach hereafter presented as the best way to analyse these kinds of data: the approach presented is merely one of the possible ways to do so. Statistics is a field that is an art as much as it is a science. Although statistical theory is founded on exact assumptions and conditions, the real world is seldom that straightforward. Consequently, the practice of statistics involves a tremendous number of choices, and the challenge is how to make those choices.

2 Methodology

2.1 A Bayesian approach to regression

In this study, a Bayesian approach is used to analyse the data. This approach is not entirely new in thermal comfort studies (e.g., [34-36]); however, it is not an established practice either. Since statistical knowledge in this field generally tends towards 'frequentist' principles, it is essential to explain the Bayesian approach and compare it with the frequentist one. Nevertheless, the aim of this paper is neither to go into details about their differences, nor to be a full introduction to either approach. For a more complete treatment, see, for example, [37] and [38].

Essentially, the divide between frequentists and Bayesians is in the definition of probability. For frequentists, probabilities are associated with frequencies of events. For Bayesian, probabilities are related to their own understanding (i.e., certainty or uncertainty) of events. This difference has important implications in the analysis of data. For instance, in a frequentist view, the parameter θ is considered a fixed (i.e., constant) but unknown quantity and only the information from the sampling data is relevant for the inference. On the

contrary, Bayesian statistics estimate the full (joint) posterior distribution of the parameters (i.e., the probability of the parameters given the observed set of data), which is generally calculated as:

$$Pr(\theta \mid Y) = \frac{Pr(Y \mid \theta) Pr(\theta)}{Pr(Y)}$$
Eq. (3)

where $Pr(Y | \theta)$ is the likelihood, $Pr(\theta)$ is the prior distribution, and Pr(Y) is the marginal likelihood. Here the parameters are considered random variables and not constant, as in the frequentist approach.

In Eq. (3), the $Pr(\theta)$ represent the prior 'belief' about the distribution of the parameters, and such a belief must be specified. Since there is no single method for choosing a prior (i.e., prior probability distribution), different priors can be introduced, leading to different posterior distributions and conclusions. This subjectivity is the main criticism of Bayesian inference. Furthermore, obtaining the posterior distribution analytically is rarely possible. Consequently, Bayesian statistics relies on Markov Chain Monte Carlo (MCMC) methods to estimate the posterior distributions of the parameters of interest. MCMC methods have a higher computational cost and fitting a model with Bayesian statistics is generally slower than the frequentist approach. However, Bayesian methods are usually more flexible and have more informative results (e.g., estimating a full posterior distribution, rather than a single point with a measure of uncertainty). Such advantages are often worth the increase in computational cost. Bayesian estimation does not have specific assumptions but relies on the model's assumptions, since those are the assumptions about the likelihood function. The fundamental assumption is that the likelihood function chosen is a reasonable representation of the data.

Generally, the assumptions behind a Bayesian model are not directly mentioned because they are stated when defining likelihood and priors. For example, Fig. 3 illustrates the formulation of a Bayesian model for simple linear regression. The corresponding mathematical formulations are added to the side for clarity.



Fig. 3 – Dependency diagram for a simple linear regression model (adapted from [39]).

This figure shows the assumptions about the random component (i.e., the conditional distribution assumptions for *y*) and the functional form of the systematic component (i.e., the expression for μ). The distributions of the parameters β_0 , β_1 , and σ are the priors. Since the standard deviation cannot be less than

zero, a half-normal distribution was selected as its prior (however, other distributions could have been chosen, such as exponential and uniform). For an introduction to Bayesian analysis or more advanced treatment, see [40] and [39], respectively.

2.2 Data preparation and software

As mentioned in Section 1.3, this study leverages the largest global thermal comfort database to date. This database, called ASHRAE Global Thermal Comfort Database II (downloaded from the University of California's DASH repository [41]), is an open-source database that includes approximately 107,500 sets of paired subjective comfort votes and objective instrumental measurements of the thermal environment. These observations were derived from field studies conducted worldwide between 1995 and 2016. A quality assurance check was performed on each dataset before its inclusion in the final database (see [42] for more details).

To achieve the aim of this study, the dependent variable needs to be measured on the ordinal scale. Unfortunately, the ASHRAE Global Thermal Comfort Database II does not distinguish between scales, and ordinal and continuous measurements are mixed. Additionally, even if all datasets composing the database went through a rigorous quality assurance process to harmonise their contents, it is reasonable to assume that each dataset has some unique peculiarities – different measurement protocols, questionnaires, or instruments. This aspect of the database would require that analysis of the entire database be carried out with an 'appropriate' method that considers these peculiarities (e.g., multilevel modelling) because, otherwise, the results may be unpredictably affected. For the purpose of this study, in order to reduce the uncertainty due to the unique peculiarities of different datasets, the following analysis was carried out on the data deriving from a single study.

Among the subjective thermal comfort votes available in ASHRAE Global Thermal Comfort Database II, the highest number of observations are thermal sensation votes (TSV). For this reason, TSV was selected as the dependent variable. However, the same analysis could be applied to the other rating scales if measured on the ordinal scale. For simplicity, only the two variables presented in Table 2 were utilised as covariates during the analysis. Indeed, thermal sensation depends on other variables, such as clothing, metabolic rate, air movement, radiant temperature, and relative humidity, and perhaps on several variables not yet clearly identified. Also, it is likely that not accounting for possible confounders affects the estimation of the models' coefficients. However, given that this study is an illustrative example, which aims to highlight the issue of analysing ordinal data as they were continuous, the issue of including/excluding important variables from the model can be overlooked.

Table 2 - List of covariates used in the model.

Variable	Code	Туре	Unit
Gender	Gender	Categorical	female (reference) / male
Air temperature	Tair	Continuous	°C

The analysis was carried out on Indraganti et al.' study [43] dataset, included in the ASHRAE Global Thermal Comfort Database II. This dataset was selected because (i) it does not have missing values for thermal

sensation votes, gender and air temperature; (ii) the full range of thermal sensation responses, measured on an ordinal scale, makes the dataset particularly suitable for this analysis (see Fig. 4); (iii) furthermore, the data were collected under a wide range of indoor air temperatures (min = $20.80 \,^{\circ}$ C; 1st quartile = $25.80 \,^{\circ}$ C; median = $26.80 \,^{\circ}$ C; mean = $27.06 \,^{\circ}$ C; 3rd quartile = $28.30 \,^{\circ}$ C; max = $36.50 \,^{\circ}$ C). The selected dataset comprised 6048 observations ($-27 \,^{\circ}$ female) collected during 14 months from 2787 individuals (all Indian nationals within the age group of 18-48 years). More details regarding the field survey can be found in Indraganti et al. [43].



Fig. 4 – Distribution of the thermal sensation vote.

All statistical analyses were performed using R [44] with the RStudio integrated development environment [45]. Regression analyses, using both the cumulative probit and classical linear regression, were performed with the *brms* package [46], and the respective graphs were created with the *ggplot2* package [47] via the *tidybayes* package [48].

2.3 Model parametrisation

Before proceeding with the analysis, it is essential to briefly explain how *brms* parameterises the cumulative probit model because this has repercussions on its interpretation. The cumulative distribution function (CDF) of an ordinal model based on cumulative probabilities with probit link (i.e., cumulative probit model) can be stated as:

$$Pr(Y_{i} \leq k \mid \{\tau_{k}\}, \eta_{i}, \sigma_{i}) = \Phi\left(\frac{\tau_{k} - \eta_{i}}{\sigma_{i}}\right)$$

$$\eta_{i} = \beta_{0} + \sum_{1}^{l} \beta_{l} x_{l,i}$$
Eq. (4)

$$\log(\sigma_i) = \delta_0 + \sum_{1}^{m} \delta_m x_{m,i}$$

where Φ indicates the cumulative normal distribution function, τ_k are the thresholds parameters, η_i is the linear predictor term and σ_i is the standard deviation. The subscript *i* is to stress the dependency on the *i*th observation. With K + 1 ordinal values, a model has (K + 1) + 1 parameters ($\tau_1, ..., \tau_k, \eta_i$ and σ_i) and is undetermined. Therefore, two parameters need to be fixed. *Brms* parametrises the model by fixing $\beta_0 = 0$ and $\delta_0 = 0$ and freely estimating all the thresholds, $\tau_1, ..., \tau_k$. When there are no predictors for η_i and σ_i in the model (i.e., unconditional model), $\eta_i = \beta_0 = 0$ and $\sigma_i = \exp(\delta_0) = 1$. Therefore, instead of estimating η_i and σ_i from a normal cumulative distribution function, *brms* uses the standard normal cumulative distribution function, *brms* uses the standard normal cumulative distribution function $\Phi(z)$. As a consequence, the parameters are expressed on the latent variable scale, that is, in units of ordered probit. Furthermore, since *brms* parametrise the model as:

a positive coefficient for β indicates that an increase of 1-unit of the associated variable x_i increases the thermal sensation vote. Stated analogously, voting in higher categories is more likely. The interpretation would have been the opposite if the model was parametrised differently (i.e., with a '+' instead of a '-'). A positive coefficient for β would have indicated that an increase of 1-unit of the associated variable x_i would decrease the thermal sensation vote.

For comparison, the CDF for the ordinary linear regression model can be stated as:

$$\Pr(Y_i \le y \mid \eta_i, \sigma_i) = \Phi\left(\frac{y - \eta_i}{\sigma_i}\right)$$

$$\eta_i = \beta_0 + \sum_{1}^{l} \beta_l x_{l,i}$$

$$\log(\sigma_l) = \delta_0 + \sum_{1}^{m} \delta_m x_{m,i}$$

Eq. (6)

where Φ indicates the cumulative normal, η_i is the linear predictor term and σ_i is the standard deviation. The subscript *i* is to stress the dependency on the *i*th observation. Here the β_0 and δ_0 are not fixed and therefore freely estimated by the model.

The following analysis was carried out for the cumulative probit model and compared with an ordinary linear regression, referred to as gaussian (ordinal-as-metric) model.

3 Results of the statistical analysis of subjective thermal comfort data

3.1 Unconditional model

The goal of a modelling strategy is to try to reproduce or predict an observable phenomenon via the lens of a model. Before incorporating a predictor, the unconditional model can be used to test the 'goodness' of the modelling technique. For example, if a model makes implausible predictions that are unobservable in reality,
perhaps a different technique should be adopted. The unconditional model for the cumulative probit and gaussian (ordinal-as-metric) models are the thresholds-only and intercept-only models, respectively. The unconditional model results are shown in Table 3, while Fig. 5 shows its posterior prediction. Here the data generated from the thresholds-only and intercept-only models are compared with the empirical data.

	Estimate	Est. Error	L-95 % CI*	U-95 % CI*		
Cumulative probit model						
Threshold 1, τ_1	-2.54	0.06	-2.66	-2.42		
Threshold 2, τ_2	-1.25	0.02	-1.29	-1.21		
Threshold 3, τ_3	-0.35	0.02	-0.38	-0.32		
Threshold 4, τ_4	0.48	0.02	0.45	0.52		
Threshold 5, τ_5	0.96	0.02	0.92	1.00		
Threshold 6, τ_6	1.47	0.02	1.42	1.52		
Gaussian (ordinal-as-metric) model						
Intercept	4.08	0.02	4.04	4.11		
Sigma	1.37	0.01	1.35	1.39		

Table 3 – Regression coefficients for the unconditional model.

* CI stands for credible interval (based on quantiles).



Fig. 5 – Posterior prediction for (a) the thresholds-only and (b) intercept-only model. Note. The green and red histograms are obtained from 8 draws from the posterior predictive distribution of the thresholds-only and intercept-only models, respectively.

The posterior predictive distribution for the cumulative probit model (Fig. 5.a) visually describes the distribution of the outcomes. Conversely, the posterior predictions for the gaussian (ordinal-as-metric) model (Fig. 5.b) are not a good fit, and they also have impossible predictive outcomes (i.e., value below the category '1' that is, 'cold' and above the category '7', that is, 'hot'). Fig. 6 shows the standard normal distribution underlying the ordinal data and the position of the estimated thresholds { τ_k } (see Table 3). The area under the curve in each bin represents the probability of the corresponding observed ordinal response (see Fig. 4).



Fig. 6 – Standard normal distribution underlying the ordinal data.

A 'pseudo' CDF is plotted in Fig. 7 for illustrative purposes only⁴ to inspect further and compare the two models. This direct contrast shows that the cumulative probit model better describes the data than the gaussian (ordinal-as-metric) model.



Fig. 7 – Superimposition of the CDF for the (a) cumulative probit and (b) gaussian (ordinal-as-metric) model.

Note. The green and red lines are obtained from 100 draws from the posterior predictive distribution of the thresholds-only and intercept-only models, respectively.

3.2 Fitting a categorical variable

In this section, the categorical variable *Gender* is added to the unconditional model. As described previously, *brms* parametrises the model by fixing $\eta_i = \beta_0 = 0$ and $\sigma_i = exp(\delta_0) = 1$. Therefore, the underlying Gaussian for the reference category of *Gender* (i.e., female) will be Normal(0,1). Thus, the parameter value for the

⁴ Strictly speaking, a cumulative distribution function is defined as a continuous function only for continuous variables. For discrete variables, it should be a step function.

other category of *Gender* (i.e., male) is the difference in means expressed on the latent variable scale for the reference category. The results of this model are shown in Table 4.

		Estimate	Est. Error	L-95 % CI*	U-95 % CI*
Cumulative prob	oit model				
Threshold 1, τ_1		-2.49	0.06	-2.62	-2.37
Threshold 2, τ_2		-1.20	0.03	-1.26	-1.14
Threshold 3, τ_3		-0.30	0.03	-0.35	-0.24
Threshold 4, τ_4		0.53	0.03	0.48	0.59
Threshold 5, $\tau_{\rm 5}$		1.01	0.03	0.96	1.07
Threshold 6, τ_6		1.52	0.03	1.46	1.58
Gender	female	reference			
	male	0.07	0.03	0.01	0.13
Gaussian (ord	inal-as-met	tric) model			
Intercept		4.03	0.03	3.96	4.10
Gender	female	reference			
	male	0.06	0.04	-0.02	0.14
Sigma		1.37	0.01	1.35	1.39

Table 4 – Regression coefficients for the model with only a categorical variable (assuming constant standard deviation).

* CI stands for credible interval (based on quantiles).

The model above presumes that the standard deviation of the latent variable is the same throughout the model (see Fig. 8 (top)). Unequal standard deviations can be included in the model by specifying an additional regression formula for the standard deviation component of the latent variable, \tilde{Y} . In the context of this example, allowing for unequal standard deviations implies inquiring whether the standard deviations for TSV differ across the two categories of *Gender*.

		Estimate	Est. Error	L-95 % CI*	U-95 % CI*	
Cumulative probit model						
Threshold 1, τ_1		-2.28	0.07	-2.41	-2.14	
Threshold 2, τ_2		-1.09	0.03	-1.16	-1.02	
Threshold 3, τ_3		-0.27	0.03	-0.32	-0.21	
Threshold 4, $ au_4$		0.48	0.03	0.43	0.54	
Threshold 5, τ_5		0.92	0.03	0.85	0.98	
Threshold 6, τ_6		1.38	0.04	1.30	1.46	
Gender	female	reference				
	male	0.06	0.03	0.00	0.12	
Disc.Male		0.14**	0.02	0.09	0.18	
Gaussian (ordi	nal-as-met	ric) model				
Intercept		4.04	0.04	3.96	4.11	
Gender	female	reference				
	male	0.06	0.04	-0.02	0.14	
Sigma.Female		0.39**	0.02	0.36	0.42	
Sigma.Male		0.28**	0.01	0.26	0.31	

Table 5 – Regression coefficients for the model with only a categorical variable (allowing the standard deviation to vary by group).

* CI stands for credible interval (based on quantiles).

** Values expressed on the logarithmic scale.

Table 5 shows the results of the fitted cumulative probit model with group-specific η_i and σ_i values for the underlying normal distributions of the ordinal variable, *Y* (the results for the gaussian (ordinal-as-metric) model are added for comparison). There is a difference in the approach that *brms* uses to model unequal standard deviation for the cumulative probit and the conventional Gaussian model. The SD of both is modelled on the log scale to constrain its value to be 0 or larger. The parameter related to the latent standard deviations is called *disc* (a contraction of 'discrimination'), following the conventions in item response theory. This parameter is not related to the standard deviation itself, but to the inverse of the SD, that is, $\sigma = 1/disc$. Consequently, the estimated SD for male is $\sigma = 1/\exp(0.14) = 0.87$ and $\sigma = \exp(0.28) = 1.32$ for the cumulative probit and gaussian (ordinal-as-metric) model, respectively (values from Table 5).



Fig. 8 – Density plot of the two underlying latent distributions for *TSV* with constant (top) and unconstraint (bottom) standard deviation for *Gender*.

Fig. 8 shows the density plot of the two underlying latent distributions for *TSV* given *Gender*, expressed in terms of the posterior means of each parameter. The underlying distribution for the reference category (i.e., female) is the standard normal, while the mean and SD for the other category (i.e., male) are estimated from the model. In Fig. 8 (bottom), the parameter value for *Male* is still the difference in means expressed on the latent variable scale for the reference group, but this time in terms of the SD of the reference group's latent variable (i.e., female). The SD for the two categories of *Gender* is not assumed to be the same, but it is allowed to vary. Also, the thresholds, { τ_k }, are on the scale of the reference category's latent variable and are assumed to be the same for the two categories of *Gender*.

Table 5 shows that the coefficient for *Disc.Male* is positive without zero overlapping the 95%-Cl. This indicates that the SD for male is smaller than the female (i.e., $\sigma_{male} = 0.87 < \sigma_{female} = 1$) and the evidence based on the data and the applied model is sufficient in terms of 'standard decision rules'. As such, in this sample, the standard deviations for TSV differ across the two categories of *Gender*.

Fig. 9 shows the marginal posterior distribution of the parameters (i.e., the means and standard deviations) and the effect sizes for the cumulative probit (green) and gaussian (ordinal-as-metric) (orange) models,

respectively. The cumulative probit model does not have a distribution for female because this is the reference category and its mean and standard deviation are fixed.



Fig. 9 – Posterior distributions for the model that include the variable *Gender*: cumulative probit (green) and gaussian (ordinal-as-metric) model (orange).

Here, the effect size is computed by dividing the difference of the means of the two groups by the pooled standard deviation given in Eq. (7):

$$\sigma_p = \sqrt{\frac{(n_1 - 1)\sigma_1^2 + (n_2 - 1)\sigma_2^2}{n_1 + n_2 - 2}}$$
 Eq. (7)

which is defined for two groups with unequal sample sizes (where n_1 and n_2 are the group-based sample sizes). In Fig. 9, the black line and dot at the bottom of each distribution represent the highest density interval (HDI) and the mode, respectively. The HDI is a way to summarise the distribution by defining an interval that spans over the distribution so that every point inside the interval has higher credibility than any point outside it. These intervals (i.e., the black lines) are defined here to span over 95% of the distribution; therefore, they represent the 95% HDIs.

Focusing on effect sizes and differences in means and standard deviations, two different results can be observed from Fig. 9. For the cumulative probit model, it can be seen that zero is outside the 95 % HDI for the effect size and the difference in means and SD. However, in the gaussian (ordinal-as-metric) model, zero is included in the 95 % HDIs for the effect size and the difference in SD while it is outside the 95 % HDI for the difference in means. As a consequence, in terms of 'standard decision rules', the two models convey different conclusions. While the cumulative probit model conveys a difference in effects size and difference in means for *Gender*, the gaussian (ordinal-as-metric) model does not.

3.3 Fitting a linear predictor

In this section, the continuous variable *Tair* was added to the previous model, that is, the model with the variable *Gender* and unconstraint standard deviation (i.e., where the standard deviation is allowed to vary by *Gender*). However, *Tair* was standardised before entering the model. Standardisation (i.e., subtracting the mean and dividing by its standard deviation) is done to improve the efficiency of MCMC sampling, that is, to reduce autocorrelation in the chains. In principle, it is unnecessary to standardise, but it would take more time for the chains to produce a reasonable, effective sample size. Furthermore, standardising does not change the parameter estimates. The results of fitting this model are presented in Table 6. Here can be seen that after adding *Tair.s* as a predictor, the upper and lower 95% CI (i.e., L-95% CI and U-95% CI) for the male coefficient of the gaussian (ordinal-as-metric) model does not include zero. Consequently, the two models now convey the same conclusions regarding *Gender*; both the Gaussian (ordinal-as-metric) and cumulative probit models show a difference between males and females.

		Estimate	Est. Error	L-95 % CI*	U-95 % CI*
Cumulative probi	t model				
Threshold 1, τ_1		-2.39	0.07	-2.53	-2.25
Threshold 2, τ_2		-1.15	0.04	-1.23	-1.08
Threshold 3, τ_3		-0.28	0.03	-0.34	-0.23
Threshold 4, τ_4		0.53	0.03	0.47	0.59
Threshold 5, τ_5		1.01	0.03	0.94	1.08
Threshold 6, τ_6		1.52	0.04	1.44	1.60
Gender	female	reference			
	male	0.09	0.03	0.03	0.14
Tair.s		0.34	0.01	0.31	0.37
Disc.Male		0.12**	0.02	0.07	0.16
Gaussian (ordi	nal-as-met	ric) model			
Intercept		4.01	0.03	3.95	4.08
Gender	female	reference			
	male	0.09	0.04	0.01	0.16
Tair.s		0.47	0.02	0.44	0.51
Sigma.Female		0.32**	0.02	0.28	0.35
Sigma.Male		0.22**	0.01	0.20	0.25

Table 6 - Regression coefficients for the model with a categorical and continuous variable (allowing the standard deviation to vary by group).

* CI stands for credible interval (based on quantiles). ** Values expressed on the logarithmic scale.

The marginal distribution of the standardised regression coefficient for Tair.s is shown in Fig. 10. As explained before, this is a standardised regression coefficient and represents a sort of effect size for air temperature. The two models give a different distribution for the coefficient, with a distinct mode and 95% HDIs. The coefficient of the cumulative probit model is expressed on the underlying latent scale, while the ordinalgaussian (ordinal-as-metric) coefficient refers to the ordinal scale. As a consequence, the gaussian (ordinalas-metric) coefficient for air temperature is overestimated.



Fig. 10 – (a) Standardised and (b) 'original' regression coefficient for air temperature for the cumulative probit (green) and gaussian (ordinal-as-metric) (orange).

3.4 Structured thresholds

In all the previous cumulative probit models, the thresholds { τ_k } were defined as 'flexible' providing the standard unstructured thresholds. However, restrictions such as equidistance can be imposed on the thresholds, which restricts the distance between consecutive thresholds to be of the same size (i.e., equally spaced). This allows assessing the assumptions that the subjects used the response scale (i.e., TSV) in such a way that the distance between adjacent response categories is the same, that is, $\tau_k - \tau_{k-1} = \text{constant}$ for $k \in \{1, ..., K\}$. The spacing of the equidistant threshold is plotted in Fig. 11.a. Here, the average distance between consecutive unstructured thresholds (i.e., $\frac{1}{k}\sum_{i=1}^{k}(\tau_k - \tau_{k-1})$) is also plotted (Fig. 11.b). It can be seen that zero is outside the 95% HDI for the difference between the spacing for structured and unstructured thresholds (Fig. 11.c), suggesting that, in terms of 'standard decision rules', the thresholds should not be approximated as equidistant.



Fig. 11 – Spacing for (a) structured and (b) unstructured thresholds and (c) their difference.

Furthermore, whether the restriction on the thresholds is warranted by the data can be assessed formally by comparing the relative fit of the computed models to the data. One method to assess relative fit is approximate leave-one-out cross-validation (LOOCV) [49], where smaller values indicate better fit. Table 7 shows the estimated LOO information criterion (LOOIC) for the two models and their differences. It can be seen that the cumulative probit model with unstructured thresholds has a significantly better fit (smaller LOOIC value) than the structured thresholds one since the difference in LOOIC (i.e., LOOIC.diff) is very large (more than 12 times the corresponding standard error, SE.diff). In the context of model selection, a LOOIC difference higher than twice its associated standard error suggests that the model with the lower LOOIC value fits the data significantly better.

Table 7 – Values of the Leave-One-Out Information Criterion (LOOIC) and their difference	for the
cumulative probit model with structured and unstructured thresholds.	

Model	10010	SF	1 OOIC diff*	SF diff**
inedel	20010	02	Looidiaiii	02.011
Cumulative probit model (unstructured thresholds)	19,449.2	100.0	0.0	0.0
Cumulative probit model (structured thresholds)	20,014.0	97.2	564.81	44.39

* LOOIC.diff is the difference between the two LOOIC scores.

** SE.diff is the standard error of the LOOIC.diff.

4 Discussion

This study aimed to highlight the ordinal-as-metric issue during the subjective thermal comfort data analysis. Here, the method used to assess the reliability of the two models (i.e., the cumulative probit and gaussian (ordinal-as-metric) approach) is the so-called posterior predictive checks, a commonly used technique in Bayesian analysis. In essence, after computing the posterior distribution of the parameters, many simulated data are generated and compared with the observed ones. Therefore, the posterior predictive check is used to look for 'systematic discrepancies that would be meaningful to address' [39]. This approach has the evident drawback of evaluating a model against the same data used to estimate its parameters. Unsurprisingly, the model predicts the data used to fit the parameters, but even this simple test fails when the model's

assumptions are severely violated. These systematic discrepancies are clearly shown in Section 3.1 when fitting the unconditional model for both the cumulative probit and gaussian (ordinal-as-metric) approach.

The influence of the statistical analysis on the conclusions has also been shown by Schweiker et al. [50]. In this study, the same thermal sensation votes were analysed with both linear and ordinal regression. The authors showed that the two statistical methods led to differences in the thermal conditions perceived as 'optimal' as well as between gender (i.e., female and male). However, compared with our study, important distinctions need to be made. To begin with, in Schweiker et al. [50], the analysis was carried out using mixedeffect models (linear and ordinal mixed-effect regression, specifically). This modelling strategy (also known as multilevel modelling) was applied to account for repeated measures (i.e., multiple observations for each subject). Moreover, the analysis was carried out within a frequentist framework. In our study, results are obtained by utilising a cumulative probit model in a Bayesian framework instead of the 'classic' frequentist approach. However, we emphasise that we are neither advocating a Bayesian approach as better than the frequentist approach nor that the cumulative probit model is the correct model to analyse ordinal data. Ordinal models in a frequentist framework provide another valid solution for analysing ordinal data (see ordinal package [51]). Also, other link functions besides probit are possible (e.g., logit or cloglog) and can be used. In addition, in Schweiker et al. [50], the linear mixed-effect regression model applied to ordinal data suggested a difference in means between genders. In contrast, the ordinal mixed-effect regression model led to the opposite conclusion. In our study, we obtained the opposite result: under given conditions (see Sections 3.2), the gaussian (ordinal-as-metric) approach inferred non-differences in gender, whereas the cumulative probit model showed a difference. Although these results contradict, they demonstrate the issue highlighted in this study: linear regression should not be used in place of ordinal regression to analyse ordinal data. It is essential to point out that we are not claiming that a difference between gender exists. In the literature (e.g., Refs [52] and [53]), many factors other than gender might lead to individual differences in thermal comfort - for example, age, circadian rhythm, physical disabilities, and fitness [52]. Here, the claim is about the difference in inference obtained from the same data (measured on an ordinal scale) analysed with two different methods (i.e., linear and ordinal regression). As shown in great detail by Liddell and Kruschke [24] (see also Section 1.2.3), analysing ordinal data with linear regression may, generally, lead to serious errors in inference. As such, it is not a problem concerning some specific variables (e.g., gender and air temperature in our illustrative example) but a more general issue.

One of the objectives of this work was to highlight that analysing ordinal data as they were continuous may lead to serious errors in inference (i.e., testing theoretical hypotheses). However, regression-type models can address different substantive goals and are therefore well suited to handle distinct purposes. For instance, Shmueli [54] separates a model's aim into descriptive, predictive, and causal explanations. Each of these distinct aims significantly impacts each step of the statistical modelling process and its consequences [54]. For instance, if the purpose is predictive modelling, the exact form of the data-generation process is not of interest, provided that it yields accurate predictions for the dependent variable. If the aim is inference (e.g., explanatory modelling), the estimate of the data-generation process is of interest, while making predictions of the dependent variable is not. In this study, it is not possible to draw specific conclusions regarding the accuracy of the prediction of TSVs. In this regard, Lai and Chen [55] analysed the predictive capability of linear regression compared with ordinal and multinomial regression. Using two separate datasets, the authors

demonstrated that ordinal and multinomial regression predicted around half of the individual TSVs, whereas the accuracy of the linear regression model was only around 20 to 40 %. Furthermore, chi-square statistics demonstrate that the ordinal and multinomial regression model outperformed the linear regression model in predicting TSV distributions.

In Section 3.4, the assumption of equidistance between the categories of the ASHRAE 7-point thermal sensation scale was checked. Fig. 11 shows that in terms of 'standard decision rules', the estimated thresholds $\{\tau_k\}$ should not be approximated as equidistant, suggesting that, in this sample, the TSV is not interval-scaled. This result was corroborated by the formal analysis presented in Table 7. Here the cumulative probit model with flexible (i.e., unstructured) thresholds fitted the data significantly better than the one with equidistant (i.e., structured) thresholds. It has to be noted that the distances between the thresholds are affected by the form of the latent distribution, which is defined by the link function used. For instance, if the thresholds were found to be equidistant under a latent symmetric distribution (e.g., probit or logit link), under a latent skew distribution (e.g., clog-log link), they will generally not be equidistant. However, since an underlying normal distribution (i.e., probit link) was used in our example, this issue did not affect the results. The inappropriateness of the assumption of equidistance between the categories of the ASHRAE 7-point scale was also found in Schweiker et al. [6]. From a large international collaborative questionnaire study (8225 questionnaires), the authors concluded that significant differences appeared between groups of participants in relation to the distances of the anchors of the thermal sensation scale (and other scales commonly used in thermal comfort studies). Nonetheless, we cannot claim that treating ordinal data as continuous always yields a different result or conclusion than treating them as ordinal. However, knowing in advance that a difference exists is impossible; a different result can be detected only if an ordinal analysis is also performed. Therefore, it is recommended to perform an ordinal analysis directly. Furthermore, since the arguments used by McIntyre [8], which are included in ISO 10551:2019 [1], to legitimise treating ordinal data from the ASHRAE 7-point scale as a continuous variable are disputable (see Section 1 for more detail), we strongly discourage the use of linear regression for analysing thermal comfort data measured on an ordinal scale. To improve the reliability of the results, we encourage researchers to use ordinal models. Moreover, ordinal models offer additional modelling possibilities that this paper has not discussed. For

instance, the proportional odds assumptions can be relaxed, and the threshold parameters can depend on some regression variables. In the context of thermal comfort studies, this can be translated to having, for example, different threshold parameters for gender or season.

4.1 Limitations

A fundamental aspect that is usually overlooked is the assumption of independence: residuals, and thus observations, are assumed to be independent. Non-independence can arise, for example, from temporal and spatial autocorrelation. When underlying spatial or temporal processes have the potential to impact a response, the data are autocorrelated – the closer the observations are in space or time, the more highly correlated they are. These sources of non-independence can be apparent or far less so. The response of one sampling unit influencing the response of other sampling units is an example of evident non-independence. The non-independence caused by non-measured confounding influences that vary spatially or temporally is less obvious to detect. Dealing with temporal (or spatial) autocorrelation or analysing temporal (or spatial)

trends is different. The former endeavours to deal with the lack of independence associated with temporal (or spatial) data, while the latter tries to model the effect of temporal (or spatial) patterns. During the data analysis stage, it was impossible to identify either spatial or temporal autocorrelation to test the assumption of independence because there was no temporal (e.g., subject ID and timestamp) or spatial (e.g., building ID) information available. As a consequence, this assumption was not checked. Given that the analysis was carried out for illustrative purposes only, this issue can be overlooked. However, in a real-world analysis, the assumption of independence needs to be verified. Furthermore, other issues, such as functional form misspecification, multicollinearity and omitted variable, were not considered during the analysis because they were outside the scope of this article. Nevertheless, when developing a model, depending on the aim of the study, these issues can play an important role and need to be considered. In addition, as stated in Section 2.2, the ASHRAE Global Thermal Comfort Database II does not distinguish between scales, and ordinal and continuous measurements are mixed. Consequently, there is a lack of homogeneity throughout the database that affects its integrity. Furthermore, there are conspicuous missing values in the ASHRAE Global Thermal Comfort Database II. This issue does not derive from the database itself but originates from the lack of explicit agreement on measuring the 'essential' variables in thermal comfort studies. If this lack of agreement continues, it could affect the future usefulness of the database because the information being added would continue to be non-homogeneous, thus limiting its usability and the new knowledge that could be extracted from it.

5 Conclusions and future perspectives

One of the aims of thermal comfort research is to establish the relationship between the thermal environment and the human sensation of warmth. Typically, this is accomplished by evaluating a subject's subjective thermal reaction to various temperature settings. Diverse rating scales are generally used to measure different aspects of thermal comfort, such as thermal sensation, thermal comfort, thermal preference, and thermal acceptability. While the problem of comparison of different scales (i.e., semantic equivalence) is an issue the thermal comfort research community is aware of, the use of reliable statistical methods to analyse the latter appears to be less discussed. In the thermal comfort domain, it is common practice to analyse subjective human thermal responses independently of how they have been measured. That is, the statistical analysis is unrelated to the modalities of the data that have been acquired. For example, even if measured on an ordinal scale, thermal sensation vote is generally treated as continuous and analysed with linear regression or other statistical tests that assume (conditional) normality. This approximation might be a concurrent factor to explain different results found in previous studies where, for example, gender was found to be or not an influential factor in explaining human responses to the thermal environment.

In this study, we first discussed why the arguments used in ISO 10551:2019 [1] to legitimise treating ordinal data from the ASHRAE 7-point thermal sensation scale as a continuous variable are disputable (see Section 1 for more detail). Secondly, to highlight the ordinal-as-metric issue during the subjective thermal comfort data analysis, the results obtained by utilising a cumulative probit and linear regression model were compared. Based on the analysis carried out on the dataset, the following conclusions can be drawn:

 Compared to the cumulative probit model, the linear regression model inferences non-differences in gender under given conditions.

- Compared to the cumulative probit model, the linear regression model distorts the estimate for the regression coefficient for the air temperature.
- The cumulative probit model shows that subjects used the ASHRAE 7-point thermal sensation in such a way that the distance between adjacent response categories is not the same; that is, they are not equidistant. Consequently, the cumulative probit model with flexible thresholds fitted the data significantly better than the one with equidistant thresholds.

As far as we know, in the field of thermal comfort research, the statistical issues highlighted in this paper are not usually mentioned because the modelling steps are rarely presented, and only the final model is described. However, this is a limitation because researchers can neither assess the reliability of the model nor completely understand the limits of its applicability. Furthermore, while not a primary output of this article, it emerged that there is a lack of homogeneity in the collection of common variables within the ASHRAE Global Thermal Comfort Database II. We recommend that guidelines be developed for defining specific variables to measure. Although there is generally no one-size-fits-all method (e.g., questionnaire) valid for all purposes, agreeing on a 'minimum set' of variables to be consistently measured, possibly with a standardised protocol, would undoubtedly benefit the thermal comfort research community.

Acknowledgements

Matteo Favero would like to thank the Research Centre on Zero Emission Neighbourhoods in Smart Cities (FME ZEN, Grant no. 257660) and the Research Council of Norway (Norges Forskingsrådet) for their support. Salvatore Carlucci and Antonio Luparelli would like to thank the European Commission for funding the Horizon 2020 project Collective Intelligence for Energy Flexibility (COLLECTIEF, Grant no. 101033683). The views and opinions expressed are those of the authors and do not reflect the views of the European Union.

Reference

- [1] ISO 10551, Ergonomics of the physical environment Subjective judgement scales for assessing physical environments, International Organization for Standardization, Geneva, Switzerland, 2019.
- [2] R.J. de Dear, G.S. Brager, D. Cooper, Developing an adaptive model of thermal comfort and preference -Final Report on ASHRAE RP-884, 1997.
- [3] P.O. Fanger, Thermal comfort: analysis and applications in environmental engineering., Danish Technical Press, Copenhagen, Denmark, 1970.
- [4] L.L. Thurstone, Attitudes Can Be Measured, Am J Sociol 33(4) (1928) 529–554. doi:10.1086/214483.
- [5] R. Likert, A technique for the measurement of attitudes, Archives of Psychology 22(140) (1932) 55.
- [6] M. Schweiker, M. André, F. Al-Atrash, H. Al-Khatri, R.R. Alprianti, H. Alsaad, R. Amin, E. Ampatzi, A.Y. Arsano, M. Azadeh, E. Azar, B. Bahareh, A. Batagarawa, S. Becker, C. Buonocore, B. Cao, J.-H. Choi, C. Chun, H. Daanen, S.A. Damiati, L. Daniel, R.D. Vecchi, S. Dhaka, S. Domínguez-Amarillo, E. Dudkiewicz, L.P. Edappilly, J. Fernández-Agüera, M. Folkerts, A. Frijns, G. Gaona, V. Garg, S. Gauthier, S.G. Jabbari, D. Harimi, R.T. Hellwig, G.M. Huebner, Q. Jin, M. Jowkar, J. Kim, N. King, B. Kingma, M.D. Koerniawan, J. Kolarik, S. Kumar, A. Kwok, R. Lamberts, M. Laska, M.C.J. Lee, Y. Lee, V. Lindermayr, M. Mahaki, U. Marcel-Okafor, L. Marín-Restrepo, A. Marquardsen, F. Martellotta, J. Mathur, I. Mino-Rodriguez, D. Mou, B. Moujalled, M. Nakajima, E. Ng, M. Okafor, M. Olweny, W. Ouyang, A.L.P.d. Abreu, A. Pérez-Fargallo, I. Rajapaksha, G. Ramos, S. Rashid, C.F. Reinhart, M.I. Rivera, M. Salmanzadeh, K. Schakib-Ekbatan, S. Schiavon, S. Shooshtarian, M. Shukuya, V. Soebarto, S. Subendri, M. Tahsildoost, F. Tartarini, D. Teli, P. Tewari, S. Thapa, M. Trebilcock, J. Trojan, R.B. Tukur, C. Voelker, Y. Yam, L. Yang, G. Zapata-Lancaster, Y. Zhai, Y. Zhu, Z. Zomorodian, Evaluating assumptions of scales for subjective assessment of thermal environments do laypersons perceive them the way, we researchers believe?, Energy and Buildings 211 (2020). doi:10.1016/j.enbuild.2020.109761.
- [7] M. Schweiker, X. Fuchs, S. Becker, M. Shukuya, M. Dovjak, M. Hawighorst, J. Kolarik, Challenging the assumptions for thermal sensation scales, Building Research & Information (2016) 1–18. doi:10.1080/09613218.2016.1183185.
- [8] D.A. McIntyre, Seven point scales of warmth, Build. Serv. Eng. Res. Technol. 45 (1978) 215–226.
- [9] A. Agresti, Categorical data analysis, third ed., Wiley, Hoboken, 2013.
- [10] J.-Y. Lee, E.A. Stone, H. Wakabayashi, Y. Tochihara, Issues in combining the categorical and visual analog scale for the assessment of perceived thermal sensation: Methodological and conceptual considerations, (2010) 282–290. doi:10.1016/j.apergo.2009.07.007.
- [11] G.A. Miller, The magical number seven, plus or minus two: some limits on our capacity for processing information, Psychol Rev 63(2) (1956) 81–97. doi:10.1037/h0043158.
- [12] Level of Measurement, in: W. Kirch (Ed.), Encyclopedia of Public Health, Springer Netherlands, Dordrecht, 2008, pp. 851–852.
- [13] S.S. Stevens, On the Theory of Scales of Measurement, Science 103(2684) (1946) 677–680. doi:10.1126/science.103.2684.677 PMID - 17750512.
- [14] S.S. Stevens, On the Averaging of Data, Science 121(3135) (1955) 113–116. doi:10.1126/science.121.3135.113 PMID - 13225751.
- [15] P.F. Velleman, L. Wilkinson, Nominal, Ordinal, Interval, and Ratio Typologies are Misleading, Am Statistician 47(1) (1993) 65–72. doi:10.1080/00031305.1993.10475938.
- [16] J.W. Tukey, Data analysis and behavioral science or learning to bear the quantitative man's burden by shunning badmandments, The collected works of John W. Tukey1986, pp. 391–484.
- [17] F.M. Lord, On the Statistical Treatment of Football Numbers, Am Psychol 8(12) (1953) 750–751. doi:10.1037/h0063675.
- [18] F. Mosteller, J.W. Tukey, Data analysis and regression : a second course in statistics, Addison-Wesley, London, 1977.
- [19] N.R. Chrisman, Rethinking Levels of Measurement for Cartography, Cartogr Geogr Inf Sc 25(4) (1998) 231–242. doi:10.1559/152304098782383043.
- [20] P. Lavrakas, Encyclopedia of Survey Research Methods, (2008). doi:10.4135/9781412963947.n461.
- [21] H.B. Rijal, M.A. Humphreys, J.F. Nicol, Towards an adaptive model for thermal comfort in Japanese offices, Building Research & Information (2017) 1–13. doi:10.1080/09613218.2017.1288450.
- [22] R. McElreath, Statistical Rethinking: A Bayesian Course with Examples in R and Stan, second ed.2020.
- [23] P.-C. Bürkner, M. Vuorre, Ordinal Regression Models in Psychology: A Tutorial, Adv Methods Pract Psychological Sci 2(1) (2018) 77–101. doi:10.1177/2515245918823199.
- [24] T.M. Liddell, J.K. Kruschke, Analyzing ordinal data with metric models: What could possibly go wrong?, J Exp Soc Psychol 79 (2018) 328–348. doi:10.1016/j.jesp.2018.08.009.
- [25] F. Zhang, R.J. de Dear, Impacts of demographic, contextual and interaction effects on thermal sensation— Evidence from a global database, Building and environment 162 (2019) 106286. doi:10.1016/j.buildenv.2019.106286.
- [26] R.J. de Dear, G.S. Brager, Developing an adaptive model of thermal comfort and preference, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., San Francisco, CA, 1998, pp. 27–49.

- [27] J. Kolarik, J. Toftum, B.W. Olesen, A. Shitzer, Occupant responses and office work performance in environments with moderately drifting operative temperatures (RP-1269), HVAC and R Research 15(5) (2009) 931–960. doi:10.1080/10789669.2009.10390873.
- [28] R.J. de Dear, J. Kim, C. Candido, M. Deuble, Adaptive thermal comfort in Australian school classrooms, Building Research & Information 43(3) (2015) 383–398. doi:10.1080/09613218.2015.991627.
- [29] F. Zhang, R.J. de Dear, C. Candido, Thermal comfort during temperature cycles induced by direct load control strategies of peak electricity demand management, Building and Environment 103 (2016) 9–20. doi:10.1016/j.buildenv.2016.03.020.
- [30] ASHRAE, ASHRAE Handbook: Fundamentals (SI), Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., Atlanta, GA, 2017.
- [31] CEN EN 15251, Indoor environmental input parameters for design and assessment of energy performance of buildings addressing indoor air quality, thermal environment, lighting and acoustics, European Committee for Standardization, Brussels, Belgium, 2007.
- [32] CEN EN 16798-1, Energy performance of buildings Ventilation for buildings Part 1: Indoor environmental input parameters for design and assessment of energy performance of buildings addressing indoor air quality, thermal environment, lighting and acoustics - Module MI-6, European Committee for Standardization, Brussels, Belgium, 2019.
- [33] ANSI/ASHRAE Standard 55, Thermal environmental conditions for human occupancy, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, GA, 2020.
- [34] S. Crosby, A. Rysanek, Predicting thermal satisfaction as a function of indoor CO2 levels: Bayesian modelling of new field data, Building and Environment 209 (2022) 108569. doi:10.1016/j.buildenv.2021.108569.
- [35] J. Langevin, J. Wen, P.L. Gurian, Modeling thermal comfort holistically: Bayesian estimation of thermal sensation, acceptability, and preference distributions for office building occupants, Building and Environment 69 (2013) 206–226. doi:10.1016/j.buildenv.2013.07.017.
- [36] L.T. Wong, K.W. Mui, C.T. Cheung, Bayesian thermal comfort model, Building and Environment 82 (2014) 171–179. doi:10.1016/j.buildenv.2014.08.018.
- [37] A. Gelman, J.B. Carlin, H.S. Stern, D.B. Dunson, A. Vehtari, D.B. Rubin, Bayesian Data Analysis, third ed.2013.
- [38] L. Wasserman, All of Statistics: a Concise Course in Statistical Inference, Springer, New York, NY2004.
- [39] J.K. Kruschke, Doing Bayesian Data Analysis, second ed.2015.
- [40] J.K. Kruschke, T.M. Liddell, Bayesian data analysis for newcomers, Psychon B Rev 25(1) (2018) 155–177. doi:10.3758/s13423-017-1272-1.
- [41] V. Földváry Ličina, T. Cheung, H. Zhang, R.J. de Dear, T. Parkinson, E. Arens, C. Chun, S. Schiavon, M. Luo, G.S. Brager, P. Li, S. Kaam, ASHRAE Global Thermal Comfort Database II, (2018). doi:10.6078/D1F671.
- [42] V. Földváry Ličina, T. Cheung, H. Zhang, R.J. de Dear, T. Parkinson, E. Arens, C. Chun, S. Schiavon, M. Luo, G.S. Brager, P. Li, S. Kaam, M.A. Adebamowo, M.M. Andamon, F. Babich, C. Bouden, H. Bukovianska, C. Candido, B. Cao, S. Carlucci, D.K.W. Cheong, J.-H. Choi, M. Cook, P. Cropper, M. Deuble, S. Heidari, M. Indraganti, Q. Jin, H. Kim, J. Kim, K. Konis, M.K. Singh, A. Kwok, R. Lamberts, D. Loveday, J. Langevin, S. Manu, C. Moosmann, F. Nicol, R. Ooka, N.A. Oseland, L. Pagliano, D. Petráš, R. Rawal, R. Romero, H.B. Rijal, C. Sekhar, M. Schweiker, F. Tartarini, S.-i. Tanabe, K.W. Tham, D. Teli, J. Toftum, L. Toledo, K. Tsuzuki, R.D. Vecchi, A. Wagner, Z. Wang, H. Wallbaum, L. Webb, L. Yang, Y. Zhu, Y. Zhai, Y. Zhang, X. Zhou, Development of the ASHRAE Global Thermal Comfort Database II, Building and Environment 142 (2018) 502–512. doi:10.1016/j.buildenv.2018.06.022.
- [43] M. Indraganti, R. Ooka, H.B. Rijal, G.S. Brager, Adaptive model of thermal comfort for offices in hot and humid climates of India, Building and Environment 74 (2014) 39–53. doi:10.1016/j.buildenv.2014.01.002.
- [44] R Core Team, R: A language and environment for statistical computing, R Foundation for Statistical Computing, Vienna, Austria, 2022. <u>https://www.R-project.org/</u>
- [45] RStudio Team, RStudio: Integrated Development Environment for R, RStudio, PBC, Boston, MA, 2022. http://www.rstudio.com/
- [46] P.-C. Bürkner, brms: An R Package for Bayesian Multilevel Models Using Stan, Journal of Statistical Software 80 (2017) 1–28. doi:10.18637/jss.v080.i01.
- [47] H. Wickham, ggplot2: Elegant Graphics for Data Analysis, Springer-Verlag, New York, 2016.
- [48] Matthew Kay, tidybayes: Tidy Data and Geoms for Bayesian Models, 2021. http://mjskay.github.io/tidybayes/
- [49] A. Vehtari, A. Gelman, J. Gabry, Practical Bayesian model evaluation using leave-one-out cross-validation and WAIC, Statistics and Computing 27(5) (2017) 1413–1432. doi:10.1007/s11222-016-9696-4.
- [50] M. Schweiker, K. Schakib-Ekbatan, X. Fuchs, S. Becker, A seasonal approach to alliesthesia. Is there a conflict with thermal adaptation?, Energy and buildings 212 (2020) 109745. doi:10.1016/j.enbuild.2019.109745.
- [51] R.H.B. Christensen, ordinal---Regression Models for Ordinal Data, 2019. <u>https://CRAN.R-project.org/package=ordinal</u>

- [52] Z. Wang, R.J. de Dear, M. Luo, B. Lin, Y. He, A. Ghahramani, Y. Zhu, Individual difference in thermal comfort: A literature review, Building and environment 138 (2018) 181–193. doi:10.1016/j.buildenv.2018.04.040.
- [53] S. Karjalainen, Thermal comfort and gender: a literature review, Indoor Air 22(2) (2012) 96–109. doi:10.1111/j.1600-0668.2011.00747.x.
- [54] G. Shmueli, To Explain or to Predict?, Statistical Science 25(3) (2010) 289–310. doi:10.1214/10-STS330.
- [55] D. Lai, C. Chen, Comparison of the linear regression, multinomial logit, and ordered probability models for predicting the distribution of thermal sensation, Energy and buildings 188–189 (2019) 269–277. doi:10.1016/j.enbuild.2019.02.027.

Paper III

Applied Energy 325 (2022) 119803

Contents lists available at ScienceDirect



Applied Energy

journal homepage: www.elsevier.com/locate/apenergy

Human-in-the-loop methods for occupant-centric building design and operation

Matteo Favero^{a,*}, Jan Kloppenborg Møller^b, Davide Calì^b, Salvatore Carlucci^c

^a Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway

^b Department of Applied Mathematics and Computer Science, Technical University of Denmark, DTU, Lyngby, Denmark

^c Energy, Environment and Water Research Center, The Cyprus Institute, Nicosia, Cyprus

HIGHLIGHTS

• Multilevel modelling was applied to predict subjects' thermal preference vote in a dynamic thermal environment.

The beta and ordinal mixed-effects models are both valid alternatives for modelling subjects' thermal preference votes.

• Two procedures were used to implement subjects' feedback within the occupant-centric building design and operation paradigm.

• The population-averaged procedure is suitable for the building design phase.

• The cluster-specific procedure is appropriate for the building operation phase.

ARTICLEINFO

Keywords: Subjective thermal preference Occupant-centric design and operation Human-in-the-loop Multilevel modelling Statistical thinking

ABSTRACT

A comfortable indoor environment should be one of the main services buildings provide. However, technical building systems are typically designed and operated according to fixed set-point temperatures determined by the 'one-size-fits-all' principle assuming universal thermal comfort requirements, which has been questioned in the last fifty years. Designing and implementing comfortable set-point modulations that consider occupant feedback would be beneficial in terms of increasing comfort, potentially reduce energy consumption and significantly support the clean energy transition. An exploratory study aimed at predicting the thermal preferences of human subjects exposed to a dynamic thermal environment is presented. Using data acquired from a laboratory experiment where subjects were exposed to precisely controlled thermal ramps in an 'office-like' climatic chamber, cluster-specific and population-averaged methods are designed to handle the group-level residual during the prediction of the thermal preference votes. The results show that both approaches are valid strategies for modelling thermal preference votes and are effective in supporting a concrete occupant-centric building design and the building's operation. Furthermore, the population-averaged approach is suitable for the occupant-centric building design phase, where the target is an 'average' occupant. The cluster-specific method is best suited to meet the needs of a specific occupant and is suitable for implementation in the operational phase of the building.

EN 15251:2007 [4], propose requirements based on Fanger model (beyond also including other approaches), which solves the heat balance

equations between the human body and its surroundings, represented as a uniform environment. Fanger defined the 'Predicted Mean Vote'

(PMV) as an index that predicts the mean thermal sensation vote on a

standard scale for a large group of persons exposed to a given combi-

nation of metabolic activity level, clothing insulation and four thermal

environmental variables characterising the indoor space: dry-bulb air

temperature, mean radiant temperature, air velocity and relative

AppliedEnerg

1. Introduction

A comfortable indoor environment should be one of the primary services buildings provide. Nowadays, all thermal comfort standards include recommendations concerning the indoor thermal conditions for both the design and operation phases of buildings. Currently, the most frequently cited thermal comfort standards, namely ASHRAE 55:2020 [1], ISO 7730:2005 [2] and EN 16798–1:2019 [3], which was formerly

https://doi.org/10.1016/j.apenergy.2022.119803

Received 29 April 2022; Received in revised form 3 August 2022; Accepted 5 August 2022

0306-2619/© 2022 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Corresponding author at: 7a Høgskoleringen, 7034 Trondheim, Norway. E-mail address: matteo.favero@ntnu.no (M. Favero).

Nomenclature

I	The identity matrix
k	The category of the dependent variable
Μ	The total number of the simulated random effect
n	The number of events
и	The vector of the random effects
Х	The design matrix of the fixed effects
х	The vector of the fixed effects
x	Indicates a generic variable
Y	The vector or matrix of the response variable
Y	Indicates a random value of the response variable (usually
	accompanied by a subscript)
Ζ	The design matrix of the random effects
β	The vector of parameters of the fixed effects
β	A scalar indicating a parameter of the fixed effects (usually
	accompanied by a subscript)
γ	The vector of the cumulative probabilities
γ	A scalar indicating a cumalive probability (usually
	accompanied by a subscript)
η	The vector of the linear predictor term
η	A scalar indicating the linear predictor term (usually

humidity [5]. The PMV model is generally considered a static model because it is only suited for predicting thermal sensation under a steady state or slowly changing indoor conditions (i.e., rate of change lower than 2.0 K/h) [2]. Based on the PMV, Fanger introduced another index called the 'Predicted Percentage of Dissatisfied' (PPD) to establish a quantitative prediction of the percentage of thermally dissatisfied people. Additionally, thermal dissatisfaction can be caused by other local factors (e.g., drafts) and is known as local discomfort. For naturally conditioned spaces, ASHRAE 55:2020 [1] prescribes the use of the adaptive model, while EN 16798-1:2019 [3] suggests it only as a possible alternative to the Fanger approach. In 1973, in the first adaptive comfort paper published, Nicol and Humphreys [6] hypothesised the presence of 'control mechanisms' (feedback loops) between the occupants' thermal comfort perception and their behaviour in buildings. After this, research activity on the topic remained muted until the turn of the century, when intensification of research interest occurred, and several papers were published (e.g., [7,8]). The hypothesis of adaptive thermal comfort predicts that contextual factors and past thermal history modify occupant's thermal expectations and preferences [9]. As a result, people in warm climate zones would prefer higher indoor temperatures than people living in cold climate zones, which contrasts with the assumptions underlying comfort standards based on the PMV/PPD model [9]. Before inclusion in the standard EN 15251, the adaptive approach was also used by McCartney and Nicol [10] to develop an adaptive control algorithm (ACA) that was intended to be 'an alternative to fixed temperature setpoint controls within buildings' and 'was also tested in two air-conditioned buildings as part of the SCATs project' with promising results consisting in energy saving without compromising occupants' perceived thermal comfort [10]. In current standards, Fanger's PMV/PPD model is the prerogative of mechanically heated and/or cooled buildings, while the adaptive thermal comfort model is reserved for free-running buildings. EN 16798-1:2019 [3], citing ISO 7730:2005 [2], defines different categories of indoor environments for mechanically heated and cooled buildings, namely I, II, III and IV, with category I being the most stringent in terms of the management of interior conditions. An upper PPD bound is associated with each of the four PMV ranges (and therefore each category level), varying from 6 % to 25 % (see Table 1). A similar schema is present in ASHRAE 55:2020 [1], where the 'acceptable thermal environment for general comfort' is defined as -0.5 < PMV < + 0.5, corresponding to category II in

	accompanied by a subscript)
μ	The vector of the expected values
μ	A scalar indicating the expectd value (usually accompanied
	by a subscript)
π	The vector of probabilities
π	A scalar indicating a probability (usually accompanied by a
	subscript)
σ^2	The variance
σ	The standard deviation
Σ	The variance–covariance matrix
τ	The latent threshold parameter
ϕ	The precision parameter
Subscript	S
d	Indicates the d^{th} day
i	Indicates the i th observation
k	Indicates the k^{th} category of the dependent variable
n	Indicates the dimension of a square matrix
р	Indicates the <i>p</i> th participant
r	Indicates the r th thermal ramp

Table 1

Default design categories for mechanically heated and cooled buildings.

Category	PMV	PPD (%)
I	-0.2 < PMV < +0.2	<6
П	-0.5 < PMV < +0.5	<10
III	-0.7 < PMV < +0.7	<15
IV	-1.0 < PMV < +1.0	<25

Table 1.

The categories described in Table 1 are recommended for designing mechanically heated and cooled buildings. In practice, assuming the occupants' clothing insulation and metabolic activity levels and the relative humidity and air velocity of the environment, the PMV ranges can be represented in terms of acceptable operative temperature ranges. Maintaining a tight PMV or temperature range demands more energy than allowing a wider operative temperature range. A large increase in energy consumption could only be justified if a tightly controlled thermal environment were to be more comfortable than one under less control. Arens et al. [11] investigated this specific aspect by examining the acceptability of the temperature ranges associated with categories I, II and III of the EN15251:2007 [4] standard via three databases on occupant satisfaction (specifically, the ASHRAE RP 884 [12], SCATs [10] and Berkeley City Center Project [13] databases). The authors found that in terms of satisfaction, building occupants do not benefit from an indoor environment that is tightly controlled (i.e., a category I environment). Furthermore, they identified only a small difference in satisfaction between categories II and III. Consequently, designing and controlling indoor environments, such as office buildings, following the strict specifications suggested, for example, for category I of EN 15251:2007 [4], is unreasonable [11]. However, the real issue can be traced back to using the PMV/PPD indexes as the theoretical basis for building control and operation in the first place. In a review paper, de Dear et al. [14] state that many rigorous field studies (e.g., [15-17]), founded by ASHRAE in the 1980 s and 90 s, have clearly found the 'onesize-fits-all' approach to achieving a universally comfortable environment 'to be a failure'. The main issue is that the PMV index represents a steady-state thermal comfort model that predicts the mean thermal sensation for a large group of people. Therefore, it fails to account for

dynamic and non-uniform thermal environments as well as individual differences. In practice, the indoor environment frequently changes abruptly across buildings or between various parts within a single building. For instance, manually- operated thermostats, windows and window shades can result in considerable and non-systematic changes across the indoor environment. Automatic controllers exhibit, to a lesser degree, a similar behaviour. Moreover, activity modifies an individual's basal metabolic rate over time, and the addition or removal of clothes affects their heat balance. In other words, the steady-state assumption at the root of the Fanger comfort model is very often violated (Ref [18] citing [19]). Building temperature ranges should therefore be based on real-time empirical evidence regarding the needs of the occupants. Measures to improve occupant feedback capabilities should be included in the routine control and operation of the building as well as specified in building designs. For example, Park et al. [20] analyse occupantcentric control (OCC) research, focusing on field-implementation case studies in buildings under realistic conditions. The authors offer a methodological analysis focusing on the various strategies utilised to integrate OCC into existing building systems. Another example can be found in Jung and Jazizadeh [21]. In this review, the authors, distinguishing between simulations and field evaluations, proposed a taxonomy for human-in-the-loop HVAC operations and reviewed methods for integrating human dynamics to control HAVC.

Furthermore, implementing dynamic modulations of the set-point temperature might help time-shift and/or reduce peak space heating and cooling needs, improving the energy flexibility of buildings.

In this context, individual differences between people play an essential role. When it comes to thermal comfort, individual differences result in situations where distinct people perceive the same thermal environment in different ways (i.e., they have inter-individual differences) and/or when the same individual assesses the same environment differently at different times or in different situations (i.e., this individual presents intra-individual differences). Humphreys and Nicol [22] suggested that inter-individual differences encompass both temperature differences to be considered neutral and differences in the interpretation of the semantic scale categories. In contrast, intra-individual differences refer to personal judgments that differ from time to time. Machine learning/data-driven algorithms used for predicting individual comfort responses have exploded in popularity recently and include the classification tree (e.g., [23]), random forest (e.g., [24]), gradient boosting method (e.g., [25]), support vector machine (e.g., [26]), Gaussian process classification (e.g., [27]) and artificial neural networks (e.g., [28]). Although these techniques appear to have the potential to improve prediction ability at the level of a single building occupant, their inherent character as 'black box' models renders them fundamentally unfit to explain their outputs. In predictive modelling, direct interpretability regarding the relationship between the predictors (Xs) and the outcome of interest (Y) is not required; however, transparency is desirable.

In summary, there is a need for a reassessment of how buildings are designed and operated. Implementing comfortable set-point modulations in buildings that consider occupant feedback would be beneficial to comfort, potentially reduce energy consumption and significantly support the clean energy transition. As a consequence, HVAC design and operation should consider both the inter- and intra-individual differences among and within the occupants, respectively.

1.1. Research aim

A paradigm shift from 'set-point-based' control to 'perception-based' human-in-the-loop control of buildings is necessary to increase comfort, reduce energy consumption, and support the transition to clean energy. However, considering these aspects in the building design phase would also be beneficial.

The present research is an exploratory study aimed at predicting the thermal preference vote of human subjects exposed to a dynamic thermal environment. Therefore, the objective of this work is to develop a model for prediction (i.e., forecasting new data points), not for inference (i.e., testing theoretical hypotheses). Here, the data-generation process is viewed as a 'transparent' tool for developing good predictions. However, the modelling strategy does not aim to model the effect of temporal patterns directly but rather to account for them (i.e., account for the lack of independence associated with temporal data). The model is developed using data acquired from a laboratory experiment, where subjects were exposed to precisely-controlled thermal ramps in an 'office-like' climatic chamber.

2. Methodology

2.1. Data acquisition

The dataset used in this study comes from an experimental study conducted by Favero et al. [29] in the ZEB Test Cell Laboratory on the Norwegian University of Science and Technology (NTNU) premises (Trondheim campus) between September 2019 and January 2020. Thirty-eight participants (29 females and 9 males) were recruited from the university campus to participate in a randomised crossover trial, that is, a longitudinal study, in which they were subjected to a randomised sequence of thermal exposures (i.e., thermal ramps). Two identical climatic chambers, furnished like typical single offices, were used to recreate the changes in the environment induced by thermal ramps. Space heating and cooling were provided by a constant air-volume system that supplied 100 % fresh air from outside that was distributed by a 2 m-long perforated fabric tube installed at the ceiling. The operative temperature set-point of 22.0 \pm 1.0 $^\circ$ C was determined using the thermal comfort limit for winter established for Category A of ISO 7730–2005 [2]. The rates of the temperature changes were: (i) \pm 4.4 K/ h, (ii) \pm 3.4 K/h, (iii) \pm 2.2 K/h and (iv) \pm 1.4 K/h, as recommended by ASHRAE 55:2017 [30].

During the experiment, participants were not asked to perform any specific tasks and were allowed to carry out their typical office activities. Nevertheless, the subjects were required to fill out computer-based questionnaires at scheduled intervals. By means of graphic categorical scales (see Fig. 1), these questionnaires were used to assess perception, evaluation, preference, and acceptability of the thermal environment. Further details on the experimental set-up, as well as the experimental conditions and procedure, can be found in Favero et al. [29].

2.2. Statical modelling

Multilevel models (also commonly referred to as mixed or hierarchical models) are a regression-based approach to dealing with clustered and nested data [31]. When individuals form groups or clusters, it is reasonable to expect that two randomly selected individuals from the same group will tend to be more alike than two individuals selected from different groups. Following similar reasoning, measurements taken on the same individual on different occasions will be more highly correlated than measurements taken from different individuals. Therefore, whenever data are clustered and/or nested, the assumption of independent errors is violated.

This experimental study examines a mixture of hierarchical and crossed relationships. As shown in Fig. 2, different measurements on the thermal environment (level 1) are nested within experiment conditions (level 2), which, in turn, are cross-classified by participant and day (level 3). It is essential to mention that the multilevel structure defined here is not the property of a model but rather the property of the experimental/study design, which is then reflected in the data, which the model then encapsulates.

Within the multilevel framework, there are different modelling strategies that can be used. In this study, two different modelling strategies were applied, namely the beta mixed-effects model (a beta model including random effects) and the ordinal mixed-effects model (an





At this precise moment, do you find the room temperature ... ? Please mark on the scale.

riease mark on the scale.



How do you judge the room

Please mark on the scale.

temperature on a personal level?

Pay attention to the dichotomy between acceptable and not acceptable.

At this moment. Would you prefer the room temperature to be ... ? Please mark on the scale.



Fig. 1. Subjective scales used to assess perception, evaluation, preference, and acceptability of the thermal environment. Note. The numerical values of the scale were not shown to the participants during data collection.

ordinal model including random effects). These two approaches are described in the following sections.

2.2.1. Beta mixed-effects model

Generalised linear models¹ (GLMs) constitute a large class of models

where the conditional distribution of the response variable Y_i is assumed to follow an exponential family distribution with mean μ_i . The latter is assumed to be some function of $\eta_i = \mathbf{x}_i^T \boldsymbol{\beta}$, where \mathbf{x}_i is the vector of covariates for the i^{th} observation and $\boldsymbol{\beta}$ is the respective vector of parameters to be estimated. However, one of the assumptions behind the model is the independence of the errors, which cannot be assumed whenever data are clustered and/or nested (see Section 2.2). To deal with dependent errors, GLMs can be extended to generalised linear mixed models, in which the linear predictor η contains random effects (i. e., Zu) in addition to the fixed effects (i.e., $X\beta$).

¹ This class of models is not to be confused with general linear model which usually refers to linear regression models – generally assuming a normal conditional distribution of the response.



Fig. 2. Schematic of the three-level hierarchical study: repeated measures within experimental conditions cross-classified by participant and day.

In this study, the conditional distribution of the response variable *Y* is assumed to follow a beta distribution, where its mean μ is linked to linear predictor η through a logit function:

$$Y \sim \text{Beta}(\mu, \phi)$$

$$\text{Logit}(\mu) = \eta$$

$$\eta = X\beta + Zu$$

$$u \sim \text{Normal}(0, \Sigma)$$
(1)

where ϕ is the precision parameter, and *X* (whose row *i* is $\mathbf{x}_i^{\mathrm{T}}$ and contains the *i*th observation of the covariates) and *Z* are the design matrices for the fixed and random effects, respectively. The reader is referred to Appendix A for more details about the mathematical notation and a practical example. For the sake of clarity and brevity, the beta mixed-effects model with the logit link will hereafter be referred to as simply the beta model.

2.2.2. Ordinal mixed-effects model

Cumulative link models (CLMs) belong to the ordinal regression model class and can be performed using GLMs. A cumulative model is used when latent variable representation is desired. Here, the dependent variable \tilde{Y} is the categorisation of a latent (unobservable) continuous variable \tilde{Y} . Therefore, there are some latent thresholds parameters τ_k , with $k \in \{1, ..., K\}$, that divide the values of \tilde{Y} into K + 1 bins, that is, the observable ordered categories of Y. CLMs assume independence of errors and are not suited for modelling clustered and/or nested data. Their extensions for dealing with the dependent errors are the cumulative link mixed models (CLMMs).

In a CLMM, the conditional distribution of the response variable Y_i for the i^{th} observation is assumed to follow a multinomial distribution with probability vector $\pi_i = {\pi_{i1}, ..., \pi_{ik}}$, where $\pi_{ik} = \Pr(Y_i = k)$. The cumulative probability corresponding to π_{ik} is $\gamma_{ik} = \Pr(Y_i \le k)$; hence, $\gamma_{ik} = \pi_{i1} + ... + \pi_{ik}$. The cumulative probabilities are then mapped to the real numbers through a link function. In this study, the logit function was chosen as that link function. The mathematical formulation of the model can be written as:

$Y \sim Multinomial(n, \pi)$	
$\text{Logit}(\boldsymbol{\gamma}_k) = 1\boldsymbol{\tau}_k - \boldsymbol{\eta}$	(2)
$\eta = X\beta + Zu$	(2)
$\boldsymbol{u} \sim \text{Normal}(\boldsymbol{0} \boldsymbol{\Sigma})$	

where the τ_k are the thresholds parameters and η is the linear predictor term with a fixed effect component (i.e., $X\beta$) without an intercept² and a random effect component (i.e., Zu). The reader is referred to Appendix A for more details about the mathematical notation and a practical example. For the sake of clarity and brevity, the ordinal mixed-effects model with the logit link will hereafter be referred to as simply the ordinal model.

Applied Energy 325 (2022) 119803

2.2.3. Computing predictions using a multilevel model

Research setting aims to make predictions for certain values of x (e. g., adjusting the values of one x at a time or for combinations of x-values that reflect 'typical' persons) rather than calculating a probability for each individual in the sample. However, for a multilevel model, the treatment of the group-level residual u (i.e., group random effect) for these 'out-of-sample'³ predictions must be considered.

In this study, two different procedures were used to handle the group-level residual during prediction. For the ordinal model, the first procedure consisted of holding the group-level residual at its mean of zero and calculating the probabilities for some specific *x*-values. It should be noted that the calculated predictions are not the mean response probabilities for the specific *x*-value because γ_{ik} is a nonlinear function of $u(\text{as is } \pi_i)$. However, since u is assumed to be normally distributed with mean = median = 0, the $\text{Logit}(\gamma_{ik})$ for $x = x^*$ and u = 0 is equal to the median γ_{ik} for $x = x^*$ across groups. This is the case because the logit transformation does not affect the rank order of the observations. The response probabilities thus calculated have a cluster-specific interpretation.

The second procedure outlined a simulation-based approach, which consisted of the following steps:

- i. Generate M values for the random effect u from the $\operatorname{Normal}(0,\Sigma)$ distribution;
- ii. For each simulated value (m = 1, ..., M) calculate, for the given x-value, the cumulative response probabilities for each K+1 ordered categories of Y;
- iii. Compute the mean of the *M* cumulative response probabilities calculated in (ii) for each of the K+1 ordered categories of *Y*;
- iv. Repeat steps (i) (iii) for a different *x*-value.

The generated *M* values for the random effect should be a large number, here fixed at $1 \cdot 10^4$. This approach results in probabilities with a population-averaged interpretation (i.e., averaged across experimental conditions, participants and days). The same two procedures were applied to the beta model, with the difference that the prediction was not a vector (i.e., probabilities of voting in each category) but rather a single number (i.e., predicted mean).

2.3. Data pre-processing and analysis

A total of 314 thermal ramps were performed, for a total of 1522 votes. There were three missing values for thermal perception and evaluation, six for thermal preference and 14 for thermal acceptability. However, since only thermal preference was of interest in this study, only the missing values for the latter were eliminated. As a result, the final sample size was reduced to 1516 observations. Fig. 3(a) illustrates the distribution of the thermal preference votes.

As shown in Fig. 1, thermal preference ratings were measured using a graphical categorical scale. Participants could cast their vote by placing a diagonal line anywhere within the limits of the scale (i.e., within 'lower' and 'higher'). Consequently, the resulting distribution of votes is on a continuous, but bounded, scale (Fig. 3(a)). In this study, the conditional distribution of the response (i.e., $Y|\mu$) is assumed to follow a beta distribution. However, since any beta distribution's probability density function (pdf) is defined only on the interval (0,1), the dependent variable needs to be rescaled. Therefore, the thermal preference votes were scaled so that the values at the boundary of the scale, -1 (i. e., 'lower') and +1 (i.e., 'higher'), were mapped to + 0.001 and + 0.999, respectively.

As an alternative approach to the beta model, the ordinal model was

 $^{^2}$ Omitting the intercept term allows the full set of thresholds $\tau_1,...,\tau_k$ to be identified.

³ The term 'out-of-sample' is used here to highlight that fixing variables at some values (e.g., their mean) may not reflect any actual person in the sample.



Fig. 3. Frequency distributions of (a) the thermal preference votes and (b) its categorisation. Note. The dotted lines represent the thresholds used for the categorisation, that is, -0.6, -0.2, +0.2, and +0.6.

chosen. However, the ordinal model requires the dependent variable to be categorical, which entails categorising the thermal preference votes. Therefore, the votes were binned into five categories according to the thresholds -0.6, -0.2, +0.2, and +0.6:

- a) thermal preference votes < -0.6 were defined as 'lower';
- b) $-0.6 \leq$ thermal preference votes <-0.2 were defined as 'slightly lower';
- c) $-0.2 \leq$ thermal preference votes $\leq +0.2$ were defined as 'without change';
- d) +0.2 < thermal preference votes $\leq +0.6$ were defined as 'slightly higher';
- e) thermal preference votes > +0.6 were defined as 'higher'.

The frequency distribution of the resulting bins can be observed in Fig. 3(b).

In a regression-type model, the shape of the distribution of a predictor has no direct impact on the model itself. Therefore, there is no real a priori need to transform or categorise a predictor based on its distribution. Of greater importance is the correlation between predictors (i.e., whether or not there is collinearity⁴). There are two types of collinearity: structural and data-based collinearity. The former is a mathematical artefact originating from composing new predictors from other predictors, such as powers (higher-order terms) or products (interaction terms) of predictors. The latter is a 'property' of the data itself, which can be the result of, for example, a poorly designed experiment. To manage the first type of collinearity, predictors lacking a meaningful zero were centred by their grand mean; it should be noted that this standardisation procedure can facilitate the interpretation of the model [32]. Data-based collinearity is more challenging and regrettably, is the most common form of the two. It is typically dealt with via the removal of one or more of the collinear predictors from the regression model. Variable selection was performed with an automated backward elimination employing the Akaike information criterion (AIC) as the selection criterion.

Table 2 presents the descriptive statistics of all the dependent variables used to infer the models. Detailed information concerning these

Table 2	
Descriptive statistics of the variables used in the mo	dels.

Variable	Code	Unit	Mean*	Frequency*	Median (1st, 25th, 75th, 99th)**
Thermal resistance of clothing	Clothing	clo	0.86	-	0.87 (0.54, 0.78, 0.97, 1.11)
Gender	Gender	female	-	0.78	-
		male	-	0.22	-
Age	Age	years	27.11	-	25 (20, 22, 30, 49)
Body Mass Index	BMI	kg/m ²	22.09	-	21.67 (17.42, 20.69, 23.94, 29.24)
Time lived in	Time.	\leq 3 years	-	0.53	-
Norway	Norway	greater than 3 years	-	0.47	-
Air velocity	Air.vel	m/s	<0.10	-	0.00 (0.00, 0.00, 0.00, 0.06)
Time of day	Time.	morning	-	0.47	-
-	day	afternoon	_	0.53	_
Vapour pressure	Vap.pre	kPa	0.70	-	0.70 (0.39, 0.58, 0.82, 1.06)
Operative temperature	Тор	°C	22.39	-	22.10 (18.79, 21.08, 23.72, 27.33)

* the mean refers to continuous variables, whereas the frequency refers to categorical variables

where 1st, 25th, 75th and 99th represent percentiles

variables can be found in Appendix B, while the instruments' accuracy can be found in Favero et al. [29].

All statistical analyses were performed using R [33] with the RStudio integrated development environment [34]. The beta model and the ordinal model were determined with the *glmmTMB* package [35] and *ordinal* package [36], respectively. Automated backward elimination was performed with the *buildmer* package [37] and all the graphs were

⁴ Collinearity is semantically equivalent to multicollinearity. In a general sense, collinearity refers to 'the condition of being collinear' and is a property of a set of explanatory variables, not just pairs of them.

created with the *ggplot2* package [38]. The significance level for all analyses was set at 0.05.

3. Results

In this section, the main results of the statical analysis are presented. Table 3 list all the variables used in the two models (i.e., the beta and ordinal models).

3.1. Testing for cluster effects

In Section 2.2, the experimental study was described as a mixture of hierarchical and crossed relationships. Nevertheless, before proceeding with the analysis, it was essential to establish that the three-level crossclassified model fit the data significantly better than the simpler three-levels models and the two-level model nested within it (see Fig. 4). The single-level model (i.e., the model without random effects) was also checked. The likelihood ratio (LR) test was used to perform this initial check.

This preliminary analysis was carried out for both the beta and ordinal models, and its results are presented in Table 4. Here, the threelevel cross-classified model is compared with the nested models. For both the beta and ordinal models, the three-level cross-classified model offers a better fit to the data.

3.2. Initial model

In this section, the initial full model is presented. The formulation of

Table 3

Covariates used in the models.

Classification (level)	Code	Variable	Туре	Unit
Days (level 3) Participants (level 3)	Day_ID Participant_ID Gender	Gender	Categorical, time-	Female (reference)/
	Age_c	Age (centred)	Continuous, time- independent	Years
	BMI_c	Body Mass Index (centred)	Continuous, time- independent	kg/m ²
	Time.Norway	Time lived in Norway	Categorical, time- independent	Less than or equal to 3 years (reference)/ More than 3 years
Experimental	Ramp_ID			J
conditions (level 2)	Time.day	Time of day	Categorical, time- independent	Morning (reference)/ Afternoon
	Clothing_c	Thermal resistance of clothing	Continuous, time- independent	clo
Measurement	Timepoint	-	-	
occasions (level 1)	Top_c	Operative temperature (centred)	Continuous, time- dependent	°C
	Vap.pre_c	Vapour pressure (centred)	Continuous, time- dependent	kPa
	Air.vel	Air velocity	Continuous, time- dependent	m/s
	Therm.pref	Thermal preference	Continuous/ Categorical, time- dependent	-

the linear predictor η_i is the same for the beta and ordinal models and can be written as:

$$\begin{aligned} \eta_i &= \beta_1 \text{clothing_c}_i + \beta_2 (\text{gender}_i) + \beta_3 \text{age_c}_i + \beta_4 \text{BMI_c}_i + \beta_5 (\text{time.norway}_i) \\ &+ \beta_6 \text{air.vel}_i + \beta_7 (\text{time.day}_i) + \beta_8 \text{vap.pre_c}_i + \beta_9 \text{top_c}_i + u_{\text{ramp_ID}(i)}^{(2)} \\ &+ u_{\text{participant_ID}(i)}^{(3)} + u_{\text{day_ID}(i)}^{(3)} \end{aligned}$$
(3)

where the subscript *i* is used to stress dependence on the i^{th} observation. Since the three-level cross-classified model fits the data better (see Section 3.1), all three random components were added to the initial model. Only the main results of applying automated backward elimination are illustrated in the following sections.

3.3. Ordinal model

Table 5 summarises the results of the ordinal model after automated backward elimination has been applied. Here, the coefficient estimates are given in units of ordered logits (or ordered log-odds). Five significant predictors were identified – thermal resistance of clothing, Body Mass Index, air velocity, time of day and operative temperature – all negatively associated with Logit(γ_k). The *ordinal* package [36] parametrises the model as:

$$Logit(\boldsymbol{\gamma}_k) = \mathbf{1}\boldsymbol{\tau}_k - \boldsymbol{\eta} = \mathbf{1}\boldsymbol{\tau}_k - \boldsymbol{X}\boldsymbol{\beta} - \boldsymbol{Z}\boldsymbol{u}$$
(4)

so a negative coefficient for β indicates that an increase of the associated variable x_i decreases the thermal preference vote. Stated analogously, votes for higher categories (e.g., prefer 'higher') are less likely. Nevertheless, the aim of this study is not inference but prediction; therefore, the specific values of the model's coefficients are not of interest. Furthermore, utilising any automated model selection procedure (e.g., automated forward selection, backward elimination or stepwise selection) should be avoided for inferential purposes. The parameter estimates are biased away from zero, the standard error and *p* values are too low and the confidence intervals are too narrow (page 68 of [39]), leading to misleading results. For prediction purposes, model selection can indeed provide a better bias-variance trade-off and improve the outof-sample error [40,41].

The estimated coefficients for a multilevel model are referred to as cluster-specific effects. For instance, the coefficient of $Top_c c$ in Table 5 is interpreted as the effect of a one-unit change in $Top_c c$ on the log-odds that $Pr(Y \le k)$ for a given cluster (i.e., while the unobserved characteristics captured by the random effects are held constant). However, considering the effects in this manner implies that individuals are compared with the exact same value for fixed and random effects. For some variables (e.g., gender) or other specific purposes, a comparison averaging across unobserved characteristics in the population is often of interest. In such a situation, population-averaged probabilities should be derived (see Section 2.2.3).

Fig. 5 shows the predicted probabilities as functions of the operative temperature for the cluster-specific and population-averaged procedures. It can be seen that the probabilities calculated with the two methods are dissimilar. For example, the maximum predictive probability for 'without change' is about 91 % for the cluster-specific approach, while it is only 55 % for the population-averaged one. Fig. 6(a) shows the probability mass for the ordinal model and cluster-specific procedure. These probabilities are plotted as a function of three different operative temperatures while holding the other covariates constant at their centred values and fixing the random effects at zero. Fig. 6(b) shows the population-averaged procedure's results.

3.4. Beta model

Table 6 summarises the results for the beta model after automated backward elimination has been applied. Here, the estimated coefficients

Applied Energy 325 (2022) 119803



Fig. 4. Schematics of the (a) single-level model, (b) two-level model and (c and d) two simpler three-level models nested within the three-level cross-classified model.

Table 4

Preliminary check

Modelling strategy	Model comparison	LR test statistic	
Ordinal model			
	Testing for multilevel model (see Fig. 4(a))	$\chi^2(3) = 534.4, p < 0.$ 001	
	Testing for participants and days (see Fig. 4(b))	$\chi^2(2) = 59.271, p < 0.001$	
	Testing for participants (see Fig. 4(c))	$\chi^2(1) =$ 12.854, $p < 0.001$	
	Testing for days (see Fig. 4(d))	$\chi^2(1) = 55.981,$ p < 0.001	
Beta model		P	
	Testing for multilevel model (see Fig. 4(a))	$\chi^2(3) = 675.77, p < 0.001$	
	Testing for participants and days (see Fig. 4(b))	$\chi^2(2) =$ 49.328, $p < 0.001$	
	Testing for participants (see Fig. 4(c))	$\chi^2(1) = 10.946, p < 0.001$	
	Testing for days (see Fig. 4(d))	$\chi^2(1) =$ 46.525, $p < 0.001$	
-			

are given in units of logits. Four significant predictors were identified – thermal resistance of clothing, Body Mass Index, time of day and operative temperature – all negatively associated with $Logit(\mu)$ (the intercept was not considered).

Fig. 7 shows the predicted responses as functions of the operative temperature using the cluster-specific and population-averaged procedures. The predicted response μ is the inverse of the link function, which in this study corresponds to the inverse of the logit:

$$\mu = \text{Logist}^{-1}(\eta) = \text{Logistic}(\eta) = \frac{1}{1 + e^{-\eta}} = \frac{1}{1 + e^{-(X\beta + Z\alpha)}}$$
(5)

More details about the mathematical formulation can be found in the

Table 5

Regression	coefficients for	the p	predictors	in	the	ordinal	model	(after	applying
automated	backward elimi	natio	n).						

Fixed Effects		coeff	se (coeff)	z	p value
Threshold $1, \tau_1$		-6.325	0.383	-16.519	_
Threshold $2, \tau_2$		-4.245	0.345	-12.307	_
Threshold $3, \tau_3$		2.065	0.306	6.761	_
Threshold 4, τ_4		4.070	0.327	12.441	_
Clothing_c		-3.483	1.375	-2.532	0.011*
BMI_c		-0.218	0.093	-2.338	0.019*
Air.vel		-12.584	5.503	-2.287	0.022*
Time.day	morning	reference			
	afternoon	-0.474	0.238	-1.988	0.047*
Top_c		-1.519	0.077	-19.719	< 0.001*
Random effects		sd	var		
Ramp_ID	(Intercept)	1.585	2.512		
Day_ID	(Intercept)	1.008	1.017		
Participant_ID	(Intercept)	1.158	1.340		

Number of groups: $Ramp_ID = 314$, $Day_ID = 68$, $Participant_ID = 38$ * indicates a significant term.

Appendix A.

All the lines in Fig. 7 are plotted as a function of the operative temperature while the other covariates (i.e., the fixed effects) are held constant at their centred values. However, these lines differ in the random effects, specifically:

- The solid black line (cluster-specific procedure) has the random effect fixed at zero;
- The dashed black lines (cluster-specific procedure) have the random effect fixed at the 16th and 84th percentiles (which correspond roughly to \pm 1 standard deviation above and below the mean);
- The solid red line (population-averaged procedure) has the random effect derived from simulation (see Section 2.2.3).



Fig. 5. Predicted probabilities of a thermal preference vote using the (a) cluster-specific and (b) population-averaged procedures.

The points in Fig. 7 are the observed thermal preference votes. While the predicted central tendency follows the general trend of the data, the predictions do not agree well with the observations, particularly close to the upper (i.e., prefer 'higher') and lower (i.e., prefer 'lower') boundaries.

Fig. 8(a) shows the pdfs generated from the beta model's estimated parameters (i.e., μ and ϕ) using the cluster-specific procedure. Each pdf is plotted as a function of three different operative temperatures while the other covariates are held constant at their centred values and the random effects are fixed at zero. It can be observed that the dispersion of the probability densities is relatively high. For instance, for an operative temperature of 26 °C, the probability of voting equal or lower 0.50 (i.e., from 'lower' to 'without change' on the continuous scale) is about 93 %.

implying a 7 % probability of voting higher than that. Fig. 8(b) shows the categorised probabilities of the predicted thermal preference votes. Fig. 9 presents the pdfs generated from the beta model with the population-averaged procedure. Here, as in Fig. 8, each pdf is plotted as a function of three different operative temperatures while the other covariates are held constant at their centred values, but the random effects are the results of simulations (see Section 2.2.3).

4. Discussion

Different approaches can be found in the literature for OCC for building (e.g., [42–44]). Futhrtmore, diverse modelling strategies have been developed to predict occupant' thermal preferences, many of



Fig. 6. Predicted probabilities of a thermal preference vote using the (a) cluster-specific and (b) population-averaged procedures for three different operative temperatures.

Table 6

Regression coefficients for the predictors in the beta model (after applying automated backward elimination).

Fixed Effects		coeff	se (coeff)	z	p value
(Intercept)		0.301	0.091	3.319	< 0.001*
Clothing_c		-1.291	0.456	-2.829	0.005*
BMI_c		-0.077	0.029	-2.634	0.008*
Time.day	morning	Reference			
	afternoon	-0.177	0.083	-2.136	0.033*
Vap.pre_c		-0.627	0.340	-1.845	0.065
Top_c		-0.413	0.018	-23.121	< 0.001*
Random effects		sd	var		
Ramp_ID	(Intercept)	0.565	0.320		
Day_ID	(Intercept)	0.311	0.096		
Participant_ID	(Intercept)	0.343	0.177		

Number of groups: $Ramp_ID = 314$, $Day_ID = 68$, $Participant_ID = 38$ Dispersion parameter: 6.38

indicates a significant term.

.....

which can be found in the review of Park et al. [20], Jung and Jazizadeh

[21] and Ngarambe et al. [45]. Among them a large portion are machine learning/data-driven algorithms. Even though there is sometimes overlap in goals and algorithms, statistical modelling and machine learning are based on two different concepts. The basic goal of stochastic modelling is to understand which probabilistic model could have generated the data observed. The usual procedure can be synthesised in the following steps: (i) choose a potential model from a plausible model family, (ii) fit the model to the data (i.e., estimate its parameters), and (iii) contrast the fitted model with other models. After selecting a model, this is used to conduct investigations, such as hypothesis testing and predicting new values. The estimated model becomes the lens used to interpret the data. Usually, a model that reasonably approximates the underlying stochastic process that has generated the data predicts well. On the contrary, machine learning is a data-driven application that is inspired by pattern recognition and focuses on regression, classification, and clustering techniques. The underlying stochastic process is frequently of secondary importance. Of course, stochastic models and procedures may be used to frame many machine learning approaches. However, the data are not regarded as having been created by that model. Instead, the main objective is figuring out which method or



Fig. 7. Predicted responses using the cluster-specific (black line) and population-averaged (red line) procedures. Note. The points are the observed thermal preference votes.

approach performs the specific task. Although these techniques appear to have the potential to improve prediction ability at the level of a single building occupant, their inherent character as 'black box' models renders them fundamentally unfit to explain their outputs. Interpretability is one of the primary problems with machine learning and can be an issue in a specific setting. For instance, understanding why a model reached a particular conclusion is fundamental in a building design setting. In this study, the statistical modelling strategies applied are a transparent tool and, as such, can be easily used in contexts where interpretability is required. Moreover, while the data used to develop the model in this study derives from a laboratory experiment with a mechanically conditioned environment, the modelling strategies are independent of this aspect. As such, they could also be applied to a naturally conditioned space.

In the literature, examples of statistical modelling can be found in the study of Daum et al. [46]. The authors create personalised thermal comfort profiles using multinomial logistic regression, a regression technique used to analyse a dependent variable measured on a categorical scale. The main difference with ordinal regression is that the categorical data are assumed to have no intrinsic ordering. Not taking an inherent order of the dependent variable makes the model more flexible than ordinal regression. However, it is essential to mention that this flexibility comes at a price. The number of parameters to estimate will drastically increase because k-1 different linear predictor term (η_k) are needed for the k category of the dependent variable (only k-1 because one category of the dependet variable is used as reference). In our study, since the dependent variable has five categories, this would have led to having, for example, four parameters for the operative temperature to estimate instead of only one. More parameters to estimate would require a larger sample size. The other difference compared with the study of Daum et al. [46] is that the diversity between subjects is directly accounted for in the model through the random effects term. Doing so makes it possible to model and predict the thermal preference of a specific (using the cluster-specific procedure) and a 'general' occupant (using the population-averaged procedure). In addition, the beta model allows doing the same when the thermal preference votes are measured on a continuous, but bounded, scale.

and interpreted. To begin with, the variables selected by the beta and ordinal model are contrasted and discussed. Subsequently, the beta and ordinal models are compared based on their predictive capabilities and the cluster-specific and population-averaged approaches are analysed. Finally, the limitations of the study are provided.

4.1. Variables selection

As explain in Section 1.1, the focus of the study is prediction and not inference; therefore, the specific value of the models' coefficients is not of interest. However, it is useful to compare variables selection across the models and contrast the relative importance of these variables. Table 5 and Table 6 show that automated backward elimination selected different sets of predictors for the two models. Four out of five predictors are shared by the two models (Clothing_c, BMI_c, Time.day and Top_c), while the fifth variables differ. For the ordinal model, automated backward elimination selected Air.vel, whereas for the beta model, Vap. pre_c was selected. In any attempt to understand the relative importance of the parameters estimated for the models, a direct comparison between their absolute values would be meaningless because the variables are measured using different units. Furthermore, several units could be used to measure the same variable. For example, if the operative temperature had been measured in degrees Fahrenheit instead of degrees Celsius, its estimated regression coefficient would have been different. However, the importance of the variable would not have changed. The relative importance of the predictors could be obtained via standardisation (i.e., subtracting the mean from each observed variable and dividing by its standard deviation) before conducting the statistical analysis. The resulting parameters estimated by the model are on the same scale and can be directly compared. The results of this procedure are show in Table 7. Here, even though the two models have different predictors, the order of relative importance of the common predictors is the same. The variables that differ between the two models are of minor relative importance. However, this importance is purely statistical. To determine the practical importance of the variables, subject-area expertise is required. Note that p values cannot be used directly to assess the importance of the predictors. A predictor can have a small p value when it has a very precise estimate, low variability, or a large sample size. As a result, even effect sizes that are small in practice might have extremely low p values. Understanding the practical importance of the predictors is beyond the scope of this study and is not pursued further. However, for inferential purposes, it is of the utmost importance.

As mentioned in Section 2.3, the Akaike information criterion (AIC) was used for variable selection. This metric is based on the maximised log-likelihood value with a penalty for including more parameters; it is a trade-off between goodness of fit (assessed by the likelihood function) and parsimony (the smaller the number of parameters, the lower the penalty). However, the AIC tends to over-parameterised, thus selecting models with a higher number of predictors, which could explain why the first four relatively important predictors were common to the two models, while their least relatively important predictors differed.

4.2. Models' comparison

The Akaike information criterion is generally used to compare different possible models and determine which one best fits the data. However, it cannot be used to compare models with different likelihood functions.⁵ For example, for a discrete distribution (e.g., ordinal response), the likelihood refers to the joint probability mass of the data, whereas for a continuous distribution (e.g., continuous response), the likelihood refers to the joint probability density of the data. Therefore, models based on continuous and ordinal responses cannot be compared directly. For this reason, the two models are compered graphically in

In the next sections, the results previously illustrated are examined

⁵ This is generally true for all probability-based statistics.



Fig. 8. (a) Probability densities and (b) categorised probabilities of the predicted response using the cluster-specific procedure for three different operative temperatures. Note. The dotted lines in (a) represent the thresholds used for categorisation.

terms of predicted probabilities (see Fig. 6, Fig. 8 and Fig. 9). However, it is important to point out that this method poses a limitation: a different categorisation of the beta distribution would lead to different probabilities. The same applies for the categorisation of the thermal preference vote used to estimate the ordinal model (see Fig. 3). Nevertheless, by comparing the probabilities estimated by the two models, the following general observations can be made. On the one hand, the ordinal model is more flexible in the sense that it can handle different probability distributions (virtually any probability distribution). For example, in Fig. 6, it can handle the spike in the probabilities for the 'without change' category for an operative temperature of 23 °C. On the other hand, the beta model is more detailed since it provides a pdf. For example, in Fig. 8(a), the predicted probability of observing a thermal preference vote between 0.45 and 0.55 for an operative temperature of 23 °C is 19.2 %. An alternative approach to comparing the two models would be to calculate the mean of the estimated probabilities for the ordinal model and contrast it with the predicted mean response of the beta model. The mean of the probabilities can be written as:

$$\operatorname{Mean}\operatorname{Pr} = \sum_{1}^{K} \pi_{k} k \tag{6}$$

where π_k is the probability of a specific category $k, k \in \{1, ..., K\}$. Here, the category prefer 'lower' was mapped to 1 and the category prefer 'higher' was mapped to 5. The resulting mean probabilities were then rescaled between + 0.001 and + 0.999 to match the predicted mean response of the beta model. Fig. 10 shows this comparison. For the ordinal model, the cumulative probability γ_k is the inverse of the link function, which in this study corresponds to the inverse of the logit:

$$\begin{aligned} \gamma_k &= \operatorname{Logit}^{-1}(\mathbf{1}\tau_k - \boldsymbol{\eta}) = \operatorname{Logistic}(\mathbf{1}\tau_k - \boldsymbol{\eta}) = \frac{1}{1 + e^{-(\mathbf{1}\tau_k - \boldsymbol{\eta})}} \\ &= \frac{1}{1 + e^{-(\mathbf{1}\tau_k - \boldsymbol{\chi}\boldsymbol{\theta} - \boldsymbol{Z}\boldsymbol{u})}} \end{aligned}$$
(7)

The probability of a specific category *k* is calculated as $\pi_k = \gamma_k - \gamma_{k-1}$. The probabilities of all the categories are then used to calculate the mean as in Eq. (6). More details about the mathematical formulation can be found in the Appendix A.



Fig. 9. (a) Probability densities and (b) categorised probabilities of the predicted response using the population-averaged procedure for three different operative temperatures. Note. The dotted lines in (a) represent the thresholds used for the categorisation.

Table 7	
---------	--

Predictors' relative importance for both the beta and ordinal models.

Modelling strategy	Predictor		Standardise coeff	Rank*
Ordinal model	Clothing		-0.464	4
	BMI		-0.567	2
	Air.vel		-0.163	5
	Time.day	morning	Reference	
		afternoon	-0.474	3
	Тор		-2.917	1
Beta model	Clothing		-0.172	4
	BMI		-0.199	2
	Time.day	morning	Reference	
		afternoon	-0.177	3
	Vap.pre		-0.102	5
	Тор		-0.793	1

 $^{\circ}\,$ the higher the absolute value of the standardise coefficient, the higher the rank.

As in Fig. 7, all the lines in Fig. 10 are plotted as a function of the operative temperature while the other covariates (i.e., the fixed effects)

are held constant at their centred values, but they differ in the random effects. Specifically:

- The solid black line (beta model, cluster-specific procedure) has the random effect fixed at zero (as in Fig. 7);
- The solid red line (beta model, population-averaged procedure) has the random effect derived from simulation (as in Fig. 7);
- The dashed black line (ordinal model, cluster-specific procedure) has the random effect fixed at zero;
- The dashed red line (ordinal model, population-averaged procedure) has the random effect derived from simulation.

It can be seen that the curve produced by using the cluster-specific procedure for the ordinal model has three inflexion points. This particular behaviour can be explained by looking at the predicted probabilities in Fig. 5(a). Between the operative temperatures of 22–24 °C, the predicted probabilities for 'without change' were much greater than all the others (from 80 % up to more than 90 %). Consequently, within this range, the calculated mean was greatly affected by these probabilities. The same behaviour can be observed for the population-averaged curve



Fig. 10. Predicted responses using the cluster-specific (black solid and dashed lines) and population-averaged (red solid and dashed lines) procedures for the beta and ordinal models, respectively. Note. The points are the observed thermal preference votes.

to a lesser extent. However, more considerable differences are visible at the tails of the curves, that is, the two extremities. Here, the beta model's mean response curve has tails that are heavier than the mean of the estimated probabilities for the ordinal model. Despite these differences, both the beta and ordinal models are both valid strategies for modelling thermal preference votes. However, the choice between the two models should be made based on how the response variable is measured. The beta model is a suitable choice when the thermal preference votes are measured on a continuous, but bounded, scale. In contrast, the ordinal model is appropriate when a categorical scale is used. Unfortunately, in the thermal comfort field, it is common practice to analyse subjective human thermal responses independently of how they were measured [47].

Furthermore, both models identified the largest random-effect component at the 'experimental condition' level, indicating that differences in conditions were the primary source of variability (see Table 5 and Table 6). Therefore, the variability in how individuals react to different dynamic conditions is higher than the variability between individuals, which may indicate that there could be unmodeled information in the variance at this level (i.e., the 'experimental condition'). For instance, the different rates of temperature change could play a role both in terms of absolute magnitude (i.e., the specific value of the rate of change) and sign (i.e., heating or cooling). To this aim, modelling strategies that model the effect of temporal patterns directly should be used (e.g., time series). Furthermore, if the variability of the environmental condition were significantly reduced (i.e., a static environment created), the inter-individual differences (i.e., difference between individuals) would be dominant.

4.3. Approaches' comparison

The predicted thermal preference votes were calculated from the two models using two different approaches: fixing the random effects at their mean of zero (cluster-specific procedure) and using a simulation approach with $M = 1 \cdot 10^4$ (population-averaged procedure). Regarding the ordinal model, from Fig. 6 it can be seen that the most evident

difference between the cluster-specific and the population-averaged procedures are the predicted probabilities for an operative temperature of 23 °C. Here, the predictive probability for 'without change' is about 91 % for the cluster-specific approach, while it is only 55 % for the population-averaged one. The reason for the discrepancy lies in the fact that the level 2 variance ($\widehat{\sigma}_{ramp_{-ID}}^2 = 2.512$) and the level 3 variances ($\widehat{\sigma}^2_{\rm day_ID}=$ 1.017 and $\widehat{\sigma}^2_{\rm participant_ID}=$ 1.340) are not close to zero. As the between-cluster variances $\hat{\sigma}_{ramp_ID}^2$, $\hat{\sigma}_{day_ID}^2$ and $\hat{\sigma}_{participant_ID}^2$ in the random-intercepts model increase, the curves will be further apart. For an example the reader is referred to Appendix C. The advantage of having predictive probabilities as outcomes is that they are their own error measures. In Fig. 6, the predicted probability of 'without change' for the cluster-specific approach is 91 %; if one decided not to choose this category as the expected outcome, the probability of this being an error is, by definition, 91 %. Following the same reasoning, for the population-averaged approach, not selecting 'without change' as the expected outcome has a 55 % probability of being an error. As a standard practice, the ordinal model regards the category with the highest probability as the predicted outcome (i.e., thermal preference vote). However, utilising a hard threshold, such as the automatic selection of the category with the higher probability, does not fully use the information contained in the probabilities. For example, in Fig. 5(b), such a threshold would lead to 'slightly lower' and 'slightly higher' never being selected. Here the necessity of defining a utility/cost function that, for example, maximises the expected utility or minimises the expected cost. With regard to the beta model, from Fig. 8 and Fig. 9 it can be seen that, for both the cluster-specific and population-averaged procedures, the distributions of the probability densities (and analogously, the categorised probabilities) for an operative temperature equal to 23 $^\circ\mathrm{C}$ are the same. The predicted mean response of the beta model intersects the thermal preference vote at 0.5, at which the prediction at $u_{\text{ramp_ID}} =$ $u_{\text{day}_\text{ID}} = u_{\text{participant}_\text{ID}} = 0$ (the median) equals the mean prediction (see Fig. 7). The median (i.e., cluster-specific) curve is lower than the population-averaged curve for a predicted thermal preference vote lower than 0.5 but is higher for a predicted thermal preference vote higher than 0.5. Consequently, the cluster-specific probability densities (i.e., the median probabilities) become skewed faster than the population-averaged ones (i.e., the mean probabilities) at operative temperatures higher or lower than 23 °C (see Fig. 8 and Fig. 9 for the operative temperatures of 20 °C and 26 °C).

4.4. Limitations

This study's limitations arise from some simplifications introduced during the statistical modelling. For both models, the functional form was assumed to be linear for simplicity (see Eq. (3)). As a consequence, the models do not account for potential nonlinearities. However, nonlinearities could be considered, for example, by using smoothing splines. Another simplification derives from assuming that all the independent variables were measured exactly, that is, 'error-free'. When covariates are measured with errors, the parameter estimates do not tend to the true values, even in extensive samples. For simple linear regression, this effect is known as the attenuation bias and leads to an underestimation of the coefficient. For more complex methods, such as multilevel models, this issue deserves a proper treatise and is beyond the scope of this study.

For a beta model, the conditional variance is $var(Y_i|U = u) = \mu_i(1 - \mu_i)/(1 + \phi)$, where μ_i is the mean and ϕ is the precision parameter. The parameter ϕ is known as the precision parameter because for fixed μ_i , the larger the ϕ , the smaller the variance of Y_i . Therefore, the variance is not constant but rather a function of the mean and the precision parameter, here assumed to be constant. However, the precision parameter can be modelled as a function of some predictors, for example, the operative temperature. In this study, this possibility was not explored and should examined in future studies.

To apply an ordinal model, the dependent variable must be cate-

gorical. For this reason, the dependent variable was binned into five categories according to the thresholds -0.6, -0.2, +0.2, and +0.6. However, these cut-off points were arbitrary and indirectly assumed to be the same for all participants. When a categorical scale is used to measure the dependent variable, this choice is made directly by single responders. Consequently, it is unlikely to be the same for all responders (see, for example, [48]). While this study used categorisation to apply the ordinal model, we do not encourage this practice in 'normal' circumstances. It is more appropriate to measure the variable directly with a categorical scale. As stated earlier, cut-off points are arbitrary and generally do not have practical/scientific meaning. Furthermore, the ordinal model has an additional assumption called 'proportional odds' (or equal slope assumptions). This assumption implies that the threshold parameters τ_k are independent of the regression variables or, equivalently, that the regression parameters are not allowed to vary with k, a specific category of Y. This restriction derives from the fact that the thresholds are theoretically linked with the response measure (and therefore assumed to be part of the measurement procedure), not to the predictor's value. However, the ordinal package [36], does not yet implement this feature when there is more than one random effect. Therefore, testing for partial and non-proportional odds (called 'nominal effects' in [36]) was not possible.

Moreover, multilevel models offer additional modelling possibilities that this paper has not discussed. For instance, in both the models used in this study, the slope coefficients of the predictors added to the models were assumed to be fixed across higher-level units. However, it is possible that the relationships between the responses and the predictors vary across these higher classification units (e.g., between participants). In multilevel models, it is possible to allow the effects of the predictors to vary randomly across higher classification units by adding a random slope to the model, which can be translated into checking whether, for example, the effect of the temperature differs across different occupants and to what degree of magnitude.

5. Conclusions and future perspectives

This study aimed to predict the thermal preference votes of human subjects exposed to a dynamic thermal environment. To this aim, two different statistical models were proposed, namely the beta and ordinal models. Based on the analyses carried out, the following points can be made concerning the two models:

- A three-level cross-classified model fit the data significantly better than the simpler three-levels models, the two-level model and the single-level model nested within it. Nevertheless, it is important to note that the multilevel structure described here is not a characteristic of the model. It is a feature of the experimental/study design represented by the data and encapsulated in the model. Therefore, it is essential to know how, where and when the data were collected (i. e., the metadata) to model them appropriately.
- In predictive modelling, direct interpretability regarding the model is not required; however, transparency is desirable. Multilevel models can model complex structures and at the same time, offer the advantage of having transparent outputs and modelling steps in contrast to machine learning/data-driven algorithms, which are basically 'black box' models.
- While likelihood-based statistics (e.g., Akaike information criterion, Bayesian information criterion) cannot be used to contrast the two models' performances directly, some qualitative observations can be made by comparing the probabilities estimated by the two models. On the one hand, the ordinal model is more flexible in the sense that

it can handle different probability distributions (virtually any probability distribution). On the other hand, the beta model is more detailed because it provides a pdf. The two models used in this study are both valid strategies for modelling thermal preference votes. However, the choice between the ordinal and the beta models should be made based on how the response variable is measured.

Furthermore, two distinct procedures were used in this study, namely the cluster-specific and the population-averaged procedures, to predict the thermal preference votes. These two methods apply directly to the concept of occupant-centric building design and operation. The population-averaged approach is suitable for the occupant-centric building design phase, where the target is the 'general' occupant. On the other hand, during the building operation phase, the notion of a 'general' occupant is pointless, and the focus should be on satisfying the needs of the specific occupant. In this case, a cluster-specific procedure is appropriate. This procedure can be carried out by measuring the specific occupant response to the environment and consequently updating the probabilities of the population-averaged procedure. These procedures could be used to design more energy-efficient and satisfying control strategies according to occupants' feedback(e.g., Ref [49]) in an occupant-centric [20] or human-in-the-loop [21] approach.

CRediT authorship contribution statement

Matteo Favero: Conceptualization, Data curation, Formal analysis, Methodology, Software, Validation, Visualization, Writing – original draft, Writing – review & editing. Jan Kloppenborg Møller: Conceptualization, Formal analysis, Methodology, Validation, Writing – review & editing. Davide Cali: Conceptualization, Methodology, Writing – review & editing. Salvatore Carlucci: Conceptualization, Funding acquisition, Supervision, Writing – review & editing.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The authors do not have permission to share data.

Acknowledgements

Matteo Favero would like to thank the Research Centre on Zero Emission Neighbourhoods in Smart Cities (FME ZEN, Grant no. 257660) and the Research Council of Norway (Norges Forskingsrådet) for their support.

Davide Calì and Jan Kloppenburg Møller received funding from the project "Human-in-the-Loop Digitalisation and Energy Management of buildings, HuiL-DEMand" (Program: EUDP, demonstration project, Journal number 64021-1021). Davide Calì also received funding from the project "Sustainable plus energy neighbourhoods, syn.ikia" (H2020 No. 869918).

This work has been performed within the framework of the International Energy Agency – Energy in Buildings and Communities Program (IEA-EBC) Annex 79 'Occupant-centric building design and operation'.

Open Access Funding by NTNU (Norges teknisk-naturvitenskapelige universitet).

(A3)

Appendix A. Mathematical background and examples

In this study-two modelling strategies are used, specifically the beta mixed-effects model with the logit link (simply referred as beta model) and the ordinal mixed-effects model with the logit link (simply referred as ordinal model). Both these models belong to a broad class of models called generalized linear mixed models.

There are four components that are common to any generalised linear mixed models:

- Random component (of the response variable): Specifies the probability distribution of the response variable, such as the normal distribution for *Y* (i.e., *Y* ~ Normal(μ, σ^2)) in the (classical) linear regression model. In general, there is no separate error term. Classical linear regression is a special case in which the error term can be extract from the distributional assumption (i.e., *Y* = $\mu + \epsilon$, where $\mu = X\beta$ and $\epsilon \sim Normal(0, \sigma^2)$);
- Link function: Specifies the link between the random and the systematic components. It denotes the relationship between the predicted response value (e.g., the mean) and the covariates;
- Systematic component: Specifies the covariates in the model, more specifically, how they are combined (usually through a linear combination);
- Random component (of the random effect): Specifies the probability distribution of the random effects, usually assuming a normal distribution with zero mean (i.e., $u \sim Normal(0, \Sigma)$).

Below, for both beta and ordinal model, the mathematical formulation and examples are provided.

Beta model

The beta model, as specified in Eq. (1), is:

 $Y \sim \text{Beta}(\mu, \phi)$ $\text{Logit}(\mu) = \eta$ $\eta = X\phi + Zu$ $u \sim \text{Normal}(0, \Sigma)$ (A1)

where the conditional distribution of the response variable Y is assumed to follow a beta distribution. Here, its expected value (i.e., its mean μ) is linked to linear predictor η through a logit function. The logit function is defined as the inverse of the cumulative distribution function (cdf) of the standard⁶ logistic distribution:

$$\operatorname{Logit}^{-1}(\eta) = \operatorname{Logistic}(\eta) = \frac{1}{1 + e^{-\eta}}$$
(A2)

Eq. (A1) is expressed in matrix notation. For a specific case it can be written as:

$$\begin{split} &Y_{pdri} \sim \text{Beta}\left(\mu_{pdri}, \phi\right) \\ &\text{Logit}(\mu_{pdri}) = \eta_{pdri} \\ &\eta_{pdri} = \mathbf{x}_{pdri}^{\mathsf{T}} \boldsymbol{\beta} + u_p + u_d + u_r \\ &u_p \sim \text{Normal}\left(0, \sigma_p^2\right); \text{ iid} \\ &u_d \sim \text{Normal}(0, \sigma_d^2); \text{ iid} \\ &u_r \sim \text{Normal}(0, \sigma_r^2); \text{ iid} \end{split}$$

where the subscript p indicates a participant, d a day, r a thermal ramp, and i is the i^{th} obsevation. If for example, we consider:

- two participants (no. 1 and 2), where participant 1 visited the lab on days 1 and 2 while participant 2 visited the lab on days 2 and 3,

- each day has two ramps (named 1–6) and only two observation per ramp.

We obtain:

the 16 \times 1 vector of the response variable

 $\boldsymbol{Y} = \left[Y_{1111}, Y_{1112}, Y_{1121}, Y_{1122}, Y_{1231}, Y_{1232}, Y_{1241}, Y_{1242}, Y_{2231}, Y_{2232}, Y_{2241}, Y_{2242}, Y_{2351}, Y_{2352}, Y_{2361}, Y_{2362}, \right]^{\mathrm{T}}$

the 11×1 vector of the random effects

 $\boldsymbol{u} = \left[u_{p(1)}, u_{p(2)}, u_{d(1)}, u_{d(2)}, u_{d(3)}, u_{r(1)}, u_{r(2)}, u_{r(3)}, u_{r(4)}, u_{r(5)}, u_{r(6)} \right]^{1}$

with variance-covariance matrix equal to

$$\Sigma = \begin{bmatrix} \sigma_p^2 I_2 & \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \sigma_d^2 I_3 & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \sigma_r^2 I_6 \end{bmatrix}$$
 where I_n is the identity matrix of dimension n

⁶ The logistic distribution is defined by two parametes: a location parameter μ and a scale parameter *s*. When $\mu = 0$ and s = 1, the logistic distribution is called standard logistic distribution, that is: Logistic($\mathbf{x}; \mu = 0, s = 1$) = $\frac{1}{1+e^{-\alpha}}$
the 16 \times 11 design matrix of the random effects

	Γ1	0	1	0	0	1	0	0	0	0	0]	
	1	0	1	0	0	1	0	0	0	0	0	
	1	0	1	0	0	0	1	0	0	0	0	
	1	0	1	0	0	0	1	0	0	0	0	
	1	0	0	1	0	0	0	1	0	0	0	
	1	0	0	1	0	0	0	1	0	0	0	
	1	0	0	1	0	0	0	0	1	0	0	
~	1	0	0	1	0	0	0	0	1	0	0	
L =	0	1	0	1	0	0	0	1	0	0	0	
	0	1	0	1	0	0	0	1	0	0	0	
	0	1	0	1	0	0	0	0	1	0	0	
	0	1	0	1	0	0	0	0	1	0	0	
	0	1	0	0	1	0	0	0	0	1	0	
	0	1	0	0	1	0	0	0	0	1	0	
	0	1	0	0	1	0	0	0	0	0	1	
	0	1	0	0	1	0	0	0	0	0	1	

Ordinal model

The ordinal model, as specified in Eq. (2), is:

 $\begin{aligned} \mathbf{Y} &\sim \text{Multinomial}(n, \boldsymbol{\pi}) \\ \text{Logit}(\boldsymbol{\gamma}_k) &= \mathbf{1}\boldsymbol{\tau}_k - \boldsymbol{\eta} \\ \boldsymbol{\eta} &= \mathbf{X}\boldsymbol{\beta} + \mathbf{Z}\boldsymbol{u} \\ \boldsymbol{u} &\sim \text{Normal}(\mathbf{0}, \boldsymbol{\Sigma}) \end{aligned}$

where the conditional distribution of the response variable *Y* is assumed to follow a multinomial distribution with probability vector $\pi = {\pi_1, ..., \pi_k}$, where $\pi_k = \Pr(Y = k)$. The cumulative probability corresponding to π_k is $\gamma_k = \Pr(Y \le k)$; hence, $\gamma_k = \pi_1 + ... + \pi_k$. Here the cumulative probabilities are then mapped to the real numbers through a logit function. In the ordinal model, the logit is function of the linear predictor η and $1\tau_k$, the vector of laten thresholds parameters τ_k , with $k \in {1, ..., K}$. That is:

$$\gamma_k = \operatorname{Logit}^{-1}(\mathbf{1}\tau_k - \eta) = \operatorname{Logistic}(\mathbf{1}\tau_k - \eta) = \frac{1}{1 + e^{-(\mathbf{1}\tau_k - \eta)}}$$
(A5)

Eq. (A4) is expressed in matrix notation. For a specific case it can be written as:

 $\begin{aligned} &\Pr(Y_{pdri}=k)=\pi_k\\ &\gamma_{pdri,k}=\Pr(Y_{pdri}\leq k)=\pi_1+\ldots+\pi_k\\ &\text{Logit}(\gamma_{pdri,k})=\tau_k-\eta_{pdri}\\ &\eta_{pdri}=\mathbf{x}_{pdri}^{\mathbf{T}}\boldsymbol{\beta}+u_p+u_d+u_r\\ &u_p\sim\text{Normal}\left(0,\sigma_p^2\right); \text{ iid}\\ &u_d\sim\text{Normal}(0,\sigma_d^2); \text{ iid}\\ &u_r\sim\text{Normal}(0,\sigma_r^2); \text{ iid} \end{aligned}$

where the subscript p indicates a participant, d a day, r a thermal ramp, i is the i^{th} observation, and k is a category of the dependent variable. If for example, we consider:

- two participants (no. 1 and 2), where participant 1 visited the lab on days 1 and 2 while participant 2 visited the lab on days 2 and 3,

- each day has two ramps (named 1-6) and only two observation per ramp.

We obtain, for a specific category *k*: the 16×1 vector of the response variable

 $\boldsymbol{Y} = [Y_{1111}, Y_{1112}, Y_{1121}, Y_{1122}, Y_{1231}, Y_{1232}, Y_{1241}, Y_{1242}, Y_{2231}, Y_{2232}, Y_{2241}, Y_{2242}, Y_{2351}, Y_{2352}, Y_{2361}, Y_{2362},]^{\mathrm{T}}$

the 11×1 vector of the random effects

 $\boldsymbol{u} = \left[u_{p(1)}, u_{p(2)}, u_{d(1)}, u_{d(2)}, u_{d(3)}, u_{r(1)}, u_{r(2)}, u_{r(3)}, u_{r(4)}, u_{r(5)}, u_{r(6)}\right]^{\mathrm{T}}$

with variance–covariance matrix equal to

$$\Sigma = \begin{bmatrix} \sigma_p^2 I_2 & \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \sigma_d^2 I_3 & \mathbf{0} \\ \mathbf{0} & \mathbf{0} & \sigma_r^2 I_6 \end{bmatrix}$$
 where I_n is the identity matrix of dimension n

the 16 \times 11 design matrix of the random effects

(A4)

(A6)

	[1]	0	1	0	0	1	0	0	0	0	0]
	1	0	1	0	0	1	0	0	0	0	0
	1	0	1	0	0	0	1	0	0	0	0
	1	0	1	0	0	0	1	0	0	0	0
	1	0	0	1	0	0	0	1	0	0	0
7	1	0	0	1	0	0	0	1	0	0	0
	1	0	0	1	0	0	0	0	1	0	0
	1	0	0	1	0	0	0	0	1	0	0
Z =	0	1	0	1	0	0	0	1	0	0	0
	0	1	0	1	0	0	0	1	0	0	0
	0	1	0	1	0	0	0	0	1	0	0
	0	1	0	1	0	0	0	0	1	0	0
	0	1	0	0	1	0	0	0	0	1	0
	0	1	0	0	1	0	0	0	0	1	0
	0	1	0	0	1	0	0	0	0	0	1
	0	1	0	0	1	0	0	0	0	0	1

Appendix B. Frequency distributions for the independent variables



Fig. B1. Frequency distributions for the continuous variables: (a) thermal resistance of clothing, (b) age, (c) Body Mass Index, (d) air velocity, (e) vapour pressure and (f) operative temperature. Note. The vertical black marks at the bottom of each figure are the rug plots.

Applied Energy 325 (2022) 119803



Fig. B2. Frequency distributions for the categorical variables: (a) gender, (b) time lived in Norway and (c) time of day.

Appendix C. Population-averaged and cluster-specific predictions

In Fig. 6, the differences between the predicted probabilities for the cluster-specific and the population-averaged procedures depend on the between-cluster variances. As the between-cluster variances $\hat{\sigma}^2_{\text{ramp_ID}}, \hat{\sigma}^2_{\text{day_ID}}$ and $\hat{\sigma}^2_{\text{participant_ID}}$ in the random-intercepts model increase, the curves will be further apart. This situation can be observed in Fig. C1 for category $k \leq 3$ (i.e., from the 'lower' to 'without change' category).

The predicted cumulative response probabilities always intersect at p = 50%, the point at which the prediction at $u_{ramp_{ID}} = u_{day_{ID}} = u_{participant_{ID}} = 0$ (the median) equals the mean prediction. The median (i.e., cluster-specific) curve is lower than the population-averaged curve for predicted probabilities lower than 50 %, while the median curve is higher for probabilities greater than 50 %. As a result, for a given range of *Top* values, the cluster-specific predicted probabilities will be greater than the population-averaged ones. In this case, for the category k = 3 (i.e., 'without change'), for *Top* values between 20.8 °C and 25.4 °C, the cluster-specific predicted probabilities are higher than the population-averaged ones (see Fig. C2). The dashed black lines in both Fig. C1 and Fig. C2 are cluster-specific effects, where the random effects $u_{ramp_{-ID}}$, $u_{day_{-ID}}$ and $u_{participant_{-ID}}$ were set to their 16th and 84th percentiles (i.e., $u_{ramp_{-ID}} = \{-1.11, 1.19\}$, $u_{day_{-ID}} = \{-0.56, 0.69\}$ and $u_{participant_{-ID}} = \{-1.05, 0.93\}$) and then added (i.e., $\{-2.71, 2.80\}$). The 16th and 84th percentiles were chosen because they correspond roughly to ± 1 standard deviation above and below the mean and encompass about 68 % of the observed random effects for *Ramp_ID*, Day_{ID} and $Participant_{ID}$.



Fig. C1. Predicted cumulative response probabilities for category $k \leq 3$ (i.e., from the 'lower' to 'without change' category) using the cluster-specific (black line) and population-averaged (red solid line) procedures.



r opulation aronagou

Fig. C2. Predicted response probabilities for category k = 3 (i.e., 'without change') using the cluster-specific (black line) and population-averaged (red line) procedures. Note. The dotted black lines represent the intersections between the cluster-specific and population-averaged probabilities.

References

- ANSI/ASHRAE Standard 55, Thermal environmental conditions for human occupancy, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, GA, 2020.
- [2] ISO 7730, Ergonomics of the thermal environment Analytical determination and interpretation of thermal comfort using calculation of the PMV and PPD indices and local thermal comfort criteria, International Organization for Standardization, Geneva, Switzerland, 20s05.
- [3] CEN EN 16798-1, Energy performance of buildings Ventilation for buildings Part 1: Indoor environmental input parameters for design and assessment of energy performance of buildings addressing indoor air quality, thermal environment, lighting and acoustics - Module MI-6, European Committee for Standardization, Brussels, Belgium, 2019.
- [4] CEN EN 15251, Indoor environmental input parameters for design and assessment of energy performance of buildings addressing indoor air quality, thermal environment, lighting and acoustics, European Committee for Standardization, Brussels, Belgium, 2007.
- [5] Fanger PO, Thermal comfort: analysis and applications in environmental engineering. Copenhagen, Denmark: Danish Technical Press; 1970. [6] Nicol JF, Humphreys MA. Thermal comfort as part of a self-regulating system
- [6] Nicol JF, Humphreys MA. Thermal comfort as part of a self-regulating system. Build Res Pract 1973;1(3):174–9. https://doi.org/10.1080/09613217308550237.
- [7] de Dear RJ, Brager RJ. Developing an adaptive model of thermal comfort and preference, ASHRAE Transactions, Amer Soc Heating Ref Air-Conditoning Eng Inc, San Francisco, CA, 1998, pp. 27–49.
- [8] Humphreys MA, Nicol JF. Understanding the adaptive approach to thermal comfort, ASHRAE Transactions, Amer Soc Heating, Ref Air-Conditoning Eng. Inc., San Francisco, CA, 1998, pp. 991–1004.
- [9] de Dear RJ, Brager GS, Cooper D. Developing an adaptive model of thermal comfort and preference -. Final Rep ASHRAE RP-884 1997.
- [10] McCartney KJ, Fergus Nicol J. Developing an adaptive control algorithm for Europe. Energy Build 2002;34(6):623–35. https://doi.org/10.1016/S0378-7788 (02)00013-0.
- [11] Arens E, Humphreys MA, de Dear RJ, Zhang H. Are 'class A' temperature requirements realistic or desirable? Build Environ 2010;45(1):4–10. https://doi. org/10.1016/j.buildenv.2009.03.014.
- [12] de Dear RJ. A global database of thermal comfort field experiments, ASHRAE Transactions, Amer. Soc. Heating, Ref. Air-Conditoning Eng. Inc., San Francisco, CA, 1998, pp. 1141–1152.
- [13] Brager GS, Paliaga G, de Dear RJ. Operable windows, personal control, and occupant comfort, ASHRAE Transactions, Am Soc Heating, Ref Air-Conditoning Eng. Inc., Nashville, TN, 2004, pp. 17–35.
- [14] de Dear RJ, Xiong J, Kim J, Cao B. A review of adaptive thermal comfort research since 1998. Energy Build 2020;214:109893. https://doi.org/10.1016/j. enbuild.2020.109893.

- [15] Cena K, de Dear RJ. Field study of occupant comfort and office thermal environments in a hot, arid climate, ASHRAE Trans Amer Soc Heating, Ref Air-Conditoning Eng. Inc., Seattle, WA, 1999, pp. 204–217.
- [16] de Dear RJ, Fountain ME. Field experiments on occupant comfort and office thermal environments in a hot-humid climate, ASHRAE Trans, Amer Soc Heating Ref Air-Conditoning Eng. Inc., Orlando, FL, 1994, pp. 457–474.
- [17] Schiller G, Arens EA, Bauman F, Benton C, Fountain M, Doherty T. A field study of thermal environments and comfort in office buildings, ASHRAE Trans Am Soc Heating, Ref Air-Conditoning Eng Inc., Ottawa, ON, 1988, pp. 280–308.
- [18] Yao R, Zhang S, Du C, Schweiker M, Hodder S, Olesen BW, Toftum J, Romana d'Ambrosio F, Gebhardt H, Zhou S, Yuan F, Li B. Evolution and performance analysis of adaptive thermal comfort models – A comprehensive literature review. Build Environ 2022;217. https://doi.org/10.1016/j.buildenv.2022.109020.
- Hensen JLM. Literature review on thermal comfort in transient conditions. Build Environ 1990;25(4):309–16. https://doi.org/10.1016/0360-1323(90)90004-B.
 Park JY, Ouf MM, Gunay B, Peng Y, O'Brien W, Kjærgaard MB, et al. A critical
- [20] Park JT, Oui Min, Guilay S, Peng T, O Brien W, Ajargaard MD, et al. A Critical review of field implementations of occupant-centric building controls. Build Environ 2019;165:106351. https://doi.org/10.1016/j.buildenv.2019.106351.
- [21] Jung W, Jazizadeh F. Human-in-the-loop HVAC operations: A quantitative review on occupancy, comfort, and energy-efficiency dimensions. Appl Energy 2019;239: 1471–508. https://doi.org/10.1016/j.apenergy.2019.01.070.
- [22] Humphreys MA, Fergus Nicol J. The validity of ISO-PMV for predicting comfort votes in every-day thermal environments. Energy Build 2002;34(6):667–84. https://doi.org/10.1016/S0378-7788(02)00018-X.
- [23] Kim J, Zhou Y, Schiavon S, Raftery P, Brager GS. Personal comfort models: Predicting individuals' thermal preference using occupant heating and cooling behavior and machine learning. Build Environ 2018;129:96–106. https://doi.org/ 10.1016/j.buildenv.2017.12.011.
- [24] Aryal A, Becerik-Gerber B, Lucas GM, Roll SC. Intelligent agents to improve thermal satisfaction by controlling personal comfort systems under different levels of automation. IEEE Internet Things J 2021;8(8):7089–100. https://doi.org/ 10.1109/JIOT.2020.3038378.
- [25] Lee J, Ham Y. Physiological sensing-driven personal thermal comfort modelling in consideration of human activity variations. Build Res Information: Int J Res Dev Demonstration 2021;49(5):512–24. https://doi.org/10.1080/ 09613218.2020 1840328
- [26] Katić K, Li R, Zeiler W. Machine learning algorithms applied to a prediction of personal overall thermal comfort using skin temperatures and occupants' heating behavior. Appl Ergon 2020;85:103078. https://doi.org/10.1016/j. apergo.2020.103078.
- [27] Guenther J, Sawodny O. Feature selection and Gaussian Process regression for personalized thermal comfort prediction. Build Environ 2019;148:448–58. https:// doi.org/10.1016/j.buildenv.2018.11.019.
- [28] Shan C, Hu J, Wu J, Zhang A, Ding G, Xu LX. Towards non-intrusive and high accuracy prediction of personal thermal comfort using a few sensitive physiological

Applied Energy 325 (2022) 119803

parameters. Energy Build 2020;207:109594. https://doi.org/10.1016/j. enbuild.2019.109594.

- [29] Favero M, Sartori I, Carlucci S. Human thermal comfort under dynamic conditions: an experimental study. Build Environ 2021;204:108144. https://doi.org/10.1016/ j.buildenv.2021.108144.
- [30] ANSI/ASHRAE Standard 55, Thermal environmental conditions for human occupancy, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, GA, 2017.
- [31] Salkind NJ. Encyclopedia of Research Design. California: Thousand Oaks; 2010.[32] Tabachnick BG, Fidell LS. Using multivariate statistics. 6th ed., new international
- ed., Harlow: Pearson; 2014.
 [33] R Core Team, R: A language and environment for statistical computing, R Foundation for Statistical Computing, Vienna, Austria, 2022. https://www.R-pro iect.org/.
- [34] RStudio Team, RStudio: Integrated Development Environment for R, RStudio, PBC, Boston, MA, 2022. http://www.rstudio.com/.
- [35] Brooks ME, Kristensen K, van Benthem KJ, Magnusson A, Berg CW, Nielsen A, et al. glmmTMB Balances Speed and Flexibility Among Packages for Zero-inflated Generalized Linear Mixed Modeling. R J 2017;9(2):378–400. https://doi.org/ 10.32614/KJ-2017-066.
- [36] Christensen RHB, ordinal—Regression Models for Ordinal Data, 2019. https://CR AN.R-project.org/package=ordinal.
- [37] Voeten CC, buildmer: Stepwise Elimination and Term Reordering for Mixed-Effects Regression, 2022. https://CRAN.R-project.org/package=buildmer. [38] Wickham H. sgploi2: Elegant Graphics for Data Analysis. New York: Springer-
- Verlag; 2016.
- [39] Harrell FEJ, Regression modeling strategies, with applications to linear models, logistic and ordinal regression, and Survival Analysis, 2015.
- [40] Hastie T, Tibshirani R, Friedman J. The Elements of Statistical Learning: Data Mining, Inference, and Prediction, Second Edition, New York, NY: Springer New York, New York, NY, 2017.
- [41] James G, Witten D, Hastie T, Tibshirani R. An introduction to statistical learning: With applications in R. New York: Springer; 2013.
- [42] Baldi S, Korkas CD, Lv M, Kosmatopoulos EB. Automating occupant-building interaction via smart zoning of thermostatic loads: A switched self-tuning approach. Appl Energy 2018;231:1246–58. https://doi.org/10.1016/j. apenergy.2018.09.188.

- [43] Korkas CD, Baldi S, Kosmatopoulos EB. 9 Grid-Connected Microgrids: Demand Management via Distributed Control and Human-in-the-Loop Optimization. In: I. Yahyaoui (Ed.), Advances in Renewable Energies and Power Technologies, Elsevier 2018, pp. 315–344.
- [44] Korkas CD, Baldi S, Michailidis I, Kosmatopoulos EB. Occupancy-based demand response and thermal comfort optimization in microgrids with renewable energy sources and energy storage. Appl Energy 2016;163:93–104. https://doi.org/ 10.1016/j.apenergy.2015.10.140.
- [45] Ngarambe J, Yun GY, Santamouris M. The use of artificial intelligence (AI) methods in the prediction of thermal comfort in buildings: energy implications of AI-based thermal comfort controls. Energy Build 2020;211:109807. https://doi. org/10.1016/j.enbuild.2020.109807.
- [46] Daum D, Haldi F, Morel N. A personalized measure of thermal comfort for building controls. Build Environ 2011;46(1):3–11. https://doi.org/10.1016/j. buildenv.2010.06.011.
- [47] Favero M, Luparelli A, Carlucci S. Analysis of subjective thermal comfort data: a statistical point of view [Manuscript submitted for publication], (2022).
- [48] Schweiker M. André M. Al-Atrash F. Al-Khatri H. Alprianti RR. Alsaad H. Amin R. Ampatzi E, Arsano AY, Azadeh M, Azar E, Bahareh B, Batagarawa A, Becker S, Buonocore C, Cao B, Choi J-H, Chun C, Daanen H, Damiati SA, Daniel L, Vecchi RD, Dhaka S, Domínguez-Amarillo S, Dudkiewicz E, Edappilly LP, Fernández-Agüera J, Folkerts M, Frijns A, Gaona G, Garg V, Gauthier S, Jabbari SG, Harimi D, Hellwig RT, Huebner GM, Jin Q, Jowkar M, Kim J, King N, Kingma B, Koerniawan MD, Kolarik J, Kumar S, Kwok A, Lamberts R, Laska M, Lee MCJ, Lee Y, Lindermayr V, Mahaki M, Marcel-Okafor U, Marín-Restrepo L Marquardsen A, Martellotta F, Mathur J, Mino-Rodriguez I, Mou D, Moujalled B, Nakajima M, Ng E, Okafor M, Olweny M, Ouyang W, Abreu ALPd, Pérez Fargallo A, Rajapaksha I, Ramos G, Rashid S, Reinhart CF, Rivera MI, Salmanzadeh M, Schakib-Ekbatan K, Schiavon S, Shooshtarian S, Shukuya M, Soebarto V, Suhendri S, Tahsildoost M, Tartarini F, Teli D, Tewari P, Thapa S, Trebilcock M, Trojan J, Tukur RB, Voelker C, Yam Y, Yang L, Zapata-Lancaster G, Zhai Y, Zhu Y, Zomorodian Z. Evaluating assumptions of scales for subjective assessment of thermal environments - do laypersons perceive them the way, we researchers believe? Energy Build 2020;211. https://doi.org/10.1016/j enbuild.2020.109761.
- [49] Cali D, Thilker CA, Specht SA, Real JP, Madsen H, Olesen BW. Human in the Loop: perceived-based control as the key to enhance buildings' performance. Build Simul 2021 Conf 2022.

Appendix B: Supplementary publications

The following six journal articles constitutes the supplementary publications of this thesis.

- (a) M. Schweiker, E. Ampatzi, M.S. Andargie, R.K. Andersen, E. Azar, V.M. Barthelmes, C. Berger, L. Bourikas, S. Carlucci, G. Chinazzo, L.P. Edappilly, M. Favero, S. Gauthier, A. Jamrozik, M. Kane, A. Mahdavi, C. Piselli, A.L. Pisello, A. Roetzel, A. Rysanek, K. Sharma, S. Zhang, Review of multi-domain approaches to indoor environmental perception and behaviour, Building and Environment 176 (2020). doi:10.1016/j.buildenv.2020.106804.
- (b) A.L. Pisello, I. Pigliautile, M. Andargie, C. Berger, P.M. Bluyssen, S. Carlucci, G. Chinazzo, Z. Deme Belafi, B. Dong, M. Favero, A. Ghahramani, G. Havenith, A. Heydarian, D. Kastner, M. Kong, D. Licina, Y. Liu, A. Luna-Navarro, A. Mahdavi, A. Nocente, M. Schweiker, M. Touchie, M. Vellei, F. Vittori, A. Wagner, A. Wang, S. Wei, Test rooms to study human comfort in buildings: A review of controlled experiments and facilities, Renewable & sustainable energy reviews 149 (2021) 111359. doi:10.1016/j.rser.2021.111359.
- (c) G. Chinazzo, R.K. Andersen, E. Azar, V.M. Barthelmes, C. Becchio, L. Belussi, C. Berger, S. Carlucci, S.P. Corgnati, S. Crosby, L. Danza, L. de Castro, M. Favero, S. Gauthier, R.T. Hellwig, Q. Jin, J. Kim, M.S. Khanie, D. Khovalyg, C. Lingua, A. Luna-Navarro, A. Mahdavi, C. Miller, I. Mino-Rodriguez, I. Pigliautile, A.L. Pisello, R.F. Rupp, A. Sadick, F. Salamone, M. Schweiker, M. Syndicus, G. Spigliantini, N.G. Vasquez, D. Vakalis, M. Vellei, S. Wei, Quality criteria for multi-domain studies in the indoor environment: critical review towards research guidelines and recommendations [Manuscript submitted for publication] (2022).
- (d) S. Carlucci, M. De Simone, S.K. Firth, M.B. Kjærgaard, R. Markovic, M.S. Rahaman, M.K. Annaqeeb, S. Biandrate, A. Das, J.W. Dziedzic, G. Fajilla, M. Favero, M. Ferrando, J. Hahn, M. Han, Y. Peng, F. Salim, A. Schlüter, C. van Treeck, Modeling occupant behavior in buildings, Building and environment 174 (2020) 106768. doi:10.1016/j.buildenv.2020.106768.
- (e) M. Vellei, E. Azar, K. Bandurski, C. Berger, S. Carlucci, B. Dong, M. Favero, A. Mahdavi, M. Schweiker, Documenting occupant models for building performance simulation: A state-of-the-art, Journal of building performance simulation 15 (2022) 634-655. doi:10.1080/19401493.2022.2061050.
- (f) B. Dong, R. Markovic, S. Carlucci, Y. Liu, A. Wagner, A. Liguori, C. van Treeck, D. Oleynikov, E. Azar, G. Fajilla, J. Drgoňa, J. Kim, M. Vellei, M. De Simone, M. Shamsaiee, M. Bavaresco, M. Favero, M. Kjaergaard, M. Osman, M. Frahm, S.

Dabirian, D. Yan, X. Kang, A guideline to document occupant behavior models for advanced building controls, Building and Environment 219 (2022) 109195. doi:10.1016/j.buildenv.2022.109195.

Paper a

Building and Environment 176 (2020) 106804



Contents lists available at ScienceDirect

Building and Environment

journal homepage: http://www.elsevier.com/locate/buildenv

Building and Environment

Review of multi-domain approaches to indoor environmental perception and behaviour



Marcel Schweiker ^{a,t,*}, Eleni Ampatzi ^b, Maedot S. Andargie ^c, Rune Korsholm Andersen ^d, Elie Azar ^e, Verena M. Barthelmes ^f, Christiane Berger ^g, Leonidas Bourikas ^h, Salvatore Carlucci ^{i,j}, Giorgia Chinazzo ^k, Lakshmi Prabha Edappilly ^{l,a}, Matteo Favero ^j, Stephanie Gauthier ^m, Anja Jamrozik ⁿ, Michael Kane ^o, Ardeshir Mahdavi ^g, Cristina Piselli ^p, Anna Laura Pisello ^p, Astrid Roetzel ^q, Adam Rysanek ^r, Kunind Sharma ^o, Shengbo Zhang ^s

- f Thermal Engineering for the Built Environment Laboratory, École Polytechnique Fédérale de Lausanne, Lausanne, Switzerland
- ⁸ Department of Building Physics and Building Ecology, TU Wien, Vienna, Austria
- ^h Lancaster Institute for the Contemporary Arts, Lancaster University, UK
- ⁱ Energy, Environment, And Water Research Center, The Cyprus Institute, Nicosia, Cyprus
- ¹ Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway
- ^k Department of Civil and Environmental Engineering, Northwestern University, Evanston, USA
- ¹ Department of Civil Engineering, Building Technology and Construction Management, Indian Institute of Technology Madras, India
- ^m Faculty of Engineering & Physical Sciences, University of Southampton, Southampton, UK
- ⁿ Independent consultant, Montreal, Canada
- ° Department of Civil and Environmental Engineering, Northeastern University, Boston, MA, USA
- ^p Department of Engineering, University of Perugia, Perugia, Italy
- ^q School of Architecture and Built Environment, Deakin University, Geelong, Australia
- ^r School of Architecture and Landscape Architecture, The University of British Columbia, Vancouver, BC, Canada
- ^s Department of Mechanical and Industrial Engineering, University of Toronto, Toronto, ON, Canada
- t Institute of Occupational, Social and Environmental Medicine, RWTH Aachen University, Aachen, Germany

ARTICLE INFO

Keywords

Comfort

Human perception

Occupant behaviour

Multi-physical

Contextual

Multi-domain

Personal

Multi-perceptual

ABSTRACT

Building occupants are continuously exposed to multiple indoor environmental stimuli, including thermal, visual, acoustic, and air quality related factors. Moreover, personal and contextual aspects can be regarded as additional domains influencing occupants' perception and behaviour. The scientific literature in this area typically deals with these multiple stimuli in isolation. In contrast to single-domain research, multi-domain research analyses at least two different domains, for example, visual and thermal. The relatively few literature reviews that have considered multi-domain approaches to indoor-environmental perception and behaviour covered only a few dozen articles each. The present contribution addresses this paucity by reviewing 219 scientific papers on interactions and cross-domain effects that influence occupants' indoor environmental perception and behaviour. The objective of the present review is to highlight motivational backgrounds, key methodologies, and major findings of multi-domain investigations of human perception and behaviour in indoor environments. The indepth review of these papers provides not only an overview of the state of the art, but also contributes to the identification of existing knowledge gaps in this area and the corresponding need for future research. In particular, many studies use "convenience" variables and samples, there is often a lack of theoretical foundation to studies, and there is little research linking perception to action.

* Corresponding author. Building Science Group, Karlsruhe Institute of Technology, Karlsruhe, Germany. *E-mail address:* marcel.schweiker@kit.edu (M. Schweiker).

Available online 23 March 2020

0360-1323/© 2020 Elsevier Ltd. All rights reserved.

^a Building Science Group, Karlsruhe Institute of Technology, Karlsruhe, Germany

^b Welsh School of Architecture, Cardiff University, Cardiff, UK

^c Department of Civil and Mineral Engineering, University of Toronto, Toronto, Canada

^d International Centre for Indoor Environment and Energy, Technical University of Denmark, Kgs. Lyngby, Denmark

^e Department of Industrial and Systems Engineering, Khalifa University, Abu Dhabi, United Arab Emirates

https://doi.org/10.1016/j.buildenv.2020.106804

Received 18 December 2019; Received in revised form 29 February 2020; Accepted 10 March 2020

1. Introduction

1.1. Background and state-of-the-art

Inhabitants of industrialized areas spend most of their time (85–96%) inside buildings [1,2]. Meanwhile, the human sensory system receives information regarding multiple indoor environmental exposures. Building energy consumption is significantly influenced by occupant perception and behaviour; that is, occupants' evaluation of thermal, visual, acoustic, and air quality stimuli and their reactions to any resulting discomfort [3]. As such, these four principal categories of environmental stimuli are integral to building design standards [4]. Not all interactions of occupants with their built environment result from dissatisfaction, but a close link between perception and behaviour exists [5].

While environmental stimuli occur simultaneously, the majority of scientific literature considers environmental influences on human perception and occupant behaviour in isolation. Literature reviews related to single-domain perceptions cover thermal [6–8], visual [9–12], indoor air quality (IAQ) [13], or acoustic [14–16] perception, as well as single-domain influences on occupants' actions [5,17–19]. An understanding of multi-domain environmental effects is lacking. ASHRAE [4]

states "current knowledge on interactions between and among factors that most affect occupants of indoor environments is limited". Addressing this knowledge gap, Torresin et al. [20] proposed a multi-domain research framework that identifies interactions and crossed effects between domains. Interactions are combined effects of two or more distinct domains (e.g., thermal and visual), on a third domain (e.g., overall environmental satisfaction). In contrast, crossed effects involve a main effect of one domain (e.g., thermal stimuli) on another domain (e.g., visual perception).

Literature reviews on multi-domain approaches are less numerous. Recently, Torresin et al. [20] identified 45 laboratory studies published after 1990 dealing with the effects of two or more environmental domains on perception and performance. Earlier reviews were based on smaller numbers of studies [21–23]. Frontczak et al. [23] reviewed nine studies focusing on the influence of individual domains on overall satisfaction. Candas et al. [21] discussed neurophysiological and behavioural findings on multisensory influences on thermal perception based on 25 publications. Centnerová et al. [22] reviewed eight papers with the same topic. The authors of this review could not identify earlier reviews addressing multi-domain approaches related to occupant behaviour.

In addition to the four principal indoor environmental domains,



Fig. 1. Contextual variables and their categorization.

contextual and personal variables influence occupants' perception and behaviour and are summarized in Figs. 1 and 2. Schweiker et al. [5] reviewed drivers of occupant behaviour, including contextual and personal factors. However, they did not examine interactions between these factors. Frontczak et al. [23] reviewed personal and contextual influences on overall satisfaction with the indoor environment. Schweiker et al. [8] included personal (psychological) and contextual factors in their review on individual differences in thermal perception. O'Brien et al. [24] concluded that most approaches analysed aggregated average models and diversity is captured through statistical approaches, without extracting personal or contextual factors.

This brief overview reveals a lack of reviews that considered multidomain influences on occupants' perception and behaviour. The current review aims to fill this gap as described in the following.

1.2. Objective, research questions, and scope

The primary objective is to examine multi-domain approaches with a much broader scope compared to previous reviews in order to enter into a new phase of conceptual developments in the field. This review aimed to identify motivations, key methods, findings, and gaps in the field of multi-domain approaches to human perception and behaviour in indoor environments.

The main research questions were (1) Why did researchers choose the domains and questions they considered?, (2) How did they approach multi-domain investigations?, (3) What were the key results?, and (4) What are limitations and gaps of their approaches?

The scope of this review covers studies applying a *multi-domain* approach to people's perception of the indoor environment and their resulting behavioural outcomes. The first categorization level made is between "perception" and "behaviour", as shown in Figs. 3 and 4, respectively. Studies without any physical predictors or with performance or health-related outcomes are beyond the scope.

Physical-perceptual independent variables cover measurable physical properties of the indoor and outdoor environment, e.g. indoor and outdoor air temperature for the thermal environment. All the physical properties of the thermal, visual, acoustic, and air quality environment are considered. *Physical multi-perceptual* approaches are defined as those covering variables from more than one domain of perception (e.g., thermal and visual perception). Studies dealing with multiple variables covering one domain only (e.g., solely air temperature and relative humidity, which are both from the thermal domain, on thermal perception) are not considered unless they included either contextual or



Fig. 2. Personal variables and their categorization.

personal variables. All *contextual and personal variables* shown in Figs. 1 and 2 are considered, except personal variables related to demographic factors (e.g. age, sex), or clothing if dealing with thermal perception.

Other behaviour and additional variables are included to cover studies that consider the status of one behaviour in the analysis of another behaviour. For example, window opening behaviour as dependent and the status of the heating system as independent variable.

This review covers laboratory studies, field studies, and questionnaire surveys. Studies related to perception or behaviour within the outdoor environment, virtual reality studies, or research based on simulations are out of scope. As such, this review provides a comprehensive overview of multi-domain approaches to understanding human perception and occupant behaviour indoors.

2. Methods

This review's approach is visualized in Fig. 5. The visualization is based on the "Preferred Reporting Items for Systematic Reviews and Meta-Analyses" (PRISMA) schema [25]. However, in contrast to a systematic review, first, we collected and reviewed known research, which returned 153 articles. This initial step included searches in author's individual reference databases as well as in bibliographic search engines (Table 1). Second, the more than 1000 articles citing these 153 articles or being cited by this initial collection were assessed. Together with their evaluation, we categorize our work as critical review [26].

2.1. Selection process

The units of analysis were the articles and their records. A record is defined as a dependent variable analysed within an article. As such, one article presenting analyses for two or more dependent variables (e.g. analyses of thermal and visual perception as dependent variable) has an equivalent number of records.

The exclusion criteria were: (1) out of scope; (2) other than English language; (3) full text unavailable, and (4) not peer-reviewed. In addition, (5) duplicates such as conference and journal articles presenting the same research were considered once; and (6) review papers without additional analyses such as meta-analysis were not considered.

2.2. Records' structure

The following data were extracted: dependent and independent variables; number (N) of participants, offices, and/or buildings; sex and age distributions; number of votes obtained or length of study; type of study (e.g. field or laboratory); type of building (e.g. residential or office); type of conditioning (e.g. naturally-ventilated (NV) or airconditioned (AC)); region in which the study was conducted; data collected; statistical approach applied, and key findings.

In addition, introduction and discussion sections were scanned for the study's motivation and gaps/future research needs mentioned.

3. Comparison between perceptual and behavioural multidomain approaches

Multi-domain approaches with perception as a dependent variable (244 records/163 articles) are three times more frequent than behavioural multi-domain studies (97 records/64 articles). Note that eight articles report results from perceptual and behavioural dependent variables (see the complete review table: https://osf.io/gnvp2/). The most frequent approach in perceptual and behavioural studies was a combination of one or more physical factors with contextual variables (Fig. 6).

In both research areas, perception and behaviour, field studies are the most frequent methods used (Fig. 7). Laboratory studies only dominate in studies that examined multi-perceptual effects without contextual or personal variables.

The sample size varies according to the type of sample analysed, i.e.



^{a)} overall perception and/or domain specific perception ^{b)} not considered as multi-variable approach and not included in this review





^{a)} not considered as multi-variable approach and not included in this review

Fig. 4. Schema of multi-variable approaches with behaviour as outcome variable. Note that the approach numbers at the bottom of this figure refer to the corresponding subsection numbers within this review.

whether authors reported buildings, rooms, or participants (Table 2). The number of participants in laboratory studies ranged from 5 to 199 with nearly half of the studies with less than 30 (mean 45.6, SD 42.2, median 30). In field studies, the largest number of participants (N = 52,980 and N = 29,632) were observed in two studies combining physical and contextual variables (subsection 4.2) using existing databases of online surveys [27,28] (mean of all field studies 824.1, SD 3178, median 138). Sample sizes below 10 participants were observed in several subsections. Arguments were for example an integral research approach triangulating between four qualitative and quantitative methods [29] or in-depth insights by gathering detailed information through interviews and discussions [30]. The number of buildings varies from 1 [31] to 351 [27].

4. Perceptual studies

This section is divided into four subsections: physical; physical and contextual; physical and personal; and physical, personal, and contextual. In each subsection, we reflect on the motivational background, the methods employed for data collection and analysis, and some of the key findings. We conclude each subsection with thoughts on the current state of the art, prevailing knowledge gaps, and future research needs.

Fig. 8 summarizes the findings on crossed main effects on thermal, visual, IAQ, and acoustic perception referred to in the following.

4.1. Physical multi-perceptual approaches

A considerable number of studies addressed the effects of multiple environmental factors on occupant perception. While not all these



Fig. 5. Schema of the review process.

Table 1

Literature searches performed during the first phase of this review.

Database/search engine	Search terms (combinations of)
Web of Science	"thermal", "visual", "acoustic", "comfort", "satisfaction", "perception", "behaviour"
Scopus	"thermal", "visual", "acoustic", "personal", "contextual", "multi-domain", "comfort"
Science Direct	"occupant behaviour", "multi-domain", "model", "combined effects"
Google Scholar	"thermal", "visual", "acoustic", "comfort", "satisfaction", "perception", "behaviour"
Google Scholar	"indoor factors", "interaction", "combination"
Google Scholar	"Occupant", "thermal", "comfort", "satisfaction", "visual", "behaviour"
Google scholar	"occupant behaviour", "multi-domain", "model", "combined effects"
Deakin University library (linked to several databases)	"thermal comfort", "visual comfort", "acoustics"

studies specifically address the combined effects of multiple indoor environmental variables, most acknowledge at least their concurrent presence [32–109]. In the following, we focus on a number of these papers and their contributions, directly relevant to the topic of multi-domain exposures.

4.1.1. Motivational background

The majority of the studies cite the need for better understanding of exposure situations involving multiple indoor environmental variables. Other studies observed effects of multiple environmental variables without a specific intent to examine their interactions [53,62]. Studies considered different combinations of environmental variables, most frequently thermal and visual [34,37–39,52,67,102–104]. A few studies investigated other combinations of variables, such as visual and acoustic [45], thermal and acoustic [56,57,66], visual and IAQ [59], acoustic and IAQ [83], visual, thermal, and acoustic [48,49,62,70], as well as IAQ, thermal, and acoustic [35,41]. Researchers were mostly interested in the

effect on dependent variables such as occupants' comfort, sensation, and preference [34,39,48,52,66,102–104]; and satisfaction [59,68].

4.1.2. Approaches

The majority of papers involved short-term laboratory studies in office settings. Only in a few studies, participants were given the opportunity to adjust certain factors of their immediate surroundings [45, 59] or exercise a choice upon experiencing different settings [40].

Experimental settings typically involved different properties of the physical environments such as air temperature (thermal environment), sound type and level (acoustic environment), illumination level, glare intensity, light colour (visual environment), and airflow rates (thermal and air quality environment). Laboratory studies typically lasted a few hours or up to a day. Typically, experiments tested one or more levels of a physical variable crossed with one or more levels of another physical variable (e.g., three levels of temperature crossed with two levels of illumination, as in Kulve et al. [104]), while holding other indoor environmental variables constant.

The majority of experiments had within-subject designs, that is, all participants experienced all experimental conditions, typically counterbalanced by randomising the order of conditions. Within-subject experiments are more sensitive to the manipulation of independent variables, which is important for studies with smaller sample sizes.

The occupancy-related implications of environmental factors were queried using techniques such as surveys and questionnaires (e.g. Ref. [48]), comfort and sensation scales (e.g. Ref. [66]), and visual observations (e.g. Ref. [70]).

As expected, data analysis involved various well-established formats and techniques from descriptive and inferential statistics. The collection of statistical methods commonly referred to as ANOVA (Analysis of Variance) was frequently deployed for processing and interpretation of measurement results [40,45,52,66,68,70], as were mixed-effects models [37,39,49,67,104].

In the majority of the less frequent field studies, the setting was a university classroom and participants were students. However, field



Fig. 6. Frequency of studies reviewed per approach.



Fig. 7. Frequency of records separated by type of study.

studies were also conducted in office, hospital (e.g. Ref. [34]), and residential settings [50]. Field studies typically lasted several months.

Environmental physical conditions were monitored and participants were asked to rate their perceptions through questionnaires on comfort, sensation and satisfaction (e.g. Ref. [53,105]). Measurements of environmental conditions were associated with participants' subjective ratings, and the subjective ratings with each other, using measures such as correlation [105] or ANOVA [106]. Field studies enabled the variation of environmental conditions for large samples of subjects (e.g., 331 students in 7 varied classrooms [105]).

Table 2

 $Number \ of \ participants, \ offices, \ or \ buildings \ by \ category. \ N = number \ of \ records, \ Min = minimum, \ SD = standard \ deviation, \ Med = median, \ Max = maximum.$

Section	Partic	ipants					Roor	ns/offic	es				Buildings/households						
	N	Min	Mean	SD	Med	Max	N	Min	Mean	SD	Med	Max	N	Min	Mean	SD	Med	Max	
4 Perception																			
4.1 Physical multi- perceptual	109	6	99.3	186.6	35	990	0						3	1	2	1	2	3	
4.2 Physical + contextual	82	7	1525.9	6674.6	168	52980	10	1	6.3	5.3	4	18	34	2	38.6	84.2	14.5	351	
4.3 Physical + personal	16	20	557.9	1852.4	93	7500	8	6	56.5	51.9	46	120	6	2	4.3	3.8	2	11	
4.4 Phys. + cont. + pers.	9	35	295.4	206.3	400	482	0						1	8	8		8	8	
5 Behaviour																			
5.1 Physical multi- perceptual	9	5	42.2	44.8	20	128	4	1	3.5	3.1	2.5	8	4	9	17.8	6.1	19.5	23	
5.2 Physical + contextual	11	17	504.9	891.3	36	2787	18	3	83.6	159.2	14	555	20	1	30.5	28.9	16.5	70	
5.3 Physical + personal	4	65	65	0	65	65	2	6	63	80.6	63	120	1	2	2		2	2	
5.4 Physical + cont. + pers.	6	32	1091.8	905.3	933	2787	2	4	4.5	0.7	4.5	5	4	13	35	14.7	42	43	
5.5 Physical + multi- behavioural	11	8	18.5	9.3	21	40	6	3	8.5	3.6	8	14	4	1	1	0	1	1	

The geographic distribution is presented in Table 3. Studies were predominantly conducted in Central Europe, North America, and Eastern Asia.

Table 3

Geographic distribution of records.

Section	Africa	Asia	Europe	North- America	Oceania	South- America
4 Perception 4.1 Physical multi- perceptual	1	44	38	9	0	0
4.2 Physical + contextual variables	0	22	34	28	2	1
4.3 Physical + personal variables	0	3	10	6	0	0
4.4 Physical + contextual + personal variables	0	2	2	0	1	0
5 Behaviour 5.1 Physical multi- perceptual	0	8	1	2	0	0
5.2 Physical + contextual	0	3	13	3	0	0
5.3 Physical + personal	0	0	5	1	0	0
5.4 Physical + contextual + personal	0	3	2	0	2	0
5.5 Physical + multi- behavioural	0	0	16	2	1	0

4.1.3. Findings

Studies described in the reviewed papers entail a host of valuable findings (Fig. 8). Tiller et al. [66] reported a slight effect of acoustic conditions on subjective ratings of thermal comfort, but no reverse effect. Nagano and Horikoshi [56] concluded that operative temperature has a slight effect on auditory comfort sensation votes and thus that the thermal environment must be taken into consideration in acoustic studies. On the other hand, they did not observe any effect of noise on reported thermal sensation. On the contrary, Pellerin et al. [107] indicated a noise effect on thermal comfort in warm conditions, but not of temperature on acoustic sensation, comfort, and preference. Yang et al. [108,109] reported that thermal comfort decreased with increased noise level, and with the noise of a fan as compared to that of babble, and that water sounds increased cold sensation and decreased thermal comfort. The authors also observed the influence of the thermal environment on acoustic comfort and sensation, but with contrasting findings, as they report a decrease of annoyance and an increase of acoustic comfort at thermoneutrality [108] as well as an increase in acoustic perception and annoyance at thermoneutrality [77,109].

Nakamura et al. [58] reported that higher colour temperature is preferred in summer and vice versa in winter. Fanger et al. [45] observed slight lighting effects on thermal comfort: people preferred a slightly lower temperature under red light than under blue light. Similar results were reported by Albers et al. [32] and by Winzen et al. [69], with electric light colour affecting thermal sensation, comfort and temperature estimation. Chinazzo et al. [39] suggested that participants' thermal sensation reports were influenced by the colour of the daylight. For instance, as compared to orange daylight exposure, a colder thermal sensation was reported in the case of blue daylight, even though the measured temperature remained the same. Daylight quantity was also reported to affect thermal perception, with increased thermal comfort under dim daylight conditions in a warm environment and under bright daylight conditions in a cold environment [103]. However, the authors indicate no effect of daylight illuminance levels on thermal sensation [103], similarly to what was reported by an earlier study with electric lighting [75]. Meanwhile, Azmoon et al. [34] observed improved thermal comfort responses because of increased light intensity.

Unexpected effects were sometimes found on variables that were not the focus of the experimental study. For example, people reported IAQ differences across temperature settings [67], or across combinations of acoustic, lighting, and temperature settings [49]. In some cases, papers noted significant effects only under restricted conditions. For example, Geng et al. [47] observed that people were less satisfied with IAQ and lighting under certain temperature settings, but not others. In some cases, papers noted statistically non-significant interactions between environmental conditions. For example, Pan et al. [83] observed that adding noise to odour mitigated the effect of odour on air-quality-related measures. However, with a sample size of N = 9, small interaction effects are unlikely detected.

Many studies observed no interactions between environmental factors tested (e.g. Ref. [37,38]), or were not designed in a way to investigate these interactions (field studies).

4.1.4. Identified gaps and future directions

The review of multi-physical perceptual research shows the extent of valuable knowledge generated over the past five decades. However, the



Fig. 8. Overview of crossed main effects related to thermal, visual, air quality and acoustic perception based on studies including significance tests.

yield is less extensive and less conclusive if we specifically query for frequent, clear, and consistent instances of cross-modal influence. The results are in many instances inconclusive, and in certain cases even contradictory. It is thus of paramount importance to reflect upon some of the key shortcomings and limitations of past research, which correspond more or less directly to requirements for future research efforts.

Given the difficulties of conducting research including real occupants in realistic settings (involving, amongst others practical, ethical, and economic issues), it is not surprising that most studies are shortterm. Moreover, the participants, often young students, are not necessarily representative of pertinent populations, for instance, of office workers. Most studies were conducted in offices, yet other building typologies such as residential buildings are practically ignored by the literature.

Researchers frequently try to establish some measure of realism in the experimental settings, but this is rarely effectual given the difficulty in concealing the inherent artificiality of the available testing facilities. As such, the reviewed studies do not truly succeed in addressing the implications of the Hawthorne effect, even though, scholars argue about its nature and suitable methods to account for it in research [110,111].

Studies often start with some reference to previous research (frequently to authors' own previous publications), but there is very little evidence of actual carryover of past studies' findings. As such, the majority of the studies appear to practically start from scratch. Perhaps consequently, different studies do not deploy standard research designs, data collection strategies, metrics, and statistical analysis techniques, making attempts toward conducting meta-analyses factually futile.

There is arguably a paucity of collaborative, multi-institutional, international, and interdisciplinary experimental studies. Specifically, few studies seem to have truly recognized the critical importance of conceptual and methodological integration of engineering and human science methods.

One fundamental problem with most research efforts is the absence of foundational theories that would facilitate the processes of hypothesis formulation and testing. This may be of course in part due to the inherent complexity of the subject. However, the chances of obtaining scalable and generalizable results remain slim if research designs do not at least make an attempt to start from a provisional general theory of the nature of the perceptual and behavioural processes involved in multidomain exposure situations.

4.2. Physical + contextual variables

This subsection examines studies investigating the combined effects of physical and contextual variables on environmental perception. These studies examined how context may interplay with single- or multisensory domain perceptions by imposing unknown or indirect influences on the physical properties of the environment or by shaping the users' perceptions and expectations in line with social or cultural experiences [27–29,42,112–192].

4.2.1. Motivational background

The drive for research varies greatly between the studies identified. Some researchers challenge the absence of an established single index for holistic comfort [124,129,167]. In other studies, the combined effects of physical and contextual variables were merely incidental rather than an intended outcome [124].

In four of the identified studies, the inclusion of contextual factors was thought to enrich environmental evaluation by factoring subjectivity into assessments typically based on only physical criteria [112, 114,132,149]. Similarly, some research aimed to improve post-occupancy evaluation techniques, from how data is collected or analysed [72,168], to examine the combined influence of suspected co-contributors to satisfaction in a single-sensory domain [169].

We identified three distinct research themes focusing on specific building attributes. One addressed the concurrent influence of

environmental and spatial factors present in open-plan office space configurations [42,113,137,142,170], a second examined limitations of green building design and rating systems [125,127,136,171], and a third concerned the impact of the presence of control opportunities [115,116,172,173].

4.2.2. Approaches

In contrast to the studies reviewed in section 4.1, the interest seems to be more in real settings, shown by the majority of studies applying field study approaches. Here, the influences of the contextual factors can be examined with limited cost and reduced difficulty in the experimental set-up.

Subjective evaluation through surveys is a common approach for data collection of comfort or satisfaction based on the self-reporting of participants [27,28,174]. Several studies involve measurements of indoor environmental quality metrics related to thermal, acoustic, and visual properties alongside with occupants' subjective votes [42,74,124, 129,136,137,142,144,145,169,170,175–177].

The most frequent building typologies were office buildings (e.g. Ref. [115,119,126,134,140,143,147,178–180]) and educational buildings (e.g. Ref. [116,120,123,181–183]), followed by residential buildings [122,153,175], hostels and student residences [42,121], restaurants and cafés [132,142], factories [118,184], a healthcare facility [150], a shopping mall [141] and airport terminal [47].

The length of data collection differed depending on the methodology and the research focus. Longitudinal studies ranged from months to years [171,185]. Studies employing structured or semi-structured interviews may span over several seasons [141,186,187]. Short survey or interview studies last usually no more than two months [148,150,153], but can be as short as a few days [124,125,140,143,174,181,188].

Summary statistics, including mean and variance, were used by nearly all studies. In addition, several types of correlational analysis, parametric and non-parametric tests are common approaches.

Overall perception was the most frequently researched dependent variable, followed closely by thermal perception and then by visual perception, acoustics, and IAQ. Metrics for overall perception ranged from mainstream choices such as overall satisfaction, acceptability or comfort (and even 'uncomfortableness') to measures of 'psychic wellbeing", preference for space and affective quality of space. The metrics used for thermal, visual, acoustic, and IAQ perceptions were more conventional, with higher variance for the visual domain, including satisfaction with lighting, glare perception, eye discomfort and appearance of the environment.

4.2.3. Findings

The influence of geographic location is not conclusive. With similar climate conditions, occupant responses to warm and cold weather tend not to differ greatly across countries [28]. Similarly, Sakellaris et al. [157] found minimal differences in multiple types of perception between two locations. In contrast, thermal and IAQ perception differed between occupants of the same country, especially for those countries with a large north-south spread [98].

The interior design and furniture in office and school settings correlated strongly with comfort [133,135,143,151,157]. Perception of illuminance level strongly depended on office layout and furniture type [123,183,189]. Furthermore, since daylight levels exhibit strong spatial dependence, visual comfort at workplaces varied greatly with proximity to the window [27,151].

The perceptual aspects of visibility in classrooms [183,189], privacy in offices [27,151], and available space in offices [27] are additional factors associated with room layout and furniture selection, which correlated with visual and overall comfort levels. Few studies recommend optimal office layout or furniture selection for comfort. This is likely due to the subjective and non-quantifiable nature of these properties.

One of the most important components of the building envelope is

the window [190]. Poor thermal comfort (e.g., cold or warm window) [175,191], daylight glare [191], and poor acoustic comfort [191] are reported by participants in large-windowed residential or office buildings. Additionally, the design of solar control devices and solar control techniques can affect occupant comfort, especially thermal and visual. For instance, Karlsen et al. [192] demonstrate that occupants prefer venetian blinds with adjustable slat angles to those with only on-off position. These handful of studies are among the few that made conclusions from surveys, while the majority of other studies use simulation approaches beyond the scope of this review.

Perception and comfort in green buildings vs. conventional buildings varied greatly among studies. Two studies demonstrated that occupants' overall comfort is higher for green buildings [127,144]. In contrast, Gou et al. [128] observed no significant difference in overall comfort between these building types. The contrasting results may be due to two reasons. First, overall comfort can be influenced by occupants' attitude towards the "green" identity of the building [171]. Second, the term "green" building is not universally defined, and used for buildings that are certified by different standards (e.g. LEED [144], LEED and GBL [128], BREEAM [171]). These standards differ significantly in their assessment criteria. Consequently, the building performance can vary largely.

NV and passively cooled buildings that allow occupants to control aspects of the indoor environment, resulted in positive thermal comfort perceptions outside the fixed temperature limits set in standards [120, 130,148,180]. Moreover, controllability strongly increases occupants' satisfaction with thermal indoor conditions in winter and summer [28, 120,130,148].

4.2.4. Identified gaps and future directions

The contextual variables discussed in this paper are those mentioned in the literature. Further research would be needed to evaluate whether the most researched dependent and independent variables are the most influential.

Among the building related parameters, façade design and interior design are crucial. Few studies use a surveying approach to evaluate façade design options. Thus, further field surveys are needed to associate occupant multi-domain perception with design decisions. Simulations alone cannot substantiate the claims, as they may not truly reflect the actual indoor environment. Spatial information is merely described in the text. For future studies, publishing this information in a visual format is desirable, e.g., with photos and architectural drawings such as floor plans, sections, or elevations, which can convey the spatial situation better. Examples of appropriately published spatial architectural information exist [29,125,142]. In general, further research on spatial characteristics would be desirable, because spatial characteristics and typologies also depend on building types and the number of studies considering each building type is currently small.

In most studies, the context was represented by one or a few variables. However, context is a complex system of multiple dynamically interacting variables. For example, visual perception varies with the location of a workplace within a floor plan [169], but the occupants' perception is further influenced by other spatial parameters such as orientation and fenestration of the façade [175], climate related parameters such as season, sun path/latitude [126], and indoor surface materials [114]. Our review identified no study that investigated the complexity and interplay of multiple contextual variables, which is likely due to methodological challenges with required data types and the needed quantity of data. New methodological approaches might be needed for future studies to describe and understand the complexity and interplay of contextual variables.

Most papers used statistics for data analysis, and these methodologies tend to require large sample sizes for higher validity. If context is evaluated at a high level of resolution, i.e. with in-depth analysis of the spatial geometric or architectural design characteristics, it is unlikely that large sample sizes exposed to identical characteristics can be obtained for all building types. Therefore, a broader variety of approaches and methodologies could expand the investigated contexts.

4.3. Physical + personal variables

This subsection concerns thirteen studies that combine the impact and mutual influence of measured indoor environmental conditions and personal variables on occupants' perception [52,168,193–203].

4.3.1. Motivational background

In some studies, the analysis of personal variables is tangential and brief, while in other studies, the main purpose and motivation is to understand how personal variables influence occupants' perception. The analysis of personal variables is important to understand the differences in perception observed among individuals or groups in similar environmental conditions [196]. Nevertheless, all experimental studies aimed to evaluate the possible correlation between personal variables and the different domains of environmental perception.

4.3.2. Approaches

Studies include one or more dependent variables related to thermal, visual, acoustic, IAQ, or overall perception. Other studies considered comfort perception as a dependent variable together with productivity [202], which is out of scope here.

Almost all studies were conducted in office or educational buildings or in controlled chambers that simulate a working environment. Only one study was set in a non-office commercial building, a shopping centre [193].

Field studies including physical measurements and questionnaires dominate in this subsection. To achieve higher control and a broader collection of physical variables, some studies used laboratories that reproduce commercial [193], educational [52,197], or office environments [194,196,198]. One study is based on questionnaires [168]. Yun [199], instead, applied a mixed methodology to evaluate the energy implications of personal variables, specifically of perceived control.

The applied statistical analysis methods largely vary among the studies, ranging from ANOVA and MANOVA [197,198,203] to regression [196], correlation analysis [52,168,193,194], and non-parametric analysis [195].

4.3.3. Findings

Overall, findings showed that personal variables significantly influence multi-domain comfort perception positively or negatively.

Occupants' perceived control and satisfaction with building management are among the key analysed personal variables significantly interacting with the overall perception. Robertson et al. [195] highlighted that workers' visual comfort and personal wellbeing are influenced by perceived control over lighting, especially in non-naturally ventilated buildings. Additionally, occupants' reduced perceived control over the indoor environment has a significant negative effect on their thermal comfort [199] and general perception of a building [168]. On the contrary, the availability of choice over lighting control were demonstrated to decrease occupants' perceived importance of lighting in offices [198] and their performance [197]. Focusing on the interaction of thermal, acoustic, and visual domains, Dang et al. [193] showed that, although thermal and acoustic personal satisfaction are not directly correlated with lighting parameters, they interact with personal lighting satisfaction. On the other hand, a significant effect of thermal variables and clothing level on visual perception was obtained only in artificially illuminated buildings, while in daylight the influence of other parameters, e.g. acoustics, became relevant [52]. Finally, Schweiker at al. [196] demonstrated that personality traits, i.e. neuroticism, extraversion, openness to new experiences, are moderating thermal perception. Focusing on physiological parameters, Pigliautile et al. [194] highlighted that a multi-domain approach is required to understand human comfort thoroughly.

4.3.4. Identified gaps and future directions

Generally, very few studies were identified that deal with the interaction of multi-domain perception and personal variables beyond demographics. Moreover, many of these studies concern the impact of perceived control on environmental conditions and less focus is given to other personal variables. In addition, many studies simply report the differences observed among occupants with different personal variables without attempting to understand their motivation, which limits their contribution to the factual understanding of the influence of personal variables. Another important gap is the small sample size and the lack of diversity of the samples. Although gender balance is generally fulfilled, many of the studies selected university students for their experiments. Finally, none of the reviewed papers focused on residential environments. While certain personal variables, such as perceived control and privacy, might be less significant in residential spaces compared to office buildings, other variables, such as the expectation of building performance and energy/money saving might be significant, and thus worthy of exploration.

4.4. Physical + contextual + personal variables

While some of the studies discussed in the previous subsections explored physical, contextual, and personal predictors of perceptions, none aimed to understand the interactions of these independent variables. The current subsection covers eleven research efforts that addressed this gap by simultaneously examining at least one predictor variable from each category [30,72,125,204-211].

4.4.1. Motivational background

All studies promote a multi-domain approach to perceptual evaluation. For instance, Jin et al. [211] highlight the need to study physical (i. e., objective) and non-physical (i.e., subjective) drivers of occupants' perceptions with their indoor environment. Pivac et al. [204] state the importance of physiological and social factors in the evaluation of perceptions. Indraganti et al. [209] focus on the role of occupants' demographic and personal characteristics while assessing thermal comfort. Hitchings et al. [30] highlight the need to study cultural, geographic, and seasonal adaptation effects. Other studies aimed to understand overall environmental satisfaction levels [72,125]. Overall, a unified and explicit goal of proving that physical, contextual, and personal variables combine to explain perceptions is lacking.

4.4.2. Approaches

Ten of the reviewed articles are field studies conducted in noncontrolled building environments, while one [206] took place in a laboratory controlled office setting. The studied environments were office [125,204–208], residential [30,72,209,210], and retail buildings [211]. Dependent variables included domain-specific comfort metrics such as thermal comfort [30,204,209], neutral temperature [206], visual comfort [205,211], and acoustic comfort [210]. Two studies [72,125] considered domain-specific comfort metrics and overall perceived comfort levels of the respondents.

Data collection was carried out through environmental sensing devices, questionnaires, walkthroughs, inspections, interviews, and diaries. The data collection duration varies from one-time surveys (e.g. Ref. [210]) to data collected over an extended period of time (e.g., 40 days in Sadeghi et al. [205]).

The data analysis approaches include qualitative and quantitative assessments. Starting with the former, Hitchings et al. [30] used a qualitative analysis of the collected data. The other studies mostly applied statistical analysis methods to derive relationships between, on the one hand, the environmental, contextual, and personal data that were collected, and on the other, the respondents' perceptions of comfort. The statistical methods include ANOVA [125,210], X²-tests [72], Mann-Whitney *U* test and the Kruskal-Wallis test [204], correlations [72,205,210,211], and linear regression [125,205,206,209–211].

4.4.3. Findings

While this subsection covers a broader scope of predictor categories than previous sections, the results are not more diverse. The results do not explicitly confirm that physical, contextual, and personal predictors collectively drive the reported perceptions. The findings of the articles mostly identify single or dual types of interacting perception drivers, which is in line with the observations of previous subsections.

Starting with thermal perception, Pivac et al. [204] found that environmental metrics, office type, and job type have a significant influence on the perceived thermal comfort. Indraganti and Rao [209] observed a strong correlation between the respondents' economic group and their reported comfort levels, and weaker relationship with the other considered variables such as season and tenure. Schweiker and Wagner [206], on the other hand, highlight a significant influence of perceived control on neutral temperature, while office type affected perceived control.

Related to visual perception, Jin et al. [211] found that the measured illuminance level is the dominant driver of visual comfort, while the existence of daylighting plays an essential role in subjective satisfaction. Sadeghi et al. [205] found a strong relationship between the occupants' perception of control and their acceptability of a broader range of visual conditions.

In Park et al. [210], the authors studied potential drivers of subjective responses to floor impact noise in residential buildings. They highlight a significant impact of noise sensitivity and floor slab thickness on the reported acoustic comfort levels.

The main observation by Xue et al. [72] and Freihoefer et al. [125] is a significant difference in the reported overall comfort levels between workspace types (open and closed). Xue et al. [72] found that the combined effect of thermal comfort and IAQ significantly influences visual comfort, while the abundance of daylight hours and illuminance levels showed strong positive correlations with reported visual perceptions. More interestingly, the authors confirm strong dependencies between pairs of variables such as IAQ/thermal comfort and room orientation, adaptive behaviours of shading/lighting and visual comfort, and finally, mental stress and acoustic comfort.

4.4.4. Identified gaps and future directions

The findings presented above do not provide a clear understanding of the interactions nor fundamentals of the combined effect of physical, contextual, and personal predictors of perception. The findings cannot be generalized given the small sample of studies that met the criterion used for inclusion in this subsection. Furthermore, the data analysis methods applied were mostly constrained to studying relationships between a limited number of variables (in many cases two variables), falling short of providing a comprehensive understanding of the influence of multi-variable predictors and their interactions. More diverse predictors and complex analysis tools (e.g., Principal Component Analysis and Artificial Neural Networks) can be considered in future research to draw more diverse and comprehensive conclusions on the drivers of occupant perceptions. Finally, except for Schweiker and Wagner [206], none of the studies were conducted in controlled environments, which is another potential avenue for exploring multi-domain predictors of perception.

5. Behaviour

This section summarizes studies considering the relationship between measurable conditions of indoor environmental quality and occupant behaviour.

Figs. 9 to 12 show the crossed main effects of multiple independent variables on different types of behaviour, which will be discussed in the following subsections.



Fig. 9. Effects of physical, contextual and personal variables on window opening behaviour.

5.1. Physical multi-perceptual approaches

The nineteen studies analysed in this subsection attempt to relate occupant behaviour to multi-perceptual physical environmental conditions [46,212–229].

5.1.1. Motivational background

The motivation behind the majority of these studies was to evaluate the drivers of occupant behaviour in the context of multiple domains of occupant comfort. In general, all of the studies aimed at a better forecasting and simulation of occupant behaviour under multiple indoor environmental performance criteria. Specifically, all but a few studies were concerned with the effect of indoor and outdoor climatic conditions on occupant control of windows, blinds, and/or lighting, as well as the derivative effect of such control on perceived thermal comfort, lighting comfort, and/or building energy use.

The underlying objective was the characterization of the relationship between measurable physical parameters and occupant behaviour. Specific objectives include the evaluation of the effect of solar insolation on perceived thermal comfort, lighting comfort, and occupant controls of window blinds [215] and the development of a data-driven personalized thermal comfort model and minimum daylight requirement model to be used for model-predictive control of window blinds [213].

5.1.2. Approaches

All reviewed papers relied to some extent on physical monitoring of indoor environmental conditions and direct monitoring or measurement of occupant control decisions (e.g., window opening behaviour). Most studies undertook some form of occupant comfort evaluation via questionnaires, and several papers undertook monitoring of outdoor climatic conditions (e.g., outdoor air temperature and air pollution concentrations).

All but a few papers described field studies of offices or dwellings. The exceptions were laboratory studies [214,218,219]. All field studies took place in regions where there are discernible heating and cooling seasons, and no studies were undertaken in climatic regions such as the Tropics or Sub-Tropics.

The duration of behavioural studies followed one of three trends:

they undertook either a short duration of measurements in a manner of days [214,219], a medium-term measurement across a single climate season [212,221,227], or a much longer-term study across several seasons up to an entire year or more [46,213,215,216,222,224–226,228, 229]. Controlled laboratory studies had the shortest measurements. A notable example is Daum et al. [213], who collected over 6800 individual survey responses over a period of 3 years.

The studies' methods of data analysis included, for example, correlations between the probability of an action and environmental variables. For example, Inkarojit [215] evaluated the correlation between the probability of occupants' opening or closing windows and received solar radiation on window surfaces. Similarly, Daum et al. [213] correlated the probability of window blinds opening/closing actions and indoor air temperature. Various forms of regression methods, such as linear regression, multiple linear regression, univariate and multivariate logistic regression, were used by all studies.

5.1.3. Findings

Given an underlying, often implied understanding across all studies that occupant behaviour is inherently stochastic, the main format of illustrated findings were probability density functions of occupant behaviour against one or more parameters.

The findings from these papers defended widely-accepted principles of thermal and visual comfort in the built environment, as opposed to revolutionising them or putting them into question. For example, the studies which evaluated the extent to which window open/close behaviour would be driven by outdoor climatic conditions, IAQ, or other parameters, broadly concluded that indoor and outdoor air temperature, coupled with IAQ and/or solar radiation, are the primary drivers of window control by occupants [46,216,217,221,222,226,228]. Outdoor air quality was identified as a moderate parameter of influence, particularly when it is considerably poor [226]. While solar radiation should be deemed a quasi-thermal parameter with a direct effect on indoor and outdoor air temperature and indoor heat gains, IAQ is related to a different domain, so that window open/close behaviour can be understood as a multi-domain problem.

All studies that evaluated the physical drivers and indicators of window blind and lighting operation [212,213,215,225,229] observed



Fig. 10. Effects of physical, contextual and personal variables on different types of thermal behaviours.

the effect of multiple environmental conditions on blind and lighting controls, but still found parameters of solar insolation to be the primary driving force of control decisions. While window blinds are a form of solar and thermal control, and electric lighting is needed in the absence of daylight, it is surprising that all studies suggested that blind and lighting control are univariate problems determined by solar insolation alone.

5.1.4. Identified gaps and future directions

Overall, meteorological conditions were not usually measured adjacent to the buildings or sites under analysis, or at least were not reported. Differences in microclimatic conditions, from what is experienced directly outside a building envelope to what is measured from a central weather station are non-negligible. This is a potential limitation of correlations made between weather and human behaviour [226].

Of the studies examining window opening/closing behaviour, works such as Jeong et al. [216] indicate that caution must be taken when data from only one or two seasons are used. In other words, drivers of behaviour in winter may not apply in summer conditions, and studies in either season may not apply to conditions under autumn and spring. The effort to observe occupant behaviour across multiple seasons was, if not a norm across the long-term works, an identified research gap across several of the medium-term studies. As observed by Naspi et al. [222], this may be the main research gap of studies in this subsection.

Despite prior evidence that circadian lighting affects occupants' perception, only the experimental studies evaluated the effect of

circadian lighting conditions on occupant behaviour. The study of circadian lighting, both natural and artificial, and its effect on human physiology and psychology warrants further attention by field studies. None of the evaluated field studies explored whether light colours, or other indicators of circadian lighting, affected occupant behaviour. In addition, noise levels were not frequently measured across studies that evaluated window open/close behaviour, even though the relationship between noise and window operation is not trivial [230].

5.2. Physical + contextual variables

This section provides insights into thirty-one studies aimed at predicting or explaining behaviours that include at least one type of physical and one type of contextual predictor variable.

5.2.1. Motivational background

Similar to the studies identified in subsection 5.1, one of the key objectives behind the majority of these papers is to account for behaviour-related uncertainty in building energy simulation and to develop models, which are hence developed to help bridge the gap between measured and predicted energy consumption [31,167,203,229, 231–243]. Some of the studies stated that their contribution was based on the need to develop models for specific geographic contexts or building types (e.g., hospital wards) [244]. Linked to this objective is the investigation of cause-effect relationships between the operation of the building by occupants and different technologies installed [245].



Fig. 11. Effects of physical, contextual and personal variables on light switch on behaviour.

Other studies investigated control interaction for providing enhanced input for building automation control [246] or the optimization of peak electricity loads [175]. Furthermore, researchers stated that the key objectives were to gain better insights into occupants' choices of adaptive opportunities for thermal comfort enhancement in specific climatic contexts [140,247–250], or into the effect of occupancy on perceived control and behavioural patterns [206]. Other papers modelled occupant interaction with certain controls to gain a better understanding on other environmental factors [251].

5.2.2. Approaches

The majority of papers addressed window control (N = 16), window blinds control (7), thermal adjustments (e.g. thermostat adjustment, switching on space heating and/or cooling systems)(7), lighting control (7), and adjustment of fan speed (2). Multi-domain independent variables were related to the thermal environment (36), the visual environment (17), and IAQ (13). Only one record included information on acoustic variables [248]. Amongst these independent variables, the most common for window control behaviour models were those related to indoor and outdoor temperatures [18,167,203,206,231,233,235–238, 240,241,244,248,252] and IAQ [18,231,235–237,241,244,248,252]. Blinds behaviour models mostly included thermal variables [167,175, 206,229,233,239,253] and visual variables [206,229,233,239,253]. The papers investigating thermal adjustments only included thermal environmental variables in combination with contextual variables [167, 206,232,238,245,249,250].

The contextual factors included the time of day or arriving/leaving times [31,203,231,233,236,237,239,240,242,246], the previous control state [203], geographical location [167,238], ventilation type [140,203, 238], building system and envelope characteristics (e.g., installed

technologies, building envelope efficiency, window opening size) [244, 245], facade orientation [175,251], dress code [249], season or cloud cover [175], socio-economics [232], and occupancy levels [206].

Most of the 26 field studies used physical measurements (24) and 11 of them also surveys. Two studies used a combination of measurements, surveys, and observations, and one field study used only observations. The duration of the data collection varied from a few days (laboratory studies such as [206]) up to several years [239].

Some of the studies combined field measurements with a questionnaire-based investigation [140,167,246], or used questionnaires [232] or interview techniques [245] independently. Most records refer to office environments (22), residential buildings (9), and hospital environments (1).

The statistical methods used were logistic regression [236,237,240, 242,244], probit analysis [167,203,238], neural networks [231], Markov processes [239], data mining approaches [237], and Bayesian networks [31,241]. Other statistical analysis included Generalized Estimation Equations [246], ANOVA analysis [115], weighted and linear trend lines [140].

5.2.3. Findings

A wide range of studies found a strong dependency between the time of day and window control patterns in offices [203,236,237] and residential buildings [31,235,240,252]. Hansen et al. [232] found that window operation in Danish households was correlated with building characteristics, such as technical installations and energy efficiency of the building envelope, while it was not correlated with the building age. Shi et al. [244] found that windows with large adjustable opening sizes are more likely to be in an ajar state and the interaction frequency is much higher. Based on questionnaires, the indoor temperature at which



Fig. 12. Effects of physical, contextual and personal variables on blind closing behaviour.

a substantial proportion of occupants start to open windows for ventilation was observed to be similar in all climates, but window use was more common in Europe than in Pakistan [167,238]. Rainfall was also found to have a significant effect on opening a window, along with the location of the office (and its relation to safety) [254].

In line with section 5.1.3, studies including physical and contextual variables found correlations between window operation and IAQ indicators (e.g. CO_2 and VOC concentrations) [31,231,235,236,252]. Stazi et al.'s review [18] found that window opening was mostly linked to CO_2 concentration in residential buildings. According to Fabi et al. [255], all papers that measured IAQ indicators found correlations with window operation.

Several studies found a strong relationship between window blind control operation and the time of day [233,239,246], while others did not [251]. Another important contextual factor influencing window blinds operation is the facade orientation [233,246,251]. Time of day and/or arrival/leaving times play an important role also for light switch behaviour [233,242].

5.2.4. Identified gaps and future directions

Although all studies included at least one contextual variable, further work needs to create a comprehensive approach including a more extensive set of contextual and potentially personal factors.

Regarding contextual physical environmental factors, further attention should be paid to the ease and convenience of using building system interfaces, the state of other devices (or controls) and the influence of building automation routines on behavioural patterns. Furthermore, contextual factors such as interior design and furniture layout, or the relation between the indoor and outdoor environments (e.g., view to the outside) need to be further investigated. Even various social factors, such as social constraints, group interactions, the presence of multiple occupants in open space versus private offices [206], and control behaviour due to safety reasons need to be further investigated. Although some studies compared a few different geographic locations, a more comprehensive approach is needed to understand the variability of occupant behaviour in different climatic zones and/or cultural backgrounds.

Related to the research methodology, relationships between indoor variables and window transitions, based purely on survey responses (e.g. Ref. [167,238]), must be treated with caution. Since the window state affects indoor variables [235,255], conditions just prior to an event are needed to understand the relationship.

5.3. Physical + personal variables

This subsection reviewed six studies looking at physical and personal predictors, which could explain some of the differences amongst adaptive behaviours. The personal predictors include clothing habits, sociocultural expectations, personality traits, and occupancy preferences.

5.3.1. Motivational background

Most studies investigating physical and personal variables concurrently aimed to develop occupants' behaviour models to control building systems.

5.3.2. Approaches

The studies consider thermal systems [256,257], lighting systems [198,258], or thermal and lighting systems [196]. These systems were generally operating in non-stressful conditions (i.e. acceptable environmental conditions). One common dimension considered in study designs is their longitudinal aspect, with studies lasting from a day to many months.

Research exploring physical and personal variables as predictors to behaviour analysed these two predictors independently or jointly. Indraganti et al. [256] applied descriptive and inferential analyses to explore the relationships between occupants' behaviours (14 control actions) and personal variables (dress habits). In parallel, the relationship between occupants' behaviours (air-conditioning and fan usage) and a physical variable (outdoor daily mean temperature) was explored through logistic regression. Schweiker et al. [196] applied mixed effect regression analysis to explore the effect of physical (thermal and visual) and personal (personality traits) variables on occupants' behaviours (clothing adjustments, window opening, blind closing, and ceiling fan usage). Gunay et al. [258] applied discrete-time Markov logistic regression to explore the effect of physical (ceiling illuminance) and personal (occupant's presence) variables on occupant's behaviours (light switching and window blind actions).

5.3.3. Findings

Most studies highlight that occupants respond to environmental discomfort, but fail to revert the state once discomfort disappears. Gunay et al. [258] observed that occupants closed blinds upon glare and switched on lights upon low daylight; but they then often failed to open the blinds and to switch off the lights. Occupants' locus of control is not a concern in non-stressful or acceptable good conditions [198]. Furthermore, occupants' interactions with building environmental systems may be linked to daily routine and habits [257] and differences in behavioural patterns between sub-populations based on personality traits are considerably high [196].

5.3.4. Identified gaps and future directions

Most studies highlighted a lack of the contextual dimension, including climate, seasonal effects, building types, building orientations, complexity of controls, interior layout, single/shared spaces, and organisational policies [196,198,257,258]. In addition, multi-domain physical predictors are missing except for one study including IAQ [196]. Finally, studies should consider the Hawthorne effect already discussed in section 4.1.4 and by Schweiker et al. [196]. In general, very few studies have systematically assessed the effect of personal variables other than age and gender on behaviour.

5.4. Physical + contextual + personal variables

This subsection summarizes eleven studies looking at the influence of physical environmental conditions and their interactions with contextual and personal factors on occupant behaviour [256,259–268].

5.4.1. Motivational background

As in previous subsections, modelling of occupants' behaviour for use in building performance simulations for office buildings is the main motivation common in studies across the world. Thereby, the main research focus is on window control behaviour, and its impact on the energy consumption.

5.4.2. Approaches

The majority of the publications were field studies, often based on or involving questionnaire surveys [256,257,262,264]. Almost all the studies used logistic regression to evaluate the crossed main effects of environmental and non-environmental factors on the occupants' behaviour. The analysis of the interactions between different predictors has not been established yet, but there is a growing body of literature with results that point out its importance [267,268].

The four commonly studied behaviours are interactions with windows, use of heating controls, electric lighting use, and interaction with shades.

The physical variables were the internal and outdoor air temperature, globe temperature and air velocity. Some studies collected additional measurements of carbon dioxide concentration, particulate matter [266], and solar radiation [267]. Contextual factors included building features and maintenance, the orientation of windows, floor level (security), the type of office, and socio-cultural aspects such as habits and dress code. Personal factors included perceived control.

The number of residential and office building studies was similar, but residential longitudinal studies usually had a longer duration. Office studies benefit mainly from a large number of respondents albeit often using cross-sectional surveys for shorter periods.

5.4.3. Findings

While all studies observed physical, contextual, and personal variables, window use was mainly analysed as a function of outdoor temperature [256,264], indoor temperature, and IAQ [268]. Often, the probability of an opened window is positively correlated with outdoor temperature, but Kim et al. [262] showed a bell-shaped relationship were above a certain ambient temperature this positive correlation is reversed and the number of closed windows increases again. This effect was observed in previous single-domain studies [269] and shows the importance of local context in the interpretation of the observed behaviours.

In an office building in the hot and humid climate region of India, window use was mostly defined by contextual factors such as the time of day, while the occupants did not interact with other building controls [256]. A study in China [266] concluded that the window use in the studied offices was affected by a combination of physical and contextual factors such as the number of sunshine hours. Wei et al. [265] revealed a seasonal effect and a significant influence of the location of the window (ground floor or not) and personal preference (habitual or not) on the "end-of-day" window state. Absence in subsequent days and contextual factors such as daylight saving time and façade orientation did not have a significant effect. Seasonal effects were also evident in a South Korean study [268]. In spring, window use was affected by the CO2-concentration, whereas in summer the indoor temperature was a significant driver. In winter, indoor temperature and CO2-concentration did not have a statistically significant effect on window use. Yun et al. [257] showed a significant relationship between comfort and perceived control over temperature in NV buildings and highlighted that a change of the windows' state is more likely with high compared to low perceived control [257].

The lighting behaviour in households was found to be influenced by the solar radiation, perceived illumination, outdoor temperature, thermal sensation and IAQ [264] showing the complexity of the interrelationships between multiple physical and personal variables.

The interaction of household occupants with the radiator thermostat set-points showed that the occupants could be classified into different behaviour categories according to the number of interactions with the heating controls [267]. The set-point changes were significantly influenced by the indoor relative humidity, outdoor ambient temperature, solar radiation, wind speed and time of day.

5.4.4. Identified gaps and future directions

The findings show the importance of contextual factors and how these non-physical factors affect occupants' perception and behaviour. They emphasize the need for systematic analysis of contextual factors and for the study of their interactions with physical and personal variables. However, there is a lack of research into the relationships and interactions amongst multi-perceptual, contextual, and personal factors and their combined influence on occupant behaviours. While there seems to be a consensus on the physical variables measured, there are still differences in the selection of contextual and personal variables and their reporting. The type of building system varied with the particularities of the location (e.g. climate, prevailing architecture and construction typologies) and seemed biased by what the sites permitted and the studies' aims. The main reason could be that these parameters are often "fixed", defined by the building and location and not directly controlled by the researchers.

Contextual factors are mainly referred to generically without systematically assessing their interactions and their impact on other predictors. Noted missing relationships include the effect of different climatic and cultural background factors on window use behaviour [257].

In relation to lighting studies, research is required to assess the effect of light on psychological factors and investigate the duration of the effects on comfort [260].

5.5. Physical + multi-behavioural approaches

The focus of the 13 studies in this subsection is on the interrelations between the indoor environmental conditions with a combination of different behavioural responses [212,225,254,258,270–278].

5.5.1. Motivational background

The aim of these studies is related again to energy savings through more realistic modelling of occupancy behaviour in simulations. The underlying objective was to characterise the relationship between physical environmental parameters and occupant behaviour including the assessment of the interactions and combined effect of multiple behaviours.

5.5.2. Approaches

Similarly to the previous subsection, the behaviours investigated were interactions with windows, heating and lighting controls including electric lights and shading.

In contrast to the previous subsection, the research in this field is focused on office buildings. Window use remains the most prominent behaviour and is studied in combination with personal adaptation behaviours (e.g. physiological responses [278], clothing adjustments, and interactions with the heating and cooling systems [274]). Responses to changes in visual conditions are discussed in light of interactions with shades and electric lighting [212].

The physical variables commonly considered were the indoor and outdoor air temperature, relative humidity, wind speed, illuminance, and the level of CO₂-concentration as an indicator of IAQ. The nonphysical variables differed again with the building characteristics and the researchers' objectives and included season, period of day, type of room and current state of controls. However, the analysis of the significance and impact of different variables followed mostly again a crossmain effects approach.

Data collection occurred through surveys with concurrent field measurements, except for one study in a controlled office-like environment [278].

5.5.3. Findings

Despite the influence of indoor and outdoor physical variables, confirming observations of previous subsections, occupancy state (arrival/departure) was the most often studied other behaviour. Langevin et al. [274] found a significant influence of indoor/outdoor temperature and arrival time on clothing, fan, heater, and window use behaviours. While the occupancy state interacted with window opening [236], the previous or next absence for more than 8 h did not have a significant effect on the opening behaviour during departure or the closing behaviour upon arrival [275]. In contrast, the closing behaviour during departure and the opening behaviour upon arrival were influenced by the absence duration. Fabi's review of the physical predictors that influence light switching behaviour identified the key drivers to be absence duration and daylight [273]. Season, light sensor control, and time spent with the light off were not significant predictors. Similarly, lighting use is a function of the daylight availability and the duration of absence before switching the lights on or after switching the lights off [212]. In the intermediate period, the only significant variable is the worktop daylight illuminance level. The same study concluded that the majority of shade adjustments take place during the first arrival or last departure of the day.

Schweiker et al.'s analysis of the interactions between behaviours

indicates a significant impact of fan operation and clothing level on window behaviour but no significant effect of sun shading [278]. In addition, the window state significantly affects fan and sun shading use. Sanati et al. [271] found no significant relationship between sunlight availability, window occlusion, and electric light usage in a single university building.

5.5.4. Identified gaps and future directions

The low number of studies in this subsection showed that the influence amongst the studied behaviours themselves is seldom thoroughly assessed. Fabi et al. [236] suggested that there is a need to investigate the correlation of behaviour responses to multiple, instantaneous factors.

6. Discussion and conclusion

Overall, this review reveals the diversity of approaches and findings of multi-domain analysis. This section compares and discusses the findings and identified gaps from individual subsections above.

6.1. Motivational background

In perceptual studies, the main motivation is a better understanding of the phenomena involved. In behavioural studies, the aim is mostly model development for predictive purposes. This does not mean that perceptual studies do not involve any aspects of prediction, but, the authors stated the focus is on understanding, rather than modelling.

6.2. Approaches

A variety of methodological approaches for research design and assessment are presented in the literature. Whereas laboratory studies are the most frequent type of perceptual multi-physical studies (subsection 4.1), field studies dominate in all other categories. New approaches using virtual environment (e.g. Ref. [272]) published promising results, but still lack sufficient evidence that they permit the reproduction of effects observed in reality.

Geographical contexts are mainly from developed countries (http s://www.un.org/en/development/desa/policy/wesp/wesp_curre

nt/2014wesp_country_classification.pdf), which likely represents the availability of research funding, rather than the contextual diversity or the population size in a particular context. Therefore, the findings presented are not necessarily representative of buildings, lifestyles, climate zones or cultural regions in developing countries.

Context is more likely considered in studies of human behaviour than perception. This may be due to the advantages of laboratory studies to control multi-physical influences on perception without contextual considerations. For example, the experimental design by Kulve et al. [104] enabled researchers to avoid natural correlations among environmental variables and to causally test the effect of variables on outcomes of interest. In addition, it allowed testing if cross-modal effects occurred at a specific level of one variable (e.g., only in comfortable thermal conditions) or were independent of the level of the other variable (i.e. the same cross-modal effect occurred at all the levels of the other variable).

Few studies considered personal variables beyond demographics despite their inclusion by means of questionnaires being an easy extension in laboratory studies. Participants in laboratory studies are not otherwise distracted from their work or leisure activities as would be the case in field studies. Still, the application of findings relating to personal factors in the building design process with generally unknown user profiles is less clear, but potentially beneficial for specific buildings (e.g. retirement homes) or individualized operation strategies.

Contextual influences and occupant behaviour are more difficult to study in an artificial setting of a laboratory environment. The low frequency of interactions (i.e., 1 to 4 actions per day) would require very

long and expensive study periods uncommon in laboratory settings. Still, more attempts would be beneficial to reveal true causalities, because field studies also have drawbacks. The lack of experimental control over environmental conditions means that the conditions cannot be causally related to human outcomes, and that environmental conditions are likely to naturally co-occur in predictable ways (e.g., a position near a window in the summer is likely warmer and brighter than one on the interior of a room).

The question of causality is also relevant to several papers addressing contextual factors, such as green vs. conventional buildings or NV vs. AC buildings in field studies with a limited number of buildings. These studies assign observed differences in perception or behavioural patterns to the type of building, while neglecting the multitude of other potential influences (e.g., non-documented contextual or personal differences). Without addressing, discussing, or eliminating potential confounding variables, assigned causalities could be mistaken. For potential meta-analyses and other comparisons, well-documented contextual elements of the environment under investigation are crucial. Unfortunately, contextual elements and spatial characteristics such as relative position to control devices are often poorly documented – if at all – in the text. Therefore, we recommend using the categories presented in Fig. 1 together with aspects mentioned in previously published ontologies [279] to describe the contextual aspects.

The assessment of the dependent variables varies largely. While there are meaningful differences in behavioural studies, the perceptual studies vary in the dimension assessed (e.g., thermal sensation, preference, or acceptability), and the type of scale (e.g., categorical, continuous). There is a tendency to ignore previous approaches and develop one's own instruments, without benchmarking them against existing ones (see also subsection 4.1.4). As discussed earlier [20], this variety impedes comparing results across studies, and understanding whether differences between outcomes of two studies are a result of the instrument or differences in (unreported) contextual or personal aspects.

In addition to the diversity in data collection approaches, the analysis approaches taken are at different levels. Studies, most often in laboratories, exist, which apply multi-domain approaches from study design to analysis. At the same time, the number of field studies reporting the collection of multi-perceptual data is increasing. However, their potential is poorly utilized, because the multi-perceptual nature of the data is not considered during analysis. The reasons for such omission can be manifold. First, limits in word counts in combination with the complexity of describing multi-physical data and their analysis might lessen the potential to report multi-domain analysis approaches first, but cannot be an argument for missing subsequent publications. Second, multi-domain interaction or cross-over statistical analyses might have been conducted, but not reported due to non-significant results; a common issue leading to scientific bias as reported earlier [5]. Third, a lack of statistical skills might have impeded the integration of interaction terms in statistical analysis.

To overcome these shortcomings, all researchers, reviewers, and editors are encouraged to demand extensive descriptions and analysis methods for multi-domain studies until there is a substantial body of evidence that certain aspects are not relevant for a specific perception or behaviour.

Further research shortcomings in all categories are small sample sizes, low diversity in participants, representativeness of samples, and environment. In contrast to previous reviews' discussions [20], which emphasize the general need for larger sample sizes, we argue that the actual number of cases is not the main problem. Examples exist throughout scientific literature in a variety of disciplines, which show the benefits of studies with small sample sizes that still increase the existing knowledge (see Flyvbjerg [280] for an extended discussion). Small sample sizes are to be criticized when lacking a clear strategy for sample selection and being based on so-called convenience samples, i.e. those at hand of the researcher. In contrast, Flyvbjerg [280] discusses information-oriented sampling strategies including the selection of critical cases or maximum variation cases, which enable the extraction of new knowledge even with small sample sizes. At the same time, he emphasizes that small sample sizes are very suitable for falsification of theories – sometimes a single case is sufficient – but less for generalizing.

6.3. Findings

Overall, results are often inconclusive and in part contradictory (see Figs. 8-12). Few observations are repeatedly shown: significant effects of visual properties on thermal perception exist, though they are partially contradictory and a comparable number of studies found no significant interactions. A general statement seems impossible due to findings, that warm light colours are perceived as satisfactory in cold environments and vice-versa. Thermal properties have been shown to influence acoustic perception, while the number of non-significant findings is again of the same magnitude. Related to occupant behaviour, most evidence was observed for the interaction of thermal and IAQ related variables on window opening behaviour. Such a finding is not surprising given that windows enable occupants to control IAQ and thermal conditions except for reasons of outdoor conditions such as high air pollution. Contradictory results are apparent in all categories of multi-domain studies. While such observation can be attributed to the low number of studies in subsections 4.4 and 5.4 and 5.5, it is more surprising for subsections 4.1, 4.2, 5.1, and 5.2, which are based on a much larger number of items.

Despite the large variety of independent variables assessed, there is a need to clarify whether those variables are the most influential ones to explain variances observed in perception or behaviour, or solely the most accessible ones. This necessity is linked to the next gap in the reviewed literature: a missing theoretical foundation. One could assume that many, if not all studies, are based on underlying theories of human physiology and perception. However, very few articles mention theories when describing their study design or discussing their findings. Not all studies need to be designed to falsify an existing theory; case studies, especially very detailed ones looking at individual cases, are also very suitable to develop new theories inductively. Nevertheless, a theoretical foundation is meaningful to link and explain potentially diverse findings and to justify the selection or exclusion of specific physical, contextual, or personal variables. Theories relevant for multi-domain approaches may originate from disciplines like psychology, sociology, but also from neurology or physiology. One of the few research items mentioning theoretical foundations is Candas et al. [21], who mention neurophysiological aspects related to multisensory integration in their introduction. However, they do not relate their review findings to such approaches. The literature on multisensory integration [281-283] outlines first explanations to what extent interactions can be additive, antagonistic, or synergetic. For example, Talsma et al. [283] propose a framework that shows the interaction between multi-physical perception and attention.

There are few studies linking perception and action. In behavioural studies, physical quantities are assessed which relate to perceptual domains. For example, the assessed indoor air temperature can be related to thermal perception. As such, the perception of such physical indoor environmental qualities is an assumed prerequisite for the action. Given the low observed correlations between observed physical variables and behavioural actions (R² are frequently below 0.2), it might be necessary to include additional variables or to consider different approaches to understand occupant behaviour. Thereby, several aspects are to be considered. First, perceptual studies show a large variance between and within individuals in the perception of the same physical stimuli. Second, theories in the field of psychology together with empirical findings suggest a difference between the intention to perform an action and the action itself [284-286]. Not surprisingly, previous research has revealed a multitude of factors influencing occupant behaviour [5], which potentially affect the relationship between intention and action (e.g., the level of perceived control, the distance to means of control, or other

work tasks that require full attention). Therefore, we recommend looking further at the relationship between perception and action and evaluating whether those contextual and personal factors affecting behaviour effect perception and vice versa.

6.4. Future directions

Based on the results and discussion presented in this review, we propose the following points to be considered by authors and reviewers of future multi-domain approaches.

The first point is easily applicable and points to a limitation of this review: keywords for multi-domain studies. Commonly, an a priori defined set of search terms is used for a systematic review. However, an initial review of keywords used by a selection of relevant multi-domain articles revealed that the keywords for the individual domains investigated are used, but that there is no specific keyword to clarify the multidomain approach. Therefore, a systematic search through a set of keywords would have required searching for all possible combinations of individual domain keywords. Given the number of authors involved and their diverse backgrounds from different domains, we decided to start with the collection of articles known to us in combination with a backward and forward search of cited or citing articles. This strategy might have failed to find all relevant research items. However, articles, which have not been cited or do not cite any of the 200+ articles considered for this review might be of minor relevance and likely not adding much to our general conclusions. Still, we suggest future studies to use a unique keyword such as "multi-domain" or "combined effects" in order to facilitate future review efforts.

Second, researchers should clarify whether their research is intended to explore new influences, i.e. supporting the development of new theories or the extension of existing ones, or test an existing theory. In addition, researchers should clearly state the limitations of their studies, especially when dealing with small samples, discuss the applicability and comparability of results in the context of existing knowledge, and be careful with false causalities arising from unobserved confounding factors. Thereby, generalization is relevant to find common patterns. However, addressing individual differences and revealing factors leading to such differences, even for single cases, is of high importance in order to consider outliers as valuable points of information. The latter assertion is valid either because these points are true outliers and explanations are available (see e.g. O'Brien et al. [287] for a qualitative approach to explain outliers). Or, because they point to methodological issues (e.g., the question asked is prone to misinterpretation under specific circumstances).

Third, advanced statistical analysis methods for capturing interactions and their complexity are recommended. Aside from the application of multiple regression including interaction terms, hierarchical modelling or structural equation modelling, which permit understanding of interdependent relationships, are appropriate methods for this task. Additionally, analysis methods derived from machine learning approaches may be useful to detect underlying patterns in large and rich datasets. When reporting statistical results, significant levels together with effect sizes are crucial information for later meta-analysis.

Fourth, missing agreement on classification of contextual and personal variables leads to the same terms being used for different aspects. Therefore, general classifications (e.g., "green buildings") should be avoided in favour of explicit descriptions (e.g., LEED Platinum certified buildings).

Fifth, interactions are complex by nature. Given the large variety of potential interactions between physical, contextual and personal variables, collective approaches, which build upon the knowledge generated, are necessary. We thus encourage researchers to join or establish collaborative activities such as those developed within international research groups like the IEA EBC Annex 79 "Occupant-Centric Building Design and Operation" (http://annex79.iea-ebc.org/), which is the basis for this review. Moreover, a common framework is necessary, which

facilitates meta-analysis efforts in the future and allows aligning one's own research with previous research. As a start, our review table is available as a dynamic open-access document permitting others to add their research related to multi-domain approaches (https://osf. io/gnvp2/). We hope that this document will serve as a growing knowledge base to increase collectively our knowledge related to multi-domain influences on perception and behaviour.

Sixth is the balance between the benefit and the risk of increasing the complexity of perceptual or behavioural models partly addressed in behavioural studies by means of statistical measures such as Akaikes Information Criterion [288]. Future studies need to investigate under which circumstances additional factors are meaningful, given issues such as over-fitting and error propagation. This question is best answered based on a solid theoretical foundation together with a clear description of the potential application of the results.

Combining all these conclusions necessitates designing studies within a framework of occupant perception and behaviour that accounts for the complexity of the physiological-perception-cognition-decisionaction-automation-building system. First examples for such frameworks have been proposed [289,290] and attempts to challenge them by means of field or laboratory studies are highly recommended. In addition, the development of guidelines on this topic is an expected future development of this work.

Declaration of competing interest

There are no known conflicts of interest.

Acknowledgements

This review was conducted within the framework of IEA-EBC Annex 79.

Marcel Schweiker would like to acknowledge funding received from the Federal Ministry for Economic Affairs and Energy (BMWi) under Grant no. 03EN1002A.

Elie Azar would like to acknowledge the financial support received from the Abu Dhabi Department of Education and Knowledge (ADEK) under Grant AARE18-063.

Salvatore Carlucci and Matteo Favero would like to acknowledge funding received from the Research Council of Norway and the Research Centre on Zero Emission Neighborhoods in Smart Cities (FME ZEN) under Grant no. 257660.

Anna Laura Pisello and Cristina Piselli would like to thank the Italian Ministry of research for supporting the follow up of this investigation through NEXT.COM PRIN 2017 project, 20172FSCH4_002 "Towards the NEXT generation of multiphysics and multidomain environmental COMfort models: theory elaboration and validation experiment".

S.G. would like to thank the Sustainable Energy Research Group (energy.soton.ac.uk) for supporting this work.

Verena M. Barthelmes would like to acknowledge the funding received from the Swiss Federal Office of Energy (SFOE) under contract no. SI/501895-01.

Rune Korsholm Andersen would like to acknowledge the funding received from the Technology Development and Demonstration Program – The Danish Energy Agency (EUDP) under Grant number 64 018-0558.

Shengbo Zhang and Maedot S. Andargie would like to acknowledge the funding support from the Natural Sciences and Engineering Research Council of Canada (NSERC), [RGPIN-2016-06325] and Ontario Early Researcher Award. Shengbo Zhang would also like to acknowledge the funding support from the RDH Building Science Inc.

References

N.E. Klepeis, W.C. Nelson, W.R. Ott, J.P. Robinson, A.M. Tsang, P. Switzer, J. V. Behar, S.C. Hern, W.H. Engelmann, The National Human Activity Pattern

Survey (NHAPS): a resource for assessing exposure to environmental pollutants, J. Expo. Sci. Environ. Epidemiol. 11 (2001) 231–252, https://doi.org/10.1038/sj. jea.7500165.

- [2] C. Schweizer, R.D. Edwards, L. Bayer-Oglesby, W.J. Gauderman, V. Ilacqua, M. Juhani Jantunen, H.K. Lai, M. Nieuwenhuijsen, N. Künzli, Indoor time-microenvironment-activity patterns in seven regions of Europe, J. Expo. Sci. Environ. Epidemiol. 17 (2007) 170–181, https://doi.org/10.1038/sj. jes.7500490.
- [3] Z. Yu, B.C.M. Fung, F. Haghighat, H. Yoshino, E. Morofsky, A systematic procedure to study the influence of occupant behavior on building energy consumption, Energy Build. 43 (2011) 1409–1417, https://doi.org/10.1016/j. enbuild.2011.02.002.
- [4] ASHRAE, Guideline 10-2016 Interactions Affecting the Achievement of Acceptable Indoor Environments, American Society of Heating, Refrigerating and Air-Conditioning Engineering, Atlanta, USA, 2016.
- [5] M. Schweiker, S. Carlucci, R.K. Andersen, B. Dong, W. O'Brien, Occupancy and occupants' actions, in: Exploring Occupant Behavior in Buildings: Methods and Challenges, Springer International Publishing, 2017, pp. 7–38, https://doi.org/ 10.1007/978-3-319-61464-9 2.
- [6] M.A. Ortiz, S.R. Kurvers, P.M. Bluyssen, A review of comfort, health, and energy use: understanding daily energy use and wellbeing for the development of a new approach to study comfort, Energy Build. 152 (2017) 323–335, https://doi.org/ 10.1016/j.enbuild.2017.07.060.
- [7] R.F. Rupp, N.G. Vásquez, R. Lamberts, A review of human thermal comfort in the built environment, Energy Build. 105 (2015) 178–205, https://doi.org/10.1016/ j.enbuild.2015.07.047.
- [8] M. Schweiker, G.M. Huebner, B.R.M. Kingma, R. Kramer, H. Pallubinsky, Drivers of diversity in human thermal perception – a review for holistic comfort models, Temperature 5 (2018) 308–342, https://doi.org/10.1080/ 23328940.2018.1534490.
- [9] L. Edwards, P. Torcellini, Literature Review of the Effects of Natural Light on Building Occupants, NREL, 2006, https://doi.org/10.2172/15000841.
- [10] A.D. Galasiu, J.A. Veitch, Occupant preferences and satisfaction with the luminous environment and control systems in daylit offices: a literature review, Energy Build. 38 (2006) 728–742, https://doi.org/10.1016/j. enbuild/2006.03.001
- [11] C. Pierson, J. Wienold, M. Bodart, Review of factors influencing discomfort glare perception from daylight, LEUKOS - J. Illum. Eng. Soc. North Am. 14 (2018) 111–148, https://doi.org/10.1080/15502724.2018.1428617.
- [12] K.G. Van Den Wymelenberg, Visual comfort, discomfort glare, and occupant fenestration control: developing a research agenda, LEUKOS - J. Illum. Eng. Soc. North Am. 10 (2014) 207–221, https://doi.org/10.1080/ 15502724.2014.93004.
- [13] J.M. Daisey, W.J. Angell, M.G. Apte, Indoor air quality, ventilation and health symptoms in schools: an analysis of existing information, Indoor Air 13 (2003) 53–64.
- [14] N.-G. Vardaxis, D. Bard, K. Persson Waye, Review of acoustic comfort evaluation in dwellings—part I: associations of acoustic field data to subjective responses from building surveys, Build. Acoust. 25 (2018) 151–170, https://doi.org/ 10.1177/1351010X18762687.
- [15] N.-G. Vardaxis, D. Bard, Review of acoustic comfort evaluation in dwellings: part II—impact sound data associated with subjective responses in laboratory tests, Build. Acoust. 25 (2018) 171–192, https://doi.org/10.1177/ 1351010X18772026.
- [16] K.W. Ma, H.M. Wong, C.M. Mak, A systematic review of human perceptual dimensions of sound: meta-analysis of semantic differential method applications to indoor and outdoor sounds, Build. Environ. 133 (2018) 123–150, https://doi. org/10.1016/j.buildenv.2018.02.021.
- [17] T. Hong, S.C. Taylor-Lange, S. D'Oca, D. Yan, S.P. Corgnati, Advances in research and applications of energy-related occupant behavior in buildings, Energy Build. 116 (2016) 694–702, https://doi.org/10.1016/j.enbuild.2015.11.052.
- [18] F. Stazi, F. Naspi, M. D'Orazio, A literature review on driving factors and contextual events influencing occupants' behaviours in buildings, Build. Environ. 118 (2017) 40–66, https://doi.org/10.1016/j.buildenv.2017.03.021.
- [19] K. Van Den Wymelenberg, Patterns of occupant interaction with window blinds: a literature review, Energy Build. 51 (2012) 165–176, https://doi.org/10.1016/j. enbuild.2012.05.008.
- [20] S. Torresin, G. Pernigotto, F. Cappelletti, A. Gasparella, Combined effects of environmental factors on human perception and objective performance: a review of experimental laboratory works, Indoor Air 28 (2018) 525–538, https://doi. org/10.1111/ina.12457.
- [21] V. Candas, A. Dufour, Thermal comfort: multisensory interactions? J. Physiol. Anthropol. Appl. Hum. Sci. 24 (2005) 33–36, https://doi.org/10.2114/jpa.24.33.
- [22] L.H. Centnerová, A.C. Boerstra, Comfort is more than just thermal comfort, adapting to change: new thinking on comfort, 9-11, http://nceub.commoncense. info/uploads/26-01-06-Centnerova.pdf, 2010.
- [23] M. Frontczak, P. Wargocki, Literature survey on how different factors influence human comfort in indoor environments, Build. Environ. 46 (2011) 922-937, https://doi.org/10.1016/j.buildenv.2010.10.021.
- [24] W. O'Brien, H.B. Gunay, F. Tahmasebi, A. Mahdavi, A preliminary study of representing the inter-occupant diversity in occupant modelling, J. Build. Perform. Simul. 10 (2017) 509–526, https://doi.org/10.1080/ 19401493.2016.1261943.
- [25] D. Moher, A. Liberati, J. Tetzlaff, D.G. Altman, T.P.R.I.S.M.A. Group, Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement, PLoS Med. 6 (2009) 1–6, https://doi.org/10.1371/journal.pmed.1000097.

- [26] M.J. Grant, A. Booth, A typology of reviews: an analysis of 14 review types and associated methodologies, Health Inf. Jbr. J. 26 (2009) 91–108, https://doi.org/ 10.1111/j.1471-1842.2009.00848.x.
- [27] M. Frontczak, S. Schiavon, J. Goins, E. Arens, H. Zhang, P. Wargocki, Quantitative relationships between occupant satisfaction and satisfaction aspects of indoor environmental quality and building design, Indoor Air 22 (2012) 119–131, https://doi.org/10.1111/j.1600-0668.2011.00745.x.
 [28] C. Huizenga, S. Abbaszadeh, L. Zagreus, E. Arens, Air quality and thermal comfort
- [28] C. Huizenga, S. Abbaszadeh, L. Zagreus, E. Arens, Air quality and thermal comfort in office buildings: results of a large indoor environmental quality survey. http ://cbe.berkeley.edu, 2006.
- [29] A. Roetzel, M. DeKay, A. Nakai Kidd, A. Klas, A.-M. Sadick, V. Whittem, L. Zinkiewicz, Architectural, indoor environmental, personal and cultural influences on students' selection of A preferred place to study, Architect. Sci. Rev. (2019). https://mc.manuscriptcentral.com/asre.
- [30] R. Hitchings, G. Waitt, K. Roggeveen, C. Chisholm, Winter cold in a summer place: perceived norms of seasonal adaptation and cultures of home heating in Australia, Energy Res. Soc. Sci. 8 (2015) 162–172, https://doi.org/10.1016/j. erss.2015.05.007.
- [31] V.M. Barthelmes, Y. Heo, V. Fabi, S.P. Corgnati, Exploration of the Bayesian Network framework for modelling window control behaviour, Build. Environ. 126 (2017) 318-330, https://doi.org/10.1016/j.buildenv.2017.10.011.
- [32] F. Albers, J. Maier, C. Marggraf-Micheel, In search of evidence for the hue-heat hypothesis in the aircraft cabin, Light. Res. Technol. 47 (2014) 483–494, https:// doi.org/10.1177/1477153514546784.
- [33] C.J. Anumba, Y. Abraham, Z. Zhao, S. Asadi, Innovations for wellbeing occupantrelated energy use: a Qatar office case study, in: World Sustainable Built Environment Conference 2017 Hong Kong, 2017.
- [34] H. Azmoon, H. Dehghan, J. Akbari, S. Souri, The relationship between thermal comfort and light intensity with sleep quality and eye tiredness in shift work nurses, J. Environ. Publ. Health (2013) 1–5, https://doi.org/10.1155/2013/ 639184, 2013.
- [35] I. Balazova, G. Clausen, D.P. Wyon, The Influence of Exposure to Multiple Indoor Environmental Parameters on Human Perception, Performance and Motivation, Clima 2007 WellBeing Indoors, 2007.
- [36] R.R. Baniya, E. Tetri, J. Virtanen, L. Halonen, The effect of correlated colour temperature of lighting on thermal sensation and thermal comfort in a simulated indoor workplace, Indoor Built Environ. 27 (2018) 308–316, https://doi.org/ 10.1177/1420326X16673214.
- [37] C.A. Bennett, P. Rey, What's so hot about red? Hum. Factors 14 (1972) 149-154.
- [38] P.C. Berry, Effect of Colored Illumination upon Perceived Temperature, vol. 1, 1961
- [39] G. Chinazzo, J. Wienold, M. Andersen, Combined effects of daylight transmitted through coloured glazing and indoor temperature on thermal responses and overall comfort, Build. Environ. (2018), https://doi.org/10.1016/j. buildenv.2018.08.045.
- [40] G. Clausen, D.P. Wyon, The combined effects of many different indoor environmental factors on acceptability and office work performance, HVAC R Res. 14 (2008) 103–113, https://doi.org/10.1080/10789669.2008.10390996.
- [41] G. Clausen, L. Carrick, P.O. Fanger, S.W. Kim, T. Poulsen, J.H. Rindel, A comparative study of discomfort caused by indoor air pollution, thermal load and noise, Indoor Air 3 (1993) 255–262.
- [42] N.D. Dahlan, Perceptive-cognitive aspects investigation in relation to indoor environment satisfaction collected from naturally ventilated multi-storey student accommodations in Malaysia, Indoor Built Environ. 24 (2015) 116–127, https:// doi.org/10.1177/1420326X13506449.
- [43] V. Fabi, S.P. Corgnati, R.V. Andersen, M. Filippi, B. Olesen, Effect of occupant behaviour related influencing factors on final energy end uses in buildings, J. Chem. Inf. Model. 53 (2013) 1689–1699, https://doi.org/10.1017/ CBO9781107415324.004.
- [44] L. Fang, G. Clausen, P.O. Fanger, Impact of temperature and humidity on perception of indoor air quality during immediate and longer whole-body exposures, Indoor Air 8 (1998) 276–284, https://doi.org/10.1111/j.1600-0668.1998.00008.x.
- [45] P.O. Fanger, N.O. Breum, E. Jerking, Can colour and noise influence man's thermal comfort? Ergonomics 20 (1977) 11–18, https://doi.org/10.1080/ 00140137708931596.
- [46] S. Gauthier, G. Giakoumis, L. Bourikas, G.M. Giakoumis, P. James, Understanding Window Behaviour in a Mixed-Mode Buildings and the Impact on Energy Performance, 2016.
- [47] Y. Geng, W. Ji, B. Lin, Y. Zhu, The impact of thermal environment on occupant IEQ perception and productivity, Build. Environ. 121 (2017) 158–167, https:// doi.org/10.1016/j.buildenv.2017.05.022.
- [48] L. Huang, Y. Zhu, Q. Ouyang, B. Cao, A study on the effects of thermal, luminous, and acoustic environments on indoor environmental comfort in offices, Build. Environ. 49 (2012) 304-309. https://doi.org/10.1016/j.buildeny.2011.07.022.
- [49] A. Jamrozik, C. Ramos, J. Zhao, J. Bernau, N. Clements, T. Vetting Wolf, B. Bauer, A novel methodology to realistically monitor office occupant reactions and environmental conditions using a living lab, Building and Environment. https:// doi.org/10.1016/j.buildenv.2017.12.024, 2018.
- [50] A.C.K. Lai, K.W. Mui, L.T. Wong, L.Y. Law, An evaluation model for indoor environmental quality (IEQ) acceptance in residential buildings, Energy Build. 41 (2009) 930–936, https://doi.org/10.1016/j.enbuild.2000.03.016.
- [51] L. Lan, P. Wargocki, D.P. Wyon, Z. Lian, Effects of thermal discomfort in an office on perceived air quality, SBS symptoms, physiological responses, and human performance, Indoor Air 21 (2011) 376–390, https://doi.org/10.1111/j.1600-0668.2011.00714.x.

- [52] C. Laurentin, V. Bermtto, M. Fontoynont, Effect of thermal conditions and light source type on visual comfort appraisal, Light. Res. Technol. 32 (2000) 223-233, https://doi.org/10.1177/096032710003200406.
- [53] M.C. Lee, K.W. Mui, L.T. Wong, W.Y. Chan, E.W.M. Lee, C.T. Cheung, Student learning performance and indoor environmental quality (IEQ) in air-conditioned university teaching rooms, Build. Environ. 49 (2012) 238-244.
- F. Martellotta, A. Simone, S. Della Crociata, M. D'Alba, Global comfort and indoor [54] environment quality attributes for workers of a hypermarket in Southern Italy, Build. Environ. 95 (2016) 355-364, https://doi.org/10.1016/j.
- [55] L. Mølhave, Z. Liu, A.H. Jørgensen, O.F. Pedersen, S.K. Kjægaard, Sensory and physiological effects on humans of combined exposures to air temperatures and volatile organic compounds, Indoor Air 3 (1993) 155–169, https://doi.org/ 10.1111/j.1600-0668.1993.t01-1-00002.x.
- [56] K. Nagano, T. Horikoshi, New comfort index during combined conditions of moderate low ambient temperature and traffic noise, Energy Build. 37 (2005) 287-294, https://doi.org/10.1016/j.enbuild.2004.08.001.
- K. Nagano, T. Horikoshi, New index of combined effect of temperature and noise [57] on human comfort: summer experiments on hot ambient temperature and traffic pise, Arch. Compl. Environ. Stud. 13 (2001).
- [58] H. Nakamura, M. Oki, Influence of air temperature on preference for color temperature of general lighting in the room, J. Hum. Environ. Syst. 4 (2000) 41-47, https://doi.org/10.1618/jhes.4.41.
- G. Newsham, S. Mancini, J. Veitch, R. Marchand, W. Lei, K. Charles, C. Arsenault, [59] Control strategies for lighting and ventilation in offices: effects on energy and occupants, Intell. Build. Int. 1 (2009) 101-121, https://doi.org/10.3763 2009 0004
- [60] P.S. Nimlyat, Indoor environmental quality performance and occupants' satisfaction [IEQPOS] as assessment criteria for green healthcare building rating, Build. Environ. 144 (2018) 598-610, https://doi.org/10.1016/j. buildenv.2018.09.003.
- [61] N. Pellerin, V. Candas, Effects of steady-state noise and temperature conditions on environmental perception and acceptability, Indoor Air 14 (2004) 129-136, https://doi.org/10.1046/j.1600-0668.2003.00221.x.
- [62] P. Ricciardi, C. Buratti, Environmental quality of university classrooms subjective and objective evaluation of the thermal, acoustic, and lighting comfort conditions, Build. Environ. 127 (2018) 23–36.
- M. Te Kulve, L. Schellen, L. Schlangen, A. Frijns, W. Van, M. Lichtenbelt, Light [63] intensity and thermal responses, in: Proceedings of Windsor Conference, 2016.
- [64] M. Te Kulve, L. Schlangen, W. van Marken Lichtenbelt, Interactions between the perception of light and temperature, Indoor Air 28 (2018) 881-891, https://doi. rg/10.1111/ina.12500.
- Y. Teramoto, H. Tokura, K. Ohkura, Y. Ohmasa, S. Suho, R. Inoshiri, M. Masuda, [65] Effects of different light intensities during the forenoon on the afternoon thermal sensation in mild cold, J. Therm. Biol. 21 (1996) 339-343, https://doi.org/ 10.1016/S0306-4565(96)00019-8
- [66] D. Tiller, L.M. Wang, A. Musser, M.J. Radik, Combined effects of noise and temperature on human comfort and performance, Build, Eng. 116 (2010) 522-540.
- [67] J. Toftum, A. Thorseth, J. Markvart, Á. Logadóttir, Occupant response to different correlated colour temperatures of white LED lighting, Build. Environ. 143 (2018) 258-268, https://doi.org/10.1016/j.buildenv.2018.07.013
- J. Varjo, V. Hongisto, A. Haapakangas, H. Maula, H. Koskela, J. Hyönä, [68] Simultaneous effects of irrelevant speech, temperature and ventilation rate on performance and satisfaction in open-plan offices, J. Environ. Psychol. 44 (2015) n 2015.08.001. 16-33, https://doi.org/10.1016/
- [69] J. Winzen, F. Albers, C. Marggraf-Micheel, The influence of coloured light in the aircraft cabin on passenger thermal comfort, Light. Res. Technol. 46 (2014) 465-475, htt doi.org/10.1177/1477153513484028
- [70] T. Witterseh, D.P. Wyon, G. Clausen, The effects of moderate heat stress and openplan office noise distraction on SBS symptoms and on the performance of office work, Indoor Air 14 (2004) 30-40, https://doi.org/10.1111/j.1600 0668.2004.00305.x.
- [71] L.T. Wong, K.W. Mui, P.S. Hui, A multivariate-logistic model for acceptance of indoor environmental quality (IEQ) in offices, Build. Environ. 43 (2008) 1-6.
- [72] P. Xue, C. Mak, Z. Ai, A structured approach to overall environmental satisfaction in high-rise residential buildings, Energy Build. 116 (2016) 181–189.
- [73] K. Yamazaki, S. Nomoto, Y. Yokota, T. Murai, The effects of temperature, light, ound on perceived work environment, Build. Eng. 104 (1998) 711.
- [74] Moon Yang, Jeon, Comparison of response scales as measures of indoor environmental perception in combined thermal and acoustic conditions, Sustainability 11 (2019) 3975, https://doi.org/10.3390/su11143975
- W. Yang, H.J. Moon, Cross-modal effects of illuminance and room temperature on [75] indoor environmental perception, Build. Environ. 146 (2018) 280-288, https:// loi.org/10.1016/j.buildenv.2018.10.007
- [76] W. Yang, H.J. Moon, M.J. Kim, Combined effects of short-term noise exposure and hygrothermal conditions on indoor environmental perceptions, Indoor Built Environ. 27 (2018) 1119–1133, https://doi.org/10.1177/1420326X17703774.
- W. Yang, M.J. Kim, H.J. Moon, Effects of indoor temperature and background [77] noise on floor impact noise perception, Indoor Built Environ. 28 (2019) 454-469, ttps://doi.org/10.1177/1420326X17753708
- [78] M.S. Andargie, E. Azar, An applied framework to evaluate the impact of indoor office environmental factors on occupants' comfort and working conditions, Sustain. Cities Soc. 46 (2019) 101447, https://doi.org/10.1016/j scs.2019.101447

- [79] W. Yang, H.J. Moon, Combined effects of acoustic, thermal, and illumination conditions on the comfort of discrete senses and overall indoor environment. Build. Environ. 148 (2019) 623-633, https://doi.org/10.1016/j. buildenv.2018.11.040.
- [80] S.G. Hodder, K. Parsons, The effects of solar radiation on thermal comfort, Int. J. Biometeorol. 51 (2007) 233-250, https://doi.org/10.1007/s00484-006-0050-y.
- W. Yang, H.J. Moon, Combined effects of sound and illuminance on indoor environmental perception, Appl. Acoust. 141 (2018) 136-143, https://doi.org/ 10.1016/j.apacoust.2018.07.008.
- S. Gauthier, B. Liu, G. Huebner, D. Shipworth, Investigating the effect of CO2 [82] concentration on reported thermal comfort, in: Proceedings of CISBAT 2015 International Conference on Future Buildings and Districts, 2015.
- [83] Z. Pan, S.K. Kjaergaard, L. Mølhave, A chamber-experiment investigation of the interaction between perceptions of noise and odor in humans, Int. Arch. Occup. Environ. Health 76 (2003) 598-604, https://doi.org/10.1007/s00420-003-04
- [84] E. Siekierski, V. Roussarie, S. Viollon, S. Ségrétain, S. Bojago, What's so hot about sound?-influence of HVAC sounds on thermal comfort, in: Proceedings of INTERNOISE 2005, 2005. https://www.researchgate.net/p
- [85] H. Ma, W. Nie, Influence of visual factors on noise annoyance evaluation caused by road traffic noise in indoor environment, in: Inter Noise 2014, Melbourne, Australia, 2014.
- [86] I. Golasi, F. Salata, E.de L. Vollaro, A. Peña-García, Influence of lighting colour temperature on indoor thermal perception: a strategy to save energy from the HVAC installations, Energy Build. 185 (2019) 112-122, https://doi.org/10.1016/ d.2018.12.0
- [87] H. Wang, G. Liu, S. Hu, C. Liu, Experimental investigation about thermal effect of colour on thermal sensation and comfort, Energy Build. 173 (2018) 710-718, oi.org/10.1016/j.enbuild.2018.06.008
- [88] H.N. Ho, D. Iwai, Y. Yoshikawa, J. Watanabe, S.Y. Nishida, Combined effects of sound and illuminance on indoor environmental perception, Appl. Acoust. 141 (2018), https://doi.org/10.1016/j.apacoust.2018.07.008, 163–143.
- N. Matsubara, A. Gassho, Y. Kurazumi, Facilitatory Effects of Environmental [89] Sounds on Hue-Heat Phenomena, The International Conference on Acoustics, 2004, pp. 1775-1778.
- [90] M. Ziat, C.A. Balcer, A. Shirtz, T. Rolison, A century later, the hue-heat hypothesis: does color truly affect temperature perception?, in: Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 2016, https://doi.org/10.1007/978-3 42321-0 25
- [91] H. Tang, Y. Ding, B. Singer, Interactions and comprehensive effect of indoor environmental quality factors on occupant satisfaction, Build. Environ. 167 (2019) 106462, https://doi.org/10.1016/j.buildenv.2019.106462
- [92] F. Haldi, D. Robinson, On the unification of thermal perception and adaptive actions, Build. Environ. 45 (2010) 2440-2457, https://doi.org/10.1016/j. 2010 05 010
- [93] B. Cao, Q. Ouyang, Y. Zhu, L. Huang, H. Hu, G. Deng, Development of a multivariate regression model for overall satisfaction in public buildings based on field studies in Beijing and Shanghai, Build. Environ. 47 (2012) 394-399, https:/ oi.org/10.1016/j.buildenv.2011.06.022
- [94] T. Mihai, V. Iordache, Determining the indoor environment quality for an educational building, in: Energy Procedia, Elsevier Ltd, 2016, pp. 566-574, https://doi.org/10.1016/j.egypro.2015.12.246.
 [95] L. Pronk, The Combined Effects of Temperature, Background Noise and Lighting
- on the Non-physical Task Performance of University Students, 2008
- [96] K. Nagano, T. Horikoshi, Development of equi-comfort charts constituted with temperature and noise at 150 and 3 lx, Proc. Indoor Air (2014).
- [97] S.C. Sekhar, S.E. Goh, Thermal comfort and IAQ characteristics of naturally/ mechanically ventilated and air-conditioned bedrooms in a hot and humid climate, Build. Environ. 46 (2011) 1905-1916, https://doi.org/10.1016/j. v.2011.03.012 uilde
- [98] C.-M. Chiang, C.-M. Lai, A study on the comprehensive indicator of indoor environment assessment for occupants' health in Taiwan, Build. Environ. 37 (2002) 387-392, https://doi.org/10.1016/S0360-1323(01)00034
- [99] M. Ncube, S. Riffat, Developing an indoor environment quality tool for assessment of mechanically ventilated office buildings in the UK - a preliminary study, Build. Environ. 53 (2012) 26-33, https://doi.org/10.1016/j. buildenv.2012.01.003.
- [100] F. Fassio, A. Fanchiotti, R.D.L. Vollaro, Linear, non-linear and alternative algorithms in the correlation of IEQ factors with global comfort: a case study, Sustainability 6 (2014) 8113-8127
- [101] A.A. Hettiarachchi, R. Emmanuel, Colour as a psychological agent to manipulate perceived indoor thermal environment for effective energy usage; cases mplemented in Sri Lanka, in: PLEA 2017 Edinburgh, 2017
- [102] G. Chinazzo, J. Wienold, M. Andersen, Variation in thermal, visual and overall comfort evaluation under coloured glazing at different temperature levels, J. Int. Colour Assoc. (2019) 45-54. https://aic-color.org/resources/Docu nents/jaic y 23 05.pdf.
- [103] G. Chinazzo, J. Wienold, M. Andersen, Daylight affects human thermal perception, Sci. Rep. 9 (2019) 13690, https://doi.org/10.1038/s41598-019-
- [104] M. te Kulve, L. Schellen, L.J.M. Schlangen, W.D. van Marken Lichtenbelt, The influence of light on thermal responses, Acta Physiol. 216 (2016) 163-185, https://doi.org/10.1111/apha.12552.

- [105] C. Buratti, E. Belloni, F. Merli, P. Ricciardi, A new index combining thermal, acoustic, and visual comfort of moderate environments in temperate climates, Build. Environ. 139 (2018) 27–37, https://doi.org/10.1016/j. buildenv.2018.04.038.
- [106] G. Chinazzo, L. Pastore, J. Wienold, M. Andersen, A field study investigation on the influence of light level on subjective thermal perception in different seasons, Windsor Conference, Rethinking Comf. (2018).
- Windsor Conference, Rethinking Comf. (2018).
 [107] N. Pellerin, V. Candas, Combined effects of temperature and noise on human discomfort, Physiol. Behav. 78 (2003) 99–106.
- [108] W. Yang, H.J. Moon, Cross-modal effects of noise and thermal conditions on indoor environmental perception and speech recognition, Appl. Acoust. 141 (2018) 1–8, https://doi.org/10.1016/j.apacoust.2018.06.019.
- [109] W. Yang, H.J. Moon, Effects of recorded water sounds on intrusive traffic noise perception under three indoor temperatures, Appl. Acoust. 145 (2019) 234–244, https://doi.org/10.1016/j.apacoust.2018.10.015.
- [110] G. Diaper, The Hawthorne Effect: a fresh examination, Educ. Stud. 16 (1990) 261–267, https://doi.org/10.1080/0305569900160305.
- [111] H.M. Parsons, What happened at Hawthorne? Science 183 (1974) 922, https:// doi.org/10.1126/science.183.4128.922, LP – 932.
- [112] N. Baker, M. Standeven, A behavioural approach to thermal comfort assessment, Int. J. Sol. Energy 19 (1997) 21–35, https://doi.org/10.1080/ 014/25019708014329.
- [113] I. Balazova, G. Clausen, J.H. Rindel, T.T. Poulsen, D.P. Wyon, Open-plan office environments: a laboratory experiment to examine the effect of office noise and temperature on human perception, comfort and office work performance. https: //www.semanticscholar.org/paper/Open-plan-office-environments%3A-A-la boratory-to-the-Balazova-Clausen/10cb448cac558b19a1fa18d052de16cce79141 fc, 2008.
- [114] D. Blankenberger, K. Van Den Wymelenberg, J. Stenson, Visual effects of wood on thermal perception of interior environments, in: ARCC Conference Repository, 2019. https://www.arccjournal.org/index.php/repository/article/view/619.
- [115] P.R. Boyce, N.H. Eklund, S.N. Simpson, Individual lighting control: task performance, mood, and illuminance, J. Illum. Eng. Soc. 29 (2000) 131–142, https://doi.org/10.1080/00994480.2000.10748488.
- [116] Z. Brown, R.J. Cole, Influence of occupants' knowledge on comfort expectations and behaviour, Build. Res. Inf. 37 (2009) 227–245, https://doi.org/10.1080/ 00613210902794135.
- [117] V. Butala, S. Muhič, Perception of air quality and the thermal environment in offices, Indoor Built Environ. 16 (2007) 302–310, https://doi.org/10.1177/ 142032606079886.
- [118] V.L. Castaldo, I. Pigliautile, F. Rosso, F. Cotana, F. De Giorgio, A.L. Pisello, How subjective and non-physical parameters affect occupants' environmental comfort perception, Energy Build. 178 (2018) 107–129, https://doi.org/10.1016/j. enbuild. 2018.08.020.
- [119] J.H. Choi, J. Moon, Impacts of human and spatial factors on user satisfaction in office environments, Build. Environ. 114 (2017) 23–35, https://doi.org/10.1016/ i.buildenv.2016.12.003.
- [120] S.P. Corgnati, M. Filippi, S. Viazzo, Perception of the thermal environment in high school and university classrooms: subjective preferences and thermal comfort, Build. Environ. 42 (2007) 951–959, https://doi.org/10.1016/j. buildenv.2005.10.027.
- [121] N.D. Dahlan, P.J. Jones, D.K. Alexander, E. Salleh, J. Alias, Evidence base prioritisation of indoor comfort perceptions in Malaysian typical multi-storey hostels, Build. Environ. 44 (2009) 2158–2165, https://doi.org/10.1016/j. buildenv.2009.03.010.
- [122] R. de Dear, J. Kim, T. Parkinson, Residential adaptive comfort in a humid subtropical climate—sydney Australia, Energy Build. 158 (2018) 1296–1305, https://doi.org/10.1016/j.enbuild.2017.11.028.
- [123] V. De Giuli, R. Zecchin, L. Corain, L. Salmaso, Measurements of indoor environmental conditions in Italian classrooms and their impact on childrens comfort, Indoor Built Environ. 24 (2015) 689–712, https://doi.org/10.1177/ 1420326X14530586.
- [124] S. Dhaka, J. Mathur, A. Wagner, G.D. Agarwal, V. Garg, Evaluation of thermal environmental conditions and thermal perception at naturally ventilated hostels of undergraduate students in composite climate, Build. Environ. 66 (2013) 42–53, https://doi.org/10.1016/J.BUILDENV.2013.04.015.
- [125] K. Freihoefer, D. Guerin, C. Martin, H.Y. Kim, J.K. Brigham, Occupants' satisfaction with, and physical readings of, thermal, acoustic, and lighting conditions of sustainable office workspaces, Indoor Built Environ. 24 (2015) 457–472, https://doi.org/10.1177/1420326X13514595.
- [126] J.Y. Garretón, R. Rodriguez, A. Pattini, Effects of perceived indoor temperature on daylight glare perception, Build. Res. Inf. 44 (2016) 907–919, https://doi.org/ 10.1080/09613218.2016.1103116.
- [127] Z. Gou, S.S.Y. Lau, Z. Zhang, A comparison of indoor environmental satisfaction between two green buildings and a conventional building in China, J. Green Build. 7 (2012) 89-104, https://doi.org/10.3992/jgb.7.2.89.
- [128] Z. Gou, D. Prasad, S. Siu-Yu Lau, Are green buildings more satisfactory and comfortable? Habitat Int. 39 (2013) 156–161, https://doi.org/10.1016/j. habitatint.2012.12.007.
- [129] M.A. Humphreys, Quantifying occupant comfort: are combined indices of the indoor environment practicable? Bulld. Res. Inf. 33 (2005) 317–325, https://doi. org/10.1080/09613210500161950.
- [130] S. Karjalainen, Thermal comfort and use of thermostats in Finnish homes and offices, Build. Environ. 44 (2009) 1237–1245, https://doi.org/10.1016/j. buildenv.2008.09.002.

- [131] J. Kim, R. De Dear, Impact of different building ventilation modes on occupant expectations of the main IEQ factors, Build. Environ. 57 (2012) 184–193, https:// doi.org/10.1016/j.buildenv.2012.05.003.
- [132] D.H. Kim, K.P. Mansfield, A cross-cultural study on perceived lighting quality and occupants' well-being between UK and South Korea, Energy Build. 119 (2016) 211–217, https://doi.org/10.1016/j.enbuild.2016.03.033.
- [133] Z. Kong, D.M. Utzinger, K. Freihoefer, T. Steege, The impact of interior design on visual discomfort reduction: a field study integrating lighting environments with POE survey, Build. Environ. 138 (2018) 135–148, https://doi.org/10.1016/j. buildenv.2018.04.025.
- [134] P. Kosmopoulos, D. Galanos, D. Anastaselos, A.M. Papadopoulos, An assessment of the overall comfort sensation in workplaces, Int. J. Vent. 10 (2012) 311–322, https://doi.org/10.1080/14733315.2012.11683958.
- [135] Y.S. Lee, D.A. Guerin, Indoor environmental quality related to occupant satisfaction and performance in LEED-certified buildings, Indoor Built Environ. 18 (2009) 293–300, https://doi.org/10.1177/1420326X09105455.
- [136] H.H. Liang, C.P. Chen, R.L. Hwang, W.M. Shih, S.C. Lo, H.Y. Liao, Satisfaction of occupants toward indoor environment quality of certified green office buildings in Taiwan, Build. Environ. 72 (2014) 232–242, https://doi.org/10.1016/j. buildenv.2013.11.007.
- [137] H. Lou, D. Ou, A comparative field study of indoor environmental quality in two types of open-plan offices: open-plan administrative offices and open-plan research offices, Build. Environ. 148 (2019) 394–404, https://doi.org/10.1016/j. buildenv.2018.11.022.
- [138] G. Liu, C. Cen, Q. Zhang, K. Liu, R. Dang, Field study on thermal comfort of passenger at high-speed railway station in transition season, Build. Environ. 108 (2016) 220–229, https://doi.org/10.1016/j.buildenv.2016.09.003.
- [139] G. Mangone, S.R. Kurvers, P.G. Luscuere, Constructing thermal comfort: investigating the effect of vegetation on indoor thermal comfort through a four season thermal comfort quasi-experiment, Build. Environ. 81 (2014) 410-426, https://doi.org/10.1016/j.buildenv.2014.07.019.
- [140] S. Manu, Y. Shukla, R. Rawal, L.E. Thomas, R. de Dear, Field Studies of Thermal Comfort across Multiple Climate Zones for the Subcontinent: India Model for Adaptive Comfort (IMAC), Building and Environment, vol. 98, 2016, pp. 55–70, https://doi.org/10.1016/j.buildenv.2015.12.019.
- [141] Q. Meng, J. Kang, H. Jin, Field study on the influence of spatial and environmental characteristics on the evaluation of subjective loudness and acoustic comfort in underground shopping streets, Appl. Acoust. 74 (2013) 1001–1009, https://doi.org/10.1016/j.apacoust.2013.02.003.
- [142] Q. Meng, S. Zhang, J. Kang, Effects of typical dining styles on conversation behaviours and acoustic perception in restaurants in China, Build. Environ. 121 (2017) 148–157, https://doi.org/10.1016/J.BUILDENV.2017.05.025.
- [143] G. Motalebi, A. Sal Moslehian, E. Hasanzadeh, The most effective indoor environmental quality factors related to worker satisfaction and performance: a case of the administrative office building at Ferdowsi University of Mashhad, Int. J. Occup. Saf. Ergon. (2019), https://doi.org/10.1080/3548.2019.1582886.
- [144] G.R. Newsham, B.J. Birt, C. Arsenault, A.J.L. Thompson, J.A. Veitch, S. Mancini, A.D. Galasiu, B.N. Gover, I.A. MacDonald, G.J. Burns, Do green buildings have better indoor environments? New evidence, Build. Res. Inf. 41 (2013) 415–434, https://doi.org/10.1080/09613218.2013.789951.
- [145] Z. Pei, B. Lin, Y. Liu, Y. Zhu, Comparative study on the indoor environment quality of green office buildings in China with a long-term field measurement and investigation, Build. Environ. 84 (2015) 80–88, https://doi.org/10.1016/j. buildenv.2014.10.015.
- [146] Y. Sakurai, T. Noguchi, G. Horie, N. Matsubara, Quantification of the synthesized evaluation of the combined environment, Energy Build. 14 (1990) 169–173, https://doi.org/10.1016/0378-7788(90)90037-J.
- [147] E.L.J. Sander, A. Caza, P.J. Jordan, Psychological perceptions matter: developing the reactions to the physical work environment scale, Build. Environ. 148 (2019) 338–347, https://doi.org/10.1016/j.buildenv.2018.11.020.
- [148] A. Wagner, E. Gossauer, C. Moosmann, T. Gropp, R. Leonhart, Thermal comfort and workplace occupant satisfaction-Results of field studies in German low energy office buildings, Energy Build. 39 (2007) 758–769, https://doi.org/ 10.1016/j.enbuild.2007.02.013.
- [149] M. Watchman, A. Potvin, C.M.H. Demers, A post-occupancy evaluation of the influence of wood on environmental comfort, BioResources 12 (2017) 8704–8724, https://doi.org/10.15376/biores.12.4.8704.48724.
- [150] X. Xuan, Study of indoor environmental quality and occupant overall comfort and productivity in LEED- and non-LEED-certified healthcare settings, Indoor Built Environ. 27 (2018) 544-560, https://doi.org/10.1177/1420326X16684007.
- [151] K. Yildirim, A. Akalin-Baskaya, M. Celebi, The effects of window proximity, partition height, and gender on perceptions of open-plan offices, J. Environ. Psychol. 27 (2007) 154–165, https://doi.org/10.1016/j.jenvp.2007.01.004.
- [152] A. Zalejska-Jonsson, M. Wilhelmsson, Impact of perceived indoor environment quality on overall satisfaction in Swedish dwellings, Build. Environ. 63 (2013) 134–144, https://doi.org/10.1016/j.buildenv.2013.02.005.
- [153] A. Zalejska-Jonsson, Parameters contributing to occupants' satisfaction: green and conventional residential buildings, Facilities 32 (2014) 411–437, https://doi. org/10.1108/F-03-2013-0021.
- [154] F.H. Rohles, W.V. Wells, Interior design. Comfort and Thermal sensitivity, J. Interior Des. 2 (1976) 36-44, https://doi.org/10.1111/j.1939-1668.1976. tb00392.x.
- [155] M. Stokkermans, I. Vogels, Y. de Kort, I. Heynderickx, Relation between the perceived atmosphere of a lit environment and perceptual attributes of light, Light. Res. Technol. 50 (2018) 1164–1178, https://doi.org/10.1177/ 1477153517722384.

- [156] R. Kelly, J. Mardaljevic, B. Painter, K. Irvine, The long-term evaluation of electrochromic glazing in an open plan office under normal use: project outline, in: Proceedings of Experiencing Light, 2012.
- [157] I.A. Sakellaris, D.E. Saraga, C. Mandin, C. Roda, S. Fossati, Y. De Kluizenaar, P. Carrer, S. Dimitroulopoulou, V.G. Mihucz, T. Szigeti, O. Hänninen, E. De Oliveira Fernandes, J.G. Bartzis, P.M. Bluyssen, Perceived indoor environment and occupants' comfort in European "Modern" office buildings: the OFFICAIR Study, Int. J. Environ. Res. Publ. Health 13 (2016), https://doi.org/10.3390/ jierph13050444.
- [158] J. Kim, R. de Dear, Nonlinear relationships between individual IEQ factors and overall workspace satisfaction, Build. Environ. 49 (2012) 33–40, https://doi.org/ 10.1016/j.buildenv.2011.09.022.
- [159] J.H. Choi, V. Loftness, A. Aziz, Post-occupancy evaluation of 20 office buildings as basis for future IEQ standards and guidelines, Energy and Buildings, 2012, pp. 167–175, https://doi.org/10.1016/j.enbuild.2011.08.009.
- [160] S. Kumar, M.K. Singh, V. Loftness, J. Mathur, S. Mathur, Thermal comfort assessment and characteristics of occupant's behaviour in naturally ventilated buildings in composite climate of India, Energy Sustain. Dev. 33 (2016) 108–121, https://doi.org/10.1016/j.esd.2016.06.002.
- [161] I.E. Bennet, W. O'Brien, Field study of thermal comfort and occupant satisfaction in Canadian condominiums, Architect. Sci. Rev. 60 (2017) 27–39, https://doi. org/10.1080/00038628.2016.1205179.
- [162] A.K. Persily, Evaluating building IAQ and ventilation with indoor carbon dioxide, Build. Eng. 103 (1997) 1–12.
- [163] K.W. Tham, H.C. Willem, Temperature and ventilation effects on performance and neurobehavioral-related symptoms of tropically acclimatized call center operators near thermal neutrality, Build. Eng. 111 (2005) 687–698.
- [164] M. Wilson, F. Nicol, in: Some Thoughts on Acoustic Comfort: a Look at Adaptive Standards for Noise, Institute of Acoustics, 2003, pp. 116–124.
- [165] R.W. Marans, X. ying Yan, Lighting quality and environmental satisfaction in open and enclosed offices, J. Architect. Plann. Res. 6 (1989) 118–131. http:// www.jstor.org/stable/43028916.
- [166] A. Astolfi, F. Pellerey, Subjective and objective assessment of acoustical and overall environmental quality in secondary school classrooms, J. Acoust. Soc. Am. 123 (2008) 163–173, https://doi.org/10.1121/1.2816563.
- [167] J.F. Nicol, M.A. Humphreys, A stochastic approach to thermal comfort–occupant behavior and energy use in buildings, Build. Eng. 110 (2004) 554–568.
- [168] G.J. Levermore, D. Meyers, Occupant questionnaire on interior environmental conditions: initial results, Build. Serv. Eng. Technol. 17 (1996) 29–36, https:// doi.org/10.1177/014362449601700105.
- [169] M. Boubekri, Appraisal of the lighting conditions in an office building: results of a survey, Indoor Environ. 4 (1995) 162–169, https://doi.org/10.1177/ 142032659500400306.
- [170] J.A. Veitch, K.E. Charles, K.M.J. Farley, G.R. Newsham, A model of satisfaction with open-plan office conditions: COPE field findings, J. Environ. Psychol. 27 (2007) 177–189, https://doi.org/10.1016/j.jenvp.2007.04.002.
- [171] I.G. Monfared, S. Sharples, Occupants perceptions and expectations of a green office building: a longitudinal case study, Architect. Sci. Rev. 54 (2011) 344–355, https://doi.org/10.1080/00038628.2011.613636.
- [172] E.M. De Korte, M. Spiekman, L. Hoes-van Oeffelen, B. van der Zande, G. Vissenberg, G. Huiskes, L.F.M. Kuijt-Evers, Personal environmental control: effects of pre-set conditions for heating and lighting on personal settings, task performance and comfort experience, Build. Environ. 86 (2015) 166–176, https://doi.org/10.1016/j.buildenv.2015.01.002.
- [173] T. Moore, D.J. Carter, A. Slater, A study of opinion in offices with and without user controlled lighting, light. Res. Technol. 36 (2004) 131-146, https://doi.org/ 10.1191/1365782804111090a.
- [174] E.V. White, B. Gatersleben, Greenery on residential buildings: does it affect preferences and perceptions of beauty? J. Environ. Psychol. 31 (2011) 89–98, https://doi.org/10.1016/j.jenvp.2010.11.002.
- [175] I. Bennet, W. O'brien, H.B. Gunay, Effect of Window Blind Use in Residential Buildings: Observation and Simulation Study, 2014.
- [176] Y. Geng, J. Yu, B. Lin, Z. Wang, Y. Huang, Impact of individual IEQ factors on passengers' overall satisfaction in Chinese airport terminals, Build. Environ. 112 (2017) 241–249, https://doi.org/10.1016/j.buildenv.2016.11.040.
- [177] S. Leder, G.R. Newsham, J.A. Veitch, S. Mancini, K.E. Charles, Effects of office environment on employee satisfaction: a new analysis, Build. Res. Inf. 44 (2016) 34–50, https://doi.org/10.1080/09613218.2014.1003176.
- [178] J. Kim, T. Hong, J. Jeong, C. Koo, M. Kong, An integrated psychological response score of the occupants based on their activities and the indoor environmental quality condition changes, Build. Environ. 123 (2017) 66–77, https://doi.org/ 10.1016/j.buildenv.2017.06.046.
- [179] I.A. Raja, J.F. Nicol, K.J. McCartney, M.A. Humphreys, Thermal comfort: use of controls in naturally ventilated buildings, Energy Build. 33 (2001) 235–244, https://doi.org/10.1016/S0378-7788(00)00087-6.
- [180] K. Voss, S. Herkel, J. Pfafferott, G. Löhnert, A. Wagner, Energy efficient office buildings with passive cooling - results and experiences from a research and demonstration programme, Sol. Energy 81 (2007) 424-434, https://doi.org/ 10.1016/j.solener.2006.04.008.
- [181] L. Chatzidiakou, D. Mumovic, A. Summerfield, Is CO2 a good proxy for indoor air quality in classrooms? Part 2: health outcomes and perceived indoor air quality in relation to classroom exposure and building characteristics, Build. Serv. Eng. Technol. 36 (2015) 162–181, https://doi.org/10.1177/0143624414566245.
 [182] F.H. Rohles, W. Wells, The role of environmental antecedents on subsequent
- [182] F.H. Rohles, W. Wells, The role of environmental antecedents on subsequent thermal comfort, Build. Eng. 83 (1977) 21–29.

- [183] Z. Yang, B. Becerik-Gerber, L. Mino, A study on student perceptions of higher education classrooms: impact of classroom attributes on student satisfaction and performance, Build. Environ. 70 (2013) 171–188, https://doi.org/10.1016/J. BUILDENV.2013.08.030.
- [184] S. Ravindu, R. Rameezdeen, J. Zuo, Z. Zhou, R. Chandratilake, Indoor environment quality of green buildings: case study of an LEED platinum certified factory in a warm humid tropical climate, Build. Environ. 84 (2015) 105–113, https://doi.org/10.1016/j.buildenv.2014.11.001.
- [185] J. Kim, Y. Zhou, S. Schiavon, P. Raftery, G. Brager, Personal comfort models: predicting individuals thermal preference using occupant heating and cooling behavior and machine learning, Build. Environ. 129 (2018) 96–106, https://doi org/10.1016/j.buildenv.2017.12.011.
- [186] L. Chatzidiakou, D. Mumovic, A. Summerfield, Is CO2 a good proxy for indoor air quality in classrooms? Part 1: the interrelationships between thermal conditions CO2 levels, ventilation rates and selected indoor pollutants, Build. Serv. Eng. Technol. 36 (2015) 129–161, https://doi.org/10.1177/0143624414566244.
- [187] C. Vásquez, F.E. Pino, A.P. Hoces, C. Aguirre Nuñez, Thermal and lighting perception in four fully glazed office buildings in Santiago, Chile, J. Facade Des. Eng. 1 (2013) 31–51, https://doi.org/10.3233/FDE-130007.
- [188] D.K. Serghides, C.K. Chatzinikola, M.C. Katafygiotou, Comparative studies of the occupants' behaviour a university building during winter and summer time, Int. J. Sustain. Energy 34 (2015) 528–551.
- [189] V. De Giuli, R. Zecchin, L. Corain, L. Salmaso, Measured and perceived environmental comfort: field monitoring in an Italian school, Appl. Ergon. 45 (2014) 1035–1047, https://doi.org/10.1016/j.apergo.2014.01.004.
- [190] G.K. Oral, A.K. Yener, N.T. Bayazit, Building envelope design with the objective to ensure thermal, visual and acoustic comfort conditions, Build. Environ. 39 (2004) 281–287, https://doi.org/10.1016/S0360-1323(03)00141-0.
- [191] G.F. Menzies, J.R. Wherrett, Windows in the workplace: examining issues of environmental sustainability and occupant comfort in the selection of multiglazed windows, Energy Build. 37 (2005) 623–630, https://doi.org/10.1016/j. enbuild.2004.09.012.
- [192] L. Karlsen, P. Heiselberg, I. Bryn, Occupant satisfaction with two blind control strategies: slats closed and slats in cut-off position, Sol. Energy 115 (2015) 166–179, https://doi.org/10.1016/j.solener.2015.02.031.
- [193] R. Dang, L. Wei, Y. Yuan, G. Liu, The impact of physical environments in satisfaction in shopping centers, in: 7th International Building Physics Conference, IBPC2018, 2018, pp. 799–804, https://doi.org/10.14305/ibpc.2018.ie-2.06.
 [194] I. Pigliautile, S. Casaccia, A. Calvaresi, N. Morresi, M. Arnesano, AL. Pisello, G.
- [194] I. Pigliautile, S. Casaccia, A. Calvaresi, N. Morresi, M. Arnesano, A.L. Pisello, G. M. Revel, A comprehensive human comfort assessment protocol based on multidomain measurements and surveys, in: 51st AiCARR International Conference the Human Dimension of Building Energy Performance, Venice, Italy, 2019, pp. 49–61.
- [195] A.S. Robertson, M. Mcinnes, D. Glass, G. Dalton, P.S. Burge, Building sickness, are symptoms related to the office lighting? Ann. Occup. Hyg. 33 (1989) 47–59, https://doi.org/10.1093/annhyg/33.1.47.
- [196] M. Schweiker, M. Hawighorst, A. Wagner, The influence of personality traits on occupant behavioural patterns, Energy Build. 131 (2016) 63–75, https://doi.org/ 10.1016/j.enbuild.2016.09.019.
- [197] J.A. Veitch, R. Gifford, Choice, perceived control, and performance decrements in the physical environment, J. Environ. Psychol. 16 (1996) 269–276, https://doi. org/10.1006/jevp.1996.0022.
- [198] J.Ä. Veitch, G.R. Newsham, Exercised control, lighting choices, and energy use: an office simulation experiment, J. Environ. Psychol. 20 (2000) 219–237, https:// doi.org/10.1006/jevp.1999.0169.
- [199] G.Y. Yun, Influences of perceived control on thermal comfort and energy use in buildings, Energy Build. 158 (2018) 822–830, https://doi.org/10.1016/j. enbuild.2017.10.044.
- [200] X. Zhang, P. Wargocki, Z. Lian, C. Thyregod, Effects of exposure to carbon dioxide and bioeffluents on perceived air quality, self-assessed acute health symptoms, and cognitive performance, Indoor Air 27 (2017) 47–64, https://doi.org/ 10.1111/ina.12284.
- [201] T.C. Greene, P.A. Bell, Additional considerations concerning the effects of warm and cool wall colours on energy conservation, Ergonomics 23 (1980) 949–954, https://doi.org/10.1080/00140138008924804.
- [202] Y. Al Horr, M. Arif, A. Kaushik, A. Mazroei, M. Katafygiotou, E. Elsarrag, Occupant productivity and office indoor environment quality: a review of the literature, Build. Environ. 105 (2016) 369–389, https://doi.org/10.1016/j. buildenv.2016.06.001.
- [203] G.Y. Yun, K. Steemers, Time-dependent occupant behaviour models of window control in summer, Build. Environ. 43 (2008) 1471–1482, https://doi.org/ 10.1016/j.buildenv.2007.08.001.
- [204] N. Pivac, S. Nižetić, V. Zanki, Occupant behavior and thermal comfort field analysis in typical educational research institution A case study, Therm. Sci. 22 (2018) 785–795, https://doi.org/10.2298/TSCII70915013P.
- [205] S.A. Sadeghi, P. Karava, I. Konstantzos, A. Tzempelikos, Occupant interactions with shading and lighting systems using different control interfaces: a pilot field study, Build. Environ. 97 (2016) 177–195, https://doi.org/10.1016/j. buildenv.2015.12.008.
- [206] M. Schweiker, A. Wagner, The effect of occupancy on perceived control, neutral temperature, and behavioral patterns, Energy Build. 117 (2016) 246–259, https://doi.org/10.1016/j.enbuild.2015.10.051.
- [207] A. Boerstra, T. Beuker, M. Loomans, J. Hensen, Impact of perceived control on comfort and health in European office buildings, in: 10th International Conference on Healthy Buildings 2012, 2012, pp. 370–375. http://nceub.org.uk.

- [208] J. Kim, S. Schiavon, G. Brager, Personal comfort models a new paradigm in thermal confort for occupant-centric environmental control, Build. Environ. 132 (2018) 114–124, https://doi.org/10.1016/j.buildenv.2018.01.023.
- [209] M. Indraganti, K.D. Rao, Effect of age, gender, economic group and tenure on thermal comfort: a field study in residential buildings in hot and dry climate with seasonal variations, Energy Build. 42 (2010) 273–281, https://doi.org/10.1016/j. enbuild.2009.09.003.
- [210] S.H. Park, P.J. Lee, Reaction to floor impact noise in multi-storey residential buildings: the effects of acoustic and non-acoustic factors, Appl. Acoust. 150 (2019) 268–278, https://doi.org/10.1016/j.apacoust.2019.02.021.
- [211] H. Jin, X. Li, J. Kang, Z. Kong, An evaluation of the lighting environment in the public space of shopping centres, Build. Environ. 115 (2017) 228–235, https:// doi.org/10.1016/J.BUILDENV.2017.01.008.
- [212] P.C. da Silva, V. Leal, M. Andersen, Occupants interaction with electric lighting and shading systems in real single-occupied offices: results from a monitoring campaign, Build. Environ. 64 (2013) 152–168, https://doi.org/10.1016/j. buildenv.2013.03.015.
- [213] D. Daum, F. Haldi, N. Morel, A personalized measure of thermal comfort for building controls, Build. Environ. 46 (2011) 3–11, https://doi.org/10.1016/j buildenv.2010.06.011.
- [214] H. Eun Kim, H. Tokura, Influence of different light intensities during the daytime on evening dressing behavior in the cold, Physiol. Behav. 58 (1995) 779–783, https://doi.org/10.1016/0031-9384(95)00129-7.
- [215] V. Inkarojrit, Monitoring and modelling of manually-controlled Venetian blinds in private offices: a pilot study, J. Build. Perform. Simul. 1 (2008) 75–89, https:// doi.org/10.1080/14901490802021012.
- [216] B. Jeong, J.-W. Jeong, J.S. Park, Occupant behavior regarding the manual control of windows in residential buildings, Energy Build. 127 (2016) 206–216, https:// doi.org/10.1016/j.enbuild.2016.05.097.
- [217] A. Kim, S. Wang, J.E. Kim, D. Reed, Indoor/outdoor environmental parameters and window-opening behavior: a structural equation modeling analysis, Buildings 9 (2019), https://doi.org/10.3390/buildings9040094.
 [218] H.-E. Kim, H. Tokura, Influence of Two Different Light Intensities from 16:00 to
- [218] H.-E. Kim, H. Tokura, Influence of Two Different Light Intensities from 16:00 to 20:30 Hours on Evening Dressing Behavior in the Cold, 2007.
- [219] H.E. Kim, H. Tokura, Influence of Light Intensities on Dressing Behavior in Elderly People, 2000.
- [220] E.S. Lee, D.L. DiBartolomeo, S.E. Selkowitz, Thermal and daylighting performance of an automated Venetian blind and lighting system in a full-scale private office, Energy Build. 29 (1998) 47–63, https://doi.org/10.1016/s0378-7788(98)00035-8.
- [221] N. Li, J. Li, R. Fan, H. Jia, Probability of occupant operation of windows during transition seasons in office buildings, Renew. Energy 73 (2015) 84–91, https:// doi.org/10.1016/j.renene.2014.05.065.
- [222] F. Naspi, M. Arnesano, L. Zampetti, F. Stazi, G.M. Revel, M. D'Orazio, Experimental study on occupants' interaction with windows and lights in Mediterranean offices during the non-heating season, Build. Environ. 127 (2018) 221–238, https://doi.org/10.1016/j.buildenv.2017.11.009.
- [223] G. Ozcelik, B. Becerik-Gerber, R. Chugh, Understanding human-building interactions under multimodal discomfort, Build. Environ. 151 (2019) 280–290, https://doi.org/10.1016/J.BUILDENV.2018.12.046.
- [224] J. Park, C.-S. Choi, Modeling occupant behavior of the manual control of windows in residential buildings, Indoor Air 29 (2019) 242-251, https://doi.org/10.1111/ ina.12522.
- [225] C.F. Reinhart, K. Voss, F. C, K. Voss, Monitoring manual control of electric lighting and blinds Reinhart, Light. Res. Technol. 35 (2003) 243–260. NRCC-45701.
- [226] S. Shi, B. Zhao, Occupants' interactions with windows in 8 residential apartments in Beijing and Nanjing, China, Build. Simul. 9 (2016) 221–231, https://doi.org/ 10.1007/s12273-015-0266-z.
- [227] M. Yao, B. Zhao, Window opening behavior of occupants in residential buildings in Beijing, Build. Environ. 124 (2017) 441–449, https://doi.org/10.1016/j. buildenv.2017.08.035.
- [228] M. Yao, B. Zhao, Factors affecting occupants' interactions with windows in residential buildings in Beijing, China, in: W. Cui, P. J. Liu, Zhang (Eds.), 10TH INTERNATIONAL SYMPOSIUM ON HEATING, VENTILATION AND AIR CONDITIONING, 2017, pp. 3428–3434, https://doi.org/ 10.1016/j.proeng.2017.09.857. ISHVAC2017.
- [229] Y. Zhang, P. Barrett, Factors influencing occupants' blind-control behaviour in a naturally ventilated office building, Build. Environ. 54 (2012) 137–147, https:// doi.org/10.1016/j.buildenv.2012.02.016.
- [230] K. Paunović, B. Jakovljević, G. Belojević, Predictors of noise annoyance in noisy and quiet urban streets, Sci. Total Environ. 407 (2009) 3707–3711.
- [231] R. Markovic, E. Grintal, D. Wölki, J. Frisch, C. van Treeck, Window opening model using deep learning methods, Build. Environ. 145 (2018) 319–329, https://doi.org/10.1016/j.buildenv.2018.09.024.
- [232] A.R. Hansen, K. Gram-Hanssen, H.N. Knudsen, How building design and technologies influence heat-related habits, Build. Res. Inf. 46 (2018) 83–98, https://doi.org/10.1080/09613218.2017.1335477.
- [233] A. Mahdavi, A. Mohammadi, E. Kabir, L. Lambeva, Occupants' operation of lighting and shading systems in office buildings, J. Build. Perform. Simul. 1 (2008) 57-65, https://doi.org/10.1080/19401490801906502.
- [234] J. Kim, F. Bauman, P. Raftery, E. Arens, H. Zhang, G. Fierro, M. Andersen, D. Culler, Occupant comfort and behavior: high-resolution data from a 6-month field study of personal comfort systems with 37 real office workers, Build. Environ. 148 (2019) 348–360, https://doi.org/10.1016/j.buildenv.2018.11.012.

- [235] R. Andersen, V. Fabi, J. Toftum, S.P. Corgnati, B.W. Olesen, Window opening behaviour modelled from measurements in Danish dwellings, Build. Environ. 69 (2013) 101–113, https://doi.org/10.1016/j.builden.2013.07.005.
- [236] V. Fabi, V. Maggiora, S.P. Corgnati, R.K. Andersen, Occupants' behaviour in office buildings: stochastic models for window opening, in: Proceedings of 8th Windsor Conference: Counting the Cost of Comfort in a Changing World Cumberland Lodge, Windsor, UK, London, 2014. https://www.researchgate.net/publication/ 261794473.
- [237] S. DOca, T. Hong, A data-mining approach to discover patterns of window opening and closing behavior in offices, Build. Environ. 82 (2014) 726–739, https://doi.org/10.1016/j.buildenv.2014.10.021.
- [238] H.B. Rijal, M.A. Humphreys, J.F. Nicol, Understanding occupant behaviour: the use of controls in mixed-mode office buildings, Build. Res. Inf. 37 (2009) 381–396, https://doi.org/10.1080/09613210902904221.
- [239] F. Haldi, D. Robinson, Adaptive actions on shading devices in response to local visual stimuli, J. Build. Perform. Simul. 3 (2010) 135–153, https://doi.org/ 10.1080/19401490903580759.
- [240] R.V. Jones, A. Fuertes, E. Gregori, A. Giretti, Stochastic behavioural models of occupants' main bedroom window operation for UK residential buildings, Build. Environ. 118 (2017) 144–158, https://doi.org/10.1016/j.buildenv.2017.03.033.
- [241] V.M. Barthelmes, Y. Heo, R.K. Andersen, V. Fabi, S.P. Corgnati, Towards A Comprehensive Model of Window Control Behaviour: A Survey-Based Investigation on Interdisciplinary Drivers in Danish Dwellings, 2018.
- [242] V. Fabi, V. Camisassi, F. Causone, S.P. Corgnati, R. Andersen, Light switch behaviour : occupant behaviour stochastic models in office buildings, in: 8th Windsor Conference: Counting the Cost of Comfort in a Changing World, 2014, pp. 10–13.
- [243] Å. Roetzel, Spatial considerations in modelling assumptions for manual lighting control in offices, in: AIRAH and IBPSA's Australasian Building Simulation 2017, 2017.
- [244] Z. Shi, H. Qian, X. Zheng, Z. Lv, Y. Li, Liu, P.V. Nielsen, Seasonal variation of window opening behaviors in two naturally ventilated hospital wards, Build. Environ. 130 (2018) 85–93, https://doi.org/10.1016/j.buildenv.2017.12.019.
- [245] L.V. Madsen, Materialities shape practices and notions of comfort in everyday life, Build. Res. Inf. 46 (2018) 71–82, https://doi.org/10.1080/ 09613218.2017.1326230.
- [246] V. Inkarojrit, Balancing Comfort : Occupants' Control of Window Blinds in Private Offices, PhD thesis, University of California, Berkeley, 2005.
- [247] M. Indraganti, R. Ooka, H.B. Rijal, G.S. Brager, Adaptive model of thermal comfort for offices in hot and humid climates of India, Build. Environ. 74 (2014) 39–53, https://doi.org/10.1016/j.buildenv.2014.01.002.
- [248] H.B. Rijal, P. Tuohy, M.A. Humphreys, F. Nicol, A. Samuel, An algorithm to represent occupant use of windows and fans including situation-specific motivations and constraints, Build. Simul. 4 (2011) 117–134, https://doi.org/ 10.1007/s12273-011-0037-4.
- [249] F. Haldi, D. Robinson, Modelling occupants' personal characteristics for thermal comfort prediction, Int. J. Biometeorol. 55 (2011) 681–694, https://doi.org/ 10.1007/s00484-01-0383-4.
- [250] M.S. Mustapa, S.A. Zaki, H.B. Rijal, A. Hagishima, M.S.M. Ali, Thermal comfort and occupant adaptive behaviour in Japanese university buildings with free running and cooling mode offices during summer, Build. Environ. 105 (2016) 332–342, https://doi.org/10.1016/j.buildenv.2016.06.014.
- [251] M.S. Rea, Window blind occlusion: a pilot study, Build. Environ. 19 (1984) 133–137, https://doi.org/10.1016/0360-1323(84)90038-6.
- [252] D. Cali, R.K. Andersen, D. Mueller, B.W. Olesen, Analysis of occupants' behavior related to the use of windows in German households, Build. Environ. 103 (2016) 54–69, https://doi.org/10.1016/j.buildenv.2016.03.024.
- [253] J. Yao, Determining the energy performance of manually controlled solar shades: a stochastic model based co-simulation analysis, Appl. Energy 127 (2014) 64–80, https://doi.org/10.1016/j.apenergy.2014.04.046.
- [254] F. Haldi, D. Robinson, Interactions with window openings by office occupants, Build. Environ. 44 (2009) 2378–2395, https://doi.org/10.1016/j. buildenv.2009.03.025.
- [255] V. Fabi, R.V. Andersen, S. Corgnati, B.W. Olesen, Occupants' window opening behaviour: a literature review of factors influencing occupant behaviour and models, Build. Environ. 58 (2012) 188–198, https://doi.org/10.1016/j. buildenv.2012.07.009.
- [256] M. Indraganti, R. Ooka, H.B. Rijal, G.S. Brager, Drivers and barriers to occupant adaptation in offices in India, Architect. Sci. Rev. 58 (2015) 77–86, https://doi. org/10.1080/00038628.2014.976539.
- [257] G.Y. Yun, K. Steemers, N. Baker, Natural ventilation in practice: linking facade design, thermal performance, occupant perception and control, Build. Res. Inf. 36 (2008) 608–624, https://doi.org/10.1086/09613210802417241.
- [258] H.B. Gunay, W. O'Brien, I. Beausoleil-Morrison, S. Gilani, Development and implementation of an adaptive lighting and blinds control algorithm, Build. Environ. 113 (2017) 185–199, https://doi.org/10.1016/j.buildenv.2016.08.027.
- [259] J. Langevin, J. Wen, P.L. Gurian, Simulating the human-building interaction: development and validation of an agent-based model of office occupant behaviors, Build. Environ. 88 (2015) 27–45, https://doi.org/10.1016/j. buildenv.2014.11.037.
- [260] G.M. Huebner, D.T. Shipworth, S. Gauthier, C. Witzel, P. Raynham, W. Chan, Saving energy with light? Experimental studies assessing the impact of colour temperature on thermal comfort, Energy Res. Soc. Sci. 15 (2016) 45–57, https:// doi.org/10.1016/j.erss.2016.02.008.

- [261] B. Gucyeter, Evaluating diverse patterns of occupant behavior regarding controlbased activities in energy performance simulation, Front. Architect. Res. 7 (2018) 167–179, https://doi.org/10.1016/j.foar.2018.03.002.
- [262] J. Kim, R. de Dear, T. Parkinson, C. Candido, Understanding patterns of adaptive comfort behaviour in the Sydney mixed-mode residential context, Energy Build. 141 (2017) 274-283, https://doi.org/10.1016/j.enbuild.2017.02.061.
- [263] Y. Song, Y. Sun, S. Luo, Z. Tian, J. Hou, J. Kim, T. Parkinson, R. de Dear, Residential adaptive comfort in a humid continental climate – tianjin China, Energy Build. 170 (2018) 115–121, https://doi.org/10.1016/j. enbuild.2018.03.083.
- [264] R.V. Andersen, J. Toftum, K.K. Andersen, B.W. Olesen, Survey of occupant behaviour and control of indoor environment in Danish dwellings, Energy Build. 41 (2009) 11–16, https://doi.org/10.1016/j.enbuild.2008.07.004.
- [265] S. Wei, R. Buswell, D. Loveday, Factors affecting 'end-of-day' window position in a non-air-conditioned office building, Energy Build. 62 (2013) 87–96, https://doi. org/10.1016/j.enbuild.2013,02.060.
- [266] S. Pan, Y. Xiong, Y. Han, X. Zhang, L. Xia, S. Wei, J. Wu, M. Han, A study on influential factors of occupant window-opening behavior in an office building in China, Build. Environ. 133 (2018) 41–50, https://doi.org/10.1016/j. buildenv.2018.02.008.
- [267] V. Fabi, R.V. Andersen, S.P. Corgnati, Influence of occupant's heating set-point preferences on indoor environmental quality and heating demand in residential buildings, HVAC R Res. 19 (2013) 635–645, https://doi.org/10.1080/ 1078666 2013 789372
- [268] G.Y. Yun, H. Kim, J.T. Kim, Thermal and non-thermal stimuli for the use of windows in offices, Indoor Built Environ. 21 (2012) 109–121, https://doi.org/ 10.1177/1420326X11420012.
- [269] M. Schweiker, F. Haldi, M. Shukuya, D. Robinson, Verification of stochastic models of window opening behaviour for residential buildings, J. Build. Perform. Simul. 5 (2012) 55-74, https://doi.org/10.1080/19401493.2011.567422.
- [270] S. Wei, R. Jones, P. De Wilde, Driving factors for occupant-controlled space heating in residential buildings, Energy Build. 70 (2014) 36–44, https://doi.org/ 10.1016/j.enbuild.2013.11.001.
- [271] L. Sanati, M. Utzinger, The effect of window shading design on occupant use of blinds and electric lighting, Build. Environ. 64 (2013) 67–76, https://doi.org/ 10.1016/j.buildenv.2013.02.013.
- [272] A. Heydarian, E. Pantazis, J.P. Carneiro, D. Gerber, B. Becerik-Gerber, Lights, building, action: impact of default lighting settings on occupant behaviour, J. Environ. Psychol. 48 (2016) 212-223, https://doi.org/10.1016/j. jenvp.2016.11.001.
- [273] V. Fabi, R.K. Andersen, S. Corgnati, Accounting for the uncertainty related to building occupants with regards to visual comfort : a literature survey on drivers and models. Buildings 6 (2016), https://doi.org/10.3390/buildings6010005.
- [274] J. Langevin, P.L. Gurian, J. Wen, Tracking the human-building interaction: a longitudinal field study of occupant behavior in air-conditioned offices, J. Environ. Psychol. 42 (2015) 94–115, https://doi.org/10.1016/j. jenvp.2015.01.007.

- [275] F. Haldi, D. Cali, R.K. Andersen, M. Wesseling, D. Mueller, Modelling diversity in building occupant behaviour: a novel statistical approach, J. Build. Perform. Simul. 10 (2017) 527–544, https://doi.org/10.1080/19401493.2016.1269245.
- [276] C. Eon, G.M. Morrison, J. Byrne, The influence of design and everyday practices on individual heating and cooling behaviour in residential homes, Energy Effic. 11 (2018) 273-293, https://doi.org/10.1007/s12053-017-9563-y.
- [277] G.Y. Yun, K. Steemers, Night-time naturally ventilated offices: statistical simulations of window-use patterns from field monitoring, Sol. Energy 84 (2010) 1216–1231, https://doi.org/10.1016/j.solener.2010.03.029.
- [278] M. Schweiker, S. Brasche, W. Bischof, M. Hawighorst, A. Wagner, Explaining the individual processes leading to adaptive comfort: exploring physiological, behavioural and psychological reactions to thermal stimuli, J. Build. Phys. 36 (2013) 438–463. http://ien.sagepub.com/content/36/4/438.short.
- [279] A. Mahdavi, M. Taheri, An ontology for building monitoring, J. Build. Perform. Simul. 10 (2017) 499–508, https://doi.org/10.1080/19401493.2016.1243730.
- [280] B. Flyvbjerg, Five misunderstandings about case-study research, Qual. Inq. 12 (2006) 219–245, https://doi.org/10.1177/1077800405284363.
- [281] S. Kanaya, Y. Matsushima, K. Yokosawa, Does seeing ice really feel cold? Visualthermal interaction under an illusory body-ownership, PloS One 7 (2012), e47293, https://doi.org/10.1371/journal.pone.0047293.
- [282] B.E. Stein, T.R. Stanford, Multisensory integration: current issues from the perspective of the single neuron, Nat. Rev. Neurosci. 9 (2008) 255–266, https:// doi.org/10.1038/nrn2331.
- [283] D. Talsma, D. Senkowski, S. Soto-Faraco, M.G. Woldorff, The multifaceted interplay between attention and multisensory integration, Trends Cognit. Sci. 14 (2010) 400–410, https://doi.org/10.1016/j.tics.2010.06.008.
- [284] I. Ajzen, The theory of planned behavior, Organ. Behav. Hum. Decis. Process. 50 (1991) 179–211.
- [285] R. Zierler, W. Wehrmeyer, R. Murphy, The energy efficiency behaviour of individuals in large organisations: a case study of a major UK infrastructure operator, Energy Pol. 104 (2017) 38–49, https://doi.org/10.1016/J. ENPOL.2017.01.033.
- [286] Z.H. Ding, Y.Q. Li, C. Zhao, Y. Liu, R. Li, Factors affecting heating energy-saving behavior of residents in hot summer and cold winter regions, Nat. Hazards 95 (2019) 193–206, https://doi.org/10.1007/s11069-018-3489-3.
- [287] W. O'Brien, M. Schweiker, J.K. Day, Get the picture? Lessons learned from a smartphone-based post-occupancy evaluation, Energy Res. Soc. Sci. 56 (2019) 101224, https://doi.org/10.1016/j.erss.2019.101224.
 [288] H. Akaike, in: B.N. Petrov (Ed.), Proceedings of the Second International
- [288] H. Akaike, in: B.N. Petrov (Ed.), Proceedings of the Second International Symposium on Information Theory, 1973, pp. 267–281.
- [289] S. D'Oca, C.-F. Chen, T. Hong, Z. Belafi, Synthesizing building physics with social psychology: an interdisciplinary framework for context and occupant behavior in office buildings, Energy Res. Soc. Sci. 34 (2017) 240–251, https://doi.org/ 10.1016/j.erss.2017.08.002.
- [290] M.B. Kane, Modeling human-in-the-loop behavior and interactions with HVAC systems, in: 2018 Annual American Control Conference, ACC), 2018, pp. 4628–4633, https://doi.org/10.23919/ACC.2018.8431913.


Renewable and Sustainable Energy Reviews 149 (2021) 111359

Contents lists available at ScienceDirect



Renewable and Sustainable Energy Reviews

journal homepage: www.elsevier.com/locate/rser



Test rooms to study human comfort in buildings: A review of controlled experiments and facilities

A.L. Pisello^{a,b,*}, I. Pigliautile^{a,b}, M. Andargie^c, C. Berger^d, P.M. Bluyssen^e, S. Carlucci^{f,g},

G. Chinazzo^h, Z. Deme Belafiⁱ, B. Dong^j, M. Favero⁸, A. Ghahramani^k, G. Havenith¹,

A. Heydarian^m, D. Kastnerⁿ, M. Kong^j, D. Licina^o, Y. Liu^j, A. Luna-Navarro^p, A. Mahdavi^d,

A. Nocente^q, M. Schweiker^r, M. Touchie^c, M. Vellei^s, F. Vittori^b, A. Wagner^t, A. Wang^m,

^a Department of Engineering, University of Perugia, Perugia, Italy

^c Department of Civil and Mineral Engineering, University of Toronto, Toronto, ON, Canada

^d Department of Building Physics and Building Ecology, TU Wien, Vienna, Austria

^e Faculty of Architecture and the Built Environment, Delft University of Technology, Delft, the Netherlands

^f Energy, Environment and Water Research Center, The Cyprus Institute, Nicosia, Cyprus

⁸ Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway

^h Department of Civil and Environmental Engineering, Northwestern University, Evanston, USA

¹ Department of Building Services and Process Engineering, Faculty of Mechanical Engineering, Budapest University of Technology and Economics, Hungary

^j Department of Mechanical and Aerospace Engineering, Syracuse University, Syracuse, NY, USA

^k Department of Building, National University of Singapore and Center for the Built Environment, UC Berkeley, USA

¹ Environmental Ergonomics Research Centre, Loughborough University, School of Design and Creative Arts, Loughborough, UK

^m Department of Engineering Systems and Environment, Link Lab, University of Virginia, USA

ⁿ Institute of Energy Efficiency and Sustainable Building E3D, RWTH Aachen University, Aachen, Germany

° Human-Oriented Built Environment Lab, School of Architecture, Civil and Environmental Engineering, École Polytechnique Fédérale de Lausanne, Lausanne,

Switzerland

^p Department of Engineering, University of Cambridge, Cambridge, UK

^q SINTEF AS, Trondheim, Norway

r Institute for Occupational, Social and Environmental Medicine, Medical Faculty, RWTH Aachen University, Aachen, Germany

^s LaSIE (UMR CNRS 7356), La Rochelle University, La Rochelle, France

t Building Science Group, Karlsruhe Institute of Technology, Karlsruhe, Germany

^u The Bartlett School of Construction and Project Management, University College London, London, UK

ARTICLEINFO

Keywords: Test room Laboratory Climate chamber Human comfort Human-centric experiments Thermal comfort Visual comfort Acoustic comfort Indoor ari quality Multi-domain comfort Energy performance Indoor environmental quality

ABSTRACT

Occupants' comfort perception affects building energy consumptions. To improve the understanding of human comfort, which is crucial to reduce energy demand, laboratory experiments with humans in controlled environments (test rooms) are fundamental, but their potential also depends on the characteristic of each research facility. Nowadays, there is no common understanding for definitions, concepts, and procedures related to human comfort studies performed in test rooms. Identifying common features would allow standardising test procedures, reproducing the same experiments in different contexts, and sharing knowledge and test possibilities. This review identifies 187 existing test rooms worldwide: 396 papers were systematically selected, thoroughly reviewed, and analysed in terms of performed experiments and related test room details. The review highlights a rising interest in the topic during the last years, since 46% of related papers has been published between 2016 and 2020. A growing interest in non-thermal sensory domains (such as visual and air quality) and multi-domain studies about occupant's whole comfort emerged from the results. These research trends have entailed a change in the way test rooms are designed, equipped and controlled, progressively becoming more realistic inhabitable environments. Nevertheless, some lacks in comfort investigation are highlighted: some continents (like Africa and South America) and climate zones are found to be underrepresented, while involved subjects are mainly

* Corresponding author. Department of Engineering, University of Perugia, Perugia, 06125, Italy. *E-mail address: anna.pisello@unipg.it* (A.L. Pisello).

https://doi.org/10.1016/j.rser.2021.111359

Received 21 December 2020; Received in revised form 27 May 2021; Accepted 10 June 2021

Available online 30 June 2021

1364-0321/© 2021 Elsevier Ltd. All rights reserved.

S. Wei^u

^b CIRIAF – Interuniversity Research Center, University of Perugia, Perugia, Italy

students performing office tasks. This review aspires to guide scientists and professionals toward the improved design or the audit of test room experimental facilities, especially in countries and climate zones where human comfort indoors is under-studied.

. .	~			
1101	^ +	o b	https://philogeneticst	• •
LIST	UI.	av	DIEVIATIO	15

IEQ WWR	Indoor Environmental Quality Window-to-wall ratio (expressed in %)
HVAC	Heating, Ventilating and Air Conditioning
ACH	Air Change per Hour
VOCs	Volatile Organic Compounds
SPL	Sound Pressure Level (expressed in dB)
SBS	Sick Building Syndrome
EEG	Electroencephalogram
EDA	Electrodermal activity
ECG	Electrocardiogram
EOG	Electrooculography

1. Introduction

People in developed countries spend 85–90% of their time indoors [1]. Notwithstanding undeniable improvements in the quality of building interiors in the past decades, a range of health risks and discomfort issues associated with exposure to the indoor environment persists. Researchers have demonstrated the strong connection between the indoor environmental quality (IEQ) of a building and occupants' comfort, health, and productivity [2,3]. Moreover, buildings' energy consumption is largely affected by occupants' behaviour [4], triggered by their perception of the surrounding environment [5]. Therefore, decoding human comfort is a crucial issue in building science for enhancing building design and operation from a sustainable perspective and through a human-centric approach [6].

The scientific community approaches human indoor comfort by coupling measurements of the physical environment (e.g., air temperature, sound pressure level, air pollutant concentrations, illuminance) and occupants' feedback collected via surveys, behavioural and/or physiological monitoring. Applied experimental protocols can be broadly categorized into (i) in-field monitoring and (ii) laboratory experiments.

In-field experiments allow researchers to observe subjects in a real environment such as workplaces [7,8], residential [9] or educational [10] buildings, or even semi-open transitional urban spaces [11]. This approach provides essential outcomes, especially for assessing the impact of real-space configurations on occupants' perception [12], the effects of building characteristics on occupants' wellbeing [13], or the impact of occupants' behaviour on buildings' energy consumption [14, 15]. However, it does not allow to directly control the environmental parameters of the investigated spaces. Indeed, it is not feasible to isolate the contribution of a single environmental factor or a specific combination of multiple environmental stimuli on subjective responses, for example, overall comfort perception or productivity [16] in in-field research, while this is fundamental to establish a cause-effect relationship related to the comprehension of human comfort and the related occupancy behaviour [17]. These issues can be solved through experiments in controlled environments where desired physical boundaries can be determined and replicated, so different subjects can be exposed to the same stimuli and the influence of subjective factors elucidated [18]. Moreover, laboratory experiments generally allow researchers to perform a more detailed investigation of human subjects and collect physiological signals less commonly monitored in-field.

Many research institutions have built their own environmentally controlled experimental facilities to perform human comfort-related experiments worldwide and throughout the years. Each facility is designed to achieve specific research goals, thus presenting different dimensions, internal layouts, envelope characteristics, energy systems, and monitoring setup. Different equipment types are also included depending on the final aim of an experimental campaign targeting a specific comfort domain. Examples include thermal manikins, commonly simulating human thermal comfort [19] or inhalation exposure [20], or different apparatus for studying the human reaction to specific environmental input such as glare discomfort [21,22]. The test room design influences the experimental design and the accuracy of related modelling. The construction and technological details of the test room decide on the extent and scope of the different stimuli that can be provided as well as the different spatial layouts that can be generated. Being an essential determinant of experimental methodology, a careful design process of these facilities is of primary importance.

Due to the rising interest in better understanding human comfort, many reviews shed light on different perspectives of the topic. Several reviews summarise visual-related studies, reporting both lab and field investigations, as well as simulation studies [24–27]. Others focus on thermal comfort and different modelling approaches [28], main experimental procedures [29,30], or its energy-related implications [31]. Nevertheless, none addresses the diversity of laboratory facilities, which is a key component in the design of human-centred comfort experiments.

The identification of standard tools for advancing knowledge in the field would be helpful for the scientific community. An accepted glossary for identifying such facilities is still missing. Many papers refer to these facilities as test rooms or chambers or test-cells or simply laboratories. Here, "test room" was chosen as the most representative definition, highlighting the differences between facilities designed for human comfort studies and laboratory equipment devoted to material testing. Moreover, we define a "test room" as an enclosed space, environmentally controlled and properly instrumented, in which human-centric comfort studies can be performed through actual occupants' presence and monitoring.

This review aims at describing existing test rooms worldwide and at summarizing experimental studies on human comfort performed in such facilities to outline trends in the field, common components, and define new research perspectives. Precise selection criteria of the papers have been identified and used for the critical review (Section 2), and common technical features and trends in construction have been taken into account (Section 3), while Section 4 focuses on the specific experiments conducted in these facilities to deepen human comfort theory. Each experiment was categorized based on the type of domain(s) of human perception involved (thermal, visual, olfactory, and aural). In this context, a distinction was made between single-domain studies, which describe experiments focusing on thermal, visual, indoor air quality or acoustical stimuli only, and multi-domain studies [18,32], which simultaneously address two or more domains; for instance, the analysis of thermal and acoustic stimuli on overall comfort perception, or the analysis of thermal perception as influenced by lighting or air quality conditions. The key findings and conclusions, including suggestions for future research agenda, are summarised and critically discussed in Sections 5 and 6, respectively.

2. Materials and methods

A systematic bibliographic search was planned and conducted to establish a database as comprehensive as possible, looking at existing test rooms for human comfort experiments according to available scientific literature and not to miss any test rooms that the authors are aware of. The final database is thus the result of two main steps: an automatic search and a supplementary hand search (Fig. 1).

The automatic search was systematically conducted through Scopus and Web of Science scientific databases to identify papers concerning human comfort investigation in test rooms, as available up to June 2020. The search was limited to journal papers written in English after 1985 to keep the search consistent between the two scientific databases due to the temporal limitation of Web of Science. To cover the scientific literature on the theme published before 1985, a further search was conducted in Google Scholar. Different typologies of documents such as books, book chapters, reviews, or conference proceedings were thus excluded from the search to improve consistency and avoid repetitions of the same study that may have been presented in different document types. Five queries were designed within these boundaries, corresponding to each aspect of indoor human comfort. The queries were structured in three parts, progressively focusing on the purpose of the review:

- (i) on the laboratory facility where human comfort experiments took place,
- (ii) on the main aim of the studies, i.e., human comfort, and
- (iii) on the specific comfort domain of interest (e.g., thermal, visual, acoustic, air-quality related).

Each part of the query was detailed after a discussion among the authors that are experts in human comfort studies and come from different countries and cultural backgrounds. These cultural differences provide a comprehensive definition of the facilities object of the review. The first two parts of the query were used for all the five queries and consisted of the following keywords: (testroom OR test-room OR chamber OR laborator* OR "test cell") AND comfort. The term "human" was not included for not missing any contributions that may fit the scope but did not explicitly mention humans' involvement. The publications not dealing with human comfort were excluded through the double-screening procedure, as specified in the following. In addition to these keywords, the five queries were distinguished by including the following specific keywords:

1. Thermal

- 2. Visual OR Lighting
- 3. Acoustic
- 4. Air quality OR Pollution
- 5. Energy



Fig. 1. Papers selection workflow.

Each specific query focused on a single comfort aspect addressed from the perspective of the provided physical stimulus, as associated with thermal, visual, aural and olfactory human perception. In contrast, the fifth query focused on the theme of energy that is commonly associated with human comfort studies aimed at improving indoor environmental quality while reducing building energy consumption.

The automatic bibliographic search resulted in 1776 papers. A cleaning procedure of the database was performed by focusing only on experiments both carried out in a controlled environment and addressing human perception and exposure. This procedure accounts for two main steps. The first screening was conducted through a specifically developed script in Python language for automatic abstract screening by excluding papers presenting specific words referred to out-of-scope disciplines such as medicine or veterinary medicine. After this first screening, 598 papers were still included in the review process. and went through the second screening phase: the papers were carefully read and selected according to the primary purpose of the review. Only papers describing experiments performed in the controlled environments (test rooms) whose internal dimensions and conditions were suitable for human experiments were considered for this review.

The hand search was carried out for reducing the automatic search biases and limiting the number of existing test rooms not covered by this review. Additional papers were included according to the previous knowledge of the authors and the selection criteria that is the usage of a controlled environment for conducting experimental research on human perception and exposure. More than half of the additional papers (49 out of the 92) concern the visual comfort domain, meaning that common keywords coming from the other domains were not suitable to catch all the visual comfort studies. The final number of analysed papers was 396.

Table 1 summarises the number of analysed papers per topic and year of publication, considering four time periods: (i) up to 2000, (ii) 2001–2010, (iii) 2011–2015, and (iv) 2016–2020. Defined time periods highlight the considerable increase in published papers on controlled test room experiments on human comfort. Indeed, the increase ratio observed during the first decade of the 21st century (1.9) is comparable to the one observed for the first (1.5) and second (1.7) part of the following decade.

The table depicts a predominant interest of the scientific community in thermal comfort investigations (conducted either in isolation or in combination with other factors) followed by energy-related studies (total of 85 papers) and visual comfort assessments. Air quality studies are less common, especially as a single stimulus for the participants involved in test room experiments. Indeed, the total amount of reviewed papers related to air quality assessment is 84. Only 18 of them were found to focus on air quality only as a single stimulus, disabling the olfactory from the thermal perception and all the other spheres of comfort. More detailed presentation of the aims and procedures of the air-quality-only studies is provided in <u>Subsection 4.4</u>.

Fig. 2 shows trends of publication for each specific domain of comfort, without distinguishing between single and multi-domains experiments, with respect to studies published before 2000. Thermal comfortrelated experiments present the slowest increasing ratio from the reference scenario. Air quality-related experiments show the greatest increase in the number of published papers, with a slight decrease in the last five years. A similar trend can be observed for energy-related studies. Visual comfort-related studies are gaining more attention with currently seven times more papers compared to available publications before 2000. Aural comfort is the least investigated domain in controlled environments. Reviewed papers including a focus on acoustic comfort are 32 in total, half of which published in the last five years.

3. The test rooms around the world

From the 396 papers selected according to the systematic review process, 187 different test rooms located in 126 research institutes around the world have been identified based on the descriptions

Table 1

Number o	f journal	s papers	published	through	out years	(up to	June	2020)	and	concerning	each	analysed	l topic.
----------	-----------	----------	-----------	---------	-----------	--------	------	-------	-----	------------	------	----------	----------

Domain(s)		Time periods						
		≤2000	2001-10	2011-15	2016–20 ^b	Total		
1 domain	Thermal	26	39	50	89	204		
	Air quality	3	3	4	8	18		
	Acoustic	0	2	2	7	11		
	Visual	5	10	23	32	70		
2 domains	Thermal + Air quality	0	10	22	19	51		
	Thermal + Acoustic	0	3	0	3	6		
	Thermal + Visual	1	0	1	17	19		
	Air quality + Acoustic	0	1	0	1	2		
	Air quality + Visual	0	0	0	0	0		
	Acoustic + Visual	0	0	1	1	2		
3 domains	Thermal + Air quality + Acoustic	1	0	1	1	3		
	Thermal + Air quality + Visual	0	1	1	0	2		
	Thermal + Acoustic + Visual	1	0	0	1	2		
	Air quality + Acoustic + Visual	0	0	0	0	0		
4 domains	Thermal + Air quality + Acoustic + Visual	0	0	1	5	6		
Total		37	69	106	181	396		
	energy related ^a	5	15	29	36	85		

^a The energy-related topic is transversal to the others.

^b The count for 2020 considers only those documents indexed until June 2020.



Fig. 2. Publication increase ratio with respect to the number of published papers before 2000 for each query.

provided in the papers.

Fig. 3 summarises the test rooms distribution across continents (a,b) and different climate conditions (c), referring to the Köppen-Geiger climate classification [33]. Nowadays, the great majority of test rooms are located in Europe and Asia (82%), and in a temperate climate, without dry seasons, characterized by hot (Cfa) and warm (Cfb) summer. 29 out of the 44 test rooms located in the Cfa climate zone are in Asia (South and coastal area of Japan and South-Eastern China mainly), while 54 out of the 57 test rooms located in Cfb zones are in Europe (North-Western countries mostly). Fig. 3b presents how the worldwide distribution of these facilities varied across time (all the test rooms were dated per the oldest related paper available in the review dataset). European countries have the oldest tradition in human-related experiments conducted in controlled test room settings: 50% of the facilities already existing before 2000 were located in Europe. The number of facilities in Asia has grown over the last 20 years from 18 to 41% of the total number worldwide in 2020, overcoming the number of facilities located in North America (13%).

The following subsections are intended to provide helpful information for researchers evaluating whether to create or buy a test room for human comfort studies. These illustrate the range of test room characteristics that enable the researcher to perform different experiments and investigate specific aspects of human comfort. An overview of construction and technical details is provided in section 3.1 and 3.2, in accordance with the available information from the reviewed papers. Then, sections 3.3 and 3.4 provide insights into the economic investment required to set up these kinds of facilities, either if these are customized or commercially available. Since none of the reviewed papers provides information on test room costs and related economic investment, data provided in sections 3.3 and 3.4 come from an additional search: an online survey was submitted to authors of the identified significant and recent literature, seeking details on key aspects of the needed economic investment (including design, construction, operation and maintenance costs). Finally, commercial test room producers (eight institutions from the U.S. and five institutions from Europe) were directly contacted to provide dedicated insights for the readers, reported in section 3.4.

3.1. Construction details

The construction details were specifically examined to determine how passive elements of the test room, including windows, shades, layout, size, and position within or external to an existing building, may allow or hinder different types of investigations. Unfortunately, comprehensive descriptions of the test rooms construction details are not always available. It was not possible to assess whether the test rooms are located inside a building or are entirely independent buildings for 10% of the 187 test rooms identified. According to the available information, only 7% of the facilities are independent buildings, external to any other building [34–47]. Five of these independent test rooms are located on a platform that allows the whole structure to rotate [34–37, 41]. The great majority are situated inside the related research institute. Among these, it is possible to distinguish between facilities completely detached from the surrounding structure (43%) and test rooms that are specifically equipped rooms within the hosting building (32%).

Some test rooms include more than one room. These rooms could be adjacent, but with independent entrances, or connected through an intermediate door. The latter configuration allows researchers to continuously monitor participants' reactions when exposed to different controlled environmental conditions [48]. Eight of the external facilities have just one room, but the possibility to work with movable internal partitions is mentioned for four of them [38,41,43,44]. The other six outdoor test rooms present two rooms, and four out of the six have movable partitions for changing the interior space layout [34,35,40,45]. For the inside test rooms, single room configurations are most common (79%), some of which can be modified through movable interior partitions (19%). More information about the number of rooms embedded in the test rooms and their dimensions are summarised in Table 2.

Renewable and Sustainable Energy Reviews 149 (2021) 111359



Fig. 3. (a) Number of test room facilities located in the seven continents; (b) amount of test rooms located in each continent for each defined time period; (c) frequency distribution of test rooms with respect to Köppen-Geiger climate classes [33].

Table 2

Test rooms composition and dimensions with respect to their position (inside or outside another building).

Test rooms position Number o		f rooms	ooms			Dimensions [m ³]				
		1	2	>2	N/A	<9	9–20	>20	N/A	
Inside	Detached ^a	53	8	1	4	5	16	36	9	66
	Integrated ^a	37	8	2	2	0	4	43	2	49
	N/A	32	2	0	6	1	5	20	14	40
Outside		8	6	0	0	0	4	10	0	14
N/A		7	1	0	10	0	1	4	13	18
Total		137	25	3	22	6	30	113	38	187

^a With respect to the building structure of the related research centre.

It was not possible to define whether the described test rooms present any type of openings for 41% of the recognized facilities, 25% of the test rooms located inside have no openings, 18% have windows facing the outside, 16% have windows to interior spaces, and just 2% have both windows to the outdoors and the indoors (Fig. 4). Among the 14 experimental facilities built outside, only one does not have windows [46]. At the same time, five include an adjustable envelope to vary the window-to-wall ratio (WWR) [35,37,41,43,45], five have a WWR lower than 0.5 [38,39,42,44,47], and three have a WWR in between 0.6 and 0.8 [34,36,40]. Concerning the shading system, it is clearly stated that there are external blinds in three test rooms [34,36,47], four present internal shading systems [38,40,42,44], while just one has both [39].

Half of the test rooms have no specific internal layout, meaning that there is no intention to simulate a real space but only to expose subjects to controlled environmental stimuli. Equipment for performing physical exercises are included in 10% of these test rooms [49–64]. All the others



Fig. 4. Overview of the most common combination of characteristics for inside test rooms, in terms of its position with respect to the main structure and the windows availability.

have no specific furniture, even if 49% are larger than 20 m³. Finally, 12% of the analysed test rooms are presented in different papers with different internal layouts, 32% are equipped as offices, 3% as classrooms [65–69], and less than 1% present other configurations [70–73].

The above presented physical characteristics of the reviewed test rooms can be associated with their capability of performing different types of experiments, focusing on different domains of human comfort. The external test rooms are more commonly devoted to visual-related experiments. Indeed, six out of the 14 exterior test rooms are associated with visual-only experiments, while only one was used for testing human comfort conditions due only to thermal boundaries. When more than one domain is explored, four test rooms hosted experiments providing combinations of thermal and visual stimuli; the air quality influence was additionally explored in one test room while all the four domains of comfort were explored in only two of the 14 external test rooms.

With respect to performed experiments, it is more complicated to deduce the most common combination of construction details for the test rooms located inside other facilities due to a lack of information on all the analysed features. Only 82 out of 155 reviewed test rooms are described in terms of both (i) their position in the hosting facility (de-tached or integrated) and (ii) windows availability facing the inside or the outside. Accounting for these two aspects, detached test rooms generally have no window (56%) and are more commonly adopted for investigating human comfort under thermal stimuli only (46%). Those test rooms that are integrated into the main structure, as specially

equipped rooms, commonly have windows facing the outside (68%) and are mainly used for experiments on visual domain only (54%).

3.2. Technical details

Similar to the presentation of construction details, the technical capabilities of the test rooms directly inform what types of experiments can be conducted. Specifically, this subsection outlines which parameters are controllable and to what degree. As a first step, an analysis of the most common parameters that could be controlled by the test room systems was conducted. For this purpose, the relevant information was extracted from the corresponding papers for each test room and categorized as presented in Table 3.

This categorization is more granular than the multi-physics domains introduced in Section 2 (thermal, visual, air quality and acoustic comfort) to better characterize the specific system types used to influence each domain parameter. Indeed, in some cases, multiple controlled parameters will impact a single domain such as air temperature, mean radiant temperature and incoming solar radiation, all impacting thermal comfort. Additionally, the controlled parameters were subdivided into centralized and personalized systems (generally located at a desktop or at a participant/manikin). In the process of this categorization, 91 test rooms were selected for further analysis because related publications provided relevant and sufficient information. Fig. 5 summarises the number of test rooms which can control each of the listed parameters. In some cases, one test room is counted multiple times in this plot, once for each parameter its system controls.

The most common centrally controlled parameter is air temperature, followed by humidity and air quality control. All these three parameters can potentially be controlled by HVAC (Heating, Ventilation and Air Conditioning) systems with a humidifier and/or dehumidifier equipment, heating and/or cooling coils, and air filtering. The common practice of controlling thermal conditions in actual buildings, together with the predominant focus on thermal comfort studies (highlighted in Section 2), is likely why these controlled parameters are found to be so common. Fig. 6 summarises the ranges for each of these three controlled parameters for all of the test rooms where ranges were reported. As shown, nearly all the test rooms can control air temperature between 15 and 30 °C, and relative humidity between 30 and 70%, but air-speed control was more variable. Almost all test rooms were able to control these parameters at least in the ranges covered by indoor comfort standards such as ISO 7730 [74] and, in many cases, well beyond this range, particularly with respect to the seven low-temperature chambers.

Only a few papers included details of the other parameter ranges. Control of air change rates in the test rooms, which is accomplished through multi- or variable speed fans, ranged from 0 to 36 air changes per hour (ACH) but generally allowed for control within the minimums required by the EN 12931 (0.5–3.6 ACH for residential buildings) [75]

Table 3

Categorization of technological systems and related controlled parameters.

Technological control system for comfort	Controlled parameters
Ventilation and space	Air temperature
conditioning	Air velocity
	Air humidity
	Air quality (gas concentration, air changes per hour)
Heating/cooling surfaces	Envelope superficial temperature
	Radiator or other element temperature (e.g., clothes,
	furniture)
Light sources	Illuminance
	Solar radiation (artificial, e.g., solar simulator)
	Solar radiation (natural, e.g., actively controlled
	blinds and shades, electrochromic glass)
Acoustic systems	Background noise level (sound intensity, sound pressure level)
	Sound typology (soundscape)



*lighter colour bars indicate uncertainty: the controlled parameter was not explicitly indicated but deduced by the experiment description in the context of the publication



and by EN 16798 part 3 for offices (1–8 ACH) [76]. Only five test rooms reported the temperature range at which their radiant wall systems (either electric or hydronic panels) could be controlled (generally between 10 and 40 °C). For rooms with reported artificial lighting, the range 100–800 lx covered and exceeded the requirements (e.g., EN 12464) [77]. A few publications also reported the ability to vary the correlated colour temperature of the artificial light (2000 K to 10, 000 K). There was insufficient information about artificial solar radiation and acoustic systems to report ranges here.

Only 11 of the reviewed test rooms included parameters that could be controlled at a personal level. Furthermore, most of these personalized systems were only temporary for specific experiments and not a fixed part of the test room. Typical setups would be ventilation tubes aimed at a desktop, heated/cooled clothing and chairs, electrical heated mats or computer equipment (mouse, keyboard), and electrical radiators.

The parameters controlled by the test rooms were also examined based on the estimated date of construction of the test room to identify trends or most prevalent innovative technologies, as shown in Fig. 7. It is also unknown if or when test rooms have been upgraded, nor do we have insight about the upgrades made. Thus, the results in Fig. 7 represent the latest built stage of the test rooms according to the publications and may differ from their technologies at the given date of construction. The graph suggests a trend towards incorporating the control of acoustic sources, artificial and natural solar radiation, illumination, and radiant heat sources, including radiators and radiant wall panels.

Furthermore, the analysis revealed that personalized control systems are becoming popular in newer test rooms constructed after 2000. Finally, in the latest test rooms built between 2011 and 2020, there also seems to be a trend for controlled multi-domain installations with six test rooms since 2013, controlling at least three domains.

3.3. Economic investment

The economics of test rooms is rarely reported. Therefore, a survey to assess key elements related to this topic was designed. All co-authors of this manuscript and authors of identified literature were invited to complete it. In total, 18 responses related to separate test rooms were



Fig. 6. Ranges of controlled air temperature, relative humidity and air velocity in the reviewed test rooms.



Fig. 7. Time distribution of implemented technologies for controlling specific parameters (see the legend) identifying trends in test room construction.

obtained, of which 14 have been completed, and four are still under construction. Except for one completed in 1990, all others have been built within the last ten years. The majority of the test rooms is either a test room constructed within an existing building (N = 8) or a building itself (6). Three test rooms are newly built test rooms within a new building, and one test room is an existing room refurbished and upgraded to serve as a test room. The vast majority is located in Europe (13), followed by Asia (3) and North America (2).

Local currencies have been converted to EURO based on currency rates from September 4th 2020. The total budget ranges from EUR 45,500 to EUR 943,000 (mean = EUR 347,000 \pm 299,000, median = EUR 240,000). For eight test rooms, information was provided in more detail. On average, shell construction costs (especially for those test rooms built as stand-alone test rooms within new buildings) are highest (mean = EUR 175,000), followed by costs for design, contracting, and commission (EUR 91,000), heating and cooling system (EUR 31,000 and 34,000), and in-built sensors and the Building Management System (EUR 33,000 and 19,000). This large variety can be explained partly by the variety in the type of construction, controlled and monitored variables and the ranges within which these variables can be controlled. In addition, it can be expected that prices vary locally and between countries. Seven out of 18 test rooms were fully funded by governmental sources, either from basic funding (N = 3) or project

funding (4). In addition to public and project funding, five test rooms were partially funded by the industry (min 5%, mean 24%, max 70%).

In addition to initial construction and installation costs, running costs (e.g., electricity, gas, water) and/or maintenance costs were assessed. Running costs were reported solely for three test rooms, but differed largely (EUR 2500 to 17,500 per year). Interestingly, the source of funding for running costs was provided for 14 test rooms, of which nine responded that the university pays for running costs, three state project funding, and the other shared funding either between the university and the lab (10/90%) or the university and project funding (20/ 80%). The large discrepancy in response numbers between actual costs and funding source may signify that researchers are not aware of the running costs. Maintenance costs were provided for eight test rooms and range between EUR 930 to EUR 10,000 per year (EUR 5100 \pm 3500). Funding sources for maintenance costs vary more than running costs for 12 out of 14 facilities, for which such information was provided. In three cases each, maintenance is paid fully from the laboratories' basic funding or project funding. In two cases, the university covers all maintenance costs. In the other cases, maintenance costs were shared between the university, basic funding of laboratory and project funding with varying degrees. Only in one case, 25% of maintenance costs are provided by industrial partners.

3.4. Commercial test rooms

Commercial test rooms are available on the market to provide researchers who want to use an already existing and tested product with an off-the-shelf option. These test rooms tend to use a similar structure and envelope materials as prefabricated foam-insulation panels with stainless steel, galvanized or coated aluminium (usually white) interior surfaces for fast and easy installation. This is for protecting the test room surface from being damaged or corroded by moisture and chemicals. The stainless-steel chamber can also help minimize the adsorption of VOCs by the surfaces, which is critical to some indoor air quality studies. However, for human-centred thermal studies, the reflective properties of the interior surfaces also determine the radiative heat exchange in the space, thus additional materials or painting are needed to simulate a 'real-life' condition. The test room usually has at least one hinged door made of the same material and optional windows of different sizes. Important differences between offerings tend to be in the type of airflow achieved in the test room. Cheaper and smaller systems tend to have the heat exchangers inside the room and achieve spatial stability by producing turbulent flows. More laminar flows are achieved with wallto-wall or floor-to-ceiling air flows across the whole wall/floor, which requires a plenum space inside the test room, thereby increasing the external size. Most of the rooms come with predesigned and prepackaged conditioning systems that can provide space heating and cooling, ventilation, humidification, and dehumidification to the room. Air temperature, relative humidity, and ventilation rate are under control and monitored. Some test rooms are even equipped with pressure, CO₂, and O₂ sensors.

The operating condition of commercial test rooms depends on their application that can be testing equipment, storing experimental materials, and also human-centric tests. Here, since we only focus on the test rooms for the human-centric test, the surveyed test rooms only include those capable of providing conditions indicated by the green box in Fig. 8.

These commercial test rooms can be as small as 1.5 m^2 and as large as up to 10 m² with a height in between 2.4 and 2.6 m. The price of the test rooms (N = 13 units personally contacted by the authors) ranges from EUR 54,600 to 210,000. The average quote from U.S. companies is around EUR 128,100 with a standard deviation of EUR 44,000, while the average quotation from Europe is around EUR 99,800 with a standard deviation of EUR 27,400. On average, the test rooms from the U.S. (8) are a little more expensive than in Europe (5). The explanation may include regional reasons such as shipping and labour, material and sensors cost, and size difference. One should note that the size and quotes obtained in this study are based on the smallest test room with the basic features of temperature, relative humidity, and ventilation control with at least one occupant. The quotes were obtained in August 2020, and for the commercial test rooms made in the U.S., the quote was converted to EUR based on the exchange rate on September 4th, 2020 [1 USD = 0.84 EUR].

4. Test room experiments on human-environmental comfort

This section focuses on the experiments conducted in the test room above presented in terms of their structure and main functionalities. Each subsection presents an overview of the main aims and procedures of test room experiments answering the question, what is the scientific



Fig. 8. Required operating conditions of the surveyed commercial test rooms.

community looking for through test room experiments? Scopes of the experiments are broadly clustered in the presented subsections with respect to (i) the comfort domain of interest (sections 4.1-4.5), (ii) the subjects' involvement (possibility to interact with the test room during an experiment, section 4.6), and (iii) the investigation of the energy related aspects (section 4.7), which are all relevant aspects for human comfort studies. Concerning the applied procedures, the main distinction is adopted between stationary and dynamic conditions.

4.1. Thermal-only experiments

This subsection reviews 204 papers on test room studies that explored the effects of thermal conditions on participants. The scope of the reviewed thermal experiments can be broadly classified into three categories: (i) fundamental research aiming at providing a better understanding of human thermal comfort; (ii) technology-oriented experiments, whose purpose is to test the thermal comfort performances of specific types of heating and/or cooling systems or newly developed clothing; (iii) predictive studies with the purpose of data collection to test and train novel predictive models. Fundamental studies are more common than technology-oriented and predictive studies, respectively 57%, 36% and 7%, and their distribution over the last four decades is shown in Fig. 9a. Fundamental studies include research focusing on a variety of different aspects influencing human thermal comfort such as thermal adaptation [78-82], thermal acclimatization [83-86], increased air velocity [87-90], relative humidity [60,91-94], gender [95-98], age [52,99-102], transient thermal conditions [93,103-107], perceived control [108], and the influence of emotional states [109, 110]. About 30% of the thermal experiments are dedicated to the study of non-uniform thermal conditions. Non-uniformities and thermal asymmetries are not seen only as a cause of discomfort; indeed, many recent studies aim to understand how comfort can be enhanced with local thermal stimuli [111-120].

The technology-oriented experiments mainly look at the thermal comfort performances of specific types of equipment, such as innovative heating and/or cooling systems (thermo-electric air cooling systems [38, 121], stratum, mixing and displacement ventilation [122–124],

underfloor air distribution systems [122,125], radiant cooling/heating panels, floors and ceilings [126-129], ceiling fans [130], etc.). In particular, the last 20 years have seen a progressive increase in the number of experiments dedicated to local heating and/or cooling systems (personal cooling with phase change materials [112,113], heated/cooled chairs [114-116], seats heated with encapsulated carbonized fabric [120], feet heaters [117,118], etc.). About 40% of the technology-oriented experiments aim to test new clothing (uniforms for heat strain or cold thermal stress attenuation in the construction industry [50,51,131,132], sports clothing [53,133-135], protective clothing systems [136,137], cooled/heated garments [138,139], etc.). The distribution of the technology-oriented experiments based on the type of system studied (heating or cooling, local heating and/or cooling, clothing) over the four different climate groups is shown in Fig. 8b. As expected, in tropical climates there is a prevalence of experiments studying cooling systems, while in continental climates, the focus is on new clothing systems.

The predictive studies provide experimental data to either develop, test and train novel data-driven predictive models. Many of them aim to predict either thermal comfort or thermal stress (e.g., heat strain indexes [140,141]). Instead, others are attempting to build models for predicting metabolic rate and clothing insulation levels [64,142].

A majority (46%) of the reviewed thermal experiments deal with both warm and cold thermal conditions, 39% of them only focus on warm conditions and the remaining 15% on cold conditions. They mainly consider sedentary activity levels (77%), only a few of them focus on high metabolic rate activities (21%) and a minority on sleeping (2%). Furthermore, most of them consider stationary thermal environments, while the experiments dealing with dynamic conditions mainly study step-change transients [93,103–107]. In the last 20 years, female and male participants have been equally represented in the thermal experiments; nevertheless, elderly and children continue to be underrepresented groups (in only 3% of the experiments). Concerning the sample size, a majority of the experiments (57%) employ between 10 and 50 participants, 31% of them recruit less than 10 participants, and only 12% more than 50 participants. In most of the experiments (about 70%), participants are passive recipients of thermal stimuli without any



Fig. 9. (a) Thermal studies aim distribution over time periods; (b) thermal technology-oriented studies, studied system over climate classification.

possibility of adaptation/control.

The ASHRAE 7-point thermal sensation scale is the most used metric of thermal perception, followed by thermal comfort, thermal acceptability, thermal preferences, and cognitive performances. Air temperature is the most frequently monitored environmental variable (in 90% of the experiments), followed by relative humidity (75%), air velocity (63%), globe temperature (36%), and wall surface temperatures (9%). Air turbulence intensity, luminance, and solar irradiation (artificially provided) are more rarely monitored. Oxygen and carbon dioxide measurements are mainly used to estimate the metabolic rate, less often as a proxy of air quality. Skin temperature is the most common personal measurement (60% of the experiments), followed by heart rate/heart rate variability (27%), rectal/body core temperature (18%), body weight for sweat rate determination (7%), skin wetness (6%), ear/oral temperature (5%), skin surface blood flow (4%), blood pressure (3%), and skin heat flux (2%). Some very recently emerging topics are the use of immersive virtual reality [143-145] and the monitoring of brain electrical activity patterns [109,146].

4.2. Acoustic-only experiments

This subsection looks at 11 test room studies exploring the effects of acoustic conditions on participants by investigating different human responses and developing or evaluating new metrics for soundscapes description (Fig. 10). The test room experiments' aims include investigating maximum heavy-weight impact sound levels for perceived comfort [147], effects of sound pressure levels (SPL) and sound types on children's task performance [148], factors that contribute to sound complexity [149], effects of speech noise and speech transmission index (STI) in offices on cognitive performance [150,151], suitable masking sound frequency distribution for offices [152], effects of low-frequency noise in offices [153], effects of various noise sources on occupants in multi-family buildings [154], useful acoustic parameters that effectively describe to perceived sensations of urban noise [157].

Many of the studies followed the general procedure of exposing participants to stimuli (recordings of sounds at various SPLs, frequencies, or decay rates) while performing cognitive tests and/or completed subjective assessments of the acoustic environments.

Test room setups and specific data collection procedures varied considerably among the studies. For instance, the provided stimuli length ranged from 10 s to 45 min, and the time that participants were given to respond to objective and subjective assessments ranged from 5 s to as long as the participants wanted to take. Most of the studies used loudspeakers to play the studied sounds, except Hermida and Pavón



Fig. 10. Distribution of acoustic studies' aim over time.

[156] and Hong et al. [157], who used headphones, and Jeon et al. [147], who used both loudspeakers and headphones. Only three studies [150–152] had test room setups that mimicked the type of real-world environment that they were investigating. Concerning the overall environmental control, three studies [150,152,153] mentioned that other indoor environmental conditions (such as temperature and lighting level) were kept constant in the test rooms. In contrast, others did not give any description of non-acoustic environmental conditions in the test rooms that could potentially affect the study outcomes.

For acoustic experiments involving human participants, it is common practice to screen participants' hearing abilities before conducting listening tests to avoid bias in the perception analysis. However, only four studies [147,149,153,157] screened their participants' hearing abilities using audiometers and other devices, and three studies [148, 152,155] used subjective assessments to determine hearing abilities. Other studies either did not do similar screening or did not specify how they determined participants' hearing abilities. In addition, only three studies included evaluation of the effects of demographics, for example, age [147,148,150] and gender, and personal factors, such as personality traits [150], on participants' responses. Finally, just one study [153] monitored the physiological responses of participants (including the electrical activity of the brain, eye activity, heart rate, and heart rate variability) to low-frequency sound exposure using electroencephalography (EEG), electrocardiogram (ECG), electromyography (EMG), and electrooculography (EOG) signals.

Regarding sample size, six out of the 11 reviewed studies involved between 10 and 50 participants, with a minimum of 23 [152], while all the others involved more than 50 participants up to a maximum of 290 [148].

4.3. Visual - lighting-only experiments

The following overview focuses on visual-related experiments aiming at studying subjective evaluations of the visual environment performed in controlled environments. Studies conducted with the use of a scale model (e.g. Refs. [158–160]), with a small apparatus (e.g. Refs. [161–163]), in a booth (e.g. Ref. [164]) or in virtual reality (e.g. Refs. [165,166]) were excluded from the analysis as they were not performed in real-scale controlled environments. Investigations on electric lighting evaluations primarily aiming at testing lamp brightness and colour rendition based on lamp characteristics (e.g. Refs. [167–170]) were also not included. The resulting sample analysed consisted of 70 papers.

As introduced in Section 2, visual-related studies in controlled experiments have increased over the last decade, with more than 77% of



Fig. 11. (a) Distribution of visual studies' aim over time; (b) investigated light source distribution over time periods.

the considered studies conducted between 2010 and 2020 (Fig. 11). The type of light source investigated has been relatively constant throughout the years, with an equal number of studies focusing on electric light and daylight (Fig. 11b). The majority of studies focused on glare (more than 50%), either to evaluate subjective perceptions due to variations of lighting conditions or other factors' influences (such as time of the day or openings and blinds features) [21,22,37,39,42,66,171-182], develop, evaluate or validate metrics, thresholds or indexes [35,68,183-193], investigate glare influence on performance and physiology [187, 194-196] or study a combination of such objectives (Fig. 11a). Other studies investigated visual perceptions of the visual environment, surface finishing preference, physiological responses, performance, sleepivitality, arousal, tension, mood, self-control ness, and cognitive-biological processes (light-reactive hormones of melatonin and cortisol) mainly related to the light quantity and correlated colour temperature (CCT), but also in relation to light uniformity, wall luminance, light source type, flicker rate, view and chromatic glazing [164, 197-216]. The majority of the studies did not allow for personal control of the environment, testing pre-defined conditions, and were conducted with 10-50 participants. Only in a few studies participants were requested or simply allowed to vary their visual environment through the operation of blinds and electric lights, either to evaluate glare conditions or to assess how occupants perceived their visual environments associated with diverse luminous ambiences created by daylight in apartment buildings [73,189,191].

Most of the investigations were conducted in re-configured office spaces located in existing buildings, transformed into experimental test rooms in which it was possible to control or at least measure visual parameters. The traditional configuration was a side-lit single office, generally bigger than 20 m³. Still, some investigations used a corner office [193], a mock-up of an open-plan office with multiple workplaces [209], a re-configured classroom [66], a full-scale mock-up conference room [208], or divided an existing office room with internal vertical partitions, resulting in smaller experimental spaces [217,218]. Some glare experiments used full-size apparatuses consisting of a semi hexagonal lighting chamber equipped with a chin rest [22,172,173,176] or of a semi-spherical screen with two halogen lamps mounted on a 1-m radius round boom [21,185]. Only fewer studies were conducted in a stand-alone test room, either located indoor [197,198,200-204, 210-213,219] or outdoor [35,42,179,188,192,194,220,221]. Some of the outdoor facilities were rotating structures [35,179,192,195,220], allowing daylight conditions to be tested with a reduced impact of the daylight variations due to the season and time of the day. Very few test spaces were designed to have a side-by-side configuration with two identical spaces, one for participants and the other for measurements [35,171,183,187,189,190,194,220]. This particular setting, aiming at decreasing interventions in lab experiments, is particularly suitable for visual-related investigations as photometric data are relatively affected by the presence of people, contrary to the other indoor factors that have to be measured close to participants. The presence of a window to the outdoors was linked to the type of experiment investigated. Almost all experimental spaces provided with a window investigated daylight, except for those studies that performed the experiments at night [222] or in which windows were shaded with a black-out fabric or blocked [164,180,199,207,217]. The studies investigating a mix of daylight and electric light were provided with shading devices [189,190,205,220, 223]. On the other hand, not all the studies on daylight were provided with a real window to the outdoor (intended as an opening with a view), but used artificial windows [37,177,181,192,204] or anidolic systems on the southern façade [224]. Non-visual factors were measured, controlled, or balanced across experimental conditions in almost all stand-alone test room experiments, and only in fewer re-configured offices [199,205,217,218,223,225]. The factors considered were primarily air temperature and humidity, but also noise [217,218] and air quality [37,197,198].

4.4. Air quality-only experiments

This subsection describes the controlled air quality-only experiments in test rooms summarised in 18 papers according to the reviewed database. Additional four papers that fall under two-domain experiments are included in the analysis since thermal and air quality aspects are hard to disentangle as the thermal analysis is ancillary to the air quality assessment [72,226-228]. Among the representative selections of 22 air quality studies in test rooms, researchers have focused on the three main topics: (i) understanding perceived air quality, productivity and health under a range of environmental parameters [71,72, 229-235]; (ii) human inhalation exposure and spatio-temporal variation of air pollution in a space [20,228,236-241]; and (iii) airflow distribution in occupied spaces and ventilation effectiveness [226,227, 242-244] (Fig. 12). These topics were pursued through a combination of questionnaire surveys, environmental measurements (near a study participant, in bulk air or ventilation ducts), and physiological measures. Discrepancies in facilities among the selected studies include test room layouts (office space, classroom, aircraft cabin, hospital room), test room volumes (small below 10 m³, medium 10-50 m³, or larger than 50 m³), surface materials (stainless steel, polytetrafluoroethylene, aluminium, glass or their combination), type of air pollutant generation (continuous or episodic), ventilation type (mechanical or mix-mode ventilation), ventilation strategy (mixing, displacement, underfloor or personalized ventilation), degree of air mixing (ventilation only or additional use of mechanical fans), operating procedure (dynamic or stationary conditions), and participant type (real occupancy or use of breathing thermal manikins).

In the reviewed air quality papers, all test rooms were located inside of the building and had control over the ventilation rate, air temperature and relative humidity. While nearly all studies reported air temperature and relative humidity values and associated uncertainties, only 12 out of 22 studies reported air change rate values (mean $= 3.89 \text{ h}^{-1}$), out of which only three described the method of estimation [237-239]. These studies used the tracer gas decay method by means of low adsorption tracer gases such as CO2. The majority of the selected studies were performed in test rooms larger than 20 m³ (mean floor area = $30 \pm 27 \text{ m}^2$), which is important for mimicking various indoor layouts occupied with people and for studying air contaminant distribution in the space. Twelve studies focused on mimicking office environments, whereas other studies focused on aircraft (2), classroom (1), hospital (1) and other unspecified environments (6). Studies involving perceived air quality, Sick Building Syndrome (SBS) symptoms and productivity under variable levels of gas-phase pollutants [71,72, 229-235] had a significantly higher number of study participants (76 ± 9.3) compared to studies focusing on human inhalation exposure and spatio-temporal variation of indoor air pollutants (8.2 \pm 13.6) [20,



Fig. 12. Distribution of air quality studies' main topic over time.

228,236–241] and airflow distribution in occupied spaces and ventilation effectiveness (2.3 \pm 2.1) [226,227,242–244]. The majority of studies focused on measurements of CO₂ (9), followed by VOCs (7), particulate matter (4), and other inorganic gasses such as NO₂, N₂O, O₃, and CO. Measurements of these air pollutants were performed with scientific instruments, which were not an integral part of the test rooms. None of those studies reported the adoption of the optimal inner coating of the test room surfaces, which is essential to determine how these coatings influence heterogeneous reactions with volatile organic compounds and other gaseous pollutants. Among the selected papers, only a fraction (2) reported issues that could arise due to pollutant uptake or emissions in the test rooms. Furthermore, in all studies, there was a lack of integration between advanced online and offline instrumentation and analytical techniques within the test rooms.

4.5. Multi-domain and whole comfort experiments

The goals of a multi-domain experiment can be categorized into (i) evaluate the effect of specific building technologies or control strategies on occupant multi-domain comfort [119,245-250]; (ii) understand cross-modal and interaction between different domains [46,72, 251-268]; (iii) model the physiological [97,100,228,269,270] or behavioural [271-273] response of occupant to combined multi-domain stimuli and to understand the effect of IEQ on stress [274,275]; (iv) identifying new multi-domain metrics such as air enthalpy [251], air distribution index [276] and bio-signals such as skin temperature [277] for the whole comfort. In some cases, the energy consequences of such multi-domain interactions are also captured, as for the studies investigating novel personalized thermostats [272,278,279] or novel visual comfort systems [39,45] to improve energy efficiency and comfort. Among the studies focusing on the effect of specific building technologies or control strategies on occupant multi-domain comfort, the development of novel personal comfort systems in buildings [113,116, 245,246,248,280-286] and vehicles [118] has received particular attention.

The interest in studying occupant response to multi-domain stimuli has increasingly grown since 2000, especially after 2010. Multi-domain experiments constitute 23% of the overall 396 occupant comfort experiments in test rooms, as given by the review database. Most of these studies investigated the relationship between two physical domains, while studies focusing on three or more physical domains were just 4% of the whole database. In terms of investigated combinations of domains, thermal and air quality represent the most studied one, followed by thermal with visual and thermal with acoustic (Fig. 13).

The majority of the studies were conducted under stationary conditions, while only a third of the studies exposed occupants to changing



Fig. 13. Multi-domain experiments by combination of each domain.

environmental or dynamic conditions. Dynamic conditions were achieved either by step changes in indoor conditions [246,248,251,257, 274,287–289] or, especially concerning thermal-related studies, by fast and long changes [79,265,275,290], meaning that a rate of change greater than 2 K per hour is provided for more than 1 h of exposure. Only a few studies investigated multi-domain effects under high-speed conditions [91,250,291] or slow and long dynamical changes [263, 271,292].

In addition to highly accurate monitoring of environmental parameters, most studies capture occupants' responses as a combination of subjective and physiological parameters. Nearly half of the studies (53%) relied only on subjective occupants' responses. Table 4 shows the subjective metrics and physiological parameters monitored in the experiments. In terms of subjective measurements, based on survey or behavioural observations, environmental sensations are the most employed, followed by environmental preference and acceptability. In terms of physiological parameters, skin temperature and heart rate are the most monitored ones, also due to the thermal domain being investigated at least in 94% of the overall multi-domain experiments. Lastly, the use of EEG, ECG, and EDA has just recently started to be adopted, mostly after 2015, to understand multi-physical occupants' responses in test rooms, especially when investigating interactions between different comfort domains.

Table 4

Different approaches for capturing occupants' responses in multi-domain experiments in test rooms.

Occupant response		References
Subjective (survey based or from behavioural observations)	Environmental sensation	[44,46,47,60,67,78,79, 85,86,89–91,99–101,118, 120,122,124–126,245, 250,253,256,257,264, 267–269,273,275–277, 290–297]
	Environmental comfort	[44,47,60,72,85,89,91, 99,100,118,120,122,125, 245,252,253,267–269, 291,294,296–299]
	Environmental preference	[44,47,79,86,99,102,122, 248,252,253,257,267, 268,291,295,297]
	Acceptability	[47,85,86,91,122,124, 248,249,253,275,294, 295,297]
	Environmental satisfaction Emotion response Alertness	[46,78,252] [46,264,289] [50]
	Stress level Work performance	[274] [90,268,273,275,289, 294,295]
	Clothing level	[125,249,261,269,277]
Physiological parameters (sensing device based)	Skin temperature	[44,46,79,86,99–102,117, 120,125,253,258,259, 292,300]
	Skin moisture	[301]
	Core temperature	[86,102,258,259]
	Electrodermal activity (EDA)	[44,46]
	Electrocardiogram (ECG)	[47,100,294]
	Electrooculography (EOG)	[297]
	Electroencephalogram (EEG)	[47,274,289,294,297]
	Acceleration	[46,302]
	Heart rate	[44,46,266,289,292,303]
	Nasal dimension by	[301]
	acoustic rhinometry	
	Photoplethysmography	[302]
	Metabolic rate	[277]
	Frequency of blinking	[303]
	Mucociliary transport	[303]
	Saliva and tear mucus film samples	[295]

4.6. Participants interacting with the environment

This section focuses on those experiments whose protocol allowed participants to freely interact with the test room components and systems. The interactions taken into account for this further classification include adjusting settings of the test room conditioning system, dimming/switching lights, opening/closing windows and shading systems, adjusting personal comfort devices. According to the reviewed scientific publications, this section is based on 21 papers (see Table 5). Nine of those 21 have been published in 2018–2020, and ten originate from European universities or institutes.

Two papers describe a test room facility developed and constructed to test all environmental factors (lighting, acoustics, air, and thermal quality) [43,304], including interactions with the environment through design and systems, making it possible to provide both input data to and output data from the occupants. Most of the publications were concerned with thermal quality in relation to thermal comfort, sensations and/or preferences [102,124,305–307], in combination with (personal) control [36,111,281–283,308,309], together with air quality [232] or visual quality [45,310]. The latter was studied in three reported studies [73,189,191], of which one was concerned with daylight, glare, shading and control [73]. Only one study included all the IEQ aspects [311].

The participants involved in the different studies mostly comprise of students and healthy young adults. Only one study was concerned with children (primary school children with an average age of 10 years) [311]. One study included a comparison between young (average23 years) and older males (average 67 years) [102], and one study looked at the impact of ethnicity [309]. In most publications, the responses or interactions of a participant with an object or variable/parameter in the environment are reported. The studied controlling devices varied from (local) heating or ventilation devices [283], light dimmers [310] or blind/solar shading control device [73], wearable conditioning devices [111], and furniture [281]. Table 5 summarises the 21 papers concerning those experiments where the building occupant is able to interact with the test room in the form of personal judgments or specific actuator-to-reaction.

4.7. Energy-related human comfort experiments

Out of 396 reviewed papers, 85 considered energy-related issues while carrying out thermal-, visual-, indoor air quality-, and acoustic-related experiments. Of these, 28 papers had a multi-domain focus with 22 papers considering both thermal and air quality-related experiments, five papers presenting thermal- and visual-related experiments [97,232,281,312–314], and only one paper discussing the effect of personal control on thermal, visual, and air quality perceived by building occupants [310]. Among the single comfort domain studies, thermal investigations are by far the most widely carried out (50), followed by visual investigation (5). Olfactory and aural comfort were studied together with energy considerations in just one article each [156,232].

The first document of the database was published in 1978. For the following 30 years, much slower growth was observed in the number of publications on energy-related human comfort experiments. After 2008, the scientific interest in this topic has progressively increased because of the increasing research interest in human-centric building design [315], personalized control strategies [316], and perceptual and behavioural environmental studies [32] (Fig. 14).

The majority of experiments have been conducted in test rooms located inside buildings with controlled environmental conditions, and only three experiments were run considering the actual outdoor weather [46,97,312]. Furthermore, 45 experimental procedures employed dynamic conditions and 32 studies used steady-state conditions. Dynamic studies are generally more recent (the average publication year is 2013), while steady-state conditions are more common in older studies (the average publication year is 2011); this can be explained by the recent

Renewable and Sustainable Energy Reviews 149 (2021) 111359

Table 5

List of reviewed studies concerning human comfort experiments in test rooms where the participants could directly interact with the facility.

Year pub.	Investigated domain	Studied parameters/object	Interaction between the participant and the test room	Reference
			the test room	
1991	Thermal	Adjust ambient temperature	Adjustment of test room temperature	[307]
1995	Thermal	Two age groups	Adjustment of test room temperature	[102]
2000	Thermal	Adjusting air movement (supplied via ceiling)	Adjustment of the Personal Comfort System (PCS)	[308]
2007	Thermal	3 task air- conditioning systems	Adjustment of the Personal Comfort System (PCS)	[306]
2009	Thermal	Control of 2 fans at chair (under seat, behind backrest)	Adjustment of the Personal Comfort System (PCS)	[282]
2009	Visual	Dimming of light; airflow from ceiling-based nozzle	Adjustment of the Personal Comfort System (PCS)	[310]
2012	Thermal	4 fans at corners chair to enhance displacement vent	Adjustment of the Personal Comfort System (PCS)	[124]
2012	Thermal & Air quality	Air movement (air terminal device), air pollution, temperature and RH	Adjustment of the Personal Comfort System (PCS)	[232]
2012	Visual	Artificial lighting and blinds control, daylight	Adjustments of shading system	[73]
2014	Thermal	Ceiling fan	Adjustment of shading system, ceiling fan, operable windows	[36]
2014	Visual	Daylight	Adjustment of shading system	[189]
2015	Thermal	Heated/cooled chair	Adjustment of the Personal Comfort System (PCS)	[281]
2018	Thermal	Control of personalized beating system	Adjustment of the Personal Comfort System (PCS)	[283]
2018	Visual & Thermal & Air quality & Acoustics	Facades, controls, interior, etc.	Adjustment of shading system, façade properties, thermal settings	[43]
2018	Visual & Thermal & Air quality & Acoustics	Walls, lighting, sound, thermal, air, interior, etc.	Control of HVAC and lighting system	[304]
2019	Thermal & Visual	Windows, blinds and ceiling lights	Adjustment of desk light, ceiling light, solar shading, operable windows	[45]
2019	Visual & Thermal & Air quality & Acoustics	IEQ in their own classroom	IEQ problems in classrooms and solutions for those problems	[311]
2020	Visual	Daylight, glare, shading	Adjustment of shading system	[191]
2020	Thermal	Thermal sensation, thermal preference	Adjustment of the Personal Comfort System (PCS)	[305]
2020	Thermal	Wearable wrist devices for warming or cooling	Adjustment of the Personal Comfort System (PCS)	[111]
2020	Thermal	Self-selected air temperature, thermal sensation, comfort and preferences; skin temperature	Adjustment of the personal comfort system	[309]



Fig. 14. Cumulative number of publications describing energy-related issues in human comfort experiments in test rooms.

availability of easier and user-friendly control interfaces and power modulation for electric motors and pumps.

Regarding the technical systems used during the experiments, only considering the documents where this information was expressed, most of the investigations used air-conditioning systems and only a few tested hypotheses under radiant systems (9 papers), controlled mechanical ventilation (8), artificial lighting (10), and sound equipment (1) [156]. Additionally, 38 papers reported experiments, which adopted personal environmental control systems, which are effective means of testing energy-saving control strategies and are well received by the occupants.

5. Summary of key findings

This review analysed a wide range of test rooms for the experimental investigation of human comfort indoors and provided an overview of scientific experiments that are conducted in such facilities and that were published in scientific papers. All reported information was deducted from reviewed papers. According to such an approach, it has to be mentioned that experimental facilities may exist which have not (yet) published any results in peer-reviewed journal articles. The reason may be because (i) it is too new to present results, or (ii) the facility is dedicated to industrial or other research not meant for public sharing of results. This limitation may affect some of our conclusions. Nevertheless, while accepting this limitation, we believe that the number of facilities not included in our review is small due to the two search strategies applied and that the knowledge generated in those facilities not publishing their work, for one reason or the other, is in any case not directly available for the scientific community and less suitable to enhance human comfort theories.

A general observation pertains to the growing number of such facilities. The total number of 187, specifically referred to in the present contribution, is about eight times higher than the number of comparable facilities before 2000. However, the geographic distribution of these facilities does not reflect the variance of climatic regions around the world: 82% are located in moderate climatic regions. Notwithstanding, the increasing number of test rooms may reflect the growing realization of the influence of indoor environments on human health, comfort, and productivity. This trend is reflected in the increasing number of publications reporting research conducted in these facilities. In this review, a total number of 396 publications were considered.

Looking at the publications from a topical standpoint reveals the scientific community's primary interest in human thermal comfort (204 papers), followed by energy-related studies (85), visual comfort (70), air quality (18), and acoustic comfort (11). Roughly a quarter of the reviewed publications explored indoor-environmental exposure situations involving more than one domain. Only a small number of publications (21) investigated circumstances in which participants could assume an active role and had the opportunity to interact with relevant features of the indoor environment.

Our findings suggest that about 92% of the test rooms were built inside of a building. This is interesting: while the performance characterization of building components has mainly been tested in outdoor testing facilities [317], the investigation of indoor comfort has been conducted either in actual occupied buildings or in dedicated test rooms located indoors with potential better controlled experimental procedures and microclimate conditions. However, based on the reviewed publications alone, it is not possible to draw up a more detailed picture of the test rooms' design and construction. For instance, in 47% of the reviewed publications, it was not possible to ascertain whether the test room envelope entailed any type of openings. Lack of such details makes it difficult to independently replicate and subsequently validate the results coming from experiments in test rooms. Our review also addresses another critical point: there is a lack of information regarding investment, operation, and maintenance costs associated with the facilities. A dedicated survey designed and distributed on our side received responses only from 18 facility owners or operators, pointing to the need for further efforts in the transferability of know-how with the test rooms.

Certain observations apply to studies that focused exclusively on thermal comfort: studies on fundamental issues dominate in this area (57%) versus technology-oriented (36%) and predictive studies (7%). An increasing number of experiments in the last 20 years focus on local heating/cooling systems. A large share of technology-oriented studies (40%) focuses on developing new, insulating, and thermally active clothing. This may indicate a shift in the industry from the traditional room-air-conditioning design perspective to a more personalized thermal comfort approach. The majority of the reviewed studies were conducted in office-like environments with small samples (10–50 people) engaging in sedentary activities. Few papers focused on the elderly or children (3%), and in 70% of the studies, participants were passive recipients only. Some studies introduced new, recently emerging methods such as immersive virtual reality and monitoring of brain activity patterns.

Studies related to acoustic comfort mostly followed a general procedure where participants were exposed, on a short-term basis, to stimuli while performing cognitive tasks or completed subjective tests. Interestingly, only four studies (less than 40%) screened the hearing abilities of the participants. This may have introduced bias in their results.

Studies on lighting and visual comfort significantly increased in the last decade, addressing both daylight and electric light: their bulk is concerned with glare problems in the workplace, primarily deal with glare perception and entail the development and evaluation of related metrics, thresholds or indexes. The investigations also pertain to various human responses related to light quantity and CCT. Most of these latter investigations focused on the non-image-forming effects of light. Only a few studies allowed participants to change the visual conditions by interacting with blinds and electric lights.

IAQ-related studies mostly addressed three topics, namely the perceived air quality's impact on productivity and health, the spatiotemporal variation of air pollution and inhalation exposure, and the airflow distribution and ventilation effectiveness. Some reviewed publications did not report the experimental conditions (e.g., ventilation rates) in detail. In contrast, none of the studies reported surface materials, which is essential concerning how they influence heterogeneous reactions with volatile organic compounds and other gaseous pollutants. The majority of the studies were conducted in sufficiently large test rooms, hence allowing for the consideration of realistic room layouts and air contaminant distribution patterns.

About investigations of multi-domain exposure situations, thermal and indoor air quality represent the most frequently studied combination, followed by thermal-visual and thermal-acoustic combinations. Only one-third of the studies exposed participants to dynamic environmental conditions. 53% of the studies relied solely on subjective responses. In the last few years, a new trend can be seen in the related scientific literature, whereby diagnostic methods from neurophysiology (such as EEG, ECG and EDA) have been applied to explore multi-domain exposure situations.

6. Research gaps and future trends in test room experiments for human comfort

Despite a growing interest in multi-domain studies, we still do not have an agreed-upon conceptual framework and a systematic methodology for a mature and holistic science of human-centric indoor environments. Common design guidelines and a shared terminology for innovative test rooms and experimental procedures would allow establishing a shared understanding of the driving phenomena and the inclusion of the non-physical (psychological and contextual factors) dimensions. This can be further supported by the deployment of lowinvasive physiological sensing techniques. A better understanding of the visual, IAQ and acoustic factors and their mutual influence on human comfort and occupants' perception requires further investigation. Future trends in test room experiments (and thus facility design) must account for a multi-domain and multi-disciplinary approach.

On a geographical and demographic basis, despite the increased interest in human comfort and the large availability of test room setups, these facilities are limited to specific climatic regions, while concerning tested subjects' composition, these are mainly students and faculty members. These sociological and geographical weak points may cause a non-negligible bias in the interpretations of experimental results and knowledge generation. We see the need for dedicated studies in those climatic and demographic contexts where experimental data are still not available to increase diversity and cross-validation.

In terms of test room design: test rooms mostly emulate office spaces with a limited number of occupants. Therefore, another research gap to close is the analysis of other settings and contexts, such as realistic openplan offices and different building typologies (educational, residential, hospitals, etc.). This factor may affect the quality of the collected data and limit the research findings to office-only investigations (difficult to replicate and extend).

Concerning experiments, increased attention is being paid to occupants, also driven by the recent trends toward human-centric building design and operation. This is also reflected in the fast growth of multidomain studies in the last decade, where the focus is the whole comfort perception analysis. Additionally, even technology-oriented studies are focused on human applications. About 40% of the technologyoriented studies aimed at developing and testing wearable systems for improved personal comfort, such as smart clothing and sensing techniques. This observation shows the necessity for a more systematic collaborative research framework whereby the environmental comfort is not handled exclusively by building physicists or engineers and architects. The topic requires a significant and proactive interaction with researchers in human factors, human-machine interaction, big data analytics, and social science, as we see more studies are focusing on psycho-physiological factors alongside IEQ and human-centric approaches.

From the operation perspective, the economic analysis showed the necessity for a better common understanding of the economic model behind test room design and construction. This may be helpful to foster local and global collaboration and connection to industry, taking advantage of the unique resources that each location provides. For this purpose, a higher transparency of existing business/economic models is recommended. Private-public partnerships may also be established with shared economic models allowing both researchers and industry partners to use these facilities to conduct controlled experimental studies, e. g., for technology development. Such models can also help sustain and expand the test rooms' role in underlining the importance of whole comfort experiments. Toward this end, funding agencies/industry partners should be informed and engaged in providing funding support to maintain/sustain and expand existing testbeds dedicated to a better understanding of human comfort in buildings.

In this context, standardization in the design and experimental

validation procedures is still missing, with the consequent limitations in error and uncertainty analysis, quality control and replicability potential. Therefore, the creation of a unified framework for keeping track of the functionality of the test room facilities is expected to establish a common ground for collaboration and cross-validation and would help to identify cultural and geographical differences and biases.

This cannot be done without a joint effort in terms of open-source research in and for society, where the resources of test room facilities and collected data are freely available for fostering the impact of these multi-domain and multi-disciplinary investigations. In this scenario, future efforts by the authors and their institutions would support research via a systematic data sharing process and a publicly available and continuously updated test room portfolio. Finally, a first Round-Robin test in test room facilities worldwide is expected to emerge as a follow up to this review.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

This review was conducted within the framework of IEA-EBC Annex 79. A.L. Pisello and I. Pigliautile thank the Italian Ministry of Research for supporting NEXT.COM (Towards the NEXT generation of multiphysics and multidomain environmental COMfort) project (Prin 2017-20172FSCH4). I. Pigliautile's acknowledgments are due to the European Union's Horizon 2020 program - GEOFIT project (G.A. 792210). M. Vellei's work is supported by the French National Research Agency under the project ANR CLEF (ANR-17-CE22-0005-01). B. Dong, M. Kong and Y.P. Liu's work is supported by U.S. National Science Foundation (Award: #1949372) and Syracuse Centre of Excellence Faculty Fellowship Award. M. Favero and A. Nocente would like to thank the Research Centre on Zero Emission Neighborhoods in Smart Cities (FME ZEN, Grant n. 257660) and the Research Council of Norway (Norges Forskingsrådet) for the support. M. Schweiker is supported by a research grant (21055) from VILLUM FONDEN. Z. Deme Bélafi has been supported by the NRDI Fund (TKP2020 IES, Grant No. BME-IE-MISC) based on the charter of bolster issued by the NRDI Office under the auspices of the Ministry for Innovation and Technology within the framework of project (no. K 128199) with the support provided from the National Research, Development and Innovation Fund of Hungary, financed under the K 18 funding scheme. A. Wagner's related work is funded by the German Ministry for Economic Affairs and Energy through various projects.

References

- [1] Klepeis NE, Nelson WC, Ott WR, Robinson JP, Tsang AM, Switzer P, et al. The National Human Activity Pattern Survey (NHAPS): a resource for assessing exposure to environmental pollutants. J Expo Anal Environ Epidemiol 2001;11: 231–52. https://doi.org/10.1038/sj.jea.7500165.
- [2] Kjellstrom T, Holmer I, Lemke B. Workplace heat stress, health and productivity an increasing challenge for low and middle-income countries during climate change. Glob Health Action 2009;2:2047. https://doi.org/10.3402/gha. v2i0.2047.
- [3] Tanabe SI, Iwahashi Y, Tsushima S, Nishihara N. Thermal comfort and productivity in offices under mandatory electricity savings after the Great East Japan earthquake. Architect Sci Rev 2013;56:4–13. https://doi.org/10.1080/ 00038628.2012.744296. Taylor & Francis.
- [4] Delzendeh E, Wu S, Lee A, Zhou Y. The impact of occupants' behaviours on building energy analysis: a research review. Renew Sustain Energy Rev 2017;80: 1061–71. https://doi.org/10.1016/j.rser.2017.05.264.
- [5] Naylor S, Gillott M, Lau T. A review of occupant-centric building control strategies to reduce building energy use. Renew Sustain Energy Rev 2018;96: 1–10. https://doi.org/10.1016/j.revr.2018.07.019.
- [6] Yan D, O'Brien W, Hong T, Feng X, Burak Gunay H, Tahmasebi F, et al. Occupant behavior modeling for building performance simulation: current state and future

Renewable and Sustainable Energy Reviews 149 (2021) 111359

challenges. Energy Build 2015;107:264-78. https://doi.org/10.1016/j.enbuild.2015.08.032.

- [7] Lassen N, Goia F, Schiavon S, Pantelic J, Field investigations of a smiley-face polling station for recording occupant satisfaction with indoor climate. Build Environ 2020;185:107266. https://doi.org/10.1016/j.buildenv.2020.107266.
- [8] Castaldo VL, Pigliautile I, Rosso F, Cotana F, De Giorgio F, Pisello AL. How subjective and non-physical parameters affect occupants' environmental comfort perception. Energy Build 2018;178:107–29. https://doi.org/10.1016/j. enbuild.2018.08.020.
- [9] Lai ACK, Mui KW, Wong IT, Law LY. An evaluation model for indoor environmental quality (IEQ) acceptance in residential buildings. Energy Build 2009;41:930–6. https://doi.org/10.1016/j.enbuild.2009.03.016.
- [10] Palacios Temprano J, Eichholtz P, Willeboordse M, Kok N. Indoor environmental quality and learning outcomes: protocol on large-scale sensor deployment in schools. BMJ Open 2020;10:31233. https://doi.org/10.1136/bmjopen-2019-031233.
- [11] Han J, Kwon S bark, Chun C. Indoor environment and passengers' comfort in subway stations in Seoul. Build Environ 2016;104:221–31. https://doi.org/ 10.1016/j.buildenv.2016.05.008.
- [12] Sakellaris I, Saraga D, Mandin C, Roda C, Fossati S, de Kluizenaar Y, et al. Perceived indoor environment and occupants' comfort in European "modern" office buildings: the OFFICAIR study. Int J Environ Res Publ Health 2016;13:444. https://doi.org/10.3390/jjerph13050444.
- [13] Roulet CA, Johner N, Foradini F, Bluyssen P, Cox C, De Oliveira Fernandes E, et al. Perceived health and comfort in relation to energy use and building characteristics. Build Res Inf 2006;34:467–74. https://doi.org/10.1080/ 09613210600822279.
- [14] Debnath KB, Jenkins DP, Patidar S, Peacock AD. Understanding residential occupant cooling behaviour through electricity consumption in warm-humid climate. Buildings 2020;10:78. https://doi.org/10.3390/BUILDINGS10040078
- [15] Földváry Ličina V, Cheung T, Zhang H, de Dear R, Parkinson T, Arens E, et al. Development of the ASHRAE global thermal comfort database II. Build Environ 2018;142:502-12. https://doi.org/10.1016/j.buildenv.2018.06.022
- [16] Franke M, Nadler C. Towards a holistic approach for assessing the impact of IEQ on satisfaction, health, and productivity. Build Res Inf 2020:1–28. https://doi. org/10.1080/09613218.2020.1788917.
- [17] Piselli C, Pisello AL. Occupant behavior long-term continuous monitoring integrated to prediction models: impact on office building energy performance. Energy 2019;176:667-81. https://doi.org/10.1016/J.ENERGY.2019.04.005.
- [18] Torresin S, Pernigotto G, Cappelletti F, Gasparella A. Combined effects of environmental factors on human perception and objective performance: a review of experimental laboratory works. Indoor Air 2018;28:525–38. https://doi.org/ 10.1111/ina.12457.
- [19] Wang D, Zhang H, Arens E, Huizenga C. Observations of upper-extremity skin temperature and corresponding overall-body thermal sensations and comfort. Build Environ 2007;42:3933–43. https://doi.org/10.1016/j. buildew. 2006.06.035
- [20] Ai Z, Hashimoto K, Melikov AK. Influence of pulmonary ventilation rate and breathing cycle period on the risk of cross-infection. Indoor Air 2019;29: 993–1004. https://doi.org/10.111/ina.12589.
- [21] Kim W, Han H, Kim JT. The position index of a glare source at the borderline between comfort and discomfort (BCD) in the whole visual field. Build Environ 2009;44:1017–23. https://doi.org/10.1016/j.buildenv.2008.07.007.
- [22] Tuaycharoen N. Windows are less glaring when there is a preferred view. Built-Environment Sri Lanka 2011;9:45. https://doi.org/10.4038/besl.v9i1-2.3064.
- [24] Van Den Wymelenberg KG. Visual comfort, discomfort glare, and occupant fenestration control: developing a research agenda. LEUKOS - J Illum Eng Soc North Am 2014;10:207–21. https://doi.org/10.1080/15502724.2014.939004.
- [25] Pierson C, Wienold J, Bodart M. Review of factors influencing discomfort glare perception from daylight. LEUKOS - J Illum Eng Soc North Am 2018;14:111–48. https://doi.org/10.1080/15502724.2018.1428617.
- [26] Galasiu AD, Veitch JA. Occupant preferences and satisfaction with the luminous environment and control systems in daylit offices: a literature review. Energy Build 2006;38:728–42. https://doi.org/10.1016/j.enbuild.2006.03.001.
- [27] Shafavi NS, Zomorodian ZS, Tahsildoost M, Javadi M. Occupants visual comfort assessments: a review of field studies and lab experiments. Sol Energy 2020;208: 249–74. https://doi.org/10.1016/j.solener.2020.07.058.
- [28] Djongyang N, Tchinda R, Njomo D. Thermal comfort: a review paper. Renew Sustain Energy Rev 2010;14:2626–40. https://doi.org/10.1016/j. rser.2010.07.040.
- [29] Van Craenendonck S, Lauriks L, Vuye C, Kampen J. A review of human thermal comfort experiments in controlled and semi-controlled environments. Renew Sustain Energy Rev 2018;82:3365–78. https://doi.org/10.1016/j. rser.2017.10.053.
- [30] Rupp RF, Vásquez NG, Lamberts R. A review of human thermal comfort in the built environment. Energy Build 2015;105:178–205. https://doi.org/10.1016/j. enbuild.2015.07.047.
- [31] Park JY, Nagy Z. Comprehensive analysis of the relationship between thermal comfort and building control research - a data-driven literature review. Renew Sustain Energy Rev 2018;82:2664–79. https://doi.org/10.1016/j. rser.2017.09.102.
- [32] Schweiker M, Ampatzi E, Andargie MS, Andersen RK, Azar E, Barthelmes VM, et al. Review of multi-domain approaches to indoor environmental perception and behaviour. Build Environ 2020;176:106804. https://doi.org/10.1016/j. buildenv.2020.106804.

[33] Peel MC, Finlayson BL, Mcmahon TA. Updated world map of the Köppen-Geiger climate classification. Hydrol Earth Syst Sci Discuss 2007;4:439–73.

- [34] Fisekis K, Davies M, Kolokotroni M, Langford P. Prediction of discomfort glare from windows. Light Res Technol 2003;35:360–9. https://doi.org/10.1191/ 13657828031i0950a.
- [35] Wienold J, Christoffersen J. Evaluation methods and development of a new glare prediction model for daylight environments with the use of CCD cameras. Energy Build 2006;38:743–57. https://doi.org/10.1016/j.enbuild.2006.03.017.
- [36] Schweiker M, Brasche S, Hawighorst M, Bischof W, Wagner A. Presenting LOBSTER, an innovative climate chamber, and the analysis of the effect of a ceiling fan on the thermal sensation and performance under summer conditions in an office-like setting. In: Proc. - wind. Conf. 2014 count. Cost comf. A chang. World, NCEUB 2014; 2019. p. 924–37.
- [37] Rodriquez R, Pattini A. Tolerance of discomfort glare from a large area source for work on a visual display. Light Res Technol 2014;46:157–70. https://doi.org/ 10.1177/147715512470386.
- [38] Irshad K, Habib K, Basrawi F, Thirumalaiswamy N, Saidur R, Saha BB. Thermal comfort study of a building equipped with thermoelectric air duct system for tropical climate. Appl Therm Eng 2015;91:1141–55. https://doi.org/10.1016/j. applthermaleng.2015.08.077.
- [39] Karlsen L, Heiselberg P, Bryn I. Occupant satisfaction with two blind control strategies: slats closed and slats in cut-off position. Sol Energy 2015;115:166–79. https://doi.org/10.1016/j.solener.2015.02.031.
- [40] Konstantzos I, Chan YC, Seibold JC, Tzempelikos A, Proctor RW, Protzman JB. View clarity index: a new metric to evaluate clarity of view through window shades. Build Environ 2015;90:206–14. https://doi.org/10.1016/j. buildenv.2015.04.005.
- [41] Ascione F, De Masi RF, De Rossi F, Ruggiero S, Vanoli GP. MATRIX, a multi activity test-room for evaluating the energy performances of "building/HVAC" systems in Mediterranean climate: experimental set-up and CFD/BPS numerical modeling. Energy Build 2016;126:424–46. https://doi.org/10.1016/j. enbuild.2016.05.044.
- [42] Kent MG, Altomonte S, Wilson R, Tregenza PR. Temporal effects on glare response from daylight. Build Environ 2017;113:49–64. https://doi.org/ 10.1016/j.buildenv.2016.09.002.
- [43] Luna-Navarro A, Meizoso M, DeBleecker H, Verhoeven Y, Donato M, Overend M, Facade impulse: experimental methods for stretching the envelope beyond human comfort. Proceeding VIII Int. Congr. Archit. Envel. 2018:21–2.
- [44] Chinazzo G, Wienold J, Andersen M. Combined effects of daylight transmitted through coloured glazing and indoor temperature on thermal responses and overall comfort. Build Environ 2018;144:583–97. https://doi.org/10.1016/j. buildenv.2018.08.045.
- [45] Lolli N, Nocente A, Brozovsky J, Woods R, Grynning S. Automatic vs manual control strategy for window blinds and celling lights: consequences to perceived visual and thermal discomfort. J Daylighting 2019;6:112–23. https://doi.org/ 10.15627/jd.2019.11.
- [46] Salamone F, Bellazzi A, Belussi L, Damato G, Danza L, Dell'aquila F, et al. Evaluation of the visual stimuli on personal thermal comfort perception in real and virtual environments using machine learning approaches. Sensors 2020;20. https://doi.org/10.3390/s20061627.
- [47] Pigliautile I, Casaccia S, Morresi N, Arnesano M, Pisello AL, Revel GM. Assessing occupants' personal attributes in relation to human perception of environmental comfort: measurement procedure and data analysis. Build Environ 2020;177: 106901. https://doi.org/10.1016/j.buildenv.2020.106901.
- [48] Zhai Y, Miao F, Yang L, Zhao S, Zhang H, Arens E. Using personally controlled air movement to improve comfort after simulated summer commute. Build Environ 2019;165:106329. https://doi.org/10.1016/j.buildenv.2019.106329.
- [49] Stacy RW, Seal E, Green J, House D. Pulmonary function in normal humans with exercise and temperature-humidity stress. J Appl Physiol Respir Environ Exerc Physiol 1982;53:1015–8. https://doi.org/10.1152/jappl.1982.53.4.1015.
- [50] Corttili G, Mognoni P, Saibene F. Work tolerance and physiological responses to thermal environment wearing protective nbc clothing. Ergonomics 1996;39: 620–33. https://doi.org/10.1080/00140139608964485.
- [51] Nishihara N, Tanabe S-I, Hayama H, Komatsu M. A cooling vest for working comfortably in a moderately hot environment, vol. 21; 2002.
- [52] Potkanowicz ES, Caine N, Otterstetter R, Glickman EL. Thermal sensation of old vs young males at 12, 18, and 27° C for 120 min. Elsevier Ergon B Ser 2005;3: 147–50. https://doi.org/10.1016/S1572-347X(05)80025-4.
- [53] Laing RM, Sims ST, Wilson CA, Niven BE, Cruthers NM. Differences in wearer response to garments for outdoor activity. Ergonomics 2008;51:492–510. https:// doi.org/10.1080/00140130701636520.
- [54] WH Y, ZW Y. Study on improving the thermal-wet comfort of clothing during exercise with an assembly of fabrics, vol. 17; 2009.
- [55] Waterhouse J, Hudson P, Edwards B. Effects of music tempo upon submaximal cycling performance. Scand J Med Sci Sports 2010;20:662–9. https://doi.org/ 10.1111/j.1600-0838.2009.00948.x.
- [56] Karlsson CM, Rosenblad EFS. Evaluating functional clothing in climatic chamber tests versus field tests: a comparison of quantitative and qualitative methods in product development. Ergonomics 1998;41:1399–420. https://doi.org/10.1080/ 001401398186171.
- [57] Zhang XH, Li J, Wang Y. Effects of clothing ventilation openings on thermoregulatory responses during exercise, vol. 37; 2012.
- [58] Kim J, Na Y. Skin temperature and predicted discomfort of women wearing sheer empire style dress. Fibers Polym 2016;17:963–70. https://doi.org/10.1007/ s12221-016-5911-3.

- [59] Wang H, Hu S. Experimental study on thermal sensation of people in moderate activities. Build Environ 2016;100:127–34. https://doi.org/10.1016/j. buildenv.2016.02.016.
- [60] Mekjavic IB, Ciuha U, Grönkvist M, Eiken O. The effect of low ambient relative humidity on physical performance and perceptual responses during load carriage. Front Physiol 2017;8:451. https://doi.org/10.3389/fphys.2017.00451.
- [61] Yi W, Chan APC, Wong FKW, Wong DP. Effectiveness of a newly designed construction uniform for heat strain attenuation in a hot and humid environment. Appl Ergon 2017;58:555–65. https://doi.org/10.1016/j.aprog.2016.04.011.
- [62] Ji W, Luo M, Cao B, Zhu Y, Geng Y, Lin B. A new method to study human metabolic rate changes and thermal comfort in physical exercise by CO2 measurement in an airtight chamber. Energy Build 2018;177:402–12. https://doi. org/10.1016/j.enbuild.2018.08.018.
- [63] Celcar D. Thermo-physiological comfort of business clothing incorporating phase change materials in a cold environment. Int J Cloth Sci Technol 2018;30:49–61. https://doi.org/10.1108/JJCST-04-2017-0050.
- [64] Na HS, Choi JH, Kim HS, Kim T. Development of a human metabolic rate prediction model based on the use of Kinect-camera generated visual data-driven approaches. Build Environ 2019;160:106216. https://doi.org/10.1016/j. buildenv.2019.106216.
- [65] Wyon DP, Andersen I, Lundqvist GR. Spontaneous magnitude estimation of thermal discomfort during changes in the ambient temperature. J Hyg 1972;70: 203–21. https://doi.org/10.1017/S0022172400022269.
- [66] Ngai P, Boyce P. The effect of overhead glare on visual discomfort. J Illum Eng Soc 2000;29:29–38. https://doi.org/10.1080/00994480.2000.10748315.
- [67] Fong ML, Lin Z, Fong KF, Chow TT, Yao T. Evaluation of thermal comfort conditions in a classroom with three ventilation methods. Indoor Air 2011;21: 231–9. https://doi.org/10.1111/j.1600-0668.2010.00693.x.
- [68] Cai H, Chung T. Evaluating discomfort glare from non-uniform electric light sources. Light Res Technol 2013;45:267–94. https://doi.org/10.1177/ 1477153512453274.
- [69] Bluyssen PM, Zhang D, Kim DH, Eijkelenboom AM, Ortiz-Sanchez M. First SenseLab studies with primary school children: exposure to different environmental configurations in the experience room. Intell Build Int 2019;1–18. https://doi.org/10.1080/17508975.2019.1661220.
- [70] Ohnaka T, Hodder S, Parsons K. The effects of simulated solar radiation to the head and trunk on the thermal comfort of seated subjects. Elsevier Ergon B Ser 2005;3:305–11. https://doi.org/10.1016/S1572-347X(05)80049-7.
- [71] Strøm-Tejsen P, Wyon DP, Lagercrantz L, Fang L. Passenger evaluation of the optimum balance between fresh air supply and humidity from 7-h exposures in a simulated aircraft cabin. Indoor Air 2007;17:92–108. https://doi.org/10.1111/ j.1600-0668.2006.00458.x.
- [72] Grün G, Trimmel M, Holm A. Low humidity in the aircraft cabin environment and its impact on well-being - results from a laboratory study. Build Environ 2012;47: 23-31. https://doi.org/10.1016/j.buildenv.2011.05.004.
- [73] Shin HY, Kim G, Kim JT. Effect of occupants' behaviour of daylight controls on residential visual environment. Indoor Built Environ 2013;22:191–202. https:// doi.org/10.1177/1420326X12469735.
- [74] International Organization for Standardization. ISO 7730:2005 ergonomics of the thermal environment – analytical determination and interpretation of thermal comfort using calculation of the PMV and PPD indices and local thermal comfort criteria 2005.
- [75] EN 12931. Chemicals used for treatment of water intended for human consumption - chemicals for emergency use - sodium dichloroisocyanurate, anhydrous n.d. 2015.
- [76] EN 16798-1:2019 energy performance of buildings ventilation for buildings -Part 1: indoor environmental input parameters for design and assessment of energy performance of buildings addressing indoor air quality, thermal environment, lighting and acoustics - module M1-6 n.d.
- [77] EN 12464-1. Light and lighting lighting of work places Part 1: indoor work places n.d. 2011.
- [78] Wu Y, Liu H, Li B, Kosonen R, Kong D, Zhou S, et al. Thermal adaptation of the elderly during summer in a hot humid area: psychological, behavioral, and physiological responses. Energy Build 2019;203:109450. https://doi.org/ 10.1016/j.enbuild.2019.109450.
- [79] Kong D, Liu H, Wu Y, Li B, Wei S, Yuan M. Effects of indoor humidity on building occupants' thermal comfort and evidence in terms of climate adaptation. Build Environ 2019;155:298–307. https://doi.org/10.1016/j.buildenv.2019.02.039.
- [80] Schweiker M, Schakib-Ekbatan K, Fuchs X, Becker S. A seasonal approach to alliesthesia. Is there a conflict with thermal adaptation? Energy Build 2020;212: 109745. https://doi.org/10.1016/j.enbuild.2019.109745.
- [81] Luo M, Cao B, Ouyang Q, Zhu Y. Indoor human thermal adaptation: dynamic processes and weighting factors. Indoor Air 2017;27:273–81. https://doi.org/ 10.1111/ina.12323.
- [82] Chun C, Kwok A, Mitamura T, Miwa N, Tamura A. Thermal diary: connecting temperature history to indoor comfort. Build Environ 2008;43:877–85. https:// doi.org/10.1016/j.buildenv.2007.01.031.
- [83] Wu Y, Yuan M, Li C, Cheng Y, Liu H. The effect of indoor thermal history on human thermal responses in cold environments of early winter. J Therm Biol 2019;86:102448. https://doi.org/10.1016/j.jtherbio.2019.102448.
- [84] Chong D, Zhu N, Luo W, Zhang Z. Broadening human thermal comfort range based on short-term heat acclimation. Energy 2019;176:418–28. https://doi.org/ 10.1016/j.energy.2019.04.007.
- [85] Zhang Y, Chen H, Wang J, Meng Q. Thermal comfort of people in the hot and humid area of China—impacts of season, climate, and thermal history. Indoor Air 2016;26:820–30. https://doi.org/10.1111/ina.12256.

- [86] Cándido C, de Dear R, Ohba M. Effects of artificially induced heat acclimatization on subjects' thermal and air movement preferences. Build Environ 2012;49: 251–8. https://doi.org/10.1016/j.buildenv.2011.09.032.
- [87] Du C, Li B, Liu H, Wei Y, Tan M. Quantifying the cooling efficiency of air velocity by heat loss from skin surface in warm and hot environments. Build Environ 2018;16:146–55. https://doi.org/10.1016/j.buildenv.2018.03.023.
- [88] Chow TT, Fong KF, Givoni B, Lin Z, Chan ALS. Thermal sensation of Hong Kong people with increased air speed, temperature and humidity in air-conditioned environment. Build Environ 2010;45:2177–83. https://doi.org/10.1016/j. buildenv.2010.03.016.
- [89] Luo M, Yu J, Ouyang Q, Cao B, Zhu Y. Application of dynamic airflows in buildings and its effects on perceived thermal comfort. Indoor Built Environ 2018; 27:1162–74. https://doi.org/10.1177/1420326X17702520.
- [90] Cui W, Cao G, Ouyang Q, Zhu Y. Influence of dynamic environment with different airflows on human performance. Build Environ 2013;62:124–32. https://doi.org/ 10.1016/j.buildenv.2013.01.008.
- [91] Li C, Liu H, Li B, Cheng Y, Du C, Sheng A. Human responses to the air relative humidity ramps: a chamber study. Build Environ 2017;123:458–68. https://doi. org/10.1016/j.buildenv.2017.07.024.
- [92] Li C, Liu H, Li B, Sheng A. Seasonal effect of humidity on human comfort in a hot summer/cold winter zone in China. Indoor Built Environ 2019;28:264–77. https://doi.org/10.1177/14203263(17751594.
- [93] Du X, Li B, Liu H, Yang D, Yu W, Liao J, et al. The response of human thermal sensation and its prediction to temperature step-change (Cool-Neutral-Cool). PloS One 2014;9:e104320. https://doi.org/10.1371/journal.pone.0104320.
- [94] Li B, Du C, Tan M, Liu H, Essah E, Yao R. A modified method of evaluating the impact of air humidity on human acceptable air temperatures in hot-humid environments. Energy Build 2018;158:393–405. https://doi.org/10.1016/j. enbuild.2017.09.062.
- [95] Lan L, Lian Z, Liu W, Liu Y. Investigation of gender difference in thermal comfort for Chinese people. Eur J Appl Physiol 2008;102:471–80. https://doi.org/ 10.1007/s60421-007-6609-2.
- [96] Hashiguchi N, Feng Y, Tochihara Y. Gender differences in thermal comfort and mental performance at different vertical air temperatures. Eur J Appl Physiol 2010;109:41–8. https://doi.org/10.1007/s00421-009-1158-7. Springer.
- [97] Irshad K, Algarni S, Jamil B, Ahmad MT, Khan MA. Effect of gender difference on sleeping comfort and building energy utilization: field study on test chamber with thermoelectric air-cooling system. Build Environ 2019;152:214–27. https://doi. org/10.1016/j.buildenv.2019.01.058.
- [98] Wenger S, Wright T, Bechtold T, Nachbauer W. Sex-related differences in clothing-microclimate and subjective perceptions while wearing two outdoor jackets during submaximal exercise in a cool environment. J Text Inst 2019;110: 1343–51. https://doi.org/10.1080/00405000.2019.1580474.
- [99] Soebarto V, Zhang H, Schiavon S. A thermal comfort environmental chamber study of older and younger people. Build Environ 2019;155:1–14. https://doi. org/10.1016/j.buildenv.2019.03.032.
- [100] Xiong J, Ma T, Lian Z, de Dear R. Perceptual and physiological responses of elderly subjects to moderate temperatures. Build Environ 2019;156:117–22. https://doi.org/10.1016/j.buildenv.2019.04.012.
- [101] Bae C, Lee H, Chun C. Predicting indoor thermal sensation for the elderly in welfare centres in Korea using local skin temperatures. Indoor Bullt Environ 2017; 26:1155–67. https://doi.org/10.1177/1420326X16664563.
- [102] Taylor NAS, Allsopp NK, Parkes DG. Preferred room temperature of young vs aged males: the influence of thermal sensation, thermal comfort, and affect. Journals Gerontol - Ser A Biol Sci Med Sci 1995;50:M216–21. https://doi.org/10.1093/ gerona/SOA.4.M216.
- [103] Jin Q, Duanmu L, Zhang H, Li X, Xu H. Thermal sensations of the whole body and head under local cooling and heating conditions during step-changes between workstation and ambient environment. Build Environ 2011;46:2342–50. https:// doi.org/10.1016/i.buildeny.2011.05.017.
- [104] Zhang Z, Zhang Y, Ding E. Acceptable temperature steps for transitional spaces in the hot-humid area of China. Build Environ 2017;121:190–9. https://doi.org/ 10.1016/j.buildenv.2017.05.026.
- [105] Chen CP, Hwang RL, Chang SY, Lu YT. Effects of temperature steps on human skin physiology and thermal sensation response. Build Environ 2011;46:2387–97. https://doi.org/10.1016/j.buildenv.2011.05.021.
- [106] Xiong J, Lian Z, Zhou X, You J, Lin Y. Effects of temperature steps on human health and thermal comfort. Build Environ 2015;94:144–54. https://doi.org/ 10.1016/j.buildenv.2015.07.032.
- [107] Dear RJ, Ring JW, Fanger PO. Thermal sensations resulting from sudden ambient temperature changes. Indoor Air 1993;3:181–92. https://doi.org/10.1111/ j.1600-0668.1993.101-100004.x.
- [108] Luo M, Cao B, Ji W, Ouyang Q, Lin B, Zhu Y. The underlying linkage between personal control and thermal comfort: psychological or physical effects? Energy Build 2016;111:56–63. https://doi.org/10.1016/j.enbuild.2016.11.004.
- [109] Kim M, Chong SC, Chun C, Choi Y. Effect of thermal sensation on emotional responses as measured through brain waves. Build Environ 2017;118:32–9. https://doi.org/10.1016/j.buildenv.2017.03.023.
- [110] Wang H, Liu L. Experimental investigation about effect of emotion state on people's thermal comfort. Energy Build 2020;211:109789. https://doi.org/ 10.1016/j.enbuild.2020.109789.
- [111] Wang Z, Warren K, Luo M, He X, Zhang H, Arens E, et al. Evaluating the comfort of thermally dynamic wearable devices. Build Environ 2020;167:106443. https:// doi.org/10.1016/j.buildenv.2019.106443.

Renewable and Sustainable Energy Reviews 149 (2021) 111359

- [112] Gao C, Kuklane K, Wang F, Holmér I. Personal cooling with phase change materials to improve thermal comfort from a heat wave perspective. Indoor Air 2012;22:523-30. https://doi.org/10.1111/j.1600-0668.2012.00778.x.
- [113] Jovanovi] DB, Karkali] A RM, Tomi] LD, Veli'kovi] B ZS, Radakovi SS. EFFICACY OF A NOVEL PHASE CHANGE MATERIAL FOR MICROCLIMATE BODY COOLING n.d. https://doi.org/10.2298/TSCI130216129J.
- [114] Luo M, Arens E, Zhang H, Ghahramani A, Wang Z. Thermal comfort evaluated for combinations of energy-efficient personal heating and cooling devices. Build Environ 2018;143:206–16. https://doi.org/10.1016/j.buildenv.2018.07.008.
- [115] Pasut W, Zhang H, Arens E, Kaam S, Zhai Y. Effect of a heated and cooled office chair on thermal comfort. HVAC R Res 2013;19:574–83. https://doi.org/ 10.1080/10789660.2013.781371.
- [116] Zhang YF, Wyon DP, Fang L, Melikov AK. The influence of heated or cooled seats on the acceptable ambient temperature range. Ergonomics 2007;50:586–600. https://doi.org/10.1080/00140130601154921.
- [117] Du C, Liu H, Li C, Xiong J, Li B, Li G, et al. Demand and efficiency evaluations of local convective heating to human feet and low body parts in cold environments. Build Environ 2020;171:106662. https://doi.org/10.1016/j. buildenv.2020.106662.
- [118] Oi H, Yanagi K, Tabat K, Tochihar Y. Effects of heated seat and foot heater on thermal comfort and heater energy consumption in vehicle. Ergonomics 2011;54: 690-9. https://doi.org/10.1080/00140139.2011.595513.
- [119] Wang Y, Hwang RL, Lian Z. A study on the parameter ranges of the locally supplied air in a task ambient conditioning system with chest exposure. Sci Technol Built Environ 2018;24:238–47. https://doi.org/10.1080/ 23744731.2017.1373701.
- [120] Brooks JE, Parsons KC. An ergonomics investigation into human thermal comfort using an automobile seat heated with encapsulated carbonized fabric (ECF). Ergonomics 1999;42:661-73. https://doi.org/10.1080/001401399185379.
- [121] Irshad K, Habib K, Kareem MW, Basrawi F, Saha BB. Evaluation of thermal comfort in a test room equipped with a photovoltaic assisted thermo-electric air duct cooling system. Int J Hydrogen Energy 2017;42:26956–72. https://doi.org/ 10.1016/j.ijhydene.2017.05.247.
- [122] Schiavon S, Rim D, Pasut W, Nazaroff WW. Sensation of draft at uncovered ankles for women exposed to displacement ventilation and underfloor air distribution systems. Build Environ 2016;96:228–36. https://doi.org/10.1016/j. buildenv.2015.11.009.
- [123] Fong ML, Hanby V, Greenough R, Lin Z, Cheng Y. Acceptance of thermal conditions and energy use of three ventilation strategies with six exhaust configurations for the classroom. Build Environ 2015;94:606–19. https://doi.org/ 10.1016/j.buildenv.2015.10.024.
- [124] Sun W, Cheong KWD, Melikov AK. Subjective study of thermal acceptability of novel enhanced displacement ventilation system and implication of occupants' personal control. Build Environ 2012;57:49–57. https://doi.org/10.1016/j. buildenv.2012.04.004.
- [125] Kawahara Y, Emura K, Nabeshima M, Bougaki K, Kadoya M. Air-conditioning with underfloor air supply: comfort in non-uniform thermal environment. Build Serv Eng Technol 1999;20:1–7. https://doi.org/10.1177/014362449902000101.
- [126] Ali AHH, Morsy MG. Energy efficiency and indoor thermal perception: a comparative study between radiant panel and portable convective heaters. Energy Effic 2010;3:283-301. https://doi.org/10.1007/s12053-010-09077-3.
- [127] Jin Q, Simone A, Olesen BW, Holmberg SKM, Bourdakis E. Laboratory study of subjective perceptions to low temperature heating systems with exhaust ventilation in Nordic countries. Sci Technol Built Environ 2017;23:457–68. https://doi.org/10.1080/23744731.2017.1251266.
- [128] Hodder SG, Loveday DL, Parsons KC, Taki AH. Thermal comfort in chilled ceiling and displacement ventilation environments: vertical radiant temperature asymmetry effects. Energy Build 1998;27:167–73. https://doi.org/10.1016/ s0378-7788(97)00038-8.
- [129] Parsons K. Displacement ventilation environments with chilled ceilings: thermal comfort design within the context of the BS EN ISO7730 versus adaptive debate. Energy Build n.d.
- [130] Pasut W, Arens E, Zhang H, Zhai Y. Enabling energy-efficient approaches to thermal comfort using room air motion. Build Environ 2014;79:13–9. https://doi. org/10.1016/j.buildenv.2014.04.024.
- [131] Guo Y, Chan AP, Wong FK, Li Y, Sun S, Han X. Developing a hybrid cooling vest for combating heat stress in the construction industry. Textil Res J 2019;89: 254-69. https://doi.org/10.1177/0040517517743685.
- [132] Chan AP, Yang Y, Guo Y, Yam MC, Song W. Evaluating the physiological and perceptual responses of wearing a newly designed construction work uniform. Textil Res J 2016;86:659-73. https://doi.org/10.1177/0040517515591773.
- [133] Pang TY, Subic A, Takla M. Evaluation of thermal and evaporative resistances in cricket helmets using a sweating manikin. Appl Ergon 2014;45:300–7. https:// doi.org/10.1016/j.apergo.2013.04.011.
- [134] Brühwiler PA. Role of the visor in forced convective heat loss with bicycle helmets. Int J Ind Ergon 2009;39:255–9. https://doi.org/10.1016/j. ergon.2008.08.001.
- [135] Davis JK, Matt Laurent C, Allen KE, Zhang Y, Stolworthy NI, Welch TR, et al. Influence of clothing on thermoregulation and comfort during exercise in the heat. J Strength Condit Res 2017;31:3435–43. https://doi.org/10.1519/ JSC.00000000001754.
- [136] Branson DH, Dejonge JO, Munson D. Thermal response associated with prototype pesticide protective clothing. Textil Res J 1986;56:27–34. https://doi.org/ 10.1177/004051758605600104.
- [137] Hayashi C, Tokura H. Improvement of thermophysiological stress in participants wearing protective clothing for spraying pesticide, and its application in the field.

Int Arch Occup Environ Health 2000;73:187–94. https://doi.org/10.1007/ s004200050026.

- [138] Ma N, Lu Y, Xu F, Dai H. Development and performance assessment of electrically heated gloves with smart temperature control function. Int J Occup Saf Ergon 2020;26:46–54. https://doi.org/10.1080/10805348.2018.1457886.
- [139] Song W, Wang F, Zhang C, Lai D. On the improvement of thermal comfort of university students by using electrically and chemically heated clothing in a cold classroom environment. Build Environ 2015;94:704–13. https://doi.org/ 10.1016/j.buildenv.2015.10.017.
- [140] Chong D, Zhu N, Luo W, Pan X. Human thermal risk prediction in indoor hyperthermal environments based on random forest. Sustain Cities Soc 2019;49: 101595. https://doi.org/10.1016/j.scs.2019.101595.
- [141] Chong D, Zhu N, Zheng G. Developing a continuous graphical index to assess heat strain in extremely hot environments. Build Environ 2018;138:283–92. https:// doi.org/10.1016/j.buildenv.2018.05.009.
- [142] Lee K, Choi H, Choi JH, Kim T. Development of a data-driven predictive model of clothing thermal insulation estimation by using advanced computational approaches. Sustainability 2019;11:5702. https://doi.org/10.3390/sul1205702.
- [143] Yeom D, Choi JH, Kang SH. Investigation of the physiological differences in the immersive virtual reality environment and real indoor environment: focused on skin temperature and thermal sensation. Build Environ 2019;154:44–54. https:// doi.org/10.1016/j.buildenv.2019.03.013.
- [144] Ibrahim A, Ali H, Zghoul A, Jaradat S. Mood state and human evaluation of the thermal environment using virtual settings. Indoor Built Environ 2019. https:// doi.org/10.1177/1420326X19880325. 1420326X1988032.
- [145] Orehounig K, Pont U, Mahdavi A. Short-term occupancy implications of digitally provided outside views in window-less rooms. In: International conference on architecture and urban design; 2012.
- [146] Yang H, Cao B, Zhu Y. Study on the effects of chair heating in cold indoor environments from the perspective of local thermal sensation. Energy Build 2018; 180:16–28. https://doi.org/10.1016/j.enbuild.2018.09.003.
- [147] Jeon JY, Hong JY, Kim SM, Lee PJ. Classification of heavy-weight floor impact sounds in multi-dwelling houses using an equal-appearing interval scale. Build Environ 2015;94:821–8. https://doi.org/10.1016/j.buildenv.2015.07.021.
- [148] Zhang D, Tenpierik M, Bluyssen PM. Interaction effect of background sound type and sound pressure level on children of primary schools in The Netherlands. Appl Acoust 2019;154:161–9. https://doi.org/10.1016/j.apaccoust.2019.05.007.
- [149] Weisser A, Buchholz JM, Keidser G. Complex acoustic environments: review, framework, and subjective model. Trends Hear 2019;23. https://doi.org/ 10.1177/2331216519881346. 233121651988134.
- [150] Haka M, Haapakangas A, Keränen J, Hakala J, Keskinen E, Hongisto V. Performance effects and subjective disturbance of speech in acoustically different office types - a laboratory experiment. Indoor Air 2009;19:454–67. https://doi. org/10.1111/j.1600-0668.2009.00608.x.
- [151] Kang S, Ou D. The effects of speech intelligibility on work performance in Chinese open-plan offices: a laboratory study. Acta Acust United with Acust 2019;105: 9-13. https://doi.org/10.3813/AAA.919281.
- [152] Hongisto V, Oliva D, Rekola L. Subjective and objective rating of spectrally different pseudorandom noises—implications for speech masking design. J Acoust Soc Am 2015;137:1344–55. https://doi.org/10.1121/1.4913273.
- [153] Abbasi AM, Motamedzade M, Aliabadi M, Golmohammadi R, Tapak L. Study of the physiological and mental health effects caused by exposure to low-frequency noise in a simulated control room. Build Acoust 2018;25:233–48. https://doi.org/ 10.1177/1351010X18779518.
- [154] Jeon JY, Ryu JK, Lee PJ. A quantification model of overall dissatisfaction with indoor noise environment in residential buildings. Appl Acoust 2010;71:914–21. https://doi.org/10.1016/j.apacoust.2010.06.001.
- [155] Yu B, Kang J, Ma H. Development of indicators for the soundscape in urban shopping streets. Acta Acust United with Acust 2016;102:462–73. https://doi. org/10.3813/AAA.918965.
- [156] Hermida L, Pavón I. Spatial aspects in urban soundscapes: binaural parameters application in the study of soundscapes from Bogotá-Colombia and Brasília-Brazil. Appl Acoust 2019;145:420–30. https://doi.org/10.1016/j. apacoust.2018.10.011.
- [157] Hong JY, Ong ZT, Lam B, Ooi K, Gan WS, Kang J, et al. Effects of adding natural sounds to urban noises on the perceived loudness of noise and soundscape quality. Sci Total Environ 2020;711:134571. https://doi.org/10.1016/j. sciintenv.2019.134571.
- [158] Ne'Eman E, Hopkinson RG. Critical minimum acceptable window size: a study of window design and provision of a view. Light Res Technol 1970;2:17–27. https:// doi.org/10.1177/14771535700020010701.
- [159] Arsenault H, Hébert M, Dubois MC. Effects of glazing colour type on perception of daylight quality, arousal, and switch-on patterns of electric light in office rooms. Build Environ 2012;56:223–31. https://doi.org/10.1016/j. buildenv.2012.02.032.
- [160] Dubois M-C, Cantin F, Johnsen K. The effect of coated glazing on visual perception: a pilot study using scaleamodels. Light Res Technol 2007;39: 283–304. https://doi.org/10.1177/1477153507080074.
- [161] Huang W, Yang Y, Luo MR. Discomfort glare caused by white LEDs having different spectral power distributions. Light Res Technol 2018;50:921–36. https://doi.org/10.1177/1477153517704996.
- [162] Kimura-Minoda T, Ayama M. Evaluation of discomfort glare from color leds and its correlation with individual variations in brightness sensitivity. Color Res Appl 2011;36:286–94. https://doi.org/10.1002/col.20630.
- [163] Osterhaus WKE, Bailey IL. Large area glare sources and their effect on visual discomfort and visual performance at computer workstations. In: Conf. Rec. - IAS

annu. Meet. (IEEE ind. Appl. Soc., vol. 1992- janua, institute of electrical and electronics engineers inc.; 1992. p. 1825-9. https://doi.org/10.1109/ 188.1992.244537.

- [164] Á Logadóttir, Christoffersen J, Fotios S. Investigating the use of an adjustment task to set the preferred illuminance in a workplace environment. Light Res Technol 2011;43:403–22. https://doi.org/10.1177/1477153511400971.
- [165] Chamilothori K, Chinazzo G, Rodrigues J, Dan-Glauser ES, Wienold J, Andersen M. Subjective and physiological responses to façade and sunlight pattern geometry in virtual reality. Build Environ 2019;150:144–55. https://doi. org/10.1016/i.buildenv.2019.01.009.
- [166] Moscoso C, Chamilothori K, Wienold J, Andersen M, Matusiak B. Window size effects on subjective impressions of daylit spaces: indoor studies at high latitudes using virtual reality. Leukos 2020:1–23. https://doi.org/10.1080/ 15502724.2020.1726183.
- [167] Fotios SA. Lamp colour properties and apparent brightness: a review. Light Res Technol 2001;33:163–78. https://doi.org/10.1177/136578280103300306.
- [168] Wei M, Houser KW, Allen GR, Beers WW. Color preference under LEDs with diminished yellow emission. LEUKOS - J Illum Eng Soc North Am 2014;10: 119–31. https://doi.org/10.1080/15502724.2013.865212.
- [169] Hu X, Houser KW, Tiller DK. Higher color temperature lamps may not appear brighter. LEUKOS - J Illum Eng Soc North Am 2006;3:69–81. https://doi.org/ 10.1582/LEUKOS.2006.03.01.004.
- [170] Royer M, Wilkerson A, Wei M, Houser K, Davis R. Human perceptions of colour rendition vary with average fidelity, average gamut, and gamut shape. Light Res Technol 2017;49:966–91. https://doi.org/10.1177/1477153516663615.
- [171] Velds M. User acceptance studies to evaluate discomfort glare in daylit rooms. Sol Energy 2002;73:95–103. https://doi.org/10.1016/S0038-092X(02)00037-3.
- [172] Kent MG, Fotios S, Altomonte S. An experimental study on the effect of visual tasks on discomfort due to peripheral glare. Leukos 2019;15:17–28. https://doi. org/10.1080/15502724.2018.1489282.
- [173] Altomonte S, Kent MG, Tregenza PR, Wilson R. Visual task difficulty and temporal influences in glare response. Build Environ 2016;95:209–26. https://doi.org/ 10.1016/j.buildenv.2015.09.021.
- [174] Boubekri M, Boyer LL. Effect of window size and sunlight presence on glare. Light Res Technol 1992;24:69–74. https://doi.org/10.1177/096032719202400203.
- [175] Geerdinck LM, Van Gheluwe JR, Vissenberg MCJM. Discomfort glare perception of non-uniform light sources in an office setting. J Environ Psychol 2014;39:5–13. https://doi.org/10.1016/j.jenyp.2014.04.002.
- [176] Kent M, Altomonte S, Tregenza P, Wilson R. Discomfort glare and time of day. Light Res Technol 2015;47:641-57. https://doi.org/10.1177/ 1477153514547221.
- [177] Shin JY, Yun GY, Kim JT. View types and luminance effects on discomfort glare assessment from windows. Energy Build 2012;46:139–45. https://doi.org/ 10.1016/j.enbuild.2011.10.036. Elsevier.
- [178] Van Den Wymelenberg K, Inanici M, Johnson P. The effect of luminance distribution patterns on occupant preference in a daylit office environment. LEUKOS - J Illum Eng Soc North Am 2010;7:103–22. https://doi.org/10.1582/ LEUKOS.2010.07.02003.
- [179] Krüger EL, Tamura C, Trento TW. Identifying relationships between daylight variables and human preferences in a climate chamber. Sci Total Environ 2018; 642:1292–302. https://doi.org/10.1016/j.scitotenv.2018.06.164.
- [180] Boyce PR, Hunter CM, Inclan C. Overhead glare and visual discomfort. J Illum Eng Soc 2003;32:73–88. https://doi.org/10.1080/00994480.2003.10748406.
- [181] Taniguchi T, Iwata T, Sato M, Honma M. Proposal OF a method for determining slat-rotation speed OF automated blinds to reduce discomfort caused BV the change IN window brightness. J Environ Eng (Transactions AIJ) 2017;82:551–9. https://doi.org/10.3130/aije.82.551.
- [182] Tuaycharoen N, Tregenza PR. View and discomfort glare from windows. Light Res Technol 2007;39:185–200. https://doi.org/10.1177/1365782807077193.
- [183] Bian Y, Ma Y. Subjective survey & simulation analysis of time-based visual comfort in daylit spaces. Build Environ 2018;131:63–73. https://doi.org/ 10.1016/j.buildenv.2018.01.007.
- [184] Bian Y, Luo T. Investigation of visual comfort metrics from subjective responses in China: a study in offices with daylight. Build Environ 2017;123:661–71. https:// doi.org/10.1016/j.buildeny.2017.07.035.
- [185] Kim W, Kim Jeong Tai. A position index formula for evaluation of glare source in the visual field. Indoor Buil: Environ 2011;20:47–53. https://doi.org/10.1177/ 1420326X10389267.
- [186] Suk JY, Schiler M, Kensek K. Investigation of existing discomfort glare indices using human subject study data. Build Environ 2017;113:121–30. https://doi. org/10.1016/j.buildenv.2016.09.018.
- [187] Yamin Garretón JA, Rodriguez RG, Ruiz A, Pattini AE. Degree of eye opening: a new discomfort glare indicator. Build Environ 2015;88:142–50. https://doi.org/ 10.1016/j.buildenv.2014.11.010.
- [188] Karlsen L, Heiselberg P, Bryn I, Johra H. Verification of simple illuminance based measures for indication of discomfort glare from windows. Build Environ 2015; 92:615–26. https://doi.org/10.1016/j.buildenv.2015.05.040.
- [189] Van Den Wymelenberg K, Inanici M. A critical investigation of common lighting design metrics for predicting human visual comfort in offices with daylight. LEUKOS - J Illum Eng Soc North Am 2014;10:145–64. https://doi.org/10.1080/ 15502724.2014.881720.
- [190] Van Den Wymelenberg K, Inanici M. Evaluating a new suite of luminance-based design metrics for predicting human visual comfort in offices with daylight. LEUKOS - J Illum Eng Soc North Am 2016;12:113–38. https://doi.org/10.1080/ 15502724.2015.1062392.

- [191] Bian Y, Dai Q, Ma Y, Liu L. Variable set points of glare control strategy for side-lit spaces: daylight glare tolerance by time of day. Sol Energy 2020;201:268–78. https://doi.org/10.1016/j.solener.2020.03.016.
- [192] Kim JT, Shin JY, Yun GY. Prediction of discomfort glares from windows: influence of the subjective evaluation of window views. Indoor Built Environ 2012;21:92–7. https://doi.org/10.1177/1420326X11423157.
- [193] Suk JY, Schiler M, Kensek K. Absolute glare factor and relative glare factor based metric: predicting and quantifying levels of daylight glare in office space. Energy Build 2016;13:0:8-19. https://doi.org/10.1016/j.enbuild.2016.08.021.
- [194] Yamín Garretón JA, Rodriguez RG, Pattini AE. Glare indicators: an analysis of ocular behaviour in an office equipped with Venetian blinds. Indoor Built Environ 2016;25:69-80. https://doi.org/10.1177/1420326X14538082.
- [195] Rodriguez R, Yamín Garretón J, Pattini A. Glare and cognitive performance in screen work in the presence of sunlight. Light Res Technol 2016;48:221–38. https://doi.org/10.1177/1477153515577851.
- [196] Hamedani Z, Šolgi E, Hine T, Skates H. Revealing the relationships between luminous environment characteristics and physiological, ocular and performance measures: an experimental study. Build Environ 2020;172:106702. https://doi. org/10.1016/j.buildenv.2020.106702.
- [197] Choi J-H. Investigation of human eye pupil sizes as a measure of visual sensation in the workplace environment with a high lighting colour temperature. Indoor Built Environ 2017;26:488–501. https://doi.org/10.1177/142032615626585.
- [198] Choi JH, Zhu R. Investigation of the potential use of human eye pupil sizes to estimate visual sensations in the workplace environment. Build Environ 2015;88: 73–81. https://doi.org/10.1016/j.buildenv.2014.11.025.
- [199] Sun C, Lian Z, Lan L. Work performance in relation to lighting environment in office buildings. Indoor Built Environ 2019;28:1064–82. https://doi.org/ 10.1177/1420326X18820089.
- [200] Smolders KCHJ, de Kort YAW. Investigating daytime effects of correlated colour temperature on experiences, performance, and arousal. J Environ Psychol 2017; 50:80–93. https://doi.org/10.1016/j.jenvp.2017.02.001.
- [201] Huiberts LM, Smolders KCHJ, de Kort YAW. Non-image forming effects of illuminance level: exploring parallel effects on physiological arousal and task performance. Physiol Behav 2016;164:129–39. https://doi.org/10.1016/j. physbeh.2016.05.035.
- [202] Huiberts LM, Smolders KCHJ, De Kort YAW. Seasonal and time-of-day variations in acute non-image forming effects of illuminance level on performance, physiology, and subjective well-being. Chronobiol Int 2017;34:827–44. https:// doi.org/10.1080/07420528.2017.1324471.
- [203] de Vries A, Souman JL, de Ruyter B, Heynderickx I, de Kort YAW. Lighting up the office: the effect of wall luminance on room appraisal, office workers' performance, and subjective alertness. Build Environ 2018;142:534–43. https:// doi.org/10.1016/j.buildenv.2018.06.046.
- [204] Liang R, Kent M, Wilson R, Wu Y. Development of experimental methods for quantifying the human response to chromatic glazing. Build Environ 2019;147: 199–210. https://doi.org/10.1016/j.buildenv.2018.09.044.
- [205] Newsham GR, Mancini S, Marchand RG. Detection and acceptance of demandresponsive lighting in offices with and without daylight. LEUKOS - J Illum Eng Soc North Am 2008;4:139–56. https://doi.org/10.1582/ LEUKOS.2007.004.03.001.
- [206] Newsham G, Arsenault C, Veitch J, Tosco AM, Duval C. Task lighting effects on office worker satisfaction and performance, and energy efficiency. LEUKOS - J Illum Eng Soc North Am 2005;1:7–26. https://doi.org/10.1582/ IJEIIKOS.01.04.001.
- [207] Chrabi S, Crommentuijn L, Loenen E van, Rosemann A. Influence of wall luminance and uniformity on preferred task illuminance. Build Environ 2017;117: 24–35. https://doi.org/10.1016/j.buildenv.2017.02.026.
- [208] Loe L, Mansfield KP, Rowlands E. Appearance of lit environment and its relevance in lighting design: experimental study. Light Res Technol 1994;26:119–33. https://doi.org/10.1177/96032719402600301.
- [209] Newsham G, Veitch J. Lighting quality recommendations for VDT offices: a new method of derivation. Light Res Technol 2001;33:97–113. https://doi.org/ 10.1177/136578280103300205.
- [210] Ru T, de Kort YAW, Smolders KCHJ, Chen Q, Zhou G. Non-image forming effects of illuminance and correlated color temperature of office light on alertness, mood, and performance across cognitive domains. Build Environ 2019;149:253–63. https://doi.org/10.1016/j.buildenv.2018.12.002.
- [211] Smolders KCHJ, de Kort YAW, Cluitmans PJM. A higher illuminance induces alertness even during office hours: findings on subjective measures, task performance and heart rate measures. Physiol Behav 2012;107:7–16. https://doi. org/10.1016/j.physbeh.2012.04.028.
- [212] Smolders KCHJ, de Kort YAW. Bright light and mental fatigue: effects on alertness, vitality, performance and physiological arousal. J Environ Psychol 2014;39:77–91. https://doi.org/10.1016/j.jenvp.2013.12.010.
- [213] Kompier ME, Smolders KCHJ, van Marken Lichtenbelt WD, de Kort YAW. Effects of light transitions on measures of alertness, arousal and comfort. Physiol Behav 2020;223:112099. https://doi.org/10.1016/j.inpsyche.2020.112909.
- 2020;223:112999. https://doi.org/10.1016/j.physbeh.2020.112999.
 [214] Hellinga H. Daylight and view the influence of windows on the visula quality of indoor spaces [Phb thesis]. 2013. https://doi.org/10.4233/uuid:2daeb534-9572-4c85-bf8f-308f3f6825fd.
- [215] Shamsul BM, Sia CC, Ng Y, Karmegan K. Effects of light's colour temperatures on visual comfort level, task performances, and alertness among students. Am J Publ Health Res 2013;1:159–65. https://doi.org/10.12691/ajbn-1.7-3.
- [216] Ortiz MA, Zhang D, Bluyssen PM. Table top surface appraisal by school children under different lighting conditions tested in the Senselab. E3S Web Conf. 2019; 111. https://doi.org/10.1051/c3sconf/201911102040. EDP Sciences.

- [217] Tonello G, Hernández de Borsetti N, Borsetti H, Tereschuk L, López Zigarán S. Perceived well-being and light-reactive hormones: an exploratory study. Light Res Technol 2019;51:184–205. https://doi.org/10.1177/147715517750714.
- [218] Gou Z, Lau SS-Y, Qian F. Comparison of mood and task performance in naturallylit and artificially-lit environments. Indoor Built Environ 2015;24:27–36. https:// doi.org/10.1177/1420326X13207792.
- [219] Cajochen C, Frey S, Anders D, Späti J, Bues M, Pross A, et al. Evening exposure to a light-emitting diodes (LED)-backlit computer screen affects circadian physiology and cognitive performance. J Appl Physiol 2011;110:1432–8. https:// doi.org/10.1152/japplohysiol.00165.2011.
- [220] Sarey Khanie M, Stoll J, Einhäuser W, Wienold J, Andersen M. Gaze and discomfort glare, Part I: development of a gaze-driven photometry. Light Res Technol 2017;49:845–65. https://doi.org/10.1177/1477153516649016.
- [221] Konstantzos I, Tzempelikos A. Daylight glare evaluation with the sun in the field of view through window shades. Build Environ 2017;113:65–77. https://doi.org/ 10.1016/j.buildenv.2016.09.009.
- [222] Linhart F, Scartezzini JL. Evening office lighting visual comfort vs. energy efficiency vs. performance? Build Environ 2011;46:981–9. https://doi.org/ 10.1016/j.buildenv.2010.10.002.
- [223] Suk JY, Luminance and vertical eye illuminance thresholds for occupants' visual comfort in daylit office environments. Build Environ 2019;148:107–15. https:// doi.org/10.1016/j.buildeny.2018.10.058.
- [224] Borisuit A, Linhart F, Scartezzini J-L, Mü M. Effects of realistic office daylighting and electric lighting conditions on visual comfort, alertness and mood n.d. https://doi.org/10.1177/1477153514531158.
- [225] Boubekri M, Hull RB, Boyer LL. Impact of window size and sunlight penetration on office workers' mood and satisfaction. Environ Behav 1991;23:474–93. https://doi.org/10.1177/0013916591234004.
- [226] Licina D, Melikov A, Sekhar C, Tham KW. Human convective boundary layer and its interaction with room ventilation flow. Indoor Air 2015;25:21–35. https://doi. org/10.1111/ina.12120.
- [227] Licina D, Pantelic J, Melikov A, Sekhar C, Tham KW. Experimental investigation of the human convective boundary layer in a quiescent indoor environment. Build Environ 2014;75:79–91. https://doi.org/10.1016/j.buildenv.2014.01.016.
- [228] Licina D, Melikov A, Sekhar C, Tham KW. Air temperature investigation in microenvironment around ahumanbody. Build Environ 2015;92:39–47. https:// doi.org/10.1016/j.buildenv.2015.04.014.
- [229] Gunnarsen L, Ole Fanger P. Adaptation to indoor air pollution. Environ Int 1992; 18:43–54. https://doi.org/10.1016/0160-4120(92)90209-M.
- [230] Johnsen CR, Heinig JH, Schmidt K, Albrechtsen O, Nielsen PA, Wolkoff P, et al. A study of human reactions to emissions from building materials in climate chambers. Part I: clinical data, performance and comfort. Indoor Air 1991;1: 377–88. https://doi.org/10.1111/j.1600-0668.1991.00002.x.
- [231] Amano K, Takahashi H, Kato S, Tani H, Ishiguro T, Higuchi M, et al. Indoor odor exposure effects on psycho-physiological states during intellectual tasks and rest. HVAC R Res 2012;18:217–24. https://doi.org/10.1080/10789669.2011.599765.
- [232] Melikov AK, Kaczmarczyk J. Air movement and perceived air quality. Build Environ 2012;47:400–9. https://doi.org/10.1016/j.buildeny.2011.06.017.
- [233] Allen JG, MacNaughton P, Satish U, Santanam S, Vallarino J, Spengler JD. Associations of cognitive function scores with carbon dioxide, ventilation, and volatile organic compound exposures in office workers: a controlled exposure study of green and conventional office environments. Environ Health Perspect 2016;124:805–12. https://doi.org/10.1289/ehp.1510037.
- [234] Wargocki P. Perceived air quality, sick building syndrome (SBS) symptoms and productivity in an office with two different pollution loads. Indoor Air 1999;9: 165–79. https://doi.org/10.1111/j.1600-0668.1999.101-1.00003.x
- [235] Armijos Moya T, Bluyssen PM. Appraisal and identification of different sources of smell by primary school children in the air quality test chamber of the SenseLab. Intell Build Int 2019;1–14. https://doi.org/10.1080/17508975.2019.1682493.
- [236] Mølhave L, Kjaergaard SK, Attermann J. Respiratory effects of experimental exposure to office dust. Indoor Air 2004;14:376–84. https://doi.org/10.1111/ j.160-0668.2004.00252.x
- [237] Pantelic J, Tham KW, Licina D. Effectiveness of a personalized ventilation system in reducing personal exposure against directly released simulated cough droplets. Indoor Air 2015;25:683–93. https://doi.org/10.1111/ina.12187.
- [238] Licina D, Tian Y, Nazaroff WW. Inhalation intake fraction of particulate matter from localized indoor emissions. Build Environ 2017;123:14–22. https://doi.org/ 10.1016/j.buildenv.2017.06.037.
- [239] Pantelic J, Liu S, Pistore L, Licina D, Vannucci M, Sadrizadeh S, et al. Personal CO2 cloud: laboratory measurements of metabolic CO2 inhalation zone concentration and dispersion in a typical office desk setting. J Expo Sci Environ Epidemiol 2020;30:328–37. https://doi.org/10.1038/s41370-019-0179-5.
- [240] Alotaibi S, Chakroun W, Habchi C, Ghali K, Ghaddar N. Effectiveness of contaminant confinement in office spaces equipped with ceiling personalized ventilation system. Build Simul 2018;11:773–86. https://doi.org/10.1007/ s12273-018-0437-9.
- [241] Bivolarova M, Kierat W, Zavrl E, Popiolek Z, Melikov A. Effect of airflow interaction in the breathing zone on exposure to bio-effluents. Build Environ 2017;152:516–56. https://doi.org/10.1016/j.buildenv.2017.08.043.
- [242] Bolashikov Z, Melikov A, Spilak M, Nastase I, Meslem A. Improved inhaled air quality at reduced ventilation rate by control of airflow interaction at the breathing zone with lobed jets. HVAC R Res 2014;20:238–50. https://doi.org/ 10.1080/10789669.2013.864919.
- [243] Cermak R, Melikov AK, Forejt L, Kovar O. Performance of personalized ventilation in conjunction with mixing and displacement ventilation. HVAC R Res 2006;12:295–311. https://doi.org/10.1080/10789669.2006.10391180.

- [244] Bivolarova MP, Melikov AK, Mizutani C, Kajiwara K, Bolashikov ZD. Bedintegrated local exhaust ventilation system combined with local air cleaning for improved IAQ in hospital patient rooms. Build Environ 2016;100:10–8. https:// doi.org/10.1016/j.buildeny.2016.02.006.
- [245] Verhaart J, Li R, Zeiler W. User interaction patterns of a personal cooling system: a measurement study. Sci Technol Built Environ 2018;24:57–72. https://doi.org/ 10.1080/23744731.2017.1333365.
- [246] Assaad D Al, Ghali K, Ghaddar N, Habchi C. Mixing ventilation coupled with personalized sinusoidal ventilation: optimal frequency and flow rate for acceptable air quality. Energy Build 2017;154:569–80. https://doi.org/10.1016/ j.enbuild.2017.08.090.
- [247] Melikov AK, Cermak R, Majer M. Personalized ventilation: evaluation of different air terminal devices. Energy Build 2002;34:829–36. https://doi.org/10.1016/ S0378-7788(02)00102-0.
- [248] Zhang H, Arens E, Kim DE, Buchberger E, Bauman F, Huizenga C. Comfort, perceived air quality, and work performance in a low-power task-ambient conditioning system. Build Environ 2010;45:29–39. https://doi.org/10.1016/j. buildenv.2009.02.016.
- [249] Li R, Sekhar SC, Melikov AK. Thermal comfort and IAQ assessment of under-floor air distribution system integrated with personalized ventilation in hot and humid climate. Build Environ 2010;45:1906–13. https://doi.org/10.1016/j. buildenv.2010.03.003.
- [250] Faulkner D, Fisk WJ, Sullivan DP, Lee SM. Ventilation efficiencies and thermal comfort results of a desk-edge-mounted task ventilation system. Indoor Air 2004; 14:92–7. https://doi.org/10.1111/j.1600-0668.2004.00295.x.
- [251] Piasecki M, Kostyrko K, Fedorczak-Cisak M, Nowak K. Air enthalpy as an IAQ indicator in hot and humid environment—experimental evaluation. Energies 2020;13:1481. https://doi.org/10.3390/en13061481.
- [252] Brambilla A, Hu W, Samangouei R, Cadorin R, Davis W. How correlated colour temperature manipulates human thermal perception and comfort. Build Environ 2020;177:106929. https://doi.org/10.1016/j.buildenv.2020.106929.
- [253] Chinazzo G, Wienold J, Andersen M. Daylight affects human thermal perception. Sci Rep 2019;9:1–15. https://doi.org/10.1038/s41598-019-48963-y.
- [254] Golasi I, Salata F, Vollaro E de L, Peña-García A. Influence of lighting colour temperature on indoor thermal perception: a strategy to save energy from the HVAC installations. Energy Build 2019;185:112–22. https://doi.org/10.1016/j. enbuild.2018.12.026.
- [255] Yang W, Moon HJ. Combined effects of acoustic, thermal, and illumination conditions on the comfort of discrete senses and overall indoor environment. Build Environ 2019;148:623–33. https://doi.org/10.1016/j. buildenv.2018.11.040.
- [256] Yang W, Moon HJ. Cross-modal effects of noise and thermal conditions on indoor environmental perception and speech recognition. Appl Acoust 2018;141:1–8. https://doi.org/10.1016/j.apacoust.2018.06.019.
- [257] Yang W, Kim M-J, Moon HJ. Effects of indoor temperature and background noise on floor impact noise perception. Indoor Built Environ 2019;28:454–69. https:// doi.org/10.1177/1420326X17753708.
- [258] te Kulve M, Schlangen LJM, Schellen L, Frijns AJH, van Marken Lichtenbelt WD. The impact of morning light intensity and environmental temperature on body temperatures and alertness. Physiol Behav 2017;175:72–81. https://doi.org/ 10.1016/j.physbeh.2017.03.043.
- [259] te Kulve M, Schlangen L, van Marken Lichtenbelt W. Interactions between the perception of light and temperature. Indoor Air 2018;28:881–91. https://doi.org/ 10.1111/ina.12500.
- [260] Bjerre L, Larsen T, Sørensen Aj, Santurette S, Jeong C-H. On-site and laboratory evaluations of soundscape quality in recreational nurban spaces. Noise Health 2017;19:183. https://doi.org/10.4103/nah.NAH_109_16.
- [261] Huebner GM, Shipworth DT, Gauthier S, Witzel C, Raynham P, Chan W. Saving energy with light? Experimental studies assessing the impact of colour temperature on thermal comfort. Energy Res Soc Sci 2016;15:45–57. https://doi. org/10.1016/j.erss.2016.02.008.
- [262] Hong JY, Jeon JY. Designing sound and visual components for enhancement of urban soundscapes. J Acoust Soc Am 2013;134:2026–36. https://doi.org/ 10.1121/j.4817924.
- [263] Pellerin N, Candas V. Combined effects of temperature and noise on human disconfort. Physiol Behav 2003;78:99–106. https://doi.org/10.1016/S0031-9384(02)00956-3.
- [264] Greene TC, Bell PA. Additional considerations concerning the effects of warm and cool wall colours on energy conservation. Ergonomics 1980. https://doi.org/ 10.1080/00140138008924804.
- [265] Clausen G, Carrick L, Fanger PO, Kim SW, Poulsen T, Rindel JH. A comparative study of discomfort caused by indoor air pollution, thermal load and noisec. Indoor Air 1993;2:255–62. https://doi.org/10.1111/j.1600-0668.1993.00006.x.
- [266] Fanger PO, Breum NO, Jerking E. Can colour and noise influence man's thermal comfort? Ergonomics 1977;20:11–8. https://doi.org/10.1080/ 00140137708931596.
- [267] Chinazzo G, Wienold J, Andersen M. Effect of indoor temperature and glazing with saturated color on visual perception of daylight. LEUKOS - J Illum Eng Soc North Am 2020. https://doi.org/10.1080/15502724.2020.1726182.
- [268] Chinazzo G, Wienold J, Andersen M. Influence of indoor temperature and daylight illuminance on visual perception. Light Res Technol 2020;52:350–70. https://doi.org/10.1177/1477153519859609.
- [269] Jing S, Li B, Tan M, Liu H. Impact of relative humidity on thermal comfort in a warm environment. Indoor Built Environ 2013;22:598–607. https://doi.org/ 10.1177/1420326X12447614.

- [270] Sunwoo Y, Chou C, Takeshita J, Murakami M, Tochihara Y. Physiological and subjective responses to low relative humidity. J Physiol Anthropol 2006;25:7–14. https://doi.org/10.2114/jpa2.25.7.
- [271] Schweiker M, Wagner A. The effect of occupancy on perceived control, neutral temperature, and behavioral patterns. Energy Build 2016;117:246–59. https:// doi.org/10.1016/j.enbuild.2015.10.051.
- [272] Chen Y, Raphael B, Sekhar SC. Individual control of a personalized ventilation system integrated with an ambient mixing ventilation system. HVAC R Res 2012; 18:1136–52. https://doi.org/10.1080/10789669.2012.710059.
- [273] Toftum J, Kolarik J, Belkowska D, Olesen BW. Influence on occupant responses of behavioral modification of clothing insulation in nonsteady thermal environments (RP-1269). HVAC R Res 2010;16:59–74. https://doi.org/10.1080/ 10789669.2010.10390892.
- [274] Choi Y, Kim M, Chun C. Measurement of occupants' stress based on electroencephalograms (EEG) in twelve combined environments. Build Environ 2015;88:65–72. https://doi.org/10.1016/j.buildenv.2014.10.003.
- [275] Kolarik J, Toftum J, Olesen BW, Shitzer A. Occupant responses and office work performance in environments with moderately drifting operative temperatures (RP-1269). HVAC R Res 2009;15:931–60. https://doi.org/10.1080/ 10789669.2009.10390873.
- [276] Almesri I, Awbi HB, Foda E, Sirén K. An air distribution index for assessing the thermal confort and air quality in uniform and nonuniform thermal environments. Indoor Built Environ 2013;22:618–39. https://doi.org/10.1177/ 1420326X12451186.
- [277] Choi JH, Loftness V. Investigation of human body skin temperatures as a biosignal to indicate overall thermal sensations. Build Environ 2012;58:258–69. https://doi.org/10.1016/j.buildenv.2012.07.003.
- [278] Hammoud M, Ghali K, Ghaddar N. The optimized operation of a solar hybrid desiccant/displacement ventilation combined with a personalized evaporative cooler. Int J Green Energy 2014;11:141–60. https://doi.org/10.1080/ 15435075.2013.771360.
- [279] Du C, Li B, Liu H, Ji Y, Yao R, Yu W. Quantification of personal thermal comfort with localized airflow system based on sensitivity analysis and classification tree model. Energy Build 2019;194:1–11. https://doi.org/10.1016/j. enbuild.2019.04.010.
- [280] Veselý M, Molenaar P, Vos M, Li R, Zeiler W. Personalized heating comparison of heaters and control modes. Build Environ 2017;112:223–32. https://doi.org/ 10.1016/j.buildenv.2016.11.036.
- [281] Pasut W, Zhang H, Arens E, Zhai Y. Energy-efficient comfort with a heated/cooled chair: results from human subject tests. Build Environ 2015;84:10–21. https:// doi.org/10.1016/j.buildenv.2014.10.026.
- [282] Watanabe S, Shimomura T, Miyazaki H. Thermal evaluation of a chair with fans as an individually controlled system. Build Environ 2009;44:1392–8. https://doi. org/10.1016/j.buildenv.2008.05.016.
- [283] Katić K, Li R, Verhaart J, Zeiler W. Neural network based predictive control of personalized heating systems. Energy Build 2018;174:199–213. https://doi.org/ 10.1016/j.enbuild.2018.06.033.
- [284] Kong M, Zhang J, Dang TQ, Hedge A, Teng T, Carter B, et al. Micro-environmental control for efficient local cooling: results from manikin and human participant tests. Build Environ 2019;160:106198. https://doi.org/10.1016/j. buildenv.2019.106198.
- [285] Chou C, Tochihara Y, Kim T. Physiological and subjective responses to cooling devices on firefighting protective clothing. Eur J Appl Physiol 2008;104:369–74. https://doi.org/10.1007/s00421-007-0665-7. Springer.
- [286] Song W, Wang F, Wei F. Hybrid cooling clothing to improve thermal comfort of office workers in a hot indoor environment. Build Environ 2016;100:92–101. https://doi.org/10.1016/j.buildenv.2016.02.009.
- [287] Boerstra A. Personal control in future thermal comfort standards. Adapt To Chang New Think 2010:9–11.
- [288] Ba M, Kang J. A laboratory study of the sound-odour interaction in urban environments. Build Environ 2019;147:314–26. https://doi.org/10.1016/j. buildenv.2018.10.019.
- [289] Zhang F, Haddad S, Nakisa B, Rastgoo MN, Candido C, Tjondronegoro D, et al. The effects of higher temperature setpoints during summer on office workers' cognitive load and thermal comfort. Build Environ 2017;123:176–88. https://doi. org/10.1016/j.buildenv.2017.06.048.
- [290] Choi JH, Yeom D. Investigation of the relationships between thermal sensations of local body areas and the whole body in an indoor built environment. Energy Build 2017;149:204–15. https://doi.org/10.1016/j.enbuild.2017.05.062.
- [291] Zhai Y, Zhang H, Zhang Y, Pasut W, Arens F, Meng Q. Comfort under personally controlled air movement in warm and humid environments. Build Environ 2013; 65:109–17. https://doi.org/10.1016/j.buildenv.2013.03.022.
- [292] Li Y, Yuan Y, Li C, Han X, Zhang X. Human responses to high air temperature, relative humidity and carbon dioxide concentration in underground refuge chamber. Build Environ 2018;131:53–62. https://doi.org/10.1016/j. buildenv.2017.12.038.
- [293] Li R, Sekhar SC, Melikov AK. Thermal comfort and indoor air quality in rooms with integrated personalized ventilation and under-floor air distribution systems. HVAC R Res 2011;17:829–46. https://doi.org/10.1080/10789669.2010.544834.

- [294] Lan L, Lian Z, Pan L. The effects of air temperature on office workers' well-being, workload and productivity-evaluated with subjective ratings. Appl Ergon 2010; 42:29–36. https://doi.org/10.1016/j.apergo.2010.04.003.
- [295] Melikov AK, Skwarczynski MA, Kaczmarczyk J, Zabecky J. Use of personalized ventilation for improving health, comfort, and performance at high room temperature and humidity. Indoor Air 2013;23:250–63. https://doi.org/10.1111/ ina.12012.
- [296] Wu YC, Mahdavi A. Assessment of thermal comfort under transitional conditions. Build Environ 2014;76:30–6. https://doi.org/10.1016/j.buildenv.2014.03.001.
- [297] Zhang N, Cao B, Zhu Y. Effects of pre-sleep thermal environment on human thermal state and sleep quality. Build Environ 2019;148:600–8. https://doi.org/ 10.1016/j.buildenv.2018.11.035.
- [298] Cui W, Cao G, Park JH, Ouyang Q, Zhu Y. Influence of indoor air temperature on human thermal comfort, motivation and performance. Build Environ 2013;68: 114–22. https://doi.org/10.1016/j.buildenv.2013.06.012.
- [299] Baniya RR, Tetri E, Virtanen J, Halonen L. The effect of correlated colour temperature of lighting on thermal sensation and thermal comfort in a simulated indoor workplace. Indoor Built Environ 2018;27:308–16. https://doi.org/ 10.1177/1420326X16673214.
- [300] Bolashikov Z, Melikov A, Spilak M. Experimental investigation on reduced exposure to pollutants indoors by applying wearable personalized ventilation. HVAC R Res 2013;19:385–99. https://doi.org/10.1080/10789669.2013.784645.
- [301] Pan Z, Kjaergaard SK, Mølhave L. A chamber-experiment investigation of the interaction between perceptions of noise and odor in humans. Int Arch Occup Environ Health 2003;76:598–604. https://doi.org/10.1007/s00420-003-0464-3
- [302] Jung W, Jazizadeh F. Vision-based thermal comfort quantification for HVAC control. Build Environ 2018;142:513–23. https://doi.org/10.1016/j. buildenv.2018.05.018.
- [303] Sunwoo Y, Chou C, Takeshita J, Murakami M, Tochihara Y. Physiological and subjective responses to low relative humidity in young and elderly men. J Physiol Anthropol 2006;25:229–38. https://doi.org/10.2114/jpa2.25.229.
- [304] Bluyssen PM, van Zeist F, Kurvers S, Tenpierik M, Pont S, Wolters B, et al. The creation of SenseLab: a laboratory for testing and experiencing single and combinations of indoor environmental conditions. Intell Build Int 2018;10:5–18. https://doi.org/10.1080/17508975.2017.1330187.
- [305] Katić K, Li R, Zeiler W. Machine learning algorithms applied to a prediction of personal overall thermal comfort using skin temperatures and occupants' heating behavior. Appl Ergon 2020;85:103078. https://doi.org/10.1016/j. apero.2020.103078.
- [306] Amai H, Tanabe S ichi, Akimoto T, Genma T. Thermal sensation and comfort with different task conditioning systems. Build Environ 2007;42:3955–64. https://doi. org/10.1016/j.buildenv.2006.07.043.
- [307] Grivel F, Candas V. Ambient temperatures preferred by young european males and females at rest. Ergonomics 1991;34:365–78. https://doi.org/10.1080/ 00140139108967320.
- [308] Xia YZ, Niu JL, Zhao RY, Burnett J. Effects of turbulent air on human thermal sensations in a warm isothermal environment. Indoor Air 2000;10:289–96. https://doi.org/10.1034/j.1600-0668.2000.010004289.x.
- [309] Havenith G, Griggs K, Qiu Y, Dorman L, Kulasekaran V, Hodder S. Higher comfort temperature preferences for anthropometrically matched Chinese and Japanese versus white-western-middle-European individuals using a personal comfort/ cooling system. Build Environ 2020;183:107162. https://doi.org/10.1016/j. buildenv.2020.107162.
- [310] Newsham G, Mancini S, Veitch J, Marchand R, Lei W, Charles K, et al. Control strategies for lighting and ventilation in offices: effects on energy and occupants. Intell Build Int 2009;1:101-21. https://doi.org/10.3763/inbi.2009.0004.
- [311] Bluyssen PM, Kim DH, Eijkelenboom A, Ortiz-Sanchez M. Workshop with 335 primary school children in The Netherlands: what is needed to improve the IEQ in their classrooms? Build Environ 2020;168:106486. https://doi.org/10.1016/j. buildenv.2019.106486. Elsevier Ltd.
- [312] Liu S, Schiavon S, Kabanshi A, Nazaroff WW. Predicted percentage dissatisfied with ankle draft. Indoor Air 2017;27:852–62. https://doi.org/10.1111/ ina.12364.
- [313] Jowkar M, Rijal HB, Brusey J, Montazami A, Carlucci S, Lansdown TC. Comfort temperature and preferred adaptive behaviour in various classroom types in the UK higher learning environments. Energy Build 2020;211. https://doi.org/ 10.1016/i.enbuild.2020.109814.
- [314] Yang Y, Li B, Liu H, Tan M, Yao R. A study of adaptive thermal comfort in a wellcontrolled climate chamber. Appl Therm Eng 2015;76:283–91. https://doi.org/ 10.1016/j.applthermaleng.2014.11.004.
- [315] O'Brien W, Wagner A, Schweiker M, Mahdavi A, Day J, Kjærgaard MB, et al. Introducing IEA EBC annex 79: key challenges and opportunities in the field of occupant-centric building design and operation. Build Environ 2020;178. https:// doi.org/10.1016/j.buildenv.2020.106738.
- [316] Kim J, Schiavon S, Brager G. Personal comfort models a new paradigm in thermal comfort for occupant-centric environmental control. Build Environ 2018; 132:114–24. https://doi.org/10.1016/j.buildenv.2018.01.023.
- [317] de Gracia A, Navarro L, Coma J, Serrano S, Romaní J, Pérez G, et al. Experimental set-up for testing active and passive systems for energy savings in buildings – lessons learnt. Renew Sustain Energy Rev 2018;82:1014–26. https://doi.org/ 10.1016/j.rser.2017.09.109.

Paper c

Quality criteria for multi-domain studies in the indoor environment: critical review towards research guidelines and recommendations

Giorgia Chinazzo^{1*}, Rune Korsholm Andersen², Elie Azar^{3,4}, Verena M. Barthelmes⁵, Cristina Becchio⁶, Lorenzo Belussi⁷, Christiane Berger⁸, Salvatore Carlucci⁹, Stefano Paolo Corgnati⁶, Sarah Crosby¹⁰, Ludovico Danza⁷, Luiza de Castro¹¹, Matteo Favero¹², Stephanie Gauthier¹³, Runa T. Hellwig⁸, Quan Jin¹⁴, Joyce Kim¹⁵, Mandana Sarey Khanie², Dolaana Khovalyg⁵, Carola Lingua⁶, Alessandra Luna-Navarro¹⁶, Ardeshir Mahdavi¹⁷, Clayton Miller¹⁸, Isabel Mino-Rodriguez¹⁹, Ilaria Pigliautile²⁰, Anna Laura Pisello^{20,21}, Ricardo Forgiarini Rupp², Abdul-Manan Sadick²², Francesco Salamone⁷, Marcel Schweiker²³, Marc Syndicus²⁴, Giorgia Spigliantini^{6,25}, Natalia Giraldo Vasquez², Donna Vakalis²⁶, Marika Vellei²⁷, Shen Wei²⁸

¹Department of Civil and Environmental Engineering, Northwestern University, Evanston, USA

²International Centre for Indoor Environment and Energy, Department of Environmental and Resource Engineering, Technical University of Denmark, Kgs. Lyngby, Denmark

³Department of Civil and Environmental Engineering, Carleton University, Ottawa, ON, Canada

⁴Department of Industrial and Systems Engineering, Khalifa University of Science and Technology, Abu Dhabi, UAE

⁵Laboratory of Integrated Comfort Engineering (ICE), School of Architecture, Civil and Environmental Engineering (ENAC), Ecole Polytechnique Fédérale de Lausanne (EPFL), Lausanne, Switzerland

⁶TEBE-IEEM Research Group, Department of Energy (DENERG), Politecnico di Torino, Corso Duca degli Abruzzi 24, 10129 Torino, Italy

⁷Construction Technologies Institute, National Research Council of Italy (ITC-CNR), San Giuliano Milanese, Italy

⁸Department of Architecture, Design and Media Technology, Human Building Interaction, Aalborg University, Aalborg, Denmark

⁹Energy, Environment and Water Research Center, The Cyprus Institute, Nicosia, Cyprus

¹⁰Department of Mechanical Engineering, University of British Columbia, Vancouver, BC, Canada

¹¹Federal University of Santa Catarina, Department of Civil Engineering, Laboratory of Energy Efficiency in Buildings, Florianópolis, Brazil

¹²Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway

¹³Faculty of Engineering & Physical Sciences, University of Southampton, Southampton, UK

¹⁴Department of Architecture and Civil Engineering, Chalmers University of Technology, Sven Hultins gata 6, SE-412 96, Göteborg, Sweden

¹⁵Civil and Environmental Engineering, University of Waterloo, Canada

¹⁶Department of Architectural Engineering and Technology, Faculty of Architecture and the Built Environment, TU Delft, The Netherlands

¹⁷Department of Building Physics and Building Ecology, Faculty of Architecture and Planning, TU Wien, Vienna, Austria

¹⁸Department of the Built Environment, College of Design and Engineering, National University of Singapore (NUS), Singapore

¹⁹Building Science Group, Faculty of Architecture, Karlsruhe Institute of Technology, Englerstr. 7, 76131 Karlsruhe, Germany

²⁰Department of Engineering, University of Perugia, Perugia, Italy

²¹Department of Civil and Environmental Engineering, Princeton University, NJ

²²School of Architecture and Built Environment, Faculty of Science, Engineering and Built Environment, Deakin University, Geelong, Australia

²³Healthy Living Spaces lab, Institute for Occupational, Social and Environmental Medicine, Medical Faculty, RWTH Aachen University, Pauwelsstr. 30, 52074 Aachen, Germany ²⁴Energy Efficiency and Sustainable Building, RWTH Aachen University, Mathieustrasse 30, 52074 Aachen, Germany

²⁵RINA Consulting S.p.a., Via Gran S. Bernardo, Palazzo R, 20089 Rozzano (MI), Italy

²⁶Building Decisions Research Group, University of British Columbia, Canada

²⁷Laboratory of Engineering Sciences for the Environment LaSIE (UMR CNRS 7356), La Rochelle University, La Rochelle, France

²⁸The Bartlett School of Sustainable Construction, Faculty of the Built Environment, University College London, London

* corresponding author:

Department of Civil and Environmental Engineering, Northwestern University, 2145 Sheridan Road, Evanston, IL 60208, USA

E-mail address: giorgia.chinazzo@northwestern.edu (G. Chinazzo)

Abstract

The perception, physiology, behavior, and performance of building occupants are influenced by multi-domain exposures; the simultaneous presence of multiple environmental stimuli, i.e., visual, thermal, acoustic, and air quality. Despite being extensive, the literature on multi-domain exposures presents heterogeneous methodological approaches and inconsistent study reporting, which hinders direct comparison between studies and meta-analyses. Therefore, in addition to carrying out more multi-domain studies, such investigations need to be designed, conducted, and documented in a systematic and transparent way. With the goal to facilitate and support future multi-domain studies and their meta-analyses, this work provides (1) a range of criteria for multi-domain study design and reporting (i.e., defined as quality criteria), and (2) a critical review of the multi-domain literature based on the described criteria, which can serve as guidelines and recommendations for future studies on the topic. The identified quality criteria encompass study set-up, study deployment and analysis, and study outcome, stressing the importance of adopting a consistent terminology and result reporting style. The developed critical review highlights several shortcomings in the design, deployment, and documentation of multi-domain studies, emphasizing the need for quality improvements of future multi-domain research. The ultimate goal of this work is to consolidate our knowledge on multi-domain exposures for its integration into regulatory resources and guidelines, which are currently dominated by single-domain knowledge.

Keywords: IEQ; Human Comfort; Combined effects; Cross-modal effects; Transparent reporting; Research quality assurance

1 Introduction

In industrialized areas, people spend about 90% of their time indoors [1], where they are simultaneously exposed to multiple indoor environmental stimuli, i.e., thermal, visual, acoustic, and air quality variables. It is well known that indoor environmental stimuli affect how people perceive the indoor environment [2], their behaviors [3], health [4], [5], and work-related matters such as real and self-estimated performance [6]–[8], job absenteeism [9], and job satisfaction [10]–[12]. Consequently, indoor environmental stimuli have indirect implications on energy consumption linked to changes in human behavior (e.g., openings/closing windows when mechanical systems are operating) [13]–[15] and on companies' financial revenues, due to the

aforementioned work-related issues [16] and health effects [17]. Therefore, it is paramount to understand occupants' responses to indoor environmental stimuli to design and operate comfortable, healthy, and productive spaces.

Over the past years, many efforts have been devoted to studying human responses to indoor environmental stimuli. Investigations were predominantly carried out for each stimulus, considering visual, thermal, acoustic, and air quality separately. These studies resulted in comfort models and metrics (e.g., Fanger's predicted mean vote model [18], daylight glare probability model [19]), which are included in technical standards and design guidelines (e.g., [20], [21]), and provide comfort requirements for temperature, light, noise and air guality separately. Consequently, buildings and current technologies devoted to controlling the indoor environment are designed on the supposedly independent effects of indoor environmental stimuli [22], [23]. From a cognitive perspective, this approach implies that human perception is a modular function, composed of independent sensory modalities processing sensory stimuli independently of each other as separate modules. For example, the underlying assumption of this mono-sensory approach is that light does not influence thermal perception, and temperature does not affect how the visual environment is perceived. However, it has been shown that human perceptual experience is not modular but is shaped by the combination and integration of a multitude of stimuli experienced simultaneously [24]-[27]. The integration of different sensory modalities is called multisensory integration and results in more robust estimates of occupants' perception [28]-[30]. Examples of multisensory integration relevant to the indoor environment can be found in Calvert et al. [29] and Bertelson & Gelder [31], while anthropological and architectural approaches in Hall [32] and Rapoport [33].

As sensory perception is inherently multimodal, so is people's perception of the indoor environment. While synesthesia (e.g., music excites the perception of color [34]) seems to be a widely known example of the underlying topic, it understates and occasionally misrepresents the nature and importance of integration and binding problems in human perception. Not always human senses are equally involved (think of a visual acuity test such as Snellen Chart), and oftentimes a specific quality of an indoor environment stands out and annoys or satisfies people predominantly. Yet, the overall impression and the effects of an indoor environment remain interwoven and holistic, which is why a multimodal and integrative approach to the investigation of indoor environments appears more valid and representative. Multisensory integration might be one of the factors explaining discrepancies observed between predicted and reported occupant satisfaction [35]–[37], as people are often not satisfied with their indoor environment although threshold values indicated by standards are met. A recent analysis of an extensive survey database shows that only two-thirds of building occupants are satisfied with their environment and multiple environmental stimuli contribute to dissatisfaction, including sound privacy, temperature, and noise level [38].

Although the explanation of how our brains integrate various sensory information is yet to be solved by neuroscience and related fields, it is a good starting point for researchers in the Indoor Environmental Quality (IEQ) domain to expand research in a multimodal manner. Research in this field is necessary considering that "current knowledge on interactions between and among factors that most affect occupants of indoor environments is limited" [39, p. 2]. Since each IEQ stimulus includes several variables, such as (relative) humidity and (air, mean radiant or operative) temperature for the thermal environment, considering all the potential interactions in a single study is unfeasible, even more, if several human responses are considered.

3

For this reason, existing studies focus on the interaction of a few stimuli with selected human responses. To gain a comprehensive understanding of the effect of all the stimuli that can be found in the built environment on all human responses, it is, therefore, necessary to conduct reviews and meta-analyses to combine the results from several studies. This *collective approach* builds upon the knowledge generated as suggested in Schweiker et al. [40].

In recent years, some studies have analyzed the existing literature to understand human responses to multiple indoor environmental stimuli. Torresin et al. [41] reviewed 45 laboratory studies that examined the effects of two or more environmental stimuli on human perception and performance. Wu et al. [42] expanded their review to include field studies and identified multi-domain effects (thermal, acoustic, and illumination) on human perception. Schweiker et al. [40] recognized the link between human perception and behavior and conducted a comprehensive review of multi-domain influences on occupant perception and behavior based on field and laboratory studies. By identifying motivations, theoretical foundations, key methods, findings, and gaps in the field of multi-domain approaches, the authors conclude that "results were often inconclusive and in part contradictory" and emphasize the need to establish a common framework to analyze diverse results, design future studies, and develop standards and guidelines. The incomplete knowledge of multi-domain effects and the inconsistencies across results have been also highlighted in other studies [43], [44]. According to Rupp et al. [45], this outcome is the result of a lack of interdisciplinary research between different disciplines within building science (i.e., visual, thermal, acoustic, and air quality), and between research fields such as psychology, physiology, engineering, and architecture. In addition, the direct comparison of the results of studies can be misleading as the great majority of them differ in terms of objectives, magnitude of considered stimuli, experimental design and setting, studied population, analysis conducted and reporting of the results. Without a common way of designing, conducting, and reporting multi-domain studies, comparisons are difficult to conduct. This is not the first field to recognize and call for the development of more rigorous study designs, transparent reporting, and quality assurance checklists (e.g., [46]).

To address this need, the present work identifies criteria covering the key research aspects that should be considered when designing, conducting, and reporting multi-domain studies and critically reviews the published studies on the basis of these criteria. It is necessary to highlight that this work does not review existing multi-domain investigations for conducting a meta-analysis of their results. In other words, this study does not focus on the questions: "is factor *x* affecting the perception of factor *y*?" or "are interactions between factors *x* and y affecting human response *z*?", but rather on the methodological aspects and characteristics of the reported information for addressing these questions. The described criteria are defined as "quality criteria" as their presence and accurate description in the literature can determine the degree of excellence of a study, which in turn allows its replicability and comparability. The quality criteria can thus be considered as research guidelines and recommendations that aim to establish a solid foundation for future multi-domain studies as a unified approach to facilitate meta-analyses on this topic, helping to untangle the complex effects of multi-domain stimuli on different human responses.

First, the methods applied in this paper are described. Then, the quality criteria are outlined in terms of (1) study set-up (dependent and independent variables, hypothesis, setting features, exposure features, experimental design quality), (2) study deployment and analysis (data collection and processing, participants, data analysis), and (3) study outcome (reporting results, study discussion and conclusion) (see details in

Figure 2). Next, the critical review of the multi-domain literature is performed based on the quality criteria. Finally, the key findings of the critical review are summarized, and future directions are highlighted.

2 Methods

Three steps were followed to define the quality criteria and carry out the critical review of existing multidomain studies: (i) selection of multi-domain studies, (ii) categorization of the studies based on the type of multi-domain effect (i.e., cross-modal or combined) and study type (i.e., laboratory or field study), (iii) definition of the quality criteria.

2.1 Multi-domain studies selection

The selection of research studies analyzed in this work is based on three recent literature reviews reporting studies on the effect of multiple indoor environmental stimuli on different human responses: Schweiker et al. [40], Torresin et al. [41], and Wu et al. [42]. Furthermore, the list of papers analyzed in the reviews was expanded to include additional studies based on forward reference searching and authors' knowledge. The list of considered papers is reported in Appendix A.

Not all studies reported in the three reviews were included in the analysis. Three main selection criteria were applied to meet the aim of the research, described as follows: (i) the study had to involve the response of people (i.e., no simulations, no physical measurements only); (ii) the study had to focus on perception, behavior, and/or performance (i.e., not on physiology only); and (iii) the study had to have as independent variables the physical measurements of two or more of the four IEQ stimuli (i.e., visual, thermal, indoor air quality, and acoustic). Papers in languages other than in English, with an unavailable full text, or not peer-reviewed are also excluded. The excluded papers are reported in Appendix B.

2.2 Multi-domain studies' categorization

The existing literature is reviewed and analyzed by distinguishing the papers according to two study features: type of effect investigated and study type.

Two types of effects are considered in this research (see Figure 1), described as follows:

- Cross-modal effect is when one stimulus influences a non-related response, which is usually triggered by another stimulus (e.g., when light influences thermal responses).
- Combined effect is when multiple stimuli, in combination, affect a response not directly related to a
 specific indoor stimulus (i.e., individual perception such as overall comfort perception and physical
 status, behavior, physiology, and performance). The stimulus can be environmental or belong to
 other domains (e.g., personal, and contextual).

A cross-modal effect can be further categorized into (i) Cross-modal *main* effect; and (ii) Cross-modal *interaction* effect. The difference between the two types of cross-modal effects depends on the levels of the considered stimuli (e.g., dim, and bright are two levels for the visual stimulus, and hot and cold are two levels for the thermal stimulus). Cross-modal main effects occur when the response to stimulus A is influenced by the presence of stimulus B, independently of the levels that they have. Cross-modal interaction effects occur when the effect of different levels of stimulus B on the response to stimulus A differs according to stimulus

A's level. See Figure 1 for a graphical representation of the cross-modal effects. The sub-categorization into main and interaction effects is reported to provide a complete description of multi-domain effects, but it is not used to analyze the reviewed literature in Section 4. However, the authors believe that a comprehensive description of the type of effects could benefit the reporting and interpretation of future multi-domain studies. Figure 1 schematizes cross-modal and combined effects (multi-domain studies), distinguishing them from the same-modality effects (single-domain studies), which are not considered in this research.



Figure 1: Top: Schematic description of the type of effect: cross-modal, combined and same-modality effects. The light gray dashed lines in the cross-modality effect refer to the influence of one domain (e.g., temperature) on the same-modality response (e.g., thermal comfort) when another domain is considered in the investigation (e.g., illuminance). In a multi-domain study, such effect does not have to be included (e.g., a study could look solely at visual influences on thermal perception without observing the effects of thermal properties (or their interactions with the visual properties) but only controlling for them). Bottom: Graphic representation of the types of cross-modal effects between two stimuli. a) Cross-modal <u>main</u> effect of stimulus A; c)

cross-modal interaction effect of stimuli A and B; d) the main effect of stimulus A, no effect of stimulus B. Adapted and expanded from Coolican [47].

The study types considered in the analysis are: (i) lab study (including test room, climate chamber, and airplane simulator) [48], and (ii) field study [49]. The living lab (i.e., a conventional space equipped with measurement tools in which occupants conduct their normal lives or work [50] is a study type not used in the considered papers and hence it is not used to categorize the papers in the following analysis.

Table 1 summarizes the distribution of the considered studies according to the effect type and study type. Lab studies outnumbered field studies, while cross-modal and combined effects were equally investigated across studies. Most of the cross-modal effects were investigated in lab studies, while an equal number of combined effects were tested in both lab and field studies. Sometimes, cross-modal and combined effect types were investigated in the same study, in the great majority of the cases in lab studies.

		Cross-modal	Combined	Combined and	Total
				cross-modal	
Study type	Lab	36%	17%	23%	76%
	Field	4%	19%	1%	24%
Total		40%	36%	24%	100%

Table 1: Distribution of the considered studies according to the effect type and study type.

2.3 Research quality criteria

The research quality criteria (Figure 2) were used to critically analyze the published studies and can serve as research guidelines and recommendations for future studies. These criteria are categorized into three groups, defined as (1) study set-up, (2) study deployment and analysis, and (3) study outcome (Figure 2). The collection of quality criteria was determined first on the basis of the authors' experience with multi-domain studies, previous review efforts, and intensive online meetings within and beyond IEA EBC Annex-79 meetings¹. Such basis was constantly reviewed during the analysis of the studies considered for this work and augmented upon necessity. The selected criteria focused on methodological and reporting features. The introductory sections with the related analysis of previous literature and reference to validated theoretical models and theoretical assumptions are not considered in the analysis as multi-domain studies have been reported to rarely carryover previous studies' findings and to lack foundational theories to formulate and test research hypothesis [40].

Some of the considered research quality criteria are common to all experimental investigations, while others are specific to multi-domain studies. However, to guide future researchers on what to consider while designing, deploying, and reporting multi-domain investigations, all quality criteria are described in the same depth in the next section followed by their application in a critical review of published literature.

¹ "Occupant-centric building design and operation" (http://annex79.iea-ebc.org/)

Quality criteria for multi-domain studies

l Study set-up	Study conduction and analysis	Study outcome
 Dependent variable human response type and details collection approach (subjective, objective) detailed response description Independent variables sensory domain and their combination environmental variable description number of levels design values Hypothesis hypothesis statement hypothesis description Setting feature environmental condition not varied as independent variable building or space type setting description and picture building systems (heating, cooling, ventilation, lighting) facade/ fenestration description human-building interaction features (control opportunities) location Exposure feature exposure characteristics exposure characteristics exposure for participants per exposure of participants per experimental conditions in a day if within subjects total length of the experiment per participant adaptation time Experimental design quality randomization and counterbalancing blind procedure confounding variable control (experimental, statistical) nul condition 	 Data collection and processing environmental stimuli measured description measurements' features (location, frequency, post-processing) design vs. measurement values comparison Participants distribution (total number and sex) personal characteristics (culture/origin, physical condition preexperiment, health status, age height, weight) experimental involvement (type of observation, task, payment, ethical approval, information sheet, consent sheet) Statistical analysis justification of choice statistics type power analysis effect size 	 Reporting results data documentation descriptive statistics inferential statistics classification of observed effect type effect size effect direction Study discussion and conclusions mechanism explanation limitations conclusions practical implications future studies
Figure 2: Research quality criteria consider	ed in the critical review of existing m	ulti-domain studies and that

can serve as guidelines and recommendations for future multi-domain research.

3 Description of research quality criteria for multi-domain studies

The quality criteria shown in Figure 2 are described in the following.

3.1 Study set-up

3.1.1 Dependent variables

A clear description of the investigated dependent variable(s) is of primary importance since they express the human responses to variations of the independent variables (i.e., the investigated stimuli).

The dependent variables in multi-domain studies refer to the different human responses that can be captured in experimental or observational settings. Figure 3 illustrates the type of human responses that can be collected and the associated methods of assessment in both field and laboratory investigations, and in relation to the type of effect considered (combined or cross-modal). Responses can be described according to the nature of the data collection approach, i.e., subjective or objective. Subjective data from occupants is collected by interviews or survey methods querying self-reported perceptions or opinions. Objective responses include physiological signals, test grades and other quantitative observations (e.g., number of interactions between occupants and building components).



Figure 3: Schematic representation of the type of human responses that can be collected in studies investigating cross-modal or combined effects.

In addition to a clear description of the human response type under consideration, studies should clearly report how these responses were gathered by specifying the method(s) of assessment and the adopted tools used (e.g., questionnaire for perception responses, test type for performance responses, sensing technology for behavioral and physiological responses). Such tools must be described in detail to allow reproducibility and a comprehensive understanding of the followed methodology. In addition to the details of the assessment method, the time and frequency of assessment must also be reported. Special attention must be given to the description of the questionnaires and the related responses when subjective evaluations are sought. Questionnaire responses, if not in an open-ended format, refer to scales that can be categorical (CS), visual analogue (VAS), categorical scale combined with VAS (graphic CS), semantic differential, or dichotomous. To get comparable data and results, agreement on specific aspects of the subjective assessment scales is of primary importance. These can be summarized in (i) adopted terminology in the questions and responses, (ii) type of scale used, and (iii) (only in case of CS) number of provided response categories. From this point of view, it is essential to report the original text of the adopted questionnaire, preferably in both the original language and English.

3.1.2 Independent variables
Multi-domain studies are characterized by the presence of more than one environmental stimulus, presented in combination. Such environmental stimuli are the independent variables of the study. Reporting the type of combination of the environmental stimuli is taken for granted as it represents the essence of each multi-domain investigation. However, the detailed description of the independent variables needs further attention. Correctly describing independent variables in multi-domain investigations is crucial for conducting replication studies and facilitating meta-analysis and comparison across studies. The way of reporting independent variables depends on the study approach, either experimental or observational. In experimental investigations, usually carried out in a climate chamber or an environmentally controlled space, the experimenter manipulates the independent variables to measure their effect on the dependent variable. In contrast, in observational studies conducted in field setups, the experimenter cannot usually control the independent variables, which are measured to observe correlations between independent and dependent variables.

In multi-domain papers reporting experimental studies, researchers should always clearly indicate the independent environmental variables in terms of type (e.g., air temperature), the number of levels (e.g., 3), and design values (e.g., 22 °C, 25 °C, and 28 °C). In experimental cross-modal studies, both same-modality and cross-modal independent variables must be clearly described. For example, in a study investigating the effect of Correlated Color Temperature (CCT) of light on thermal perception, the cross-modal independent variable CCT must be reported together with the air temperature, representing the same-modality variable. In multi-domain papers reporting observational studies, as the independent variables are usually not controlled but measured, researchers must clearly report the measured variables' descriptive statistics, i.e., measures of central tendency and variability. This information is critical for evaluating the external validity of the study's findings, and whether the findings are generalizable to the study's source population of people and buildings. If the researchers cut the independent variables' continuous values into bins for analysis (e.g., [51]), then each bin must be described in terms of value counts and mean or median. Such description is necessary as the choice of bin number and position is arbitrary and generally do not have practical/scientific meaning and could influence the results. Analyzing solely with the described bin method may lead to some loss of information. Therefore, it must be complementary to other descriptions of the independent variables. It is recommended to opt for continuous and numerical design values (rather than categorical ones such as "blue" and "red" when colors are assessed) that enable replication studies and facilitate meta-analysis. When several levels of the same independent variable are considered, it is a good practice to assign different labels to the different levels, facilitating the comprehension of both the experimental design and the results. Another good practice is the consideration of possible covariates (e.g., summarized for thermal comfort by Wang et al. [52] or Schweiker et al. [53]) that are not environmental, for example, gender, age, and body mass index. Refer to 3.1.4 and 3.2.2 for further discussion on the topic.

3.1.3 Research hypothesis

Stating and describing the research hypotheses of a study leads to a better comprehension of the work, even if it consists of an exploratory study searching for discoveries, trends, correlations, or relationships between the measurements in which outcomes would generate new ideas or hypotheses (that need to be confirmed in follow-up studies). When conducting a causal research based on a pre-existing theory and aimed to determine what occurs to one measurement on average when another measurement is changed, it is possible

to state causal hypotheses. The causal hypothesis should be stated in all cases where the scientific literature reasonably sounds or where the current state of knowledge on the topic makes it possible. This makes it easier to determine the research scope and establish a correlation between the initial assumptions and the results.

Research hypotheses should be described in terms of "directions." A hypothesis with direction expresses a direct or inverse relationship between dependent and independent variables: if the independent variables increase (or decrease), the dependent variable increases (or decreases). An example is the Hue-Heat Hypothesis, posing that warm-appearing colors, such as red or yellow, make people feel warmer, while the opposite effect is obtained with cold-appearing colors [54], [55]. A hypothesis "without direction" expresses a general relationship between dependent and independent variables regarding the influence one may have on the other (e.g., thermal conditions influence acoustic sensation and perception [56]).

3.1.4 Setting feature

Experimental setting features play a fundamental role in the combination and interaction of the variables investigated in multi-domain studies. The following paragraphs summarize the importance of collecting and reporting information regarding: the environmental conditions not varied as independent variables, the building and space type, the space layout, equipment, ventilation, and lighting, the control opportunity, and the experimental location.

Along with a detailed description of the environmental stimuli investigated as independent variables, it is necessary to include a comprehensive description of all the environmental features, considering as well those that were not included as independent variables. Such a comprehensive description of the indoor environment might help to understand potential differences across studies and detect confounding factors.

The type of building or space (i.e., office, educational, residential, or others) determines several aspects of the experimental setting, e.g., indoor space layout, furniture, occupants' activity, or interaction with other people. Specifying the building or space type in a field study but also the "emulated" space in an experimental lab setting is crucial.

Besides indicating building and space typology, a description of the space layout, equipment, ventilation, and lighting gives a comprehensive and immediate overview of the space. Layout description should include dimensions and photos for furniture type and disposition, for instance, the distance between the seats and relevant building elements (e.g., windows). Describing relevant equipment (such as HVAC, artificial lighting etc.) and building elements (windows, shadings etc.) is also important, as these influence indoor environmental conditions and occupants' interaction with available interfaces [57]. Lighting type and related details should be described, that is, electric, natural, or a combination of the previous, and possibly specifying if electric lighting was designed to obtain extreme conditions (e.g., a poorly lit environment). Related to lighting, fenestration systems should be detailed with reference to shadings (internal or external) or, if present, advanced technologies (e.g., smart windows, low-emissivity coatings etc.).

Another relevant feature involves the interactions that occupants can have with building interfaces (i.e., occupant control). Occupant control can influence not only human interactions but also the satisfaction and behavior of users in different domains [58], [59]. Thus, reporting exhaustive information on control opportunities within the indoor space is highly recommended.

Despite not being directly related to the experimental indoor space, knowing the location brings insights into the climatic conditions and indicates participants' cultural approach, including their habits, perception, and reporting attitudes.

3.1.5 Exposure feature

This section covers the conditions to which subjects are exposed (i.e., exposure features). Such information is essential when analyzing the results and ease the replicability of the experiments. The first aspect to consider when defining and describing the exposure features is whether the experimenter measures human responses to different exposures within-subjects (i.e., all participants are exposed to all conditions), between-subjects (i.e., each participant is exposed to some of the conditions), or a mix of the two (i.e., participants are exposed to all conditions of one experimental variable and to some of the conditions of another experimental variable). An important aspect to provide clearly in this latter case is the number and combination of tested experimental conditions (e.g., mixed design with one between-subjects condition and two within-subjects conditions).

Each study must then define and report the characteristics of exposure, which we can divide into (i) the "exposure type" (i.e., steady-state, dynamic, or combined), (ii) the length of exposure for each experimental condition (e.g., exposure to warm temperature for 30 minutes and to each light condition for 10 minutes), (iii) the number (and demographic information) of participants per experimental condition, and (iv) the timing of the exposure (during the day and the year). For example, seasonal variations are important to be recorded given their impact on several human responses [60]. In addition, publications suggest the potential variations of human responses during the day [61]–[63], highlighting the importance of recording and reporting the exact time of the day during which the experiment is conducted. In the case of within-subjects design, it is also necessary to report the number of experimental conditions experienced in a day by the same participant and their potential distribution over several days. It is also good practice to report the total length of the experiment for each participant, especially in the presence of within-subjects design when each participant is exposed to a series of experimental conditions.

The adaptation time (or acclimation time, i.e., is the time given to the participants to adapt to the experimental conditions) is another exposure feature that should be considered and reported in all studies. The consideration of the adaptation time is more relevant in studies involving the thermal domain since the human body requires longer time to reach a steady-state thermal response in a new thermal condition and/or at a different activity level [64] and strongly depends on the temperature difference between experimental and pre-experimental conditions.

3.1.6 Experimental design quality

Recently, a replication crisis has been in the spotlight of the scientific community [65], [66]. This crisis is mainly attributed to selective reporting bias (i.e., reporting only significant results and omitting non-significant results) and poor experimental design quality (e.g., lack of a random assignment of subjects). A quality experimental design should follow several principles commonly reported in statistics books (e.g., [47], [67], [68]): (i) randomly assigned or counterbalanced experimental conditions; (ii) blinded (single- or double-blind) experimental procedure; (iii) controlled confounding variables (experimentally or statistically); (iv) reported study null condition; and (v) repeated one or more experimental conditions.

Besides the recommendations above for specific experimental design elements, a pre-design step for countering the replication crisis trend of underreporting results that did not reach significance is pre-registration. In pre-registration, before beginning to run an experiment or study, the authors outline their hypotheses, methods, and analyses in a public registry (e.g., https://aspredicted.org/). If this step had been taken, the reporting of randomization, blinding, controls, and hypotheses in the analyzed multi-domain studies would have also been accomplished. None of the reviewed studies were pre-registered, as far as could be determined. The lack of pre-registration is a common feature of all the experiments conducted in the Building Science field and not only for multi-domain experiments. A noteworthy exception in this field is the study by Schweiker et al. [69], which had been registered on osf.io.

3.2 Study deployment and analysis

3.2.1 Data collection and processing

A comprehensive reporting of the data collected and the way such data is processed before the statistical analysis is essential as it facilitates comparison, meta-analysis, and the reproduction of an experiment. In multi-domain studies, it is important to measure and report all the environmental stimuli – not only the investigated independent variables but additional factors that are hypothesized to be relevant. For example, in a study on the cross-modal effect of light on thermal perception, the air quality, and acoustic conditions should be reported as well (at the best of the researcher's knowledge). Without measuring the possible confounders, the analysis necessarily excludes them, and therefore the results of the analysis are less valid. Besides the type of environmental stimuli collected, it is important to report how the measurements were performed and the data processed before the statistical analysis. More specifically, the measurements' location, frequency, processing (e.g., "is data averaged over a specific period of time? How is missing data treated?"), and differences from the design conditions should always be reported or discussed. Regarding the measurement location, it is important to highlight that measurements, whenever possible, should be taken in proximity to the occupant, based on the recommendations of the domain-specific guidelines, to correctly evaluate the effect of one environmental stimulus on another domain perception or behavior since those are the actual environmental conditions that affect the occupant.

3.2.2 Participants

Like all studies involving human subjects, multi-domain studies should include a concise but exhaustive description of participants' characteristics to (i) demonstrate the representativeness of the research findings (sample size and confidence interval), (ii) provide insights on the generalizability of the findings as well as possible limitations of the study (external and internal validity), and (iii) test the impact of these confounding factors on the hypothesis testing and provide confidence of the results. Sufficiently detailed information, as far as possible by obeying privacy issues (e.g. following the General Data protection regulation, GDPR [70]) on the distribution of participants (e.g., total number, number of males/females/not disclosed gender), the personal characteristics of the subjects (e.g., culture/origin, age/height/weight, health status, and verification of physical conditions before the experiment), as well as information related to their experimental involvement (e.g., direct observation, described task, participation payment, detail on the ethical approval and consent), is required for reviewing research findings and aid future replication studies.

3.2.3 Statistical analysis

Statistical methods are fundamental instruments in experimental studies to support the interpretation of the results and develop accurate, reliable, and representative experimental designs. To this end, statistical tools are used for characterizing the recorded observations, testing for differences among data series, quantifying the effect size, developing, and validating models, and identifying the sample size required to detect an effect in an experiment given the desired significance level, effect size, and statistical power. The adoption of a specific statistical method should always be justified. Also, the studies should communicate clearly the hypotheses tested and the assumptions set together with the adopted statistical tests and significance levels. Although publication space is scarce, and journals often urge authors to draft their manuscripts as concise as possible, detailed reporting of statistical analyses is mandatory if authors wish to present their results in a replicable fashion and to make their findings amenable to meta-analytical efforts [71]. To rely on the results of statistical methods, to promote transparency and reproducibility of experiments, and to ensure robustness to systematic errors, it is essential that studies clearly state the sample size, identified through an a priori power analysis or justified by any other method (e.g., resource constraints, accuracy, heuristics) to provide evidence of representativity. Effect sizes are important as a measure of how meaningful the difference between different variables or groups is to demonstrate the actual real-life significance of the experimental outcomes. It not only indicates the strength of the statistical results, but also puts a study into perspective by facilitating the comparison across different studies and helping to determine sample sizes for future studies. Beyond the basic descriptive findings such as measures of central tendency, error, and dispersion as well as data distribution characteristics, a detailed summary of the statistical results also includes the reporting of non-significant results, degrees of freedom (related to sample size), missing data, and potential exclusions of data points as well as imputation methods, if applied. Lastly, any changes and adaptations applied to the statistical models and tests need to be stated [72].

In case the full report of these figures is not possible in a paper's results section or may appear redundant to the reviewers (indeed, some statistics can be reverse-engineered and checked for plausibility from reported results with tools such as G*Power or statcheck, see [73], [74]), authors are encouraged to seek online supplemental publication possibilities which are provided by a growing number of journals. Lastly, although the full extent of how strong various research fields are plagued by publication bias remains unknown, selective publication of only significant findings bears the threat of misrepresenting findings and puts the burden of detailed checks and evaluations on the researchers conducting the research synthesis [75], [76]. Finally, it is recommended that the statistical method is decided before the experimental design, guided by the aim of the study. In this way, the experiment is designed to get the data needed to support the data analysis and aim of the study.

3.3 Study outcome

3.3.1 Reporting results

This section does not focus on the specific results obtained in the considered papers (e.g., "is temperature affecting visual sensation?") but on the content that should be reported in the result section of each study and the way such content should be presented. In general, for reasons of transparency, comparability, and general advancements in a particular research area, the results must contain sufficient information regarding

each individual outcome to facilitate replication or metanalysis efforts. This is especially true for the case of multi-domain studies due to a large number of potentially dependent and independent variables, which cannot be addressed through a single study. Given the need to report on each permutation of possible interactions between variables, the number of reported outcomes increases exponentially when compared to single variable studies. As space is usually limited, documenting data alongside the paper – including a detailed description of the number of data points excluded and argument (statistical, thematic) for exclusion can be done in a separate document, such as data descriptors, e.g. [77], [78].

While the section about results in general reports problem-specific findings intended to answer a specific research hypothesis, the following basic information needs to be provided²: (1) descriptive statistics for each individual variable collected (depending on data type, e.g. measures of central tendency and variability alongside with sample size); and (2) results from inferential statistics, disregarding whether they are statistically significant or not (see reporting bias in science and the potential of misrepresentation of scientific results [79]). From this perspective, it is of utmost importance to report all main and interaction effects, the exact level of significance (i.e., p = .04 and not p < .05) [80], and the effect size, whenever it is possible to determine it. The results should be in line with the type of the statistical test and its purpose described in the paper (most likely in the Methods section). The observed effects, but not stated as primary or secondary research hypotheses, need to be flagged as "*explorative*".

Specifically for multi-domain studies, a classification of the expected and observed effect is recommended, that is, whether it is a cross-modal effect, or a combined effect. In addition, further classification of the results should be reported according to the effect type. For cross-modal effects, it is necessary to indicate the "direction" (i.e., positive, negative or no effect) of the effect instead of merely reporting the presence of an interaction. The direction should be described according to the level(s) of the same-modality independent variable. For example, if temperature influences visual perception, the study should clarify if the effect of a specific visual level (e.g., dim illuminance) is positive or negative according to a specific thermal level (e.g., cold temperature).

Figure 4 schematizes the possible cross-modal effects between two stimuli and the resulting directions. As illustrated, the presence of a stimulus B can result in a negative effect (i.e., strengthen a negative or weaken a positive response of stimulus A alone as in Figure 4a and Figure 4b), positive effect (i.e., weaken a negative or strengthen a positive response of stimulus A alone as in Figure 4d and Figure 4d and Figure 4e) or no effect (i.e., response to stimulus A is not affected by the presence of stimulus B as in Figure 4c) on the response to stimulus A.

² For some readers, some of these points may appear as common knowledge. However, our review showed that there are still a substantial number of papers published without including even basic information such as measures of dispersion like standard deviations.





Table 2 illustrates a possible scheme for summarizing the results of a cross-modal effect between two stimuli, with three levels each. The number of columns and rows can be adapted to the number of levels tested for each stimulus. The following descriptions of the results are suggested as examples:

- Significant *negative* effect: the presence of stimulus B at level x (e.g., illuminance, dimmer condition) *strengthen the negative* or *weaken the positive or neutral* response of stimulus A (e.g., thermal comfort) at y level (or all levels) of stimulus A (e.g., colder and warmer). In Table 2, this effect is shown in the first column of stimulus B.
- Significant *positive* effect: the presence of stimulus B at level x (e.g., illuminance, brighter condition) weaken the negative or strengthen the positive response of stimulus A (e.g., thermal comfort) at y level (or all levels) of stimulus A (e.g., colder and warmer). Table 2, this effect is shown in the last column of stimulus B.

Contrary to the example described, note that the effects can be different according to the level of stimulus A (e.g., be positive at low level and negative at high level). Results could also be represented graphically as in Figure 1.

Concerning combined effects, when not described as a combined index, they can be further specified into additive, synergistic, or antagonistic, with reference to the *medical analogies* described in the ASHRAE Guideline 10-2016 [39, p. 7]. Figure 5 describes the possible combined effect types, according to the following definitions reported in the standard:

• Additive: when each of the stimuli affects the human response and their combined presence results in the sum of their separate effects (no effect of interactions);

• Synergistic: when the combined presence of two or more stimuli results in a greater effect than the sum of their separate effects (enhancement effect of interactions);

• Antagonistic: when the effect of the combined presence of two or more stimuli is less than the sum of their separate effects (diminishing effect of interactions).



Figure 5: Schematic example of combined effects of stimuli A and B on human response and the resulting effect description.

Also, in the case of combined effects, results could be described as illustrated in Table 3 according to the levels of the considered stimuli.

Table 2: Template for results reporting for cross-modal effects of stimulus A + stimulus B on the response to stimulus A.

			Original effect of stimulus A on the response to	Effect of Stimulus A + Stimulus B on the response to stimulus A			
				Stimulus B levels (e.g., visual – illuminance)			
			stimulus A (same- modality)	(e.g., dimmer)	Connorriever	(e.g., brighter)	
els (e.g., air	thermal – air	Lower level (e.g., colder)	discomfort	e.g., negative		e.g., positive	
A lever		Comfort level	comfort	e.g., negative		e.g., positive	
Stimulus the		Higher level (e.g., warmer)	discomfort	e.g., negative		e.g., positive	

Table 3: Template for results reporting for combined effects of stimulus A and B on human responses.

			Effect of Stimulus A + Stimulus B on human response "x"				
			Stimulus B level				
			Lower level	Comfort level	Higher level		
Stimulus A	level	Lower level	e.g., additive				
		Comfort level					
		Higher level					

3.3.2 Study discussion and conclusion

As for all research papers, it is obvious that multi-domain studies should present the discussion and conclusion sections. They should naturally follow and comment on the results of the study (hence being datainformed and not speculative), with reference to the results of previous studies on the topic. These sections should also include future studies, study limitations, mechanism explanations, and practical implications. With the declaration of future studies and the identification of the limitations, authors provide food for thought for the scientific communities pointing out the direction of the research highlighting the way to create a shared opinion. The tentative explanation of the mechanisms related to the results can be used as the basis for future research. Finally, the identification of practical implications of the research creates a direct link between the experiment and the impact on human life and society.

4 Critical review of existing multi-domain research

The following sections review existing multi-domain research based on the quality criteria defined in Section 3, presenting a transversal analysis of the percentage of studies reporting the aspects whenever a specific quality criterion is not present in all studies.



4.1 Review of study set-up

4.1.1 Dependent variables: human responses

Figure 6: Distribution of human response types in multi-domain studies according to the type of effect (combined, cross-modal) and the study type (field or lab).

Figure 6 summarizes the distribution of human response types investigated in the considered multi-domain studies and shows that most of the studies investigated perceptual responses. In most of the studies, perceptual responses were the only human responses considered, and only in a few studies were human responses considered in combination with performance (e.g., [81]–[85]), physiology (e.g., [86]–[88]), or behavior [89]–[91]. Behavior and performance alone were investigated in fewer studies compared to perception. The limited number of studies reporting physiological responses may be due to the papers considered in this research, although it included studies with physiological responses in combination with other human responses only. Physiological responses were collected in lab studies only. This outcome may suggest that sensing techniques for collecting physiological signals are still too invasive or too expensive to be used in field studies. Similarly, performance studies were only conducted in lab environments. Behavioral responses were primarily collected in field studies, unless they were investigated with other human responses in lab studies [90]–[94]. Additionally, behavioral investigations in field studies were based on the collection of data on windows and blinds operations [95]–[102], thermostat setpoints [81] and ventilation speed settings [94]. Perception responses were equally collected in both lab and field studies.

When observing the distribution of data collection approaches (objective and subjective) adopted for gathering human responses, performance, behavior, and physiological responses were primarily collected via objective approaches. Performance was objectively assessed through dedicated performance tests while exploring different cognitive dimensions (e.g., proofreading, arithmetic, problem solving, creative thinking, etc.), which were generally quantified through the number of correct answers provided [103]–[106], the

associated response times [84], [107], or both [108]. The subjective assessment of the performance was conducted through questionnaires [81]–[83], [109]. Methods for the objective evaluation of human behavior were based on the experimenter's observations of subjects' clothing adaptation throughout the test (e.g., number of clothing items put on/off) [92], sensors to assess windows and blinds state [102], or equipment settings (e.g., selected fan speed level) [94]. Information on windows state was also commonly collected through physical measurements by means of sensors, especially in long-term field studies [95], [97], [101], [102]. The objective approach for physiological aspects relied upon the use of wearable sensing technologies. The most investigated signals were heart rate, skin temperature, and blood pressure, while other signals such as core temperature, electroencephalography (EEG), electrooculography (EOG), blink measurement, eye movement, respiration rate and skin conductance (also through the use of an algorithm for the detection of artifacts [110]) were rarely included in multi-domain studies [55], [108]. The subjective approach to collecting physiological observations focused on direct questions about subjects' perception of health symptoms (e.g., eye irritation, throat irritation, and skin dryness) via questionnaires [87], [88], [111].

When studying human perception through subjective assessment, the top five assessment categories were perception, comfort, satisfaction, acceptability, and preference. They were primarily assessed through categorical scales. Perception, satisfaction, and preference were most often expressed through a 7-points scale, while comfort was mainly investigated on a 5-points scale, and acceptance on a 3-points, 4-points, or dichotomous scale (acceptable, not acceptable). Some trends can be identified, most likely as a result of questionnaires referencing standards pertaining to human perception research [21], [112], [113]. It must be noted that sometimes, despite evaluating the same assessment category (e.g., thermal sensation) and indicating the same number of response categories, the labels used can vary [114]. Similarly, visual analogue scales can have varying ranges (e.g., 0 to 100 or 0 to 60) [111], [115]. These differences may increase the difficulty of comparing results across studies.

4.1.2 Independent variables: combined environmental stimuli

Figure 7 reports the distribution of independent variable combinations in the considered papers. In general, thermal and visual stimuli were the most investigated combination of independent variables, mainly studied in cross-modal investigations. Thermal and IAQ, and thermal and acoustic, were the second and third most common combinations in cross-modal studies, highlighting the dominant interest in thermal studies. In contrast, combined effect papers tended to focus more on all four environmental stimuli and their effect on overall perception and performance. Behavior and physiological responses were primarily studied in response to thermal and visual, and thermal and IAQ combinations. The least studied combinations were visual and IAQ, and acoustic and IAQ, both in cross-modal and combined investigations.



Figure 7: Distribution of independent and dependent variables, according to the type of effect (combined, cross-modal), study type (lab, field) in the reviewed papers. Acu = Acoustics; IAQ = Indoor Air Quality; The = Thermal; Vis = Visual.

All the reviewed studies reporting experimental investigations about cross-modal effects indicated the type of cross-modal independent variables. Only a few studies did not report the number of levels (3%) or the design values (6%). In contrast, the same-modality independent variable was sometimes not described in terms of type (12%) and design value (18%). Figure 8 summarizes the design values of the independent variables (facet headings) used in experimental cross-modal investigations. The sensory domains on which their effect was tested are indicated on the x-axis. For example, the first box on the top-left of the graph illustrates the values of indoor air temperature tested to assess their influence on acoustic, IAQ and visual responses. The "thermal" response is not indicated as it is a same-modality response. Each dot represents a tested condition, while the box-plots illustrate the overall ranges of values for each independent variable (i.e., mean and interquartile range). It can be observed that the thermal independent variables (air temperature, relative humidity, air velocity and Fanger's Predicted Mean Vote - PMV) were the most considered independent variables, investigated to assess their influence on all the non-thermal domains. This outcome can be expected given the strong interest in thermal studies previously highlighted. For all independent variables, extreme values were commonly used in experimental investigations. The choice of extreme stimuli can be justified because if a cross-modal effect is not observed for extreme stimuli, it is unlikely that it will occur in normal conditions (if it assumed that the relationship between the dependent and independent variable is linear). Interestingly, when the same independent variable was tested on different sensory domains (e.g., air temperature effect on acoustic, IAQ and visual perception), the distribution and median of its values were consistent across domains.

Figure 8 highlights the least and more explored combinations or range of independent variables tested in cross-modal investigations, an information that can be used to guide future multi-domain studies. Ventilation rate was not represented, as only one value (30 l/s influence on thermal and acoustic responses) was present in the considered studies.



Domain affected by the independent variable

Figure 8: Number of cross-modal studies depending on the domain and reported value of the Independent variable (IV). Each dot represents the value of the independent variable investigated in the literature. Dots are randomly jittered to discern more values and their left or right position with respect to the vertical line in each boxplot has no additional meaning.

The great majority of the reviewed studies reporting experimental investigations about combined effects clearly indicated the type of independent environmental variables. Design values were specified in most of the studies as well (98% of studies reporting thermal stimuli, 88% visual, 95% IAQ and acoustic). Level values were less frequently reported for each environmental variable: 71% for thermal and 76% for visual, IAQ, and acoustic. Figure 9 shows the combination of the independent environmental variables reported in experimental studies investigating combined effects. It can be observed that air temperature and illuminance were the most studied variables. Figure 9 also indicates the dependent variables, confirming the overwhelming focus on overall comfort and performance as discussed before and highlighted in Figure 7. The range of considered values broadly varied between variables and the investigated human response. Figure 9, similarly as Figure 8 for cross-modal effects, highlights the least and most explored environmental stimuli combinations tested in combined effects research, a piece of information that could be used for future multi-domain studies.



Figure 9: Combination of independent variable values employed in experimental investigations studying combined effects, considering the dependent variables (indicated with colors). Ventilation rate, air velocity, relative humidity, CCT, and VOC (Volatile Organic Compounds) are not represented as only a few data points

were present in the considered studies. Some outliers of the represented independent variables were excluded as well (i.e., 3000, 4000 lx). Each dot represents an investigated independent variable.

Finally, in most of the reviewed multi-domain investigations, the independent variable values were continuous, only rarely categorical (e.g., natural versus electrical light, wall colors, good vs. bad light comfort conditions).

4.1.3 Research hypothesis

Only 53% of the considered articles reported the research hypothesis, divided into those where the hypothesis was "with direction" or "without direction" (Figure 10). Studies carried out in laboratories had the highest percentage of hypothesis statements (59%) with almost the same percentage for "with direction" (28%) and "without direction" (30%). In addition, most articles with a hypothesis statement belonged to "cross-modal" experiments, mainly carried out in laboratories. In research on the combined effect, only 26% of the papers stated the hypothesis. Among them, the study by Lin [116] can be considered as a best practice of the category "with direction" because the author clearly stated the hypothesis of the work: "higher noise intensity and either too low or too high illumination intensity will reduce visual performance". The papers on the combined effect not reporting the hypothesis might be due to the lack of research and data on the topic [84], [107], [108], [117]–[119].

More than 60% of field studies did not state the hypothesis. This may be due to the number of uncontrolled variables that make it difficult to formulate a clear hypothesis. In this case, the research was based on generic assumptions that needed to be verified in the current conditions.



Figure 10: Relative frequencies of hypothesis statements in the considered studies.

4.1.4 Setting feature

The considered setting features and their presence in the literature are summarized in Figure 11 and discussed in the following.



Figure 11: Relative frequencies of the experimental setting features in the considered studies.

Despite the fundamental importance of a comprehensive description of all the environmental conditions (besides the ones varied as independent variables), most of the considered studies did not report them. Only about *20%* reported a comprehensive overview (e.g., [54], [55], [120]) (i.e., with all environmental stimuli described), while an additional *37%* included a partial description. The description was often present in lab studies, especially when it was comprehensive (Figure 11). Many studies described only the features that were relevant in the investigated domains, neglecting the potential cross-modal influence of other domain-related features.

The building or space type was not reported in 47% of all the studies. In laboratory studies, the "emulated" space type was not reported in 60% of cases. From the studies that did report the building or space type, it can be observed that multi-domain studies were mainly carried out in offices (34%), followed by educational (10%) and residential buildings (5%).

In the investigated studies, *48%* omitted space pictures and *35%* reported it without participants (ethical issues may play a role in this case). A good reference for description can be found in several studies [83], [121]–[124]. Best practices of pictures can be found in [94], [105], [125]. *55%* of studies did not report any information about the heating and cooling systems, and only *50%* provided information on ventilation type. Examples of these systems descriptions can be found in Tiller et al. [84] and Yang and Moon [126]. Description of ventilation type can be found in Skwarczynski et al. [127]. Lighting information was provided in *55%* of studies on daylight could be explained by the challenging experimental conditions that such an environmental variable entails. Winzen et al. [131] and Chinazzo et al. [55] described the lighting system. Among the investigated studies, information about the fenestration system was only accounted for in *22%* of papers. Reference descriptions can be found in Haldi and Robinson [124] and Garretón et al. [132].

Among the studies considered in this work, 86% provided information on control opportunities, especially in laboratory experiments. Discrepancies between laboratory and field studies can also be recognized in terms of occupants' level of control over the environment since lab experiments were largely characterized by the lack of control by occupants (92%). The same situation can be found in only 11% of field experiments.

Reporting exhaustive information on occupants' possible interactions with all available interfaces was not common since studies usually provided insights solely about actions that affected the investigated variables. Most of the studies (*88%*) provided details on the experiment location (e.g., reporting the city and country), offering the possibility of interpreting results with a more accurate consideration of local climatic conditions as well as sociological attitudes of the population [54], [114], [133]. Europe and Asia hosted most of the studies (*38%* and *35%*, respectively), followed by North America (*11%*), South America and Oceania (about *1%*).

4.1.5 Exposure feature

In the analyzed studies, the most frequent design was within-subjects (40% of studies), particularly common in lab studies, followed by between-subjects (18%) and mixed designs (15%). However, many papers did not clearly report on their study design (27%), especially in field studies, which might be due to the fact that field studies normally work with between-subjects-design as they measure existing environmental conditions without modifying them. A within-subject design in a field study would be called an intervention study. It is challenging to summarize the number and combination of tested experimental conditions in a concise manner as they vary highly across studies. For example, Huebner et al. [92] reported several conditions tested withinsubjects, with all subjects experiencing dynamic temperature variations, and two conditions experienced between subjects (two CCTs). Laurentin et al. [134] tested six conditions within-subjects (two temperatures and three light types).

What is important to notice was the lack of reporting of exposure characteristics in many studies. While some missing information can be justified by the study type (e.g., the length of exposure for each experimental condition was rarely reported in field studies due to the lack of clear exposures), others should be reported in all studies to increase replicability and better understand study results. It was the case of the total length of the experiment per participant, not reported in *82*% of field studies and *15*% of lab studies, which greatly influences the outcome of the experiment given the potential fatigue of longer experimental sessions, especially in laboratory settings. It is surprising to see that the timing of the experiment during both the day and the year was not reported in many field studies (*40*% and *25*%, respectively), and even less in lab studies in which the time of the day was not reported in about *55*% of the studies and the time of the year in *61*% of them.

The last analyzed aspect of the exposure feature is the adaptation time. More than half of the studies reported the adaptation time, especially in lab settings (Figure 12). There was a tendency in most of the laboratory studies to use 30 minutes as an adaptation time; however, the time ranged from 5 to 55 minutes among the experiments, indicating a lack of consensus regarding this parameter.



Figure 12: Adaptation time in minutes reported in the studies (y-axis) in laboratory experiments.

4.1.6 Experimental design quality

As shown in Figure 13, there were substantive gaps in reporting across all the elements of good experimental design analyzed here. Half of the reviewed papers did not report how the participants were assigned to the experimental conditions, especially in field studies where 82% did not indicate this information. The risk of bias due to participants' expectations during the experimental sessions was reduced through single-blind and double-blind procedures in 34% and 2%, respectively. The rest, mostly field studies, did not mention blinding. In IEQ studies, a procedure can be considered blind when the experimental conditions are not directly explained to participants (i.e., another goal is introduced instead of presenting the study as "the effect of x conditions on y human response"). It must be highlighted that it is very challenging to make some conditions blinded (e.g., temperature or light conditions), especially in repeated measures. Hence, a truly blind procedure might be hard to achieve in IEQ studies, especially with extreme environmental conditions.

To reach the internal validity of the results, the experimental design must account for confounding variables. The most common variables controlled during data collection involved thermal stimuli (clothing insulation, relative humidity, and air velocity), followed by illuminance. Such variables can be controlled during the experiments or in the subsequent statistical analysis. The number of studies that did not report this information is high, especially in field studies. This is a surprising result considering the more numerous confounding factors present in real buildings than those found while performing controlled experiments. A null condition was reported in *28%* of the studies, all of them developed in a laboratory setting. Depending on the type of stimuli investigated, different null conditions were used, such as comfortable indoor temperature or daylight transmitted through uncolored filters [55]. In a repeated stimulus design, the consistency of the responses to the same stimuli can be tested, which is a good practice to verify the reliability of the results [47], [67]. Yet, this approach did not seem to be common in the considered studies.



Figure 13: Relative frequencies of information about the experimental design quality reported in the considered studies by effect type and study type.

4.2 Review of study deployment and analysis

4.2.1 Data collection and processing

Table 4 shows the frequency of the studies reporting the measured environmental parameters. The thermal parameters (i.e., temperature and relative humidity) were the most frequently measured and reported for both field and lab studies. The predominance of thermal measurements is linked to the numerous experiments concerning thermal aspects. However, it must be noted that such measurement was also present in other studies (approximately in *71*% of all the reviewed studies). This outcome is potentially due to the great influence that thermal conditions play on occupants' experience of space and the relative ease of

measuring thermal parameters due to the availability of low-cost sensors [135]. The visual domain was the second most frequently measured aspect, with 47% of studies reporting illuminance values.

	Effect type			Study	Study type	
Environmental measurements	Combined	Cross-modal	Combined and cross-modal	Field	Lab	
Air temperature	26	33	21	22	58	80
Air velocity/speed	9	16	10	12	23	35
Global radiation	0	1	0	1	0	1
Globe temperature	5	4	3	4	8	12
Humidity	1	3	0	0	4	4
Local outdoor temperature	0	1	0	1	0	1
Mean radiant temperature	3	1	0	4	0	4
Operative temperature	0	0	4	0	4	4
Outdoor relative humidity	4	0	0	4	0	4
Relative humidity	18	22	17	18	39	57
Surface temperature	0	0	3	0	3	3
Wet bulb temperature	0	0	3	0	3	3
Clearness index	0	1	0	1	0	1
Correlated colour temperature	0	4	0	0	4	4
Illuminance level	21	17	17	13	42	55
Illumination intensity	1	0	0	0	1	1
Luminance	1	0	0	0	1	1
CO2 concentration	13	2	7	15	7	22
Particulate matter	4	0	0	4	0	4
Ventilation rate	0	3	1	1	3	4
Sound/noise level	12	3	3	3	15	18

Table 4: Number of reviewed studies reporting to measure the environmental parameters.

Overall, general information for reproducibility was scarcely reported in the considered papers, as shown in Figure 14. For instance, most studies did not report the frequency with which measurements were taken (78% of studies), the processing method used after data collection (59%), or the comparison between measured and design conditions (83%). These results include both field and lab studies. This lack of information on environmental measurements is a severe limitation of existing multi-domain studies. The location of the measurements was the only information that was reported more often, presented in 66% of the studies. Environmental measurements at the proximity of each participant were more common in lab studies than in field experiments, where sensors were usually deployed to measure the average room condition.



Figure 14: Relative frequency of reviewed studies reporting information on environmental stimuli measurements. The label "other" includes, e.g., per desk row, and in the corner desk.

4.2.2 Participants

Figure 15 highlights the percentage of studies reporting the participant characteristics. Overall, field studies had a higher number of participants (mean = 141) compared to laboratory studies (mean = 35). Distribution by sex was reported in most laboratory studies (91%) and approximately half of the field studies (46%). As shown in Figure 15, age is reported more than most other single characteristics (73%). The origin of the participants was reported in only 18% of the analyzed papers. The verification of physical conditions before the experiment (e.g., sleep, vision, food/alcohol/caffeine intake) was reported in more than half of the laboratory papers (55%), but rarely in field studies (11%). Indications about the subjects' health status, height, weight, and origin were reported in one-third or fewer of the papers. Finally, participants' involvement in experiments was more frequently reported in laboratory studies than in field studies across all measures. Of the measures, the description of tasks/activities was the most commonly reported (77%). In the described tasks/activities, the predominant activities were office activity (29%) and class activity (7%), while in laboratory studies, the most reported activities were reading (17%), sitting (15%), and conducting performance tests (13%). Participants' payment for taking part in the project is reported in 44% of the papers. None of the field studies foresaw a payment to the participants, while 47% of the laboratory studies remunerated the participants. Surprisingly, information on the ethical approval and whether a tailored information sheet was distributed to the participants was reported only in 21% of the analyzed studies (7% and 25% in field and laboratory studies, respectively).



Figure 15: Relative frequencies of information about participants reported in the considered studies.

While most studies provided detailed information on the number of participants, most of the described participant-related aspects seem to be underreported or not clearly stated in the papers. This leads to the risk that readers will make assumptions about certain aspects (e.g., assume that all participants were nationals from the country where the study was conducted). In future studies, researchers should report participants' characteristics in detail to clearly define to whom the study's findings apply. As best practice references, the studies that, according to our review, reported most of the relevant aspects related to participants' characteristics and their involvement were Kim and Tokura [93], Chinazzo et al. [128], Golasi et al. [136], and Wang et al. [137].

4.2.3 Statistical analysis

Among the considered publications, 44 statistical methods were used to analyze the combined and crossmodal effects. Figure 16 shows the main statistical methods used in the reviewed studies and the percentage of the studies adopting different methods. The most used statistical methods were the analysis of variance (ANOVA), linear regression, and *t*-test. The least used methods were categorized in the "other" group, which includes, for example, Mann-Whitney-U test [138], change-point regression [97], and permutation test [105]. The methods were also analyzed and categorized based on their "appropriate" use. For example, the *t*-test was deemed as "not appropriate" if multiple *t*-tests were applied directly as the primary test and not as a follow-up test of a "higher-order" test such as ANOVA. In addition, the absence of a statistical analysis was categorized as "not appropriate".

In Figure 16, it can be seen that mixed-effect models (also commonly referred to as multilevel or hierarchical models), although used, are not applied often. However, these models are valuable tools developed to address the violation of the independence assumption (required by traditional statistical analyzes such as ANOVA and ordinary least-squares regression). This assumption is violated whenever the observations are nested and/or clustered. For example, nested and/or clustered observations can arise from temporal and spatial autocorrelation. In the context of multi-domain studies, an application of the statistical methods used (14 out of 236) perform preliminary tests to assess the collected data (e.g., Shapiro-Wilk to check the data distribution).

Adopting a specific statistical method was justified in only 40% of the papers. This result shows that the authors either assumed the readers could infer the statistical reasoning or considered it not an important aspect of the manuscript.



Figure 16: The percentage of different statistical methods used (thereinto, ANOVA includes repeated measures ANOVA, one-way ANOVA, two-way ANOVA, three-way ANOVA, factorial ANOVA, mixed model ANOVA, Welch's ANOVA, ANCOVA, MANOVA, MANCOVA; generalized estimating equations includes a

model assuming a binomial distribution with logistic link, a model assuming a normal distribution with identity link function; correlation includes partial correlation analysis, Pearson's correlation coefficient, Spearman's rank correlation coefficient).

Only 4% of the studies reported a power analysis, both in field and lab studies. This aspect attracts a quite interesting outcome because, in most of the cases, either the experimental design was not entirely reported in the publication, or the minimum number of observations of an experiment was simply stated but not justified.

Despite its importance, only 22% of the studies explicitly reported the effect size, both in field and lab studies. It means that most of the studies referred only to statistical significance testing to evaluate their results. While the domain outcomes of the reviewed studies are of paramount importance and contribute to the development of the field knowledge, unfortunately, the description of the statistical methodology was often approximate or missing. In most cases, statistical methods were applied without a dedicated description of data acquisition, analysis, curation, storage, and usage. In some cases, even validity and representativity of outcomes cannot be inferred due to missing information on data accuracy, completeness, consistency, relevance, and uniformity.

4.3 Review of study outcome

4.3.1 Reporting results

To our knowledge, the definitions of the results and results reporting style are described for the first time in this study. Therefore, it is difficult to analyze the presence of such information in the considered papers. Most of the time, we observed that results were reported in an incomplete way (e.g., only statistically significant results were described, or not all the effects of all the considered stimuli were reported). In addition, the direction of the effect in cross-modal studies and the type of combined effect were rarely stated. Finally, a graphical representation of the cross-modal and combined effects was reported in only a few studies (e.g., [130], [140], [141]).

4.3.2 Study discussion and conclusion

Figure 17 shows the relative frequency of data-informed conclusions and frequency of reporting future studies, study limitations, mechanism explanations, and practical implications.



Figure 17: Relative frequencies of information about conclusions and discussions reported in the considered studies, according to the effect type and study type.

Most of the articles (88%) presented conclusions based on data, while the remainder seems to be speculative. Such distribution was similar across effect types, but not across laboratory versus field studies: the percentage of conclusions based on data is higher in research carried out in laboratory than in-field, 90% against 82%, respectively. This difference can be related to the opportunity to control some potential confounding factors in laboratory that are not always detectable in a real case.

Roughly half of the studies did not identify future studies, limitations or practical implications, although there were some differences within the sub-types. For instance, limitations were related to the study type, with some of them only relevant for field studies (e.g., limited control) and others for lab studies (e.g., limited exposure time). Also, in studies carried out in the field, the percentage of papers with mechanisms explanation was lower than those in a laboratory, 64% and 86%, respectively. The results of these studies can be influenced by variables that cannot be controlled, making it difficult to reach an unambiguous result [142]. However, in many cases, authors were able to provide a description of the mechanism [97], [121], even with a step forward from the initial hypothesis [143]. In research carried out in laboratory, the use of a wide range of sensors and the control of variables allowed some authors to explain the results while also considering a physiological point of view [90], [93], [127], [144], [145]. In other studies, authors deepened the effect of specific stimuli on the perception of comfort [146] or on performance [116].

In the articles presenting future studies several authors propose further research in the form of new configurations of existing independent variables or expanding their ranges and identifying new parameters

of the same independent variables or new environmental factors. These future developments are more visible in cross-modal analysis where the interaction among the independent variables are often partials. Other future studies include the investigation of different size, age, and origins of the participant group, of new building typologies or settings, different exposure time and experimental length, and of new computational models.

Practical implications were not reported in 43% of the analyzed papers. Examples of good descriptions of practical implications can be found in several papers [85], [116], [147].

5 Conclusions and recommendations

5.1 Key observations

The premise of this paper, as well as that of most of the work it assesses, is that people's experience of and response to indoor environmental conditions involve multiple domains. Nonetheless, the bulk of regulatory resources for building professionals is single-domain. This may be attributed to the complexity of multidomain exposures and the mechanisms by which they influence buildings' occupants and implies a need for increased multi-domain research. Moreover, while additional studies are necessary, they are not sufficient for progress in this area. To achieve a deeper understanding of the nature of multi-domain exposure implications for occupants' health, comfort, and productivity, the related research must also satisfy several qualitative requirements. Such research must be designed, conducted, and documented in a systematic and transparent manner, such that the results are reproducible and suitable for meta-analyses. This paper's assessment of the past research efforts in this area identified several shortcomings, notwithstanding the studies' general relevance, importance, and in some cases, pioneering significance. Therefore, as the following summary of the observed key challenges implies, necessary quality improvements of future multi-domain research need to address both the studies' design, deployment, and reporting. The key observations are divided into those related to each aspect of the critical review and those associated with a transversal analysis of the results.

Key observations of the critical review:

- Dependent variables: existing studies mainly focused on the investigation of subjective perceptual responses, most commonly through numeric scales (including 3-point, 5-point, and 7-point scales) to capture test participants' responses regarding perception, comfort, satisfaction, and preference. At times, a different number of points and different labels were used, even though the same assessment category was involved. This, as well as the inconsistent use of dimensions in analogue scales, disables the comparison of results from different studies and poses a problem for conducting large-scale meta-analyses. Performance, behavior, and physiology are still untapped research venues that could lead to new breakthroughs in multi-domain studies.
- Independent variables: thorough documentation of the prevailing values of the independent variables is
 a basic requirement for doing multi-domain studies. Most reviewed research generally provided such
 documentation, even though the types and design values in some same-modality independent variables
 were not reported. Future comparative studies and meta-analyses could benefit from a more consistent

choice of the design values for independent variables. It is recommended to always adopt numerical design values which will enable future replication studies and meta-analyses. Moreover, documentation of non-environmental independent variables (e.g., relevant information on participants and outdoor conditions) could strengthen the interpretation scope of the studies' findings.

- Research hypothesis: the comprehension and utility of results from experimental research, would be
 arguably higher when research hypotheses are explicitly stated, including their "direction." Surprisingly,
 about 40% of the laboratory studies and 60% of field studies did not state the research hypotheses.
 Whenever the hypothesis was stated, only a fraction indicated the direction a very small one in field
 studies.
- Setting features: the description of the settings is a key aspect, yet not sufficiently reported in most reviewed studies. Such information includes building location, type, space layout, HVAC, building elements (e.g., windows and shades), control interfaces, and lighting systems. Consequently, confounding factors and potential cross-modal effects of other features could be overlooked.
- Exposure features: in many instances, characteristics of the exposure situation (e.g., type, timing, and length of exposure) were not reported in many studies. This represents a problem when trying to replicate a study or include its findings in an overarching meta-analysis of multiple investigations. The analysis of previous studies also shows a lack of consistency regarding the adaptation time, which might influence the results of the experiments.
- Experimental design quality: the consideration of experimental design criteria/principles is of critical importance to assure high standards of scientific quality. The reviewed studies were analyzed regarding randomization, counterbalance, experimental procedure (single or double blind, at least when explaining the goal of the study), experimental and statistical confounding variables, reporting of null condition, and repetition of certain experimental conditions. The reviewed studies did not consistently report these aspects. For instance, *82%* of the reviewed field studies did not include information on how participants were assigned to specific experimental conditions.
- Data collection and processing: the measurement and data processing of environmental conditions (not only explicitly targeted independent variables, but other elements of the experiments' boundary conditions) in the course of multi-domain studies is of high importance, especially in view of reproducibility criteria. A sufficient level of reporting on environmental measurements and their analysis was provided only in a small number of multi-domain studies. This implies the need for streamlined assessment and reporting procedures for both environmental conditions and human responses.
- Participants: studies involving human participants should provide detailed information on their distribution, relevant personal characteristics, and their role/involvement in the experiments. The assessment of the reviewed studies regarding this criterion yields a rather unsatisfactory picture. Aside from their number (almost always reported), essential information regarding participants was either underreported or not clearly stated. This circumstance undermines the credibility of the studies concerning, among other things, their representativeness and generalizability. In addition, information

about the ethical approval and the related documentation (consent and information sheet) is lacking in almost 80% of the publications, raising concerns about the ethics of the performed studies.

- Statistical analysis: a considerable number (N = 44) of different statistical methods were employed in the reviewed studies (mostly ANOVA). Among the formal tests of the distribution of the data, it is striking that the Shapiro-Wilk test, although recommended among the possible formal tests (e.g., Kolmogorov-Smirnov, Lilliefors and Anderson-Darling) [148], is only used in 5.9% of the cases, while the tests where the normality distribution should be verified are much more (more than 55% if we consider the sum of the papers where ANOVA, t-test and linear regression are used). Future studies should adopt a statistical approach that first checks the distribution of the sample and then applies tests where the normal distribution is an underlying assumption that corresponds to the main hypothesis. About 60% of the studies conducted a power analysis (4%) and reported effect sizes (22%). This hampers the reproducibility of experiments, feasibility of meta-analyses, and review of collective insights.
- Reporting results: the reporting of the results in published studies is inconsistent and sometimes incomplete (e.g., not all the results are reported, graphical representations are missing). The use of the same terminology to describe the type of effect investigated (i.e., cross-modal or combined) and their results is paramount to conduct future meta-analyses on multi-domain studies. For cross-modal effects, the direction of the effect (i.e., positive, negative or no effect) must be reported for each of the levels of the considered stimuli. For combined effects, the results can be described following the terminology described in the ASHRAE Guideline 10-2016 [39]. In future studies, researchers are invited to describe the results comprehensively and adopt the suggested reporting style for both cross-modal and combined effects (including terminology and the suggested tabular representation). In addition, considering that understanding cross-modal and combined effects solely based on the outcome of statistical analysis (e.g., model coefficients) may be a complex task for those without a solid background in statistics, we advise the complimentary usage of as simple as possible graphical representations of the cross-modal and combined effects (as depicted in Figure 1).
- Study discussion and conclusions: despite always presenting the conclusions (mostly based on the data), a large part of the considered papers does not include future studies (43%), limitations (55%), explanation of the mechanisms behind the results (69%), and practical implications (43%). The lack of such information reduces the possibility to advance the knowledge on the topic and understanding its relevance for people and society.

Transversal key observations:

- Multi-domain studies have been reported to rarely carryover previous studies' findings and to lack foundational theories to formulate and test research hypothesis [40]. Therefore, introductory sections were not reviewed in this study. Future multi-domain investigations should build upon previous findings to generate theoretical assumptions or start from theory-based motivations based on human perceptual and behavioral processes to formulate their research hypotheses.
- Field studies were less likely to report features (e.g., site, location, equipment etc.), hypotheses, assumptions, and variables. Laboratory and field experiments have intrinsic differences, but this is not a

justification for leaving out the information required for valid, generalizable, replicable, and reproducible studies.

• The low fraction of the studies that conducted a power analysis, followed a good experimental design, described sufficient population characteristics, and effect sizes, suggests a possible replication crisis identified elsewhere [65], [66]. The adjacent field of psychology serves as a reservoir of a decade's worth of scientific discussion and proposed methodological improvements (e.g., pre-registration prior to the start of the study, transparent data processing practices, and reporting effects sizes) [149] that should serve as example in future studies. It has been suggested that the social-structural factors that contribute to the replication crisis are not limited to psychology [150] and may apply to other fields [151].

It can be concluded that multi-domain studies were often not thoroughly documented and reported in a systematic and detailed manner or did not adhere to paramount research quality criteria. These issues may be rooted in the lack of robust schemes for conceptualizing and reporting both cross-modal and combined effects. This study aimed to establish sound guidelines and recommendations for designing, deploying and reporting multi-domain studies for addressing this challenge and foster more structured and coherent future multi-domain studies. Standardizing methods and reporting formats for multi-domain studies will enhance the rigor in reviewing these studies and enable future meta-analyses.

5.2 Future multi-domain studies

Although the provided recommendations were developed for investigations about (indoor) environmental stimuli, their application can be extended to studies investigating personal (e.g., sex, age, culture) and contextual aspects (e.g., time of the day, season, building typology, control opportunities). These aspects can be considered as additional domains influencing human responses in multi-domain studies [40].

The publications and context covered by this work outline momentum towards characterizing the multidimensional impact of the built environment on occupants. This foundation and the lessons learned provide the context for future work. Research in this area going forward could focus on filling the gaps of information about indoor environmental stimuli and human responses through innovative technologies and methods. For example, the use of continuous, field-based biosensing methods, like those being developed in mobile health research, can enable the detection of a broader range of human physiological responses [152]. The human response can be captured in a more scalable way using innovative interfaces that are integrated specifically into mobile devices and wearables [153]. There are, moreover, relatively new statistical techniques for testing causal claims relevant to multi-domain studies from a properly designed field study setup. For an overview of some of the recent developments in techniques, see [154]. Many of the proposed quality criteria are complementary to the rigorous study design required for a causal framework. The quality criteria summarized in Figure 2 and their description in section 3 can therefore serve as guidance for study design and reporting in future multi-domain studies.

During the reviewing process, we uncovered a wide range of possible interdisciplinary research opportunities through collaboration with the research communities of machine learning, building controls, wellness, public health, and real estate communities, as well as between research fields such as psychology, physiology, engineering, and architecture. The methodological best quality criteria uncovered during the review process can be further enhanced by these interdisciplinary collaborations to create hybrid approaches that accelerate

the transfer of IEQ research results into actionable outputs, such as the amendment of building design and operation standards and guidelines. Future work may also consider the increasingly dynamic nature in which buildings are used, especially in office spaces where a larger diversity of activities can occur due to the enhanced workplace flexibility.

Declarations of competing interest

There are no known conflicts of interest.

Acknowledgements

This work was conducted within the framework of IEA-EBC Annex 79 "Occupant-Centric Building Design and Operation" (<u>https://annex79.iea-ebc.org/</u>).

Matteo Favero would like to thank the Research Centre on Zero Emission Neighbourhoods in Smart Cities (FME ZEN, Grant no. 257660) and the Research Council of Norway (Norges Forskingsrådet) for their support. Stephanie Gauthier would like to thank the support of UK Engineering and Physical Sciences Research Council under Grant EP/T023074/1.

RunaT. Hellwig was supported by the Obelske Familiefond.

Ilaria Pigliautile acknowledgements go to the PON project "Red-To-Green" financed by the Italian Ministry of Research (D.M. 1062 PON Ricerca e Innovazione).

Anna Laura Pisello's acknowledgements go to the PRIN 2017 (Prot. 20172FSCH4) project entitled "Towards the NEXT generation of multiphysics and multidomain environmental COMfort models: theory elaboration and validation experiment".

Isabel Mino-Rodriguez is funded by the German Federal Ministry for Economic Affairs and Climate Action (BMWK), promotional reference 03EN1002A and 03EN1002B.

Marcel Schweiker is funded by a research grant (21055) by VILLUM FONDEN.

Rune Korsholm Andersen acknowledges funding by the Danish EUDP programme, project: 64018-0558

References

- [1] N. E. Klepeis *et al.*, "The National Human Activity Pattern Survey (NHAPS): a resource for assessing exposure to environmental pollutants," *J. Expo. Sci. Environ. Epidemiol.*, vol. 11, no. 3, p. 231, 2001.
- [2] P. M. Bluyssen, "Towards new methods and ways to create healthy and comfortable buildings," *Build. Environ.*, vol. 45, no. 4, pp. 808–818, Apr. 2010, doi: 10.1016/j.buildenv.2009.08.020.
- [3] F. Stazi, F. Naspi, and M. D'Orazio, "A literature review on driving factors and contextual events influencing occupants' behaviours in buildings," *Build. Environ.*, vol. 118, pp. 40–66, Jun. 2017, doi: 10.1016/j.buildenv.2017.03.021.
- [4] M. Aries, M. Aarts, and J. van Hoof, "Daylight and health: A review of the evidence and consequences for the built environment," *Light. Res. Technol.*, vol. 47, no. 1, pp. 6–27, Feb. 2015, doi: 10.1177/1477153513509258.
- [5] M. Amundadottir, S. Lockley, and M. Andersen, "Unified framework to evaluate non-visual spectral effectiveness of light for human health," *Light. Res. Technol.*, vol. 49, no. 6, pp. 673–696, Oct. 2017, doi: 10.1177/1477153516655844.
- [6] A. Leaman and B. Bordass, "Are users more tolerant of 'green' buildings?," *Build. Res. Inf.*, vol. 35, no. 6, pp. 662–673, Sep. 2007, doi: 10.1080/09613210701529518.
- [7] D. Clements-Croome, Creating the Productive Workplace. Taylor & Francis, 2006.
- [8] D. Clements-Croome and L. Baizhan, "Productivity and indoor environment," in *Proceedings of Healthy Buildings*, 2000, vol. 1, pp. 629–634. Accessed: Nov. 09, 2015. [Online]. Available: http://www.senseair.se/wp-content/uploads/2011/05/31.pdf
- [9] A. Soriano, M. W. Kozusznik, J. M. Peiró, and C. Mateo, "Mediating role of job satisfaction, affective well-being, and health in the relationship between indoor environment and absenteeism: Work patterns matter!," *Work*, vol. 61, no. 2, pp. 313–325, Jan. 2018, doi: 10.3233/WOR-182802.
- [10] J. A. Veitch, K. E. Charles, K. M. J. Farley, and G. R. Newsham, "A model of satisfaction with openplan office conditions: COPE field findings," *J. Environ. Psychol.*, vol. 27, no. 3, pp. 177–189, Sep. 2007, doi: 10.1016/j.jenvp.2007.04.002.

- [11] G. Newsham, J. Brand, C. Donnelly, J. Veitch, M. Aries, and K. Charles, "Linking indoor environment conditions to job satisfaction: a field study," *Build. Res. Inf.*, vol. 37, no. 2, pp. 129–147, Apr. 2009, doi: 10.1080/09613210802710298.
- [12] G. Chinazzo, "Investigating the indoor environmental quality of different workplaces through webscraping and text-mining of Glassdoor reviews," *Build. Res. Inf.*, vol. 49, no. 6, pp. 695–713, Aug. 2021, doi: 10.1080/09613218.2021.1908879.
- [13] Z. Yu, B. C. M. Fung, F. Haghighat, H. Yoshino, and E. Morofsky, "A systematic procedure to study the influence of occupant behavior on building energy consumption," *Energy Build.*, vol. 43, no. 6, pp. 1409–1417, 2011, doi: 10.1016/j.enbuild.2011.02.002.
- [14] H. B. Rijal, P. Tuohy, M. A. Humphreys, J. F. Nicol, A. Samuel, and J. Clarke, "Using results from field surveys to predict the effect of open windows on thermal comfort and energy use in buildings," *Energy Build.*, vol. 39, no. 7, pp. 823–836, Jul. 2007, doi: 10.1016/j.enbuild.2007.02.003.
- [15] O. Guerra-Santin and L. Itard, "Occupants' behaviour: determinants and effects on residential heating consumption," *Build. Res. Inf.*, vol. 38, no. 3, pp. 318–338, Jun. 2010, doi: 10.1080/09613211003661074.
- [16] W. Fisk, O. Seppanen, and O. Fisk Seppanen, "Providing Better Indoor Environmental Quality Brings Economic Benefits," 2007.
- [17] J. G. Allen and J. D. Macomber, *Healthy buildings: How indoor spaces drive performance and productivity*. Harvard University Press, 2020.
- [18] P. O. Fanger, Thermal comfort: Analysis and applications in environmental engineering. Copenhagen: Danish Technical Press, 1970. [Online]. Available: http://books.google.ch/books?id=S0FSAAAAMAAJ
- [19] J. Wienold, Daylight glare in offices. Fraunhofer-Verlag, 2009.
- [20] ASHRAE, "ANSI/ASHRAE Standard 62.1-2016: Ventilation for Acceptable Indoor Air Quality," American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, USA, 2016.
- [21] ASHRAE, "ANSI/ASHRAE 55-2020: Thermal Environmental Conditions for Human Occupancy," American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, USA, Standard 55-2020, 2020.
- [22] H. Levin and Emmerich, "Dissecting Interactions Among IEQ Factors," ASHRAE J., vol. 55, no. 9, pp. 66–72, 2013.
- [23] ASHRAE, "ANSI/ASHRAE Guideline 10P: Interactions affecting the achievement of acceptable indoor environments," American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, USA, 2011.
- [24] S. Shimojo and L. Shams, "Sensory modalities are not separate modalities: plasticity and interactions," *Curr. Opin. Neurobiol.*, vol. 11, no. 4, pp. 505–509, 2001.
- [25] R. B. Welch, Intersensory interactions, vol. 1. New York: John Wiley and Sons, 1986.
- [26] J. Driver and C. Spence, "Multisensory perception: Beyond modularity and convergence," Curr. Biol., vol. 10, no. 20, pp. R731–R735, Oct. 2000, doi: 10.1016/S0960-9822(00)00740-5.
- [27] D. Alais, F. N. Newell, and P. Mamassian, "Multisensory processing in review: from physiology to behaviour," *Seeing Perceiving*, vol. 23, no. 1, pp. 3–38, 2010.
- [28] G. M. Wyburn, *Human Senses and Perception*. University of Toronto Press, 1964.
- [29] G. Calvert, C. Spence, and B. E. Stein, *The handbook of multisensory processes*. MIT press, 2004.
- [30] M. O. Ernst and H. H. Bülthoff, "Merging the senses into a robust percept," *Trends Cogn. Sci.*, vol. 8, no. 4, pp. 162–169, 2004.
- [31] P. Bertelson and B. De Gelder, "The psychology of multimodal perception," *Crossmodal Space Crossmodal Atten.*, pp. 141–177, 2004.
- [32] E. T. Hall, *The hidden dimension*, vol. 609. Garden City, NY: Doubleday, 1966.
- [33] A. Rapoport and S. el Sayegh, *Culture, architecture, and design*. Locke science publishing Company Washington, 2005.
- [34] J. Ward, "Synesthesia," *Annu. Rev. Psychol.*, vol. 64, no. 1, pp. 49–75, 2013, doi: 10.1146/annurev-psych-113011-143840.
- [35] C. Karmann, S. Schiavon, and E. Arens, "Percentage of commercial buildings showing at least 80% occupant satisfied with their thermal comfort," presented at the Proceedings of 10th WIndsor Conference: Rethinking Comfort, Windsor, UK, Apr. 2018.
- [36] C. Huizenga, S. Abbaszadeh, L. Zagreus, and E. A. Arens, "Air quality and thermal comfort in office buildings: Results of a large indoor environmental quality survey," *Proceeding Healthy Build. 2006*, vol. 3, Jan. 2006, Accessed: Oct. 22, 2015. [Online]. Available: http://escholarship.org/uc/item/7897g2f8
- [37] G. Brager and L. Baker, "Occupant satisfaction in mixed-mode buildings," *Build. Res. Inf.*, vol. 37, no. 4, pp. 369–380, Aug. 2009, doi: 10.1080/09613210902899785.

- [38] L. T. Graham, T. Parkinson, and S. Schiavon, "Lessons learned from 20 years of CBE's occupant surveys," *Build. Cities*, vol. 2, no. 1, Art. no. 1, Feb. 2021, doi: 10.5334/bc.76.
- [39] ASHRAE, "ANSI/ASHRAE Guideline 10P: Interactions affecting the achievement of acceptable indoor environments," American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, USA, 2016.
- [40] M. Schweiker *et al.*, "Review of multi-domain approaches to indoor environmental perception and behaviour," *Build. Environ.*, vol. 176, p. 106804, Jun. 2020, doi: 10.1016/j.buildenv.2020.106804.
- [41] S. Torresin, G. Pernigotto, F. Cappelletti, and A. Gasparella, "Combined effects of environmental factors on human perception and objective performance: A review of experimental laboratory works," *Indoor Air*, vol. 28(4), pp. 525–538, 2018.
- [42] H. Wu, Y. Wu, X. Sun, and J. Liu, "Combined effects of acoustic, thermal, and illumination on human perception and performance: A review," *Build. Environ.*, vol. 169, p. 106593, 2020.
 [43] J. Van Hoof, M. Mazej, and J. L. Hensen, "Thermal comfort: research and practice," *Front. Biosci.*, vol.
- [43] J. Van Hoof, M. Mazej, and J. L. Hensen, "Thermal comfort: research and practice," *Front. Biosci.*, vol. 15, no. 2, pp. 765–788, 2010.
- [44] A. Mahdavi et al., "Necessary Conditions for Multi-Domain Indoor Environmental Quality Standards," Sustainability, vol. 12, no. 20, Art. no. 20, Jan. 2020, doi: 10.3390/su12208439.
- [45] R. F. Rupp, N. G. Vásquez, and R. Lamberts, "A review of human thermal comfort in the built environment," *Energy Build.*, vol. 105, pp. 178–205, 2015.
- [46] S. C. Landis *et al.*, ⁴A call for transparent reporting to optimize the predictive value of preclinical research," *Nature*, vol. 490, no. 7419, Art. no. 7419, Oct. 2012, doi: 10.1038/nature11556.
- [47] H. Coolican, *Research Methods and Statistics in Psychology*, 6th ed. Psychology Press, 2014. [Online]. Available: https://books.google.ch/books?hl=it&Ir=&id=8DJFAwAAQBAJ&oi=fnd&pg=PP1&dq=hugh+coolican &ots=BSeeshl8eW&sig=--W_waAGHMrmDl4VsNsd6ZuOFaA
- [48] A. L. Pisello et al., "Test rooms to study human comfort in buildings: A review of controlled experiments and facilities," *Renew. Sustain. Energy Rev.*, vol. 149, p. 111359, Oct. 2021, doi: 10.1016/j.rser.2021.111359.
- [49] A. K. Mishra and M. Ramgopal, "Field studies on human thermal comfort An overview," Build. Environ., vol. 64, pp. 94–106, Jun. 2013, doi: 10.1016/j.buildenv.2013.02.015.
- [50] R. Jacoby Cureau *et al.*, "Bridging the gap from test rooms to field-tests for human indoor comfort studies: A critical review of the sustainability potential of living laboratories," *Energy Res. Soc. Sci.*, vol. 92, p. 102778, Oct. 2022, doi: 10.1016/j.erss.2022.102778.
- [51] G. Chinazzo, L. Pastore, J. Wienold, and M. Andersen, "A field study investigation on the influence of light level on subjective thermal perception in different seasons," p. 12, 2018.
- [52] Z. Wang *et al.*, "Individual difference in thermal comfort: A literature review," *Build. Environ.*, vol. 138, pp. 181–193, Jun. 2018, doi: 10.1016/j.buildenv.2018.04.040.
- [53] M. Schweiker, G. M. Huebner, B. R. M. Kingma, R. Kramer, and H. Pallubinsky, "Drivers of diversity in human thermal perception – A review for holistic comfort models," *Temperature*, vol. 5, no. 4, pp. 308–342, Oct. 2018, doi: 10.1080/23328940.2018.1534490.
- [54] A. Brambilla, W. Hu, R. Samangouei, R. Cadorin, and W. Davis, "How correlated colour temperature manipulates human thermal perception and comfort," *Build. Environ.*, vol. 177, p. 106929, Jun. 2020, doi: 10.1016/j.buildenv.2020.106929.
- [55] G. Chinazzo, J. Wienold, and M. Andersen, "Combined effects of daylight transmitted through coloured glazing and indoor temperature on thermal responses and overall comfort," *Build. Environ.*, vol. 144, pp. 583–597, Oct. 2018, doi: 10.1016/j.buildenv.2018.08.045.
- [56] W. Yang and H. J. Moon, "Combined effects of acoustic, thermal, and illumination conditions on the comfort of discrete senses and overall indoor environment," *Build. Environ.*, vol. 148, pp. 623–633, Jan. 2019, doi: 10.1016/j.buildenv.2018.11.040.
- [57] S. A. Sadeghi, P. Karava, I. Konstantzos, and A. Tzempelikos, "Occupant interactions with shading and lighting systems using different control interfaces: A pilot field study," *Build. Environ.*, vol. 97, pp. 177–195, Feb. 2016, doi: 10.1016/j.buildenv.2015.12.008.
- [58] J. K. Day et al., "A review of select human-building interfaces and their relationship to human behavior, energy use and occupant comfort," *Build. Environ.*, vol. 178, p. 106920, Jul. 2020, doi: 10.1016/j.buildenv.2020.106920.
- [59] R. T. Hellwig, "Perceived control in indoor environments: a conceptual approach," *Build. Res. Inf.*, vol. 43, no. 3, pp. 302–315, May 2015, doi: 10.1080/09613218.2015.1004150.
- [60] L. M. Huiberts, K. C. H. J. Smolders, and Y. A. W. D. Kort, "Seasonal and time-of-day variations in acute non-image forming effects of illuminance level on performance, physiology, and subjective wellbeing," *Chronobiol. Int.*, vol. 34, no. 7, pp. 827–844, Aug. 2017, doi: 10.1080/07420528.2017.1324471.
- [61] M. G. Kent, "Temporal effects on glare response from daylight," Build. Environ., p. 16, 2017.

- [62] M. Vellei, G. Chinazzo, K.-M. Zitting, and J. Hubbard, "Human thermal perception and time of day: A review," *Temperature*, vol. 0, no. 0, pp. 1–22, Oct. 2021, doi: 10.1080/23328940.2021.1976004.
- [63] A. Wagner, E. Gossauer, C. Moosmann, Th. Gropp, and R. Leonhart, "Thermal comfort and workplace occupant satisfaction—Results of field studies in German low energy office buildings," *Energy Build.*, vol. 39, no. 7, pp. 758–769, Jul. 2007, doi: 10.1016/j.enbuild.2007.02.013.
- [64] T. Goto, J. Toftum, R. de Dear, and P. O. Fanger, "Thermal sensation and thermophysiological responses to metabolic step-changes," *Int. J. Biometeorol.*, vol. 50, no. 5, pp. 323–332, May 2006, doi: 10.1007/s00484-005-0016-5.
- [65] Open Science Collaboration, "Estimating the reproducibility of psychological science," *Science*, vol. 349, no. 6251, p. aac4716, Aug. 2015, doi: 10.1126/science.aac4716.
- [66] M. Baker, "1,500 scientists lift the lid on reproducibility," *Nature*, vol. 533, no. 7604, pp. 452–454, May 2016, doi: 10.1038/533452a.
- [67] A. Field, J. Miles, and Z. Field, *Discovering Statistics Using R. SAGE Publications*, 2012.
- [68] A. Dean, D. Voss, and D. Draguljić, Design and analysis of experiments. Springer, 2017.
- [69] M. Schweiker *et al.*, "Evaluating assumptions of scales for subjective assessment of thermal environments – Do laypersons perceive them the way, we researchers believe?," *Energy Build.*, vol. 211, p. 109761, Mar. 2020, doi: 10.1016/j.enbuild.2020.109761.
- [70] P. Voigt and A. Von dem Bussche, "The eu general data protection regulation (gdpr)," Pract. Guide 1st Ed Cham Springer Int. Publ., vol. 10, no. 3152676, pp. 10–5555, 2017.
- [71] A. P. Siddaway, A. M. Wood, and L. V. Hedges, "How to do a systematic review: a best practice guide for conducting and reporting narrative reviews, meta-analyses, and meta-syntheses," *Annu. Rev. Psychol.*, vol. 70, pp. 747–770, 2019.
- [72] H. Cooper, L. V. Hedges, and J. C. Valentine, *The Handbook of Research Synthesis and Meta-Analysis*. Russell Sage Foundation, 2019.
- [73] F. Faul, E. Erdfelder, A. Buchner, and A.-G. Lang, "Statistical power analyses using G*Power 3.1: Tests for correlation and regression analyses," *Behav. Res. Methods*, vol. 41, no. 4, pp. 1149–1160, Nov. 2009, doi: 10.3758/BRM.41.4.1149.
- [74] M. B. Nuijten and J. R. Polanin, "'statcheck': Automatically detect statistical reporting inconsistencies to increase reproducibility of meta-analyses," *Res. Synth. Methods*, vol. 11, no. 5, pp. 574–579, 2020, doi: 10.1002/jrsm.1408.
- [75] S. Duval and R. Tweedie, "Trim and Fill: A Simple Funnel-Plot–Based Method of Testing and Adjusting for Publication Bias in Meta-Analysis," *Biometrics*, vol. 56, no. 2, pp. 455–463, 2000, doi: 10.1111/j.0006-341X.2000.00455.x.
- [76] R. C. M. van Aert, J. M. Wicherts, and M. A. L. M. van Assen, "Conducting Meta-Analyses Based on p Values: Reservations and Recommendations for Applying p-Uniform and p-Curve," *Perspect. Psychol. Sci.*, vol. 11, no. 5, pp. 713–729, Sep. 2016, doi: 10.1177/1745691616650874.
- [77] J. Langevin, "Longitudinal dataset of human-building interactions in U.S. offices," *Sci. Data*, vol. 6, no. 1, p. 288, Nov. 2019, doi: 10.1038/s41597-019-0273-5.
- [78] M. Schweiker *et al.*, "The Scales Project, a cross-national dataset on the interpretation of thermal perception scales," *Sci. Data*, vol. 6, no. 1, p. 289, Nov. 2019, doi: 10.1038/s41597-019-0272-6.
- [79] T. D. Sterling, "Publication Decisions and Their Possible Effects on Inferences Drawn from Tests of Significance--Or Vice Versa," J. Am. Stat. Assoc., vol. 54, no. 285, pp. 30–34, 1959, doi: 10.2307/2282137.
- [80] J. D. Perezgonzalez, "Fisher, Neyman-Pearson or NHST? A tutorial for teaching data testing," Front. Psychol., vol. 6, 2015, Accessed: Apr. 28, 2022. [Online]. Available: https://www.frontiersin.org/article/10.3389/fpsyg.2015.00223
- [81] M. S. Andargie and E. Azar, "An applied framework to evaluate the impact of indoor office environmental factors on occupants' comfort and working conditions," *Sustain. Cities Soc.*, vol. 46, p. 101447, Apr. 2019, doi: 10.1016/j.scs.2019.101447.
- [82] I. Balazova, G. Clausen, and D. P. Wyon, "The influence of exposure to multiple indoor environmental parameters on human perception, performance and motivation," p. 8, 2007.
- [83] M. C. Lee, K. W. Mui, L. T. Wong, W. Y. Chan, E. W. M. Lee, and C. T. Cheung, "Student learning performance and indoor environmental quality (IEQ) in air-conditioned university teaching rooms," *Build. Environ.*, vol. 49, pp. 238–244, Mar. 2012, doi: 10.1016/j.buildenv.2011.10.001.
- [84] D. Tiller, L. Wang, A. Musser, and M. Radik, "AB-10-017: Combined effects of noise and temperature on human comfort and performance (1128-RP)," *Archit. Eng. -- Fac. Publ.*, Jan. 2010, [Online]. Available: https://digitalcommons.unl.edu/archengfacpub/40
- [85] T. Witterseh, D. P. Wyon, and G. Clausen, "The effects of moderate heat stress and open-plan office noise distraction on SBS symptoms and on the performance of office work," *Indoor Air*, vol. 14, no. s8, pp. 30–40, Dec. 2004, doi: 10.1111/j.1600-0668.2004.00305.x.

- [86] I. Knez and S. Hygge, "Irrelevant speech and indoor lighting: effects on cognitive performance and self-reported affect," *Appl. Cogn. Psychol.*, vol. 16, no. 6, pp. 709–718, 2002, doi: 10.1002/acp.829.
- [87] L. Mølhave, Z. Liu, A. H. Jørgensen, O. F. Pedersen, and S. K. Kjægaard, "Sensory And Physiological Effects On Humans Of Combined Exposures To Air Temperatures And Volatile Organic Compounds," *Indoor Air*, vol. 3, no. 3, pp. 155–169, 1993, doi: 10.1111/j.1600-0668.1993.t01-1-00002.x.
- [88] A. K. Melikov and J. Kaczmarczyk, "Air movement and perceived air quality," *Build. Environ.*, vol. 47, pp. 400–409, Jan. 2012, doi: 10.1016/j.buildenv.2011.06.017.
- [89] A. Kabanshi, H. Wigö, M. K. van de Poll, R. Ljung, and P. Sörqvist, "The Influence of Heat, Air Jet Cooling and Noise on Performance in Classrooms," *Int. J. Vent.*, vol. 14, no. 3, pp. 321–332, Dec. 2015, doi: 10.1080/14733315.2015.11684089.
- [90] H. E. Kim and H. Tokura, "Influence of Light Intensities on Dressing Behavior in Elderly People," J. Physiol. Anthropol. Appl. Human Sci., vol. 19, no. 1, pp. 13–19, 2000, doi: 10.2114/jpa.19.13.
- [91] C. Kim and H. Tokura, "Influence of Two Different Light Intensities from 16:00 to 20:30 Hours on Evening Dressing Behavior in the Cold," 2007.
- [92] G. M. Huebner, D. T. Shipworth, S. Gauthier, C. Witzel, P. Raynham, and W. Chan, "Saving energy with light? Experimental studies assessing the impact of colour temperature on thermal comfort," *Energy Res. Soc. Sci.*, vol. 15, pp. 45–57, May 2016, doi: 10.1016/j.erss.2016.02.008.
- [93] H. Eun Kim and H. Tokura, "Influence of different light intensities during the daytime on evening dressing behavior in the cold," *Physiol. Behav.*, vol. 58, no. 4, pp. 779–783, Oct. 1995, doi: 10.1016/0031-9384(95)00129-7.
- [94] A. Hettiarachchi and R. Emmanuel, "Colour as a psychological agent to manipulate perceived indoor thermal environment for effective energy usage; cases implemented in Sri Lanka.," Edinburgh, 2017, p. 9.
- [95] G. Giakoumis and S. Gauthier, "Understanding window behaviour in a mixed-mode buildings and the impact on energy performance," Sep. 2016. Accessed: Jul. 23, 2020. [Online]. Available: https://eprints.soton.ac.uk/414904/
- [96] V. Inkarojrit, "Monitoring and modelling of manually-controlled Venetian blinds in private offices: a pilot study," *J. Build. Perform. Simul.*, vol. 1, no. 2, pp. 75–89, Jun. 2008, doi: 10.1080/19401490802021012.
- [97] B. Jeong, J.-W. Jeong, and J. S. Park, "Occupant behavior regarding the manual control of windows in residential buildings," *Energy Build.*, vol. 127, pp. 206–216, Sep. 2016, doi: 10.1016/j.enbuild.2016.05.097.
- [98] N. Li, J. Li, R. Fan, and H. Jia, "Probability of occupant operation of windows during transition seasons in office buildings," *Renew. Energy*, vol. 73, pp. 84–91, Jan. 2015, doi: 10.1016/j.renene.2014.05.065.
- [99] F. Naspi, M. Arnesano, L. Zampetti, F. Stazi, G. M. Revel, and M. D'Orazio, "Experimental study on occupants' interaction with windows and lights in Mediterranean offices during the non-heating season," *Build. Environ.*, vol. 127, pp. 221–238, Jan. 2018, doi: 10.1016/j.buildenv.2017.11.009.
- [100] J. Park and C.-S. Choi, "Modeling occupant behavior of the manual control of windows in residential buildings," *Indoor Air*, vol. 29, no. 2, pp. 242–251, 2019, doi: https://doi.org/10.1111/ina.12522.
- [101] S. Shi and B. Zhao, "Occupants' interactions with windows in 8 residential apartments in Beijing and Nanjing, China," *Build. Simul.*, vol. 9, no. 2, pp. 221–231, Apr. 2016, doi: 10.1007/s12273-015-0266z.
- [102] M. Yao and B. Zhao, "Window opening behavior of occupants in residential buildings in Beijing," Build. Environ., vol. 124, pp. 441–449, Nov. 2017, doi: 10.1016/j.buildenv.2017.08.035.
- [103] S. Hygge, "The interaction of noise and mild heat on cognitive performance and serial reaction time," *Environ. Int.*, vol. 17, no. 4, pp. 229–234, Jan. 1991, doi: 10.1016/0160-4120(91)90007-D.
- [104] G. Clausen and D. P. Wyon, "The Combined Effects of Many Different Indoor Environmental Factors on Acceptability and Office Work Performance," *HVACR Res.*, vol. 14, no. 1, pp. 103–113, Jan. 2008, doi: 10.1080/10789669.2008.10390996.
- [105] W. H. Ko *et al.*, "The impact of a view from a window on thermal comfort, emotion, and cognitive performance," *Build. Environ.*, vol. 175, p. 106779, May 2020, doi: 10.1016/j.buildenv.2020.106779.
- [106] J. A. Veitch, "Office noise and illumination effects on reading comprehension," J. Environ. Psychol., vol. 10, no. 3, pp. 209–217, Sep. 1990, doi: 10.1016/S0272-4944(05)80096-9.
- [107] A. Liebl, J. Haller, B. Jödicke, H. Baumgartner, S. Schlittmeier, and J. Hellbrück, "Combined effects of acoustic and visual distraction on cognitive performance and well-being," *Appl. Ergon.*, vol. 43, no. 2, pp. 424–434, Mar. 2012, doi: 10.1016/j.apergo.2011.06.017.
- [108] A. M. Abbasi, M. Motamedzade, M. Aliabadi, R. Golmohammadi, and L. Tapak, "Combined effects of noise and air temperature on human neurophysiological responses in a simulated indoor environment," *Appl. Ergon.*, vol. 88, p. 103189, Oct. 2020, doi: 10.1016/j.apergo.2020.103189.

- [109] R. Andersen, V. Fabi, J. Toftum, S. P. Corgnati, and B. W. Olesen, "Window opening behaviour modelled from measurements in Danish dwellings," *Build. Environ.*, vol. 69, pp. 101–113, Nov. 2013, doi: 10.1016/j.buildenv.2013.07.005.
- [110] F. Salamone *et al.*, "Evaluation of the Visual Stimuli on Personal Thermal Comfort Perception in Real and Virtual Environments Using Machine Learning Approaches," *Sensors*, vol. 20, no. 6, p. 1627, Mar. 2020, doi: 10.3390/s20061627.
- [111] Z. Pan, S. K. Kjaergaard, and L. Mølhave, "A chamber-experiment investigation of the interaction between perceptions of noise and odor in humans," *Int. Arch. Occup. Environ. Health*, vol. 76, no. 8, pp. 598–604, Oct. 2003, doi: 10.1007/s00420-003-0464-3.
- [112] EN ISO, "EN ISO 2802-2012: Ergonomics of the physical environment -- Assessment of environments by means of an environmental survey involving physical measurements of the environment and subjective responses of people," CEN, Brussels, 2012.
- [113] EN ISO, ISO 10551-2019: Ergonomics of the Thermal Environment—Assessment of the Influence of the Thermal Environment Using Subjective Judgement Scales. ISO, 2019.
- [114] C. A. Bennett and P. Rey, "What's So Hot about Red?," *Hum. Factors*, vol. 14, no. 2, pp. 149–154, Apr. 1972, doi: 10.1177/001872087201400204.
- [115] K. Nagano and T. Horikoshi, "New index of combined effect of temperature and noise on human comfort: summer experiments on hot ambient temperature and traffic noise," 2001.
- [116] C.-C. Lin, "Effect of Noise Intensity and Illumination Intensity on Visual Performance," *Percept. Mot. Skills*, vol. 119, no. 2, pp. 441–454, Oct. 2014, doi: 10.2466/26.24.PMS.119c20z1.
- [117] O. Alm, T. Witterseh, G. Clausen, J. Toftum, and P. O. Fanger, "The impact of human perception of simultaneous exposure to thermal load, low-frequency ventilation noise and indoor air pollution," in *Proc. of 8th International Conference on Indoor Air Quality and Climate*, 1999, pp. 270–275. Accessed: Aug. 14, 2020. [Online]. Available: https://orbit.dtu.dk/en/publications/the-impact-of-humanperception-of-simultaneous-exposure-to-therma
- [118] P. C. Berry, "Effect of colored illumination upon perceived temperature.," J. Appl. Psychol., vol. 45, no. 4, p. 248, 1961.
- [119] S. Hygge and I. Knez, "Effects of noise, heat and indoor lighting on cognitive performance and self-reported affect," *J. Environ. Psychol.*, vol. 21, no. 3, pp. 291–299, Sep. 2001, doi: 10.1006/jevp.2001.0222.
- [120] P. M. Bluyssen, D. Zhang, D. H. Kim, A. Eijkelenboom, and M. Ortiz-Sanchez, "First SenseLab studies with primary school children: exposure to different environmental configurations in the experience room," *Intell. Build. Int.*, vol. 0, no. 0, pp. 1–18, Sep. 2019, doi: 10.1080/17508975.2019.1661220.
- [121] J. Choi, A. Aziz, and V. Loftness, "Decision support for improving occupant environemtnal satisfacton in office buildings: The relationship between sub-set of IEW satisfaction and overall environemntal satisfaction," *Proc. Healthy Build. Conf. 2009*, pp. 1302–1305.
- [122] K. Yamazaki, S. Nomoto, Y. Yokota, and T. Murai, "The effects of temperature, light, and sound on perceived work environment," ASHRAE Trans., vol. 104, no. 711, 1998.
- [123] L. Fang, G. Clausen, and P. O. Fanger, "Impact of Temperature and Humidity on Chemical and Sensory Emissions from Building Materials," *Indoor Air*, vol. 9, no. 3, pp. 193–201, 1999, doi: 10.1111/j.1600-0668.1999.t01-1-00006.x.
- [124] F. Haldi and D. Robinson, "On the unification of thermal perception and adaptive actions," *Build. Environ.*, vol. 45, no. 11, pp. 2440–2457, Nov. 2010, doi: 10.1016/j.buildenv.2010.05.010.
- [125] Y. Geng, W. Ji, B. Lin, and Y. Zhu, "The impact of thermal environment on occupant IEQ perception and productivity," *Build. Environ.*, vol. 121, pp. 158–167, Aug. 2017, doi: 10.1016/j.buildenv.2017.05.022.
- [126] W. Yang and H. J. Moon, "Effects of recorded water sounds on intrusive traffic noise perception under three indoor temperatures," *Appl. Acoust.*, vol. 145, pp. 234–244, 2019.
- [127] M. A. Skwarczynski, A. K. Melikov, J. Kaczmarczyk, and V. Lyubenova, "Impact of individually controlled facially applied air movement on perceived air quality at high humidity," *Build. Environ.*, vol. 45, no. 10, pp. 2170–2176, Oct. 2010, doi: 10.1016/j.buildenv.2010.03.017.
- [128] G. Chinazzo, J. Wienold, and M. Andersen, "Daylight affects human thermal perception," *Sci. Rep.*, vol. 9, no. 1, Art. no. 1, Sep. 2019, doi: 10.1038/s41598-019-48963-y.
- [129] G. Chinazzo, J. Wienold, and M. Andersen, "Influence of indoor temperature and daylight illuminance on visual perception," *Light. Res. Technol.*, vol. 52, no. 3, pp. 350–370, May 2020, doi: 10.1177/1477153519859609.
- [130] G. Chinazzo, J. Wienold, and M. Andersen, "Effect of Indoor Temperature and Glazing with Saturated Color on Visual Perception of Daylight," *LEUKOS*, vol. 0, no. 0, pp. 1–22, Mar. 2020, doi: 10.1080/15502724.2020.1726182.
- [131] J. Winzen, F. Albers, and C. Marggraf-Micheel, "The influence of coloured light in the aircraft cabin on passenger thermal comfort," *Light. Res. Technol.*, vol. 46, no. 4, pp. 465–475, Aug. 2014, doi: 10.1177/1477153513484028.
- [132] J. Y. Garretón, R. Rodriguez, and A. Pattini, "Effects of perceived indoor temperature on daylight glare perception," *Build. Res. Inf.*, vol. 44, no. 8, pp. 907–919, Nov. 2016, doi: 10.1080/09613218.2016.1103116.
- [133] C. Buratti, E. Belloni, F. Merli, and P. Ricciardi, "A new index combining thermal, acoustic, and visual comfort of moderate environments in temperate climates," *Build. Environ.*, vol. 139, pp. 27–37, Jul. 2018, doi: 10.1016/j.buildenv.2018.04.038.
- [134] C. Laurentin, V. Bermtto, and M. Fontoynont, "Effect of thermal conditions and light source type on visual comfort appraisal," *Light. Res. Technol.*, vol. 32, no. 4, pp. 223–233, Dec. 2000, doi: 10.1177/096032710003200406.
- [135] F. Salamone, G. Chinazzo, L. Danza, C. Miller, S. Sibilio, and M. Masullo, "Low-Cost Thermohygrometers to Assess Thermal Comfort in the Built Environment: A Laboratory Evaluation of Their Measurement Performance," *Buildings*, vol. 12, no. 5, Art. no. 5, May 2022, doi: 10.3390/buildings12050579.
- [136] I. Golasi, F. Salata, E. de L. Vollaro, and A. Peña-García, "Influence of lighting colour temperature on indoor thermal perception: A strategy to save energy from the HVAC installations," *Energy Build.*, vol. 185, pp. 112–122, Feb. 2019, doi: 10.1016/j.enbuild.2018.12.026.
- [137] H. Wang, G. Liu, S. Hu, and C. Liu, "Experimental investigation about thermal effect of colour on thermal sensation and comfort," *Energy Build.*, vol. 173, pp. 710–718, Aug. 2018, doi: 10.1016/j.enbuild.2018.06.008.
- [138] G. Grün, R. T. Hellwig, M. Trimmel, and A. H. Holm, "Interrelations of comfort parameters in a simulated aircraft cabin," 2008.
- [139] M. te Kulve, L. Schlangen, and W. van M. Lichtenbelt, "Interactions between the perception of light and temperature," *Indoor Air*, vol. 28, no. 6, pp. 881–891, 2018, doi: 10.1111/ina.12500.
- [140] C. K. C. Lam *et al.*, "Cross-modal effects of thermal and visual conditions on outdoor thermal and visual comfort perception," *Build. Environ.*, vol. 186, p. 107297, Dec. 2020, doi: 10.1016/j.buildenv.2020.107297.
- [141] S. Lechner, C. Moosmann, A. Wagner, and M. Schweiker, "Does thermal control improve visual satisfaction? Interactions between occupants' self-perceived control, visual, thermal, and overall satisfaction," *Indoor Air*, vol. 31, no. 6, pp. 2329–2349, 2021, doi: 10.1111/ina.12851.
- [142] L. Fang, D. P. Wyon, G. Clausen, and O. Fanger, "Impact of indoor air temperature and humidity in an office on perceived air quality, SBS symptoms and performance.," *Indoor Air*, vol. 14 Suppl 7, pp. 74– 81, Jan. 2004, doi: 10.1111/j.1600-0668.2004.00276.x.
- [143] G. Chinazzo, L. Pastore, J. Wienold, and M. Andersen, "A field study investigation on the influence of light level on subjective thermal perception in different seasons," 2018.
- [144] T. Liu, C.-C. Lin, K.-C. Huang, and Y.-C. Chen, "Effects of noise type, noise intensity, and illumination intensity on reading performance," *Appl. Acoust.*, vol. 120, pp. 70–74, May 2017, doi: 10.1016/j.apacoust.2017.01.019.
- [145] Y. Teramoto *et al.*, "Effects of different light intensities during the forenoon on the afternoon thermal sensation in mild cold," *J. Therm. Biol.*, vol. 21, no. 5, pp. 339–343, Oct. 1996, doi: 10.1016/S0306-4565(96)00019-8.
- [146] H. Tang, Y. Ding, and B. Singer, "Interactions and comprehensive effect of indoor environmental quality factors on occupant satisfaction," *Build. Environ.*, vol. 167, p. 106462, Jan. 2020, doi: 10.1016/j.buildenv.2019.106462.
- [147] W. Yang and H. J. Moon, "Cross-modal effects of illuminance and room temperature on indoor environmental perception," *Build. Environ.*, vol. 146, pp. 280–288, Dec. 2018, doi: 10.1016/j.buildenv.2018.10.007.
- [148] N. M. Razali and Y. B. Wah, "Power comparisons of shapiro-wilk, kolmogorov-smirnov, lilliefors and anderson-darling tests," J. Stat. Model. Anal., vol. 2, no. 1, pp. 21–33, 2011.
- [149] P. E. Shrout and J. L. Rodgers, "Psychology, Science, and Knowledge Construction: Broadening Perspectives from the Replication Crisis," *Annu. Rev. Psychol.*, vol. 69, no. 1, pp. 487–510, 2018, doi: 10.1146/annurev-psych-122216-011845.
- [150] N. C. Nelson, J. Chung, K. Ichikawa, and M. M. Malik, "Psychology Exceptionalism and the Multiple Discovery of the Replication Crisis," *Rev. Gen. Psychol.*, vol. 26, no. 2, pp. 184–198, Jun. 2022, doi: 10.1177/10892680211046508.
- [151] F. Romero, "Philosophy of science and the replicability crisis," *Philos. Compass*, vol. 14, no. 11, p. e12633, 2019, doi: 10.1111/phc3.12633.

- [152] E. Rich and A. Miah, "Mobile, wearable and ingestible health technologies: towards a critical research agenda," *Health Sociol. Rev.*, vol. 26, no. 1, pp. 84–97, Gennaio 2017, doi: 10.1080/14461242.2016.1211486.
- [153] D. Sheikh Khan, J. Kolarik, and P. Weitzmann, "Design and application of occupant voting systems for collecting occupant feedback on indoor environmental quality of buildings – A review," *Build. Environ.*, vol. 183, p. 107192, Ottobre 2020, doi: 10.1016/j.buildenv.2020.107192.
- [154] T. A. Glass, S. N. Goodman, M. A. Hernán, and J. M. Samet, "Causal Inference in Public Health," Annu. Rev. Public Health, vol. 34, no. 1, pp. 61–75, 2013, doi: 10.1146/annurev-publhealth-031811-124606.
- [155] G. Clausen, L. Carrick, P. O. Fanger, S. W. Kim, T. Poulsen, and J. H. Rindel, "A Comparative Study Of Discomfort Caused By Indoor Air Pollution, Thermal Load And Noisec," *Indoor Air*, vol. 3, no. 4, pp. 255–262, 1993, doi: 10.1111/j.1600-0668.1993.00006.x.
- [156] L. Danza, L. Belussi, and F. Salamone, "A multiple linear regression approach to correlate the Indoor Environmental Factors to the global comfort in a Zero-Energy building," *E3S Web Conf.*, vol. 197, p. 04002, 2020, doi: 10.1051/e3sconf/202019704002.
- [157] L. T. Wong, K. W. Mui, and P. S. Hui, "A multivariate-logistic model for acceptance of indoor environmental quality (IEQ) in offices," *Build. Environ.*, vol. 43, no. 1, pp. 1–6, Jan. 2008, doi: 10.1016/j.buildenv.2007.01.001.
- [158] L. Huang, Y. Zhu, Q. Ouyang, and B. Cao, "A study on the effects of thermal, luminous, and acoustic environments on indoor environmental comfort in offices," *Build. Environ.*, vol. 49, pp. 304–309, Mar. 2012, doi: 10.1016/j.buildenv.2011.07.022.
- [159] L. Danza et al., "A weighting procedure to analyse the Indoor Environmental Quality of a Zero-Energy Building," Build. Environ., vol. 183, p. 107155, Oct. 2020, doi: 10.1016/j.buildenv.2020.107155.
- [160] A. C. K. Lai, K. W. Mui, L. T. Wong, and L. Y. Law, "An evaluation model for indoor environmental quality (IEQ) acceptance in residential buildings," *Energy Build.*, vol. 41, no. 9, pp. 930–936, Sep. 2009, doi: 10.1016/j.enbuild.2009.03.016.
- [161] O. Fanger, N. O. Breum, and E. Jerking, "Can Colour and Noise Influence Man's Thermal Comfort?," *Ergonomics*, vol. 20, no. 1, pp. 11–18, Jan. 1977, doi: 10.1080/00140137708931596.
- [162] W. Yang, H. J. Moon, and M.-J. Kim, "Combined effects of short-term noise exposure and hygrothermal conditions on indoor environmental perceptions," *Indoor Built Environ.*, vol. 27, no. 8, pp. 1119–1133, Oct. 2018, doi: 10.1177/1420326X17703774.
- [163] W. Yang and H. J. Moon, "Combined effects of sound and illuminance on indoor environmental perception," *Appl. Acoust.*, vol. 141, pp. 136–143, Dec. 2018, doi: 10.1016/j.apacoust.2018.07.008.
- [164] N. Pellerin and V. Candas, "Combined effects of temperature and noise on human discomfort," *Physiol. Behav.*, vol. 78, no. 1, pp. 99–106, 2003.
- [165] S. Crosby and A. Rysanek, "Correlations between thermal satisfaction and non-thermal conditions of indoor environmental quality: Bayesian inference of a field study of offices," *J. Build. Eng.*, p. 102051, Dec. 2020, doi: 10.1016/j.jobe.2020.102051.
- [166] W. Yang and H. J. Moon, "Cross-modal effects of noise and thermal conditions on indoor environmental perception and speech recognition," *Appl. Acoust.*, vol. 141, pp. 1–8, 2018.
- [167] Choi Joon-Ho, A. Aziz, and V. Loftness, "Decision support for improving occupant environmental satisfaction in office buildings: The relationship between sub-set of IEQ satisfaction and overall environmental satisfaction.," presented at the Conference Healthy Buildings, Syracuse, NY USA.
- [168] T. Mihai and V. Iordache, "Determining the indoor environment quality for an educational building," *Energy Procedia*, vol. 85, pp. 566–574, 2016.
- [169] M. Ncube and S. Riffat, "Developing an indoor environment quality tool for assessment of mechanically ventilated office buildings in the UK – A preliminary study," *Build. Environ.*, vol. 53, pp. 26–33, Jul. 2012, doi: 10.1016/j.buildenv.2012.01.003.
- [170] B. Cao, Q. Ouyang, Y. Zhu, L. Huang, H. Hu, and G. Deng, "Development of a multivariate regression model for overall satisfaction in public buildings based on field studies in Beijing and Shanghai," *Build. Environ.*, vol. 47, pp. 394–399, Jan. 2012, doi: 10.1016/j.buildenv.2011.06.022.
- [171] K. Nagano and T. Horikoshi, "Development of Equi-comfort Charts Constituted with Temperature and Noise at 150 and 3 lx," Proc. Indoor Air, 2014.
- [172] W. Yang, M.-J. Kim, and H. J. Moon, "Effects of indoor temperature and background noise on floor impact noise perception," *Indoor Built Environ.*, vol. 28, no. 4, pp. 454–469, Apr. 2019, doi: 10.1177/1420326X17753708.
- [173] P. A. Bell, "Effects of Noise and Heat Stress on Primary and Subsidiary Task Performance," Hum. Factors, vol. 20, no. 6, pp. 749–752, Dec. 1978, doi: 10.1177/001872087802000614.
- [174] N. Pellerin and V. Candas, "Effects of steady-state noise and temperature conditions on environmental perception and acceptability," *Indoor Air*, vol. 14, no. 2, pp. 129–136, Apr. 2004, doi: 10.1046/j.1600-0668.2003.00221.x.

- [175] L. Lan, P. Wargocki, D. P. Wyon, and Z. Lian, "Effects of thermal discomfort in an office on perceived air quality, SBS symptoms, physiological responses, and human performance," *Indoor Air*, vol. 21, no. 5, pp. 376–390, 2011.
- [176] N. Matsubara, A. Gassho, and Y. Kurazumi, "Facilitatory effects of environmental sounds on hue-heat phenomena," in *18th Int. Congr. Acoust*, 2004, vol. 2, pp. 1775–1778.
- [177] Y. Geng, J. Yu, B. Lin, Z. Wang, and Y. Huang, "Impact of individual IEQ factors on passengers' overall satisfaction in Chinese airport terminals," *Build. Environ.*, vol. 112, pp. 241–249, Feb. 2017, doi: 10.1016/j.buildenv.2016.11.040.
- [178] L. Fang, G. Clausen, and P. O. Fanger, "Impact of Temperature and Humidity on Perception of Indoor Air Quality During Immediate and Longer Whole-Body Exposures," *Indoor Air*, vol. 8, no. 4, pp. 276– 284, 1998, doi: 10.1111/j.1600-0668.1998.00008.x.
- [179] L. Fang, G. Clausen, and P. O. Fanger, "Impact of Temperature and Humidity on the Perception of Indoor Air Quality," *Indoor Air*, vol. 8, no. 2, pp. 80–90, 1998, doi: 10.1111/j.1600-0668.1998.t01-2-00003.x.
- [180] F. Albers, J. Maier, and C. Marggraf-Micheel, "In search of evidence for the hue-heat hypothesis in the aircraft cabin," *Light. Res. Technol.*, vol. 47, no. 4, pp. 483–494, 2015.
- [181] A. J. Xu and A. A. Labroo, "Incandescent affect: Turning on the hot emotional system with bright light," J. Consum. Psychol., vol. 24, no. 2, pp. 207–216, Apr. 2014, doi: 10.1016/j.jcps.2013.12.007.
- [182] H. Nakamura and M. Oki, "Influence of Air Temperature on Preference for Color Temperature of General Lighting in the Room," 人間-生活環境系学会英文誌, vol. 4, no. 1, pp. 41–47, 2000, doi: 10.1618/jhes.4.41.
- [183] H. Ma and W. Nie, "Influence of visual factors on noise annoyance evaluation caused by road traffic noise in indoor environment," in *INTER-NOISE and NOISE-CON Congress and Conference Proceedings*, 2014, vol. 249, no. 1, pp. 6308–6315.
- [184] E. C. Poulton and R. S. Edwards, "Interactions and range effects in experiments on pairs of stresses: Mild heat and low-frequency noise," *J. Exp. Psychol.*, vol. 102, no. 4, pp. 621–628, 1974, doi: 10.1037/h0036319.
- [185] S. Gauthier, B. Liu, G. Huebner, and D. Shipworth, "Investigating the effect of CO2 concentration on reported thermal comfort," in *Proceedings of CISBAT 2015 International Conference on Future Buildings and Districts - Sustainability from Nano to Urban Scale - Vol. I*, Sep. 2015, pp. 315–320. Accessed: Jul. 23, 2020. [Online]. Available: https://eprints.soton.ac.uk/381850/
- [186] H. Wu, X. Sun, and Y. Wu, "Investigation of the relationships between thermal, acoustic, illuminous environments and human perceptions," *J. Build. Eng.*, vol. 32, p. 101839, Nov. 2020, doi: 10.1016/j.jobe.2020.101839.
- [187] Y. T. Jiang and C. Z. Yang, "Investigation of the subjective evaluation of indoor illumination level on perceived air quality," *Proc. ISHVAC 2011*, vol. Vol. 2, pp. 453–458.
- [188] M. te Kulve, L. Schellen, L. Schlangen, A. J. H. Frijns, and W. D. van Marken Lichtenbelt, "Light intensity and thermal responses," in *Making Comfort Relevant: Proceedings of 9th Windsor Conference*, 2016, pp. 1–8.
- [189] F. Fassio, A. Fanchiotti, and R. D. L. Vollaro, "Linear, Non-Linear and Alternative Algorithms in the Correlation of IEQ Factors with Global Comfort: A Case Study," *Sustainability*, vol. 6, no. 11, Art. no. 11, Nov. 2014, doi: 10.3390/su6118113.
- [190] K. Nagano and T. Horikoshi, "New comfort index during combined conditions of moderate low ambient temperature and traffic noise," *Energy Build.*, vol. 37, no. 3, pp. 287–294, Mar. 2005, doi: 10.1016/j.enbuild.2004.08.001.
- [191] J. Kim and R. de Dear, "Nonlinear relationships between individual IEQ factors and overall workspace satisfaction," *Build. Environ.*, vol. 49, pp. 33–40, Mar. 2012, doi: 10.1016/j.buildenv.2011.09.022.
- [192] J. Toftum, A. Thorseth, J. Markvart, and Á. Logadóttir, "Occupant response to different correlated colour temperatures of white LED lighting," *Build. Environ.*, vol. 143, pp. 258–268, Oct. 2018, doi: 10.1016/j.buildenv.2018.07.013.
- [193] L. Bellia, F. R. d'Ambrosio Alfano, F. Fragliasso, B. I. Palella, and G. Riccio, "On the interaction between lighting and thermal comfort: An integrated approach to IEQ," *Energy Build.*, vol. 231, p. 110570, Jan. 2021, doi: 10.1016/j.enbuild.2020.110570.
- [194] L. G. Berglund and W. S. Cain, "Perceived air quality and the thermal environment," in *Proceedings* of IAQ, 1989, vol. 89, pp. 93–99.
- [195] Y. Sakurai, T. Noguchi, G. Horie, and N. Matsubara, "Quantification of the synthesized evaluation of the combined environment," *Energy Build.*, vol. 14, no. 3, pp. 169–173, 1990.
- [196] J. Varjo, V. Hongisto, A. Haapakangas, H. Maula, H. Koskela, and J. Hyönä, "Simultaneous effects of irrelevant speech, temperature and ventilation rate on performance and satisfaction in open-plan offices," J. Environ. Psychol., vol. 44, pp. 16–33, Dec. 2015, doi: 10.1016/j.jenvp.2015.08.001.

- [197] Z. Tu, S. Geng, and Y. Li, "Study on human responses under different CO2 concentration and illuminance in underground refuge chamber," *IOP Conf. Ser. Earth Environ. Sci.*, vol. 510, p. 052026, Jul. 2020, doi: 10.1088/1755-1315/510/5/052026.
- [198] G. Clausen and D. Wyon, "The combined effects of many different indoor environmental factors on acceptability and office work performance," in *Proc of Indoor Air 2005*, 2005, pp. 351–356.
- [199] S. K. Bhattacharya, S. R. Tripathi, and S. K. Kashyap, "The Combined Effects of Noise and Illumination on the Performance Efficiency of Visual Search and Neuromotor Task Components," *J. Hum. Ergol.* (*Tokyo*), vol. 18, no. 1, pp. 41–51, 1989, doi: 10.11183/jhe1972.18.41.
- [200] L. Pronk, "THE COMBINED EFFECTS OF TEMPERATURE, BACKGROUND NOISE AND LIGHTING ON THE NON-PHYSICAL TASK PERFORMANCE OF UNIVERSITY STUDENTS," p. 209.
- [201] R. R. Baniya, E. Tetri, J. Virtanen, and L. Halonen, "The effect of correlated colour temperature of lighting on thermal sensation and thermal comfort in a simulated indoor workplace," *Indoor Built Environ.*, vol. 27, no. 3, pp. 308–316, Mar. 2018, doi: 10.1177/1420326X16673214.
- [202] J. Winzen, F. Albers, and C. Marggraf-Micheel, "The influence of coloured light in the aircraft cabin on passenger thermal comfort," *Light. Res. Technol. Lond.*, vol. 46, no. 4, pp. 465–475, Aug. 2014, doi: http://dx.doi.org/10.1177/1477153513484028.
- [203] H. Azmoon, H. Dehghan, J. Akbari, and S. Souri, "The Relationship between Thermal Comfort and Light Intensity with Sleep Quality and Eye Tiredness in Shift Work Nurses," *J. Environ. Public Health*, vol. 2013, p. e639184, Feb. 2013, doi: https://doi.org/10.1155/2013/639184.
- [204] J. Toftum, A. S. Jørgensen, and P. O. Fanger, "Upper limits of air humidity for preventing warm respiratory discomfort," *Energy Build.*, vol. 28, no. 1, pp. 15–23, Aug. 1998, doi: 10.1016/S0378-7788(97)00018-2.
- [205] W. S. Cain *et al.*, "Ventilation requirements in buildings—I. Control of occupancy odor and tobacco smoke odor," *Atmospheric Environ. 1967*, vol. 17, no. 6, pp. 1183–1197, Jan. 1983, doi: 10.1016/0004-6981(83)90341-4.
- [206] D. Blackenberger, K. V. D. Wymelenberg, and J. Stenson, "Visual effects of wood on thermal perception of interior environments," *ARCC Conf. Repos.*, May 2019, Accessed: Nov. 30, 2020. [Online]. Available: https://www.arcc-journal.org/index.php/repository/article/view/619
- [207] R. D. Pepler, "Warmth, glare and a background of quiet speech: A comparison of their effects on performance," *Ergonomics*, vol. 3, no. 1, pp. 68–73, 1960.
- [208] V. Roussarie, E. Siekierski, S. Viollon, S. Ségrétain, and S. Bojago, "What's so hot about sound? influence of HVAC sounds on thermal comfort," p. 11.
- [209] G. Newsham *et al.*, "Control strategies for lighting and ventilation in offices: effects on energy and occupants," *Intell. Build. Int.*, vol. 1, no. 2, pp. 101–121, Feb. 2009, doi: 10.3763/inbi.2009.0004.
- [210] K. Ishibashi, S. Arikura, T. Kozaki, S. Higuchi, and A. Yasukouchi, "Thermoregulatory Effect in Humans of Suppressed Endogenous Melatonin by Pre-Sleep Bright-Light Exposure in a Cold Environment," *Chronobiol. Int.*, vol. 27, no. 4, pp. 782–806, May 2010, doi: 10.3109/07420521003794069.
- [211] S. Lamb and K. C. S. Kwok, "A longitudinal investigation of work environment stressors on the performance and wellbeing of office workers," *Appl. Ergon.*, vol. 52, pp. 104–111, Jan. 2016, doi: 10.1016/j.apergo.2015.07.010.
- [212] A. Astolfi and F. Pellerey, "Subjective and objective assessment of acoustical and overall environmental quality in secondary school classrooms," *J. Acoust. Soc. Am.*, vol. 123, no. 1, pp. 163– 173, Jan. 2008, doi: 10.1121/1.2816563.
- [213] N. D. Dahlan, "Perceptive-cognitive aspects investigation in relation to indoor environment satisfaction collected from naturally ventilated multi-storey student accommodations in Malaysia," *Indoor Built Environ.*, vol. 24, no. 1, pp. 116–127, Feb. 2015, doi: 10.1177/1420326X13506449.
- [214] M. Frontczak, R. V. Andersen, and P. Wargocki, "Questionnaire survey on factors influencing comfort with indoor environmental quality in Danish housing," *Build. Environ.*, vol. 50, pp. 56–64, Apr. 2012, doi: 10.1016/j.buildenv.2011.10.012.
- [215] M. A. Humphreys, "Quantifying occupant comfort: are combined indices of the indoor environment practicable?," *Build. Res. Inf.*, vol. 33, no. 4, pp. 317–325, 2005, doi: 10.1080/09613210500161950.
- [216] F. Martellotta, A. Simone, S. Della Crociata, and M. D'Alba, "Global comfort and indoor environment quality attributes for workers of a hypermarket in Southern Italy," *Build. Environ.*, vol. 95, pp. 355–364, Jan. 2016, doi: 10.1016/j.buildenv.2015.09.029.
- [217] P. S. Nimlyat, "Indoor environmental quality performance and occupants' satisfaction [IEQPOS] as assessment criteria for green healthcare building rating," *Build. Environ.*, vol. 144, pp. 598–610, Oct. 2018, doi: 10.1016/j.buildenv.2018.09.003.
- [218] R. M. Reffat and E. L. Harkness, "Environmental Comfort Criteria: Weighting and Integration," J. Perform. Constr. Facil., vol. 15, no. 3, pp. 104–108, Aug. 2001, doi: 10.1061/(ASCE)0887-3828(2001)15:3(104).

- [219] C.-M. Chiang and C.-M. Lai, "A study on the comprehensive indicator of indoor environment assessment for occupants' health in Taiwan," *Build. Environ.*, vol. 37, no. 4, pp. 387–392, 2002.
- [220] S. C. Sekhar and S. E. Goh, "Thermal comfort and IAQ characteristics of naturally/mechanically ventilated and air-conditioned bedrooms in a hot and humid climate," *Build. Environ.*, vol. 46, no. 10, pp. 1905–1916, Oct. 2011, doi: 10.1016/j.buildenv.2011.03.012.
- [221] A. Jamrozik *et al.*, "A novel methodology to realistically monitor office occupant reactions and environmental conditions using a living lab," *Build. Environ.*, vol. 130, pp. 190–199, Feb. 2018, doi: 10.1016/j.buildenv.2017.12.024.
- [222] G. Ozcelik, B. Becerik-Gerber, and R. Chugh, "Understanding human-building interactions under multimodal discomfort," *Build. Environ.*, vol. 151, pp. 280–290, Mar. 2019, doi: 10.1016/j.buildenv.2018.12.046.
- [223] G. Chinazzo, K. Chamilothori, J. Wienold, and M. Andersen, "Temperature–Color Interaction: Subjective Indoor Environmental Perception and Physiological Responses in Virtual Reality," *Hum. Factors*, p. 0018720819892383, Jan. 2020, doi: 10.1177/0018720819892383.

Appendix A

Ref.	List of considered papers								
[111]	A chamber-experiment investigation of the interaction between perceptions of noise and odor in humans								
[155]	A comparative study of discomfort caused by indoor air pollution, thermal load and noise								
[51]	A field study investigation on the influence of light level on subjective thermal perception in different seasons								
[156]	A multiple linear regression approach to correlate the Indoor Environmental Factors to the global comfort in a Zero-Energy building								
[157]	A multivariate-logistic model for acceptance of indoor environmental quality (IEQ) in offices								
[133]	A new index combining thermal, acoustic, and visual comfort of moderate environments in temperate climates								
[158]	A study on the effects of thermal, luminous, and acoustic environments on indoor environmental comfort in offices								
[159]	A weighting procedure to analyse the Indoor Environmental Quality of a Zero-Energy Building								
[88]	Air movement and perceived air quality								
[81]	An applied framework to evaluate the impact of indoor office environmental factors on occupants' comfort and working conditions								
[160]	An evaluation model for indoor environmental quality (IEQ) acceptance in residential buildings								
[161]	Can colour and noise influence man's thermal comfort?								
[94]	Colour as a psychological agent to manipulate perceived indoor thermal environment for effective energy usage								
[107]	Combined effects of acoustic and visual distraction on cognitive performance and well-being								
[56]	Combined effects of acoustic, thermal, and illumination conditions on the comfort of discrete senses and overall indoor environment								
[55]	Combined effects of daylight transmitted through coloured glazing and indoor temperature on thermal responses and overall comfort								
[108]	Combined effects of noise and air temperature on human neurophysiological responses in a simulated indoor environment								
[84]	Combined Effects of noise and temperature on human comfort and performance								
[162]	Combined effects of short-term noise exposure and hygrothermal conditions on indoor environmental perceptions								
[163]	Combined effects of sound and illuminance on indoor environmental perception								
[164]	Combined effects of temperature and noise on human discomfort								

[165]	Correlations between thermal satisfaction and non-thermal conditions of indoor environmental quality: Bayesian inference of a field study of offices								
[147]	Cross-modal effects of illuminance and room temperature on indoor environmental perception								
[166]	Cross-modal effects of noise and thermal conditions on indoor environmental perception and speech recognition								
[140]	Cross-modal effects of thermal and visual conditions on outdoor thermal and visual comfort perception								
[128]	Daylight affects human thermal perception								
[167]	Decision support for improving occupant environmental satisfaction in office buildings: The relationship between sub-set of IEQ satisfaction and overall environmental satisfaction								
[168]	Determining the Indoor Environment Quality for an Educational Building								
[169]	Developing an indoor environment quality tool for assessment of mechanically ventilated office buildings in the UK – A preliminary study								
[170]	Development of a multivariate regression model for overall satisfaction in public buildings based on field studies								
[171]	Development of equi-comfort charts constituted with temperature and noise at 150 and 3 lx								
[118]	Effect of colored illumination upon perceived temperature								
[130]	Effect of Indoor Temperature and Glazing with Saturated Color on Visual Perception of Daylight								
[116]	Effect of noise intensity and illumination intensity on visual performance								
[134]	Effect of thermal conditions and light source type on visual comfort appraisal								
[145]	Effects of different light intensities during the forenoon on the afternoon thermal sensation in mild cold								
[172]	Effects of indoor temperature and background noise on floor impact noise perception								
[173]	Effects of noise and heat stress on primary and subsidiary task performance								
[144]	Effects of noise type, noise intensity, and illumination intensity on reading performance								
[119]	Effects of noise, heat and indoor lighting on cognitive performance and self-reported affect								
[132]	Effects of perceived indoor temperature on daylight glare perception								
[126]	Effects of recorded water sounds on intrusive traffic noise perception under three indoor temperatures								
[174]	Effects of steady-state noise and temperature conditions on environmental perception and acceptability								
[175]	Effects of thermal discomfort in an office on perceived air quality, SBS symptoms, physiological responses, and human performance								
[110]	Evaluation of the Visual Stimuli on Personal Thermal Comfort Perception in Real and Virtual Environments Using Machine Learning Approaches								
[137]	Experimental investigation about thermal effect of colour on thermal sensation and comfort								
[99]	Experimental study on occupants' interaction with windows and lights in Mediterranean offices during the non-heating season								
[176]	Facilitatory effects of environmental sounds on hue-heat phenomena								
[120]	First SenseLab studies with primary school children: exposure to different environmental								
[120]	configurations in the experience room								
[54]	How correlated colour temperature manipulates human thermal perception and comfort								
[177]	Impact of individual IEQ factors on passengers ' overall satisfaction in Chinese airport terminals								
[127]	Impact of individually controlled facially applied air movement on perceived air quality at high humidity								
[142]	Impact of indoor air temperature and humidity in an office on perceived air quality, SBS symptoms and performance								

[178] Impact of Temperature and Humidity on Perception of Indoor Air Quality During In									
[]	and Longer Whole-Body Exposures								
[179]	Impact or temperature and humidity on the perception of indoor air quality								
[180]	In search of evidence for the hue-heat								
[]	hypothesis in the aircraft cabin								
[181]	Incandescent affect: turning on the hot emotional system with bright light								
[182]	Influence of air temperature on preference for color temperature of general lighting in the								
	10011								
[93]									
[120]	Influence of indoor temperature and daylight illuminance on visual percention								
[90]	Influence of Light Intensities on Dressing Behavior in Elderly People								
[00]	Influence of lighting colour temperature on indoor thermal perception: A strategy to save								
[136]	energy from the HVAC installations								
	Influence of Two Different Light Intensities from 16:00 to 20:30 Hours on Evening Dressing								
[91]	Behavior in the Cold								
	Influence of visual factors on noise annovance evaluation caused by road traffic noise in								
[183]	indoor environment								
	Interactions and comprehensive effect of indoor environmental quality factors on occupant								
[146]	satisfaction								
140.43	Interactions and range effects in experiments on pairs of stresses: mild heat and low								
[184]	frequency noise								
[139]	Interactions between the perception of light and temperature								
[138]	Interrelations of Comfort Parameters in a Simulated Aircraft Cabin								
[185]	Investigating the effect of CO2 concentration on reported thermal comfort								
[406]	Investigation of the relationships between thermal, acoustic, illuminous environments and								
[186]	human perceptions								
[187]	Investigation of the subjective evaluation of indoor illumination level on perceived air quality								
1961	Irrelevant speech and indoor lighting: effects on cognitive performance and self-reported								
[00]	affect								
[188]	Light intensity and thermal responses								
[189]	Linear, non-linear and alternative algorithms in the correlation of IEQ factors with global								
[100]	comfort: a case study								
[100]	Modeling occupant behavior of the manual control of windows in residential buildings								
[96]	Monitoring and modelling of manually-controlled venetian blinds in private offices: A pilot								
[00]	study								
[190]	New comfort index during combined conditions of moderate low ambient temperature and								
[100]	traffic noise								
[115]	New index of combined effect of temperature and noise on human								
	comfort: summer experiments on hot ambient temperature and traffic noise								
[191]	Nonlinear relationships between individual IEQ factors and overall workspace satisfaction								
[97]	Occupant behavior regarding the manual control of windows in residential buildings								
[192]	Occupant response to different correlated colour temperatures of white LED lighting								
[101]	Occupants' interactions with windows in 8 residential apartments in Beijing and Nanjing,								
[106]	Crima								
[100]	Once hoise and illumination effects on reading comprehension								
[193]	On the interaction between lighting and thermal comfort: an integrated approach to IEQ								
[124]	On the uninication of thermal perception and adaptive actions								
[194]	Prerceived air quality and the thermal environment								
[98]	Probability of occupant operation of windows during transition seasons in office buildings								
[195]									

[92]	Saving energy with light? Experimental studies assessing the impact of colour temperature on thermal comfort							
[87]	Sensory and physiological effects on humans of combined exposures to air temperatures and volatile organic compounds							
[196]	Simultaneous effects of irrelevant speech, temperature and ventilation rate on performance and satisfaction in open-plan offices							
[83]	Student learning performance and indoor environmental quality (IEQ) in air-conditioned university teaching rooms							
[197]	Study on human responses under different CO2 concentration and illuminance in underground refuge chamber							
[198]	The combined effects of many different indoor environmental factors on acceptability and office work performance							
[199]	The combined effects of noise and illumination on the performance efficiency of visual search and neuromotor task components							
[200]	The combined effects of temperature, background noise and lighting on the non-physical task performance of university students							
[201]	The effect of correlated colour temperature of lighting on thermal sensation and thermal comfort in a simulated indoor workplace							
[85]	The effects of moderate heat stress and open-plan office noise distraction on SBS symptoms and on the performance of office work							
[122]	The effects of temperature, light, and sound on perceived work environment							
[105]	The impact of a view from a window on thermal comfort, emotion, and cognitive performance							
[117]	The impact of human perception of simultaneous exposure to thermal load, low frequency ventilation noise and indoor air pollution							
[125]	The impact of thermal environment on occupant IEQ perception and productivity							
[202]	The influence of coloured light in the aircraft cabin on passenger thermal comfort							
[82]	The influence of exposure to multiple indoor environmental parameters on human perception							
[89]	The influence of heat, air jet cooling and noise on performance in classrooms							
[103]	The interaction of noise and mild heat on cognitive performance and serial reaction time							
[203]	The Relationship between Thermal Comfort and Light Intensity with Sleep Quality and Eye Tiredness in Shift Work Nurses							
[95]	Understanding window behaviour in a mixed-mode buildings and the impact on energy performance							
[204]	Upper limits of air humidity for preventing warm respiratory discomfort							
[205]	Ventilation requirements in buildings-I. Control of occupancy odor and tobacco smoke odor							
[206]	Visual effects of wood on thermal perception of interior environments							
[207]	Warmth, glare and a background of quiet speech: A comparison of their effects on performance							
[114]	What's So Hot About Red?							
[208]	What's so hot about sound?-influence of HVAC sounds on thermal comfort							
[102]	Window opening behavior of occupants in residential buildings in Beijing							

Appendix B

The described exclusion criteria lead to the exclusion of studies involving contextual, personal or other behavior (all sections besides 4.1 and 5.1 in Schweiker et al. [40]). In particular, the following studies were excluded from the analysis:

• Studies focusing on the effect of personal control [209];

- Studies focusing on physiological responses only³ (e.g., [210]);
- Studies in which the independent variables are not physical measurements such as those in which overall comfort/index or performance are evaluated on the basis of subjective evaluations of the indoor environmental stimuli (e.g., [211], [211]–[218]);
- Studies reporting results of experts' questionnaires [219];
- Studies where interactions are analyzed just looking at the correlation between human responses [220];
- Studies investigating the effect of the combined presence of multiple indoor stimuli on the measurements of another factor [123];
- Studies focusing neither on cross-modal nor on combined effects [133];
- Preliminary studies in which the quantitative results described are not the goal of the study [189];
- Proof-of-concept studies [221]
- Experiments in Virtual Reality [222], [223].

³ Physiological responses are analyzed in papers where this type of response is reported together with other perceptual, behavioral, and cognitive responses.

Paper d

Building and Environment 174 (2020) 106768



Contents lists available at ScienceDirect

Building and Environment

journal homepage: http://www.elsevier.com/locate/buildenv



Modeling occupant behavior in buildings

Salvatore Carlucci^{a,b,*}, Marilena De Simone^c, Steven K. Firth^d, Mikkel B. Kjærgaard^e, Romana Markovic^f, Mohammad Saiedur Rahaman^g, Masab Khalid Annaqeeb^h, Silvia Biandrate^{b,i}, Anooshmita Das^e, Jakub Wladyslaw Dziedzic^h, Gianmarco Fajilla^c, Matteo Favero^b, Martina Ferrandoⁱ, Jakob Hahn^j, Mengjie Han^k, Yuzhen Peng¹, Flora Salim^g, Arno Schlüter¹, Christoph van Treeck^f

^h Department of Energy and Process Engineering, Norwegian University of Science and Technology, Trondheim, Norway

ⁱ Department of Energy, Politecnico di Milano, Milan, Italy

^j CENERGIE - Research Center Energy Efficient Buildings and Districts, Munich University of Applied Sciences, Germany

k School of Technology and Business Studies, Dalarna University, Falun, Sweden

ARTICLE INFO

Keywords: Occupant behavior Data-driven methods Deep learning Machine learning Stochastic methods PRISMA

ABSTRACT

In the last four decades several methods have been used to model occupants' presence and actions (OPA) in buildings according to different purposes, available computational power, and technical solutions. This study reviews approaches, methods and key findings related to OPA modeling in buildings. An extensive database of related research documents is systematically constructed, and, using bibliometric analysis techniques, the scientific production and landscape are described. The initial literature screening identified more than 750 studies, out of which 278 publications were selected. They provide an overarching view of the development of OPA modeling methods. The research field has evolved from longitudinal collaborative efforts since the late 1970s and, so far, covers diverse building typologies mostly concentrated in a few climate zones. The modeling approaches in the selected literature are grouped into three categories (rule-based models, stochastic OPA modeling, and data-driven methods) for modeling occupancy-related target functions and a set of occupants' actions (window, solar shading, electric lighting, thermostat adjustment, clothing adjustment and appliance use). The explanatory modeling is conventionally based on the model-based paradigm where occupant behavior is assumed to be stochastic, while the data-driven paradigm has found wide applications for the predictive modeling of OPA, applicable to control systems. The lack of established standard evaluation protocols was identified as a scientifically important yet rarely addressed research question. In addition, machine learning and deep learning are emerging in recent years as promising methods to address OPA modeling in real-world applications

1. Introduction

In the last four decades several methods have been used to model occupants' presence and actions (OPA) in buildings to meet different

research objectives given available computational power and technical solutions. Often the purpose has been to understand how people use a space and how their behavior impacts on a building's energy performance. Indeed, occupant behavior is also one of the main sources of

https://doi.org/10.1016/j.buildenv.2020.106768

^a Energy, Environment and Water Research Center, The Cyprus Institute, Nicosia, Cyprus

^b Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway

^c Department of Environmental Engineering, University of Calabria, Rende, Italy

^d School of Architecture, Building and Civil Engineering, Loughborough University, Loughborough, UK

e Center for Energy Informatics, University of Southern Denmark, Odense, Denmark

f E3D - Institute of Energy Efficiency and Sustainable Building, RWTH Aachen University, Aachen, Germany

⁸ School of Science, Computer Science & IT, RMIT University, Melbourne, Victoria, Australia

¹ Architecture and Building Systems, Institute of Technology in Architecture, Department of Architecture, ETH Zürich, Switzerland

^{*} Corresponding author. 20 Konstantinou Kavafi Street, 2121, Nicosia, Cyprus. *E-mail address:* s.carlucci@cyi.ac.cy (S. Carlucci).

Received 16 December 2019; Received in revised form 9 February 2020; Accepted 24 February 2020

Available online 28 February 2020

^{0360-1323/© 2020} Elsevier Ltd. All rights reserved.

uncertainty in building's energy modeling [1]. In particular, the oversimplification of the OPA description can introduce a large discrepancy between the simulated and actual energy consumption of a building [2, 3]. These and other issues have driven the exploitation of various approaches to explain and predict OPA in order to accurately model OPA in building energy simulation tools and to improve building management systems to decrease building's energy consumption. In order to address these issues, the attention of the building research community on OPA modeling has increased in recent years [4]. One initiative approved by the International Energy Agency (IEA) in 2013 is the Energy in Buildings and Communities (EBC) Annex 66 [5] that aimed to study the importance of occupant behavior in buildings and its modeling techniques and to formalize simulation approaches regarding occupant behavior. Following this, in 2017 IEA approved the EBC Annex 79 "Occupant-centric building design and operation", which aims to explore open issues on the implementation and application of occupant modeling into practice [6]. In the context of IEA-EBC Annex 79, this review aims at providing a thorough and carefully-designed overview of the methods and techniques used for modeling OPA in buildings in order to create the current state-of-the-art and identify the latest trends in this research sector. Given these ambitious objectives, a systematic approach is used to review the scientific literature to reduce the risk of missing important contributions in the field, and bibliometric analysis tools are adopted to extract patterns and information from the identified database of documents. In the scope of this work, the existing OPA studies are grouped into three paradigms: rule-based models, stochastic OPA models, and data-driven methods. The first paradigm includes, but is not restricted to, the time-dependent users' profiles as defined, for example, in the ASHRAE standard 90.1 [7]. The second paradigm considers the occupant behavior to be stochastic since behavior varies between occupants and may evolve over time [8] and is the result of complex relationships between contextual factors, adaptive triggers, and non-adaptive triggers [9]. The third paradigm refers to data-driven methods where a black-box model is derived from relating input and output data [10] so that, the modeling is conducted without an explicit aim to understand the OPA [11] and/or only with the limited inclusion of the domain engineering knowledge [12]. Resultantly, the data-driven OPA modeling, for the scope of this study, can be defined as "an approach to modeling that focuses on using the computational intelligence and particularly machine learning (ML) methods in building models that would complement or replace the "knowledge-driven" models describing physical behavior" [12]. The present study aims at describing the features of methods used for OPA modeling in buildings rather than reporting their mathematical formulation that can be found in statistical and ML handbooks. A summary of a few modelling techniques is available in Ref. [13].

1.1. Related work

Numerous reviews about OPA modeling have tried to categorize and formalize the different approaches to OPA modeling [9]. However, they are usually limited in the covered time span, in the building typology investigated or in the OPA under study. For example, Gunay et al. [14] have reviewed the modeling approaches developed for the simulation engine EnergyPlus regarding occupant presence, window and shading operations, lighting, and clothing adjustment developed since 2014. Yang et al. [15], focusing on institutional buildings, have studied the available estimation, detection and modeling methods to assess presence and movement of occupants. Gilani and O'Brien [16] have reviewed the estimation and detection methods to study OPA in office buildings. Chen et al. [17] have studied presence estimation and detection methods developed between 2005 and 2017. Zhang et al. [4] have reviewed the modeling methods for OPA regarding residential and commercial buildings. Balvedi et al. [18] focused on residential buildings in the temporal coverage from 2006 to 2017. Dong et al. [19] did an extensive literature review including all typologies of buildings, but without considering any modeling method regarding occupants' movement and activity or their clothing adjustment. Li et al. [20] covered a large period, till 2018, and all typologies of buildings, however, clothing adjustment was not considered. Finally, Salimi and Hammad [21] covered all OPA aspects, considering a time coverage from 2008 till 2018 and focusing on office buildings. Table 1 compares the main features of analyzed literature reviews and identifies the main gaps that the present study aims to fill.

1.2. Motivation and objectives

The overview of the state-of-the-art presented in Table 1 reveals a lack of review studies that cover thoroughly the different aspects of OPA modeling and the different building typologies, as well as the latest developments in this field. Therefore, standing as an addition to the work done in the IEA-EBC Annex 66 and embracing the new propositions of the IEA-EBC Annex 79, the main purpose of this study is (1) building an updated biographical database of the studies that have developed models on OPA, (2) based on analysis of this database, providing an overview of the scientific production and the current scientific landscape on OPA modeling, (3) identifying the key methods adopted in OPA modeling by considering different OPAs and by comparing documents that propose rule-base methods, data-driven methods, and a stochastic description of OPA, and (4) drawing a future outlook in OPA modeling.

2. Methodology

The purpose of this work is enabling a comprehensive analysis of the existing literature in the field of occupant behavioral modeling in building performance analysis. The presented systematic literature review is conducted following the PRISMA methodology, and the research question and the related literature search are built according to the guidelines proposed by Denyer and Tranfield [22]. Although the PRISMA methodology is a useful guideline for a critical development of systematic reviews, it is not an instrument that can automatically guarantee their quality [23]; thus, a large pool of experts from the IEA-EBC Annex 79 community has been involved in the planning, development and execution of this study. As such, the authors are aware of the possibility of relevant articles that might be missing in the review but are confident that the identified bibliographic database represents the main tendencies and approaches adopted into the field so far.

The PRISMA methodology considers four main phases: (1) identification, (2) screening, (3) eligibility, and (4) inclusion of studies. The summary of the PRISMA methodology is presented with a flow chart that shows the number of bibliographic records initially identified by the search query and subsequently included in this study (Fig. 1).

2.1. Identification of studies

The first step consists in constructing the research question. In this work, the CIMO-logic [24] is adopted, where CIMO stands for Context, Intervention, Mechanism and Output, and the research question is: "How do we model (M) the occupant presence and actions (I) to simulate the performance (O) of buildings (C)?" (Table 2).

Next, a comprehensive list of keywords is populated for each of the CIMO terms, and a research query is construct using the Boolean operators AND, OR and NOT and exploiting the list of keywords (1) to include all the keywords that have the same root but different declinations (e.g., for considering both British and American spelling), (2) to consider precise technical wording, (3) to exclude some divergent terms. Afterwards, exclusion criteria are applied to limit the search to usable documents in order to limit the search only to journal articles, conference papers, reviews, books, book chapters and articles in press written in English. Old articles and conferences proceeding not available anymore were also excluded. Finally, the search query is executed in the Scopus, Web of Science and EI Compendex databases. However, due to

Com	naricon	of	litoratura	roviowe	on	Occur	aant	Droconco	and	Actions	cinco	2015
COIII	Dalison	OI.	interature	reviews	on	Occu	pant.	Presence	anu	ACTIONS	since	2013

Authors	Year	Temporal	Typology of buildings	Occupant presence and actions							
		coverage		Presence	Movement activity	Window operation	Shading operation	Lighting operation	Thermostat adjustment	Appliance use	Clothing adjustment
Gunay, O'Brien, Beausoleil- Morrison	2015	Up to 2014	All	•		•	•	•			•
Yang, Santamouris, Lee	2016	Up to 2016	Institutional	•	•						
Gilani, O'Brien	2016	Up to 2015	Office	•		•	•	•	•	•	•
Chen, Jiang, Xie	2018	2005-2017	All	•							
Zhang, Bai, Mills, Pezzey	2018	Up to 2016	Residential and Commercial	•	•	•		•	•	•	•
Balvedi, Ghisi, Lamberts	2018	2006-2017	Residential	•	•	•	•	•	•	•	
Dong, Yan, Li, Jin, Feng, Fontenot	2018	Up to 2017	All	•		•	•	•	•		
Li, Yu, Haghighat, Zhang	2019	Up to 2018	All	•	•	•	•	•	•	•	
Salimi, Hammad	2019	2008–2018 +adding	Office	•	•	•	•	•	•	•	•



Fig. 1. Literature screening process following the PRISMA framework (Moher et al., 2009).

compatibility issues with the bibliometric tools, the file exported by EI Compendex could not be used. Furthermore, the files exported by Scopus and Web of Science could not be merged and, given the wider coverage, the Scopus file was eventually used for the literature search.

During the screening phase, the titles and abstracts of the identified documents were read, and several publications were excluded because not relevant. Afterwards, only studies with full-text were considered eligible for further analysis. Then, quality and consistency assessments were conducted by reading all the full-texts of the eligible documents. Those documents (i) not matching the research question, (ii) not relevant, (iii) without sufficient data, and (iv) presenting overlaps were also removed from the final database. Also, a few studies were removed due to overlap (e.g., the same set of data or models presented in both journal articles and conference papers). Finally, the bibliographic database was consolidated, and the bibliometric analysis were executed in Bibliometrix [25] to identify relationships between topics, patterns in the

Table 2

The CIMO-logic for studying modeling of Occupant Presence and Actions in buildings.

Context (Where? In which context the intervention is embedded?)	Intervention (What? Which is the main topic?)	Mechanism (How? Which is the medium?)	Outcome (To get what? What is the wanted information?)
Buildings (all building types)	Occupant presence and actions: • Presence and activity • Window operation • Shading operation • Lighting operation • Thermostat adjustment • Appliance use • Clothing adjustment	Modeling techniques: • Rule-based models • Stochastic OPA modeling • Data-driven methods	Outputs: • Energy performance • Indoor comfort

metadata of publications and thematic evolution.

2.2. Bibliometric analysis

The bibliometric analysis provides information on the relevance of the identified bibliographic records and uses science mapping to extract knowledge at the nexus among conceptual, intellectual and social structures.

2.2.1. Collaboration network

A collaboration network involves the analysis of authors' productivity, affiliations, and countries (of their affiliated organizations) and is represented on a map. It specifically deals with the scientific production disaggregated by country and the collaboration between authors with affiliations in each country. When a document is written by two authors whose affiliations belong to different countries, it is considered a collaboration.

2.2.2. Co-word analysis

A co-word analysis is a quantitative method for mapping the structure of a science field [26]. This technique analyzes the pattern of co-occurrence of pairs of words, which is the simultaneous occurrence of two words in a piece of text. The co-word analysis is performed by adopting clustering algorithms that identify the main themes characterizing the work under study. Outcomes of the co-word analysis are typically displayed with a co-occurrence network. The dimension of the node representing a keyword is proportional to its frequency of appearance in the analyzed bibliographic database, while the thickness of the connecting lines is proportional to the equivalent index e_{ij} is defined as $e_{ij} = c_{ij}^2/(c_i \, c_j)$, where c_{ij} represents the number of the documents in which beat the keyword sco-occur, c_i and c_j are the numbers of the documents in which each keyword appears.

3. Analysis of bibliographic metadata

In recent years, the interest on OPA modeling and the related scientific production have increased (Fig. 2) [5,27]. It should be noted that the literature search in this article was conducted in August 2019, therefore, the count for 2019 does not account for the documents published in the second half of the year.

The median of the publication year is 2015 and the average is 2013, in other words, a large share of the collected documents has mainly been published in the last four to six years. Specifically, there is a strong rise in published documents on OPA since 2010. By consequence, this review may be considered as an assessment of the current practice in OPA modeling in buildings. Looking at the temporal evolution of the published documents by source, the journals that have published more documents regarding OPA modeling in the latest years are *Building and Environment* and *Energy and Buildings*, followed by the *Journal of Building Performance Simulation* (Fig. 3).

Regarding the document production by country, the United States of America is the most productive country with 74 published documents from 1979 onwards. In addition, its collaborations are the most numerous (with 20 co-authored documents) and the most spread around the world (11 collaborations involve multiple countries) (Fig. 4). Europe, as a whole, is very productive with eight out of 16 countries having more than 10 publications (UK, Italy, Switzerland, Germany, Denmark, France, Belgium and Netherlands). European collaborations



Fig. 2. Annual scientific production of documents presenting Occupant Presence and Actions models. The count for 2019 considers only those documents indexed until August 2019.



Fig. 3. Chronological development of publication by sources.

are mostly internal, but there are also connections with countries from all continents.

4. Analysis of the documents on OPA modeling

The bibliographic collection is composed of 278 documents from 146 sources published from 1979 to nowadays. On average, each document is cited 46.1 times. The documents were written by 809 authors who appeared 1003 times as co-authors within this document collection with an average of 3.54 co-authors per document. These figures show a consolidated and spread international collaboration on this topic.

After the screening phase and having read all the full-texts, contextual data was extracted from the 278 documents and used to characterize the overall production of OPA models. Few documents propose more than one model and address more occupant actions; therefore, the number of models analyzed is up to 310. Fig. 5 displays aggregated figures on the number and percentage over the total number of collected models.

For the OPA model development, measurements are the most frequent data source. They represent a reliable manner to gather data and control uncertainty, but privacy issues may be encountered during the execution of measurement campaigns [28,29], typically when data collection happens in large buildings with general visitors for people-count purpose. From the analysis of the building use, offices are the most studied building type followed by residential units. In particular, the number of documents related to offices is around 60% higher than for residential buildings. This imbalance may be due to a more predictable occupant behavior in offices, an easier experimental setting, and a more direct transferability of models and results. In addition, the experiments on occupant behavior in offices can be less affected by privacy concerns when compared to the residential buildings. Naturally ventilated buildings are the most commonly researched building type and control strategy. This could be a result of the wider availability of collected data and the high variability of people interacting with a building and its devices, resulting more interesting from a model developmental perspective. However, several documents do not report explicit contextual information on the above three aspects and, hence, these descriptive statistics must be read as indicative figures.

All documents are also categorized on the base of the modeling approach used to develop the OPA models. It emerged that, in the last years, thanks to extended measurement campaigns and a higher wealth of available data, data-driven models are attracting increased interest for their capability to manage large data sources without missing the aleatory nature of OPA in buildings [30], followed by stochastic OPA modeling techniques, and rule-based methods. Next, the documents were grouped according to the Köppen-Geiger's climate classification system [31]. A high proportion of models are developed in temperate and continental climates identified respectively by the letters C and D



Fig. 4. The collaboration network map shows country collaborations and production.



Fig. 5. Graphical description of the Occupant Presence and Actions models collected in the bibliographic database (number of OPA models; percentage).

with 50% and 21% out of the total number of models respectively. Follow tropical climates (A) with 5% and arid climates (B) with 2%. In about 22% of the models, the climate condition was not mentioned.

The first five climatic zones by the number of developed OPA models represent almost the whole Europe, the USA and most populated portion of China (Fig. 6), which are also the most productive countries per number of publications.

4.1. Scientific landscape

Two main analyses are performed to describe the scientific landscape drawn by the bibliographic database: the three-field plot and the co-

occurrence network map. These analyses help to understand the research trends and the connections among the themes rising from the state of the art.

The three-field plot displayed in Fig. 7, shows the number of connections (size of the boxes) and strength of the connection (size of the connection lines) between most frequent words in abstracts (left field), Authors' Keywords (middle field) and scientific journals (right field).

The most frequent words in the abstracts point out the main and general terms of the research questions (like 'energy', 'building control'). In the middle field of the author's keywords, the main concepts on which the domain is built (like 'occupant behavior', 'thermal comfort', 'windows opening', 'lighting control', 'machine learning' and 'office





Fig. 6. Number of available OPA models by Köppen-Geiger's climate zones in the bibliographic database.

building') is presented. Finally, the main keywords as available in the journals are shown. For example, 'occupant behavior' is a very general term that is present in all the most representative journals, but 'thermal comfort' is mostly present in *Building and Environment* and *Energy and Buildings*, 'lighting control' is mostly related to *Solar Energy and Energy and Buildings*, and 'machine learning' is more present in *Applied Energy* and *Building and Environment*. This analysis provides insights to researchers new to the field to aid identifying the most suitable journals for publishing their studies.

The co-occurrence network in Fig. 8 shows the different clusters of Authors' Keywords, which are identified by the Walktrap clustering algorithm assuming 50 nodes and normalizing the relationships by the association strength [25].

The largest cluster (in green) collects the most traditional keywords (e.g., 'occupant behavior', 'office building', 'energy efficiency', 'thermal comfort') and some satellite terms typical of stochastic modeling. The second cluster (in red) pivots on 'neural network' and includes several data-driven topics like 'machine learning', 'data mining', 'prediction' and other term referring to widely used application like 'building management systems' and 'smart buildings'. The third cluster (in brown) is somewhat distant from the other terms and is very concentrated. It deals primarily with 'occupant presence' and includes terms like 'presence detection', 'number estimation', 'building occupancy' and 'cross-space modeling'. The orange cluster pivots on 'building automation' for 'building energy efficiency' together with 'occupancy detection' and 'activity recognition'. The blue cluster vertex on 'demand side management' and includes the terms 'demand response' and 'oc-cupancy'. The purple cluster focusses on 'intelligent lighting control', with terms like 'daylight harvesting' and 'smart lighting'. The keyword 'daylighting' is isolated but connected with 'lighting control' while 'vi-sual comfort' and 'indoor positioning' are isolated and not connected.

5. Explanatory and predictive power for occupant presence and actions modeling

In contrast to other scientific disciplines, the research on OPA requires models with both explanatory and predictive power, which represents a particular challenge. Motivated by the latter need for dual modeling objective, this section provides a comparison of the existing modeling formalisms for both causal explanation and predictive modeling.

OPA models were developed (1) to optimize the building design, (2) to represent the occupants in building performance simulation (BPS), and (3) to predict the human behavior for the inclusion in building control systems. The first two goals may be achieved by explaining the relationship between OPA and a set of objective measurements. For instance, by knowing the fixed working hours it may be understood the reason why an occupant was present at the workspace. Alternatively, the



Fig. 7. Evolution of the most frequent words in the abstracts (left field) to the keywords (middle field) and to the journal sources (right field) for the papers in the bibliometric database.

causal explanation of the intervention on sunshades may be visual discomfort that can be correlated with the solar radiation on the window surface. Here an important property of the chosen methods is to possess high exploratory power.

Regarding the third goal, OPA models for the application in building control require predictive power, in order to forecast the events or states on the future time-steps with satisfactory accuracy. In this place, models that possess high explanatory power are often assumed to inherently possess predictive power [32]. However, the research on statistical modeling pointed out that the distinctive models are required for prediction and causal explanation [32,33]. The need for non-identical methods for representing the impact of occupants in BPS and for predictive modeling has already been pointed out by Mahdavi and Tahmasebi [34], hence, this distinction has sometimes been overlooked by the modeling studies.

The causal explanation can be addressed using statistical and linear models [32]. The research on explaining occupant behavior has a longer tradition when compared to the predictive OPA in buildings modeling. Therefore, the set of statistical and linear models in use widely overlaps with the established general modeling formalisms that were reviewed by D'Oca et al. [35]. In addition to the methods proposed by the latter study (namely Bernoulli models, generalized linear models, and survival models), the generalized class of probabilistic graphical models, which also includes discrete Markov models, showed to be powerful tools for the research on human-building interaction. For instance, logistic regression and linear models have been applied to investigate the relationship between the thermal conditions and the resulting occupants' actions [36,37]. Furthermore, the results of the past exploratory studies on the human-building interactions led to a better hypothesis formulation regarding the drivers of occupant behavior as well as defining the baseline predictive OPA models.

The prediction of OPA has been commonly addressed using ML-

based methods. The literature screening has pointed out that the occupants' presence, activity recognition, and movement detection have been widely researched in the context of predictive modeling. For that purpose, the well-established modeling formalisms relied on probabilistic modeling, probabilistic graphical models, and conventional ML such as Support Vector Machine (SVM) and k-nearest neighbors (k-NN) algorithm. In the case of occupants' action prediction, different NN architectures have been investigated to model adaptive actions such as the use of lighting, solar shadings, windows, appliances, and clothing adjustment. The alternative widely explored methods include the conventional ML methods, such as k-NN, SVMs for classification and regression, as well as the variations of decision trees and ensembles of decision trees. The application of probabilistic methods and probabilistic graphical models led to promising modeling results for the application in the built environment. Hence, these classes of methods have not been comprehensively explored in the scope of existing OPA research. Moreover, stochastic models were also explored for their predictive capabilities for OPA. As a result, the logistic regression has been established as a baseline predictive model for window opening behavior, while in the scope of the recent study, the logistic regression showed promising results for learning the thermostat setpoints [38].

A first significant difference between the stochastic methods for the causality explanation and for the predictive modeling lies in the required data split. In the case of stochastic modeling, a set of data points is used to establish the hypothesis, while a set of distinct data points is eventually used to test the goodness of the hypothesis. Commonly, these two data sets were collected on the same occupant or on the same building, and the amount of available data is constrained by the design in terms of extent of the monitoring campaign [11]. Since these hypotheses widely address the relationship between the unique building design and the behavior, there are no strong requirements of the sample size.



visual comfort Fig. 8. Co-occurrence network of Author Keywords from papers in the bibliographic database.

An additional significant difference between the stochastic and ML modeling is the interpretability of models. Here, we refer to interpretability as the description of the internal rules of a system in a way that is understandable to humans [39]. Commonly, the ML models are developed to maximize the prediction accuracy and the results are often not interpretable using domain knowledge. This lack of interpretability has been seen as a major drawback for considering the ML approaches in the building design phase. However, as already pointed out by existing research, the most accurate explanations are not easily interpretable to people; and conversely, the most interpretable descriptions often do not provide predictive power [39]. Therefore, human interpretability is not a crucial property of the OPA models for inclusion in building control systems. Rather, the strict evaluation protocols in terms of models' effectiveness and the critical analysis of the predictive powers may be seen as the necessary components for the consideration of the ML methods in building control.

6. Modeling occupant presence

Human occupancy information is crucial for any modern building management system. The retrieved information can be utilized to understand both space utilization and building energy optimization, which enables informed decision making. Occupant presence is commonly declined in three sub-domains: occupancy detection, estimation and prediction; activity prediction and room occupation; and people movement between zones.

In this section, 53 documents published between 2004 and 2019 were analyzed. According to the developed bibliographic database, the annual scientific production in occupant presence modeling research reaches its peak (11 documents) during the period 2016–2018. The documents with most impact (in terms of a total number of citations) were published in Energy and Buildings. Next, there are documents published in journals with diverse scopes that do not belong to the core sources identified by Bradford's law, like *Energy Conversion and Management* and *Geodesy and Cartography*. These results point out that occupant presence modeling is a topic not exclusively related to energy

and indoor environmental research in buildings.

The data-driven models represent 56% of the total, followed by stochastic OPA modeling techniques (30%) and the rule-based models (14%). In particular, 27% of the data-driven models use NN techniques, 13% SVMs, and 11% Hidden Markov model (HMM). Regarding the stochastic OPA modeling techniques, 42% make use of Markov chain models, 17% of linear time series models, while 13% of the Monte Carlo method.

Fig. 9 shows the percentage of documents using a typology of methods on the overall documents published in that year considered in this review. In the last years, data-driven models are emerging compared to the other two typologies. A cause for that could be the increase of data wealth due to the digitalization of the building lifecycle, large sensors installation campaigns, and availability of smart meters.

6.1. Occupancy detection, estimation, and prediction

Occupancy detection usually refers to the binary inference of occupant presence and absence in different zones of an indoor or outdoor space while occupancy estimation usually refers to the occupancy count. Occupancy prediction is to forecast the in a future time window. Occupancy detection, estimation, and prediction are challenging tasks due to many reasons. For instance, there is a wide variety of sites of interest (such as individual and open plan workplaces, shopping malls, cinemas, etc.), which differ in size and operation mode. Hence, the appropriate contextual information must be considered for effective deployment of any system for occupancy detection, estimation, and prediction. Recent technological developments and the proliferation of pervasive technologies have opened up many opportunities to detect, estimate, and predict indoor occupancy leveraging various sensors and smart devices [40].

Many sensor-based technologies are available to detect and estimate occupancy in different types of sites [41]. A comprehensive review that compares the capabilities of different sensor types and their fusion for occupancy detection and estimation is presented in Ref. [17]. However, these technologies require extensive installation of hardware and continuous maintenance. Moreover, their accuracy can be influenced by specific physical orientation (i.e. seating, standing, walking styles) of occupants since the sensors are usually placed under the desk or overhead. To reduce the cost of extensive sensor installation, a probabilistic method for room-level occupancy counting is presented in Ref. [42]. This model utilizes common sensors available at different rooms for disaggregating accurate building-level occupancy counts to room-level occupancy counts. Another probabilistic fusion technique to estimate indoor occupancy from 3D camera counts is presented in Ref. [43]. Data from smart electricity meters is also used to detect the occupant presence [44,45]. The basic idea is to conduct cluster analysis on continuous variables, like power load, carbon dioxide (CO2) concentration, and estimate occupant presence. Another research highlights the use of different sensing systems including radio frequency, infrared, ultrasound, video cameras, and wireless local area network in recent literature [15]. However, these technologies are susceptible to surrounding electromagnetic conditions, inconsistent connections and may raise privacy concerns [15].

From the analysis of the developed bibliographic database, many state-of-the-art ML tools have been deployed to develop smart building applications which include occupancy detection, estimation, and prediction. Several classification models including Linear Discriminant Analysis, Classification and Regression Trees, and Random Forest models are evaluated for occupancy detection utilizing data from light, temperature, humidity and CO_2 measurements. The data coming from various smart sensors are utilized to provide real-time as well as future



Fig. 9. Yearly percentage of presence models with respect to the total number of published models belonging to the bibliographic database in each year.

predictions of occupancy status. However, it shall be mentioned that, since sensor data varies in dimensions and frequencies from one domain to another, a model trained for one domain cannot be applied effectively in another domain. To address this challenge, a semi-supervised domain adaptation method for CO₂-based human occupancy counter is presented in Ref. [46].

Finally, several evaluation metrics are used to validate the occupancy detection and estimation models including prediction accuracy, precision, recall, f-1 score, mean average error (MAE), mean average percentage error (MAPE), and root mean squared error (RMSE). However, it would not be fair to quantify the widespread use of a model and evaluation metric as the performance of a model generally depends on the specific application, size and quality of the data. For example, the deep learning-based models require a large dataset for better performance while compromising the interpretability. If the purpose is casualty analysis, it is possible that the statistical and ML models are a better choice over deep learning (DL).

The models discussed above are mainly developed and deployed using data from a specific site. Given the variety application scenarios, one of the key challenges is to transfer such models build for one site to another site as it may require extensive parameter tuning. In the future, efficient transfer learning methods could be adapted to mitigate this gap and more research effort needs to be given towards the adaptation of explainable machine learning and DL techniques. This will allow the research community and beyond to better understand the outcomes of the deployed models.

6.2. Occupant activity recognition

To adjust and operate control systems based on indoor occupant behavior, it becomes crucial for a building management system to recognize the indoor occupants' presence and its associated activities. The ability to identify or forecast a particular activity can minimize the exhaustion of unnecessary energy resources. Indeed, the difference in occupant activity might have a significant effect on the building's energy performance. Conservative behavior by occupants has been shown to save up to 30% of the building's energy consumption, while careless or reckless behavior can increase that amount by one-third [47]. Proper modeling of occupant activities is necessary to estimate building energy consumption and adjust the building's energy demands to optimize it [48]. Other notable uses of activity recognition and prediction include their use in health monitoring, to provide automated assistance and detect uncommon situations [49].

Activity recognition constitutes the monitoring of OPA along with the change of state in their environment. It is based on two main types of approaches, vision-based activity recognition and physical measurement-based or maybe environmental sensor-based. The former uses surveillance-based systems such as cameras [50], 3D-stereo vision systems [42], infra-red or depth registration [51], while the latter uses wearable or deployed sensors or RFID tags [52]. The typical solution for the detection of the occupant's activity involves a fusion of different environment monitoring techniques [53-56]. Most of the developed models are built on a foundation on quantity data but there are few examples that used quality-based data as the main development source [57]. Earlier works regarding the prediction of the occupant activities made use of probabilistic models and Bayesian belief networks [58]. Recent research efforts have also focused on Markov-chain models and HMM to estimate and forecast occupant activity levels [59,60]. Usually, most of the developed models are validated by ground truth data, obtained from visual observation via video recordings or notebook reporting [61]. Another development in the field of activity recognition and prediction is the use of DL methods for human activity recognition, where models are making use of Convolutional Neural Networks (CNN) [62-64] SVM [65,66], and Recurrent Neural Networks [67,68].

The main gaps for activity recognition are having a wider range of activities, since most of the research efforts to date have targeted a selected number of pre-defined activities [58,62–67]. In addition, the interdependence between activities has to be recognized as well [69]. Future efforts can be outlined to incorporate the personalization perspective for accurate activity recognition, along with adaptation with evolving activities, and context aware recognition [70].

6.3. People movement between zones

People's movement between zones is intended as the transition of occupants from one room to another inside a building. Occupants with their movement change also the sensible and latent loads between zones and so influence the temperature and humidity in rooms. This topic is fundamental for detailed building models, in which the spaces are described at room-level and, on average, occupancy probability assigned to all the rooms are too simplistic.

The bibliometric analysis suggests that the topic of detection and modeling of indoor movement of occupants is gaining momentum as it is strictly related to the topic of smart buildings. The indoor tracking of occupants is not a new field of research [71]. However, only in the last years, some descriptive and predictive models are emerging aiming specifically the better description of occupants for buildings energy modeling [72]. The description of the localization of occupants in real-time is fundamental for a large variety of smart buildings services; specifically, energy management and indoor environmental control [73]. For example, the proper load calculation due to occupants and their spatial distribution could avoid over-heating/cooling or under-heating/cooling of areas which is of a major importance especially for large public spaces [72,74]. Furthermore, these models could help to track and learn inhabitant's daily routine unobtrusively with the aim to optimize energy usage without affecting occupants' comfort [75]. Moreover, although satellite-based radio navigation systems are the common method that provides accurate track and modeling of movements outside buildings [76] and their use for positioning inside buildings is theoretically possible [77], it is difficult with traditional Global Positioning System (GPS) receivers to locate occupants in buildings [71]. Firstly, because the signal must be unobstructed, indeed conservative models suggest that the attenuation in buildings can reach levels of 2.9 dB per meter of structure [76]. Secondly, because this typology of systems requires the user to carry a tag.

Generalizing, the overall research process can be summarized into two consecutive tasks: people movement detection, identification, and localization, and people movement modeling for forecasting and simulation.

The literature relates mainly to the first task, in which arrays of binary sensors [78], environmental sensors [79], cameras [80], pressure sensors [81], inertial and vibration sensors [82,83], radio-frequency identification sensors [84], Bluetooth [75,85] and Wireless Local Area Network (WLAN) [86–88] are used to detect occupants and track their movements [89]. Generally, environmental sensors are the cheapest solution, but they provide less information about human movement, unless densely spread in the indoor space. Cameras or infrared sensors provide good accuracy, but they are usually expensive sensors with high maintenance costs and privacy issues. Pressure sensors, inertial and vibration sensors are usually employed under the floor, making the maintenance and the installation to be planned. Finally, the sensors like relying on Bluetooth or WLAN provide very detailed results, however, often they need that the occupant carries constantly a device.

The second task is usually performed with machine-learning algorithms that are able to learn representation from the data and use them to forecast, simulate and model the occupants' presence in rooms and their movements [74,75,90,91]. Some studies solve the simulation and forecasting via stochastic models, due to the lack of surveys and statistical information with proper detail [72,92].

To summarize, the topic of modeling people's presence, movement between zones and activity is relatively new, and ML methods are emerging as a promising approach to forecast, simulate, and model the occupants' presence in rooms and their movements inside buildings.

7. Modeling occupant actions

People interact with a building and its devices in various manners to meet individual needs. Occupant actions have a role in modulating energy fluxes exchanged by a space and the outdoors and, hence, have an important impact on the actual energy use in buildings and perceived occupants' comfort. In this study, considered occupant actions are windows operation, solar shading operation, electric lighting operation, thermostat adjustment, appliance use, and clothing adjustment.

7.1. Window operation

Window operation is an important control mechanism that, enabling physical connection with the outdoors, provides occupants with the ability to control the local indoor environment (i.e. regulate the indoor air quality and room air temperature). Moreover, since the '70s, building regulations are progressively increasing the energy conservation requirements of the building envelope with a reduction of infiltrations and conductive heat losses. Thus, the share of the ventilation losses on a building's overall energy balance is enlarging. In this context, window operations become even more important, and there is a high demand for window operation models that create realistic patterns for use in building energy simulations and for the predictive modeling for building control systems.

In this section, 43 documents published since 1990 were analyzed. According to the analysis of the developed bibliographic database, the control mechanisms, even though clearly influenced by physical conditions, tend to be governed by a stochastic rather than a deterministic relationship [93]. Stochastic models estimate an outcome by assuming a probabilistic relationship with one or more predictor variables. For modeling window opening behavior, the most common approach used so far are logit models and logistic regressions. These models can be used to predict the probability of a window's state (i.e. open or closed) [36, 94-104] or the probability that a certain action will occur (i.e. window opening or closing action) [105-108]. The former has been typically implemented with a Bernoulli process while the latter with a Markov process. A Bernoulli process [37] is a sequence of independent binary random variables where the current state has no impact on the future state; by definition, it ignores the actual dynamic processes leading occupants to perform actions. This limitation can be overcome using a Markov process [37,94,103,109-111], since it is a random process where future states are dependent only on a current state together with the probabilities of the state changing. However, to integrate these simulation approaches in a conventional BPS tool, since the time advances in fixed time steps, they have to be discrete (discrete-time random process). Therefore, the temporal resolution of predictions is limited (e.g., short duration openings could be ignored if they last less than the given time step). Furthermore, the time in which the active state (e.g., window closed) will be reversed is not predicted. To pose a solution, Haldi and Robinson [112] developed a hybrid approach: state transitions were predicted as Markov processes, while a continuous-time approach was employed through a survival analysis to estimate the time to reversal of the state.

Several studies implement NN and also DL has been used so far [113]. NNs are capable of learning the relationship between input signals and capturing key information through the training process based on historical records. Furthermore, they also possess a number of other strengths such as fault tolerance, robustness, and noise immunity [114, 115]. However, the architecture choice and hyperparameters optimization in the current NNs are still developed on an ad hoc basis. This implies that NNs applications are usually case dependent [116]. They have to be designed and validated each time for every different applications.

From the analysis of the bibliographic database, it was observed that

other ML techniques adopted to analyze window-opening behavior are based on a Gaussian distribution model (e.g. Ref. [95]), a Bayesian network (e.g. Ref. [117]), a cluster analysis and mining association rules (e.g. Ref. [118]).

Researchers have adopted different indices to evaluate the performance of their models, such as the true positive rate (TPR), true negative rate (TNR), the accuracy of the model (ACC), the mean absolute error (MAE), the mean signed deviation (MSD), and area under the curve (AUC). Consequently, there is a lack of horizontal comparison among these models. The motivation behind this difference is due to the fact that a convergence towards a systematic set of statistics for the prediction of the performance of behavioral models is missing. In this regard, Mahdavi and Tahmasebi [34] suggest two categories of indicators: indicators addressing agregate aspects of models' predictions, and indicators addressing the interval-by-interval congruence between predictions and measurements.

Following the Köppen climate classification scheme, the majority of the analyzed window opening models were developed in temperate climate zones Cfb (43%), Cfa (23%), Csa (2%), while the remaining in continental climate Dwa (16%) and Dfb (16%). Furthermore, most published studies referring to occupant window behavior have been carried out in European countries [37,94,102,105–107,109–113,117, 119–133]. Since window operation enables physical connection with the outdoor environment, it can be directly influenced by different conditions such as the atmospheric environment but also contextual factors such as routine/habits [134] and individual preferences [135]. It is therefore evident that in-depth research of window behavior in other climates and contexts is necessary.

While statistical models are a quite consolidate approach to model window operation (Fig. 10), data-driven models still requires further exploration, although DL has been recently used to investigate window operation [113].

7.2. Solar shading operation

Solar shading devices coupled with electric lighting are fundamental instruments to provide indoor thermal and visual comfort. The use of solar shading controls the internal daylight and influences the resulting solar heat gains. On one hand, solar shading can allow solar radiation to enter and passively heat the indoor environment, and on the other hand, it influences the operation of electric lighting that contributes to indoor sensible heat gains. Furthermore, solar shading is also used to provide privacy by blocking the view into a room from the outside.

In this section, 20 documents published since 1979 were analyzed. Solar shading operation is mainly modeled by predicting a shading state (or its change) as a binary variable (i.e. open or closed) [122,123,132, 136–142] or by estimating a shading device multi state [143–147]. Moreover, there are some specific models that predict the Venetian blind slat angle [148,149] and some others that couple the slat angle with the blind multi state [150,151].

The occupant-controlled shading devices has become of great interest in building performance simulation for different reasons.

A fundamental role in OPA models is played by the choice of the predictor variables. From the bibliometric analysis, it emerges that the most used predictors in shading control models are indoor and/or outdoor air temperatures [122,123,132,140,151], work plane daylight level [137,139,142,143], indoor illuminance [138,146,150], external radiation [146,150], and rainfall [122]. Since most of the models use external conditions as predictors, the climate in which the data for model construction are gathered is of great interest. In the analyzed bibliographic database, almost all models for shading operation come from temperate [123,137,139,140,146] and Continental [122,132,136, 138,141–143,147–151] climates, except for Kurian et al. [145] that worked in the tropics. Next, except from Andersen et al. [140] that predict shading movements in residential building, all other models are built for offices [122,123,132,136–139,141–143,147,149–151].



Fig. 10. Timeline of window operation models.

From the performed analysis came that the first shading control model was developed by Hunt in 1979 who used a stochastic method (Fig. 11). Since 2000, even data-driven methods have been used as accurate tools to predict the occupant-driven use of solar shading, with fuzzy logic and regularized logistic regression as the most used methods. NNs have been used for controlling the slat angle of Venetian blinds to optimize the energy consumption for lighting, and space heating and cooling [143,148,149], and also reinforced learning has been adopted to develop a controller to adjust both electric lighting and blind position [150].

In summary, models reflecting the operation of solar shading on tropical and arid climates are missing. Furthermore, more investigations should address residential and other types of buildings, providing a wider support for building energy modeling.

7.3. Lighting operation

In this section, 77 documents published between 1994 and 2019 and focused on electric lighting operation were analyzed. The analysis of the collected bibliographic records shows that, in the last 20 years, smart lighting control systems have been proposed to simultaneously satisfy personalized lighting levels and harvest natural daylight reducing energy consumption [152–154]. The first lighting controls were created such as on/off switch control or dimming by using sensors' outputs. Also, user-centric models based on occupants' location and their activities were used to define optimal lighting intensity level as a balance between user satisfaction and energy cost [155–157]. Lighting models that use sensor input (mostly occupancy and illuminance level) were primarily applied in office buildings. These models aimed to optimize the lighting conditions with respect to the work satisfaction and productivity [158,159]. NN technique was adopted in dwellings to

implement programming schedules of lighting control in Ref. [160]. With regard to the climatic conditions, the majority of the analyzed

investigations were developed in temperate climate zones Cfa (18%), Cfb (16%), Csa (12%), and some studies fall into the continental climate Dfb (12%). The main percentage of investigations (55%) was conducted in office buildings, followed by houses (17%) and laboratories (9%). The less analyzed building types are dormitories, hotels, and commercial buildings. Analyzing the type of data adopted for the models' development, it appears that the most common sources come from measurements (42%) and simulations (26%). Some documents adopt both measurements and simulations (18%). Surveys are rarely adopted alone, but they are typically coupled with measurements (9%) or with both measurements and simulations (4%). Regarding the models' categories, the highest percentage of identified documents belongs to the category of discriminative ML models (66%) followed by stochastic OPA modeling techniques and deterministic models that present similar applications. Some studies implement more than one model that falls into the same or into different typologies.

The most frequent category is the data-driven models [121,144,148, 149,152,155–158,160–194]], followed by the stochastic OPA modeling methods [153,159,161,184–190,195–212] and, then, the rule-based methods [139,157,160,206,213–223].

NNs allow forecasting multiple continuous variables based on design parameters because they are able to predict unique light use schedules for each design variant [172]. Furthermore, nonlinear transformation from input variables to output variables enables the designer to make predictions or classifications with regard to lighting controls [161,193]. However, their main drawbacks are that it takes too much time for the training phase [161] and needs to be trained again if the layout of any lamp is changed [163,165,166]. Regression models can help in predicting the lighting consumption of buildings [210] by providing an



Fig. 11. Timeline of solar shading operation models.

accurate estimation of the energy consumption compared to the results obtainable with extrapolation methods that use data from office lighting systems [191]. Furthermore, regression models were used to predict a state (i.e. on/off) (e.g. Ref. [212]), to estimate the probability of light switch actions (e.g. Ref. [207]), and the interactions with window shades (e.g. Ref. [206]). Rule-based models are a simpler manner to set a lighting control strategy and, in the case of large datasets, they provide acceptable results when compared with stochastic OPA models [206].

The historical overview shows an increasing development of models since 2004 (Fig. 12). Rule-based models like schedules and profiles were implemented for this intervention [215,216]. Logit model [224] was the first technique used to describe stochastically OPA behavior in European countries and Pakistan [195], but its application was time limited. Successively, there was the implementation of Markov chain model [159,205]. Since 2005, NNs [225] have become the most used data-driven method due to their abilities to learn from input data and the breakthroughs made in computing power at the beginning of the 20th century. Other methods for lighting modeling, for example, SVMs and decision tree, have emerged since 2010, but are relatively less used than NNs. As a prediction method, linear regression is easy to use, and the historical use rate is similar to SVMs and decision trees.

Researchers validated their models by means of different evaluation metrics: error or accuracy [149,155,164,166,170,177,179,180,186, 201–204]; comparison between the performances of the proposed system and the existing system in terms of energy saving or illumination level [139,153,173,174,181,191,192,194,199,206,221]; MSE [155–157,161,163,168,193,208]; RMSE [121,152,166,183,184,208, 210,211]; statistical parameters such as standard deviation, kurtosis, and skewness [165,167,197,209].

The analysis of the existing literature showed that the research about electric lighting modeling was mainly conducted in locations

characterized by temperate climatic conditions. Nevertheless, the user's interaction with electric lighting is influenced by the daylight availability that depends on local sky conditions and latitude. This limitation can negatively affect model's generalization and suggests future studies in diverse geographical contexts.

Also, offices were the most investigated indoor environments due to the easiness to apply sensors and collect measured data. Thus, research should be dedicated to residential, educational, and commercial buildings.

Discriminative ML models were widely developed and tested, stochastic and deterministic models require more investigation in order to verify their efficacy. Generally, accurate analyses about user's habits, preferences, and perceptions of indoor conditions are missing and so investigations could be improved by administrating targeted surveys during the monitoring phase.

7.4. Thermostat adjustment

Thermostat adjustment behavior is a key component of building performance modeling as it directly influences the amount of energy used for space Heating, Ventilation, and Air-Conditioning (HVAC) systems. Thermostats are used as control devices to determine when space heating, cooling, or ventilation should be applied to a building thermal zone. Thermostats typically include sensors that measure the air temperature or humidity of the building thermal zone and will request space heating, cooling, or ventilation if the indoor climate is above or below a set-point value. The occupants within buildings interact with a thermostat by adjusting the set-points for temperatures and humidity and by setting schedules for when the HVAC systems should be active and inactive. Thus, the occupant behavior (setting the set-points and the schedules) is one factor determining when an HVAC system switches on



Fig. 12. Timeline of models for operate electric lighting.

and off; other factors include the many thermal processes which influence the indoor climate such as the thermal properties of the building envelope, the internal heat gains and the capacity of the HVAC.

The choice of thermostat set-points and operation schedules by the building simulation modeler will have a significant impact on the predictions of energy use and occupants' thermal comfort. This is a key factor of the performance gap as international and national building performance standards and calculations often assume constant, simplistic occupant behavior for the thermostat control. In reality, many occupants will continually adjust the thermostat set-points and schedules depending on when they are at home or at work, the external weather conditions and for occasions such as holidays. The difference between these assumptions and the actual occupant behavior may lead to significant uncertainty in the predictions of building energy use [226].

In this section, 44 documents published after 1989 are analyzed. The occupant behavior modeling methods have been identified in the developed bibliographic database (Fig. 13). The most used methods include General/generalized linear model (33%) [227,228], Markov chain models (23%) [229,230] and logit analysis (20%) [121,132]. The studies are based on a wide range of buildings such as residential buildings (54%), offices (26%), commercial buildings, educational buildings (7%), and commercial buildings (6%). Measurement campaigns are used to collect training and calibration data for model development, including internal temperatures (set-point and indoor air temperature), occupancy/presence, heating/cooling/ventilation energy demand, and outdoor weather. For residential applications, it can be difficult to directly measure thermostat set-points and schedules (as this requires a direct interface with the control equipment) and often indirect measurements are used as a proxy such as estimating thermostat settings using the zone air temperature [132,231,232]. This further adds

to the uncertainty of the model predictions. In the numerous studies in this field, there is no agreement on the choice or amount of measurement variables that are required to construct the occupant behavior models or the choice of evaluation metrics which should be employed to validate the models.

In connection to the previously described thermostat set point adjustments, the occupants' interactions with the HVAC systems have also been explored in residential [227,233-235] as well as in office and commercial contexts [236,237]. As a result, the use of the HVAC in residential buildings has been conducted using approaches such as Markov transfer probabilities [227] and descriptive statistics [233-235]. In the case of commercial buildings, the application of logistic regression and rule-based agent models have been identified as a suitable modeling approach [236]. This study pointed out that both logistic regression-based models and the agent-based framework could identify approximately 50% of the fan use or heater use events correctly, while the proportion of the false positive rate remained around 20%. In another study of office and commercial buildings, NN was evaluated for performance among four different machine-learning algorithms [237], which actively learned occupants' interactions with thermostats under dynamic, time and space varied contexts. For a period of five months, the interaction model was conducted to an HVAC system in the case study building. The results reported 4%-25% energy consumption reduction as compared to static temperature set points at the low values of the preferred temperature range.

Significant further work is required in this area. The field of OPA thermostat set-point modeling is underdeveloped in relation to other OPA areas because of the challenges in collecting thermostat data (in residential settings) and in modeling the complex interrelated effects of occupant thermal comfort, building thermal response and dynamic external conditions. A clear data collection methodology and



Fig. 13. Timeline of models for thermostat adjustment.

standardized model testing framework needs to be developed, with clear reporting criteria and evaluation metrics. To maximize the potential of existing and future datasets, a common data collection vocabulary or ontology should be created which would enable data reuse and ultimately meta-analysis of multiple datasets across different building types, sample sizes and country of origin.

7.5. Appliance use

Appliance are electrical devices that support people's daily life, ranging from small machines (like laptop computers, air purifiers, coffeemakers and microwaves) to large ones (like fridges, clothes washers and dryers). Especially in the residential sector, appliances become one type of key electricity consumers. The energy demand for household appliances is growing as rising living standards worldwide [238]. Human behavior has an impact on appliance operation and spurs the associated energy consumption within buildings. Better understanding such activities offers potentials to operate appliances and their energy supplies (including the power grid and renewable energy) in an efficient way. Measuring and modeling appliance usages triggered by occupants, if properly visualized and communicated to together with suggestions, can promote energy-saving awareness [239]. Yu et al. [240] proposed a data mining-based method for estimating the saving potentials related to standby energy use considering the occupant behavior. Meanwhile, energy/load management based on appliance operation minimizes the variation of power supply [241], shifts appliance operation from the peak electricity demand [242] and makes appliance adapt to changes in electricity price [243].

In this section, 36 documents published since 1994 were analyzed. They describe models for identifying and modeling appliance states that were based on measurements, simulation, and surveys. Overall, the majority of the data used is measured data from field studies and home applications (69%). The studies were undertaken mostly in temperate climates (Cfa 39%, Cfb 33%, Csb 6%) with some models in continental (Dfb 11%) and arid climates (Bsh 6%, Bwh 6%). Sensing infrastructure for the data collection differed for the individual studies. It included four distinct groups of sensing devices: energy-related measurement (power, voltage, and current meters); communications technology (barcode and Bluetooth); environmental sensing (temperature, carbon monoxide, and acoustic sensors); and activity-related sensing (triaxial accelerometer and gyroscope, motion, door, and ultrasonic positioning sensors). Among them, power meters installed at the main power inlet of households were widely used by the studies as predictors.

Appliances are operated in on/off or multi states. Identifying their states was mainly described stochastically or predicted with data-driven methods. The former approaches use Bayesian networks [244,245] and hierarchical clustering models [246]. The latter use two different machine-learning-based algorithms: HMMs [239,247-249] and NNs [250-252]. To model occupants' indoor behavior and activities in interaction with appliances, diverse algorithms were employed in the studies, such as pedestrian dead reckoning [253], Bayesian network mode and linear regression [254], k-means and Gaussian mixture [69], random forest [255], and SVMs [256]. According to power usage of appliances, Gaussian mixture [257], k-means [258], optimization based on defined objective function [243] were used to infer load distribution and scheduling for systems. Similarly, power data showed potentials to extract building occupancy using data-driven approaches, such as decision trees [259] and NNs [260]. Two studies used both power data and occupant surveys [261,262]. Based on such data, the former study aimed to identify occupant behavioral predictors using a linear method, and the latter employed a Gaussian mixture method to model load patterns of the appliance in offices. For appliance controls in

households, NNs [263] and stochastic sliding mode control [241] were utilized. As shown in Fig. 14, most of the studies were based on recognition of appliance states and associated occupant activities using data-driven models.

Evaluation metrics applied to verify the above behavior modeling for appliance uses included precision, recall, F-score, RMS, RMSE, NRMSE, MAE, distance and positioning accuracy, and variances of positioning errors.

Most of the studies focused on one type of data (i.e. total electricity consumption of individual buildings or households) or one case study with several specified appliances. In actual buildings, diverse appliances are used by occupants which are affected by the purposes of the buildings (for example, residential and commercial buildings), and occupants' requirements. Meanwhile, occupant behavior interacting with appliances differs from device to device and person to person. In future research, one of the key research questions could be how to generalize methodologies for different appliance applications.

7.6. Clothing adjustment

Clothing has been considered as a critical interface between humans and their surrounding environmental settings [264–267] and is an influential input parameter in a few thermal comfort models. According to current knowledge, age, gender, and relative humidity have no significant effect on the clothing insulation levels chosen by people [267]. However, Humphreys [267] stated that the outdoor daily mean temperature to be the most crucial parameter affecting clothing insulation levels. Studies before this one had studied clothing insulation using conventional linear regression approaches. Deng and Chen [264] argued that the association between clothing and potential factors that affects clothing behavior might not be linear, hence, they developed clothing prediction models using ordinal logistic regression and NN using data collected in offices. The training accuracies of the NN model for three kinds of actions (lowering the set point or reducing the clothing level, no response, and raising the set point or adding clothing) were 89.4%, 87.3%, and 91.2%, respectively, and its overall training accuracy in predicting all three kinds of behaviors was 87.5%, resulting in an accurate tool for predicting occupants' behavior in the offices.

The main predictors used for the clothing adaptation in the existing literature include indoor air and operative temperature, relative humidity, CO_2 , air velocity, outdoor air temperature, skin temperature, human activities and time of the day [268].

The common evaluation metrics used in the existing literature are R², RMSE, MAE, MAPE for the regression models, and accuracy, F1-score, precision, and recall for the classification models.

Most of the existing methods in clothing insulation estimation assume the values to be fixed by using in-situ clothing estimation methods, thermal models, or depending on the outdoor air and indoor operative temperatures [265–267,269,270]. Also, some data-driven methods (e. g., NNs, SVMs, and regression models) have been used to establish thermal comfort inside a built environment [264–267,269–271]. These data-driven models reflect the occupants' responses and interactions



Fig. 14. Timeline of models for appliance operation.

with the building utilities and management facilities. Recently, the focus has shifted toward applying ML and DL models for predicting indoor clothing levels [264,267].

From the literature, it is affirmative that the clothing adaptation to any given situation is associated to three influencing factors: occupant behavioral adjustment, physiological factors, and psychological factors [265–267]. Therefore, for future research directions, the interrelationships and correlations between different influencing factors can be studied meticulously in those building types not already analyzed and under different individual conditions, like different metabolic activity levels.

7.7. Combined occupant actions

Researchers have also developed models that combine more than one user's actions with the aim of analyzing the multiple aspects of comfort and energy consumption in buildings. Lighting operation is one of the most co-modeled aspects due to its impact on both visual comfort, thermal comfort, and electricity demand. For example, schedules/profiles, stochastic OPA modeling techniques, and data-driven models were implemented by combining lighting operation with shading control [139,141,144,148-151,154,192,272]. Regression models were also exploited for modeling different combined actions: light switching with window operation [121], window and solar shading operation [123], and light switching with both window and solar shading operations [122]. With the aim of analyzing visual discomfort, data-driven techniques were used in Refs. [176,177] to model lighting switch in combination with blinds operation and change of the space heating set-point temperature. Furthermore, light switching and window operation combined with space heating and cooling operation were also modeled by means of schedule/profile and stochastic OPA models [185,215]. Moreover, data-driven models [210] and stochastic OPA models [196] were implemented to predict the energy consumption of buildings by considering both lighting and appliances use. More recently, Haldi et al. [122] investigated the combined operation of windows, solar shading, and light switching and developed logit models for residential buildings and offices. These models included random effects for all predictors that account the inter-individual variability in behavior among different occupants. This attempt allows overcoming the issue of modeling an occupants' average behavior and explicitly considering diversity and variability in occupant behavior.

Modeling combined actions seems a more effective approach providing a wider view of human actions and their impact in terms of energy consumption and occupants' comfort. The available literature still demonstrates gaps in this development and intersectional studies should be encouraged.

8. Future outlook in OPA modeling

Among the studies grouped under the data-driven models, there is a subset of studies recently published [54,62–64,67,113,252,264,267, 273–275], which use DL techniques. DL is adopted for obtaining rich information about occupant behavior and is proven to be competent in extrapolating discriminatory features from raw sensor data accumulated from building management systems [273–276]. Traditional ML approaches perform tasks without exploiting the correlations between diverse input sensor data. For example, CNN tries to overcome this issue by implementing convolution across n-dimensional temporal sequence to apprehend the dependencies in the input sensor data. However, the size of the kernel is an important parameter that can restrict the range of captured dependencies in the input sensor data for the CNN model [276]. Other advances in embracing dDL methods are:

 ML classifiers rely heavily upon heuristic handcrafted features (i.e. the manual selection of features) and require expertise in domain knowledge. The manual selection of features could lead to inductive bias, because the algorithm uses inputs that it has not yet encountered to predict the target outputs. Typically, such bias is supplied by hand through the dexterity and insights of domain experts. Advancements in DL make it possible for automated feature extraction and selection, thus overcoming the inductive bias [273–276].

- Shallow features can be recognized well with ML but a difficulty in identifying context-aware activities of occupant behavior (e.g., cooking a meal) or extracting other dimensions of occupant behavior [20,277–279].
- 3) In traditional approaches, extensive training data and labeled annotations are mandatory for supervised learning, but in real-world applications, most of the data remain unlabeled (unsupervised). Due to this, typical models are unadaptable to a diverse range of context-aware occupant actions and model configurations [20,45, 275–279].
- 4) Another significant difference between DL and ML methods is the problem-solving capability and critical analysis approach. DL tends to solve the issue end-to-end, whereas ML needs the problem statement to be broken into stages/parts and explained separately and combined at the final phase.

In summary, unlike ML approaches, DL classifiers are trained through feature learning rather than distinct task-specific algorithms [276]. However, DL is applicable when the task indented has a large dataset to work with; for smaller datasets, ML algorithms performs well with high accuracy. In general, when there is a lack and inadequacy of domain knowledge for feature introspection, DL outperforms most of the existing ML techniques [20,275–278].

9. Conclusions

In this study, the PRISMA methodology is exploited to conduct a systematic literature review on the topic of Occupant Presence and Actions (OPA) modeling in buildings. The identified documents were collected in a bibliographic database and analyzed. The analysis was supported by a data-driven bibliometric tool to provide an extended investigation of the methods and findings on the topic and to draw insights into the current state and future prospects of OPA modeling. This work, in the context of IEA EBC Annex 79, aimed to systematically cover all aspects of OPA modeling in different typologies of buildings.

The bibliometric analysis showed that the most productive geographic regions are North America, Europe, and China and that the intensity of the collaborations is large and well established between research groups in such regions. The documents analyzed in the database mainly involved measurement data in office buildings located in temperate and continental climates. Therefore, there is a need to develop new research studies outside these consolidated domains to provide a wider coverage of the knowledge domain, specially, in those climate contexts where models are missing, and it is expected a substantial increase of population and the construction rate (e.g., Africa, Indo-China region, Latin America). Regarding the methods, data-driven models are emerging as the most used modeling methods in recent years, which may be due to the large wealth of data coming from sensors installation. In particular, there is a recent interest in adopting deep learning techniques to model some OPA aspects for both explaining and predicting purposes. Most of the studies on occupant presence and activity detection aim at understanding occupant behavior, while the majority of studies on occupant actions are aiming at predicting occupants' interaction with given building devices for adaptive controls' development. It is highly appreciated the development of combined occupant behavioral models that provide a wider and closer-to-reality description of occupant use of the building and its systems. This is a domain where newer research is needed to increase accuracy of behavioral modeling.

In general, to maximize the potential of existing and future datasets, a common data collection vocabulary or ontology should be created

which would enable data reuse and ultimately meta-analysis of multiple datasets across different building types, sample sizes and country of origin.

This review has to be intended as a work to be regularly updated and expanded with the rise in number and detail of the OPA modeling methods to provide information on developments and new tendencies in the field. To facilitate this task, this article provides a dynamic openaccess review table as a supplementary material (https://osf. io/gnvp2/?view_only=00b08233881f471795d1d8dee79e9828), which can be expanded by other researchers to include future studies in order to represent an updated overview on the scientific production on occupant presence and action modeling.

Limitations of the current work are the possible and involuntary omission of OPA modeling documents not spotted by the literature search and not at the knowledge of the authors. However, the PRISMA methodology is designed to keep such oversights to a minimum.

Acknowledgments

This work has been performed within the framework of the International Energy Agency – Energy in Buildings and Communities Program (IEA-EBC) Annex 79 "Occupant-centric building design and operation". The authors would like to thank Prof. William Tolone for the useful inputs to the analysis.

Salvatore Carlucci and Matteo Favero would like to thank the Research Centre on Zero Emission Neighborhoods in Smart Cities (FME ZEN, Grant n. 257660) and the Research Council of Norway (Norges Forskingsrådet) for the support. Romana Markovic's and Christoph van Treeck's participation was funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) - TR 892/4-1). Gianmarco Fajilla would like to thank the Calabria Region Government for his Ph.D. scholarship (POR Calabria FSE/FESR 2014-2020, Grant n. H21G18000170006). Mikkel Baun Kjærgaard and Anooshmita Das would like to acknowledge funding by the Energy Technology Development and Demonstration Program (Energistyrelsen, Grant n. 64018-0558). Jakob Hahn gratefully acknowledges the financial support provided by the Federal Ministry for Economic Affairs and Energy (Bundesministerium für Wirtschaft und Energie, Grant n. 03ET1299A--EnEff Stadt: +EQ-Net and Grant n. 03ET1648A-EnOB: NuData Campus). During the time of writing, Flora Salim is supported by Humboldt and Bayer Foundation. Any opinions, findings, conclusions, and recommendations expressed in this article are those of the authors and do not necessarily represent the views of the funding agencies.

References

- I. Gaetani, P.J. Hoes, J.L.M. Hensen, Estimating the influence of occupant behavior on building heating and cooling energy in one simulation run, Appl. Energy 223 (2018) 159–171, https://doi.org/10.1016/j.apenergy.2018.03.106
- [2] W. O'Brien, I. Gaetani, S. Gilani, S. Carlucci, P.J. Hoes, J. Hensen, International survey on current occupant modelling approaches in building performance simulation[†], J. Build. Perform. Simul. 10 (2017) 653–671, https://doi.org/ 10.108/19401493.2016.1243731.
- [3] K.U. Ahn, D.W. Kim, C.S. Park, P. de Wilde, Predictability of occupant presence and performance gap in building energy simulation, Appl. Energy 208 (2017) 1639–1652, https://doi.org/10.1016/j.apenergy.2017.04.083.
- [4] Y. Zhang, X. Bai, F.P. Mills, J.C.V. Pezzey, Rethinking the role of occupant behavior in building energy performance: a review, Energy Build. 172 (2018) 279–294, https://doi.org/10.1016/j.enbuild.2018.05.017.
- [5] D. Yan, T. Hong, B. Dong, A. Mahdavi, S. D'Oca, I. Gaetani, X. Feng, IEA EBC Annex 66: definition and simulation of occupant behavior in buildings, Energy Build. 156 (2017) 225-8270, https://doi.org/10.1016/j.enbuild.2017.09.084.
- [6] International Energy Agency, IEA EBC Annex 79 Occupant-Centric Building Design and Operation, 2017.
- [7] ANSI/ASHRAE/IES Standard 90.1-2016: Energy Standard for Buildings except Low-Rise Residential Buildings, 2016.
- [8] D. Yan, W. O'Brien, T. Hong, X. Feng, H. Burak Gunay, F. Tahmasebi, A. Mahdavi, Occupant behavior modeling for building performance simulation: current state and future challenges, Energy Build. 107 (2015) 264–278, https://doi.org/ 10.1016/j.enbuild.2015.08.032.
- [9] M. Schweiker, S. Carlucci, R.K. Andersen, B. Dong, W. O'Brien, Occupancy and occupants' actions, in: Explor. Occupant Behav. Build. Methods Challenges,

Building and Environment 174 (2020) 106768

Springer International Publishing, 2018, pp. 7–38, https://doi.org/10.1007/978-3-319-61464-9 2.

- [10] S. Formentin, K. Van Heusden, A. Karimi, A comparison of model-based and datadriven controller tuning, Int. J. Adapt. Contr. Signal Process. 28 (2014) 882–897, https://doi.org/10.1002/acs.2415.
- [11] R. Markovic, D. Wölki, J. Frisch, Treeck van, Transaction from Stochastic Modelling to Supervised Learning of Occupant's Behavior, Bauphysiktage Kaiserslautern. 2017.
- [12] D.P. Solomatine, A. Ostfeld, Data-driven modelling: some past experiences and new approaches, in: J. Hydroinformatics, IWA Publishing, 2008, pp. 3–22, https://doi.org/10.2166/hydro.2008.015.
- [13] D. Yan, T. Hong, Definition and Simulation of Occupant Behavior in Buildings Annex 66 Final Report, 2018.
- [14] H.B. Gunay, W. O'Brien, I. Beausoleil-Morrison, Implementation and comparison of existing occupant behaviour models in EnergyPlus, J. Build. Perform. Simul. 9 (2016) 567–588, https://doi.org/10.1080/19401493.2015.1102960.
- [15] J. Yang, M. Santamouris, S.E. Lee, Review of occupancy sensing systems and occupancy modeling methodologies for the application in institutional buildings, Energy Build. 121 (2016) 344–349, https://doi.org/10.1016/j. enbuild/2015.12.019.
- [16] S. Gilani, W. O'Brien, Review of current methods, opportunities, and challenges for in-situ monitoring to support occupant modelling in office spaces, J. Build. Perform. Simul. 10 (2017) 444–470, https://doi.org/10.1080/ 19401493.2016.1255258.
- [17] Z. Chen, C. Jiang, L. Xie, Building occupancy estimation and detection: a review, Energy Build. 169 (2018) 260–270, https://doi.org/10.1016/j. enbuild/2018.03.084.
- [18] B.F. Balvedi, E. Ghisi, R. Lamberts, A review of occupant behaviour in residential buildings, Energy Build. 174 (2018) 495–505, https://doi.org/10.1016/j. enbuild/2018.06.049
- [19] B. Dong, D. Yan, Z. Li, Y. Jin, X. Feng, H. Fontenot, Modeling occupancy and behavior for better building design and operation—a critical review, Build. Simul. 11 (2018) 899–921. https://doi.org/10.1007/s12273-018-0452-x.
- [20] J. Li, Z. (Jerry) Yu, F. Haghighat, G. Zhang, Development and improvement of occupant behavior models towards realistic building performance simulation: a review, Sustain. Cities Soc. 50 (2019), https://doi.org/10.1016/j. soc.2019.101685
- [21] S. Salimi, A. Hammad, Critical review and research roadmap of office building energy management based on occupancy monitoring, Energy Build. 182 (2019) 214-241, https://doi.org/10.1016/j.enbuild.2018.10.007.
- [22] D. Denyer, D. Tranfield, Producing a systematic review, Sage Handb. Organ. Res. Methods. (2009) 671–689.
- [23] A. Liberati, D.G. Altman, J. Tetzlaff, C. Mulrow, P.C. Gøtzsche, J.P.A. Ioannidis, M. Clarke, P.J. Devereaux, J. Kleijnen, D. Moher, The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration, PLoS Med. 6 (2009), e1000100, https://doi.org/10.1371/journal.pmed.1000100.
- [24] D. Denyer, D. Tranfield, J.E. Van Aken, Developing design propositions through research synthesis, Organ. Stud. 29 (2008) 393–413, https://doi.org/10.1177/ 0170840607088020.
- [25] M. Aria, C. Cuccurullo, bibliometrix: an R-tool for comprehensive science mapping analysis, J. Informetr. 11 (2017) 959–975, https://doi.org/10.1016/j. joi.2017.08.007.
- [26] H. Qin, Knowledge discovery through Co-word analysis, Libr. Trends 48 (1999) 133–159.
- [27] H. Polinder, M. Schweiker, A. Van Der Aa, K. Schakib-Ekbatan, V. Fabi, R. Andersen, N. Morishita, C. Wang, S. Corgnati, P. Heiselberg, D. Yan, Total Energy Use in Buildings (Final Report Annex53 - Occupant Behavior and Modeling), 2013.
- [28] K. Amasyali, N.M. El-Gohary, A review of data-driven building energy consumption prediction studies, Renew. Sustain. Energy Rev. 81 (2018) 1192–1205, https://doi.org/10.1016/j.rser.2017.04.095.
- [29] M.J.N. Oliveira Panão, M.C. Brito, Modelling aggregate hourly electricity consumption based on bottom-up building stock, Energy Build. 170 (2018) 170–182, https://doi.org/10.1016/j.enbuild.2018.04.010.
- [30] A. Capozzoli, M.S. Piscitelli, S. Brandi, Mining typical load profiles in buildings to support energy management in the smart city context, in: Energy Proceedia, Elsevier Ltd, 2017, pp. 865–874, https://doi.org/10.1016/j.egypro.2017.09.545.
- [31] W.P. Köppen, R. Geiger, Handbuch der Klimatologie, Gebrüder Borntraeger, Berlin, 1930.
- [32] G. Shmueli, To explain or to predict? Stat. Sci. 25 (2010) 289–310, https://doi. org/10.1214/10-STS330.
- [33] L. Breiman, Statistical modeling: the two cultures, Stat. Sci. 16 (2001) 199–215, https://doi.org/10.1214/ss/1009213726.
- [34] A. Mahdavi, F. Tahmasebi, On the quality evaluation of behavioural models for building performance applications, J. Build. Perform. Simul. 10 (2017) 554–564, https://doi.org/10.1080/19401493.2016.1230148.
- [35] S. D'Oca, H.B. Gunay, S. Gilani, W. O'Brien, Critical review and illustrative examples of office occupant modelling formalisms, Build. Serv. Eng. Technol. 40 (2019) 732–757, https://doi.org/10.1177/0143624419827468.
- [36] H.B. Rijal, P. Tuohy, M.A. Humphreys, J.F. Nicol, A. Samuel, J. Clarke, Using results from field surveys to predict the effect of open windows on thermal comfort and energy use in buildings, Energy Build. 39 (2007) 823–836, https:// doi.org/10.1016/j.enbuild.2007.02.003.

- [37] M. Schweiker, F. Haldi, M. Shukuya, D. Robinson, Verification of stochastic models of window opening behaviour for residential buildings, J. Build. Perform. Simul. 5 (2012) 55–74, https://doi.org/10.1080/19401493.2011.567422.
- [38] H.B. Gunay, W. O'Brien, I. Beausoleil-Morrison, J. Bursill, Development and implementation of a thermostat learning algorithm, Sci. Technol. Built Environ. 24 (2018) 43–56, https://doi.org/10.1080/23744731.2017.1328956.
- [39] L.H. Gilpin, D. Bau, B.Z. Yuan, A. Bajwa, M. Specter, L. Kagal, Explaining Explanations: an Approach to Evaluating Interpretability of Machine Learning, 2018.
- [40] M.B. Kjærgaard, B. Dong, S. Carlucci, F.D. Salim, J. Yang, C.J. Andrews, O. Ardakanian, Poster abstract: data-driven occupant modeling strategies and digital tools enabled by IEA EBC Annex 79, in: BuildSys 2018 - Proc. 5th Conf. Syst. Built Environ, Association for Computing Machinery, Inc, 2018, pp. 188–189, https://doi.org/10.1145/3276774.3281015.
- [41] B. Dong, M.B. Kjærgaard, M. De Simone, H.B. Gunay, W. O'Brien, D. Mora, J. Dziedzic, J. Zhao, Sensing and data acquisition, in: Explor. Occupant Behav. Build. Methods Challenges, Springer International Publishing, 2017, pp. 77–105, https://doi.org/10.1007/978-3-319-61464-9 4.
- [42] M.B. Kjærgaard, M. Werner, F.C. Sangogboye, K. Arendt, DCount a probabilistic algorithm for accurately disaggregating building occupant counts into room counts, in: Proc. - IEEE Int. Conf. Mob. Data Manag, Institute of Electrical and Electronics Engineers Inc., 2018, pp. 46–55, https://doi.org/10.1109/ MDM.2018.00021.
- [43] F.C. Sangoboye, M.B. Kjærgaard, PLCount: a probabilistic fusion algorithm for accurately estimating occupancy from 3D camera counts, in: Proc. 3rd ACM Conf. Syst. Energy-Efficient Built Environ. BuildSys 2016, Association for Computing Machinery, Inc, 2016, pp. 147–156, https://doi.org/10.1145/2993422.2993575.
- [44] D. Mora, G. Fajilla, M.C. Austin, M. De Simone, Occupancy patterns obtained by heuristic approaches: cluster analysis and logical flowcharts. A case study in a university office, Energy Build. 186 (2019) 147–168, https://doi.org/10.1016/j. enbuild.2019.01.023.
- [45] F. Causone, S. Carlucci, M. Ferrando, A. Marchenko, S. Erba, A data-driven procedure to model occupancy and occupant-related electric load profiles in residential buildings for energy simulation, Energy Build. 202 (2019), https:// doi.org/10.1016/j.enbuild.2019.109342, 109342.
- [46] I.B. Arief-Ang, F.D. Salim, M. Hamilton, DA-HOC: Semi-supervised domain adaptation for room occupancy prediction using CO2 sensor data, in: BuildSys 2017 - Proc. 4th ACM Int. Conf. Syst. Energy-Efficient Built Environ, Association for Computing Machinery, Inc, 2017, https://doi.org/10.1145/ 3137133.3137146.
- [47] T.A. Nguyen, M. Aiello, Energy intelligent buildings based on user activity: a survey, Energy Build. 56 (2013) 244–257, https://doi.org/10.1016/j. enbuild.2012.09.005.
- [48] T. Hong, S.C. Taylor-Lange, S. D'Oca, D. Yan, S.P. Corgnati, Advances in research and applications of energy-related occupant behavior in buildings, Energy Build. 116 (2016) 694–702, https://doi.org/10.1016/j.enbuild.2015.11.052.
- [49] A. Fleury, M. Vacher, N. Noury, SVM-based multimodal classification of activities of daily living in health smart homes: sensors, algorithms, and first experimental results, IEEE Trans. Inf. Technol. Biomed. 14 (2010) 274–283, https://doi.org/ 10.1109/TTB.2009.2037317.
- [50] J. Zou, Q. Zhao, W. Yang, F. Wang, Occupancy detection in the office by analyzing surveillance videos and its application to building energy conservation, Energy Build. 152 (2017) 385–398, https://doi.org/10.1016/j. enbuild.2017.07.064.
- [51] J.W. Dziedzic, Y. Da, V. Novakovic, Indoor occupant behaviour monitoring with the use of a depth registration camera, Build. Environ. 148 (2019) 44–54, https:// doi.org/10.1016/j.buildenv.2018.10.032.
- [52] L. Chen, C.D. Nugent, J. Biswas, J. Hoey, Activity Recognition in Pervasive Intelligent Environments, 2011, https://doi.org/10.2991/978-94-91216-05-3
- [53] M. Aprea, E. Nothen, L. Uberman, Home Automation: proposal for a low-cost and noninvasive approach, in: Congr. Argentino Ciencias La Inform. Y Desarro. Investig. CACIDI 2018, Institute of Electrical and Electronics Engineers Inc., 2018, https://doi.org/10.1109/CACIDI.2018.8584341.
- [54] Z. Wang, T. Hong, M.A. Piette, Data fusion in predicting internal heat gains for office buildings through a deep learning approach, Appl. Energy (2019) 386–398, https://doi.org/10.1016/j.apenergy.2019.02.066.
- [55] P. Viertel, M. Küper, J. Wiegmann, Konig, J. Wiegmann, M. König, Determining room usage through a multi-sensor system and machine learning, in: Proc. Des Work. "Lernen Und Arb. Im Wandel,", 2018.
- [56] S.S. Shetty, D.C. Hoang, M. Gupta, S.K. Panda, Learning desk fan usage preferences for personalised thermal comfort in shared offices using tree-based methods, Build. Environ. 149 (2019) 546–560, https://doi.org/10.1016/j. buildenv.2018.12.040.
- [57] A. Ozawa, Y. Kudoh, Y. Yoshida, A new method for household energy use modeling: a questionnaire-based approach, Energy Build. 162 (2018) 32–41, https://doi.org/10.1016/j.enbuild.2017.12.032.
- [58] S. Mahmoud, A. Lotfi, C. Langensiepen, Behavioural pattern identification and prediction in intelligent environments, Appl. Soft Comput. J. 13 (2013) 1813–1822, https://doi.org/10.1016/j.asoc.2012.12.012.
- [59] S. Wolf, J.K. Møller, M.A. Bitsch, J. Krogstie, H. Madsen, A Markov-Switching model for building occupant activity estimation, Energy Build. 183 (2019) 672–683, https://doi.org/10.1016/j.enbuild.2018.11.041.
- [60] Z. Wang, Y. Ding, An occupant-based energy consumption prediction model for office equipment, Energy Build. 109 (2015) 12–22, https://doi.org/10.1016/j. enbuild.2015.10.002.

Building and Environment 174 (2020) 106768

- [61] D. Yan, C. Wang, X. Feng, B. Dong, Validation and ground truths, in: Explor. Occupant Behav. Build. Methods Challenges, Springer International Publishing, 2017, pp. 239–260, https://doi.org/10.1007/978-3319-61464-9.9.
- [62] A. Ignatov, Real-time human activity recognition from accelerometer data using Convolutional Neural Networks, Appl. Soft Comput. J. 62 (2018) 915–922, https://doi.org/10.1016/j.asoc.2017.09.027.
- [63] A. Elboushaki, R. Hannane, K. Afdel, L. Koutti, MultiD-CNN, A multi-dimensional feature learning approach based on deep convolutional networks for gesture recognition in RGB-D image sequences, Expert Syst. Appl. 139 (2020), https:// doi.org/10.1016/j.eswa.2019.112829, 112829.
- [64] H. Cho, S.M. Yoon, Divide and conquer-based 1D CNN human activity recognition using test data sharpening, Sensors 18 (2018) 1055, https://doi.org/10.3390/ s18041055.
- [65] M. Rawashdeh, M.G.H. Al Zamil, S. Samarah, M.S. Hossain, G. Muhammad, A knowledge-driven approach for activity recognition in smart homes based on activity profiling, Future Generat. Comput. Syst. (2017), https://doi.org/ 10.1016/j.future.2017.10.031.
- [66] A. Das, F.C. Sangogboye, E.S.K. Raun, M.B. Kjærgaard, HeteroSense: an occupancy sensing framework for multi-class classification for activity recognition and trajectory detection, in: Soc. 2019 - Proc. 2019 4th Int. Work. Soc. Sens, Association for Computing Machinery, Inc, 2019, pp. 12–17, https:// doi.org/10.1145/3313294.3313383.
- [67] M.Z. Uddin, M.M. Hassan, A. Alsanad, C. Savaglio, A body sensor data fusion and deep recurrent neural network-based behavior recognition approach for robust healthcare, Inf. Fusion 55 (2020) 105–115, https://doi.org/10.1016/j. inffus.2019.08.004.
- [68] M.Z. Uddin, A wearable sensor-based activity prediction system to facilitate edge computing in smart healthcare system, J. Parallel Distr. Comput. 123 (2019) 46–53, https://doi.org/10.1016/j.jpdc.2018.08.010.
- [69] S. Ahmadi-Karvigh, A. Ghahramani, B. Becerik-Gerber, L. Soibelman, Real-time activity recognition for energy efficiency in buildings, Appl. Energy 211 (2018) 146–160, https://doi.org/10.1016/j.apenergy.2017.11.055.
- [70] Z.S. Abdallah, M.M. Gaber, B. Srinivasan, S. Krishnaswamy, Activity recognition with evolving data streams: a review, ACM Comput. Surv. 51 (2018), https://doi. org/10.1145/3158645.
- [71] J. Collin, O. Mezentsev, G. Lachapelle, Indoor positioning system using accelerometry and high accuracy heading sensors, in: Proc. ION GPS/GNSS 2003 Conf, 2003, pp. 9–12.
- [72] C. Wang, D. Yan, Y. Jiang, A novel approach for building occupancy simulation, Build. Simul. 4 (2011) 149–167, https://doi.org/10.1007/s12273-011-0044-5.
- [73] P. Mohebbi, E. Stroulia, I. Nikolaidis, Indoor localization: a cost-effectiveness vs. Accuracy study, in: Proc. - IEEE Symp. Comput. Med. Syst, Institute of Electrical and Electronics Engineers Inc., 2017, pp. 552–557, https://doi.org/10.1109/ CBMS.2017.126.
- [74] W. Wang, J. Chen, T. Hong, Modeling occupancy distribution in large spaces with multi-feature classification algorithm, Build. Environ. 137 (2018) 108–117, https://doi.org/10.1016/j.buildenv.2018.04.002.
- [75] A. Alhamoud, A.A. Nair, C. Gottron, D. Böhnstedt, R. Steinmetz, Presence detection, identification and tracking in smart homes utilizing bluetooth enabled smartphones, in: Proc. - Conf. Local Comput. Networks, LCN, IEEE Computer Society, 2014, pp. 784–789, https://doi.org/10.1109/LCNW.2014.6927735.
- [76] R. Mautz, Overview of current indoor positioning systems, Geod. Cartogr. 35 (2009) 18–22, https://doi.org/10.3846/1392-1541.2009.35.18-22.
- [77] G. Dedes, G. Dedes, A.G. Dempster, Indoor GPS positioning challenges and opportunities, in: IEEE FALL Veh. Technol. CONF, 2005, pp. 412–415.
- [78] P. Zappi, E. Farella, L. Benini, Enhancing the spatial resolution of presence detection in a pir based wireless surveillance network, in: 2007 IEEE Conf. Adv. Video Signal Based Surveillance, AVSS 2007 Proc., 2007, pp. 295–300, https:// doi.org/10.1109/AVSS.2007.4425326.
- [79] L.M. Candanedo, V. Feldheim, Accurate occupancy detection of an office room from light, temperature, humidity and CO2 measurements using statistical learning models, Energy Build. 112 (2016) 28–39, https://doi.org/10.1016/j. enbuild.2015.11.071.
- [80] Y. Benezeth, H. Laurent, B. Emile, C. Rosenberger, Towards a sensor for detecting human presence and characterizing activity, Energy Build. 43 (2011) 305–314, https://doi.org/10.1016/j.enbuild.2010.09.014.
- [81] A. Bränzel, C. Holz, D. Hoffmann, D. Schmidt, M. Knaust, P. Lühne, R. Meusel, S. Richter, P. Baudisch, GravitySpace: tracking users and their poses in a smart room using a pressure-sensing floor, in: Conf. Hum. Factors Comput. Syst. - Proc, 2013, pp. 725–734, https://doi.org/10.1145/2470654.2470757.
- [82] S. Pan, T. Yu, M. Mirshekari, J. Fagert, A. Bonde, O.J. Mengshoel, H.Y. Noh, P. Zhang, FootprintID: indoor pedestrian identification through ambient structural vibration sensing, in: Proc. ACM Interactive, Mobile, Wearable Ubiquitous Technol vol. 1, 2017, pp. 1–31, https://doi.org/10.1145/3130954.
- [83] X. Tang, M.C. Huang, S. Mandal, An "Internet of Ears" for crowd-aware smart buildings based on sparse sensor networks, in: Proc. IEEE Sensors, Institute of Electrical and Electronics Engineers Inc., 2017, pp. 1–3, https://doi.org/10.1109/ ICSENS.2017.8234263.
- [84] S. Depatla, A. Muralidharan, Y. Mostofi, Occupancy estimation using only WiFi power measurements, IEL J. Sel. Area. Commun. 33 (2015) 1381–1393, https:// doi.org/10.1109/JSAC2015.2430272.
- [85] P. Dickinson, G. Cielniak, O. Szymanezyk, M. Mannion, Indoor positioning of shoppers using a network of Bluetooth Low Energy beacons, in: 2016 Int. Conf. Indoor Position. Indoor Navig. IPIN 2016, Institute of Electrical and Electronics Engineers Inc., 2016, https://doi.org/10.1109/IPIN.2016.7743684.

- [86] G. Jekabsons, V. Kairish, V. Zuravlyov, An analysis of wi-fi based indoor positioning accuracy, Sci. J. Riga Tech. Univ. Comput. Sci. 44 (2011) 131–137.
- [87] R. Jia, M. Jin, H. Zou, Y. Yesilata, L. Xie, C. Spanos, Poster abstract: mapsentinel: Map-aided non-intrusive indoor tracking in sensor-rich environments, in: BuildSys 2015 - Proc. 2nd ACM Int. Conf. Embed. Syst. Energy-Efficient Built, Association for Computing Machinery, Inc, 2015, pp. 109–110, https://doi.org/ 10.1145/2821650.2830302.
- [88] S. Gansemer, S. Hakobyan, S. Püschel, U. Großmann, 3D WLAN indoor positioning in multi-storey buildings, in: Proc. 5th IEEE Int. Work. Intell. Data Acquis. Adv. Comput. Syst. Technol. Appl. IDAACS'2009, 2009, pp. 669–672, https://doi.org/10.1109/IDAACS.2009.5342893.
- [89] R. Mautz, The challenges of indoor environments and specification on some alternative positioning systems, in: Proc. - 6th Work. Positioning, Navig. Commun vol. 2009, WPNC, 2009, pp. 29–36, https://doi.org/10.1109/ WPNC, 2009.4907800.
- [90] X. Tang, S. Mandal, Indoor occupancy awareness and localization using passive electric field sensing, IEEE Trans. Instrum. Meas. 68 (2019) 4535–4549, https:// doi.org/10.1109/TIM.2018.2890319.
- [91] K. Al Nuaimi, H. Kamel, A survey of indoor positioning systems and algorithms, in: 2011 Int. Conf. Innov. Inf. Technol. IIT 2011, 2011, pp. 185–190, https://doi. org/10.1109/INNOVATIONS.2011.5893813.
- [92] C. Liao, Y. Lin, P. Barooah, Agent-based and graphical modelling of building occupancy, J. Build. Perform. Simul. 5 (2012) 5–25, https://doi.org/10.1080/ 19401493.2010.531143.
- [93] J.F. Nicol, Characterising occupant behaviour in buildings: towards a stochastic model of occupant use of windows, lights, blinds, heaters and fans, in: Proc. 7th Int. IBPSA Conf. Rio de. Janeiro, Brazil, 2001, 2001.
- [94] F. Haldi, D. Robinson, A comparison of alternative approaches for the modeling of window opening and closing behaviour, in: Proc. Conf. Air Cond. Low Carbon Cool. Chall, 2008, pp. 27–29.
- [95] S. Pan, Y. Han, S. Wei, Y. Wei, L. Xia, L. Xie, X. Kong, W. Yu, A model based on Gauss Distribution for predicting window behavior in building, Build. Environ. 149 (2019) 210–219, https://doi.org/10.1016/j.buildemv.2018.12.008.
- [96] F. Stazi, F. Naspi, M. D'Orazio, Modelling window status in school classrooms. Results from a case study in Italy, Build. Environ. 111 (2017) 24–32, https://doi. org/10.1016/j.buildenv.2016.10.013.
- [97] S. Pan, Y. Xiong, Y. Han, X. Zhang, L. Xia, S. Wei, J. Wu, M. Han, A study on influential factors of occupant window-opening behavior in an office building in China, Build. Environ. 133 (2018) 41–50, https://doi.org/10.1016/j. buildeny.2018.02.008.
- [98] Z. Shi, H. Qian, X. Zheng, Z. Lv, Y. Li, L. Liu, P.V. Nielsen, Seasonal variation of window opening behaviors in two naturally ventilated hospital wards, Build. Environ. 130 (2018) 85–93, https://doi.org/10.1016/j.buildenv.2017.12.019.
- [99] N. Li, J. Li, R. Fan, H. Jia, Probability of occupant operation of windows during transition seasons in office buildings, Renew. Energy 73 (2015) 84–91, https:// doi.org/10.1016/j.renene.2014.05.065.
- [100] S. Dutton, L. Shao, Window opening behaviour IN a naturally ventilated school, in: Proc. SimBuild, 2010, pp. 260–268.
- [101] S. Shi, B. Zhao, Occupants' interactions with windows in 8 residential apartments in Beijing and Nanjing, China, Build. Simul. 9 (2016) 221–231, https://doi.org/ 10.1007/s12273-015-0266-z.
- [102] F. Stazi, F. Naspi, G. Bernardini, M. D'Orazio, Comparing real and predicted window use in offices. A POE-based assessment, in: Energy Procedia, Elsevier Ltd, 2017, pp. 141–150, https://doi.org/10.1016/j.egypro.2017.09.552.
- [103] Y. Wei, H. Yu, S. Pan, L. Xia, J. Xie, X. Wang, J. Wu, W. Zhang, Q. Li, Comparison of different window behavior modeling approaches during transition season in Beijing, China, Build, Environ. Times 157 (2019) 1–15, https://doi.org/10.1016/ i.buildenv.2019.04.040.
- [104] M. Yao, B. Zhao, Factors affecting occupants' interactions with windows in residential buildings in Beijing, China, in: Proceedia Eng., Elsevier Ltd, 2017, pp. 3428–3434, https://doi.org/10.1016/j.proeng.2017.09.857.
- [105] R. Andersen, V. Fabi, J. Toftum, S.P. Corgnati, B.W. Olesen, Window opening behaviour modelled from measurements in Danish dwellings, Build, Environ. Times 69 (2013) 101–113, https://doi.org/10.1016/j.buildenv.2013.07.005.
- [106] D. Calì, R.K. Andersen, D. Müller, B.W. Olesen, Analysis of occupants' behavior related to the use of windows in German households, Build. Environ. 103 (2016) 54–69, https://doi.org/10.1016/j.buildenv.2016.03.024.
- [107] R.V. Jones, A. Fuertes, E. Gregori, A. Giretti, Stochastic behavioural models of occupants' main bedroom window operation for UK residential buildings, Build. Environ. 118 (2017) 144–158, https://doi.org/10.1016/j.buildenv.2017.03.033.
- [108] F. Naspi, M. Arnesano, L. Zampetti, F. Stazi, G.M. Revel, M. D'Orazio, Experimental study on occupants' interaction with windows and lights in Mediterranean offices during the non-heating season, Build. Environ. 127 (2018) 221–238, https://doi.org/10.1016/j.buildenv.2017.11.009.
- [109] R. Fritsch, A. Kohler, M. Nygård-Ferguson, J.L. Scartezzini, A stochastic model of user behaviour regarding ventilation, Build. Environ. 25 (1990) 173–181, https:// doi.org/10.1016/0360-1323(90)90030-U.
- [110] D. Cali, M.T. Wesseling, D. Müller, WinProGen: a Markov-Chain-based stochastic window status profile generator for the simulation of realistic energy performance in buildings, Build. Environ. 136 (2018) 240–258, https://doi.org/10.1016/j. buildenv.2018.03.048.
- [111] V. Fabi, R.K. Andersen, S. Corgnati, Verification of stochastic behavioural models of occupants' interactions with windows in residential buildings, Build. Environ. 94 (2015) 371–383, https://doi.org/10.1016/j.buildenv.2015.08.016.

Building and Environment 174 (2020) 106768

- [112] F. Haldi, D. Robinson, Interactions with window openings by office occupants, Build. Environ. 44 (2009) 2378–2395, https://doi.org/10.1016/j. buildenv.2009.03.025.
- [113] R. Markovic, E. Grintal, D. Wölki, J. Frisch, C. van Treeck, Window opening model using deep learning methods, Build. Environ. 145 (2018) 319–329, https://doi.org/10.1016/j.buildenv.2018.09.024.
- [114] O.I. Abiodun, A. Jantan, A.E. Omolara, K.V. Dada, N.A.E. Mohamed, H. Arshad, State-of-the-art in artificial neural network applications: a survey, Heliyon 4 (2018), https://doi.org/10.1016/j.heliyon.2018.e00938.
- [115] M. Jamil, S.K. Sharma, R. Singh, Fault detection and classification in electrical power transmission system using artificial neural network, SpringerPlus 4 (2015) 334, https://doi.org/10.1186/s40064-015-1080-x.
- [116] S.A. Kalogirou, M. Bojic, Artificial neural networks for the prediction of the energy consumption of a passive solar building, Energy 25 (2000) 479–491, https://doi.org/10.1016/S0360-5442(99)00086-9.
- [117] V.M. Barthelmes, Y. Heo, V. Fabi, S.P. Corgnati, Exploration of the Bayesian Network framework for modelling window control behaviour, Build. Environ. 126 (2017) 318–330, https://doi.org/10.1016/j.buildenv.2017.10.011.
- [118] S. D'Oca, T. Hong, A data-mining approach to discover patterns of window opening and closing behavior in offices, Build. Environ. 82 (2014) 726–739, https://doi.org/10.1016/j.buildenv.2014.10.021.
- [119] Z. Deme Belañ, F. Naspi, M. Arnesano, A. Reith, G.M. Revel, Investigation on window opening and closing behavior in schools through measurements and surveys: a case study in Budapest, Build, Environ. Times 143 (2018) 523–531, https://doi.org/10.1016/j.buildenv.2018.07.022.
- [120] N. Nord, T. Tereshchenko, L.H. Qvistgaard, I.S. Tryggestad, Influence of occupant behavior and operation on performance of a residential Zero Emission Building in Norway, Energy Build. 159 (2018) 75–88, https://doi.org/10.1016/j. enbuild.2017.10.083.
- [121] F. Naspi, M. Arnesano, F. Stazi, M. D'Orazio, G.M. Revel, Measuring occupants' behaviour for buildings' dynamic cosismulation, J. Sensors. 2018 (2018), https:// doi.org/10.1155/2018/2756542.
- [122] F. Haldi, D. Calì, R.K. Andersen, M. Wesseling, D. Müller, Modelling diversity in building occupant behaviour: a novel statistical approach, J. Build. Perform. Simul. 10 (2017) 527–544, https://doi.org/10.1080/1940/1943.2016.1269245.
- [123] M. Schweiker, A. Wagner, The effect of occupancy on perceived control, neutral temperature, and behavioral patterns, Energy Build. 117 (2016) 246–259, https://doi.org/10.1016/j.enbuild.2015.10.051.
- [124] Y. Rafiq, S. Wei, R. Guest, R. Stone, P. de Wilde, Applying Artificial Neural Networks to Promote Behaviour Change for Saving Residential Energy, 2014, pp. 3–10, https://doi.org/10.5220/0005049600030010.
- [125] S. D'Oca, V. Fabi, S.P. Corgnati, R.K. Andersen, Effect of thermostat and window opening occupant behavior models on energy use in homes, Build. Simul. 7 (2014) 683–694, https://doi.org/10.1007/s12273-014-0191-6.
- [126] G.M. Stavrakakis, P.L. Zervas, H. Sarimveis, N.C. Markatos, Optimization of window-openings design for thermal comfort in naturally ventilated buildings, Appl. Math. Model. 36 (2012) 193–211, https://doi.org/10.1016/j. apm.2011.05.052.
- [127] G.Y. Yun, K. Steemers, Night-time naturally ventilated offices: statistical simulations of window-use patterns from field monitoring, Sol. Energy 84 (2010) 1216–1231, https://doi.org/10.1016/j.solener.2010.03.029.
- [128] G.Y. Yun, P. Tuohy, K. Steemers, Thermal performance of a naturally ventilated building using a combined algorithm of probabilistic occupant behaviour and deterministic heat and mass balance models, Energy Build. 41 (2009) 489–499, https://doi.org/10.1016/j.enbuild.2008.11.013.
- [129] R. Markovic, S. Wolf, J. Čao, E. Spinnräker, D. Wölki, J. Frisch, C. van Treeck, Comparison of different classification algorithms for the detection of user's interaction with windows in office buildings, in: Energy Procedia, Elsevier Ltd, 2017, pp. 337–342.
- [130] G.Y. Yun, K. Steemers, Time-dependent occupant behaviour models of window control in summer, Build. Environ. 43 (2008) 1471–1482, https://doi.org/ 10.1016/j.buildenv.2007.08.001.
- [131] R. Markovic, J. Frisch, C. van Treeck, Learning short-term past as predictor of window opening-related human behavior in commercial buildings, Energy Build, 188 (2019) 1–11, https://doi.org/10.1016/j.enbuild.2018.12.012.
- [132] F. Haldi, D. Robinson, On the behaviour and adaptation of office occupants, Build. Environ. 43 (2008) 2163–2177, https://doi.org/10.1016/j. buildenv.2008.01.003.
- [133] S. Herkel, U. Knapp, J. Pfafferott, Towards a model of user behaviour regarding the manual control of windows in office buildings, Build. Environ. 43 (2008) 588–600, https://doi.org/10.1016/j.buildenv.2006.06.031.
- [134] F. Stazi, F. Naspi, M. D'Orazio, A literature review on driving factors and contextual events influencing occupants' behaviours in buildings, Build. Environ. 118 (2017) 40-66, https://doi.org/10.1016/j.buildenv.2017.03.021.
- [135] V.M. Barthelmes, Y. Heo, R.K. Andersen, V. Fabi, S.P. Corgnati, Towards A comprehensive model of window control behaviour: a survey-based investigation on interdisciplinary drivers in Danish dwellings, in: Build. Simul. Optim, 2018, pp. 649–656.
- [136] Y.S. Lee, A.M. Malkawi, Simulating multiple occupant behaviors in buildings: an agent-based modeling approach, Energy Build. 69 (2014) 407–416, https://doi. org/10.1016/j.enbuild.2013.11.020.
- [137] D.R.G. Hunt, The use of artificial lighting in relation to daylight levels and occupancy, Build. Environ. 14 (1979) 21–33, https://doi.org/10.1016/0360-1323(79)90025-8.

- [138] F. Haldi, D. Robinson, Adaptive actions on shading devices in response to local visual stimuli, J. Build. Perform. Simul. 3 (2010) 135–153, https://doi.org/ 10.1080/1940149003580759.
- [139] C.F. Reinhart, Lightswitch-2002: a model for manual and automated control of electric lighting and blinds, Sol. Energy 77 (2004) 15–28, https://doi.org/ 10.1016/j.solener.2004.04.003.
- [140] R.V. Andersen, J. Toftum, K.K. Andersen, B.W. Olesen, Survey of occupant behaviour and control of indoor environment in Danish dwellings, Energy Build. 41 (2009) 11–16, https://doi.org/10.1016/j.enbuild.2008.07.004.
- [141] A. Mahdavi, A. Mohammadi, E. Kabir, L. Lambeva, Occupants' operation of lighting and shading systems in office buildings, J. Build. Perform. Simul. 1 (2008) 57-65, https://doi.org/10.1080/19401490801906502.
- [142] G.R. Newsham, Manual control of window blinds and electric lighting: implications for comfort and energy consumption, Indoor Environ. 3 (1994) 135–144, https://doi.org/10.1177/142032654900300307.
- [143] S. Yeon, B. Yu, B. Seo, Y. Yoon, K.H. Lee, ANN based automatic slat angle control of Venetian blind for minimized total load in an office building. Sol. Energy (2019) 133–145, https://doi.org/10.1016/j.solener.2019.01.027.
- [144] D. Makkar, P. Syal, Simulation of intelligent room lighting illuminance control, in: 2017 IEEE Int. Conf. Comput. Intell. Comput. Res. ICCIC 2017, Institute of Electrical and Electronics Engineers Inc., 2018, https://doi.org/10.1109/ ICCIC.2017.8524356.
- [145] C. Kurian, R. Aithal, J. Bhat, V. George, Robust control and optimisation of energy consumption in daylight—artificial light integrated schemes, Light. Res. Technol. 40 (2008) 7–24, https://doi.org/10.1177/147715350079511.
- [146] M. Trobec Lah, B. Zupančić, J. Peternelj, A. Krainer, Daylight illuminance control with fuzzy logic, Sol. Energy 80 (2006) 307–321, https://doi.org/10.1016/j. solener.2005.02.002.
- [147] A. Guillemin, N. Morel, Innovative lighting controller integrated in a self-adaptive building control system, Energy Build. 33 (2001) 477–487, https://doi.org/ 10.1016/s0378-7788(00)00100-6.
- [148] Y. Chen, H. Li, X. Chen, Venetian blind control system based on fuzzy neural network for indoor daylighting, in: 2009 Int. Conf. Comput. Electr. Eng. ICCEE, 2009, pp. 269–273, https://doi.org/10.1109/ICCEE.2009.18, 2009.
- [149] Y.F. Chen, H. Li, H. Li, Indoor dynamic lighting control system for a full-scale private office, in: Proc. 2012 24th Chinese Control Decis. Conf. CODC 2012, 2012, pp. 1691–1695, https://doi.org/10.1109/CCDC.2012.6244270.
- [150] Z. Cheng, Q. Zhao, F. Wang, Y. Jiang, L. Xia, J. Ding, Satisfaction based Qlearning for integrated lighting and blind control, Energy Build. 127 (2016) 43–55, https://doi.org/10.1016/j.enbuild.2016.05.067.
- [151] D. Daum, N. Morel, Assessing the total energy impact of manual and optimized blind control in combination with different lighting schedules in a building simulation environment, J. Build. Perform. Simul. 3 (2010) 1–16, https://doi.org/ 10.1080/19401490903362810.
- [152] N.K. Kandasamy, G. Karunagaran, C. Spanos, K.J. Tseng, B.H. Soong, Smart lighting system using ANN-IMC for personalized lighting control and daylight harvesting, Build. Environ. 139 (2018) 170–180, https://doi.org/10.1016/j. buildenv.2018.05.005.
- [153] X. Ding, J. Yu, Y. Si, Office light control moving toward automation and humanization: a literature review, Intell. Build. Int. (2018), https://doi.org/ 10.1080/17508975.2018.1555087.
- [154] V. Singhvi, A. Krause, C. Guestrin, J.H. Garrett, H. Scott Matthews, Intelligent light control using sensor networks, in: SenSys 2005 - Proc. 3rd Int. Conf. Embed. Networked Sens. Syst, Association for Computing Machinery, 2005, pp. 218–229, https://doi.org/10.1145/1098918.1098942.
- [155] I. Hidayat, Faridah, S. Sesotya Utami, Activity based smart lighting control for energy efficient building by neural network model, in: E3S Web Conf. vol 43, 2018, 0107, https://doi.org/10.1051/e3sconf/20184301017.
- [156] A.H. Khalili, C. Wu, H. Aghajan, Hierarchical preference learning for light control from user feedback, in: 2010 IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit. - Work. CVPRW 2010, 2010, pp. 56–62, https://doi.org/10.1109/ CVPRW.2010.5543265.
- [157] C.A. Hernandez S, R. Romero, D. Giral, Optimization of the use of residential lighting with neural network, in: 2010 Int. Conf. Comput. Intell. Softw. Eng. CiSE 2010, 2010, https://doi.org/10.1109/CISE.2010.5677018.
- [158] S.A.M. Offermans, van H.A. Essen, J.H. Eggen, Exploring a hybrid control approach for enhanced user experience of interactive lighting, in: Proc. 27th Int. BCS Hum. Comput. Interact. Conf. (HCI 2013), British Computer Society (BCS), 2013, pp. 1–9.
- [159] J.Y. Park, T. Dougherty, H. Fritz, Z. Nagy, LightLearn: an adaptive and occupant centered controller for lighting based on reinforcement learning, Build. Environ. 147 (2019) 397–414, https://doi.org/10.1016/j.buildenv.2018.10.028.
- [160] D.A. Giral, R.R. Romero, C. Hernández, Analysis and review of the contribution of neural networks to saving electricity in residential lighting by a design in MATLAB, in: Proc. - 2012 9th Electron. Robot. Automot. Mech. Conf. CERMA, 2012, pp. 182–190, https://doi.org/10.1109/CERMA.2012.37, 2012.
- [161] V. Lobato-Rios, V. Del Rocio Hernandez-Castanon, J.A. Carrasco-Ochoa, J. F. Martinez-Trinidad, Linear model optimizer vs Neural Networks: a comparison for improving the quality and saving of LED-Lighting control systems, in: Proc.-Int. Conf. Pattern Recognit, Institute of Electrical and Electronics Engineers Inc., 2016, pp. 2664–2669, https://doi.org/10.1109/ICPR.2016.7900037.
- [162] S. Atis, N. Ekren, Development of an outdoor lighting control system using expert system, Energy Build. 130 (2016) 773–786, https://doi.org/10.1016/j. enbuild.2016.08.066.
- [163] A. Mohagheghi, M. Moallem, A. Khayatian, Neural network-based LED lighting control with modeling uncertainty and daylight disturbance, in: Proc. IECON

2017 - 43rd Annu. Conf. IEEE Ind. Electron. Soc, Institute of Electrical and Electronics Engineers Inc., 2017, pp. 3627–3632, https://doi.org/10.1109/ IECON.2017.8216615.

- [164] W. Si, X. Pan, H. Ogai, K. Hirai, A novel lambertian-RBFNN for office light modeling, in: IEICE Trans. Inf. Syst. E99D, 2016, pp. 1742–1752, https://doi.org/ 10.1587/transinf.2015EDP7411.
- [165] J. Hu, E. Shen, Y. Gu, Evaluation of lighting performance risk using surrogate model and EnergyPlus, in: Proceedia Eng, Elsevier Ltd, 2015, pp. 522–529, https:// doi.org/10.1016/j.proeng.2015.08.471.
- [166] W. Si, X. Pan, H. Ogai, K. Hirai, N. Yamauchi, T. Li, Illumination modeling method for office lighting control by using RBFNN, IEICE Trans. Info Syst. E97.D (2014) 3192–3200, https://doi.org/10.1587/transinf.2013EDP7384.
- [167] Y. Gao, Y. Lin, Y. Sun, A wireless sensor network based on the novel concept of an I-matrix to achieve high-precision lighting control, Build. Environ. 70 (2013) 223–231, https://doi.org/10.1016/j.buildenv.2013.08.011.
- [168] W. Si, H. Ogai, T. Li, K. Hirai, A novel energy saving system for office lighting control by using RBFNN and PSO, in: IEEE 2013 Tencon - Spring, TENCONSpring 2013 - Conf. Proc., 2013, pp. 347–351, https://doi.org/10.1109/ TENCONSpring.2013.6584469.
- [169] Z. Wang, Y.K. Tan, Illumination control of LED systems based on neural network model and energy optimization algorithm, Energy Build. 62 (2013) 514–521, https://doi.org/10.1016/j.embuild.2013.03.029.
- [170] R.R. Cardoso, A.F. Brandao, Artificial neural network application to a building hybrid illumination system, in: IEEE Int. Symp. Ind. Electron, 2008, pp. 866–871, https://doi.org/10.1109/ISIE.2008.4677011.
- [171] C. Kurian, S. Kuriachan, J. Bhat, R. Aithal, An adaptive neuro-fuzzy model for the prediction and control of light in integrated lighting schemes, Light. Res. Technol. 37 (2005) 343–351, https://doi.org/10.1191/13657828051i150oa.
- [172] M.M. Ouf, H.B. Gunay, W. O'Brien, A method to generate design-sensitive occupant-related schedules for building performance simulations, Sci. Technol. Built Environ. 25 (2019) 221–232, https://doi.org/10.1080/ 23744731.2018.1514855.
- [173] F.H. Sakaci, E. Cetiner, H. Chaouch, S.C. Yener, Smart office for managing energy of lighting control system, in: Proc. - 2018 6th Int. Istanbul Smart Grids Cities Congr. Fair, ICSG 2018, Institute of Electrical and Electronics Engineers Inc., 2018, pp. 101–104, https://doi.org/10.1109/SGCF.2018.8408951.
- [174] K. Rijal Wagiman, M. Noor Abdullah, Intelligent lighting control system for energy savings in office building, Indones. J. Electr. Eng. Comput. Sci. 11 (2018) 195–202, https://doi.org/10.11591/ijeccs.v11.i1.pp195-202.
- [175] S.A. Razlei, H. Mohscnian-Had, Optimal demand response capacity of automatic lighting control, in: 2013 IEEE PES Innov. Smart Grid Technol. Conf. ISGT 2013, 2013, https://doi.org/10.1109/ISGT.2013.6497854.
- [176] D. Lindelöf, N. Morel, Bayesian estimation of visual discomfort, Build. Res. Inf. 36 (2008) 83–96, https://doi.org/10.1080/09613210701544061.
- [177] D. Lindelöf, N. Morel, A field investigation of the intermediate light switching by users, Energy Build. 38 (2006) 790–801, https://doi.org/10.1016/j. enbuild.2006.03.003.
- [178] Z.N. Zhen, Q.S. Jia, C. Song, X. Guan, An indoor localization algorithm for lighting control using RFID, in: 2008 IEEE Energy 2030 Conf. ENERGY 2008, 2008, https://doi.org/10.1109/ENERGY.2008.4781041.
- [179] S.N. Patel, T. Robertson, J.A. Kientz, M.S. Reynolds, G.D. Abowd, At the flick of a switch: detecting and classifying unique electrical events on the residential power line, in: UbiComp 2007 Ubiquitous Comput., Springer Berlin Heidelberg, 2007, pp. 271–288, https://doi.org/10.1007/978-3-540-74853-3_16.
- [180] P. Kar, A. Shareef, A. Kumar, K.T. Harn, B. Kalluri, S.K. Panda, ReViCEE: a recommendation based approach for personalized control, visual comfort & energy efficiency in buildings, Build. Environ. 152 (2019) 135–144, https://doi. org/10.1016/j.buildenv.2019.01.035.
- [181] L. Bandić, J. Kevrić, Near zero-energy home prediction of appliances energy consumption using the reduced set of features and random decision tree algorithms, in: Lect. Notes Networks Syst, Springer, 2019, pp. 164–171, https:// doi.org/10.1007/978-3-030-02574-8_13.
- [182] A.P. Chandrasiri, D. Geekiyanage, Real-time object detection system for building energy conservation: an IP camera based system, in: 34th Annu. Assoc. Res. Constr. Manag. 2018.
- [183] S. Borile, A. Pandharipande, D. Caicedo, L. Schenato, A. Cenedese, A data-driven daylight estimation approach to lighting control, IEEE Access 5 (2017) 21461–21471, https://doi.org/10.1109/ACCESS.2017.2679807.
- [184] R. Paulson, R. Paulson, R. Basu, A.M. Agogino, S. Poll, Inverse Modeling Using a Wireless Sensor Network (WSN) for Personalized Daylight Harvesting, SENSORNETS. (n.d.) 213–221.
- [185] J.F. Nicol, M.H.-A. Transactions, U. 2004, A Stochastic Approach to Thermal Comfort-Occupant Behavior and Energy Use in Buildings/DISCUSSION, Build. Eng. vol. 110 (2004) 554.
- [186] C. Wang, D. Yan, X. Ren, Modeling individual's light switching behavior to understand lighting energy use of office building, Energy Procedia 88 (2016) 781-787, https://doi.org/10.1016/j.egypro.2016.06.128.
- [187] A. Galatioto, M. Beccali, Assessment of the entropy of spatial and time distributions of rooms daylighting: a possible tool for a sustainable design, J. Sustain. Dev. Energy, Water Environ. Syst. 3 (2015) 425–435, https://doi.org/ 10.13044/j.sdwwes.2015.03.0032.
- [188] H.-O. Kim, J. Cha, A Study on LED Lights Dimming Control for BEMS Based on Detection People Staying, 2013.
- [189] M. Fischer, K. Wu, P. Agathoklis, Intelligent illumination model-based lighting control, in: Proc. - 32nd IEEE Int. Conf. Distrib. Comput. Syst. Work. ICDCSW, 2012, pp. 245–249, https://doi.org/10.1109/ICDCSW.2012.75, 2012.

- [190] K. Onobayashi, M. Miki, Y. Sakakibara, H. Ikegami, H. Aida, Comparison of Power Consumption Reduce Effect of Intelligent Lighting System and Lighting Control System by Using Motion Sensor, 2014.
- [191] P. Lee, P.T.I. Lam, W.L. Lee, Performance risks of lighting retrofit in Energy Performance Contracting projects, Energy Sustain. Dev. 45 (2018) 219–229, https://doi.org/10.1016/j.esd.2018.07.004.
- [192] J.R. Villar, E. De La Cal, J. Sedano, M. García-Tamargo, Analysing the low quality of the data in lighting control systems, in: Int. Conf. Hybrid Artif. Intell. Syst, 2010, pp. 421–428, https://doi.org/10.1007/978-3-642-13769-3_51.
- [193] Y. Chen, Q. Sun, Artificial intelligent control for indoor lighting basing on person number in classroom, in: 2013 9th Asian Control Conf. ASCC 2013, 2013, https:// doi.org/10.1109/ASCC.2013.6606030.
- [194] J. Feng, Y. Yang, Design and implementation of lighting control system for smart rooms, in: 2017 2nd IEEE Int. Conf. Comput. Intell. Appl. ICCIA 2017, Institute of Electrical and Electronics Engineers Inc., 2017, pp. 476–481, https://doi.org/ 10.1109/CIAPP.2017.8167263.
- [195] A. Ahuja, K. Mosalam, T.I. Zohdi, K.M. Mosalam, T.I. Zohdi, An illumination model for translucent concrete using radiance, in: Proc., 14th Conf., Int. Build. Perform. Simul. Assoc, 2015.
- [196] J. Widén, E. Wäckelgård, A high-resolution stochastic model of domestic activity patterns and electricity demand, Appl. Energy 87 (2010) 1880–1892, https://doi. org/10.1016/j.jenergy.2009.11.006.
- [197] P. Faria, Å. Pinto, Z. Vale, M. Khorram, F.B. De Lima Neto, T. Pinto, Lighting consumption optimization using fish school search algorithm, in: 2017 IEEE Symp. Ser. Comput. Intell. SSCI 2017 - Proc, Institute of Electrical and Electronics Engineers Inc., 2018, pp. 1–5, https://doi.org/10.1109/SSCI.2017.8285441.
- [198] C. De Bakker, T. Van de Voort, A. Rosemann, The energy saving potential of occupancy-based lighting control strategies in open-plan offices: the influence of occupancy patterns, Energies 11 (2017) 2, https://doi.org/10.3390/en11010002.
- [199] C. De Bakker, T. Van De Voort, J. Van Duijhoven, A. Rosemann, Assessing the energy use of occupancy-based lighting control strategies in open-plan offices, in: Proc. 2017 IEEE 14th Int. Conf. Networking, Sens. Control. ICNSC 2017, Institute of Electrical and Electronics Engineers Inc., 2017, pp. 476–481, https://doi.org/ 10.1109/ICNSC.2017.8000139.
- [200] M.H.T. Imam, S. Afshari, S. Mishra, An experimental survey of feedback control methodologies for advanced lighting systems, Energy Build. 130 (2016) 600–612, https://doi.org/10.1016/j.embuild.2016.08.088.
- [201] M. Stokes, M. Rylatt, K. Lomas, A simple model of domestic lighting demand, Energy Build. 36 (2004) 103–116, https://doi.org/10.1016/j. enbuild.2003.10.007.
- [202] I. Richardson, M. Thomson, D. Infield, A high-resolution domestic building occupancy model for energy demand simulations, Energy Build. 40 (2008) 1560–1566. https://doi.org/10.1016/j.enbuild.2008.02.006.
- 1560–1566, https://doi.org/10.1016/j.enbuild.2008.02.006.
 [203] J. Widén, A.M. Nilsson, E. Wäckelgård, A combined Markov-chain and bottom-up approach to modelling of domestic lighting demand, Energy Build. 41 (2009) 1001–1012, https://doi.org/10.1016/j.enbuild.2009.05.002.
- [204] X. Zhou, D. Yan, T. Hong, X. Ren, Data analysis and stochastic modeling of lighting energy use in large office buildings in China, Energy Build. 86 (2015) 275–287, https://doi.org/10.1016/j.enbuild.2014.09.071.
- [205] I. Richardson, M. Thomson, D. Infield, A. Delahunty, Domestic lighting: a highresolution energy demand model, Energy Build. 41 (2009) 781–789, https://doi. org/10.1016/j.enbuild.2009.02.010.
- [206] O.K. Larsen, R.L. Jensen, T. Antonsen, I. Strømberg, Estimation methodology for the electricity consumption with daylight- and occupancy-controlled artificial lighting, in: Energy Procedia, Elsevier Ltd, 2017, pp. 733–738, https://doi.org/ 10.1016/j.egypro.2017.07.388.
- [207] D. Caicedo, A. Pandharipande, Energy performance prediction of lighting systems, in: IEEE Int. Work. Mach. Learn. Signal Process. MLSP, IEEE Computer Society, 2016, https://doi.org/10.1109/MLSP.2016.7738858.
- [208] D. Catcedo, A. Pandharipande, Sensor data-driven lighting energy performance prediction, IEEE Sensor. J. 16 (2016) 6397–6405, https://doi.org/10.1109/ JSEN.2016.2579663.
- [209] T.M. Sanjeev Kumar, C.P. Kurian, K. Shreeya, A. Amulya, Data analytic models for lighting energy sensitivity analysis of building, in: 2018 Int. Conf. Control. Power, Commun. Comput. Technol. ICCPCCT 2018, Institute of Electrical and Electronics Engineers Inc., 2018, pp. 118–122, https://doi.org/10.1109/ ICCPCCT.2018.8574270.
- [210] L.M. Candanedo, V. Feldheim, D. Deramaix, Data driven prediction models of energy use of appliances in a low-energy house, Energy Build. 140 (2017) 81–97, https://doi.org/10.1016/j.enbuild.2017.01.083.
- [211] S. Gilani, W. O'Brien, A preliminary study of occupants' use of manual lighting controls in private offices: a case study, Energy Build. 159 (2018) 572-586, https://doi.org/10.1016/j.enbuild.2017.11.055.
- [212] V.R.M. Lo Verso, A. Pellegrino, F. Pellerey, A multivariate non-linear regression model to predict the energy demand for lighting in rooms with different architectural features and lighting control systems, Energy Build. 76 (2014) 151–163, https://doi.org/10.1016/j.enbuild.2014.02.063.
- [213] Yong-Jing Wu, Jie Yang, Pan Xiao-hui, Yu Chang-qing, Design and simulation of the auto-control system of classroom lights, in: Int. Conf. Autom. Control Artif. Intell. (ACAI 2012), Institution of Engineering and Technology, 2012, pp. 794–798, https://doi.org/10.1049/cp.2012.1097.
- [214] P. Rajaram, R. Rawal, Potential savings in lighting energy due to advancement in Indian standard time: an enquiry in context of commercial office spaces in India, in: Proc. Build. Simul. 2011 12th Conf. Int. Build. Perform. Simul. Assoc, 2011, pp. 14–16.

Building and Environment 174 (2020) 106768

- [215] L. Brackney, R. Shoureshi, Fuzzy-based self-organizing control for building systems, in: Proc. Am. Control Conf, American Automatic Control Council, 1994, pp. 3060–3064, https://doi.org/10.1109/acc.1994.735133.
 [216] N. Khera, A. Khan, P. Biswal, C. Likhith, Development of an intelligent light
- [216] N. Khera, A. Khan, P. Biswal, C. Likhith, Development of an intelligent light intensity control system for LED lighting, in: 2018 Int. Conf. Power Energy, Environ. Intell. Control. PEEIC 2018, Institute of Electrical and Electronics Engineers Inc., 2019, pp. 141–144, https://doi.org/10.1109/ PEEIC.2018.8665595.
- [217] L. Gomes, C. Ramos, A. Jozi, B. Serra, L. Paiva, Z. Vale, IoH: a platform for the intelligence of home with a context awareness and ambient intelligence approach, Future Internet 11 (2019) 58, https://doi.org/10.3309/fil1030058.
- [218] S. Gilani, W. O'Brien, H.B. Gunay, Simulating occupants' impact on building energy performance at different spatial scales, Build. Environ. 132 (2018) 327–337, https://doi.org/10.1016/j.buildenv.2018.01.040.
- [219] J. Coutaz, A. Carlesso, N. Bonnefond, R. Balzarini, Y. Laurillau, N. Mandran, J. L. Crowley, "Will the Last one out, please turn off the lights": promoting energy awareness in public areas of office buildings, in: Lect. Notes Comput. Sci. (Including Subser. Lect. Notes Artif. Intell. Lect. Notes Bioinformatics), Springer Verlag, 2018, pp. 20–36, https://doi.org/10.1007/978-3-030-03062-9_2.
- [220] S. Niu, W. Pan, Y. Zhao, A virtual reality supported approach to occupancy engagement in building energy design for closing the energy performance gap, in: Procedia Eng, Elsevier Ltd, 2015, pp. 573–580, https://doi.org/10.1016/j. proeng.2015.08.487.
- [221] A.A. Kim, S. Wang, L.J. McCunn, Building value proposition for interactive lighting systems in the workplace: combining energy and occupant perspectives, J. Build. Eng. 24 (2019), https://doi.org/10.1016/j.jobe.2019.100752.
- [222] Y. Huo, H. Wang, X. Zuo, Z. Cui, A college teaching building lighting control system based on power line carrier, in: Proc. - 2013 Int. Conf. Inf. Sci. Cloud Comput. Companion, ISCC-C 2013, Institute of Electrical and Electronics Engineers Inc., 2014, pp. 441–445, https://doi.org/10.1109/ISCC-C.2013.11.
- [223] N. Verma, A. Jain, Optimized automatic lighting control in a hotel building for energy efficiency, in: 2018 Int. Conf. Power Energy, Environ. Intell. Control. PEEIC 2018, Institute of Electrical and Electronics Engineers Inc., 2019, pp. 168–172, https://doi.org/10.1109/PEEIC.2018.8665417.
- [224] X. Pan, B. Lee, An approach of reinforcement learning based lighting control for demand response, in: PCIM Eur. 2016; Int. Exhib. Conf. Power Electron. Intell. Motion, Renew. Energy Energy Manag, VDE, 2016, pp. 1–8.
- [225] D. Finney, Probit Analysis, Cambridge University Press, Cambridge Univ. Press, 1971.
- [226] P. de Wilde, The gap between predicted and measured energy performance of buildings: a framework for investigation, Autom. ConStruct. 41 (2014) 40–49.
- [227] X. Ren, D. Yan, C. Wang, Air-conditioning usage conditional probability model for residential buildings, Build, Environ. Times 81 (2014) 172-182, https://doi.org/ 10.1016/j.buildenv.2014.06.022.
- [228] C. Fan, F. Xiao, C. Yan, C. Liu, Z. Li, J. Wang, A novel methodology to explain and evaluate data-driven building energy performance models based on interpretable machine learning, Appl. Energy (2019) 1551–1560, https://doi.org/10.1016/j. apenergy.2018.11.081.
- [229] R. Baeens, D. Saelens, Modelling uncertainty in district energy simulations by stochastic residential occupant behaviour, J. Build. Perform. Simul. 9 (2016) 431–447, https://doi.org/10.1080/19401493.2015.1070203.
- [230] D. Urieli, P. Stone, A learning agent for heat-pump thermostat control, in: Proc. 12th Int'l Conf Auton. Agents Multiagent Syst. (AAMAS 2013), 2013, pp. 1093–1100.
- [231] M.M. Gouda, S. Danaher, C.P. Underwood, Application of an artificial neural network for modelling the thermal dynamics of a Building?s space and its heating system, Math. Comput. Model. Dyn. Syst. 8 (2003) 333–344, https://doi.org/ 10.1076/mcmd.8.3.333.14097.
- [232] G.S. Virk, D. Azzi, A.E. Gegov, B.P. Haynes, K.I. Alkadhimi, Intelligent softcomputing based modelling of naturally ventilated buildings, Int. J. Sol. Energy 22 (2002) 131–140, https://doi.org/10.1080/0142591031000091112.
- [233] V. Fabi, R.V. Andersen, S.P. Corgnati, Influence of occupant's heating set-point preferences on indoor environmental quality and heating demand in residential buildings, in: HVAC R Res., 2013, pp. 635–645, https://doi.org/10.1080/ 10789669.2013.789372.
- [234] S. Chen, Y. Zhuang, J. Zhang, Y. Fu, H. Zhang, Statistical characteristics of usage behavior of air conditioners in the university students' dormitories, in: Procedia Eng, Elsevier Ltd, 2017, pp. 3593–3598, https://doi.org/10.1016/j. proceng.2017.10.209.
- [235] M. Schweiker, M. Shukuya, Comparison of theoretical and statistical models of air-conditioning-unit usage behaviour in a residential setting under Japanese climatic conditions, Build. Environ. 44 (2009) 2137–2149, https://doi.org/ 10.1016/j.buildenv.2009.03.004.
- [236] J. Langevin, J. Wen, P.L. Gurian, Simulating the human-building interaction: development and validation of an agent-based model of office occupant behaviors, Build. Environ. 88 (2015) 27–45, https://doi.org/10.1016/j. buildenv.2014.11.037.
- [237] Y. Peng, Z. Nagy, A. Schlüter, Temperature-preference learning with neural networks for occupant-centric building indoor climate controls, Build. Environ. 154 (2019) 296–308, https://doi.org/10.1016/j.buildenv.2019.01.036.
- [238] International Energy Agency, World Energy Outlook 2016, 2016, https://doi.org/ 10.1787/weo-2016-en.
- [239] N. Pathak, M.A.A.H. Khan, N. Roy, Acoustic based appliance state identifications for fine-grained energy analytics, in: 2015 IEEE Int. Conf. Pervasive Comput. Commun. PerCom 2015, Institute of Electrical and Electronics Engineers Inc., 2015, pp. 63–70, https://doi.org/10.1109/PERCOM.2015.7146510.

- [240] Z. (Jerry) Yu, B. Hu, Y. Sun, A. Li, J. Li, G. Zhang, Standby energy use and saving potentials associated with occupant behavior of Chinese rural homes, Energy Build. 154 (2017) 295-304, https://doi.org/10.1016/j.embuild.2017.08.070.
- [241] R. McCann, A.T. Le, D. Traore, Stochastic sliding mode arbitration for energy management in smart building systems, in: Conf. Rec. - IAS Annu. Meet, IEEE Ind. Appl. Soc., 2008, https://doi.org/10.1109/08IAS.2008.8.
- [242] C. Ganesh Kumar, S. Januja Josephine, P.V. Chandramani, Mining-based device control for home automation, in: Adv. Intell. Syst. Comput, Springer Verlag, 2016, pp. 1049–1058, https://doi.org/10.1007/978-81-322-2656-7_97.
- [243] S. Chen, F. Gao, X. Guan, T. Liu, Y. Che, Y. Liu, A residential load scheduling approach based on load behavior analysis, in: IEEE Int. Conf. Autom. Sci. Eng., IEEE Computer Society, 2014, pp. 954–959, https://doi.org/10.1109/ CoASE.2014.6899441.
- [244] S. Spiegel, Optimization of in-house energy demand, Adv. Comput. Vis. Pattern Recognit. 66 (2015) 217–289, https://doi.org/10.1007/978-3-319-14178-7_10.
- [245] G.Y. Lin, S.C. Lee, J.Y.J. Hsu, W.R. Jih, Applying power meters for appliance recognition on the electric panel, in: Proc. 2010 5th IEEE Conf. Ind. Electron. Appl. ICIEA 2010, 2010, pp. 2254–2259, https://doi.org/10.1109/ ICIEA.2010.5515385.
- [246] F. Jazizadeh, Building energy monitoring realization: context-aware event detection algorithms for non-intrusive electricity disaggregation, in: Constr. Res. Congr. 2016 Old New Constr. Technol. Converg. Hist. San Juan - Proc. 2016 Constr. Res. Congr, CRC, 2016, https://doi.org/10.1061/9780784479827.085, 2016.
- [247] O. Van Cutsem, G. Lilis, M. Kayal, Automatic multi-state load profile identification with application to energy disaggregation, in: IEEE Int. Conf. Emerg. Technol. Fact. Autom. ETFA, Institute of Electrical and Electronics Engineers Inc., 2018, pp. 1–8, https://doi.org/10.1109/ETFA.2017.8247684.
- [248] D. Egarter, V.P. Bhuvana, W. Elmenreich, PALDi: online load disaggregation via particle filtering, IEEE Trans. Instrum. Meas. 64 (2015) 467–477, https://doi.org/ 10.1109/TIM.2014.2344373.
- [249] A. Ridi, C. Gisler, J. Hennebert, Appliance and state recognition using hidden Markov models, in: DSAA 2014 - Proc. 2014 IEEE Int. Conf. Data Sci. Adv. Anal, Institute of Electrical and Electronics Engineers Inc., 2014, pp. 270–276, https:// doi.org/10.1109/DSAA.2014.7058084.
- [250] X. Zhang, Y. Wang, L. Chao, C. Li, L. Wu, X. Peng, Z. Xu, IEHouse: a non-intrusive household appliance state recognition system, in: 2017 IEEE SmartWorld Ubiquitous Intell. Comput. Adv. Trust. Comput. Scalable Comput. Commun. Cloud Big Data Comput. Internet People Smart City Innov. SmartWorld/ SCALCOM/UIC/ATC/CBDCom/IOP/SCI 2017, 2018, https://doi.org/10.1109/ UIC-ATC.2017.8397510.
- [251] A.R. Rababaah, E. Tebekaemi, Electric load monitoring of residential buildings using goodness of fit and multi-layer perceptron neural networks, in: CSAE 2012 -Proceedings, 2012 IEEE Int. Conf. Comput. Sci. Autom. Eng, 2012, pp. 733–737, https://doi.org/10.1109/CSAE.2012.6272871.
- [252] A.F. Ebrahim, O.A. Mohammed, Energy disaggregation based deep learning techniques: a pre-processing stage to enhance the household load forecasting, in: 2018 IEEE Ind. Appl. Soc. Annu. Meet. IAS 2018, Institute of Electrical and Electronics Engineers Inc., 2018, https://doi.org/10.1109/IAS.2018.8544664.
- [253] Q. Liu, X. Yang, L. Deng, An IBeacon-based location system for smart home control, Sensors 18 (2018), https://doi.org/10.3390/s18061897.
- [254] D. Frazzetto, B. Neupane, T.B. Pedersen, T.D. Nielsen, Adaptive user-Oriented direct load-Control of residential flexible devices, in: E-energy 2018 - Proc. 9th ACM Int. Conf. Futur. Energy Syst, Association for Computing Machinery, Inc, 2018, pp. 1–11, https://doi.org/10.1145/3208903.3208924.
- [255] Y. Kashimoto, K. Hata, H. Suwa, M. Fujimoto, Y. Arakawa, T. Shigezumi, K. Komiya, K. Konishi, K. Yasumoto, Low-cost and device-free activity recognition system with energy harvesting PIR and door sensors, in: ACM Int. Conf. Proceeding Ser, Association for Computing Machinery, 2016, pp. 6–11, https:// doi.org/10.1145/3004010.3006378.
- [256] K. Ueda, M. Tamai, K. Yasumoto, A method for recognizing living activities in homes using positioning sensor and power meters, in: 2015 IEEE Int. Conf. Pervasive Comput. Commun. Work. PerCom Work. 2015, Institute of Electrical and Electronics Engineers Inc., 2015, pp. 354–359, https://doi.org/10.1109/ PERCOMW.2015.7134062.
- [257] H.A. Cao, T.K. Wijaya, K. Aberer, N. Nunes, Temporal association rules for electrical activity detection in residential homes, in: Proc. - 2016 IEEE Int. Conf. Big Data, Big Data 2016, Institute of Electrical and Electronics Engineers Inc., 2016, pp. 3097–3106, https://doi.org/10.1109/BigData.2016.7840964.

Building and Environment 174 (2020) 106768

- [258] A. Albert, R. Rajagopal, Cost-of-service segmentation of energy consumers, in: IEEE Trans. Power Syst, 2014, https://doi.org/10.1109/TPWRS.2014.2312721.
- [259] S. D'Oca, T. Hong, Occupancy schedules learning process through a data mining framework, Energy Build. 88 (2015) 395-408, https://doi.org/10.1016/j. enbuild.2014.11.065.
- [260] M. Wang, X. Wang, G. Zhang, C. Li, Occupancy detection based on Spiking Neural Networks for green building automation systems, in: Proc. World Congr. Intell. Control Autom, Institute of Electrical and Electronics Engineers Inc., 2015, pp. 2681–2686, https://doi.org/10.1109/WCICA.2014.7053149.
- [261] R.M. Tetlow, C. van Dronkelaar, C.P. Beaman, A.A. Elmualim, K. Couling, Identifying behavioural predictors of small power electricity consumption in office buildings, Build. Environ. 92 (2015) 75–85, https://doi.org/10.1016/j. buildenv.2015.04.009.
- [262] H.B. Gunay, W. O'Brien, I. Beausoleil-Morrison, S. Gilani, Modeling plug-in equipment load patterns in private office spaces, Energy Build 121 (2016) 234–249, https://doi.org/10.1016/j.enbuild.2016.03.001.
- [263] Y.L. Hsu, P.H. Chou, H.C. Chang, S.L. Lin, S.C. Yang, H.Y. Su, C.C. Chang, Y. S. Cheng, Y.C. Kuo, Design and Implementation of a Smart Home System Using Multisensor Data Fusion Technology, 2017, https://doi.org/10.3390/s17071631. Sensors (Switzerland).
- [264] Z. Deng, Q. Chen, Artificial neural network models using thermal sensations and occupants' behavior for predicting thermal comfort, Energy Build. 174 (2018) 587–602, https://doi.org/10.1016/j.enbuild.2018.06.060.
 [265] ISO, ISO - ISO 7730:1994, Moderate Thermal Environments — Determination of
- [265] ISO, ISO ISO 7730:1994, Moderate Thermal Environments Determination of the PMV and PPD Indices and Specification of the Conditions for Thermal Comfort. 1994.
- [266] T. Chaudhuri, D. Zhai, Y.C. Soh, H. Li, L. Xie, Thermal comfort prediction using normalized skin temperature in a uniform built environment, Energy Build. 159 (2018) 426–440, https://doi.org/10.1016/j.enbuild.2017.10.098.
- [267] S. Lu, E. Cochran Hameen, INTEGRATED IR VISION SENSOR FOR ONLINE CLOTHING INSULATION MEASUREMENT, 2018.
- [268] S. Schiavon, K.H. Lee, Dynamic predictive clothing insulation models based on outdoor air and indoor operative temperatures, Build. Environ. 59 (2013) 250–260.
- [269] R. Rana, B. Kusy, R. Jurdak, J. Wall, W. Hu, Feasibility analysis of using humidex as an indoor thermal comfort predictor, Energy Build. 64 (2013) 17–25, https:// doi.org/10.1016/j.enbuild.2013.04.019.
- [270] K. Ahmed, J. Kurnitski, B. Olesen, Data for occupancy internal heat gain calculation in main building categories, Data Br 15 (2017) 1030–1034, https:// doi.org/10.1016/j.db.2017.10.036.
- [271] S. Atthajariyakul, T. Leephakpreeda, Neural computing thermal comfort index for HVAC systems, Energy Convers, Manag. 46 (2005) 2553–2565, https://doi.org/ 10.1016/j.enconman.2004.12.007.
- [272] F. Gugliermetti, G. Passerini, F. Bisegna, Climate models for the assessment of office buildings energy performance, Build. Environ. 39 (2004) 39–50, https:// doi.org/10.1016/S0360-1323(03)00138-0.
- [273] J. Redmon, A. Farhadi, YOLO9000: better, faster, stronger, in: Proc. 30th IEEE Conf. Comput. Vis. Pattern Recognition, CVPR 2017, 2017.
- [274] H.J. Jeong, K.S. Park, Y.G. Ha, Image preprocessing for efficient training of YOLO deep learning networks, in: Proc. - 2018 IEEE Int. Conf. Big Data Smart Comput. BigComp 2018, Institute of Electrical and Electronics Engineers Inc., 2018, pp. 635–637, https://doi.org/10.1109/BigComp.2018.00113.
- [275] Z. Wang, T. Hong, M.A. Piette, Predicting plug loads with occupant count data through a deep learning approach, Energy 181 (2019) 29-42, https://doi.org/ 10.1016/j.nergy.2019.05.138.
- [276] J. Redmon, S. Divvala, R. Girshick, A. Farhadi, You only look once: unified, realtime object detection, in: IEEE Conf. Comput. Vis. Pattern Recognit, 2016, pp. 779–788.
- [277] M. Jia, R.S. Srinivasan, R. Ries, N. Weyer, G. Bharathy, A systematic development and validation approach to a novel agent-based modeling of occupant behaviors in commercial buildings, Energy Build. 199 (2019) 352–367, https://doi.org/ 10.1016/i.enbuild.2019.07.009.
- [278] T. Csoknyai, J. Legardeur, A.A. Akle, M. Horváth, Analysis of energy consumption profiles in residential buildings and impact assessment of a serious game on occupants' behavior, Energy Build. 196 (2019) 1–20, https://doi.org/10.1016/j. enbuild.2019.05.009.
- [279] M. Arslan, C. Cruz, D. Ginhac, Understanding occupant behaviors in dynamic environments using OBiDE framework, Build. Environ. 166 (2019), https://doi. org/10.1016/j.buildenv.2019.106412, 106412.


This paper is not icncluded due to copyright restrictions Available in Journal of Building Performance Simulation 2022 ;Volum 15.(5) s. 634-655 https://doi.org/10.1080/19401493.2022.2061050

Paper f

Building and Environment 219 (2022) 109195



Contents lists available at ScienceDirect

Building and Environment

journal homepage: www.elsevier.com/locate/buildenv



A guideline to document occupant behavior models for advanced building controls

Check for updates

Bing Dong^{a,*}, Romana Markovic^b, Salvatore Carlucci^c, Yapan Liu^a, Andreas Wagner^b, Antonio Liguori^d, Christoph van Treeck^d, Dmitry Oleynikov^e, Elie Azar^f, Gianmarco Fajilla^g, Ján Drgoňa^h, Joyce Kimⁱ, Marika Vellei^j, Marilena De Simone^g, Masood Shamsaiee^k, Mateus Bavaresco¹, Matteo Favero^m, Mikkel Kjaergaardⁿ, Mohamed Osman^o, Moritz Frahm^P, Sanam Dabirian^o, Da Yan^q, Xuyuan Kang^q

^a Department of Mechanical & Aerospace Engineering, Syracuse University, 263 Link Hall, Syracuse, NY, 13244, United States

^b Building Science Group, Karlsruhe Institute of Technology, Englerstr. 7, 76131, Karlsruhe, Germany

^c Energy, Environment and Water Research Center, The Cyprus Institute, Nicosia, Cyprus

^d E3D - Institute of Energy Efficiency and Sustainable Building, RWTH Aachen University, Mathieustr. 30, 52074, Aachen, Germany

e Katernberger Straβe 107, Essen, Nordrhein-Westfalen, 45327, Germany

^f Department of Industrial and Systems Engineering, Khalifa University, Abu Dhabi, United Arab Emirates

⁸ Department of Environmental Engineering, University of Calabria, P. Bucci 44A, 87036, Rende, CS, Italy

h Pacific Northwest National Laboratory, Richland, WA, USA

ⁱ Department of Civil and Environmental Engineering, University of Waterloo, Waterloo, ON, Canada

^j LaSIE (UMR CNRS 7356), La Rochelle University, La Rochelle, France

^k School of Mechanical Engineering, Iran University of Science and Technology, Tehran, Iran

¹ Laboratory of Energy Efficiency in Buildings, Department of Civil Engineering, Federal University of Santa Catarina, Brazil

^m Department of Civil and Environmental Engineering, Norwegian University of Science and Technology, Trondheim, Norway

ⁿ The Maersk Mc-Kinney Moller Institute, University of Southern Denmark, Odense, Denmark

^o Department of Building, Civil and Environmental Engineering, Concordia University, Montreal, QC, H3G 1M8, Canada

^p Karlsruhe Institute of Technology, Institute for Automation and Applied Informatics, Hermann-von-Helmholtz-Platz 1, 76344, Eggenstein-Leopoldshafen, Germany

^q Building Energy Research Center, School of Architecture, Tsinghua University, Beijing, 100084, China

ARTICLE INFO

Keywords: Guideline documentation Occupant behavior model Building control Model Predictive control (MPC)

ABSTRACT

The availability of computational power, and a wealth of data from sensors have boosted the development of model-based predictive control for smart and effective control of advanced buildings in the last decade. More recently occupant-behavior models have been developed for including people in the building control loops. However, while important objectives of scientific research are reproducibility and replicability of results, not all information is available from published documents. Therefore, the aim of this paper is to propose a guideline for a thorough and standardized occupant-behavior model documentation. For that purpose, the literature screening for the existing occupant behavior models in building control was conducted, and the occupant behavior models in building control was conducted, and model implementation. The literature screening pointed out that the current state-of-the-art on model documentation shows little unification, which poses a particular burden for the model application and replication in field studies. In addition to the standardized model documentation, this work presented a model-evaluation schema that enabled benchmarking of different models in field settings as well as the recommendations on how OB models are integrated with the building system.

* Corresponding author.

E-mail address: bidong@syr.edu (B. Dong).

https://doi.org/10.1016/j.buildenv.2022.109195

Received 1 April 2022; Received in revised form 6 May 2022; Accepted 11 May 2022

Available online 20 May 2022 0360-1323/© 2022 Elsevier Ltd. All rights reserved.

1. Introduction

Building energy consumption has been proven to be a systematic procedure comprehensively influenced by not only engineering technologies but also cultural concepts, occupant behavior, social equity, etc. Occupant behavior (OB), discussed in this paper, refers to occupancy presence and the number of people in the spaces of a building, and human building interactions, such as window and blind operations, turning on/off lighting, as well as thermostat adjustment and use of electric appliances. As, occupants are one of the major factors that influence energy consumption [1], p. 79], depending on the level of building automation, the inclusion of the occupant-behavior modeling in building interactions [2,3]. Furthermore, the inclusion of human-building interaction [4] or OB in the control loop [5] could lead to a higher thermal comfort level and a general increase in occupants' satisfaction with the indoor environment.

However, OB models are rarely included in building controls, despite the vast scientific evidence that considering OB in building energy management could lead to optimized building performance [6]. Prior research studies show that the reason for the limited field applications of OB models could be the lack of OB model standardization and clear documentation [7,8], which results in models' limited replicability. The development of OB model standardization can enable easy integration and compatibility with existing or new building automation system (BAS). For instance, the inputs and outputs of OB models can be mapped with sensors and objects in BAS to enable occupant-centric building controls. Additionally, this standardization will ensure that the functionalities and requirements of an OB model are in alignment with building control requirement.

1.1. Existing reviews

The state-of-the-art, as well as an overview of related reviews that focus on OB modeling, are presented by Refs. [9–12], while the human dimension of energy consumption is reviewed by Ref. [13]. As concluded in the work by Carlucci et al. the predictive OB models are emerging, and this trend is evolving in parallel with the rise in the number of data-driven OB models. In this place, such predictive nature of data-driven OB models makes them promise for the application in advanced building controls such as model-based and model predictive controls. For the detailed revision of OB in the context of building control the reader is referred to Refs. [7,14–16]. Furthermore [17], reviewed occupant-centric control strategies, while the OB modeling was not in the particular focus of the latter work.

Complementary to the reviews of general OB modeling, the OB in the context of building simulation and in the context of building control has also been the focus of several recent studies [18–22].

1.2. Contribution of this paper

This work aims to fill the gaps required for the inclusion of OBs into building control, by proposing a guideline for model documentation and evaluation based on a comprehensive review of the scientific evidence and current state of technology. Since the building control and OB modeling were researched separately during the past years, the literature evidence did not provide a clear set of OB models that are developed and implemented in building control. For instance, OB models were commonly developed with a mentioned practical application for HVAC control, general building automation of smart buildings. However, the existing literature evidence does not provide clear recommendations, on which OB models can be used in building control and how to document these models for their real-world deployment. In order to bridge the gap between the two communities, we relied on our best domain expert knowledge and considered the OB models that are applicable to the building control.

From the control side, we put the spotlight on the OB models for the application in rule-based and more advanced control such as model based predictive controls. Further adaptive control paradigms that could include, but are not restricted to reinforcement learning, are not considered in the scope of this work. In this place, comprehensive and unified model documentation is required for model standardization and wide applicability. This model documentation also includes the guideline for suitable model performance evaluation, which is of crucial importance for the realistic presentation of the model's capabilities. In summary, this study aims to: (1) standardize OB model documentation to promote transparency through clear communication among researchers, reproducibility of experiments, (2) help researchers to select and adopt suitable models to fit their research needs, and (3) help researchers to understand the prerequisite, performance, application of the models they intend to use. In order to fulfill these goals, this work focus on the following research questions:

- (1) How are occupant models for real-time/predictive controls currently documented in the scientific literature?
- (2) How should occupant models be documented and implemented?
- (3) What are the evaluation metrics for different occupant behavior models?
- (4) What are the software platforms for future researchers to evaluate/validate their models?

2. Methodology

2.1. Guideline development

This section documents the development of the proposed guideline. As Fig. 1 shows, four major parts are included in this guideline: 1) Model description and applications which describes information representation, model inputs and output, and domain of applicability; 2) Model development detailed out data preparation, modeling formalism, and gaps in current model development documentation; 3) Model evaluation provides guidelines of selecting performance metrics which include absolute metrics, domain metrics, and indirect performance metrics; 4) Finally, in model implementation, we discussed the computational environment, computational time, experiment setup, and integration into MPC.



Fig. 1. Overview of the guideline development.

2.2. Review approaches and structure of the article

Based on an in-depth literature review process, this article aims to provide a guideline for a thorough and standardized occupant-behavior model documentation. The review focuses on six different categories of OBs, including Appliance Use, Lighting Operation, Occupancy Estimation and Prediction, Shading Operation, Thermostat Adjustment, and Window Operation. The literature search was conducted in Google Scholar with "Building" plus the aforementioned categories as keywords. Following the pre-defined categories, all the related literature was selected. Among those literature, OB models were reviewed from different perspectives, such as model description and applications (Section 3), model development (Section 4), model evaluation (Section 5), and model implementation (Section 6). We discussed our findings and future challenges in Section 7, and Section 8 concludes this paper.

3. Model description and applications

In order to answer the first research question, we have conducted a review on how current OB models are presented. The description of an occupant model typically includes three parts: information representation, model inputs, and model outputs. We will discuss those elements in the following sections. In addition, the domain of applicability is reviewed as well.

3.1. Information representation

Several ontologies and schemas, such as Industry Foundation Class (IFC), Green Building XML (gbXML), BPD Ontology, Brick, ASHRAE 201,. Have been developed to organize knowledge and structure data by describing both static (e.g. building geometry) and dynamic building data (e.g. time series temperature data) about building technical system, equipment, sensors, and corresponding relationships [23]. Each ontology or schema has its own focus area. For example, gbXML has been used to represent mostly energy performance simulation models with detailed material properties and geometries. BPD Ontology and Brick focus more on building operation data, which is typically measured by physical sensors and has a relationship with its location and measurement type. All ontologies or schemas aim to describe data and their relationship with building's devices. However, it is concluded that there is a lack of detailed documentation on existing data sets and models mostly due to the lack of guidelines as described in the discussion section. In addition, there is no metadata schema or ontology that can represent the full spectrum of occupant behavior models. For example, the occupant presence potentially could be represented by IFC as the "Timeseries" and attached to the "IfcOccupant" Class. However, the description of other occupant behaviors is very limited. Another necessary part of the occupant behavior modeling is a systematic description of input variables and parameters, prediction horizon or time interval. According to the review done by Na [8], eight out of 24 selected data tools can represent indoor and outdoor environmental data; however, none of the existing tools can store occupant behavior model parameters, unfortunately.

Furthermore, the terminology is an essential part of information representation. Na [8] concludes that only three data tools have defined terminology for occupant behavior-related data, which are ADI [24], Brick [25], and Project Haystack [26]. Within those three different metadata schemas, they have different naming for the same building component. The current lack of standardization in the names of sensors in commercial buildings creates challenges not only for occupant behavior modeling but also for building data integration and interoperability in general. There is a need to adapt different naming from the schemas and ontologies to have a unified naming guideline when documenting occupant behavior models.

3.2. Model inputs and outputs

The current section details the inputs (independent variables) and outputs (dependent variables) used in the OB models reviewed in the current work. The information is presented for the six main categories of behaviors considered in the models: (1) Appliance use, (2) Lighting operation, (3) Occupancy estimation and prediction, (4) Thermostat adjustment, (5) Shading operation, and (6) Window operation.

Fig. 2 presents a count of the inputs and outputs used when modeling appliance use (left side of the figure) and lighting operation (right side of the figure). Starting with the former, the most commonly studied outputs are predicting the multi-state of appliances, or their energy consumption levels. The most frequent inputs used to predict the mentioned output are mostly the plug-load energy (i.e., historical data used to predict future use), followed by the space's occupancy status (i.e., using occupancy presence/absence information to predict appliance usage). As for lighting operation (Fig. 2, right), the most frequent outputs are the state (either binary or multi-state) and the operation time. Illuminance levels, occupancy status, and power consumption of other systems (e.g., plug-loads) are the main inputs used to model lighting operation.

Fig. 3 summarizes the inputs and outputs used in occupancy estimation and prediction (left side) and thermostat adjustment (right side). Starting with occupancy models, the two most dominant outputs are the presence status (binary) and the number of occupants. Unlike the previous target behaviors, a wide variety of inputs is used to predict occupancy, including historical occupancy patterns, motion detection, power usage, and indoor environmental measures (e.g., illuminance, temperature, relative humidity, CO2, and VOC levels). As for thermostat adjustment (Fig. 3, right), the temperature setpoint setting is the most frequent target variable. Other outputs include indoor temperature, the probability of adjusting thermostat settings, and energy consumption. Here again, various parameters are used as predictors for these models, such as indoor/outdoor temperatures and humidity, solar radiation, CO2 levels, hour of the day, and electricity load and price.

Fig. 4 presents the count of inputs and outputs used in models of shading operation (left side) and window operation (right side). For shading operation, the listed outputs are all well represented in the reviewed models. They include the shading state (binary or multi-state), the probability of having blinds up or down, and the portion of blinds up or down. The predictors of the stated outputs are primarily environmental in this case, namely indoor/outdoor temperature, illuminance, and solar radiation. Moving to the right side of Fig. 4, the most considered output of window operation is the probability of window state, followed by the probability of taking action (e.g., opening/closing a window) and the portion of a window open, respectively. Here again, the inputs to such models are mostly environmental, namely indoor/ outdoor temperature and humidity, wind speed and direction, solar radiation, rainfall, and concentrations of CO2 and particulate matter.

3.3. Domain of applicability

This section details how the spatio-temporal domain is documented in the OB models reviewed in the current work. The temporal dimension is represented by time granularity, prediction horizon, and control horizon. While the spatial dimension is represented by the space (e.g., room, floor, building level) the OB model is addressing.

The time granularity is the time-step or shortest time window operation from which the information regarding the occupant's behavior is used for prediction (e.g., presence model in 15-min resolution). Fig. 5 shows the distribution of the availability of this information in the reviewed papers according to each target behavior. Models of lighting operation are the least documented in terms of time granularity. For the other target behaviors, about 30% of the papers do not report the used time discretization information. Sometimes, the time-step is not explicitly documented because the authors imply that the time granularity of the model is the same as that of the sensed data. The

Appliance Use Inputs↓ Outputs→	Multi-State	Binary-State	Energy Consumption	Prob. of On/Off	Presence	Schedule	Lighting Operation Inputs↓ Outputs→	Binary	Multi-State	Portion of Light On	Visual Comfort	Probab. of Switching	Power Consumption	Operation Time
Visual Data	1	0	1	1	2	0	Tin	0	0	0	0	0	1	0
Acoustic data	2	0	0	0	0	0	Tout	1	0	0	1	0	1	0
Plug Load	16	0	11	0	1	1	RH(in)	0	0	0	0	0	1	0
Price	1	0	0	0	0	0	RH(out)	0	0	0	1	0	1	0
Wireless Signal	1	0	1	0	2	0	Wind Speed	1	0	0	1	0	0	0
Tin	1	0	0	0	0	0	Illuminance	11	21	1	1	1	3	6
Tout	1	0	1	0	0	0	Solar Radiation	1	1	1	1	0	1	1
Door Operation	1	0	0	0	0	0	Occupancy	5	5	1	2	0	1	11
Presence	2	0	7	0	1	2	Power	2	4	2	1	1	2	10
RH(in)	0	0	1	0	2	0	Color	0	1	0	0	0	0	0
CO2	1	0	0	0	2	0	Visual Comfort	0	2	1	0	1	1	0
Wind Speed	0	0	1	0	0	0								

Fig. 2. Count of inputs and outputs used in models of appliance use (left) and lighting operation (right) behaviors.

Occupancy Estimation and Prediction Inputs↓ Outputs→	Presence (Binary)	Occupant Number	Percent Occupied	Position	Thermostat Adjustment Inputs↓ Outputs→	Multi-State	Binary-State	Energy Consumption	Cycle Idle Time	Prob. of Adjustment	COP	Tin	Tsetpoint	Portion of Adjustment	Tsupply
Occupant Entry/Leave	0	2	0	0	Tin	1	1	5	0	5	1	4	12	2	0
Occupant Number	0	3	0	0	Tout	1	1	3	0	4	1	5	10	3	1
Presence	6	2	1	0	Presence	0	1	2	0	0	0	2	6	0	1
Motion	3	5	0	0	RH(in)	0	1	0	0	2	0	1	6	0	0
Illuminance(in)	3	3	0	0	RH(out)	0	1	0	0	1	1	1	4	0	0
Tin	4	2	0	0	Solar Radiation	0	1	2	0	1	0	2	3	0	1
RH(in)	4	2	0	0	Wind Speed	0	0	1	0	1	0	0	1	0	0
LED Reading (Volt)	0	1	0	0	CO2	1	0	1	0	2	0	1	3	0	0
CO2 concentration	5	7	0	0	Tsetpoint	0	0	1	0	0	0	0	0	0	0
VOC	2	0	0	0	WinOpen	0	0	1	0	1	0	0	1	0	0
Power Usage	6	1	0	0	ShadeOpen	0	0	0	0	1	0	0	0	0	0
echo intensity	1	0	0	0	Hour	0	0	0	0	0	0	0	3	0	0
Acoustic Level	1	3	0	0	Electricity Load	0	0	1	0	0	0	0	4	0	0
Telephone State	1	0	0	0	Electricity Price	0	0	0	0	0	0	0	3	0	0
Keyboard/Mouse State	0	1	0	0	Air Movement	0	0	0	0	0	0	0	1	0	0
Chair State	0	1	0	0	Occupant thermal	0	0	0	0	0	0	1	1	0	1
AC State	1	0	0	0	Clothing thermal	0	0	0	0	0	0	0	1	0	0
Door State	1	1	0	0											
Light State	0	1	0	0											
Window State	1	0	0	0											
Signal Strength	0	0	0	2											
Number of Device on	0	1	0	0											
MAC address	0	3	0	0											
Device Location	0	1	0	0]										

Fig. 3. Count of inputs and outputs used in models of occupancy (left) and thermostat adjustment (right) behaviors.

documented time granularities cover a broad range from less than 1 min to hours (Fig. 6). This depends on several factors, including the granularity of the sensed data available, the temporal range in which the change of behavior in question occurs, and the envisaged predictive horizon. A time resolution between 10 and 19 min is the most frequently adopted.

The predictive horizon is the time horizon over which the OB is modeled. The predictive horizon is much less documented than the time

granularity (Fig. 7) and covers a wide range of values, from less than 1–24 h (Fig. 8). This is to be expected since it is strongly dependent on the controlled variable. For example, the predictive horizon for building predictive HVAC control is related to the type of heating and cooling system and can be relatively long for radiant floor heating systems compared to air-based ones, the response time of which is very fast.

The control horizon is the time horizon over which the control variable is modeled. The control horizon is commonly equal to or longer Li

Portion of window open

10

10

3

4

1

5 3

0

1 3

2

1

5

2

1 0

Shading Operation Inputs↓ Outputs→	Shading (Binary-State)	Shading (Multi-State)	Prob. of Blind Up/down	Portion of Blind Up/down		Window Operation Inputs↓ Outputs→	Multi-State	Prob. of window state	Prob. Pf Action	Time of Opening
Tin	4	4	4	3		Tin	2	20	15	1
RH(in)	1	2	4	2		Tout	2	20	15	0
Illuminance(in)	3	6	3	2		RH(in)	1	10	9	1
VOC(in)	0	0	2	1		RH(out)	1	13	8	0
CO2	0	0	1	0		Air Velocity	0	0	0	0
Air Velocity(in)	1	0	1	2		Wind Speed	1	12	8	0
Heating/Cooling	1	0	0	1		Wind Direction	1	9	2	0
Lighting Load	1	0	0	1		Illuminance (in)	0	0	0	0
Tout	3	5	5	3		Illuminance (out)	1	2	1	1
RH(out)	0	3	2	0		Solar Radiation (out)	1	5	6	0
Illuminance(out)	1	0	0	1		Sunshine Hour/ Date/ Time	0	4	1	0
Solar radiation	3	5	4	4		Rainfall	0	6	2	0
Solar	0	2	1	0		CO2	0	12	8	0
Wind Speed(out)	0	1	2	0		PM2.5/PM10 (out)	0	6	2	0
Heating/Cooling	1	0	0	1		VOC	0	0	1	0
Lighting Load	1	0	0	1]	Image	0	1	0	0
						Position sun protection	0	1	0	0
						Occupancy	1	2	1	0
						Person charactersitic	1	1	1	0
						Room charactersitic	1	1	1	0
						Time of day	1	1	1	0

Fig. 4. Count of inputs and outputs used in models of shading operation (left) and window operation (right) behaviors.



Fig. 5. Availability of the time granularity according to the target behavior.

than the predictive horizon. The control horizon is explicitly documented in only 5% of the reviewed papers. The lack of this information can be explained by the fact that the great majority of the reviewed papers propose OB models as an input for model predictive control (MPC) but do not actually apply it in predictive control. However, the practical implementation of these OB models in the field is currently lacking.

Regarding the space granularity, i.e. the space (room, floor, building level) the OB model is addressing, a majority of occupancy, shading, and lighting models have been developed only at room level (Fig. 9). Instead, appliance use, window operation, and thermostat adjustment models



Fig. 6. Distribution of the time granularity in minutes according to the target behavior. (If a model was developed with more than one time-step; only the smallest was considered in the figure).

have been mostly addressing the building level that, for the residential case, corresponds to an entire house. Only a minority of OB models have been addressing a lab-based installation, such as a test cubicle.

4. Model development

To develop an occupant model, one needs to prepare raw data into a format that can be used for modeling data (4.1) and identify a modeling method or algorithm that is applicable and practical for a particular



Fig. 7. Availability of the predictive horizon according to the target behavior.



Fig. 8. Distribution of the predictive horizon in hours according to the target behavior. (If a model was developed with more than 1 predictive horizon; only the smallest value was considered in the figure).



Fig. 9. Distribution of the space granularity for occupancy models.

problem (4.2). In this section, we will review diverse approaches and techniques employed in occupant models and discuss gaps in current research and development in this field (4.3).

4.1. Data preparation

Data used in occupant modeling is often collected in different

structures, granularities, and volumes as described in section 2. Hence, it is important to prepare or preprocess the raw data into a format that is suitable for intended analysis and modeling [27]. Preprocessing can include, but is not limited to, the following steps:

- Cleaning and imputing the missing and corrupt data, outliers by discarding or replacing them with inferenced values (e.g., moving average, mean, or median) to be easily parsed by machine;
- Reducing data dimensions using row-wise for data sample reduction or column-wise for data variable reduction and random sampling methods;
- Data scaling using min-max normalization [28], distribution-based standardization [29], or structure-based techniques [30] to scale the data into a consistent range;
- Feature creation to construct new variables of existing features for data analysis;
- 5. Data partitioning using supervised (e.g., decision tree [31]) and unsupervised techniques (e.g., k-means [32], Gaussian Mixture Models [33]) to divide the dataset into the test and training subsets to evaluate the trained model based on the test set;
- Merging data from different sources with various time intervals within time-series data.

Preparing data is often the most time-consuming portion of datadriven modeling. Yet, only a few studies in the reviewed literature describe the data preprocessing methods used for occupant models. The examples are as follows. Jin et al. [34] used the confusion matrix to evaluate the quality of PIR sensor data and remove inaccurate occupancy states. Q-test was used to identify outliers [10]. To handle the missing values, Yu et al. [29] used the moving average method to fill in missing entries. Also, they calculated the SHapley Additive exPlanation (SHAP) values of each feature to reduce the HVAC operation data dimensions. Ashouri et al. [28] employed min-max normalization to standardize energy consumption data. K-means clustering was used to recognize distinct air handling unit operation patterns and group the BAS data accordingly [29]. In another study, hierarchical clustering was conducted to extract the occupancy patterns in the building and create new features for occupancy models [35]. Given that methods and assumptions used in the preprocessing stage can affect data analysis and prediction outcome, there needs to be a concerted effort to document detailed preprocessing steps in future studies.

4.2. Modeling formalism

In this section, the models' category distribution is presented for each target behavior. The modeling categories were defined based on the state of the research: deterministic rule-based models, statistical/ stochastic, and data-driven [36]. Rule-based models are the deductive models that use an a priori set of rules for describing occupant behaviors in building models, including deterministic models and schedules. The statistical/stochastic is stochastically modeled the OB to represent the various behaviors among the population [9], potential change over time [18], and complexity [10]. These models are commonly represented by statistical models such as a-priori probability density functions [37]. The third modeling formalism is the data-driven modeling, where the focus was put on computational intelligence or machine learning without an explicit aim to explain the relationship between the input variables and the OB [38]. It includes the ML models and ABM.

The reviewed OB models for building control are screened for the used modeling formalism, and the results are presented in Fig. 10.

The data-driven methods are the most used. The second most implemented category is represented by statistical models that have been applied especially to model windows operation (80%). Rule-based models' category has been applied mainly for shading and has not been tested for appliance use and window operation behavior. Some documents did not provide information about the implemented models. It



Fig. 10. Distribution of the model categories according to the target behavior.

happened for occupancy, lighting, and shading behavior in a percentage of 8%, 17%, and 8%, respectively.

Fig. 11 deepens the most adopted models in the target behavior, considering those with a percentage higher than 10%. Regarding the rule-based category, most of the reviewed studies do not provide detailed information on the models used, merely defining their belonging category. Schedules have been implemented in around 40% of the cases related to lighting usage.

About the statistical models, the Markov chain appeared the most adopted for occupancy and thermostat adjustment; Markov chain Monte Carlo models have been often implemented for modeling lighting use. Regression models have been mainly found in the case of window and shading operations with a percentage of around 74% and 100%, respectively. Often, information on the adopted model was not provided, as in the case of appliance usage (67%).

Regarding the data-driven models, neural networks (NN) have been the most common. Control logic and fuzzy logic were utilized with the same percentage of neural networks (around 20%) for shading operation. Similarly, also in the case of appliance use, clustering and long short term memory (LSTM) were implemented with the same percentage of neural networks.

Fig. 12 provides the distribution of models' categories in six building types. Rule-based modeling has been mainly applied in commercial buildings for occupancy, lighting, and shading operation. Furthermore, they have been implemented in residential buildings for occupancy, in educational buildings for lighting, and in institutional edifices to model shading operation. The statistical approach has been principally used in residential buildings and prevailed in the case of thermostat, appliance use, and window operation; the application in commercial buildings was diffuse in case of occupancy, shading, and window behavior. In the case of lighting, the implementation of statistical models was equally distributed in residential and commercial buildings. Moreover, statistical models are diffuse to predict occupancy and window operation in



Fig. 11. Most adopted models according to the target behavior and the model categories (rule-based, statistical/stochastic, and data-driven).





Residential Commercial Educational Religious Laboratory Institutional

Fig. 12. Model categories (rule-based, statistical/stochastic, and data-driven) according to the target behavior and building typology.

educational edifices, whereas appliances use in laboratories.

The data-driven category found notable implementations in commercial buildings for lighting, shading, and window operation in both residential and commercial buildings for thermostat adjustment and appliance use. Educational buildings prevail for occupancy modeling, and laboratory edifices for shading operation. Other buildings typologies, such as religious and institutional buildings, have been rarely investigated for occupancy, lighting, and appliance use.

Fig. 13 presents the model categories distribution considering the space granularity. In general, shading and occupancy behavior have been investigated at room level and lighting in offices.

On the other hand, thermostat adjustment and appliance use have been modeled in apartments as they are principally studied in residential buildings.

Occupancy is also frequently detected in buildings, whereas lighting operation has been modeled considering more varied space typologies: controlled environments, such as laboratories and test chambers, classrooms, and offices.

Fig. 14 presents the model categories distribution considering the time granularity. Generally, rule-based models were developed collecting occupancy data with a time step minor than 10 min, whereas the temporal time step was prolonged and longer than 60 min in case of shading behavior. In statistical models, occupancy, thermostat, and



Fig. 13. Spatial granularity in model categories (rule-based, statistical/stochastic, and data-driven) according to the target behavior.

window were commonly detected with a time step minor than 20 min. Lighting operation behavior was mainly modeled by data collected with a temporal step minor than 10 min. In the data-driven category, the time step used for appliances usage detection was generally equal to 45 min. For the other behaviors, brief time steps, less than 20 min, were also utilized.

4.3. Gaps in model development documentation

Scientific research should be 'reproducible and replicable'. Reproducibility 'means obtaining consistent results using the same input data, code, computational steps, and conditions' while replicability 'means obtaining consistent results across studies aimed at answering the same scientific questions using different data (https://www.nap.edu/catalo g/25303/reproducibility-and-replicability-in-science). Reproducibility is challenging to attain because it involves sharing the data, which may be nowadays limited by, for instance, personal data protection needs and privacy issues. Replicability could be easily achievable compared with reproducibility. However, it demands documenting the steps undertaken in the development of the model in a transparent and detailed way. Therefore, the critical aspect is to detail the entire process – the complete workflow – rather than a specific part of it (e.g., the results).

With these premises, the model development should start with the explicit formulation of its problem; that is the modeling purpose. This implies that if a model is developed for control purposes, it should be

Building and Environment 219 (2022) 109195



Fig. 14. Time granularity in model categories (rule-based, statistical/stochastic, and data-driven) according to the target behavior.

ready to be directly implemented in control logic. Consequently, this aspect should be already addressed in the model development. For example, for application in model based control or MPC, the control horizon depends on the kind of heating and cooling system used: it can be rather long for radiant floor heating systems compared to air-based systems. Subsequently, the modeling formalism that provides the OB sequence during the whole predictive horizon should be chosen and described. Reasons for the selection of a particular method should be given from both a practical and theoretical point of view. It should also explain how the predictors included in the model were chosen. This includes stating whether feature selection was used to reduce the data dimension and which approach was used. Also, it is important that the variables included in the model as a predictor but is not measured in the building, the model is not applicable in practice.

From the reviewed literature, only 35% of the papers state that the aim was building/zone/HVAC control application, and only 19% offer a formal integration of the model into a control logic. Therefore, the majority of the available occupant models are not designed with control purposes in mind, which directly impacts the modeling formalism used. This kind of model aimed to represent what the behaviors were based on the data collected and not what the behavior will be. In models for control purposes, time becomes a critical factor that a model should directly account for. This translates to recognizing and considering the model's input variables as a function of time.

Another issue regards the integration of real-time data into the model to be updated when new information is collected. This would also require measures of the dependent variable, that is, the behavior that the model aims to predict. The ability to self-update and adapt to real-time data strongly affects the modeling strategy's choice.

Furthermore, most of the behaviors are modeled independently from each other. However, in reality, this is rarely the case. For instance, the shading operation can affect both lighting operation and thermostat adjustment behavior. Moreover, the necessary condition for most of the occupant behavior is occupancy estimation and prediction. This implies being able to measure it or predict it. In the latter case, this will result in ulterior uncertainty in predicting behavior. For example, if window operation behavior is the dependent variable to predict, its prediction will be affected by the model error plus the prediction error for the occupancy status.

5. Model evaluation

Mode evaluation is a logic and necessary next step after model

development. This section focuses on the model evaluation and documenting the model's performance. It consists of the reviewed literature evidence on the OB model evaluation and a proposed guideline for the standardized documentation of model performance. For that purpose, the performance metrics are structured into absolute, domain-specific, and indirect metrics, and their purpose is briefly elaborated. Finally, the sensitivity analysis is introduced, as an additional tool to quantify the model performance and document the uncertainties.

5.1. Performance metrics

The model evaluation [39] is structured into absolute evaluation, domain metrics, and indirect metrics as showing in Fig. 15. The absolute metrics relate to the performance indicators used for general statistical or data-driven modeling. Here, we quantify how often an OB model provides a correct prediction, or we use the absolute metrics to evaluate the performance in case of data imbalance. The domain metrics are defined using the OB and buildings physics knowledge. For instance, in the case of window operation modeling, we are not only interested in what percentage of window states is modeled correctly, but also how often a window operation occurs or what the median duration of sequences with open windows is. Lastly, the indirect metrics quantify the impact of the modeled OB on the data modeling objectives of building control: does the use of the window operation model lead to improved thermal comfort, or how does it affect the HVAC system, such as the resulting impact on energy consumption or thermal comfort.

5.1.1. Absolute metrics

The absolute metrics is based on the definition proposed by Ref. [40], namely "the metrics that are based on the absolute error calculation". The main goal is to assess the goodness of the model for fulfilling a particular task, to compare alternative approaches, and to quantify if the design updates made on an OB model led to the model's improvement. The initial step in the selection of the evaluation metrics is the assessment of the nature of the modeling objective; for example, whether the target variable is categorical or continuous.

Since the absolute metrics may result in bias in the interpretation of the results [41], these should be selected based on the nature of the modeled data and target function formulation. In this regard, the target function formulations considered for the OB modeling are continuous, categorical and the special case of binary categorical variables. The resulting absolute evaluation metrics should be defined based on the target function formulation and the particular challenge in each OB model.



The literature screening showed that there is a significant portion of the modeling studies in which the validation and testing performance was not performed. Furthermore, the conducted literature review pointed out the lack of standardized model evaluation metrics. However, similarly to the generic data-driven or statistical modeling, the performance is commonly reported using mean average error (MAE), mean squared error (MSE). Additionally, the precision, recall, and F1 scores were used by some of the existing studies that focused on classification tasks.

Based on the empirical evidence on the target variable formulation and the nature of OB data, the minimal requirements on the set of absolute metrics for each OB are summarized in Table 1. In summary, most of the OB modeling should be treated as classification problems. The multi-class categorical target functions include the shading operation, while the window operation, occupancy estimation and prediction, lighting operation, and appliance use should be modeled as the binary classification. At the same time, the occupancy count can be modeled as a regression problem.

We use modeling window status as an example. The fundamental issue to be addressed is the imbalanced prior probabilities of the window states. Therefore, the model evaluation should include the MAE, confusion matrix, and F1 scores. Similar to the window states, the occupancy estimation and prediction, appliance use, and lighting operation are also commonly formulated as binary classification problems, and the model's performance can be quantified using the same metrics. To this end, the model goodness quantified using confusion matrix and F1 score could also be reported using precision and recall. The models that represent shading operation can be evaluated using MAE, confusion matrix, and F β score. The reader is referred to Li et al. [42] for further elaboration on the choice of the evaluation metrics.

In the case of the thermostat setpoint modeling, there are very limited studies on model validation. However, we argue that the setpoint modeling should be treated as a continuous problem since the setpoint changes could be treated as rare events. By treating the thermostat set point modeling as a regression problem, the relative value of the thermostat set point is to be modeled, while the setpoint changes would be addressed in an implicit fashion.

5.1.2. Domain metrics

The domain metrics are defined as a fit-for-purpose metric that evaluates the competence of the model in representing a certain OB considering the stochasticity of results. The intention of developing these metrics is to provide comparable means for assessing how well do OB models represent particular forms of human behavior. Moreover, considering the main purpose of building as a comfortable and productive space for people [43,44], these metrics standardize building performance from the perspective of its occupants. In the existing work, Tahmasebi and Mahdavi [45] presented domain metrics for comparing

Table 1

Recommended target formulation and minimal set of evaluation metrics for each modeled OB.

	Data type				Absolute metrics									
	Continuous	Categorical	Binary	ACC	Balanced accuracy	Confusion matrix	F1	Fβ	MAE	MSE	RMSE	N- RMSE		
Window Operation		Х	Х	Х		Х	х		Х					
Thermostat Adjustment	х								х	Х	Х	Х		
Occupancy Estimation and	х	Х	Х	х		Х	Х		Х	Х	Х			
Prediction														
Shading Operation		Х		х		Х		Х	Х					
Lighting Operation		Х	Х	х		Х	Х		Х					
Appliance Use		Х	Х	Х	Х	Х	х		Х					

the performance of OB models in building simulation. However, the related literature in the context of building control is sparse and there is a need for developing consistent domain metrics [46-48].

The domain metrics are categorized based on types into aggregated and interval-by-interval groups as summarized in Table 2. The aggregated domain metrics stand out as proper evaluation criteria when the tested OB model is used for long-term purposes such as (estimating total energy saving, benchmarking building performance, etc.). On the other hand, the interval-by-interval metrics are preferable for evaluating OB models in short-term applications such as (building control, MPC, demand response studies, etc.). Table 1 summarizes the metrics defined for each distinct behavior, and it highlights the research gap in developing novel interval-to-interval metrics for thermostat domains. To fill this

Table 2

Domain metrics for each OB type.

	Purpose	Domain	Metric	Refs
Aggregated	Long-term purposes (estimate energy saving, model building performance, etc.)	Lighting Operation	Typical lighting operation profile Frequency of switching-on actions	[48]
		Window Operation	Overall fraction of open state [%] Mean number of actions per day averaged over the observation time Open state durations' median and interquartile range [hour] Closed state durations' median and interquartile range [hour].	[47]
		Occupancy Estimation and Prediction	Occupancy State Matching (SM) error Occupancy Duration (OD) error [h] Number of Transitions (NT) error	[49]
		Appliance Use Thermostat	Appliance's daily turn on times Appliance's average use duration Accumulated on- state duration N. A	[50]
Interval-by- interval	Short-term purposes (demand response, MPC, etc.)	Adjustment Lighting Operation Window Operation	The stepwise energy use N. A	[48]
		Occupancy Estimation and Prediction	First Arrival time (FA) error [h] Last Departure time (LD) error [h]	[49]
		Annliance Use	Prediction interval (PI) Coverage width- based criterion (CWC) N A	[46]
		Thermostat Adjustment	Prediction interval (PI) ^a Coverage width- based criterion (CWC) ^a	[49]

^a Metrics are suitable to be shifted from different domain.

gap, we argue that some of the developed metrics could be transferred between different domains with minor adjustments. For example, the occupancy state matching error is the percentage of false state predictions which indicate the mismatch between actual and predicted occupancy; this metric could be adjusted for evaluating appliance use models. Chong et al. [46] used the coverage width-based criterion that comprehensively evaluates the quality of prediction interval to evaluate the performance of occupancy estimation and prediction models which can easily be adjusted to evaluate the performance of OB thermostat adjustment models.

5.1.3. Indirect performance metrics

Indirect performance metrics evaluate to which extent the OB model contributes to fulfilling the control goal, such as energy consumption reduction or improving thermal comfort. For example, the integration of occupancy estimation and prediction into temperature control can minimize the heating demand. In that case, the used energy could be defined as the control metric together with absolute and direct metrics in the comprehensive model evaluation. One of the examples of jointly used domain and indirect evaluation metrics is presented by Peng et al. [51] the occupancy model was evaluated using both domain metrics (probability and duration of room occupancy) and indirect metrics (total consumed energy consumption).

A summary of literature about control metrics applied for the different OB models can be obtained from Appendix I. The literature screening pointed out that there is only limited evidence of the documented indirect control metrics for OB models, which could be a result of the rare availability of control use cases. When included, control metrics are frequently used to compare the performance of control including OB modeling versus one without it. The control metrics can be absolute or relative, for example, the energy consumption or saving in kWh (absolute) or the energy reduction in percent (relative).

In the literature, where control metrics are available, the main focus of most control algorithms is to minimize energy consumption while maintaining comfort constraints. Naylor et al. [17] reviewed occupant-centric building control strategies in regard to their energy reduction and obtained between 20 and 50% reduction in most cases. Despite the energy reduction, the comfort should usually remain in a certain range, e.g. an indoor temperature between 20 and 23 °C. The most dominant comfort control metric obtained from literature is thermal comfort, as most cases of including OB (occupancy, thermostat, windows, shading) into control are related to HVAC systems [52,53]. Shading operation is not only relevant for thermal but also visual comfort metrics. Indeed, previous work has combined parameters from both domains in an algorithm to control window blinds [54]. Lightning is only related to visual comfort, mainly by guaranteeing appropriate illuminance levels at workstations. By doing so, visual comfort and energy efficiency metrics may be combined - even during non-office hours, using algorithms able to minimize illuminance targets for unoccupied workstations [55]. For thermal comfort, most authors use the control metrics indoor air temperature, the predicted percentage of dissatisfied (PPD), or the predicted mean vote (PMV); for visual comfort, illuminance, or false-off frequency are typically used.

6. Model implementation

This section aims to provide a guideline on the best practices for implementing OB models in a software environment. This part assumes that the creator has completed the model development phase for the OB models so that these models are in the form of a stand-alone application. In this context, we refer to a stand-alone application as "a set of a USER's information processing requirements" [56]. In that context, this section provides recommendations regarding the model's computational environment, runtime analysis and scalability analysis for the real-time capabilities, and the experimental hardware settings.

Primarily, we focus on the OB models' implementation as the

outcome of the academic or general research activities. The aim of this section is to propose the documentation that enables model's reproducibility in building control systems that may have different software architectures [57].

6.1. Computational environment

The documentation on the computational environment in the sense of OB models in building control should include requirements for specific operating systems (OS), programming languages, and library or other software dependencies. Beyond this information, the used versions should also be reported. The details required are relevant both for the model's reproducibility in form of the application and due to the copyright requirements of each dependency in case of the (potentially commercial) field deployment. Furthermore, the operating system and programming language should be documented in the context of the runtime evaluation that is a crucial component of the OB model documentation and that is a programming language and OS-specific.

When selecting a suitable computational environment for model development, the evidence regarding the widely adopted environments could be beneficial, and this information is summarized in Table 3. Among others, R and Matlab/Simulink are the most commonly utilized programming language and software packages that can be employed for almost every type of OB. Oppositely, in terms of programming languages C/C++ and VBA and for software packages IBM SPSS and Weka are less used for developing already existing OB models. In practice, one can observe a large variability for different platforms and the computational efficiency of OB algorithms [57]. Nevertheless, information related to the utilized operating system is sparsely documented in existing studies.

6.2. Computation time

Commonly, the OB models are developed with the aim to be included in the end-systems such as building control that typically operates in real-time. Since the computation resources within building control and related end-systems are limited, an estimate of the required resources is required to assure the real-time operation. In this place, we refer to the computation resources of the OB models, which are defined as standalone modules that can be coupled with an end-system in various distributed manners. We focus on the runtime of the developed final models where the executed steps include the data-preprocessing and computing the value of the OB target variable. In the case of the machine learning-based models, this would correspond to a model test, while the model training and validation are considered to be previously completed. The hosting of these models is taking place within the building control, using a cloud-edge solution or on a remote cloud. In case of any of the listed computation settings, the following information should be provided:

- In which computation environment is the runtime analysis conducted?
- What is OB model inference runtime?
- Inference memory requirements?
- Optional: total required training runtime

- How does the OB model scale in space and time with the number of modeled OB instances?

The runtimes should be expressed either in core hours or the clock time, given the standard setting. The runtime should be documented together with the used hardware model. Since the majority of the OB models were created in the scope of academic research efforts, there is limited literature evidence on the runtime documentation. Among the others, the reader is referred to Refs. [57–59], and [60] for some best practices.

Additionally, the model's scalability is of particular importance and should be documented. Namely, OB models can be applied to a large number of occupants within a building and therefore the model's scalability in space and time (footnote: for further information regarding the space and time complexity, the reader is referred to Ref. [61] should be documented and expressed using "big O" notation with respect either to the number of occupants, rooms or buildings (further information about the "big O" notation is summarized by Ref. [62]. Additionally, in case an O model is intended to be used for varied temporal resolution and predictive horizon, the time complexity should also be documented with respect to these two parameters.

6.3. Experiment setup

With the computational environment guidelines discussed above, this section focuses on presenting the experiment setup by summarizing the findings from the literature review. Sensor choices and implementation location will be discussed in the following subsections. The discussions are based on six main categories of occupant behavior models, as followed throughout this paper. The sensor choices subsection offers information of sensors that have been deployed among different studies, implementation locations subsection presents different locations of deployed sensors in different research experiments. This section aims to provide guidelines for future occupant behavior researchers to deploy sensors and set up experiments.

6.3.1. Sensor choices

From the reviewed literature, in total 85% of the studies have explicitly provided information about sensors that have been deployed. Table 4 summarizes the commonly used types of sensors and the aggregated frequency that they have been picked in the literature. The color scales in the table represent how often the specific sensors were adopted. It can be observed that, for "Appliance Use" studies, current/ power sensors and smart meters are very commonly used; for "Light Operation" studies, lighting sensors and PIR sensors are primarily adopted; for "Occupancy Estimation and Prediction" studies, PIR sensor and CO2 sensor are commonly used; for "Shading Operation" studies, lighting sensors and indoor temperature sensors are commonly used; for "Thermostat Adjustment", indoor temperature sensors, sound sensors, and airspeed sensors are primarily used; for "Window Operation" studies, indoor temperature sensors and window state sensors are commonly used. Apart from the aforementioned most commonly used sensors, other sensors are also summarized in the table.

Table 3

The most common computational environments for each OB model.

	Pro	gramming l	anguages			Software pa	ckages/tools				
Domain		Python	C/C++	Java	VBA	IBM SPSS	Modelica/Dymola	Matlab/Simulink	Weka	LabVIEW	RapidMiner
Window Operation	х	Х				х	Х	х			х
Thermostat Adjustment	Х		Х					Х			
Occupancy Estimation and Prediction	Х	Х		Х			Х	Х		х	
Shading Operation	Х							Х			
Lighting Operation	Х			Х	х			Х			
Appliance Use								Х	х	Х	Х

Table 4

Sensor Choices of the Reviewed Studies.

Sensor Type	Appliance Use	Lighting Operation	Occupancy Estimation and Prediction	Shading Operation	Thermostat Adjustment	Window Operation
AC State	2	0	1	0	1	0
Air Pressure	0	0	1	0	5	2
Air Speed	0	1	2	0	10	8
Airflow Rate	0	1	3	0	5	0
Bluetooth Beacon	1	0	1	0	0	0
Camera	4	3	11	0	2	2
Chair	0	0	1	0	0	0
СО	0	0	1	0	1	1
CO2	1	0	18	0	7	16
Current/Power	10	2	5	1	5	0
Door State	1	0	4	0	0	1
GPS Location	0	0	1	0	0	0
Keyboard&Mouse	0	0	1	0	0	0
LED	0	1	1	0	0	0
Light	2	36	8	11	1	5
Light Switch	0	5	1	0	0	0
Motion (Unspecified)	2	6	4	1	1	0
PIR	3	11	19	2	1	6
RF	0	1	1	0	0	0
RH	1	1	12	1	9	18
Smart Meter	9	2	5	0	1	0
Smart Plug	4	0	0	0	0	0
Solar Irradiance	0	3	2	5	10	7
Sound (Acoustic)	0	0	6	0	0	0
Sound (Echo-Based)	1	0	2	0	0	0
Sound recording device	1	0	0	0	0	0
Telephone(State)	0	0	1	0	0	0
Temperature (Indoor)	0	3	13	6	16	33
VOC	0	0	2	0	0	1
WiFi Connection/Probe	1	3	5	0	0	0
Window State	0	2	0	0	2	32

6.3.2. Implementation location

Table 5 summarized the common locations of sensors deployed from the literature. The locations are categorized into two levels: Space Level and Building Level. Under each level of locations, detailed locations and related sensor types are provided. Based on the review work, this table provides a guideline for future researchers to refer to when deploying sensors in an experimental set up.

6.4. Integration into MPC

OB models can be used in MPC for setpoint/reference scheduling or for including measurable and predictable disturbances and for shaping the constraints. An illustrative example for the consideration of OB in MPC is presented in Fig. 16. In terms of disturbances, the OB should be considered as a cause of thermal gains from people, thermal losses from appliances or ventilation gain, and losses during window operations. This includes information about the number of occupants, time of use, and possibly information about used equipment. As measuring a direct heat dissipation is difficult, the forecasted behavior has to be used to infer the information about internal heat gains. Additionally, the OB could be considered by the constraints, such as by setting different upper and lower indoor air temperature bounds during occupancy hours. In case a specific setpoint instead of bounds is desired, the setpoint can also be explicitly defined in the cost function.

As reviewed in subsection 4.2.3, the indirect metrics evaluate the control outputs comfort and energy consumption. For the integration of OB models into MPC, we focus on the most relevant and most common use case in literature, HVAC control. The relevance results from the significant energy savings potential. Additionally, most OB models could be meaningfully coupled with HVAC control (such as Appliance Use, Lighting Operation, Occupancy Estimation and Prediction, Shading Operation, Thermostat Adjustment, Window Operation models). Most relevantly, the thermostat adjustment and attendance profiles shape the occupants' demand for thermal satisfaction by HVAC. Knowledge about absences and reduced thermal demands can significantly reduce the energy demand. The other OB models, shading and windows operation, have an impact on the thermal energy balance.

There are several requirements that need to be documented with OB

Table 5

The locations of sensors deployed from the reviewed s	tudies.
---	---------

Level of Locations	Locations	Sensor Type
Space Level	Ceiling	 CO2 Sensor
		 CO Sensor
		 Light Sensor
	Chair	 Chair Sensor
	Desk	 Keyboard & Mouse
		 Telephone (State)
	Door Frame	 Door State Sensor
	Wall	 Air Pressure Sensor
		 Air Speed Sensor
		 Bluetooth Beacon
		 GPS Location
		 LED Sensor
		 Light Switch
		 Motion (Unspecified)
		 Smart Plug
		 Sound Sensor (Acoustic)
		 Sound Sensor (Echo-Based)
		 Sound Recording Device
		 PIR Sensor
		 RF Sensor
		 RH Sensor
		 Temperature Sensor (Indoor)
		 VOC Sensor
		 WiFi Connection/Probe
	Window Frame	 Window State Sensor
Building Level	Electrical Panel	 Current/Power Sensor
		 Smart Meter
	HVAC Equipment	 AC State Sensor
		 Airflow Rate Sensor
	Main Entrance	Camera
	Rooftop	 Solar Irradiance Sensor

model to ensure the use in advanced optimal control methods such as MPC. Firstly, the models must provide a forecast of the occupancy behavior over the length of the prediction horizon of the optimization problem, which is typically between 1 and 24 h. The quality of the presence and OB forecasts also depends on the type of measurement sensors used to gather the occupancy data [63]. Most accurate predictions are obtained from occupancy dedicated-sensor data such as PIR and cameras. However, as pointed out in Ref. [64], other sensors such as CO₂ [65] or plug power can also provide sufficiently accurate data for control-oriented occupancy models.

7. Discussion and future challenges

Based on the previous review, to define a standard guideline to document occupant behavior for building controls poses the following challenges:

7.1. Model description and formal representation

The current model description varies among different schemas and formalization methods in terms of naming schema, description structures, and presentation of OB models. In addition, neither the OB model nor the building control model lacks a standard representation for model inputs, outputs, and model description. Hence, it creates a gap between OB and building control models. This results in a customized working process for every OB-driven building control study in the literature. In addition, such a process is not consistent and creates a very different performance (e.g. energy savings) even using the same type of OB model. As a matter of fact, the various inputs for the same OB model reflect this inconsistency. Prior researchers were using different sensors and instruments to develop different mathematical models to model and simulate the same behavior over decades. There is a need to standardize the model description and representation based on one formal language. Recently, a review paper [8] on data tools for building information and performance also concludes that ontologies or schemas represent the need to be developed. An effort to extend the current Brick schema to represent the OB model is ongoing.

7.2. Model development

Currently, the model development is not fully described in the scientific articles. Information is missing on preprocessing procedures and model selection limiting the reproducibility and even replicability of the results of the studies. To overcome these challenges in model development documentation, the authors should clearly state the model purpose, and the practical and theoretical arguments supporting the choice of a given modeling technique. Also, to foster transparency and clarity, they should explicitly document the adopted cleaning and imputing procedures for the missing and corrupt data, the outliers treatment chosen, the data dimensional reduction process implemented, the data scaling method used, the techniques used for feature creation, the approach adopted for partitioning the original dataset for the definition of the test and training subsets, and the anonymization techniques used, if any. Furthermore, an important challenge is to develop newer multidomain models, which can consider the multi-exposure of occupants to indoor environmental conditions and a multitude of controlling opportunities for a better and tailored adjustment of the building devices.

7.3. Model evaluation

The standardized and comprehensive model evaluation is crucial for OB model deployment in real-world scenarios of building control products. Therefore, a standardized evaluation schema is proposed for each distinct form of OB. With the ambition to provide a comprehensive evaluation, this work proposes joint use of three sets of evaluation metrics, namely absolute metrics, domain metrics, and indirect metrics. The absolute metrics were derived based on the vast literature evidence on the evaluation of general data-driven methods and their existing applications for OB modeling. Furthermore, we argue that the OB model evaluation has to include specific domain metrics. These are based on domain expertise in OB modeling and are supported by the existing literature evidence.



Fig. 16. An overview of the MPC structure with the proposed inclusions of the OB.

As human behavior is highly diverse and sensitive to unpredictable events, sensitivity analysis could be used as a supportive tool to assess the uncertainties related to the used OB model. Sensitivity analysis (SA) is a statistical technique that assesses the effects that changes in input or design variables have on the model output variables [16,66]. There are two approaches to SA that could be applied when a model can be also used for building control purposes. The simpler is the local sensitivity analysis, where the impact of an input variable's variation on a model response is estimated while keeping the values of the other input factors constant. The global sensitivity analysis, on the opposite, tests simultaneously all the input variables and enables assessing the impact of both individual input variables and their interactions on the model output. However, the existing applications of the sensitivity analysis for OB in building control are sparse, further adoption would provide useful insights on testing control algorithm robustness against noise or uncertainty in input variables and parameters.

7.4. Benefits of the inclusion of OB models in building control

Finally, the indirect metrics quantify how well the model contributes to fulfilling the higher goal of building control, such as maintaining comfort or optimizing energy consumption. Up to date, the impact of the OB model on the end-system has been limitedly explored. Namely, most of the existing OB models were not tested in field studies, and therefore, the relationship between whole system performance and the OB model is rarely analyzed. In order to come one step closer to filling this gap, this work proposed a set of indirect metrics for evaluating the impact of OB models for inclusion in HVAC control. The future challenge includes assessing the suitability of proposed metrics in field studies. Furthermore, the indirect metrics for alternative systems, such as shadings should be explored.

7.5. Model implementation

The documented model implementation should include the information about the used computational environment in which the OB model is tested, the experimental setting, and the recommendations for the intended application in the building control. Here, a particular challenge is that the buildings are commonly a one-time product. As there is limited literature evidence on documenting the model implementation, future research should focus on how to standardize information related to implementation in different buildings or HVAC systems.

Furthermore, the future model documentation should include the estimated OB model inference time. As highlighted in Ref. [67], state-of-the-art OB models are too computationally expensive to be included in real-time control applications, such as MPC. In order to obtain stable and reliable results, the computation of the next control signal should be indeed completed before the start of the actual period of observation. Based on the previous literature review, the time related to a single forward pass of the proposed data-driven models, i.e. inference time, is however rarely documented.

7.6. Model integration into advanced building control

One of the challenges of leveraging occupancy estimation and prediction models is their integration into advanced building control algorithms, e.g. MPC. These controllers are typically based on HVAC and building envelope thermal dynamics models and consider future system dynamics and future control inputs or constraints. Special care needs to be given to properly couple the occupancy estimation and prediction models into these dynamic equations. The OB models can serve as additional control variables (setpoint, constraint, or disturbances) for the building dynamics equations. These dynamic equations are usually represented by a set of first-order differential equations. As a common practice in the control engineering field, these equations can be Building and Environment 219 (2022) 109195

reformulated into a state space model and into discrete time [68,69].

8. Conclusion

In this paper, we evaluated current documentation of OB models for advanced building controls from four different perspectives: model description and formal representation, model development and evaluation, inclusion of OB model into building controls, and modeling implementation. During the literature review, we found that the building control and OB modeling were mostly researched distinctly. Most of the OB models were developed as stand-alone models. In that context, there is only a spoonful of publications that proposed a formal integration of the OB models into building control (e.g. Refs. [70,71]). Based on a comprehensive review and analysis on current documentation of OB model for advanced building controls, it can be concluded that: 1) There is no standard representation of various OB models; 2) no unified guidelines of OB model development; 3) a standardized evaluation schema is proposed for each distinct form of OB models; 4) a set of indirect metrics for evaluating the impact of OB models for the inclusion in HVAC control is defined; and 5) a systematic documentation of indented model implementation is proposed; and 6) OB models can be integrated into MPC for HVAC as predicted setpoints, constraints, or disturbances.

Given the current review and discussions, this paper also provides following future research opportunities: a) A formal representation of OB models based on the same schema and semantics. While there is an on-going effort in the Brick schema [25], such presentation can be further enriched with more common data sets [72]; b) open sourcing a library with OB model documentation that follows this guideline, c) deployment of existing OB models in building control studies.

Limitation of the study: 1) In this paper, the "occupant" is referred to as office workers in general. The review does not cover other occupant types such as elderly, who has different interactions in response to thermal stimuli; 2) The review study found very limited or no papers considering how to integrate sensor drifting into controls. Although it is an important issue for the control implementation, the paper focuses mainly on documenting occupant behavior. Future studies could further explore this topic; and 3) The guideline paper does not cover occupant behavior of personalized cooling and heating systems. This could be included in the future studies.

CRediT authorship contribution statement

Bing Dong: Writing - review & editing, Writing - original draft, Project administration, Methodology, Investigation, Funding acquisition, Conceptualization. Romana Markovic: Conceptualization, Investigation, Methodology, Writing - original draft, Writing - review & editing. Salvatore Carlucci: Writing - review & editing, Writing original draft, Resources, Methodology, Investigation, Conceptualization. Yapan Liu: Investigation, Methodology, Visualization, Writing original draft, Writing - review & editing, Conceptualization. Andreas Wagner: Writing - original draft, Resources, Methodology, Investigation, Conceptualization. Antonio Liguori: Conceptualization, Investigation, Methodology, Writing - original draft. Christoph van Treeck: Writing - original draft, Resources, Investigation, Conceptualization. Dmitry Oleynikov: Conceptualization, Investigation, Methodology, Writing - original draft. Elie Azar: Writing - original draft, Visualization, Methodology, Investigation, Conceptualization. Gianmarco Fajilla: Conceptualization, Investigation, Methodology, Writing - original draft. Ján Drgoňa: Writing - original draft, Visualization, Meth-Investigation, Conceptualization. Joyce odology, Kim: Conceptualization, Investigation, Methodology, Visualization, Writing original draft, Writing - review & editing. Marika Vellei: Writing original draft, Visualization, Methodology, Investigation, Conceptualization. Marilena De Simone: Visualization, Writing - original draft, Writing - review & editing, Conceptualization, Investigation, Methodology. Masood Shamsaiee: Writing - original draft, Methodology,

Investigation, Conceptualization. Mateus Bavaresco: Conceptualization, Investigation, Methodology, Writing – original draft. Matteo Favero: Writing – original draft, Methodology, Investigation, Conceptualization. Mikkel Kjaergaard: Conceptualization, Investigation, Methodology, Writing – original draft. Mohamed Osman: Writing – original draft, Methodology, Investigation, Conceptualization. Moritz Frahm: Conceptualization, Investigation, Methodology, Visualization, Writing – original draft. Sanam Dabirian: Writing – review & editing, Writing – original draft, Visualization, Methodology, Investigation, Conceptualization. Da Yan: Conceptualization, Investigation, Methodology, Writing – original draft. Xuyuan Kang: Writing – original draft, Methodology, Investigation, Conceptualization.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgment

Bing Dong and Yapan Liu would like to thank the support from the U. S. National Science Foundation (Award No. 1949372). Romana Markovic and Andreas Wagner gratefully acknowledge the financial support by the German Federal Ministry for Economic Affairs and Climate Action (BMWK), promotional reference 03EN1002A. Dmitry Oleynikov gratefully acknowledges the financial support by the German Federal Ministry for Economic Affairs and Climate Action (BMWK), promotional reference 03EN1002B. Moritz Frahm gratefully acknowledges the financial support from the German Federal Ministry for Economic Affairs and Climate Action (BMWK), promotional reference 03ETW016A. Gianmarco Fajilla would like to thank the Calabria Region Government for his Ph.D. scholarship (POR Calabria FSE/FESR 2014- 2020, Identification Project Number H21G18000170006). Antonio Liguori and Christoph van Treeck were funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) - TR 892/4-1. This paper benefited greatly from discussions with members of IEA EBC Annex 79.

Appendix

Appendix I. Summary of the indirect metrics for each OB modeling objective

Model	Comfort-related metrics	Energy-related metrics
Occupancy Estimation and Prediction	Thermal comfort and indoor air quality: hours of setpoints not met	energy consumption/saving, start/stop time, in most cases related to HVAC
Thermostat Adjustment	Thermal comfort: indoor temperature, PMV, PPD	energy consumption/saving, monetary savings, related to HVAC, duration of unnecessary heating [h], peak load change (energy shifting for DR), energy use during peak, setpoint reduction, HVAC coefficient of performance
Window Operation	-	energy consumption/saving, related to HVAC
Shading Operation	Thermal comfort: air temperature, PPD, overheated hours; Visual comfort: illuminance	energy consumption/saving, related to HVAC and lighting, optimal dimming
Lighting Operation	Visual confort: illuminance, false off frequency, discomfort probability	energy consumption/saving, optimal dimming (% of lighting power used needed on daylight availability), peak power, illuminance reduction in unoccupied workstations
Appliance Use	-	energy consumption/saving, related to HVAC and the appliances

References

- W. O'Brien, et al., Introducing IEA EBC annex 79: key challenges and opportunities in the field of occupant-centric building design and operation, Build. Environ. 178 (Jul. 2020), 106738, https://doi.org/10.1016/j.builden.2020.106738.
- [2] M. Kong, B. Dong, R. Zhang, Z. O'Neill, HVAC energy savings, thermal comfort and air quality for occupant-centric control through a side-by-side experimental study, Appl. Energy 306 (Jan. 2022), 117987, https://doi.org/10.1016/j. apenergy.2021.117987.
- [3] Z. Pang, Y. Chen, J. Zhang, Z. O'Neill, H. Cheng, B. Dong, Nationwide HVAC energy-saving potential quantification for office buildings with occupant-centric controls in various climates, Appl. Energy 279 (Dec. 2020), 115727, https://doi. org/10.1016/j.apenergy.2020.115727.
- [4] D. Li, C.C. Menassa, V.R. Kamat, Personalized human comfort in indoor building environments under diverse conditioning modes, Build. Environ. 126 (Dec. 2017) 304–317, https://doi.org/10.1016/j.buildenv.2017.10.004.
- [5] J. Xie, H. Li, C. Li, J. Zhang, M. Luo, Review on occupant-centric thermal comfort sensing, predicting, and controlling, Energy Build. 226 (Nov. 2020), 110392, https://doi.org/10.1016/j.enbuild.2020.110392.
- [6] J.Y. Park, et al., A critical review of field implementations of occupant-centric building controls, Build. Environ. 165 (Nov. 2019), 106351, https://doi.org/ 10.1016/j.buildenv.2019.106351.
- [7] B. Dong, D. Yan, Z. Li, Y. Jin, X. Feng, H. Fontenot, Modeling occupancy and behavior for better building design and operation—a critical review, Build. Simulat. 11 (5) (Oct. 2018) 899–921, https://doi.org/10.1007/s12273-018-0452-
- [8] N. Luo, M. Pritoni, T. Hong, An overview of data tools for representing and managing building information and performance data, Renew. Sustain. Energy Rev. 147 (Sep. 2021), 111224, https://doi.org/10.1016/j.rser.2021.111224.
- [9] S. Carlucci, et al., Modeling Occupant Behavior in Buildings, 174, Building and Environment, May 2020, 106768, https://doi.org/10.1016/j. buildenv.2020.106768.

- [10] A. Wagner, W. O'Brien, B. Dong (Eds.), Exploring Occupant Behavior in Buildings, Springer International Publishing, Cham, 2018, https://doi.org/10.1007/978-3-319-61664-9.
- [11] Y. Cardinale, Occupant Activity Detection in Smart Buildings: A Review," Social Science Research Network, Jun. 2020, https://doi.org/10.2139/ssrn.3671533. Rochester, NY, SSRN Scholarly Paper ID 3671533.
- [12] S. Norouziasi, A. Jafari, Y. Zhu, Modeling and simulation of energy-related humanbuilding interaction: a systematic review, J. Build. Eng. 44 (Dec. 2021), 102928, https://doi.org/10.1016/j.jobe.2021.102928.
- [13] S. D'Oca, T. Hong, J. Langevin, The human dimensions of energy use in buildings: a review, Renew. Sustain. Energy Rev. 81 (Jan. 2018) 731–742, https://doi.org/ 10.1016/j.issre.2017.08.019.
- [14] M. Han, J. Zhao, X. Zhang, J. Shen, Y. Li, The reinforcement learning method for occupant behavior in building control: a review, Energy. Built. Env. 2 (2) (Apr. 2021) 137–148, https://doi.org/10.1016/j.enbenv.2020.08.005.
- [15] E.T. Maddalena, Y. Lian, C.N. Jones, Data-driven methods for building control a review and promising future directions, Control Eng. Pract. 95 (Feb. 2020), 104211, https://doi.org/10.1016/j.conengprac.2019.104211.
- [16] A. Mirakhorli, B. Dong, Occupancy behavior based model predictive control for building indoor climate—a critical review, Energy Build. 129 (Oct. 2016) 499–513, https://doi.org/10.1016/j.enbuild.2016.07.036.
- [17] S. Naylor, M. Gillott, T. Lau, A review of occupant-centric building control strategies to reduce building energy use, Renew. Sustain. Energy Rev. 96 (Nov. 2018) 1-10, https://doi.org/10.1016/j.rser.2018.07.019.
- D. Yan, et al., Occupant behavior modeling for building performance simulation: current state and future challenges, Energy Build. 107 (Nov. 2015) 264–278, https://doi.org/10.1016/j.enbuild.2015.08.032.
 G. Happle, J.A. Fonseca, A. Schlueter, A review on occupant behavior in urban
- [19] G. Happle, J.A. Fonseca, A. Schlueter, A review on occupant behavior in urban building energy models, Energy Build. 174 (Sep. 2018) 276–292, https://doi.org/ 10.1016/j.enbuild.2018.06.030.
- [20] T. Hong, J. Langevin, K. Sun, Building simulation: ten challenges, Build. Simulat. 11 (5) (Oct. 2018) 871–898, https://doi.org/10.1007/s12273-018-0444-x.

- [21] E. Azar, et al., Simulation-aided occupant-centric building design: a critical review of tools, methods, and applications, Energy Build. 224 (Oct. 2020), 110292, https://doi.org/10.1016/j.enbuild.2020.110292.
- [22] Y. Zhang, X. Bai, F.P. Mills, J.C.V. Pezzey, Rethinking the role of occupant behavior in building energy performance: a review, Energy Build. 172 (Aug. 2018) 279–294, https://doi.org/10.1016/j.enbuild.2018.05.017.
- [23] M.B. Kjærgaard, et al., Current practices and infrastructure for open data based research on occupant-centric design and operation of buildings, Build. Environ. 177 (Jun. 2020), 106848, https://doi.org/10.1016/j.buildenv.2020.106848.
- [24] ARM research facility." https://www.arm.gov/(accessed Mar. 18, 2022).
 [25] B. Balaji, et al., Brick: towards a unified metadata schema for buildings, in: Proceedings of the 3rd ACM International Conference on Systems for Energy-Efficient Built Environments, Nov. 2016, pp. 41–50, https://doi.org/10.1145/ 2993422.2993577. Palo Alto CA USA.
- [26] Home Project Haystack." https://project-haystack.org/(accessed Mar. 18, 2022).
 [27] J. Kim, S. Schiavon, G. Brager, Personal comfort models a new paradigm in thermal comfort for occupant-centric environmental control, Build. Environ. 132
- (Mar. 2018) 114–124, https://doi.org/10.1016/j.buildenv.2018.01.023.
 [28] M. Ashouri, B.C.M. Fung, F. Haghighat, H. Yoshino, Systematic approach to
- provide building occupants with feedback to reduce energy consumption, Energy 194 (Mar. 2020), 116813, https://doi.org/10.1016/j.energy.2019.116813.
 [29] X. Yu, S. Ergan, G. Dedemen, A data-driven approach to extract operational
- signatures of HVAC systems and analyze impact on electricity consumption, Appl. Energy 253 (Nov. 2019), 113497, https://doi.org/10.1016/j. apenergy.2019.113497.
- [30] S. D'Oca, T. Hong, A data-mining approach to discover patterns of window opening and closing behavior in offices, Build. Environ. 82 (Dec. 2014) 726–739, https:// doi.org/10.1016/j.buildenv.2014.10.021.
- [31] S. D'Oca, T. Hong, Occupancy schedules learning process through a data mining framework, Energy Build. 88 (Feb. 2015) 395–408, https://doi.org/10.1016/j. enbuild.2014.11.065.
- [32] O. Van Cutsem, G. Lilis, M. Kayal, Automatic multi-state load profile identification with application to energy disaggregation, in: 2017 22nd IEEE International Conference on Emerging Technologies and Factory Automation (ETFA), Sep. 2017, pp. 1–8, https://doi.org/10.1109/ETFA.2017.8247684.
- [33] B. Dong, K.P. Lam, A real-time model predictive control for building heating and cooling systems based on the occupancy behavior pattern detection and local weather forecasting, Build. Simulat. 7 (1) (Feb. 2014) 89–106, https://doi.org/ 10.1007/s12273-013-0142-7.
- [34] Y. Jin, D. Yan, X. Zhang, J. An, M. Han, A data-driven model predictive control for lighting system based on historical occupancy in an office building: Methodology development, Build. Simulat. 14 (1) (Feb. 2021) 219–235, https://doi.org/ 10.1007/s12273-020-0638-x.
- [35] D. Mora, G. Fajilla, M.C. Austin, M. De Simone, Occupancy patterns obtained by heuristic approaches: cluster analysis and logical flowcharts. A case study in a university office, Energy Build. 186 (Mar. 2019) 147–168, https://doi.org/ 10.1016/j.enbuild.2019.01.023.
- [36] S. Dabirian, K. Panchabikesan, U. Eicker, Occupant-centric urban building energy modeling: approaches, inputs, and data sources - a review, Energy Build. 257 (Feb. 2022), 111809, https://doi.org/10.1016/j.enbuild.2021.111809.
- [37] C. Wang, D. Yan, Y. Jiang, A novel approach for building occupancy simulation, Build. Simulat. 4 (2) (Jun. 2011) 149–167, https://doi.org/10.1007/s12273-011-0044-5.
- [38] M.B. Kjærgaard, M. Werner, F.C. Sangogboye, K. Arendt, DCount a probabilistic algorithm for accurately disaggregating building occupant counts into room counts, in: 2018 19th IEEE International Conference on Mobile Data Management (MDM), Jun. 2018, pp. 46–55, https://doi.org/10.1109/MDM.2018.00021.
- [39] R. Markovic, E. Azar, M.K. Annaqeeb, J. Frisch, C. van Treeck, Day-ahead prediction of plug-in loads using a long short-term memory neural network, Energy Build. 234 (Mar. 2021), 110667, https://doi.org/10.1016/j.enbuild.2020.110667.
- [40] M. Shcherbakov, A. Brebels, N.L. Shcherbakova, A. Tyukov, T.A. Janovsky, V. A. Kamaev, A survey of forecast error measures, World Appl. Sci. J. 24 (Jan. 2013) 171–176, https://doi.org/10.5829/idois.wasi.2013.24.timies.80032.
- [41] M.Z. Naser, A.H. Alavi, Error metrics and performance fitness indicators for artificial intelligence and machine learning in engineering and sciences, Archit. Struct. Constr., Nov. (2021), https://doi.org/10.1007/s44150-021-00015-8.
- [42] Z. Li, H. Zhu, B. Dong, X. Xu, Development of a systematic procedure to establish customized shading behavior identification model, Energy Build. 239 (May 2021), 110793, https://doi.org/10.1016/j.enbuild.2021.110793.
- [43] W. Kampel, S. Carlucci, B. Aas, A. Bruland, A proposal of energy performance indicators for a reliable benchmark of swimming facilities, Energy Build. 129 (Oct. 2016) 186–198, https://doi.org/10.1016/j.embuild.2016.07.033.
- [44] W. O'Brien, I. Gaetani, S. Carlucci, P.-J. Hoes, J.L.M. Hensen, On occupant-centric building performance metrics, Build. Environ. 122 (Sep. 2017) 373–385, https:// doi.org/10.1016/j.buildenv.2017.06.028.
- [45] F. Tahmasebi, A. Mahdavi, The sensitivity of building performance simulation results to the choice of occupants' presence models: a case study, J. Build. Perform. Simul. 10 (5–6) (Nov. 2017) 625–635, https://doi.org/10.1080/ 19401493.2015.1117528.
- [46] A. Chong, G. Augenbroe, D. Yan, Occupancy data at different spatial resolutions: building energy performance and model calibration, Appl. Energy 286 (Mar. 2021), 116492, https://doi.org/10.1016/j.apenergy.2021.116492.
- [47] A. Mahdavi, F. Tahmasebi, On the quality evaluation of behavioural models for building performance applications, J. Build. Perform. Simul. 10 (5–6) (Nov. 2017) 554–564, https://doi.org/10.1080/19401493.2016.1230148.

- [48] D. Yan, X. Feng, Y. Jin, C. Wang, The evaluation of stochastic occupant behavior models from an application-oriented perspective: using the lighting behavior model as a case study, Energy Build. 176 (Oct. 2018) 151–162, https://doi.org/ 10.1016/j.enbuild.2018.07.037.
- [49] A. Mahdavi, F. Tahmasebi, Predicting people's presence in buildings: an empirically based model performance analysis, Energy Build. 86 (Jan. 2015) 349–355, https://doi.org/10.1016/j.enbuild.2014.10.027.
- [50] Y. Jin, et al., Appliance use behavior modelling and evaluation in residential buildings: a case study of television energy use, Build. Simulat. 13 (4) (Aug. 2020) 787–801, https://doi.org/10.1007/s12273-020-0648-8.
- [51] C. Peng, D. Yan, R. Wu, C. Wang, X. Zhou, Y. Jiang, Quantitative description and simulation of human behavior in residential buildings, Build. Simulat. 5 (2) (Jun. 2012) 85–94, https://doi.org/10.1007/s12273-011-004-0.
- [52] I.T. Michailidis, et al., Energy-efficient HVAC management using cooperative, selftrained, control agents: a real-life German building case study, Appl. Energy 211 (Feb. 2018) 113–125, https://doi.org/10.1016/j.apenergy.2017.11.046.
- [53] K.X. Perez, M. Baldea, T.F. Edgar, Integrated smart appliance scheduling and HVAC control for peak residential load management, in: 2016 American Control Conference (ACC), Jul. 2016, pp. 1458–1463, https://doi.org/10.1109/ ACC.2016.7525122.
- [54] T. Sanjeev Kumar, C. Kurian, S. Shetty, A data-driven approach for the control of a daylight-artificial light integrated scheme, Light. Res. Technol. 52 (2) (Apr. 2020) 292–313, https://doi.org/10.1177/1477153519841104.
- [55] Z. Wang, Y.K. Tan, Illumination control of LED systems based on neural network model and energy optimization algorithm, Energy Build. 62 (Jul. 2013) 514–521, https://doi.org/10.1016/j.enbuild.2013.03.029.
- [56] ANSI/ASHRAE Standard 135-2016," p. 48.
- [57] M. Aleksy, P. Bauer, Ensuring the practical applicability of algorithms for user behavior modeling through integration into building automation systems, in: Advanced Information Networking and Applications, 2020, pp. 715–721, https:// doi.org/10.1007/978-3-030-44041-1_63. Cham.
- [58] R. Markovic, Generic Occupant Behavior Modeling for Commercial Buildings, 2020.
- [59] X. Zhang, et al., IEHouse: a non-intrusive household appliance state recognition system, in: 2017 IEEE SmartWorld, Ubiquitous Intelligence Computing, Advanced Trusted Computed, Scalable Computing Communications, Cloud Big Data Computing, Internet of People and Smart City Innovation (SmartWorld/ SCALCOM/UIC/ATC/CBDCom/IOP/SCI), Aug. 2017, pp. 1–8, https://doi.org/ 10.1109/UIC-ATC.2017.8397510.
- [60] D. Egarter, V.P. Bhuvana, W. Elmenreich, PALDi: online load disaggregation via particle filtering, IEEE Trans. Instrum. Meas. 64 (2) (Feb. 2015) 467-477, https:// doi.org/10.1109/TIM.2014.2344373.
- [61] W. Kuo, M.J. Zuo, Optimal Reliability Modeling: Principles and Applications, John Wiley & Sons, 2003.
- [62] P.E. Black, 'big-O notation', in dictionary of algorithms and data structures. htt ps://xlinux.nist.gov/dads/HTML/bigOnotation.html, Sep. 2019 accessed Mar. 24, 2022.
- [63] A. Capozzoli, M.S. Piscitelli, A. Gorrino, I. Ballarini, V. Corrado, Data analytics for occupancy pattern learning to reduce the energy consumption of HVAC systems in office buildings, Sustain. Cities Soc. 35 (Nov. 2017) 191–208, https://doi.org/ 10.1016/j.scs.2017.07.016.
- [64] F.C. Sangogboye, K. Arendt, A. Singh, C.T. Veje, M.B. Kjærgaard, B.N. Jørgensen, Performance comparison of occupancy count estimation and prediction with common versus dedicated sensors for building model predictive control, Build. Simulat. 10 (6) (Dec. 2017) 829–843, https://doi.org/10.1007/s12273-017-0397-5
- [65] F. Jorissen, W. Boydens, L. Helsen, Simulation-based occupancy estimation in office buildings using CO2 sensors, in: Proceedings of the 15th IBPSA Conference, 2017, pp. 1073–1082. Accessed: Mar. 24, 2022. [Online]. Available: http://hdl. handle.net/1854/LU-8580815.
- [66] S. Goyal, H.A. Ingley, P. Barooah, Effect of various uncertainties on the performance of occupancy-based optimal control of HVAC zones, in: 2012 IEEE 51st IEEE Conference on Decision and Control (CDC), Dec. 2012, pp. 7565–7570, https://doi.org/10.1109/CDC.2012.6426111.
- [67] J. Drgoña, et al., All you need to know about model predictive control for buildings, Annu. Rev. Control 50 (Jan. 2020) 190–232, https://doi.org/10.1016/j. arcontrol.2020.09.001.
- [68] I. Hazyuk, C. Ghiaus, D. Penhouet, Optimal temperature control of intermittently heated buildings using Model Predictive Control: Part I – building modeling, Build. Environ. 51 (May 2012) 379–387, https://doi.org/10.1016/j. buildenv.2011.11.009.
- [69] I. Hazyuk, C. Ghiaus, D. Penhouet, Optimal temperature control of intermittently heated buildings using Model Predictive Control: Part II – control algorithm, Build. Environ. 51 (May 2012) 388–394, https://doi.org/10.1016/j. huildew.2011.10.09
- [70] W. Turner, T. Hong, A Technical Framework to Describe Occupant Behavior for Building Energy Simulations, LBNL-6671E, Lawrence Berkeley National Lab.

(LBNL), Berkeley, CA (United States), Dec. 2013. Accessed: Mar. 24, 2022.
[Online]. Available: https://www.osti.gov/biblio/1164208.
[71] B. Dong, K. Lam, C. Neuman, Integrated building control based on occupant behavior pattern detection and local weather forecasting, in: In Twelfth International IBPSA Conference, Nov. 2011, pp. 14–17. Accessed: Mar. 24, 2022.
[Online]. Available: https://www.semanticscholar.org/paper/Integrated-build

ing-control-based-on-occupant-and-Dong-Lam/d9c366355f9fef0678e47f448fc 50014ad77814b.

[72] G. Fierro, et al., Formalizing tag-based metadata with the Brick ontology, Frontiers in Built Environment 6 (2020). Accessed: Mar. 28, 2022. [Online]. Available: htt ps://www.frontiersin.org/article/10.3389/fbuil.2020.558034.



ISBN 978-82-326-6325-5 (printed ver.) ISBN 978-82-326-6909-7 (electronic ver.) ISSN 1503-8181 (printed ver.) ISSN 2703-8084 (online ver.)

